NASA CONTRACTOR REPORT

NASA CR-111919



N71-32690

CASE FILE COPY

MASTER AGREEMENT TASK ORDER FIVE

Analyses and Limited Evaluation of Payload and Legged Landing System Structures For The Survivable Soft Landing of Instrument Payloads

By O. R. Otto, R. M. Laurenson, R. A. Melliere, and R. L. Moore

Prepared by MCDONNELL DOUGLAS ASTRONAUTICS COMPANY – EAST St. Louis, Missouri 63166 (314) 232-0232

for Langley Research Center

NATIONAL AERONAUTICS AND SPACE ADMINISTRATION • WASHINGTON, D.C. • JULY 1971

NASA CR-111919

MASTER AGREEMENT TASK ORDER FIVE

Analyses and Limited Evaluation of Payload and Legged Landing System Structures For

The Survivable Soft Landing of Instrument Payloads

By O. R. Otto, R. M. Laurenson, R. A. Melliere, and R. L. Moore

Distribution of this report is provided in the interest of information exchange. Responsibility for the contents resides in the author or organization that prepared it.

Prepared Under Contract No. NAS 1–8137 by

MCDONNELL DOUGLAS ASTRONAUTICS COMPANY – EAST

Saint Louis, Missouri

for

NATIONAL AERONAUTICS AND SPACE ADMINISTRATION

THIS PAGE INTENTIONALLY LEFT BLANK

ABSTRACT

This report describes two computer programs developed for the investigation of legged planetary landers. Analytical methods incorporated in the programs, operating instructions, and examples of program utilization are described. One program developed is the Structural Analysis Program, containing a Center Body Option, which can be used to analyze a structure by utilizing a finite element idealization. It also contains a Landing Gear Option for determination of the energy absorption and load-stroke characteristics of either a cantilever or inverted tripod landing gear configuration. The second program is the Landing Loads and Motions Program, used to predict spatial landing dynamics of a legged lander containing up to five landing gears. This program also contains options for including the effects of a flexible center body and determining lander stability.

Several exemplary computer runs are discussed to aid in interpretation of the operating instructions and to illustrate various available program options. Analytical results for several of these cases are compared to test data obtained during model test programs conducted at NASA Langley Research Center. THIS PAGE INTENTIONALLY LEFT BLANK

TABLE OF CONTENTS

	 1 5 5 5 5 11
· · ·	 5 5 5 5 11
· · ·	 5 5 5 11
•••	• 5 • 5 • 11
• • •	• 5 • 11
•••	• 11
• •	• 12
• • •	• 17
, 	• 17
	• 19
	• 21
	• 21
	• 22
	• 45
• • •	• 75
	• 75
	• 78
• • •	• 78
, • • •	• 78
	• 95
	• 111
	. 111
• • •	. 111
· · · · · · · · · · · · · · · · · · ·	

TABLE OF CONTENTS (Continued)

	PAGE
	5.1.2 Equations of Motion
	5.1.3 Landing Gear Strut Idealization
	5.1.4 Soil Mechanics
	5.1.5 Footpad Attenuation System
	5.1.6 Footpad Sliding Motion
	5.1.7 Lander Stability
5.2	Program Description
	5.2.1 Subroutines
	5.2.2 Flow Diagram
5.3	Program Operation
	5.3.1 Input Data
	5.3.2 Output Data
	5.3.3 Example of Program Operation
DEMO	NSTRATION OF PROGRAM CAPABILITIES
6.1	Large Displacement Analysis of Landing Gears
	6.1.1 Inverted Tripod Gear
	6.1.2 Cantilever Gear
6.2	Analysis of Task Order Three Lander
	6.2.1 Modal Analysis of Center Body Structure
	6.2.2 Correlation with Third Drop Test
	6.2.3 Center Body Landing Loads
	6.2.4 Center Body Internal Load Distribution
6.3	Analysis of Modified Task Order Three Lander
6.4	Landing Analysis of Full-Mass Task Order Three Lander 196

6.

TABLE OF CONTENTS (Continued)

																														PAGE
7.	CONCLUSIONS	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	203
8.	REFERENCES	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	205

APPENDIX

A	Example of Input and Output, Center Body Option, Structural Analysis Program	20 7
В	Example of Input and Output, Modal Analysis Routine, Structural Analysis Program	219
С	Example of Input and Output, Landing Gear Option, Structural Analysis Program	241
D	Example of Input and Output, Landing Loads and Motions Program \cdot .	249
E	Operating Instructions, Center Body Landing Loads Program • • • • •	257
F	Numerical Integration Routine, Landings Loads and Motions Program	275
G	Center Body Internal Load Distribution for Task Order Three Lander Using Refined Idealization	281
H	Program Listing, Structural Analysis Program	285
I	Program Listing, Landing Loads and Motions Program	401

LIST OF FIGURES

FIGURE	TITLE	PAGE
1-1	Task Order Three Lander	2
1-2	Task Order Five Program Usage Diagram	3
2-1	Legged Lander	6
2-2	Legged Lander Components	7
2-3	Landing Gear Concepts	8
4-1	Displacements of General Bar Element in Local Coordinate System	23
4–2	Displacements of General Bar Element in Global Coordinate System	25
4-3	Force-Displacement Relations for General Bar Element	26
4-4	Stiffness Matrix for General Bar Element	28
4-5	Stiffness Factors Applied to Original Matrix to Account for Elemental Bar Shear Strain	29
4-6	Example Calculation of Shear Form Factor	30
4-7	Shear Web Replacement with Diagonals	31
4-8	Substitute Factors in K _S Matrix to Allow Elimination of Rotational Restraint in Z Direction at Point P	32
4-9	Substitute Factors in K _S Matrix to Allow Elimination of Rotational Restraint in Z Direction at Point Q	33
4-10	Substitute Factors in K _S Matrix Allowing Elimination of Rotational Restraint in Y Direction at Point P	34
4-11	Substitute Factors in K Matrix Allowing Elimination of Rotational Restraint in Y Direction at Point Q	35
4-12	Direction Cosines of Local Coordinate System Relative to Global Coordinate System	37
4-13	Assembled Stiffness Matrix Format	38

4-14	Large Displacements of a Typical Two-Dimensional Gear	46
4-15	Incremental Stiffness Method Applied to a Typical Nonlinear Load-Stroke Curve	48
4-16	Typical Incremental Step of Modified Incremental Stiffness Method	49
4-17	Landing Gear Idealizations	52
4-18	Typical Strut Load-Stroke Curve	55
4-19	Strut Displacements in Local Coordinate System	57
4-20	Landing Gear Coordinate Systems	60
4-21	Strut Displacements in Surface Coordinate System	62
4-22	Small Displacement Stiffness Matrix of An Axial Strut	63
4-23	Geometric Stiffness Matrix of an Axial Strut	64
4-24	Small Displacement Stiffness Matrix of a Bending Strut .	66
4-25	Geometric Stiffness Matrix of a Bending Strut	67
4-26	Internal Bending Loads in Main Strut of Cantilever Gear .	69
4-27	Assembled Incremental Stiffness Matrix Format	71
4-28	Assumed Load-Stroke Curve for Footpad Honeycomb	74
4-29	Example Calculation of Strut Crush Energy	76
4-30	Structural Analysis Program Subroutines	77
4-31	Flow Diagram Structural Analysis Program	79
4-32	Input Data Format - Center Body Option	82
4-33	Input Data Format - Center Body Option	84
4-34	Input Data Format - Center Body Option	87
4-35	Namelist (\$INDATA) Data Option Indicators	89
436	Input Data Format - Landing Gear Option	97

4-37	Input Data Format - Landing Gear Option	100
4-38	Input Data Format - Landing Gear Option	103
4-39	Input Data Format - Landing Gear Option	106
4-40	Input Data Format - Landing Gear Option	109
5-1	Coordinate Systems	112
5-2	Position of Point on Center Body in Surface Coordinate System	118
5-3	Position of Point on Center Body in Lander Coordinate System	119
5-4	Position of Noncontacting Footpad in Surface Coordinate System	124
5-5	Deflection of Cantilever Gear Main Strut	130
5-6	Modified Drag Strut Load-Stroke Curve	131
5-7	Symbols for Soil Mechanics Routine	133
5-8	Footpad Representation	134
5-9	Soil Forces Primary Soil Mechanics Method	136
5-10	Properties of Soils	137
5-11	Empirical Soil Relationships	138
5-12	Normal Soil Force Secondary Soil Mechanics Method	139
5-13	Footpad Attenuation System	142
5-14	Evaluation of In-Plane Soil Force Magnitude	143
5-15	Lander Stability Angle	1.45
5-16	Landing Loads and Motions Program Subroutines and Subprograms	147
5-17	Flow Diagram Landing Loads and Motions Program	149
5-18	Input Data Format Landing Loads and Motions Program	153

5-19	Input Data Control Indicators Landing Loads and Motions Program	156
5-20	Input Data Landing Loads and Motions Program	157
6-1	Load-Stroke Curve for Main Strut of Inverted Tripod Gear - Task Order Three Lander	164
6-2	Drag Strut Load Correlation - Inverted Tripod Gear	165
6-3	Anticipated Drag Strut Loads for Inverted Tripod Gear	166
6-4	Load-Stroke Curve for Main Strut of One-Sixth Scale LM Cantilever Gear	167
6-5	Load-Stroke Curve for Drag Strut of One-Sixth Scale LM Cantilever Gear	169
6-6	Maximum Energy Absorption for One-Sixth Scale LM Cantilever Gear	170
6-7	Geometry of Task Order Three Lander	171
6-8	Center Body Structural Idealization	174
6-9	Center Body Mode Shape Associated with 34.6 ${\rm H}_{\rm Z}$ Mode	175
6-10	Center Body Mode Shape Associated with 48.2 $_{ m z}^{ m H}$ Mode	176
6-11	Center Body Mode Shape Associated with 101 H $_{ m z}$ Mode \ldots .	177
6-12	Initial Conditions for Drop Test Three	178
6-13	Footpad Idealization	179
6-14	3/8 Mass Task Order Three Lander Mass Breakdown	180
6-15	Center of Gravity Accelerations with Rigid Center Body .	181
6-16	Outside Drag Strut Load with Rigid Center Body	182
6-17	Center of Gravity Accelerations with Flexible Center Body	183
6-18	Main Strut Loads with Flexible Center Body	184
6-19	Main Strut Strokes with Flexible Center Body	185

6-20	Outside Drag Strut Load with Flexible Center Body	186
6-21	Inside Drag Strut Load with Flexible Center Body	187
6-22	Leg 1 Drag Strut Load with Flexible Center Body	188
6-23	Center Body Idealization Including Tension Straps	189
6-24	Loading Condition for Evaluation of Modified Task Order Three Lander Center Body Structure	192
6-25	Comparison of Various Tension Strap Configurations	193
6-26	Center of Gravity Accelerations with Modified Center Body Structure	194
6-27	Outside Drag Strut Load with Modified Center Body Structure	195
6-28	Comparison of Computer Results for Original and Modified Task Order Three Lander	197
6-29	Main Strut Time Histories for Various Landing Surface Soil Idealizations	199
6-30	Full-Mass Task Order Three Lander Center of Gravity Acceleration Time Histories	200
6-31	Full-Mass Task Order Three Lander Main Strut Load Time Histories	201
E-1	Input Data Format - Center Body Landing Loads Program	259
E2	Input Data - Center Body Landing Program	260
F-1	Integration Control Tolerance For Footpad Equations of Motion	279
G-1	Refined Center Body Structural Idealization	283
G - 2	Comparison of Computer Results for 33 and 72 Joint Idealizations of Task Order Three Lander Center Body	284

LIST OF PAGES

- i thru xiii
- 1 thru 479

THIS PAGE INTENTIONALLY LEFT BLANK

1. INTRODUCTION

This report describes two computer programs developed by McDonnell Douglas Astronautics Company - East under NASA Contract NAS-1-8137(U) for investigation of legged type planetary landers. They are the Structural Analysis Program and the Landing Loads and Motions Program. The Structural Analysis Program contains options for the determination of internal center body load distributions, center body modal information, and the large displacement behavior of either an inverted tripod or a cantilever gear configuration. The Landing Loads and Motions Program can be employed for the determination of the spatial landing motions of a lander idealized either as a rigid body or as a flexible body. In addition, options are available for the determination of overall landing loads and lander stability boundaries.

Results of analyses conducted on typical legged lander configurations are presented and compared with test results obtained at NASA Langley Research Center. Test results were obtained using the Task Order Three Lander(Reference (1)) shown in Figure 1-1. These analyses demonstrate the capabilities of the developed computer programs.

The programs developed during Task Order Five provide the capability of conducting a complete design study on a legged lander beginning at the preliminary design phase and continuing through a final evaluation of the lander's structural configuration. A flow diagram of this capability is presented in Figure 1-2. Program options permit detailed analyses to be conducted at any point during the evolution of a legged lander. For example, analytical investigations for boiler plate models, scale models, or final lander configurations can be conducted.

During the preliminary design phase, various landing gear configurations are evaluated with the Landing Gear Option of the Structural Analysis Program. At the same time the landing loads and stability of a lander idealized with a rigid center body are investigated with the Landing Loads and Motions Program. Results of these analyses are used to select a final landing gear configuration. In addition, the predicted rigid body landing loads are used in conjunction with the Center Body Option of the Structural Analysis Program to determine internal load distributions in the center body structure. Based on



Preliminary Design Phase



FIGURE 1-2 TASK ORDER FIVE PROGRAM USAGE DIAGRAM

the results of this preliminary design phase, the lander's structural arrangement is selected. At this time, the programs are used to determine center body flexibility information, landing loads, and stability conditions for the flexible lander, and internal load distributions for the final center body structural configuration.

2. COMPUTER PROGRAM CAPABILITIES

The two computer programs developed during Task Order Five contain many options which permit analyses of a wide variety of legged lander configurations. Legged lander configurations considered and the associated program capabilities and options are discussed in the following sections.

2.1 <u>Legged Lander Description</u> - The general arrangement of a typical legged lander configuration is shown in Figure 2-1. The legged lander is composed of three basic components: (1) center body, (2) payload and auxiliary equipment, and (3) landing gear. Items included in each component are shown in Figure 2-2.

The two landing gear configurations reflected in the analysis are shown in Figure 2-3. They are the inverted tripod gear and the cantilever gear. The main strut of both gears incorporates an energy absorption system. Drag struts may also include an energy absorption system or for the inverted tripod gear, they may simply be frame members stabilizing the main strut.

2.2 <u>Structural Analysis Program</u> - Two major options are available in the Structural Analysis Program. The first of these is the Center Body Option for the small displacement, finite element analysis of a legged lander center body structure. In formulating the Center Body Option, the Structural Analysis Program written for Task Order Two, Reference (2), was employed as a nucleus. During Task Order Five, many improvements and additions were made to this baseline program. In addition, a Landing Gear Option was developed for determining the large displacement stroking behavior of an individual landing gear. This is a major addition to the capabilities of the Task Order Two version of the Structural Analysis Program.

2.2.1 <u>Center Body Option</u> - The Center Body Option of the Structural Analysis Program is formulated for solution of redundant structures based on the small displacement finite element stiffness method and utilizing iterative methods of solution. The program can determine internal and external load distributions and deflection patterns for any system of forces or deflections impressed on a network of structural bar members and shear webs. Plastic



FIGURE 2-1 LEGGED LANDER

CENTER BODY LANDING GEAR CARRY-THROUGH STRUCTURE PAYLOAD SUPPORT STRUCTURE AUXILIARY EQUIPMENT SUPPORT STRUCTURE

PAYLOAD AND AUXILIARY EQUIPMENT FACSIMILE CAMERAS ATMOSPHERIC SENSORS SOIL ACQUISITION MECHANISM SCIENCE INSTRUMENTS COMMUNICATIONS EQUIPMENT ELECTRONIC EQUIPMENT POWER SUPPLY TERMINAL PROPULSION SYSTEM THERMAL CONTROL SYSTEM

LANDING GEAR FOOTPAD MAIN STRUT DRAG STRUTS

FIGURE 2-2 LEGGED LANDER COMPONENTS





behavior of support structure can be simulated with the program. The program also allows simulation of cables, or any support members with restricted load carrying capability in certain directions. Mode shapes and natural frequencies may be obtained using this program. For modal analysis, a routine is incorporated to allow reduction in the size of the stiffness matrix for large structures, thus permitting a reduction in the required computer storage requirements and run time.

New capabilities available in this option of the Structural Analysis Program over the Task Order Two version follow. An optional iterative technique, the Conjugate Gradient Method, was added to increase program efficiency for solution of nodal deflections. A more efficient data input routine was programmed reducing core size required by this routine. The maximum number of bars which can be used to idealize a structure was increased from 74 to 130. A rectangular shear web element was added to expand idealization capability. An additional 2500 words of storage was provided for storing nonzero terms of the stiffness matrix. An improved matrix reduction routine was developed to increase efficiency when employing the modal analysis routine. The maximum number of lowest natural frequencies and associated mode shapes (excluding rigid body modes) that can be requested from the modal analysis routine was increased from 5 to 95. In addition, the capability for handling multiple data cases was added. These capabilities result in a program whose core storage requirements at execution utilize the maximum allowable storage (70K) specified for computer programs by NASA Langley Research Center.

The iterative method programmed for solution of nodal deflections was selected to minimize computer core storage allocated to the solution routine. It also permits minimization of computer time for a preliminary design problem, since accuracy requirements can be specified.

Both straight bar members and rectangular shear webs are available for idealization of the center body structure. As explained in Section 4.1.1.2, the program internally replaces a shear web by two diagonal bar elements. Hence the bar element is the basic element of the Center Body Option.

Each bar element is capable of carrying axial load, shear in two directions, bending in two directions, and torsion. Bar properties are

assumed to be constant between nodes (symmetry about one of the principal axes is assumed). The bar's longitudinal axis is assumed to pass through the centroidal axis and the bending neutral axes are assumed to align with centroidal axes. Torsional shear center of the bar is assumed to be on the bar's longitudinal axis.

Cables, which have one end attached to a support and the other to a structural joint, may be idealized by restricting the compression load carrying capability to zero, thus exercising the plastic analysis program logic.

Plastic behavior of support structure, such as crushable attenuators, may be simulated by inputting the force causing plastic deformation of each support member. An upper and lower limit must be input on the particular force or moment being constrained.

The program is based on small deflection theory with Hooke's Law applying, except with regard to the plastic support option of the program. Buckling of members is not considered, and coupling effects, such as occur with beam-columns, are neglected. Bars are assumed to be rigidly connected to each other, unless otherwise specified. Bars pinned at both ends may be simulated by setting the appropriate moments of inertia equal to zero. Bars pinned at one end and fixed at the other may be idealized through the use of special input indicators. Loads are applied to the joints as concentrated forces in global coordinates.

Computer core storage restrictions necessitated a number of limitations on the problem size which may be considered with the Center Body Option. As described in the operating instructions, this storage can be increased to accommodate larger problems. Within the current limitations, the allowable problem size is governed by the following considerations: up to 74 joints with six degrees of freedom at each joint, or a maximum assembled stiffness matrix 444 by 444; 130 bars maximum; 30 shear webs maximum, each of which reduces the allowable number of bars by two; 26 reference points maximum; 300 support constraints maximum; 88 plastic force constraints maximum (88 upper and 88 lower limits); and 88 nonzero input deflections maximum. Mode shapes and natural frequencies can be determined for a maximum assembled stiffness matrix size of 102 by 102 which corresponds to 17 joints with six

degrees of freedom or many joints with fewer degrees of freedom. A reduction routine permits modal analysis of large, complex structures because the associated large stiffness matrix (444 x 444 maximum) may be reduced to the allowable size (102 by 102) by selectively eliminating degrees of freedom. The program utilizes the mode shapes of the reduced system to generate mode shapes for all degrees of freedom of the original large stiffness matrix. The number of lowest natural frequencies and associated mode shapes requested from the modal analysis routine for any structure with free-free support must be a multiple of five and less than the order of the reduced stiffness matrix minus six.

2.2.2 Landing Gear Option - The Landing Gear Option of the Structural Analysis Program provides the capability for investigating the large displacement stroking behavior, energy absorption characteristics, and internal load distribution in an inverted tripod or cantilever landing gear. These characteristics are obtained by solving the displacement equations of equilibrium at specified steps of applied footpad joint parameters. These parameters can be specified in either of two ways: (1) a displacement normal to an arbitrarily oriented landing surface and the friction coefficient of the surface can be specified or (2) all three orthogonal displacement components in the Surface Coordinate System can be specified. The number of steps to be used for investigating the displaced state must be specified. The landing surface can be arbitrarily oriented with respect to the lander by specifying three Euler angles which orient the Surface Coordinate System relative to the Lander Coordinate System. A least squares solution technique is employed to solve the equations of equilibrium for the displacements at each step. The gear geometry and strut properties are arbitrary input quantities.

The program determines the external forces and displacements at the node points of the gear in the Surface Coordinate System, the internal loads in all struts, and the total energy components absorbed by the gear in both the Surface and Lander Coordinate Systems. The energy associated with crushing of footpad attenuation material can also be included for a footpad with up to three honeycomb crush levels.

The inverted tripod landing gear is idealized in the program with four nodes and three pin-ended elements representing the main strut and two drag struts. These elements have only axial load-carrying capability and may contain attenuation material with up to five crush levels in both tension and compression.

The cantilever landing gear is idealized in the program with five nodes and four elements. The main strut is idealized with two elements both of which have bending and axial load-carrying capability. These members are assumed to be rigidly connected at the center junction of the main strut. The junctions of the main strut with the center body and the footpad are assumed to be pinned. The drag struts are pin-ended with axial load-carrying capability only. The drag struts and the lower element of the main strut contain attenuation material with up to five crush levels in both tension and compression.

For both the inverted tripod and cantilever gear idealizations, all element properties are assumed constant between nodes. None of the elements of either gear is assumed to be capable of carrying torsional loads. The bending moment of inertia about any axis normal to the longitudinal axis of the main strut of the cantilever gear is assumed to be constant and beamcolumn effects are not included. Each strut of either gear may be of a different material.

2.3 Landing Loads and Motions Program - The Landing Loads and Motions Program predicts the landing dynamics of a legged lander. The lander is idealized as a center body structure to which the landing gears are attached. A small footpad which contacts the landing surface is located at the base of each landing gear. Program options are available for obtaining landing motions, landing loads, and stability information for planar or spatial landings on many different types of landing surfaces.

The lander center body may be idealized as either a rigid body or the effects of a flexible structure may be included. For a rigid center body up to six rigid body degrees of freedom may be included in the analysis. To conserve computer run time, any combination of center body rigid degrees of freedom may be suppressed when running planar landing cases. The flexible

center body is represented by the superposition of a number of free-free vibratory modes on the rigid body motion. From one to five modes may be included in the analysis. Flexible center body information may be obtained from either the Center Body Option of the Structural Analysis Program or some other eigenvalue program.

For a given legged lander configuration, up to five gears may be considered which may be either inverted tripod or cantilever gears. Each gear consists of a main strut and two drag struts which have pinned ends; thus, no moments or torques may be introduced at their ends. Both the main strut and drag struts are capable of carrying tension and compression loads and may possess velocity dependent force characteristics, elastic-plastic load-stroke characteristics, or a combination of the two. Five plastic load levels are available in both tension and compression for all of the landing gear struts. The load-stroke characteristics of all main struts in a given lander configuration are the same. Likewise, these characteristics for all the drag struts are the same, however, they may be different than the main struts. For a cantilever gear, the effect of main strut bending is included by modifying the elastic portion of the drag strut load-stroke relationship. Any combination of a constant magnitude Coulomb friction force, or a damping force dependent on the magnitude of the strut stroking velocity may be included in either the main struts, drag struts, or both. These friction forces are directed opposite to the strut stroking velocity. Relative motion between the center body and each footpad is employed in determining the magnitudes and directions of the landing gear strut loads.

Each footpad is represented as a single mass with three rigid body translational degrees of freedom. One degree of freedom is normal to the landing surface and the other two are in the plane of the landing surface. On an optional basis, a plastic load attenuation material, with up to three crush levels, may be located on the bottom of each footpad. For footpads whose equations of motion are not being integrated, the associated gears are assumed to be extensions of the center body structure and their inertia effects are included in the center body equations of motion.

13

ļ

Two soil mechanics routines are available for studying the footpad-soil interaction phenomenon. One method is similar to the footpad-soil interaction analysis developed during the Lunar Module Soil Mechanics Study. In this case, the soil is represented in terms of a number of semiempirical relationships. The second method determines the soil force through a simple elasticplastic relationship between soil pressure and depth of soil penetration in conjunction with a coefficient of friction.

In addition to the standard program output defining the time histories of the footpad and center body motions and the load-stroke time histories in each landing gear strut, a number of additional output options are available. Accelerations at as many as ten points on the center body may be determined. This option allows the determination of landing accelerations for equipment items throughout the center body or correlation with test data obtained from accelerometers located at points other than the center of gravity.

The time history of the lander stability angle and pitching velocity of the lander in the direction of minimum stability may be obtained. This option, in conjunction with the spatial capabilities of the program and the two soil mechanics routines, allows a comprehensive study of the lander's stability characteristics to be made.

On an optional basis, all of the time history quantities required for defining the landing loads and acceleration patterns throughout the center body structure may be output on a magnetic tape. Points in time corresponding to possible high center body landing loads are determined by evaluating the printed output data obtained from the Landing Loads and Motions Program. The data on the magnetic tape may then be retrieved and the total landing load distribution throughout the center body structure determined at these specific time points of interest. The resulting loads are a combination of the inertia loads, gravity loads, and landing gear loads acting on the center body structure. These landing loads can then be input to the Center Body Option of the Structural Analysis Program and the internal loads in the individual structural members obtained.

Two numerical integration methods are incorporated in the program. These consist of a constant step Runge-Kutta method and a variable step Runge-Kutta

method. The constant step method allows definite user control over integration step size. The variable step method often results in less computer time for long runs such as required for stability cases.

THIS PAGE INTENTIONALLY LEFT BLANK

3. STRUCTURAL DESIGN CRITERIA

In the design of a legged lander configuration, specific constraints and criteria are imposed on the design dependent on the mission to be accomplished by the vehicle. Presented in the following sections are typical design constraints and factors of safety to be used during the design process of a legged lander. These were used as guidelines in determining the capabilities and options to be incorporated in the two computer programs developed during Task Order Five. However, it is emphasized that these do not represent limitations on the computer programs.

3.1 <u>Design Constraints</u> - The following factors must be considered in landing system and payload structure design: simplicity, reliability, stowability, structural compatibility, environmental compatibility, weight and sterilizability. Methods must be provided for accomplishing postlanding payload exposure to permit operations of experiments such as bioscience and imagery; measurements of wind velocity and direction, ambient pressure, temperature, and humidity; determination of soil composition; and operation of systems such as power, communication, and thermal control. In addition, the following is a typical list of specific constraints which should be considered.

- (1) Mass of the landed vehicle shall be 584 kilograms.
- (2) The landed vehicle (center body structure and landing gear structure) shall be compatible with an 3.35 meter base diameter, 120° blunted cone entry vehicle.
- (3) Touchdown shall occur at a vertical velocity of 6.1 meters per second (relative to the gravity vector), and a maximum horizontal velocity of 3.7 meters per second.
- (4) The total mass of the scientific payload shall be a minimum of 175 kilograms and the payload packing density shall not exceed a maximum of 943 dynes/cm³.
- (5) The landing vehicle shall not be restricted in orientation about the roll axis.
- (6) Pitch and yaw attitudes at touchdown may vary as much as $\pm 10^{\circ}$ from a plane normal to the gravity vector.

- (7) The landing vehicle shall have as a goal the capability of successfully landing on slopes of 30° to the local horizontal.
- (8) The landing system shall have the capability of performing satisfactorily when landing on surfaces containing particles varying in size from sand to 12.7 centimeter diameter rocks.
- (9) The atmospheric pressure at the surface shall be assumed to be nine mb.
- (10) The drag force on the footpad shall vary with penetration and applied normal forces.
- (11) The landing surface shall be assumed to have an average crushing stress of 41 x 10⁸ dynes/m² for penetrations to depths of 15.2 centimeters, a constant density of 1414 dynes/cm³ for penetrations to depths of 15.2 centimeters and an angle of internal friction of 39 degrees.
- (12) The coefficient of sliding friction between the surface and the footpad shall be assumed to be 0.3.
- (13) Payload deceleration at any point in the payload shall be limited to a maximum of 20 earth g-units and landing deceleration of the footpads shall be limited to a maximum of 250 earth g-units.
- (14) Methods shall be available for accomplishing post landing payload exposure to permit operation of experiments such as bioscience and imagery; measurements of wind velocity and direction, ambient pressure, temperature, and humidity; determination of soil composition; and operation of systems such as power, communication, and thermal control.
- (15) Post landing orientation required by certain individual equipment items will be accomplished by aligning the item's positive X axis within <u>+5</u> degrees to an axis perpendicular to the local surface slope.
- (16) Materials considered for use in the structures shall be compatible with space environment and a maximum temperature of 500°K and shall have minimum outgassing characteristics when exposed to a vacuum. Organic materials shall not be used in areas which may be subject to abrasion with the landing surface or to fragmentation with subsequent scattering of fragments on the landing surface.

(17) Surface gravitational acceleration shall be assumed to be 375 cm/sec^2 .

3.2 <u>Factors of Safety</u> - The following factors shall be applied to the maximum loads (limit loads) encountered in planetary landing within the constraints specified above.

Energy Absorbing Material for Landing Gear Struts1.00All Other Structures1.25

The load obtained by multiplying limit load by the appropriate factor of safety is the ultimate load used in sizing the structure.

The landing gear system shall have capability for stroke greater than that required for landings within constraints defined herein. This additional stroke shall be available to provide clearance between the bottom of the center body and a rock lying on the surface. THIS PAGE INTENTIONALLY LEFT BLANK

4. STRUCTURAL ANALYSIS PROGRAM

The Structural Analysis Program contains a Center Body Option and a Landing Gear Option. The Center Body Option determines internal load distributions in the center body structure, and generates modal data for use in the Landing Loads and Motions Program. Energy absorption capability of a single landing gear undergoing large displacement stroking motion is investigated with the Landing Gear Option.

In the Center Body Option, internal and external load distributions and deflection patterns are determined for any system of forces or deflections impressed on an elastic network of structural bar members and rectangular shear webs. Problems for which the support structure behaves plastically can be solved. In addition, mode shapes and natural frequencies can be obtained for a center body with free-free support.

The Landing Gear Option of the program is used to analyze the large displacement stroking behavior of a single inverted tripod or cantilever landing gear. Energy absorption characteristics of the gear and internal strut loads are obtained for any number of applied footpad displacement steps. The footpad can be required to stroke a given distance along any arbitrarily oriented straight line, or a given distance normal to an arbitrarily oriented frictional plane. In the latter case, the coefficient of friction between the footpad and the frictional plane determines the sliding path of the footpad in the plane.

Analytical methods employed in the linear small displacement and nonlinear large displacement finite element portions of the Structural Analysis Program are presented in Section 4.1. A discussion of the normal mode method for obtaining modal data is also included in this section. Organization of the program is described in Section 4.2 and information necessary to operate the program is presented in Section 4.3. A listing of the program is contained in Appendix H.

4.1 <u>Analytical Methods</u> - The landing system and payload structures are highly redundant space frames consisting of a network of members possessing extensional, flexural, and torsional stiffness. Several methods for solving complex structural problems are in use today that effectively utilize the
computer. Two methods considered for the Structural Analysis Program were the finite difference method and the finite element method. The finite element method was selected because of simplicity in dealing with nonhomogeneous, anisotropic structural applications. In addition, the elements can be changed easily in shape and size to follow complex boundary conditions or to allow for regions of rapid changes in stress or deflection.

A fundamental part of the finite element method is the technique used to obtain displacements at junctions of elements. Two approaches considered for the Structural Analysis Program were the force (flexibility) method and the stiffness (displacement) method. The stiffness method was selected primarily because the stiffness matrix developed may be used directly to generate the centerbody normal modes required to include centerbody flexibility in the Landing Loads and Motions Program. Also, the stiffness method lends itself readily to the large displacement technique necessary for landing gear analysis.

4.1.1 <u>Center Body Option</u> - The Center Body Option of the Structural Analysis Program can be used to analyze a structure idealized with an elastic network of structural bar elements and rectangular shear webs. The program internally replaces a rectangular shear web by two diagonal bar elements whose stiffness characteristics are equivalent to the shear web. Hence, the basic element of the program is the bar element.

4.1.1.1 <u>Coordinate System</u> - Two coordinate systems employed in the Center Body Option are the Local Coordinate System and the Global Coordinate System. Each bar member has its set of local coordinates as shown in Figure 4-1. Displacement notation of a general bar element capable of carrying axial load, shear in two directions, bending in two directions, and torsion is also indicated in Figure 4-1 for the Local Coordinate System. Subscript 1 refers to a displacement due to axial load, subscripts 2 and 3 are for displacement due to shear loads, subscript 4 is for a rotation due to torque, and subscripts 5 and 6 are for rotations due to moments. The Local Coordinate System origin for each bar is located at joint "p," with the X_{g} axis aligned along the member axis. Positive X_{g} is on the side of joint "p" towards joint "q." The Y_{g} axis is perpendicular to X_{g} and is located in the "pqr" plane. Positive Y_{g} is on the side of X_{g} towards point "r." The Z_{g}



FIGURE 4-1 DISPLACEMENTS OF GENERAL BAR ELEMENT IN LOCAL COORDINATE SYSTEM axis is then established using the right hand rule. Local Coordinate Systems are established by identifying each bar's origin (joint "p"), end (joint "q"), and orientation of the bending axis Y_{ℓ} (defined by joint "r"). Moment of inertia I_T about the Y_{ℓ} axis, moment of inertia I_N about the Z_{ℓ} axis, cross-sectional area A, torsional constant J, modulus of elasticity E, and shear modulus G are specified for each bar relative to its Local Coordinate System.

A common coordinate system (global) for all structural elements must be established so that element forces and displacements may be related to a common frame of reference. This Global Coordinate System may be any convenient orthogonal right hand system. Displacement notation of the bar element in the Global Coordinate System is shown in Figure 4-2. Joint locations, external load distributions, and joint deflections are all specified in the Global Coordinate System.

4.1.1.2 <u>Stiffness Matrices</u> - A stiffness matrix for each bar element is generated in its Local Coordinate System based on small deflection theory. This is done by applying a unit displacement or rotation to one end of the bar (while restraining all other rotations and displacements) and determining the induced forces and moments. Displacements and rotations are applied sequentially until all degrees of freedom at each end of the bar element have been included. Displacement notation of a general bar element capable of carrying axial load, shear in two directions, bending in two directions, and torsion was shown in Figure 4-1.

Force-displacement relationships for the three unit displacements and three unit rotations possible at each end of the bar element are shown in Figure 4-3. Forces and displacements in the Local Coordinate System are related in matrix form by

$$\{\mathbf{F}\} = [\mathbf{K}] \{\delta_{\mathsf{T}}\} \tag{4-1}$$

In Equation (4-1), [K] is the matrix of stiffness coefficients; {F} is a column matrix of applied forces at the joints, and $\{\delta_L\}$ is the column matrix of joint displacements. Application of Maxwell's Reciprocal Law allows formulation of a symmetric stiffness matrix when an orthogonal coordinate system is employed. The element stiffness matrix for a general bar is given in



FIGURE 4-2 DISPLACEMENTS OF GENERAL BAR ELEMENT IN GLOBAL COORDINATE SYSTEM



FIGURE 4-3 FORCE - DISPLACEMENT RELATIONS FOR GENERAL BAR ELEMENT

Figure 4-4. Terms in this matrix are the stiffness coefficients given in Figure 4-3.

If it is desired to include the effects of shearing strain on elemental beam deflection, the program applies the stiffness factors given in Figure 4-5 to the corresponding terms in Figure 4-4. Input shear form factors K_N and K_T are determined by dividing the total cross-sectional area of the bar by the area effective in carrying shear loads. The area used in determining K_N should be the area effective in carrying shear in the local Y_{ℓ} direction (bending about the Z_{ℓ} axis), while K_T is determined based on effective area for shear in the Z_{ℓ} direction. An example calculation of shear form factors K_N and K_T is given in Figure 4-6.

In addition, a rectangular shear web element is incorporated in the Center Body Option. The shear web idealization used by the program is indicated in Figure 4-7. Here the shear web is replaced by two diagonal bar elements whose stiffness characteristics are equivalent to the shear web. Cross-sectional properties of diagonals are derived by requiring equivalent strain energies in shear web and diagonals for statically equivalent internal loads.

Both ends of a member may be pinned (allowed to freely rotate) in a given direction by setting the appropriate moment of inertia equal to zero. If it is desired to pin one end of a member, while the other end remains fixed, the program utilizes the terms shown in Figures 4-8 through 4-ll in place of the corresponding terms in Figure 4-5. Terms used depend on the joint (p or q) pinned and the bending axis (Y_{l} or Z_{l}) about which the end is allowed to rotate. For example, if it is desired to pin the q end of a member with respect to bending about the Y_{l} axis, terms in Figure 4-11 would be used in place of the corresponding terms in Figure 4-5.

To determine the stiffness matrix for the completely assembled structure, all element forces and displacements are related to the Global Coordinate System. Displacement notation of the bar element in Global Coordinate System is defined in Figure 4-2.

Transformation matrices are needed to change the frame of reference of each element from the Local to the Global Coordinate System. This transformation is expressed by the linear matrix equation

$$\{\delta_{I}\} = [\lambda]\{\delta_{G}\}$$
(4-2)

FIGURE 4-4 STIFFNESS MATRIX FOR GENERAL BAR ELEMENT

$$\begin{array}{c} {}^{u_{1p} \quad u_{2p} \quad u_{3p} \quad u_{1q} \quad u_{2q} \quad u_{3q} \quad u_{4p} \quad u_{5p} \quad u_{6f} \quad u_{4q} \quad u_{5q} \quad u_{6q}} \\ F_{1q} \\ F_{2p} \\ F_{3p} \\ F_{3p} \\ F_{3p} \\ F_{1q} \\$$

FIGURE 4-5 STIFFNESS FACTORS APPLIED TO ORIGINAL MATRIX TO ACCOUNT FOR ELEMENTAL BAR SHEAR STRAIN



FIGURE 4-6 EXAMPLE CALCULATION OF SHEAR FORM FACTOR



GIVEN SHEAR WEB:

MODULUS OF ELASTICITY E, POISSON'S RATIO v, AND THICKNESS t. CORNER JOINTS A, B, C, D NUMBERED COUNTERCLOCKWISE.

DIAGONAL REPLACEMENT OF WEB BY PROGRAM:

ADD 2 BARS AS SHOWN. BARS ARE PINNED AT BOTH ENDS AND NOT ATTACHED TO EACH OTHER. BAR PROPERTIES ARE AS FOLLOWS:

AREA = t. $(a^2 + b^2)^{3/2}/4ab (1 + v)$. E= E_{web} OTHER PROPERTIES = 0 WHERE a = LENGTH AB b = LENGTH AD

INTERPRETATION OF PROGRAM OUTPUT:

SHEAR FLOW IN WEB; $q = (P_{\overline{2}} - P_{\overline{1}}) \sqrt{a^2 + b^2}$

WHERE $P_{\overline{2}} = AXIAL LOAD IN BAR \overline{2}$.

 $P_{\overline{1}} = AXIAL LOAD INBAR \overline{1}.$

SIGN CONVENTION; + 9 CORRESPONDS TO SHEAR FLOW ON SIDE AB FROM A TO B.

+ P CORRESPONDS TO TENSION IN BAR.

FIGURE 4-7 SHEAR WEB REPLACEMENT WITH DIAGONALS



$$a_{N} = \frac{12 E I_{N} / \ell^{3}}{GA / \ell} K_{N}$$

FIGURE 4–8 SUBSTITUTE FACTORS IN ${\sf K}_S$ MATRIX TO ALLOW ELIMINATION OF ROTATIONAL RESTRAINT IN Z DIRECTION AT POINT P







FIGURE 4-10 SUBSTITUTE FACTORS IN K_S MATRIX ALLOWING ELIMINATION OF ROTATIONAL RESTRAINT IN Y DIRECTION AT POINT P



$$a_{T} = \frac{12E I_{T} / l^{3}}{GA / l} K_{T}$$

FIGURE 4–11 SUBSTITUTE FACTORS IN ${\rm K}_{\rm S}$ MATRIX ALLOWING ELIMINATION OF ROTATIONAL RESTRAINT IN Y DIRECTION AT POINT Q

where $[\lambda]$ is a matrix of direction cosines (cosines of angles between Local and Global Coordinate Systems) obtained by resolving global displacements in the direction of local coordinates. For the general bar, $[\lambda]$ is given in Equation (4-3).

$$\lambda = \begin{bmatrix} \lambda & 0 & 0 & 0 \\ 0 & \overline{\lambda} & 0 & 0 \\ 0 & 0 & \overline{\lambda} & 0 \\ 0 & 0 & 0 & \overline{\lambda} \end{bmatrix}$$
(4-3)

In Equation (4-3), each $\overline{\lambda}$ is a 3 by 3 matrix of the direction cosines of the local coordinate axes relative to the global system, as shown in Equation (4-4).

$$\overline{\lambda} = \begin{bmatrix} {}^{\ell} 1 & {}^{m} 1 & {}^{n} 1 \\ {}^{\ell} 2 & {}^{m} 2 & {}^{n} 2 \\ {}^{\ell} 3 & {}^{m} 3 & {}^{n} 3 \end{bmatrix}$$
(4-4)

The row format of λ corresponds to the sequence of displacements in the Local Coordinate System specified for the stiffness matrix in Figure 4-4. The column format of λ corresponds to a similar sequence in the Global Coordinate System (i.e., $\overline{u_1}$, $\overline{u_2}$, $\ldots \overline{u_6}$). Values of the direction cosines relating the Local Coordinate System to the Global Coordinate System are determined as indicated in Figure 4-12.

Each element stiffness matrix is transformed from the Local to the Global Coordinate System by Equation (4-5)

$$[\overline{K}] = [\lambda]^{T}[K][\lambda]$$
(4-5)

where $[\overline{K}]$ is the element stiffness matrix transformed to Global Coordinate System and $[\lambda]^{T}$ is the transpose of transformation matrix $[\lambda]$.

The total stiffness matrix (K_A) for the assembled structure (in the Global Coordinate System) as shown in Figure 4-13, is generated by systematically adding the transformed element stiffness matrices. Nodal points on



DIRECTION COSINE	COSINE OF ANGLE BETWEEN		FOUNTION FOR DIRECTION COSINE
	LOCAL AXIS	GLOBAL AXIS	EQUATION FOR DIRECTION COSINE
ℓ_1	×L	x	$(\overline{X}_q - \overline{X}_p)/d_{pq}$
^m l	×L	Ÿ	$(\overline{Y}_{q} - \overline{Y}_{p}) d_{pq}$
۲	×Į	τ Ξ	$(\overline{Z}_{q} - \overline{Z}_{p}) d_{pq}$
l ₂	۲L	x	$[(\bar{\mathbf{X}}_{r} - \bar{\mathbf{X}}_{p}) - \mathbf{l}_{l}d_{ps}]/d_{rs}$
^m 2	۲Ł	Ţ	$[(\overline{Y}_r - \overline{Y}_p) - m_1 d_{ps}]/d_{rs}$
ⁿ 2	۲Ł	Ī	$[(\overline{Z}_r - \overline{Z}_p) - n_1 d_{ps}]/d_{rs}$
ℓ_3	zl	x	^m 1 ⁿ 2 ^{-m} 2 ⁿ 1
^m 3	zſ	Ϋ́	$- l_{1 n_2} + l_{2 n_1}$
ⁿ 3	zſ	Ī	$\ell_1 m_2 - \ell_2 m_1$

NOTE: POINT F IS AN ARBITRARILY SELECTED POINT LYING IN THE $\chi l - \chi l$ PLANE. IT IS USED TO IDENTIFY THE ORIENTATION OF MEMBER BENDING AXES χl AND χl RELATIVE TO THE GLOBAL COORDINATE SYSTEM. (IN IS THE MOMENT OF INERTIA ABOUT THE χl AXIS AND IT IS THE MOMENT OF INERTIA ABOUT THE χl AXIS.) FIGURE 4-12 DIRECTION COSINES OF LOCAL COORDINATE SYSTEM RELATIVE TO GLOBAL COORDINATE SYSTEM



SYMBOLS † AND r INDICATE TRANSLATIONAL AND ROTATIONAL DISPLACEMENTS RESPECTIVELY.

FIGURE 4-13 ASSEMBLED STIFFNESS MATRIX FORMAT

the idealized structure are numbered consecutively from 1 to n_J . The stiffness matrix K_A is assembled with a row and column format corresponding to the three translational followed by the three rotational degrees of freedom in the global system at each node in sequence. In this general case, the size of the stiffness matrix is $6n_J \propto 6n_J$.

4.1.1.3 <u>Elastic Analysis</u> - The assembled stiffness matrix $[K_A]$ is related to the column matrices of global forces and displacements at each node by Equation (4-6) which represents a combination of Equation (4-1) through (4-5). The word "forces" in this discussion implies both forces and moments, and the word "displacements" implies both deflections and rotations.

$$\{\mathbf{F}\} = [\mathbf{K}_{\mathbf{A}}]\{\delta_{\mathbf{G}}\}$$
(4-6)

The stiffness matrix is singular. That is, its determinant vanishes and its inverse does not exist. Boundary conditions (supports) must be defined to prevent rigid body motion. Once $[K_A]$ has been determined, a solution can be obtained for any set of support conditions.

As described in Reference (3), the unknown nodal displacements and support reactions are normally obtained by partitioning Equation (4-6) according to the location and orientation of supports as indicated in Equation (4-7).

$$\begin{cases} F_{(m-n)} \\ F_{n} \end{cases} = \begin{bmatrix} A_{(m-n) \times (m-n)} & B_{(m-n) \times n} \\ B_{n \times (m-n)} & D_{n \times n} \end{bmatrix} \begin{pmatrix} \delta_{(m-n)} \\ \delta_{n} \\ \delta_{n} \end{cases}$$
(4-7)

Subscripts n and m are

n = number of support boundary conditions (i.e., $\delta_n = 0$),

m = order of stiffness matrix.

This partitioning results in two sets of equations: Equation (4-8) relates unknown nodal displacements to known applied forces, and Equation (4-9) relates unknown support reactions to unknown nodal displacements.

$$\{F_{(m-n)}\} = [A]\{\delta_{(m-n)}\}$$
(4-8)

$$\{F_{n}\} = [B^{\prime}]\{\delta_{(m-n)}\}$$
(4-9)

The inverse of matrix [A] in Equation (4-8) is the flexibility matrix of the structure. Equation (4-8) may be rewritten to give unknown nodal displacements in terms of the flexibility matrix and known applied forces as shown in Equation (4-10).

$$\{\delta_{(m-n)}\} = [A]^{-1}\{F_{(m-n)}\}$$
(4-10)

The unknown reactions may be obtained by combining Equations (4-9) and (4-10) as shown in Equation (4-11).

$$\{F_n\} = [B^{-1}][A]^{-1}\{F_{m-n}\}$$
(4-11)

The preceding discussion outlines the method normally used to obtain nodal displacements and reactions. Due to computer core limitations and the desire to minimize program running time, an iterative method of determining nodal deflections and reactions, once the stiffness matrix was established, was programmed. An iterative method eliminates the need for matrix inversion.

The requirement for plastic supports calls for multiple solutions of sets of equations very nearly the same. These sets of simultaneous equations are represented by matrix Equation (4-8), relating unknown nodal deflections to known applied forces. This type of problem is best handled by an iterative solution, since the required nodal deflections are approximately known after the initial elastic solution is determined, thus minimizing computing time. Four iterative techniques were programmed; Gauss-Siedel, Gauss-Siedel with Aitkens Delta Squared improvements, the Overrelaxation method, and the Conjugate Gradient technique. Reference (4) describes the first three methods and Reference (5) the latter. Any one of these techniques may be best suited for a specific structural problem. However, experience dictates that for highly redundant space frames, such as the legged lander, the Conjugate Gradient technique appears to be the best suited of the above solution methods. After nodal displacements have been determined using the iteration procedure, the unknown support reactions are obtained by substitution of these nodal displacements into matrix Equation (4-9). Forces on elements in the Local Coordinate System are found by transforming the nodal displacements into the Local Coordinate System using Equation (4-2) and applying the appropriate force-displacement relationships using Equation (4-1).

The program also is capable of solving problems wherein the nodal displacements are known, as in problems where some supports settle. Combinations of known applied forces and known nodal displacements may be input, and all forces and displacements will be determined. Boundary conditions imposed on the problem must be sufficient to prevent rigid body motion. Partitioning of Equation (4-7) for this case results in Equation (4-12), relating unknown nodal displacements to known applied forces and known nodal displacements, and in Equation (4-13), which relates unknown forces and reactions to the now known (determined in Equation (4-12)) nodal displacements. Subscript n for this case implies either zero or nonzero known boundary conditions.

$$\{F_{(m-n)}\} = [A]\{\delta_{(m-n)}\} + [B]\{\delta_n\}$$
(4-12)

$$\{F_n\} = [B^{\prime}]\{\delta_{(m-n)}\} + [D]\{\delta_n\}$$
(4-13)

This solution is simply a more general case of the initial partitioning (Equation (4-8) and (4-9)), where the only known nodal displacements were zero and terms involving δ_n were therefore not included.

4.1.1.4 <u>Plastic Support</u> - When analyzing a space frame it may be desirable to idealize support members with restricted load-carrying capability. An upper (positive) and lower (negative) limit can be set on the magnitude of any force or moment component in such a support member. Initial solution of the space frame redundancy assumes elastic deformation, but the magnitude of the load components at the support are compared with input limits to ascertain if these have been exceeded. Those load components (reactions) which exceed the input limits are then assumed to be at the input cutoff values. The new set of boundary conditions (new column matrix of deflections and forces) are then employed with the stiffness matrix to obtain a new solution. The load components at the supports and the limits are again compared. If the limits are exceeded, the process is repeated until no limits are exceeded and equilibrium is achieved.

A plastic attenuator support can be modeled with this feature by setting the appropriate input limit to the force causing plastic deformation. Cables with one end attached to a support and the other to a structural joint can be idealized by making one of the limits on each reaction force component zero, allowing the cable to carry only tension and no compression. Determination of the proper limit (positive or negative) to use on a particular force component depends on whether the desired limit is in the positive or negative global coordinate direction. If a desired tension force in a cable results in a force component in the positive global direction at the support, the lower (negative) limit is set equal to zero. If the desired tension force component is in the negative global direction, the upper (positive) limit is set equal to zero.

4.1.1.5 <u>Modal Analysis</u> - The frequencies and mode shapes for the freefree center body structure are determined once the unrestrained stiffness matrix of the center body (Section 4.1.1.2) has been obtained. This modal analysis is performed in the optional Modal Analysis Routine in the Center Body Option of the Structural Analysis Program.

Free vibrations of the center body structure are defined by Equation (4-14)

••

$$[M]{q(s, t)} + [K]{q(s, t)} = 0$$
(4-14)

where

The above representation of the stiffness and inertia characteristics of the center body are input data for the Modal Analysis Routine. An eigenvalue routine is used to obtain the vibratory free-free mode shapes and corresponding frequencies.

The large order center body stiffness matrix results in an eigenvalue problem which is too large to solve practically. The computer run time required to obtain the frequencies and mode shapes would be excessive. In addition, the eigenvalue problem would exceed the allotted computer core storage requirements. For these reasons, the size of the center body's structural stiffness matrix is reduced before solving the eigenvalue problem. This reduction technique is discussed in Reference (6).

In the reduction procedure, a number of degrees of freedom, corresponding to various displacements and rotations at the center body joints, are removed. To remove these, it is assumed that the inertia forces and/or moments associated with these degrees of freedom are negligible. In this procedure, all of the strain energy associated with the removed degrees of freedom is retained. This reduction procedure is outlined as follows.

$$\left\{ \begin{array}{c} \mathbf{P} \\ \mathbf{0} \end{array} \right\} = \left[\begin{array}{c} \mathbf{K}_{11} & \mathbf{K}_{12} \\ \mathbf{K}_{21} & \mathbf{K}_{22} \end{array} \right] \left\{ \begin{array}{c} \mathbf{q}_{1} \\ \mathbf{q}_{2} \end{array} \right\}$$
(4-15)

where

K₁₁, K₁₂, K₂₁, K₂₂ = segments of total center body stiffness matrix. q₁ = degrees of freedom on which forces and moments (P's) exist. q₂ = degrees of freedom on which negligible forces and moments exist.

In the above, the total stiffness matrix has been reordered such that the elements associated with the degrees of freedom to be retained appear first.

Equation (4-15) is equivalent to the two following expressions

 $\{P\} = [K_{11}]\{q_1\} + [K_{12}]\{q_2\}$ (4-16)

$$\{0\} = [K_{21}]\{q_1\} + [K_{22}]\{q_2\}$$
(4-17)

Solving the second of these for $\{q_2\}$ gives

$$\{q_2\} = -[K_{22}]^{-1}[K_{21}]\{q_1\}$$
(4-18)

Substituting this into Equation (4-16) results in the reduced form

$$\{P\} = \left[[K_{11}] - [K_{12}] [K_{22}]^{-1} [K_{21}] \right] \{q_1\}$$
(4-19)

from which the reduced stiffness matrix, [K*], is defined as

$$[K^*] = [K_{11}] - [K_{12}][K_{22}]^{-1}[K_{21}]$$
(4-20)

The eigenvalue problem associated with the reduced system is

$$[M*]{q_1} + [K*]{q_1} = 0$$
 (4-21)

where [M*] is a diagonal mass matrix whose elements represent the distribution of the center body mass at the degrees of freedom to be retained. Frequencies and mode shapes of the total elastic center body structure are obtained using Equations (4-18) and (4-21). Reference (7) summarizes the Householder-Ortega-Wilkinson Method used to determine mode shapes and natural frequencies. An example of the use of this routine is given in Section 4.3.1.3. 4.1.2 <u>Landing Gear Option</u> - The Landing Gear Option of the Structural Analysis Program can be used to investigate the energy absorption characteristics and internal loads in inverted tripod or cantilever landing gears. The finite element stiffness method employed in this option of the program is based on large displacement (nonlinear) finite element theory.

In linear finite element theory, changes in the stiffness matrix due to displacements are assumed to be negligible. Landing gear struts experience large rotational and extensional displacements during stroking of the gear. These displacements require that the finite element idealization of the struts include "rotational" and "extensional" nonlinearities. To illustrate the nonlinear nature of the problem, consider a planar gear subjected to the displacements shown in Figure 4-14. Stiffness of the gear changes appreciably during stroking from the original to the displaced position. This change is due to the following displacement nonlinearities:

- rotational nonlinearities large rotations cause significant changes in strut orientation and the resulting stiffness;
- (2) extensional nonlinearities compressive (or tensile) strut crushing changes the strut length and slope of the axial load-stroke curve

(referred to as axial stiffness) both of which alter the stiffness. For these reasons linear theory cannot be used in the analysis of landing gears since changes in the stiffness matrix due to displacements are not negligible.

4.1.2.1 <u>Modified Incremental Stiffness Method</u> - To account for the nonlinearities associated with landing gear analysis, the incremental stiffness finite element method is employed. A modification of this method is made to insure that load unbalances do not result in the gear due to the linear approximations made in each step of the solution technique.

In the incremental stiffness method, nonlinear behavior is approximated through a sequence of linear solutions. The loading is divided into a number of incremental steps. For each step, an increment of the external load is applied and incremental displacements determined. These displacements, when added to the structural positions at the conclusion of the previous step, define updated geometry. The incremental stiffness matrix, used to determine these incremental displacements, is updated at the conclusion of each step to reflect changes due to displacement nonlinearities. Because small increments



(c) TYPICAL NONLINEAR LOAD-STROKE CURVE

FIGURE 4-14 LARGE DISPLACEMENTS OF A TYPICAL TWO-DIMENSIONAL GEAR

are used, a linear small displacement problem is solved at each step. The method is illustrated in Figure 4-15 for a typical nonlinear load-stroke curve. For each step n, the incremental displacement $(\delta_n - \delta_{n-1})$ corresponding to the applied incremental load $(F_n - F_{n-1})$ is determined from the equation

$$\delta_n = \delta_{n-1} + (F_n - F_{n-1})/S_n, n = 1, \dots, m$$
 (4-22)

where m is the total number of incremental steps. In this equation, S_n is the incremental stiffness of the structure corresponding to the displaced state, δ_{n-1} . For example, at the conclusion of step 1 (point A₁ in Figure 4-15), the incremental stiffness S_2 (at point P₁) is determined for the displaced state, δ_1 . This stiffness and the next incremental load (F_2-F_1) are then used in Equation (4-22) to determine δ_2 (defining point A₂),

$$\delta_2 = \delta_1 + (F_2 - F_1)/S_2 \tag{4-23}$$

This process is repeated until all load increments have been applied. For the gear of Figure 4-14 the above method would be applied simultaneously to the nonlinear load-stroke curves in the X and Y directions.

Due to the linear approximation at each step of the incremental stiffness method, error in the solution accumulates as a function of step size. This can be seen in Figure 4-15 where at the conclusion of step 4 the difference between points A_4 and P_4 is significant. This error, the difference between the applied load (F_4) and the true load corresponding to the displaced state (the force at point P_4), is defined as the load unbalance. To eliminate this unbalance, the incremental stiffness method is combined with an iteration procedure to assure convergence to the correct solution. For each incremental step, iteration is employed to insure that the internal loads, corresponding to the displaced state, are in equilibrium with the applied loads.

The iteration approach as applied to the nth incremental step of Figure 4-15 is illustrated in Figure 4-16. The incremental displacement $\delta_n - \delta_{n-1}$, corresponding to the incremental applied load $F_n - F_{n-1}$, is sought. Knowing the incremental stiffness S_1 at the beginning of the nth incremental step, the first estimate of the incremental displacement, $\Delta \delta_1$, is found from

$$\Delta \delta_{1} = (F_{n} - F_{n-1}) / S_{1}$$
 (4-24)



FIGURE 4–15 INCREMENTAL STIFFNESS METHOD APPLIED TO A TYPICAL NONLINEAR LOAD-STROKE CURVE



INCREMENTAL STIFFNESS METHOD

This yields point A_1 . The true incremental stiffness S_2 and the resultant strut reaction load $F_n + \Delta F_2$ corresponding to the displacement state $\delta_{n-1} + \Delta \delta_1$ are defined by point P_1 . The force unbalance ΔF_2 , the difference between the total applied load, F_n , and the reaction load $F_n + \Delta F_2$, is applied to the structure whose incremental stiffness is S_2 . Thus, the new estimate of the incremental displacement, $\delta_1 + \delta_2$, is determined where δ_2 is defined by

$$\Delta \delta_2 = \Delta F_2 / S_2 \tag{4-25}$$

This process is repeated and a new estimate of the incremental displacement, $\stackrel{\ell}{\Sigma} \Delta \delta_{i}$, is determined using the relation i=1

$$\Delta \delta_{\varrho} = \Delta F_{\varrho} / S_{\varrho}, \ \ell = 3, \ \dots \tag{4-26}$$

until ΔF_{ℓ} is arbitrarily small (convergence at point A). When this occurs, the applied load is balanced by the reaction load and equilibrium has been attained.

For the gear of Figure 4-14, the above method would be applied simultaneously to the nonlinear load-stroke curves in the X and Y directions. For each of these directions, the force unbalance would be the difference between the applied load in that direction and the sum of the internal drag strut and main strut loads in that direction.

The modified incremental stiffness method is well suited to the analysis of landing gears. When employing this method, errors will not be introduced when structural stiffness properties change abruptly as is common in landing gear members containing attenuation. The error at each step of the process is known since the system is in equilibrium within a predetermined tolerance. In addition, the method provides a displacement-load history for any desired number of increments of applied gear stroke. This is essential for determining energy absorption characteristics of landing gears.

The modified incremental stiffness method, as applied to three dimensional inverted tripod or cantilever gears, requires the solution of the matrix equation

$$\Delta{\mathbf{F}} = [\mathbf{S}] \ \Delta{\{\delta\}} \tag{4-27}$$

Equation (4-27) is solved several times for each step of the loading. In this equation, Δ F } is a column matrix of incremental forces applied at the nodes of the gear; Δ { δ } is a column matrix of incremental nodal displacements; and [S] is the instantaneous incremental stiffness of the gear. For landing gear analysis, the independent variables in Equation (4-27) for each step of the process will be the applied displacements. The method of employing Equation (4-27) for each step of the process is as follows:

- (i) Equation (4-27) is solved <u>once</u> for the case where Δ{δ} is the total applied displacement vector divided by the number of incremental steps. The incremental stiffness matrix [S] of the gear corresponds to the displaced conditions existing at the conclusion of the pre-vious step. This resulting equation is analogous to Equation (4-22). The nodal locations are then updated.
- (ii) Equation (4-27) is then solved <u>repeatedly</u> (iteration is employed) until the largest load unbalance component is less than a predetermined tolerance. At the conclusion of each iteration, nodal locations are updated. For the first of these iterations, [S] corresponds to the displaced condition existing at the conclusion of (i). For all succeeding iterations, [S] corresponds to the displaced condition existing at the conclusion of the previous iteration. Δ {F} is a vector representing the difference between the applied load components and the components of internal load corresponding to the displaced condition existing at the conclusion of the previous iteration. Hence, Δ {F} represents the load unbalances. Each of the above equations is analogous to Equation (4-26).

4.1.2.2 <u>Structural Idealization</u> - The Landing Gear Option employs a fixed idealization for inverted tripod and cantilever landing gears. This idealization (node point and element numbering) must be adhered to when employing this option of the program.

The inverted tripod landing gear is idealized in the program with four nodes and three pin-ended elements (the main strut and two drag struts) as shown in Figure 4-17(a). These elements are capable of carrying axial loads only and may contain honeycomb attenuation for both tension and compression. Each element may be made of a different material. Each node is assumed to be

(a) INVERTED TRIPOD GEAR



FIGURE 4-17 LANDING GEAR IDEALIZATIONS

a pin which can support three force components but no moments. Each element of a gear has associated with it a set of three nodes (p, q, and r) which defines the Local Coordinate System, as explained in Section 4.1.2.3, for that element. The fixed idealization of the inverted tripod gear employs the following nodal point numbering system for the elements:

Element			
No.	р	q	r
1	1	4	2
2	2	4	3
3	3	4	2

The cantilever landing gear is idealized with five nodes and four elements as shown in Figure 4-17(b). The main strut is idealized with two elements, both of which are capable of carrying bending as well as axial loads. These members (elements 2 and 4 in Figure 4-17(b)) are assumed to be rigidly connected at node 4. This provides moment continuity along the main strut. For elements 2 and 4 of the main strut, the moment of inertia about any axis normal to the element is assumed to be constant. However, the moment of inertia for element 2 may be different than that for element 4. For each of these elements, the modulus of elasticity for bending displacements may be different than the modulus for axial displacements. The junctions of the main strut with the center body (node 2) and the footpad (node 5) are assumed to be pinned. The drag struts (elements 1 and 3) are pin-ended with axial load-carrying capability only. Thus, a drag strut cannot carry bending moments at either end. Both drag struts (elements 1 and 3) and the lower element of the main strut (element 4) may contain honeycomb attenuation for both tension and compression. Each strut of the cantilever gear may be made of a different material. The fixed idealization of the cantilever gear employs the following nodal point numbering system for the elements:

Element			
No.	Р	q	r
1	1	4	2
2	2	4	RP
3	3	4	2
4	5	4	RP

Point r (RP) for elements 2 and 4 is a floating reference point whose coordinates are continually changing. Initially, this point is selected as node 1. As the main strut bends, the floating reference point is located as described in Section 4.1.2.5.

For either of the above gears, struts which can carry axial loads only (all elements of the inverted tripod gear and elements 1 and 3 of the cantilever gear) are defined as "axial struts." Struts which are capable of carrying bending as well as axial loads (elements 2 and 4 of the cantilever gear) are defined as "bending struts."

For a strut which contains honeycomb attenuation, the material defining the strut properties is assumed to have a load-stroke curve for axial displacements similar to that shown in Figure 4-18. This curve represents the stroking characteristics of stacked honeycomb cartridges housed within a landing gear strut. Each cartridge is assumed to crush at constant load as the strut is stroked. Cartridges possessing different crushing strengths may be stacked in series to form a desired load-stroke characteristic. Up to five cartridges can be used to attenuate compression loads and up to five to attenuate tension loads.

For the typical load-stroke curve shown in Figure 4-18, determination of the axial load corresponding to stroking causing compression in the strut is explained in the following discussion. The axial load corresponding to tensile stroking of the strut would be determined in a similar manner.

As the strut initially begins to stroke, the strut load increases linearly with stroke to point 1 where the first crush load is reached. The load then remains constant with stroke until either the stroke reverses direction or a second elastic portion is reached. If the direction of stroke reverses, point 2, one of the following load-stroke sequences is possible:

- (1) Elastic unloading to an intermediate point between 2 and 4, such as point 3, at which time the compressive stroke again increases. This results in the load increasing elastically to point 2 and then following the original load-stroke curve.
- (2) Elastic unloading through point 3 to point 4. A continued decrease in stroke to point 5 occurs at a zero strut load. With a reversal of stroke the strut will compress with zero load until point 4 is



FIGURE 4–18 TYPICAL STRUT LOAD-STROKE CURVE

again reached. The load will then increase linearly to point 2 and continue to follow the original load-stroke curve.

(3) Elastic unloading through points 3, 4, and 5, followed by a continued decrease in stroke until the strut goes into tension. At this point, the strut load is governed by its tension load-stroke characteristics.

If at point 2 the stroke had not reversed direction, the load would remain constant until point 6 was reached. The force would then increase linearly with stroke and continue to follow the load-stroke curve until unloading took place. When the compressive stroke exceeds the maximum allowable stroke (SCMAX) the strut is assumed to have bottomed out.

Two options are available for selecting the slope of the strut's elastic unloading characteristic. This slope may be either the slope of the next elastic portion of the load-stroke curve, or it may be some input value. The first option is governed by the assumption that at any point in the stroking of a strut, the elastic slope of the load-stroke curve is determined by the elastic properties of the uncrushed cartridges acting in series. Thus, when a cartridge crushes, its elastic characteristic is no longer reflected in the strut load-stroke curve.

For a bending strut which may or may not contain honeycomb attenuation, the bending properties (modulus of elasticity for bending displacements and the moment of inertia) are assumed to remain constant.

4.1.2.3 <u>Coordinate Systems</u> - Two types of right-hand orthogonal coordinate systems employed in the Landing Gear Option are the Local Coordinate System and the Global Coordinate System. Each axial strut or bending strut of either gear has its set of local coordinates as illustrated in Figure 4-19. The Local Coordinate System origin for each strut is located at node "p," with the X_{l} axis aligned along the member axis. Positive X_{l} is on the side of node "p" towards node "q." The Y_{l} axis is perpendicular to X_{l} and is located in the "pqr" plane. Positive Y_{l} is on the side of X_{l} towards "r." The Z_{l} axis is then established using the right hand rule. Nodes p, q, and r, which establish each strut's local axis system, were preselected for all struts of the inverted tripod and cantilever gears.



(a) Axial Struts



FIGURE 4-19 STRUT DISPLACEMENTS IN LOCAL COORDINATE SYSTEM
Unless noted otherwise, in the following the terms "displacement" and "load" refer to incremental displacement or incremental load occurring during a step of the modified incremental stiffness method. The symbol " Δ " signifying an incremental quantity (see Equation (4-27)) is dropped for convenience.

Displacement notation of a typical axial strut capable of carrying axial load only is indicated in Figure 4-19(a) for the Local Coordinate System. Subscript 1 refers to displacement due to axial load while subscripts 2 and 3 are for displacements due to shear loads. Although axial struts can carry axial load only, displacements due to shear loads must be included in large displacement analysis as will be seen in the following sections. Cross-sectional area A, and modulus of elasticity E_A for axial displacements, must be specified for each axial strut relative to its Local Coordinate System. For an axial strut which contains honeycomb attenuation, this information is supplied in the form of the load-stroke curve shown in Figure 4-18.

Displacement notation of a typical bending strut capable of carrying axial load, shear in two directions, and bending in two directions is indicated in Figure 4-19(b) for the Local Coordinate System. Subscript 1 refers to a displacement due to axial load, subscripts 2 and 3 are for displacements due to shear loads, subscript 4 is for rotation due to torque, and subscripts 5 and 6 are for rotations due to moments. Although bending struts are not capable of carrying torsional loads, it is advantageous to include the torsional degrees of freedom since matrix transformation to a common coordinate system will be accomplished later and in this system all nodes will have six degrees of free-Cross-sectional area A, and modulus of elasticity ${\rm E}_{\rm A}$ for axial displacedom. ments, must be specified for each bending strut relative to its Local Coordinate System. For a bending strut which contains honeycomb attenuation, this information is supplied in the form of the load-stroke curve shown in Figure 4-18. Moment of inertia I, and the modulus of elasticity $E_{\rm B}$ for bending displacements, are specified for each bending strut relative to its Local Coordinate System. The moment of inertia is assumed to be constant about any axis normal to X ..

Two Global Coordinate Systems employed in the Landing Gear Option are the Lander Coordinate System and the Surface Coordinate System as shown in

Figure 4-20. The origins of these coordinate systems are assumed coincident. Coordinates of all nodes of the gear are input in the Lander Coordinate System.

The Surface Coordinate System must be chosen such that the axis normal to the surface, X_S in Figure 4-20, is pointed outward. Three Euler angles defined as yaw (ψ), pitch (θ), and roll (ϕ), are used to orient the Surface Coordinate System with respect to the Lander Coordinate System. That is, the orientation of the Surface Coordinate System can be found by rotations in the order ψ , θ , and ϕ about the rotated Z_S , Y_S and X_S surface axes, respectively. The transformation matrix $[T_G]$ relating vector components in the Surface Coordinate System (V X_S , V Y_S , V Z_S) to vector components in the Lander Coordinate System (V X_L , V Y_L , V Z_L) is defined by Equation (4-28).

$$\begin{cases} VX_{S} \\ VY_{S} \\ VZ_{S} \end{cases} = \begin{bmatrix} T_{G} \end{bmatrix} \begin{cases} VX_{L} \\ VY_{L} \\ VZ_{L} \end{cases}$$

$$(4-28)$$

Terms in the transformation matrix $[T_G]$ are defined in Equation (4-29).

$$[T_G] = \begin{bmatrix} D_{11} & D_{12} & D_{13} \\ D_{21} & D_{22} & D_{23} \\ D_{31} & D_{32} & D_{33} \end{bmatrix}$$

$$\begin{array}{c} D_{11} = \cos \theta & \cos \psi \\ D_{12} = \cos \theta & \sin \psi \\ D_{12} = -\sin \theta \\ D_{13} = \sin \phi & \sin \theta & \cos \psi - \cos \phi & \sin \psi \\ D_{21} = \sin \phi & \sin \theta & \sin \psi + \cos \phi & \cos \psi \\ D_{22} = \sin \phi & \sin \theta & \sin \psi + \cos \phi & \cos \psi \\ D_{23} = \cos \phi & \sin \theta & \cos \psi + \sin \phi & \sin \psi \\ D_{31} = \cos \phi & \sin \theta & \sin \psi - \sin \phi & \cos \psi \\ D_{32} = \cos \phi & \sin \theta & \sin \psi - \sin \phi & \cos \psi \\ D_{33} = \cos \phi & \cos \theta \end{array}$$

$$(4-29)$$

Coordinates of all node points of the gear are transformed to surface coordinate orientation using Equation (4-28).

The Surface Coordinate System is established as the common system for all elements of a gear so that element total and incremental loads and displacements may be related to a common frame of reference. The term "total",when applied to loads or displacements, refers to the sum of incremental quantities which have occurred during the incremental steps which have been taken to the



FIGURE 4-20 LANDING GEAR COORDINATE SYSTEMS

current point of interest. Displacement notation for the axial strut and bending strut in the Surface Coordinate System is shown in Figure 4-21. In this figure and in the following discussion, the Surface Coordinate System X_S , Y_S , Z_S will be referred to as \overline{X} , \overline{Y} , \overline{Z} . Footpad loading conditions (see Section 4.1.2.7) are specified in this system. Matrix assembly and solution of equations for each step of the modified incremental stiffness method (see Section 4.1.2.6) are also accomplished in the surface system. That is, external loads and displacements at all nodes of the gear, as well as the total energy absorbed by the gear, are determined in surface coordinate components.

4.1.2.4 <u>Incremental Stiffness Matrix</u> - The incremental stiffness matrix, employed in the modified incremental stiffness method, is generated for both an "axial" strut and "bending" strut in the Local Coordinate System. This stiffness matrix will be shown to be the sum of the standard small displacement stiffness and the large displacement geometric stiffness. The geometric stiffness matrix is derived in Reference (8) using basic nonlinear theory in conjunction with the incremental stiffness method.

For each strut of a landing gear, incremental loads and displacements in the Local Coordinate System are related in matrix form by

$$\Delta\{\mathbf{F}_{g}\} = [\mathbf{S}_{g}] \ \Delta\{\delta_{g}\} \tag{4-30}$$

In Equation (4-30), $[S_{\ell}]$ is the incremental stiffness matrix; $\Delta\{F_{\ell}\}$ is a column matrix of applied incremental loads at the nodes; and $\Delta\{\delta_{\ell}\}$ is the column matrix of incremental displacements at the nodes. In the following, for convenience, the Δ symbol will be dropped. Accordingly, unless otherwise noted, the terms displacement and load refer to incremental quantities.

A. Axial Struts - As defined in Section 4.1.2.2, all elements of the inverted tripod gear and elements 1 and 3 of the cantilever gear (Figure 4-17(b)) are "axial" struts with load-carrying capability in the axial direction only. Displacement notation in the Local Coordinate System for a typical axial strut was shown in Figure 4-19(a).

The incremental stiffness matrix $[S_{\ell}]$ of an axial strut (in local coordinates) is the sum of the small displacement stiffness matrix shown in Figure 4-22 and the geometric stiffness matrix shown in Figure 4-23. The small displacement stiffness matrix of an axial strut is equivalent to the stiffness



FIGURE 4-21 STRUT DISPLACEMENTS IN SURFACE COORDINATE SYSTEM



FIGURE 4-22 SMALL DISPLACEMENT STIFFNESS MATRIX OF AN AXIAL STRUT



FIGURE 4-23

GEOMETRIC STIFFNESS MATRIX OF AN AXIAL STRUT

matrix of the general bar element in the Center Body Option (see Figure 4-4) with $I_N = I_T = J = 0$ and only the translational degrees of freedom included. In Figure 4-22, A is the cross-sectional area; E_A the modulus of elasticity for axial displacements; and dpq the current length of the strut. For an axial strut composed of honeycomb attenuation, the terms (A E_A/dpq) in the small displacement stiffness matrix are replaced by the slope of the load-stroke curve (see Figure 4-18) corresponding to the current axial stroke of the strut.

The geometric stiffness matrix, shown in Figure 4-23, includes the effect of "total" internal axial loads on the equilibrium equations in the presence of large rotational displacements. In the stiffness terms, P_0 is the total internal axial load in the strut corresponding to the current total axial displacement (stroke) of the strut. Positive values for P_0 signify that the strut is in tension.

B. Bending Struts - As defined in Section 4.1.2.2, the main strut of the cantilever gear (elements 2 and 4 in Figure 4-17(b)) is composed of "bending" struts with bending capability as well as axial capability. Displacement notation in the Local Coordinate System for a typical bending strut was shown in Figure 4-19(b).

The incremental stiffness matrix $[S_{l}]$ of a bending strut (in local coordinates) is the sum of the small displacement stiffness matrix shown in Figure 4-24 and the geometric stiffness matrix shown in Figure 4-25. The small displacement stiffness matrix is equivalent to the stiffness matrix of the general bar element in the Center Body Option (see Figure 4-4) with J = 0 and $I_N = I_T = I$. In Figure 4-24, A is the cross-sectional area; E_A the modulus of elasticity for axial displacements; E_B the modulus of elasticity for bending displacements; I the moment of inertia about any axis normal to the strut; and dpq the current length of the strut. For a bending strut whose axial capability is defined by honeycomb attenuation, the terms (A E_A/dpq) in the small displacement stiffness matrix are replaced by the slope of the load-stroke curve (see Figure 4-18) corresponding to the current axial stroke of the strut.

The geometric stiffness matrix, shown in Figure 4-25, includes the effect of total internal axial loads on the equilibrium equations in the presence of large rotational displacements. In Figure 4-25, P_0 is the total internal axial load in the strut corresponding to the current total axial displacement (stroke). Positive values for P_0 signify tension.

_	^u 1p	^u 2p	^и з _р	^U 1q	^u 2q	^u 3q	^u 4р	^u 5p	^ս ճր	^u 4q	^u 5q	^u 6q
F1p	AE _A d _{pq}			-AE _A d _{pq}	-							
F2p		$\frac{12 \text{ E}_{\text{B}}\text{I}}{(\text{d}_{\text{pq}})^3}$			$\frac{-12 E_{B}}{(d_{pq})^{3}}$				$\frac{6 E_{B^{1}}}{(d_{pq})^{2}}$			$\frac{6 E_{B}}{(d_{pq})^{2}}$
F3p			$\frac{12 \text{ E}_{\text{B}}\text{I}}{(\text{d}_{\text{pq}})^3}$			$\frac{-12 \text{ E}_{\text{B}}\text{I}}{\left(\text{d}_{\text{pq}}\right)^3}$		$\frac{-6 \text{ E}_{\text{B}}\text{l}}{(d_{\text{pq}})^2}$			$\frac{-6 \text{ E}_{\text{B}}!}{(d_{\text{pq}})^2}$	
F1q	-A E _A	1 	- 	A E _A								
F2q		$\frac{-12 \text{ E}_{\text{B}}\text{I}}{(\text{d}_{\text{pq}})^3}$			12 E _B I (d _{pq}) ³			 	-6 E _B I (d _{pq}) ²			$\frac{-6 \text{ E}_{\text{B}}\text{I}}{(d_{\text{pq}})^2}$
F3q		 	2 E_BI(d_pq)^3	 		$\frac{12 \text{ E}_{\text{B}}\text{I}}{\left(\text{d}_{\text{pq}}\right)^3}$	 	<u>6 E_Bl</u> (d _{pq}) ²	 		6 E _B I (d _{pq}) ²	
M4p		 										
M5p		 1	$\frac{-6 \text{ E}_{\text{B}}\text{I}}{(d_{\text{pq}})^2}$			$\frac{6 E_{B}l}{\left(d_{pq}\right)^{2}}$		4 E _B I			2 E _B I d _{pq}	
^М бр		6 E _B I (d _{pq}) ²	 	 	$\frac{-6 \text{ E}_{\text{B}}\text{I}}{\left(\text{d}_{\text{pq}}\right)^2}$	 	 	 	4 E _B I d _{pq}	 		2 E _B I d _{pq}
M4q		 	 	 	 	 	 	; 	} 	 -	 	
M5q		 	$\frac{-6 \text{ E}_{\text{B}}}{(d_{\text{pq}})^2}$		 	$\frac{6 \text{ E}_{\text{B}}}{(d_{\text{pq}})^2}$	 	2 E _B I	 	 	4 E _B 1 d _{pq}	
M6q		$\frac{\overline{6 E_{B}}}{(d_{pq})^{2}}$	 		$\frac{-6 \text{ E}_{\text{B}}}{(d_{\text{pq}})^2}$		 	 	2 E _B I			4 E _B I

FIGURE 4-24 SMALL DISPLACEMENT STIFFNESS MATRIX OF A BENDING STRUT

-	"1 _p	^u 2 _p	^u 3p	^u 1q	^u 2 _q	u3q	^u 4 _p	^u 5p	^и 6 _р	u ₄ q	^u 5q	^u 6q
F1p		i 										
F _{2p}		6 P _o			-6 P ₀ 5 d _{pq}				P ₀ 10			P ₀ 10
F _{3p}			6 P _o 5 d _{pq}			-6 P ₀ 5 d _{pq}		_P ₀ 10			-P ₀ 10	
F1q			-									
F _{2q}		-6 P ₀ 5 d _{pq}			6 P _o 5 d _{pq}			; 	<u>-P₀</u> 10			$\frac{-P_{0}}{10}$
F3q			-6.P ₀ 5 d _{pq}	 		6 P _o 5 d _{pq}		<u>P</u> o 10		 	P ₀ 10	
M4p	1											
M _{5p}			P ₀		} 	P ₀ 10		2 P _o d _{pq}		► 	-P _o d _{pq} 	
M _{6p}		P _o 10	r 		-P ₀	 	 	r 	2 P _o d _{pq} 	 		-P _o d _{pq}
M ₄ q		 	• 	 			 1					
M5g		•	P ₀ <u>-</u> P ₀ <u>10</u>	+ 	; 	P ₀ <u>10</u>		-P _c d _{pq} 	 		2 P _o d _{pq} 	
M6q		P ₀ 10	 		-P ₀ <u>-</u> 10		 		-P _c d _{pq}			2 P _o d _{pq} 15

FIGURE 4-25 GEOMETRIC STIFFNESS MATRIX OF A BENDING STRUT

4.1.2.5 <u>Internal Loads</u> - For each iteration at each step of the modified incremental stiffness method, the total internal loads in each strut must be determined for the current displaced state of the strut. The total internal loads must be calculated so that load unbalances can be determined (see Section 4.1.2.1).

For axial struts, the total internal loads are simply the axial tension or compression forces which would have to be applied to the ends of the strut to extend or compress it to its present state. The extension or compression of a strut (the stroke) is the difference between the current length and the original length. Total axial force is the product of stroke and the small displacement stiffness term (A E_A/dpq) defined in Section 4.1.2.4. A positive stroke corresponds to extension of the strut and yields a positive (tensile) total axial force. For an axial strut composed of honeycomb attenuation, the total axial force is determined by the load-stroke curve (see Figure 4-18).

The total internal loads for bending struts consist of axial forces, shear forces, and bending moments. The axial forces are calculated in the same way as for axial struts. Calculation of bending loads (shear forces and bending moments) in the deflected main strut of the cantilever gear (elements 2 and 4 in Figure 4-17) is shown in Figure 4-26. Because the main strut is pinned at both ends, loads causing bending are shear loads $(F2p_{[2]} \text{ and } F2p_{[4]})$ applied at the ends in the plane of the deformed strut and directed along the Y_{g} axis. The Y_{g} direction is defined by a line passing through node 4 perpendicular to the line connecting nodes 2 and 5 and intersecting it at reference point RP. Since there is bending continuity at node 4, internal loads in elements 2 and 4 are shears $(F2q_{[2]} \text{ and } F2q_{[4]})$ and equal bending moments $(M6q_{[2]} \text{ and } M6q_{[4]})$. Coordinates of the floating reference point (RP) for elements 2 and 4 are also calculated in Figure 4-26.

4.1.2.6 <u>Matrix Assembly and Solution of Equations</u> - As explained in Section 4.1.2.3, the Surface Coordinate System is used for all elements of a gear to relate total and incremental loads and displacements to a common frame of reference. For each element of the gear, the incremental stiffness matrix is transformed from Local to Surface Coordinate System as described in Section 4.1.1.2.



FIGURE 4-26 INTERNAL BENDING LOADS IN MAIN STRUT OF CANTILEVER GEAR

Before matrix assembly can be accomplished, the incremental stiffness matrix in the Surface Coordinate System for all elements of the gear must be of the same order. All elements of the inverted tripod gear are axial struts and hence have incremental stiffness matrices of the same order. For the cantilever gear, the incremental stiffness matrices of the axial struts (elements 1 and 3 in Figure 4-17) must be expanded to the row and column format of the bending strut stiffness matrices of elements 2 and 4. This is accomplished by adding rows and columns of zeros corresponding to the six rotational degrees of freedom.

The total incremental stiffness matrix $[\overline{S}]$ (in the Surface Coordinate System) for the assembled gear is generated by systematically adding the transformed element stiffness matrices (according to the p and q nodes of the element) as shown in Figure 4-27. Node points on the inverted tripod gear are numbered consecutively from 1 to 4 (see Figure 4-17). The incremental stiffness matrix $[\overline{S}]$ of this gear is assembled with a row and column format corresponding to the three translational degrees of freedom in the Surface Coordinate System at each node in sequence. In this case, the size of the stiffness matrix is 12 by 12. For the cantilever gear, node points are numbered consecutively from 1 to 5. The incremental stiffness matrix $[\overline{S}]$ of this gear is assembled with a row and column format corresponding to the three translational followed by the three rotational degrees of freedom in the Surface Coordinate System at each node in sequence. In this case, the size are numbered consecutively from 1 to 5. The incremental stiffness matrix $[\overline{S}]$ of this gear is assembled with a row and column format corresponding to the three translational followed by the three rotational degrees of freedom in the Surface Coordinate System at each node in sequence. In this case, the size of the stiffness matrix is 30 by 30.

The assembled incremental stiffness matrix [S] is related to the column matrices of incremental surface coordinate loads and displacements at each node by Equation (4-31).

$$\Delta\{\overline{\mathbf{F}}\} = [\overline{\mathbf{S}}] \ \Delta\{\overline{\mathbf{\delta}}\} \tag{4-31}$$

For the fixed idealization of the inverted tripod gear (see Figure 4-17), nodes 1, 2, and 3 are pinned and thus have zero translational displacements. Since only translational displacements are included in the stiffness formulation for this gear, the degrees of freedom of interest are the translations of node 4. Therefore, for the inverted tripod, Equation (4-31) reduces to

$$\Delta\{\overline{\mathbf{F}'}\} = [\overline{\mathbf{S}'}] \ \Delta\{\overline{\delta'}\}$$
(4-32)



FIGURE 4-27 ASSEMBLED INCREMENTAL STIFFNESS MATRIX FORMAT

In this equation, $\Delta\{\overline{F'}\}$ is a column matrix of three incremental force components at node 4; $\Delta\{\overline{\delta'}\}$ a column matrix of three incremental translations at node 4; and $[\overline{S'}]$ a 3 by 3 matrix of stiffness terms, extracted from $[\overline{S}]$, corresponding to the degrees of freedom retained at node 4.

For the fixed idealization of the cantilever gear (see Figure 4-17), nodes 1, 2, and 3 are pinned and hence have zero translational displacements. Accordingly, equations associated with the 3 translational degrees of freedom at each of these nodes are not of interest. Elements 1 and 3 have zero bending capability. Therefore, the six equations associated with the rotational degrees of freedom at nodes 1 and 3 are not of interest. The remaining degrees of freedom for a cantilever gear are three rotations at node 2, three translations and three rotations at node 4, and three translations and three rotations at node 5. Thus, for the cantilever gear Equation (4-31) reduces to

$$\Delta\{\overline{\mathbf{F}}''\} = [\overline{\mathbf{S}}''] \quad \Delta\{\overline{\delta}''\} \quad (4-33)$$

In Equation (4-33), $\Delta\{\overline{F}''\}$ is a column matrix containing three incremental moment components at node 2, three incremental force components and three incremental moment components at node 4, and three incremental force components and three incremental moment components at node 5; $\Delta\{\overline{\delta}''\}$ a column matrix of the fifteen incremental displacements described above; and $[\overline{S}'']$ a 15 by 15 matrix of stiffness terms, extracted from $[\overline{S}]$, corresponding to the 15 degrees of freedom retained.

Equation (4-32) or (4-33) is solved repeatedly at each step of the modified incremental stiffness method (see Section 4.1.2.1). A least squares solution technique is programmed for solution of these equations. The independent variables for each of these solutions are discussed in Section 4.1.2.7. After the incremental nodal displacements have been determined, the updated nodal locations can be found as the sum of the previous nodal coordinates and the incremental nodal displacements. For the updated nodal locations, the total internal loads can then be determined as described in Section 4.1.2.5.

4.1.2.7 <u>Applied Displacement Options</u> - For each step of the modified incremental stiffness method, the governing matrix equation, Equation (4-32) for an inverted tripod gear or Equation (4-33) for a cantilever gear, is solved as outlined below. Both of these equations are analogous to Equation

(4-27) and the solution technique is the same as discussed in Section 4.1.2.1.

- (i) Equation (4-32) (or Equation (4-33)) is solved <u>once</u> for the case where Δ {δ' } (or Δ {δ" }), the total applied displacement vector divided by the number of incremental steps, is the (known) independent variable.
- (ii) Equation (4-32) (or Equation (4-33)) is solved <u>repeatedly</u> for the case where $\Delta\{\overline{F'}\}$ (or $\Delta\{\overline{F''}\}$), the load unbalance vector, is the (known) independent variable. The load unbalance vector is calculated as the difference between applied external nodal loads and total internal nodal loads. Calculation of internal loads is described in Section 4.1.2.5.

Two options are available for specifying the total applied displacement vector discussed in (i) above. In the first option, total displacement of the footpad joint (node 4 for the inverted tripod gear and node 5 for cantilever gear) normal to a landing surface, and the coefficient of friction of the surface are specified. For example, suppose it is desired to investigate the displaced state of the gear at each of N steps for an applied normal displacement value of D units. For this case, the program determines N equilibrium configurations for the gear. For each of these configurations, the footpad joint is located in one of N different planes spaced D/N units apart. Whether the footpad slides on the landing surface as the footpad joint is displaced from the (n-1)st plane to the nth plane (n = 1, ..., N) depends on the friction coefficient. As explained in Section 4.1.2.3, the landing surface can be arbitrarily oriented with respect to the Lander Coordinate System.

In the second option available for specifying the applied displacement vector, three components of footpad joint displacement are specified in the Surface Coordinate System $(\overline{X}, \overline{Y}, \overline{Z})$. For example, if it is desired to investigate the displaced state of the gear at each of N steps when the applied footpad joint displacement components are \overline{X}_f , \overline{Y}_f , and \overline{Z}_f , the program determines N equilibrium configurations for the gear. In each of these configurations, the footpad joint is located at one of the positions $(\overline{X}_0 + n \ \overline{X}_f/N, \ \overline{Y}_0 + n \ \overline{Y}_f/N, \ \overline{Z}_0 + n \ \overline{Z}_f/N)$ where n = 1, ..., N. The original position of the footpad joint (see Figure 4-17) is assumed to be $(\overline{X}_0, \ \overline{Y}_0, \ \overline{Z}_0)$.

4.1.2.8 <u>Energy Absorption</u> - For each step of the landing gear displacement history, energy absorbed by the gear during the step is calculated in both Surface and Lander Coordinate System components.

Although an element representing the footpad is not included in the landing gear idealization, energy associated with crushing of footpad honeycomb can be included in the analysis. Footpad honeycomb attenuation is idealized with up to three levels of crushing in compression as shown in Figure 4-28.



FPCRLD – FOOTPAD HONEYCOMB CRUSH LOAD FPCRST – FOOTPAD HONEYCOMB CRUSH STROKE

FIGURE 4-28 ASSUMED LOAD-STROKE CURVE FOR FOOTPAD HONEYCOMB

Each surface coordinate energy component absorbed at the footpad joint during a step is the sum of strut crush energy and footpad crush energy. Strut crush energy in a given surface coordinate direction is calculated as the product of the average force component (in that direction) over the step and the incremental displacement component (in the same direction) which occurred during the step. Figure 4-29 presents an example calculation of strut crush energy component absorbed in the \overline{X} surface coordinate direction during the nth incremental step. For a gear which has footpad attenuation with up to three levels of crush force, footpad crush energy may be added into the surface coordinate energy component in the direction normal to the landing surface. For a given incremental step, footpad crush energy is added into this energy component if the force component in the normal direction (see Figure 4-20) at the footpad joint exceeds one of the footpad honeycomb crush forces (Figure 4-28) for the first time. When this occurs, the associated level of honeycomb crush is assumed to have bottomed out during the step. Footpad crush energy component associated with this crushing is the product of crush load (FPCRLD) and the crush stroke (FPCRST). For example, the surface coordinate direction normal to the landing surface is X (Figure 4-20). If during incremental step n, the force in the \overline{X} direction at the footpad joint exceeds the first crush force FPCRLD(1) in Figure 4-28, and this has not occurred for any previous step, the energy term FPCRLD(1) times FPCRST(1) will be added to the X strut crush energy component for this step.

Components of energy absorption during a given incremental step are obtained similarly in Lander Coordinate System components. In both coordinate systems, total energy absorption components are determined after n incremental steps as the sum of the energy components obtained in each of the n steps taken.

4.2 <u>Program Description</u> - Organization of the Structural Analysis Program is described in this section. Included are flow diagrams and discussions of subroutine functions for both the Center Body Option and Landing Gear Option.

4.2.1 <u>Subroutines</u> - The Structural Analysis Program is divided into nine Fortran OVERLAY segments. OVERLAY organization and description of the function of each subroutine are shown in Figure 4-30. This organization is required to stay within the allotted core storage requirements. The main OVERLAY (0, 0)

acts as the principal executive routine and determines the order in which primary OVERLAYs (1, 0) through (5, 0) are called. OVERLAY (1, 0) processes the data set header cards and handles the data sets and initialization relating to the Center Body Option. OVERLAY (2, 0) generates the total stiffness matrix for the Center Body Option. OVERLAY (3, 0) determines the displacements and rotations for the Center Body Option and prints the results. The modal analysis portion of the Center Body Option is handled in OVERLAY (4, 0). OVERLAY (5, 0) contains the routines for the Landing Gear Option and has three supporting secondary overlays; OVERLAY (5, 1) which reads the landing gear data cases, OVERLAY (5, 2) which contains the executive routine for the inverted tripod gear, and OVERLAY (5, 3) which contains the executive routine for the cantilever gear.



FOR STEP n:

 $(E_{\overline{X}})_n = AREA UNDER CURVE BETWEEN POINTS n-1 AND n$

$$=\frac{(F_{n}+F_{n-1})}{2}(u_{n}-u_{n-1})$$

FIGURE 4-29 EXAMPLE CALCULATION OF STRUT CRUSH ENERGY

OVERLAY	SUBROUTINE	FUNCTION
(0, 0)	MAIN ERPNT1 ERPNT2 WRSTRK	PROGRAM EXECUTIVE ROUTINE ERROR PROCESSING ROUTINE FOR ALL OVERLAYS SECOND ENTRY POINT TO ERPNTI SPARSE MATRIX PRINT ROUTINE
(1,0)	INITAL DATSET RDDATA SHRPAN CROSSX	EXECUTIVE ROUTINE HEADER CARD PROCESSOR AND CENTER BODY OPTION DATA CASE READER CENTER BODY OPTION DATA CASE PROCESSOR CONVERT SHEAR PANELS TO EQUIVALENT DIAGONAL BAR ELEMENTS CALCULATE THE CROSS PRODUCT OR DOT PRODUCT OF TWO VECTORS
(2, 0)	STIFF SETSTF STFTRN STORSM TRASMK WRBDAT BRSTRA WRSTDK	EXECUTIVE ROUTINE INITIALIZE STRUCTURAL STIFFNESS MATRIX STORAGE ARRAY IN A SPARSE BLOCKED FORMAT CALCULATE ELEMENT STIFFNESS AND TRANSFORMATION MATRICES CONSTRUCT ASSEMBLED STIFFNESS MATRIX TRANSFORM ELEMENT STIFFNESS MATRICES FROM LOCAL TO GLOBAL COORDINATE SYSTEM PRINT ELEMENT STIFFNESS AND TRANSFORMATION MATRICES SECOND ENTRY POINT TO WRBDAT, PRINTS TRANSFORMED ELEMENT STIFFNESS MATRICES WRITE ELEMENT STIFFNESS AND TRANSFORMATION MATRICES ON FILE 2
(3, 0)	FINIAL PANDTK SOLVE PNTFMV FMBARS	EXECUTIVE ROUTINE PRINT AND STORE ON FILE 9 THE ASSEMBLED STIFFNESS MATRIX SET-UP FOR SOLUTION DISPLACEMENTS AND ROTATIONS DETERMINED USING ITERATIVE METHOD. GLOBAL FORCES AND MOMENTS ON JOINTS CALCULATED PRINT DISPLACEMENTS, ROTATIONS, AND GLOBAL FORCES AND MOMENTS ELEMENT BAR FORCES AND MOMENTS CALCULATED USING RESULTS OF SOLVE
(4, 0)	NLMDAL PUNPACK REDUCE STFMAS FNALEV EEPNT TRIDIA EIGVAL EIGVEC PRNT1 PRNT2 P1A721	EXECUTIVE ROUTINE DATA HANDLING ROUTINE FOR SUBROUTINE REDUCE PRINT ASSEMBLED STIFFNESS MATRIX AND REDUCE TO DESIRED SIZE (REDUCED MATRIX 102 x 102 OR LESS) CREATES MATRIX SYSTEM, USING INPUT DIAGONAL MASS MATRIX AND REDUCED STIFFNESS MATRIX, FOR WHICH EIGENVALUES CAN BE FOUND EIGENVECTORS OF REDUCED SYSTEM TRANSFORMED INTO EIGENVECTORS OF FULL SYSTEM WRITE EIGENVALUES AND EIGENVECTORS MATRIX SYSTEM IS TRI-DIAGONALIZED USING HOUSEHOLDER'S METHOD SPECIFIED NUMBER OF SMALLEST EIGENVALUES OF MATRIX SYSTEM ARE CALCULATED USING ORTEGA'S METHOD (PLUS SIX RIGID BODY NATURAL FREQUENCIES OF ZERO) CALCULATES EIGENVECTORS ASSOCIATED WITH LOWEST EIGENVALUES USING WILKINSON'S METHOD MODAL DATA OUTPUT ROUTINE FOR REDUCED SYSTEM SECOND ENTRY POINT OF PRNT1 TO OUTPUT MODAL DATA OF FULL SYSTEM PRINT SUPPORT ROUTINE FOR PRNT1 AND PRNT2
(5, 0)	GEAREX OUTPUT ENERGY TRNFSM TRALMG STFMIT TRANSM DMFSS DMLSS STRUT	EXECUTIVE ROUTINE LANDING GEAR OPTION OUTPUT COMPUTE AND SUM ENERGY QUANTITIES TRANSFORM ELEMENT STIFFNESS MATRIX TRANSFORM FORCE-MOMENTS FROM LOCAL TO GLOBAL COORDINATES COMPUTE NON-BENDING ELEMENT STIFFNESS MATRIX COMPUTE ELEMENT LENGTH AND TRANSFORMATION MATRIX DETERMINE THE RANK OF A SYMMETRIC POSITIVE SEMI-DEFINITE MATRIX CALCULATE THE LEAST SQUARES SOLUTION OF A SYSTEM OF SIMULTANEOUS EQUATIONS WITH SYMMETRIC POSITIVE SEMI-DEFINITE COEFFICIENT MATRIX WHOSE RANK IS KNOWN COMPUTE ELEMENT AXIAL LOAD AND STIFFNESS
(5, 1)	INPUT Geom	READ LANDING GEAR OPTION DATA CASES COMPUTE LANDER COORDINATE TO SURFACE COORDINATE TRANSFORMATION MATRIX FROM EULER ANGLES
(5, 2)	INVTRP	EXECUTE INVERTED TRIPOD DATA COMMANDS
(5, 3)	CANTIL BNDLDS STFMCF	EXECUTE CANTILEVER DATA COMMANDS COMPUTE BENDING LOADS FOR CANTILEVER GEAR COMPUTE BENDING ELEMENT STIFFNESS MATRIX

FIGURE 4-30 STRUCTURAL ANALYSIS PROGRAM SUBROUTINES

4.2.2 <u>Flow Diagram</u> - The Structural Analysis Program is shown schematically in Figure 4-31. While detailed steps, such as those required in the iteration loop of the Center Body Option, are not presented, the basic sequence of events is shown for both options of the program. A listing of the program is given in Appendix H.

4.3 <u>Program Operation</u> - Information necessary to operate both options of the Structural Analysis Program is contained in this section.

4.3.1 <u>Center Body Option</u> - This section includes definition of input requirements and format, and output interpretation for the Center Body Option of the program. Examples of input and output data for a typical problem are contained in Appendices A and B.

4.3.1.1 <u>Input Data</u> - For each Center Body Option data set, a data case header card and information describing geometry, supports, applied loads and displacements, member properties, and indicators needed to control program operation are required as input data. Multiple data cases may be run by stacking data sets. Each data set must consist of the following:

Header Card

Data Cards

Data Option Indicator (NAMELIST) Card(s)

A. <u>Header Card</u> - The first card of a data set must be a header card. This card indicates that the Center Body Option will be employed and must contain the characters STRUCTURE in columns 1 through 9 as indicated in Figure 4-32.

B. <u>Data Cards</u> - The input format for data cards employs a system of code numbers, located in column 1, to identify the type of data being read. The nine code numbers and corresponding type of information being input are as follows:

- (0) Comment cards
- (1) Joint information cards
- (2) Reference point information cards
- (3) Force and moment limits or specified displacement and rotation cards
- (4) Specified force vector cards
- (5) Specified moment vector cards
- (6) Bar information cards
- (7) Rectangular shear web information cards
- (8) Data terminator card





FIGURE 4-31 FLOW DIAGRAM STRUCTURAL ANALYSIS PROGRAM (Continued)

B.0 <u>Comment Cards</u> - As many comment cards as desired may be input by leaving columns 1 through 6 blank and entering the comments in columns 7 through 80. These comment cards may be located anywhere within the data cards as long as they precede the data terminator.

B.1 <u>Joint Information Cards</u> - Joint information is input on data cards in which a 1 is placed in column 1. Location of joint data in the fields of the card is indicated in Figure 4-32. Joints must be numbered sequentially (right-justified in columns 2 through 5) from 1 through the total number of joints (maximum of 74). The global coordinates, \overline{X} , \overline{Y} , and \overline{Z} , may be any right hand orthogonal coordinate system (see Figure 4-2). If the structure is planar, it will be simpler to choose a Global Coordinate System such that two of the axes lie in the plane of the structure. Advantage should be taken of any structural symmetry which may exist in selecting global axes.

Six data fields located between columns 36 and 53 are used to indicate various constraints at a particular joint. These constraints may be combinations of specified displacements (global deflections and/or rotations) and limits on reaction loads (global forces and/or moments) at a joint. Leaving these columns blank causes the program to assume that there are no displacement constraints at a joint; however, there are applied loads at the joint. These applied loads are discussed in the next paragraph. Constraints are defined by inserting nonzero integer identifying indicators (right-justified) in the fields of this region. The absolute value of these identifying indicators corresponds to the limit number on the code 3 data card defining the magnitudes of the particular constraint. When positive identifying indicators are used, the program assumes that upper and lower plastic limits are placed on the appropriate loads at a support and that the corresponding displacement or rotation is zero as defined on the related code 3 data card. For negative identifying indicators, the program assumes that no restraints are placed on the loads, but the displacements are specified on the related code 3 data card. The same code 3 data card may be used to define identical constraints at a number of joints. The maximum number of plastic loads and specified displacements is 88.

Applied forces are indicated by placing a separate set of integer identifying indicators in columns 54 through 57 (right-justified). The program looks for the three global components of force on cards with code number 4 in



column 1, and the corresponding identifying indicator in columns 2 through 5. Identifying force indicators must be sequential with a maximum of 74. Applied moments are numbered separately, with this set of identifying indicators being placed in columns 58 through 61, and the three corresponding global moment components being located on cards with a 5 in column 1. A total of 74 identifying moment indicators may be used. If either of these identifying indicators is zero (or blank), the appropriate applied load is zero and no corresponding code 4 or 5 data card is required. If displacements and/or rotations have already been restrained in specific directions, forces and/or moments may not be applied in these same directions.

Since free-free modes are obtained with the modal analysis routine, no constraints or applied loads should be specified when employing this program option. Thus, for modal analysis data cases, columns 36 through 61 should be left blank on all code 1 data cards.

B.2 <u>Reference Point Information Cards</u> - A code number 2 in column 1 indicates the data represents global coordinates and identifying joint number of a reference point. Reference points are provided to permit orientation of bar local coordinate (bending) axes, where structural joints will not suffice. Numbering of these reference points begins with one greater than the number of joints and is sequential through the total number of reference points. For example if there are 36 structural joints, the first reference point must be 37.

B.3 <u>Force and Moment Limit Cards</u> - A 3 in column 1 indicates data on the card is either a known displacement or rotation (columns 26 through 35), or upper and lower plastic force or moment constraints (columns 6 through 15 for an upper bound and 16 through 25 for a lower bound). The integer identifying number (limit number) in columns 2 through 5 corresponds to an identifying indicator or indicators on code 1 data cards, previously discussed. These limit numbers must be sequential in order. The sign of the identifying indicator on the code 1 data card determines what data are being read: A plastic force constraint and zero displacement (positive sign) or known displacement (negative sign). The known displacement may be nonzero or zero, simulating a conventional support. If the force is to be limited in the positive global direction, use the upper limit. A lower limit indicates a constraint on the

FIGURE 4-33 INPUT DATA FORMAT - CENTER BODY OPTION

(Data Codes 3 Thru 6)

force in the negative global direction.

For modal analysis data cases, code 3 data cards should not be input.

B.4 <u>Specified Force Vector Cards</u> - Applied forces at a joint are input as three or less global force components on cards with a 4 in column 1. The identifying number (force number), right-justified in columns 2 through 5, corresponds to an identifying indicator on code 1 data cards thereby defining the joint or joints where forces will be applied.

For modal analysis data cases, code 4 data cards should not be input.

B.5 <u>Specified Moment Vector Cards</u> - Applied moments at a joint are input as three or less global moment components on cards with a 5 in column 1. The identifying number (moment number), right-justified in columns 2 through 5, corresponds to an identifying indicator on code 1 data cards to define the joint or joints where the moments will be applied.

Code 5 data cards should not be input for modal analysis data cases.

B.6 <u>Bar Information Cards</u> - A card with a 6 in column 1 indicates bar data are being input. Bar number is entered in columns 2 through 5, rightjustified, with numbering being sequential from one through the total number of bars (130 maximum). Bar origin (point "p"), end (point "q"), and direction of the Y_{l} bending axis (point "r"), locating the bar Local Coordinate System (Figure 4-1), are determined by specifying joint numbers in the appropriate locations in columns 6 through 14 (all right-justified). A reference point may be used for point "r," but points "p" and "q" must be nodes (joints) on the structure. Point "r" must not be on the line "pq" or its extension.

Bar cross-sectional area perpendicular to the "pq" (X_{ℓ}) axis, A; moment of inertia about the Z_{ℓ} axis, I_N ; moment of inertia about the Y_{ℓ} axis, I_T ; and torsional constant, J, are entered in the appropriate columns as indicated in Figure 4-33. Members pinned at both ends for bending about the Z_{ℓ} axis can be simulated by making the moment of inertia I equal to zero. Similarly making I_T equal to zero simulates a member pinned at both ends for bending about the Y_{ℓ} axis. If A is set equal to zero, no axial load will be carried by the member, and setting J equal to zero prevents the member from carrying torsion. Any combination of these may be utilized. Modulus of elasticity, E, and shear modulus, G, for the bar material are entered in columns 53 through 61 and 62 through 70, respectively. If E-format is used it must be right-justified.

Caution must be exercised when using pinned members or members with A, I_T , I_N , or J equal to zero, because a joint may be left with no load carrying capability in one of the three global directions or with no rotational restraint about one of the three global axes. An example of this would be two bars in the same plane pinned together. An artificial restraint (support) must be placed on the joint in the global direction in which there is no load carrying capability, even though there will be no force induced in this support direction. This instruction is input on code 1 data cards. The program will automatically reject any problem in which load carrying capability (translational or rotational) does not exist at any joint.

The program has the ability to account for the effect of bar shear strain on bending deflections. If a zero (blank) is placed in column 15, shear strain is not accounted for. A 1 in column 15 causes the program to read the values of shear form factors, K_N and K_T , and account for the effect of shear strain. The factor K_N is the total cross-sectional area divided by the area effective in carrying shear in the Y_{ℓ} direction, while K_T is the total area divided by the area effective in carrying shear in the Z_{ℓ} direction (see Figure 4-6).

The program can also idealize a bar pinned at one end and rigidly attached at the other. To exercise this option a zero must be placed in column 15. If the rotational release is to be for bending about the Y_{ℓ} axis (I_T) , a 1 or 2 is placed in column 16. A 1 indicates the "p" end is pinned while a 2 indicates the "q" end is pinned. The same number code (1 - "p" end pinned, 2 -"q" end pinned) is used in column 17 if the rotational release is for bending about the Z_{ℓ} axis (I_N) . Values inserted for K_N and K_T for these cases will be automatically utilized.

B.7 <u>Rectangular Shear Web Information Cards</u> - Shear web information is input on data cards in which a 7 is placed in column 1. Location of shear web data in the fields of the card is illustrated in Figure 4-34. Shear webs must be numbered sequentially from 1 through the total number of shear webs (maximum of 30). The integer representing the shear web number is right-justified in columns 2 through 5. (Card Codes 7 and 8 and Data Option Indicator Card(s))





FIGURE 4-34 INPUT DATA FORMAT - CENTER BODY OPTION

Four data fields located between columns 6 and 17 are used to specify the joints A, B, C, and D, numbered counterclockwise (see Figure 4-7), which represent the corners of the rectangular web. These integers must be right-justified in their respective fields. Care must be taken to insure that the joints A, B, C, and D define a plane and form a rectangle. The program will reject any problem in which this is not true.

Three data fields located between columns 18 and 47 are used to specify the material properties and thickness of the web. The Poisson's ratio, v, modulus of elasticity, E, and thickness, t, of the web are input in these fields as indicated in Figure 4-34. These values must be right-justified in their fields if E-Format is used.

B.8 <u>Data Terminator Card</u> - After all data needed on code 1 through 7 data cards and the desired comment cards are input, a data terminator card is required to signify the end of the data. This card must contain an 8 in column 1 and may contain any desired comments in columns 7 through 80.

C. <u>Data Option Indicator (NAMELIST) Card(s)</u> - Following the previously described input data, a NAMELIST system of input indicated by \$INDATA is used to define option indicator cards for selection of output options, iteration method, modal analysis routine, program termination, and tape options. Figure 4-34 presents the card format to be used. \$INDATA is entered in columns 2 through 8 of the first data option card. Columns 10 through 68 on the first and any succeeding cards are used to specify optional data indicator values. \$END must be entered in columns 69 through 72 of the last data option card.

Data option indicators and their nominal and optional values are defined in Figure 4-35. If nominal values are acceptable for all NAMELIST items, \$INDATA is entered in columns 2 through 8, \$END is entered in columns 69 through 72, and the rest of the card left blank. By listing any of the option data indicators, such as INDNMA = 1, the nominal control is overridden, as in this case where normal mode analysis would be utilized. A data option indicator need be listed only if a value other than the nominal value is desired.

	1			and the state of the				(The second	1
CONTROL FUNCTION	DO NOT WRITE BAR LOCAL STIFFNESS AND TRANSFORMATION MATRICES.	DO NOTWRITE BAR TRANSFORMED STIFFNESS MATRICES	WILL NOT (INPUT AN INITIAL SOLUTION OF DISPLACEMENTS AND (INPUT AN INITIAL SOLVEC", WILL (INPUT AN INITIAL SOLVEC",	THE ITERATIVE SOLUTION OF THE SIMULTANEOUS EQUATIONS WILL BE STOPPED AFTER (''INDITR'' + 1) ITERATIONS	THE ABSOLUTE VALUE OF "ERRTOL" IS USED AS THE ERROR TOLERANCE TO TERMINATE THE ITERATIVE SOLUTION FOR DISPLACEMENTS AND ROTATIONS.	THE INITIAL SOLUTION OF DISPLACEMENTS AND ROTATIONS IS ZERO. THE A i's (i = 1, 6X(NO. JOINTS)) ARE INPUT VALUES (SEPARATED BY COMMAS) DEFINING AN INITIAL SOLUTION OF DISPLACEMENTS AND ROTATIONS. (SOLVEC IS DIMENSIONED 444) ie. $A_{(1)} = \overline{X}$ DISPLACEMENT OF JOINT 1	$A_{(4)} = \overline{X}$ rotation of Joint 1 $A_{(i-1)X6+1} = \overline{X}$ displacement of Joint j	A ((j-1)X6+ 4) = \overline{X} rotation of joint j.	ELIST (SINDATA) DATA OPTION INDICATORS
OPTIONAL VALUE	0	o	-	K (ANY INTEGER)	A (ANY REAL NUMBER)	A (i)			IIRF 4-35 NAMF
NOMINAL VALUE	-	-	o	3X(NO. OF ROWS IN STRUCTURAL STIFFNESS MATRIX)	1000.	0. (i)			SIA
INDICATOR	INDSFL	INDSFG	INDISL	INDITR	ERRTOL	SOLVEC			

INDICATOR	NOMINAL VALUE	OPTIONAL VALUE	CONTROL FUNCTION
RELAXF	•		EMPLOY GAUSS-SIEDEL SOLUTION METHOD
		A (1.0 ≤ A ≤ 2.0)	THE RELAXATION FACTOR EMPLOYED IF THE OVERRELAXA- TION METHOD OF SOLUTION IS USED. RECOMMENDED OPTIONAL VALUE = 1.2
INDRLX	2		EMPLOY CONJUGATE GRADIENT SOLUTION TECHNIQUE.
		-	EMPLOY GAUSS-SIEDEL METHOD OF SOLUTION WITH NOMINAL RELAXATION FACTOR OF 1., OR USE OVERRELAXATION SOLUTION METHOD WITH OPTIONAL RELAXATION FACTOR DEFINED IN "RELAXF."
		0	EMPLOY GAUSS-SIEDEL SOLUTION METHOD WITH AITKEN'S Δ^{2} improvements.
INDRKT	0		DO NOT
			TAPE 9.
INDWKT	0		DO NOT(WRITE STRUCTURAL STIFFNESS MATRIX AND BAR LOCAL
		-	DO (TAPE 9.
INDPLS	0		DO NOT
		-	DO FORCES/MOMENTS.
MINRST	v	K (ANY INTEGER)	LESS THAN "MINRST" RESTRAINTS WILL CAUSE PROGRAM TERMINATION DURING SOLUTION OF SIMULTANEOUS EQUATIONS.
AMNONI	0		D0 N0T
	,		DORUN NORMAL MODE ANALYSIS.

INDICATOR	NOMINAL VALUE	OPTIONAL VALUE	CONTROL FUNCTION
IRWKP	0 ⁽ⁱ⁾	K(i) (INTEGERS)	ROW NUMBERS IN ASCENDING ORDER (SEPARATED BY COMMA'S) TO BE KEPT IN REDUCED STIFFNESS MATRIX WHEN RUNNING NORMAL MODE ANALYSIS (i=1, $2 \dots n$ WHERE $n \le 102$;"'IRWKP" IS DIMENSIONED 103).
AMASS	0 ^{.(i)}	₹	DIAGONAL MASS MATRIX ELEMENTS (SEPARATED BY COMMAS) IN ASCENDING ORDER, INPUT WHEN RUNNING NORMAL MODE ANALYSIS ($i = 1, 2, \ldots, n$ where $n \leq 102$, "AMASS" IS DIMENSIONED 102).
IREDTO	ORDER OF STIFFNESS MATRIX	K (K ≤ 102)	THE REQUIRED ORDER OF REDUCED STIFFNESS MATRIX WHEN RUNNING NORMAL MODE ANALYSIS.
WNMQNI	0	-	DO NOT WRITE NORMAL MODE DATA ON TAPE FOR USE IN TASK ORDER TWO LANDING LOADS AND MOTIONS PROGRAM OR THE CENTER DO BODY LOADS PROGRAM.
ТМАХ	6666	A (ANY REAL NUMBER)	NUMBER OF SECONDS AFTER WHICH MACHINE WILL PRINT PRESENT SOLUTION AND STOP, IF CONVERGENCE TO PRESCRIBED TOLERANCE HAS NOT BEEN REACHED (RECOMMENDED OPTIONAL VALUE = .9X(CP TIME)).
ISFDIM	12,904	K (INTEGER)	THE NUMBER OF WORDS OF STORAGE RESERVED TO STORE THE TOTAL STIFFNESS MATRIX. THE LENGTH OF LABELED COMMON, "CMAIN", MUST BE RE. DEFINED TO ALLOW STORAGE OF A STIFFNESS MATRIX WHICH
INDWTS	0		REQUIRES "K" WORDS OF STORAGE SPACE, WHERE K > 12904. DO NOT PRINT THE NONZERO ELEMENTS OF THE TOTAL STIFFNESS MATRIX.
NEIGVL	ъ	1 M (5 ≤ M ≤ IŘEDTO-6)	DO

The ERRTOL control is important, since it affects computer time. The value specified, multiplied by 100, is the maximum acceptable percent error in the solution. The error for the Gauss-Siedel method, the Gauss-Siedel method with Aitken's Delta Squared improvements, and the overrelaxation method is defined as the largest relative difference between consecutive displacement and rotation solutions. This relative difference is calculated by taking the difference between given solution values for two successive iterations and dividing by the last value of the solution.

The error for the Conjugate Gradient Method is defined as the largest term found when the residual vector (see Reference (5)) is divided by the largest element in the force/moment vector. The program will run until it achieves the accuracy prescribed by ERRTOL or exceeds the maximum number of iterations (INDITR).

An initial set of nodal deflections may be input using the indicator SOLVEC. This option allows the program to start with these values for deflections and achieve convergence more rapidly. This option is also useful if minor changes are made in a structure after an initial solution has been obtained.

The maximum number of iterations allowed (INDITR), nominally three times the number of rows in the stiffness matrix, should be selected based on the size and sparsity of the stiffness matrix and desired length of computer run time. The nominal iteration method used is the Conjugate Gradient Method. If INDRLX is set equal to 0 the Gauss-Siedel method with Aitken's Delta Squared Improvements is employed. If INDRLX is set equal to 1, the Overrelaxation Method is employed and a value of RELAXF (the relaxation factor) between 1.0 and 2.0 should then be input. A nominal relaxation factor of 1.0 is set in the program, which is equivalent to a standard Gauss-Siedel solution (without Aitken's Delta Squared Improvements). For employing the Overrelaxation Method, a RELAXF equal to 1.2 has been found to result in minimum machine time and rapid convergence for a number of typical problems. Experience dictates that for highly redundant space frames, such as the legged lander, the nominal solution technique (the Conjugate Gradient Method) is the best suited of the above methods and is recommended.

The MINRST option is selected based on the minimum number of restraints (supports) which must exist for stability (nominally six). If, at any time, less than this number of supports exist, the program will automatically terminate with an error message.

If it is desired to perform a normal mode analysis, INDNMA is set equal to 1. If reduction is to be employed, the required order of the reduced stiffness matrix is input in IREDTO. The row numbers to be retained in the reduced stiffness matrix must be listed in ascending order in IRWKP if reduction is employed. Values for the diagonal mass matrix terms associated with the degrees of freedom retained in the reduced stiffness matrix, are input in AMASS. If all six degrees of freedom at a particular joint are retained, the first three numbers in AMASS would represent the mass associated with this joint. The next three numbers would represent the mass moment of inertia associated with this joint relative to the directions of the global \overline{X} , \overline{Y} and \overline{Z} axes. If degrees of freedom at a joint are removed in the reduced stiffness matrix, the corresponding mass or inertia items should not be input in AMASS. If reduction is not required, no data is required for IRWKP and IREDTO and the number of terms in AMASS must equal the order of the system stiffness matrix.

The number of nonrigid body modes to be calculated is input in NEIGVL. NEIGVL, which has a nominal value of 5, can optionally be any multiple of five, and must be less than or equal to IREDTO less six.

Care must be exercised in the use of the reduction routine. It is possible to eliminate so many rows from the stiffness matrix that a number of the remaining degrees of freedom are no longer independent. This has the effect of eliminating one or more of the system's six rigid body modes. To insure that this does not occur for a general space frame, at least two translational degrees of freedom in each of three global coordinate directions should remain following reduction. For the special case of a straight beam, two torsional rotations should remain in addition to these translations.
Error messages are printed if the user accidentally specifies a reduction which leads to the elimination of too many rows from the stiffness matrix.

Should more than 12904 words be required to store the stiffness matrix in a given problem, ISFDIM must be set equal to the required number of words and labeled COMMON must be redimensioned accordingly.

4.3.1.2 <u>Output Data</u> - Center Body Option output includes all input data, NAMELIST indicator values used, nodal displacements and rotations, number of iterations required for convergence, maximum error, nodal forces and moments, bar forces and moments, shear flows in shear webs, and a CPU time summary. Modal analysis output includes all input data, NAMELIST indicator values used, natural frequencies, mode shapes, generalized inertia properties, and a CPU time summary. All input data is printed out in a block format. Classification of input data is by code number in column 1, as described in Section 4.3.1.1.

For problems not requiring modal analysis, nodal displacements and rotations are printed out. The number of iterations required for the solution as well as the maximum error in the solution are then printed. Global forces and moments acting on the joints are also output. Forces and moments acting on both ends of all bars are printed. The p, q, and r joint numbers used for each bar are also listed to aid in interpretation of the direction of these forces and moments (positive local sign convention is indicated in Figure 4-1). Shear flows and identifying joints are printed for all shear webs. Positive local sign convention for establishing directions of shear flows is given in Figure 4-7.

When plastic force constraints are violated, the program prints out the elastic solution, which constraints were violated, and the plastic solution (nodal deflections and rotations, and nodal forces and moments). Forces and moments acting on both ends of all bars as well as shear flows in all shear webs are printed out for the final solution. The last output item is a CPU time summary showing a breakdown of CPU time usage.

When employing the modal analysis routine, output data include: input data, the specified number of lowest natural frequencies (excluding rigid body modes), corresponding mode shapes (normalized) for both the reduced and complete systems, generalized inertia properties, and a CPU time summary. On an optional basis, this modal analysis output may be placed on magnetic tape. This information is required by the Center Body Landing Loads Program, as discussed in Appendix E. In addition, the data on this tape is compatible with the Task Order Two Landing Loads and Motion Program, Reference (2). In this case, the Task Order Five modal analysis could be employed for a footpad structure of a platform lander. Thus the improved matrix reduction routine could be taken advantage of.

4.3.1.3 <u>Example of Program Operation</u> - Examples of typical structural and modal analysis problems are included to illustrate interpreting input instructions and output data.

An example of obtaining deflections and internal loads for a center body structure subject to external loads is presented in Section 6.2.4 for the Task Order Three lander center body subject to typical landing loads. Input data and output listing for this problem are presented in Appendix A.

Input data and resulting output data when employing the modal analysis routine of the Center Body Option is given in Appendix B. This analysis was conducted on the center body structure of the Task Order Three lander. The idealization of the center body structure and plots of a number of the mode shapes are presented in Section 6.2.1. This modal data was input to the Landing Loads and Motions Program when studying the effects of center body flexibility on correlation with drop test data for this lander. This landing analysis is discussed in Section 6.2.2.

The data set shown in Appendix B is for a structural idealization with 33 joints and 53 bar members. This results in a stiffness matrix of order 198. The matrix reduction routine was employed to reduce this to order 99. This was accomplished by removing the rotations at all of the joints. Thus, the input mass matrix contained terms associated with each of the three translations at each joint. Twenty modes were requested from the modal analysis routine. Only the mode shapes for the complete system are shown in Appendix B.

4.3.2 <u>Landing Gear Option</u> - This section includes definition of input requirements and format, and output interpretation for the Landing Gear Option of the Structural Analysis Program. Examples of input and output data for a typical problem are contained in Appendix C.

4.3.2.1 Input Data - For each Landing Gear Option data set a data case

header card and information describing landing gear type, loading information, strut properties, gear geometry, footpad information, energy absorption requirements, and indicators needed to control program operation are required as input data. Multiple data cases may be run by stacking data sets. Each data set must consist of a header card followed by the appropriate data cards.

A. <u>Header Card</u> - A header card is required as the first card of a data set to indicate that the Landing Gear Option will be employed. This card must contain the characters GEAR in columns 1 through 4 as indicated in Figure 4-36.

B. <u>Data Cards</u> - The input format for data cards employs a system of code numbers, right-justified in columns 1 and 2, to identify the type of data being read. The seventeen code numbers and corresponding type of input information are as follows:

- (0) Comment cards
- (1) Gear and load card
- (2) Friction card
- (3) Applied displacement card
- (4) Strut material cards
- (5) Solution parameter card
- (6) Nodal point cards
- (7) Footpad card
- (8) Material parameter cards
- (9) Material crush cards
- (10) Compression crush distance cards
- (11) Tension crush distance cards
- (12) Compression spring rate cards
- (13) Tension spring rate cards
- (14) Compression plastic force cards
- (15) Tension plastic force cards
- (16) Data terminator card

B.0 <u>Comment Cards</u> - As many comment cards as desired may be input by leaving columns 1 through 6 blank and entering comments in columns 7 through 80 as shown in Figure 4-36. These comment cards may be located anywhere within the data cards and must precede the data terminator card.



B.1 <u>Gear and Load Card</u> - Information describing landing gear type, type of applied loading, strut parameter for the cantilever gear, and the orientation of the landing surface is input on a data card in which a 1 is placed in column 2. Location of data fields on this card is illustrated in Figure 4-36. A 1 right-justified in columns 6 through 10 signifies that the gear being analyzed is an inverted tripod. Similarly, a 2 in this field indicates a cantilever gear.

A 1 right-justified in columns 11 through 15 indicates that the applied loading is of the frictional type in which the displacement normal to a landing surface is specified. For this type of loading, a code 2 data card is required as input data to specify the applied normal displacement and the friction coefficient. A 2 right-justified in columns 11 through 15 indicates that three applied displacement components are specified. For this case an associated code 3 data card must be input. As explained in Section B.2, a code 2 data card must also be input to specify the direction normal to the landing surface if the footpad has attenuation.

For the cantilever gear only, the length of strut 2 (see Figure 4-17) must be input in columns 21 through 30. This information is required to locate node 4 on the cantilever gear. For the inverted tripod gear this data field is left blank.

Three data fields located between columns 31 and 60 are used to specify the Euler angles ψ , θ , and ϕ which orient the Surface Coordinate System relative to the Lander Coordinate System. These Euler angles must be input in degrees. A definition of these angles and the associated sign convention is given in Figure 4-20.

B.2 <u>Friction Card</u> - Information describing the direction and magnitude of applied displacement (normal to the landing surface) at the footpad joint and the coefficient of friction of the surface is input on a data card in which a 2 is placed in column 2. Location of data fields on this card is illustrated in Figure 4-36. A 1, 2, or 3 right-justified in columns 6 through 10 of this card signifies that the positive direction normal to the landing surface is the X, Y, or Z surface coordinate direction, respectively. This information as well as the information supplied in the remaining data fields of this card must be specified if loading of the friction type is selected

on the code 1 data card. Should the three applied displacement components case be specified on the code 1 data card and the footpad have at least 1 level of attenuation, the normal displacement direction indicator must be supplied on the code 2 data card to indicate the surface coordinate direction to check for footpad crushing. For this case, the remaining information on the data card is not used. Care must be exercised when selecting the coordinate system to insure that the normal surface coordinate direction is out of the surface.

The magnitude of the applied normal displacement in the surface coordinate direction indicated in columns 6 through 10 is supplied in columns 21 through 30. The coefficient of friction of the landing surface is entered in columns 31 through 40.

B.3 <u>Applied Displacement Card</u> - For the case in which three displacement components at the footpad joint are specified, the three surface coordinate components of applied displacement are entered on a data card with a 3 in column 2. Location of data fields on this card is illustrated in Figure 4-37. Three data fields located between columns 21 and 50 are used to enter the three components of applied displacement at the footpad joint. These displacement components must be given in the Surface Coordinate System.

B.4 <u>Strut Material Cards</u> - Each strut of a gear must be assigned a material identification number to indicate which material, input on code 8 through 15 data cards, makes up the strut. The material identification number of each element (strut) is input on data cards with a 4 in column 2. Location of data fields on this card is illustrated in Figure 4-37. One card must be input for each element of the gear, i.e., three for the inverted tripod and four for the cantilever.

The element number must be right-justified in columns 6 through 10. The number identifying the material of which this element is made is right-justified in columns 11 through 15. More than one element may have the same material identification number. This is convenient in the case of a gear which has identical drag struts since then only one set of material properties has to be defined for these struts.



B.5 <u>Solution Parameter Card</u> - The number of steps for investigating the applied loading, the number of iterations to be allowed, the number of different materials used in the gear idealization, an indicator to specify in which direction to check the energy absorbed, the energy absorption cutoff, and the solution tolerance are input on a data card with a 5 in column 2. Location of data fields on this card is illustrated in Figure 4-37.

Four data fields located between columns 6 and 25 are used to specify the number of steps to divide the applied loading, the number of iterations allowed before the solution is assumed to be nonconvergent, the number of different materials employed in idealizing the struts of the gear, and an indicator to specify in which surface coordinate direction to check the energy absorption associated with the loads acting on the footpad. These integers must be right-justified in their respective fields. The energy check indicator is set to 1, 2, or 3 if the energy absorption is to be checked in the X, Y, or Z surface coordinate direction, respectively.

The energy cutoff used to make the energy absorption check is input in columns 31 through 40. In assigning a sign to this cutoff it must be remembered that a plus sign signifies that the force and corresponding displacement components of interest are in the same direction while a minus sign signifies that they are opposite in direction. The tolerance used to check for solution convergence is input in columns 41 through 50. For input data employing units of centimeters and dynes, the solution would be assumed converged to a tolerance specified as 4.0×10^4 when the magnitude of the largest unbalance, between any internal and external load component at any joint of the gear, was less than 4.0×10^4 dynes or 4.0×10^4 dyne-cm as the case may be. For each loading step, the program will run until the solution converges to the tolerance of the maximum number of iterations is exceeded.

B.6 <u>Nodal Point Cards</u> - For each node point (joint) of the gear, except for node 4 of the cantilever gear, the initial node coordinates in the Lander Coordinate System are input on cards with a 6 in column 2. The location of data fields on these cards is illustrated in Figure 4-37. Four of these cards must be input for both the inverted tripod and cantilever gears. On each card, the node point number must be right-justified in columns 6 through 10. Three data fields located between columns 21 and 50 are used to input the X, Y, and Z coordinates of the node in the Lander Coordinate System. B.7 <u>Footpad Card</u> - Information regarding attenuation on the bottom of the footpad is input on a card with a 7 in column 2. Location of data fields on this card is illustrated in Figure 4-38. An attenuation indicator to signify whether or not the footpad has attenuation is right-justified in columns 6 through 10. A 1 signifies that the footpad has attenuation while a 2 indicates that there is none. The number of attenuation crush levels is entered, right-justified, in columns 11 through 15.

Three data fields located between columns 21 and 50 are used to input the magnitudes of up to three attenuation crush forces (see Figure 4-28). These crush forces must be entered in the order of increasing magnitude. The crush stroke magnitudes, associated with up to three crush levels, are entered in three data fields located between columns 51 and 80. Each stroke entry represents the stroke of the footpad honeycomb during crushing of the level of interest and does not include the stroking that may have occurred due to crushing of lower levels. For a footpad without attenuation the information in columns 21 through 80 will be ignored.

B.8 <u>Material Parameter Cards</u> - For each different material, the material identification number, plasticity indicator for axial displacements, moduli of elasticity, and cross-sectional properties are input on a card with an 8 in column 2. Location of data fields on this card is illustrated in Figure 4-38. Material identification number is entered, right-justified, in columns 6 through 10. A plasticity indicator for this material is entered, right-justified, in columns 11 through 15. A plasticity indicator of 1 signifies that for axial displacements the material behaves elastically. A plasticity indicator of 2 indicates that for axial displacements the material behaves plastically. That is, the axial characteristics of this material are those of honeycomb attenuation.

Two data fields located between columns 21 and 40 are used to input the moduli of elasticity for bending and axial displacements of the strut composed of the material identified in columns 6 through 10. Two data fields located between columns 41 and 60 are used to input the moment of inertia for bending displacements and the cross-sectional area of the strut composed of the material identified in columns 6 through 10.

For a material with plastic axial capability (a plasticity indicator of 2), the modulus of elasticity for bending displacements and the moment of

6	F00TF	AD CARD					•			
KZPL0	SALB.	- ×0 - ∪ ∢ ⊢ 0 × 4 ⊢ ⊢ ü z ⊐ ∢ ⊢ -	NO. CRUSH LEVELS	8 JAZX	ATTEN	IUATION CRUSH FOF	RCES	ATTE	NUATION CRUSH STR	DKES
2	۲	• o z		4	IST	2ND	3RD	IST	ZND	3RD
-	2 3 4 5	6 7 8 9 10	11 12 13 14 15	16 17 18 19 20	21 22 23 24 25 26 27 28 29 30	31 32 33 34 35 36 37 38 39 40	0 41 42 43 44 45 46 47 48 49 50	1 51 52 53 54 55 56 57 58 59 60	0 61 62 63 64 65 66 67 68 69 70 7	1 72 73 74 75 76 77 78 79 80
(8)	MATEF	RIAL PAR	AMETER CI	ARDS						
<u> </u>	8	₽⋖⊦	D X - V L b	B.	MODULI OF	ELASTICITY	CROSS SECTION	AL PROPERTIES		
<u> </u>	⊣≺z	zо -ш≈-	- 1 A - C	」⋖z	BENDING	AXIAL	MOMENT OF INERTIA	AREA		
.×	× ×	- 4 -	· O & > – H >-	:×	EB	EA		A		
-	2 3 4 5	6 7 8 9 10	11 12 13 14 15 1	16 17 18 19 20	21 22 23 24 25 26 27 28 29 30	31 32 33 34 35 36 37 38 39 40	41 42 43 44 45 46 47 43 49 50	51 52 53 54 55 56 57 58 59 60	61 62 63 64 65 66 67 68 69 70 71	72 73 74 75 76 77 78 79 80
_6	MATER	NAL CRUS	H CARDS			1				
SPLB	Z P L B	20 - ₩142 - 402	-ZD-UQH JZJOZQ-	-ZO-04+ 0+KOX	ALLOWABLE	E STROKE	UNLOADING \$	SPRING RATE		
Х С	×	لـ 🖌 ن	0 K 2 U	.0 & uu	COMP	TENS	COMP	TENS		
-	3 4 5	6 7 8 9 10	11 12 13 14 15 1	16 17 18 19 20	21 22 23 24 25 26 27 28 29 30	31 32 33 34 35 36 37 38 39 40	41 42 43 44 45 46 47 48 49 50	51 52 53 54 55 56 57 58 59 60	61 62 63 64 65 66 67 68 69 70 71	72 73 74 75 76 77 78 79 80
-9	COMPF	RESSION C	RUSH DIST.	ANCE CA	RDS		-			-
	ZÞLØ	202 - 402 - 402 - 402	BLA	NK		CRUSH DI	IST ANCES IN COMPR	ESSION		
	×	ĊĂ			1ST	2ND	3RD	4TH	5TH	
		0116 8 / 19	rieri bri erizi i ri	07 61 81 /1 9	ne 62 02 1/2 02 02 12 12 12 12 12	Tat lac lac / re lac / re fee 17e 11e	00 04 04 /4 04 04 04 74 74 74	nater loc the factor the for the in		no c/ o/ // o/ c/ b/ c/ 5/
_		-		Ē	IGURE 4–38 INF	UT DATA FOR	MAT - LANDIN	IG GEAR OPTIO	N	-

(Card Codes 7 Thru 10)

inertia must be entered if the strut composed of this material can carry bending loads. These parameters should not be entered if the strut cannot carry bending loads. For either of the above materials, the modulus of elasticity for axial displacements and the cross-sectional area will be ignored since this information regarding axial capability is input in the crush property cards (card codes 9 through 15) for a material with plastic axial capability. The material for the lower member of the main strut (element 4 in Figure 4-17(b)) of the cantilever gear is an example of a material with plastic axial capability for which the bending properties must be input. The material for the main strut (element 2 in Figure 4-17(a)) of the inverted tripod gear is an example of a material with plastic axial capability for which bending properties would not be input.

For a material with elastic axial capability (a plasticity indicator of 1), the modulus of elasticity for axial displacements and the cross-sectional area must be entered. The modulus of elasticity for bending displacements and the moment of inertia are input only if the associated strut has bending capability. The material for the upper member of the main strut (element 2 in Figure 4-17(b)) of the cantilever gear is an example of a material with elastic axial capability for which bending properties must be input. For an inverted tripod gear with rigid drag struts, the material for the drag struts (elements 1 and 3 in Figure 4-17(a) is an example of a material with elastic axial capability for which bending properties would not be input.

B.9 <u>Material Crush Cards</u> - For each material with plastic axial capability (a material which has a plasticity indicator of 2 on the code 8 data card), the material identification number, crush indicators, allowable strokes, and unloading spring rates must be input on a data card with a 9 in column 2. The location of data fields on this card is illustrated in Figure 4-38. The material identification number is right-justified in columns 6 through 10. An unloading indicator is right-justified in columns 11 through 15. A value of 0 for this indicator signifies that unloading of this material will be along a slope specified by the spring rates input in columns 41 through 60 of this card. An unloading indicator of 1 signifies that the material will unload along a slope specified by the next spring rate associated with the load-stroke curve. An initial stroke indicator is input, right-justified, in columns 16 through 20.

A value of 1 for this indicator signifies that the strut which is composed of this material will initially stroke in tension. A value of -1 indicates that this strut will initially stroke in compression. This information is needed for a crushable material to indicate whether the stiffness of the strut at the start of the loading should reflect the first spring rate in tension or that in compression.

Two data fields located between columns 21 and 40 are used to input the magnitudes of allowable (bottoming) strokes in compression and tension. For the material of a strut which can stroke in one direction only, the appropriate bottoming stroke should be input as 1.E-20. Two data fields located between columns 41 and 60 are used to input the unloading spring rate magnitudes in compression and tension. These values should be input only if the unloading indicator is input as 0 in columns 11 through 15.

B.10 <u>Compression Crush Distance Cards</u> - For each material with plastic axial capability, the crush distances in compression (see Figure 4-18) must be input on a data card with a 10 in columns 1 and 2. Data fields on this card are defined in Figure 4-38. The material identification number is rightjustified in columns 6 through 10.

Five data fields, located between columns 21 and 70, are used to define the magnitudes of up to five crush distances in compression. Any number of crush distances between one and five may be input. The last crush distance input must be greater than the allowable compression stroke input on a code 9 data card for this material. This crush distance should have a fictitious value of 1.E+20 to insure that it will not be exceeded during program iteration prior to final convergence. For the material of a strut which can stroke only in tension, one compressive crush distance should be input at a value of 1.E+20.

B.11 <u>Tension Crush Distance Cards</u> - For each material with plastic axial capability, the crush distances in tension (see Figure 4-18) must be input on a data card with an 11 in columns 1 and 2. Data fields on this card are defined in Figure 4-39. The material identification number is right-justified in columns 6 through 10.

Five data fields, located between columns 21 and 70, are used to define the magnitudes of up to five crush distances in tension. Any number of crush distances between one and five may be input. The last crush distance input



FIGURE 4-39 INPUT DATA FORMAT - LANDING GEAR OPTION

(Card Codes 11 thru 14)

must be greater than the allowable tension stroke input on a code 9 data card for this material. This crush distance should have a fictitious value of 1.E20 to insure that it will not be exceeded during program iteration prior to final convergence. For the material of a strut which can stroke only in compression, one tension crush distance should be input at a value of 1.E+20.

B.12 <u>Compression Spring Rate Cards</u> - For each material with plastic axial capability, the spring rates in compression (see Figure 4-18) must be input on a card with a 12 in columns 1 and 2. Data fields on this card are defined in Figure 4-39. The material identification number is right-justified in columns 6 through 10.

Five data fields located between columns 21 and 70 are used to define the magnitudes of up to five spring rates in compression. Any number of spring rates between 1 and 5 may be input. For the material of a strut which can only stroke in tension, one spring rate in compression should be input at a value equal to the bottoming spring rate.

B.13 <u>Tension Spring Rate Cards</u> - For each material with plastic axial capability, the spring rates in tension (see Figure 4-18) must be input on a data card with a 13 in columns 1 and 2. Data fields on this card are defined in Figure 4-39. The material identification number is right-justified in columns 6 through 10.

Five data fields located between columns 21 and 70 are used to define the magnitude of up to five spring rates in tension. Any number of spring rates between 1 and 5 may be input. For the material of a strut which can only stroke in compression, one spring rate in tension should be input at a value equal to the bottoming spring rate.

B.14 <u>Compression Plastic Force Cards</u> - For each material with plastic axial capability, the plastic forces in compression (see Figure 4-18) must be input on a data card with a 14 in columns 1 and 2. Data fields on this card are defined in Figure 4-39. Material identification number is entered, rightjustified, in columns 6 through 10.

Five data fields located between columns 21 and 70 are used to define the magnitudes of up to five plastic forces in compression. Any number of plastic forces between 1 and 5 may be input. For the material of a strut which can only stroke in tension, one plastic force in compression should be input at

a value slightly less than the product of SRC(1) and CDC(1).

B.15 <u>Tension Plastic Force Cards</u> - For each material with plastic axial capability, the plastic forces in tension (see Figure 4-18) must be input on a data card with a 15 in columns 1 and 2. Data fields on this card are defined in Figure 4-40. Material identification number is right-justified in columns 6 through 10.

Five data fields located between columns 21 and 70 are used to define the magnitudes of up to five plastic forces in tension. Any number of plastic forces between 1 and 5 may be input. For the material of a strut which can only stroke in compression, one plastic force in tension should be input at a value slightly less than the product of SRT(1) and CDT(1).

B.16 <u>Data Terminator Card</u> - After all data needed on code 1 through 15 data cards and the desired comment cards are input, a data terminator card is required to signify the end of the data set. This card must contain a 16 in columns 1 and 2.

4.3.2.2 <u>Output Data</u> - Output data from the Landing Gear Option includes all input data, initial conditions, displaced conditions for the gear at each step of the specified loading, and a CPU time summary.

The initial conditions printed for the gear include the X, Y, and Z coordinates of all nodes in the Surface Coordinate System.

The displaced conditions, printed for the gear at each step of loading, include the following: step of loading; number of iterations required to reach convergence; coordinates of all nodes in the Surface Coordinate System; external forces and moments at the nodes in the Surface Coordinate System; the internal axial force, stroke, and axial stiffness of each element; the internal shear force and moment at each end of all elements; the total energy absorbed at the footpad in both surface and lander coordinate components; the number of remaining (noncrushed) footpad attenuation levels; and surface and lander coordinate components of the energy absorbed by footpad attenuation during the loading step concerned.

Should a particular loading step fail to converge, the solution existing in the program, after the maximum number of iterations had been employed, would be printed out and program termination would occur. Should the allowable stroke in any of the members be exceeded or the load normal to the footpad be



(Card Codes 15 and 16)

tensile for any loading step, the current solution and an appropriate error message would be printed, and program termination would occur. The program will continue through all loading steps unless one of the above errors occurs causing termination. If at any loading step the energy cutoff in the appropriate surface coordinate direction is exceeded, a message will be printed to indicate this fact and the program will continue.

4.3.2.3 <u>Example of Program Operation</u> - To illustrate interpretation of input instructions and output data, a typical example of the Landing Gear Option is included. An example problem where the inverted tripod gear of the Task Order Three lander was analyzed is discussed in Section 6.1.1. Input data and output listing for this example are presented in Appendix C.

5. LANDING LOADS AND MOTIONS PROGRAM

The Landing Loads and Motions Program provides the capability of predicting the landing dynamics of a legged lander. Program output presents the time histories of the landing gear loads and the spatial positions, velocities, and accelerations of the lander. Options are available for determining the distribution of landing loads throughout the lander structure and lander stability information.

The lander is comprised of up to five landing gears connecting the center body to the footpads which make contact with the landing surface. The center body may be treated as either a rigid structure with up to six rigid body degrees of freedom, or the effects of a flexible structure may be superimposed on these rigid body motions. Inverted tripod or cantilever landing gears having elastic-plastic load-stroke characteristics, velocity dependent energy absorption characteristics, or a combination of the two may be considered. Two soil mechanics routines are available and, on an optional basis, footpad attenuation material may be located on the bottom of each footpad.

Analytical methods developed for this program are presented in Section 5.1. Organization of the computer program is presented in Section 5.2 and operating instructions are discussed in Section 5.3. A program listing is given in Appendix I.

5.1 <u>Analytical Methods</u> - Analytical methods associated with the Landing Loads and Motions Program are presented below. Included are discussions of coordinate systems, equations of motion, structural idealization, soil mechanics routines, and footpad attenuation system.

5.1.1 <u>Coordinate Systems</u> - Four types of coordinate systems, as shown in Figure 5-1, are used to locate the lander as a function of time. These consist of two axis systems fixed relative to the planet, and two systems moving with the lander. These coordinate systems are all right-handed, and each has three orthogonal axes.

The coordinate systems are defined as follows:

Surface Coordinate System (X_s, Y_s, Z_s) - A coordinate system fixed in the planet and oriented with respect to the slope of the local



The X axis of this system is perpendicular to the ground surface. surface. The Z axis corresponds to the principle direction along the ground slope and the Y axis is 90 degrees across the slope. Gravity Coordinate System (X_g, Y_g, Z_g) - A coordinate system fixed relative to the planet's local acceleration of gravity vector. The X_{o} axis is directed opposite to the local gravitational vector and the Y_{o} axis is parallel to the Surface Coordinate System Y_{s} axis. The $\boldsymbol{Z}_{_{\boldsymbol{O}}}$ axis is perpendicular to the gravitational vector and in the same general direction as the Surface Coordinate System \mathbf{Z}_{c} axis. Lander Coordinate System (X, Y, Z) - A coordinate system moving with the lander and fixed at the center of gravity of the center body. The lander's angular positions are defined in terms of the three Euler angles ψ , θ , and ϕ . Definition of the lander's angular position relative to the landing surface is based on a specific order for these Euler angle rotations. The order required for these rotations consists of an initial rotation, ψ , about the Z_c axis, followed by a rotation, θ , about the displaced Y axis and finally by a rotation, ϕ , about the direction of the displaced X axis resulting from the previous two rotations. Note, that when all three of these Euler angles are zero, the Lander Coordinate System is aligned with the Surface Coordinate System.

Footpad Coordinate System (X_i, Y_i, Z_i) - A coordinate system moving with the ith footpad and fixed at the footpad-main strut pivot point. The $Y_i - Z_i$ plane remains parallel to the bottom of the footpad as the footpad impacts the landing surface. Footpad 1 may be located anywhere relative to the Lander Coordinate System. The remaining footpads are numbered consecutively in a positive angular direction about the X axis.

The Lander Coordinate System is related to the Surface Coordinate System by the following expression:

$$\begin{pmatrix} \mathbf{X}_{\mathbf{s}} \\ \mathbf{Y}_{\mathbf{s}} \\ \mathbf{Z}_{\mathbf{s}} \end{pmatrix} = \begin{bmatrix} \mathbf{DC11} & \mathbf{DC12} & \mathbf{DC13} \\ \mathbf{DC21} & \mathbf{DC22} & \mathbf{DC23} \\ \mathbf{DC31} & \mathbf{DC32} & \mathbf{DC33} \end{bmatrix} \begin{pmatrix} \mathbf{X} \\ \mathbf{Y} \\ \mathbf{Z} \end{pmatrix}$$
(5-1)

The elements in the lander's direction cosine matrix, [DC] are given as follows:

DC11 = Cos
$$\theta$$
 Cos ψ
DC12 = Sin ϕ Sin θ Cos ψ -Cos ϕ Sin ψ
DC13 = Cos ϕ Sin θ Cos ψ +Sin ϕ Sin ψ
DC21 = Cos θ Sin ψ
DC22 = Sin ϕ Sin θ Sin ψ +Cos ϕ Cos ψ (5-2)
DC23 = Cos ϕ Sin θ Sin ψ -Sin ϕ Cos ψ
DC31 = -Sin θ
DC32 = Sin ϕ Cos θ
DC33 = Cos ϕ Cos θ

In addition to the above transformations, relationships between the time derivatives of the Euler angles and the angular velocity components about the lander coordinate axes are required. Integration of these define the lander's angular position as a function of time. The required relationships are expressed as

$$\dot{\phi} = \dot{\Omega}_{x} + \text{Tan } \theta(\dot{\Omega}_{y} \text{Sin } \phi + \dot{\Omega}_{z} \text{Cos } \phi)$$

$$\dot{\theta} = \dot{\Omega}_{y} \text{Cos } \phi - \dot{\Omega}_{z} \text{Sin } \phi \qquad (5-3)$$

$$\dot{\psi} = (\dot{\Omega}_{z} \text{Cos } \phi + \dot{\Omega}_{y} \text{Sin } \phi)/\text{Cos } \theta$$

where Ω_x , Ω_y , and Ω_z are the components of the lander's angular velocity in the Lander Coordinate System.

5.1.2 <u>Equations of Motion</u> - The lander's equations of motion are discussed in two parts. The first presents the rigid body equations for the footpads considered to be in contact with the landing surface. The second part presents the development of the center body equations of motion, including the effects of a flexible center body structure.

5.1.2.1 <u>Contacting Footpads</u> - A footpad is termed "contacting" for two conditions. In the first, a footpad is considered to be contacting only after the footpad actually makes contact with the landing surface. Footpads not considered in contact with the landing surface are included in the center body equations as discussed in Section 5.1.2.2.

Secondly, on an optional basis, all of the lander's footpads may be treated as contacting footpads even if some of the footpads have not yet impacted the landing surface. In this case, inertia loading of the landing gear struts, due to the footpad mass, is simulated.

Each footpad considered to be in contact with the landing surface has three rigid body translation degrees of freedom. Two of these are in the plane of the landing surface and the third is normal to the landing surface. Equations of motion for each contacting footpad are written in the form

$$\begin{bmatrix} m_{i} & 0 & 0 \\ 0 & m_{i} & 0 \\ 0 & 0 & m_{i} \end{bmatrix} \begin{pmatrix} X_{i} \\ \vdots \\ Y_{i} \\ \vdots \\ Z_{i} \end{pmatrix} = \begin{pmatrix} F_{x_{i}} \\ F_{y_{i}} \\ F_{z_{i}} \end{pmatrix} + \begin{pmatrix} -m_{i} & g & \cos \zeta \\ 0 & & \\ m_{i} & g & \sin \zeta \end{pmatrix}$$
(5-4)

Terms in Equation (5-4) are defined as follows:

 $F_{x_i}, F_{y_i}, F_{z_i} = sum of forces acting on ith footpad in Surface Coordinate System.$

g = local acceleration of gravity.

 ζ = local surface slope.

Two sets of forces are acting on each contacting footpad. These are the loads due to the stroking of the landing gear struts and the loads resulting from the interaction between the soil and the footpad attenuation system. These soil loads are zero when the footpad is clear of the landing surface.

5.1.2.2 <u>Center Body</u> - The effects of a flexible center body structure on the center body's motion have been included in the analysis. This is an optional feature of the Landing Loads and Motions Program and may be suppressed through input data. If suppressed, the center body is treated as a rigid body with up to six degrees of freedom.

To completely describe the dynamic motion of the elastic center body, a continuous elastic body must be considered. However, for the analysis of complex structures, the structure is often idealized as a network of finite elements. Motions of the idealized structure are determined at a finite number of arbitrarily selected control points distributed throughout the body. The motion of each control point on this body is expressed in terms of three translational and three rotational displacements. This idealization results in a total number of equations of motion equal to six times the number of control points selected.

In order to reduce the number of equations to be solved in the Landing Loads and Motions Program, the normal mode method, References (9) and (10), was employed. In this method the motion of the center body is approximated by the combination of a limited number of vibratory modes plus the six rigid body modes. By selecting the vibratory modes which will be **e**xcited for a given landing condition, the behavior of the flexible center body structure is obtained.

For the analysis of the legged lander, the center body's free-free (unrestrained) modes were chosen as the vibratory modes. The rigid body modes were assumed to be the three translational displacements defining the position of the center body center of gravity in the Surface Coordinate System and three angular displacements defined in the Lander Coordinate System.

In developing the center body equations of motion, expressions defining the motion of a point on the center body were obtained. These were used to evaluate the kinetic and potential energy of the center body. The final form of the center body's equations of motion were obtained by applying the Lagrangian equations to these energy expressions.

The total displacement of a point on the center body, point j in Figure 5-2, is defined as

$$\overline{r}_{j} = \overline{R} + \overline{\rho}_{j}$$
 (5-5)

 \overline{r}_{j} = position vector of point j relative to the Surface Coordinate System.

- \overline{R} = position vector of center body center of gravity relative to Surface Coordinate System.
- $\overline{\rho}_{j}$ = position vector of point j relative to the Lander Coordinate System.

Because the center body is experiencing angular velocities and accelerations, the velocity and acceleration of point j are expressed as

$$\dot{\vec{r}}_{j} = \dot{\vec{R}} + \dot{\vec{\rho}}_{j} + (\dot{\vec{\Omega}} \times \vec{\rho}_{j})$$

$$\dot{\vec{r}}_{j} = \ddot{\vec{R}} + \ddot{\vec{\rho}}_{j} + (\ddot{\vec{\Omega}} \times \vec{\rho}_{j}) + \dot{\vec{\Omega}} \times (\dot{\vec{\Omega}} \times \vec{\rho}_{j}) + 2(\dot{\vec{\Omega}} \times \vec{\rho}_{j})$$

$$(5-6)$$

In the above, the following definitions apply

 $\frac{1}{\overline{\Omega}}$ = angular velocity of center body in the Lander Coordinate System. $\frac{1}{\overline{\Omega}}$ = angular acceleration of center body in the Lander Coordinate System. $\frac{1}{r_{j}}$ = velocity vector of point j relative to the Surface Coordinate System. $\overline{r_{j}}$ = acceleration vector of point j relative to the Surface Coordinate System. \overline{R} = velocity vector of center body center of gravity relative to the Surface Coordinate System. $\frac{\ddot{R}}{R}$ = acceleration vector of center body center of gravity relative to the

 $\frac{\dot{\rho}}{\dot{\rho}_{j}}$ = velocity vector of point j relative to the Lander Coordinate System. $\frac{\dot{\rho}}{\dot{\rho}_{j}}$ = acceleration vector of point j relative to the Lander Coordinate System.

It is assumed that the position vector locating the point on the center body can be separated into a term which varies with time and a term which remains constant with time. Thus, the location of point j relative to the Lander Coordinate System is

$$\overline{\rho}_{j}(s_{j}, t) = \overline{\rho}_{oj}(s_{j}) + \overline{\rho}_{ej}(s_{j}, t)$$
(5-7)

where

 $s_i = coordinates$ of point j in Lander Coordinate System ($s_i = X_i, Y_i$, and Z_i).

- $\overline{\rho}_{oj}$ = undeformed position of point j in Lander Coordinate System. $\overline{\rho}_{ej}$ = deformed position of point j in Lander Coordinate System measured from the undeformed position of that point.

These position vectors are shown in Figure 5-3.



FIGURE 5-2 POSITION OF POINT ON CENTER BODY IN SURFACE COORDINATE SYSTEM



FIGURE 5–3 POSITION OF POINT ON CENTER BODY IN LANDER COORDINATE SYSTEM

Employing the assumption that the elastic deformation is represented by the superposition of a limited number of vibratory modes, the deformed position of point j is

$$\overline{\rho}_{ej} = \sum_{n}^{N} \overline{\phi}_{nj} q_{n}(t)$$
(5-8)

where

N = number of modes included.

- $\overline{\phi}_{nj}$ = magnitude of nth elastic mode shape at point j. These are a function of the coordinates s_i.
 - q = generalized coordinate associated with nth mode. These are a
 function of time.

Expressing the deformed position of point j as components in the three axes of the Lander Coordinate System results in the following

$$\overline{\rho}_{ej} = \rho^{j} ex^{\overline{i}} + \rho^{j} ey^{\overline{j}} + \rho^{j} ez^{\overline{k}}$$
(5-9)

In the above, \overline{i} , \overline{j} , and \overline{k} are the unit normal vectors in the Lander Coordinate

System. Combining Equations (5-8) and (5-9) results in the following expression for the deformed position of point j.

$$\overline{\rho}_{ej} = \sum_{n}^{N} (\phi^{j} \overline{i} + \phi^{j} \overline{j} + \phi^{j} \overline{k}) q_{n}$$
(5-10)

The terms, ϕ^j , ϕ^j , and ϕ^j are the components of the nth mode shape at point j in the three lander axes directions.

The total velocity of point j is obtained by combining Equations (5-6) and (5-7).

$$\frac{\dot{r}}{\dot{r}_{j}} = \frac{\dot{R}}{R} + \frac{\dot{\rho}}{e_{j}} + \frac{\dot{\alpha}}{\alpha} \times (\bar{\rho}_{oj} + \bar{\rho}_{ej})$$
(5-11)

The total kinetic energy of the center body, T, is obtained by summing the kinetic energy of all control points on the center body each having a mass \overline{m}_{j}

$$T = \frac{1}{2} \int_{j}^{J} \vec{m}_{j} \cdot \vec{r}_{j} \cdot \vec{r}_{j}$$
(5-12)

where J equals the total number of mass points on the center body. Combining Equation (5-12) with the definitions given in Equations (5-10) and (5-11), results in the following kinetic energy expression

$$T = \frac{1}{2} M(\dot{q}_{x}^{2} + \dot{q}_{y}^{2} + \dot{q}_{z}^{2}) + \frac{1}{2} (I_{xx}\dot{q}_{rx}^{2} + I_{yy}\dot{q}_{ry}^{2} + I_{zz}\dot{q}_{rz}^{2})$$

$$- (I_{xy}\dot{q}_{rx}\dot{q}_{ry} + I_{xz}\dot{q}_{rx}\dot{q}_{rz} + I_{yz}\dot{q}_{ry}\dot{q}_{rz})$$

$$+ \frac{1}{2} \dot{q}_{rx}^{2} \sum_{n}^{N} [(P_{yn} + P_{zn})q_{n} + \frac{1}{2} N_{xn}q_{n}^{2}]$$

$$+ \frac{1}{2} \dot{q}_{ry}^{2} \sum_{n}^{N} [(P_{xm} + P_{zn})q_{n} + \frac{1}{2} N_{yn}q_{n}^{2}]$$

$$+ \frac{1}{2} \dot{q}_{rz}^{2} \sum_{n}^{N} [(P_{xm} + P_{yn})q_{n} + \frac{1}{2} N_{zn}q_{n}^{2}]$$

$$+ \frac{1}{2} \sum_{n}^{2} \sum_{n}^{N} [(P_{xm} + P_{yn})q_{n} + \frac{1}{2} N_{zn}q_{n}^{2}]$$

Terms in Equation (5-13) are defined as follows, where the q_k 's and q_{rk} 's are the generalized coordinates of the center body's rigid body modes.

$$\begin{split} M &= \text{ center body mass.} \\ I_{xx}, I_{yy}, I_{zz} &= \text{ center body moments of inertia} \\ I_{xy}, I_{xz}, I_{yz} &= \text{ center body products of inertia.} \\ q_k, q_k, \ddot{q}_k &= \text{ rigid body translational displacement, velocity, and} \\ &= \text{ acceleration of center body center of gravity in the} \\ &= \text{ Surface Coordinate System (for k = X_s, Y_s, \text{ or } Z_s \text{ axes}).} \\ q_{rk}, \dot{q}_{rk}, \ddot{q}_{rk} &= \text{ rigid body angular displacement, velocity, and acceleration relative to Lander Coordinate System (for k = X, Y, or Z).} \\ &= m_n = \text{ generalized mass of nth elastic mode,} \\ &= \sum_{j}^{J} \overline{m_j} \ \overline{\phi}_{nj} \ \cdot \ \overline{\phi}_{nj} \ . \end{split}$$

In addition, the following center body generalized inertia properties are expressed as

$$P_{xn} = \sum_{j}^{J} X_{j} \overline{m}_{j} \phi^{j}_{xn}$$

$$P_{yn} = \sum_{j}^{J} Y_{j} \overline{m}_{j} \phi^{j}_{yn}$$

$$P_{zn} = \sum_{j}^{J} Z_{j} \overline{m}_{j} \phi^{j}_{zn}$$

$$N_{xn} = \sum_{j}^{J} \overline{m}_{j} (\phi^{j}_{yn}^{2} + \phi^{j}_{zn}^{2})$$

$$N_{yn} = \sum_{j}^{J} \overline{m}_{j} (\phi^{j}_{xn}^{2} + \phi^{j}_{zn}^{2})$$

$$N_{zn} = \sum_{j}^{J} \overline{m}_{j} (\phi^{j}_{xn}^{2} + \phi^{j}_{zn}^{2})$$

The potential energy of the center body consists of the potential due to the planet's gravity field and the strain energy due to the center body's elastic deformation.

The potential due to the gravity field is

$$U_{q} = gM(\cos \zeta q_{v} - \sin \zeta q_{z})$$
 (5-14)

and the center body's strain energy is

$$U_{s} = \frac{1}{2} \sum_{n}^{N} m_{n} \omega_{n}^{2} q_{n}^{2}$$
(5-15)

where ω_n is the natural frequency at the nth free-free mode. Combining these expressions, the total potential energy of the center body is

$$U = gM(\cos \zeta q_x - \sin \zeta q_z) + \frac{1}{2} \sum_{n=1}^{N} m_n \omega_n^2 q_n^2$$
(5-16)

The final form of the center body's equations of motion were obtained by applying the Lagrangian equations to the energy terms of Equations (5-13) and (5-16). The Lagrangian equations are expressed as

$$\frac{d}{dt}\left(\frac{\partial T}{\partial q_c}\right) - \frac{\partial T}{\partial q_c} + \frac{\partial U}{\partial q_c} = Q_c.$$
(5-17)

where

q_c, q_c = cth generalized coordinate and generalized velocity (either rigid body or elastic modes). T = center body's kinetic energy. U = center body's potential energy. Q_c = generalized force or moment in cth mode.

The generalized forces of the rigid body translational modes are the combination of the inertia effects of noncontacting footpads and the sum of the landing gear strut forces in the respective axis directions. The generalized moments of the rigid body rotational modes are the sum of the moments at the center body center of gravity, about the center body's axes. These consist of the inertia effects of the noncontacting footpads and moments due to the landing gear strut loads. For the elastic modes, the generalized force in the nth mode is

$$Q_{n} = \sum_{p}^{P} (F_{x}^{p} \phi_{nx}^{p} + F_{y}^{p} \phi_{ny}^{p} + F_{z}^{p} \phi_{nz}^{p})$$
(5-18)

In the above, p refers to the pth point on the center body where a force is applied. There are a total of P forces, each of which has been resolved into components in the Lander Coordinate System axes.

To evaluate the inertia effects of a noncontacting footpad, consider the position vectors shown in Figure 5-4. The position of a footpad in question is given as

$$\overline{r}_{f_i} = \overline{R} + \overline{f}_i$$
 (5-19)

where

- rf = position vector of ith noncontacting footpad relative to Surface
 Coordinate System.
 - R = position vector of center body center of gravity relative to Surface Coordinate System.

With the assumption that the noncontacting footpad is on an extension of the center body structure, its acceleration, \ddot{r}_{f} , is expressed as

$$\ddot{\vec{r}}_{f_{i}} = \ddot{\vec{R}} + (\ddot{\vec{\Omega}} \times \vec{f}) + \dot{\vec{\Omega}} \times (\dot{\vec{n}} \times \vec{f})$$
(5-20)

In the above, the terms $\tilde{\overline{R}}$, $\tilde{\overline{\alpha}}$, and $\dot{\overline{\alpha}}$ are defined in Figure 5-2 and Equation (5-6). Thus, the total inertia loads of all the noncontacting footpads at the center body center of gravity are

$$\overline{F}_{f} = -\sum_{i}^{I} m_{i} \overline{F}_{f_{i}}$$

$$\overline{T}_{f} = -\sum_{i}^{I} m_{i} (\overline{f}_{i} \times \overline{F}_{f_{i}})$$
(5-21)

and



Equations (5-21) may be written as

$$\begin{cases} F_{xf} \\ F_{yf} \\ F_{zf} \end{cases} = -\sum_{i}^{I} m_{i} \begin{pmatrix} \ddot{q}_{x} \\ \ddot{q}_{y} \\ \ddot{q}_{z} \end{pmatrix} - \sum_{i}^{I} m_{i} [DC] \begin{bmatrix} 0 & -f_{z} & f_{y} \\ f_{z} & 0 & -f_{x_{i}} \\ -f_{y_{i}}^{i} & f_{x_{i}} & 0 \end{bmatrix} \begin{pmatrix} \ddot{q}_{rx} \\ \dot{q}_{ry} \\ \ddot{q}_{rz} \end{pmatrix}$$

$$- \sum_{i}^{I} m_{i} [DC] \begin{pmatrix} \dot{q}_{ry} & \dot{q}_{rx} & f_{y} & -\dot{q}_{ry} & f_{x} \\ \dot{q}_{ry} & \dot{q}_{rx} & f_{y} & -\dot{q}_{rz} & \dot{q}_{rz} & f_{x_{i}} \\ \dot{q}_{rz} & \dot{q}_{ry} & f_{z_{i}}^{i} & -\dot{q}_{rz} & f_{y} \end{pmatrix} - \dot{q}_{rz} & \dot{q}_{rz} & f_{x} & -\dot{q}_{rx} & f_{z} \\ \dot{q}_{rz} & \dot{q}_{ry} & f_{z_{i}}^{i} & -\dot{q}_{rz} & f_{y} \end{pmatrix}$$

$$(5-22)$$

and

$$\begin{pmatrix} {}^{T}_{xf} \\ {}^{T}_{yf} \\ {}^{T}_{zf} \end{pmatrix} = -\sum_{i}^{I} m_{i} \begin{bmatrix} (f_{y_{i}}^{2} + f_{z_{i}}^{2}) & -f_{x_{i}}f_{y_{i}} & -f_{x_{i}}f_{z_{i}} \\ -f_{x_{i}}f_{y_{i}} & (f_{x_{i}}^{2} + f_{z_{i}}^{2}) & -f_{y}f_{z_{i}} \\ -f_{x_{i}}f_{z_{i}} & -f_{y}f_{z_{i}} & (f_{x_{i}}^{2} + f_{z_{i}}^{2}) \\ -f_{x_{i}}f_{z_{i}} & -f_{y}f_{z_{i}} & (f_{x_{i}}^{2} + f_{z_{i}}^{2}) \end{bmatrix} \begin{pmatrix} \ddot{q}_{rx} \\ \ddot{q}_{ry} \\ \ddot{q}_{rz} \end{pmatrix}$$

$$-\sum_{i}^{I} m_{i} \begin{bmatrix} 0 & -f & f \\ 0 & -f & f \\ 0 & -f & y_{i} \\ f & 0 & -f \\ 0 & -f & y_{i} \\ -f & y_{i} & y_{i} \\ y_{i} & y_{i} \end{bmatrix} \begin{bmatrix} 0 & 0 & 0 \\ 0 & y_{i} \\ 0 & y_{i} \\ 0 & y_{i} \end{bmatrix} \begin{bmatrix} 0 & 0 & 0 \\ 0 & y_{i} \\ 0 & y_{i} \\ 0 & y_{i} \end{bmatrix}$$
(5-22) (Continued)

$$= \sum_{i}^{I} m_{i} \left\{ \begin{array}{c} f_{x_{i}}\dot{q}_{rx}(f_{y_{i}}\dot{q}_{rz} - f_{z_{i}}\dot{q}_{ry}) + (f_{y_{i}}^{2} - f_{z_{i}}^{2})\dot{q}_{ry}\dot{q}_{rz} + f_{y_{i}}f_{z_{i}}(\dot{q}_{rz}^{2} - \dot{q}_{ry}^{2}) \\ f_{y_{i}}\dot{q}_{ry}(f_{z_{i}}\dot{q}_{rx} - \dot{f}_{x_{i}}\dot{q}_{rz}) + (f_{z_{i}}^{2} - f_{z_{i}}^{2})\dot{q}_{rx}\dot{q}_{rz} + f_{x_{i}}f_{z_{i}}(\dot{q}_{rx}^{2} - \dot{q}_{rz}^{2}) \\ f_{z_{i}}\dot{q}_{rz}(f_{x_{i}}\dot{q}_{ry} - f_{y_{i}}\dot{q}_{rx}) + (f_{z_{i}}^{2} - f_{y_{i}}^{2})\dot{q}_{rx}\dot{q}_{ry} + f_{x_{i}}f_{y_{i}}(\dot{q}_{ry}^{2} - \dot{q}_{rz}^{2}) \\ \end{array} \right\}$$

where

$$\begin{cases} \ddots \\ f_{\nabla x_{i}} \\ f_{\nabla y_{i}} \\ f_{z_{i}} \\ \end{cases} = [DC] \begin{cases} f_{x_{i}} \\ f_{y_{i}} \\ f_{z_{i}} \\ \end{cases}$$

are the components of the ith footpad location, from the center body center of gravity, in the Surface Coordinate System and

- f_{x_i}, f_{y_i}, f_{z_i} = components of position vector locating ith noncontacting footpad in the Lander Coordinate System
 - [DC] = direction cosine matrix, Equation (5-1), relating Lander Coordinate System to Surface Coordinate System.
- F_{xf} , F_{yf} , F_{zf} = inertia forces due to noncontacting footpads, relative to Surface Coordinate System.
- T_{xf}, T_{yf}, T_{zf} = inertia moments due to noncontacting footpad, relative to Lander Coordinate System.

Using the center body energy expressions, Equations (5-13) and (5-16), and the noncontacting footpad inertia loads, Equation (5-22), in conjunction with the Lagrangian equation leads to the center body equations of motion given in Equation (5-23).

5.1.3 Landing Gear Strut Idealization - The two landing gear configuration options available in the Landing Loads and Motions Program are applicable to the inverted tripod gear and the cantilever gear. Each landing gear consists of a main strut and two drag struts. It is assumed that these are pinned struts and thus moments or torsion are not introduced at their ends. Both the main strut and the drag struts are capable of carrying tension and compression loads and may possess either velocity dependent energy absorption characteristics, stroke dependent characteristics, or a combination of the two. The energy absorption characteristics of all main struts in a given lander are the same. Similarly, characteristics of all the drag struts are the same, however they may be different than those of the main struts. For a cantilever gear, the effect of bending in the main strut is included by altering the energy absorption properties of the drag struts.

Relative motion between each footpad and the center body is used to determine the stroke and velocity of stroke in each strut. This information is used to determine the load in a landing gear strut employing the subroutine STRUT.

5.1.3.1 <u>Stroke Dependent Attenuation</u> - The primary type of energy absorption mechanism for the landing gear system consists of crushable

$ \begin{bmatrix} M + \frac{1}{2} m_{1} & 0 & 0 \\ 0 & M + \frac{1}{2} m_{1} & 0 \\ 0 & M + \frac{1}{2} m_{1} & 0 \\ 0 & M + \frac{1}{2} m_{1} & 0 \\ 0 & M + \frac{1}{2} m_{1} & 0 \\ 0 & M + \frac{1}{2} m_{1} & M_{1} \\ 0 & M + \frac{1}{2} m_{1} & M_{1} \\ 0 & -\frac{1}{2} m_{1}^{2} Y_{1} & \frac{1}{2} m_{1}^{2} Y_{1} \\ \frac{1}{2} m_{1}^{2} Y_{1} & \frac{1}{2} m_{1}^{2} Y_{1} \\ \frac{1}{2} m_{1}^{2} Y_{1} & \frac{1}{2} m_{1}^{2} H_{1} \\ -I X + \frac{1}{2} m_{1}^{2} H_{1}^{2} H_{2}^{2} H_{2}^{2} H_{1} H_{1} H_{1} \\ -I X + \frac{1}{2} m_{1}^{2} H_{1}^{2} H_{1}^{2} H_{1} \\ -I X + \frac{1}{2} m_{1}^{2} H_{1}^{2} H_{1}^{2} H_{1}^{2} H_{1} \\ -I X + \frac{1}{2} m_{1}^{2} H_{1}^{2} H_{1}^{2} H_{1} \\ -I X + \frac{1}{2} m_{1}^{2} H_{1}^{2} H_{1}$	$\begin{bmatrix} & & & & \\ & & & & \\ & & & & \\ & & & & $	x - g M COS { z + gM SIN { z
$\left\{\begin{array}{l} \frac{1}{2} m_{1} \dot{a}_{1} \left(\dot{a}_{1} x_{1}^{} y_{1}^{} - \dot{a}_{1} x_{1}^{} \left(\dot{a}_{1} x_{1}^{} y_{1}^{} \right) - \dot{a}_{1} x_{1}^{} \left(\dot{a}_{1} x_{1}^{} y_{1}^{} - \dot{a}_{1} x_{1}^{} y_{1}^{} \right) - \dot{a}_{1} x_{1}^{} \dot{a}_{1} y_{1}^{} y_{1}^{} - \dot{a}_{1} x_{1}^{} \dot{a}_{1} y_{1}^{} y_{1}^{} - \dot{a}_{1} x_{1}^{} \dot{a}_{1} y_{1}^{} y_{1}^{} - \dot{a}_{1} x_{1}^{} \dot{a}_{1} y_{1}^{} z_{1}^{} - \dot{a}_{1} x_{1}^{} \dot{a}_{1} y_{1}^{} z_{1}^{} \right) - \dot{a}_{1} x_{1} \dot{a}_{1} y_{1}^{} z_{1}^{} + \dot{a}_{1} y_{1}^{} z_{1}^{} + \dot{a}_{1}^{} x_{1}^{} \dot{a}_{1}^{} + \dot{a}_{1} x_{1}^{} \dot{a}_{1}^{} z_{1}^{} + \dot{a}_{1} x_{1}^{} \dot{a}_{1}^{} + \dot{a}_{1} x_{1}^{} \dot{a}$	$\begin{bmatrix} 2 & (P_{yn} - P_{zn}) + (N_{zn} - N_{yn}) \\ 1 & (P_{zn} - P_{zn}) + (N_{zn} - N_{zn}) \\ 2 & (P_{zn} - P_{xn}) + (N_{yn} - N_{xn}) \\ 1 & (P_{xn} - P_{yn}) + (N_{yn} - N_{xn}) \end{bmatrix}$	

AN

 $m_{11}\ddot{q}_{11} + \omega_{11}^{2} m_{11} q_{11} = Q_{11} + \dot{q}_{1x}^{2} (P_{yn} + P_{zn}) + N_{xn} q_{11}^{1} + \dot{\dot{q}}_{1y}^{2} | (P_{xn} + P_{zn}) + N_{yn} q_{n}^{1} + \dot{q}_{1z}^{2} | (P_{xn} + P_{yn}) + N_{zn} q_{n}^{1} + FOR n = 1, 2, \dots N$

cartridges housed inside the landing gear struts. Each cartridge crushes at a constant load when the landing gear strut is stroked. Several cartridges, possessing different crushing strengths, may be stacked in series to form a desired load-stroke characteristic. In the present study, all struts may contain a maximum of five cartridges to attenuate compression loads and five to attenuate tension loads.

The idealization of the crushable landing gear struts in the Landing Loads and Motions Program is the same as is employed in the Landing Gear Option of the Structural Analysis Program. A discussion explaining the procedure for determining strut loads during strut stroking is presented in Section 4.1.2.2. In addition, a typical load-stroke relationship for a strut is shown in Figure 4-18.

5.1.3.2 <u>Main Strut Bending</u> - For a cantilever gear, the drag strut loads acting normal to the main strut in combination with the lateral loads on the footpad cause lateral deflections of the main strut. The assumption is made that the effect of this main strut deflection can be approximated through a modification of the drag strut load-stroke relationship.

The main strut is idealized as a simply supported beam whose elastic axis is defined by simple beam theory. Neglecting the effect of axial loads on the lateral deflection of the main strut, the deflected main strut is shown in Figure 5-5. The force F_n is the component of the drag strut force normal to the main strut axis. This force is defined as

$$F_{n}\bar{n} = F(\bar{s}\cdot\bar{n})\bar{n} \qquad (5-24)$$

where F is the force in the drag strut, \overline{n} is a unit vector normal to the main strut, and \overline{s} is a unit vector in the direction of the drag strut. The flexural stiffness of the main strut on either side of the drag strut attach point is defined as EI₁ and EI₂.

The lateral deflection of the main strut due to the load $\ensuremath{\textbf{F}}_n$ is expressed as

$$\Delta \bar{n} = F_{n} \left[\frac{1}{EI_{1}} \left(\frac{a^{3}b}{3l} - \frac{a^{4}b}{3l^{2}} \right) + \frac{1}{EI_{2}} \left(\frac{a^{4}}{l} - \frac{a^{5}}{3l^{2}} + \frac{a^{2}l}{3} - a^{3} \right) \right] \bar{n}$$
(5-25)

The magnitude of this deflection, $\boldsymbol{\lambda}$, in the direction of the drag strut

axis is

$$\lambda \overline{s} = \Delta(\overline{s \cdot n}) \overline{s}$$
 (5-26)

A spring constant reflecting the bending deflection of the main strut is defined as

$$K_{\rm B} = \frac{F}{\lambda} \tag{5-27}$$

This expression is evaluated with the aid of Equations (5-24), (5-25) and (5-26) and results in

$$K_{\rm B} = \frac{1}{\left(\bar{s} \cdot \bar{n}\right)^2 \left[\frac{1}{{\rm EI}_1} \left(\frac{{\rm a}^3{\rm b}}{3\ell} - \frac{{\rm a}^4{\rm b}}{3\ell^2}\right) + \frac{1}{{\rm EI}_2} \left(\frac{{\rm a}^4}{\ell} - \frac{{\rm a}^5}{3\ell^2} + \frac{{\rm a}^2\ell}{3} - {\rm a}^3\right)\right]}$$
(5-28)

This linear spring is assumed to be in series with the spring defining each elastic portion of the load-stroke diagram of the drag strut. A modified spring constant for these elastic portions of the load-stroke curve is calculated to reflect the main strut bending. This modified spring constant is used to define a modified load-stroke relationship for the drag strut. A typical drag strut modified load-stroke curve is shown in Figure 5-6. This modified curve is then employed to define the loads in the drag strut.

During stroking of the main strut, its bending characteristics change due to the change in strut length. These changes in bending characteristics are incorporated by continually modifying the spring constant of the drag strut load-stroke curve. As a result the drag strut load-stroke relationship is continually updated to reflect these changes in main strut bending characteristics.

5.1.3.3 <u>Velocity Dependent Attenuation</u> - Provisions are available for the inclusion of a constant friction force and a velocity dependent damping force in each strut. Both of these are applied in a direction opposite to the velocity of the stroking motion in the strut. The combination of these force terms is expressed as

$$F_{v} = -\frac{\dot{S}}{|\dot{S}|} [F_{r} + C\gamma |\dot{S}|^{\gamma}]$$
(5-29)


FIGURE 5-5 DEFLECTION OF CANTILEVER GEAR MAIN STRUT



FIGURE 5-6 MODIFIED DRAG STRUT LOAD-STROKE CURVE

where

- F_{i} = total velocity dependent force along axis of strut.
- S = stroking velocity of strut.
- F_r = magnitude of constant friction force (input quantity).
- Cy = coefficient of velocity proportional force (input quantity).

 γ = power of velocity in velocity proportional force (input quantity). This force is superimposed on the crushing force or may be included by itself.

5.1.4 <u>Soil Mechanics</u> - Two methods of representing the footpad-soil interaction are incorporated in the Landing Loads and Motions Program. The first of these, referred to as the Primary Soil Mechanics Method, is a modification of the soil mechanics analyses developed during the Lunar Module (LM) soil mechanics studies. This modification is similar to that employed during the Surveyor analysis. An alternate soil mechanics method, referred to as the Secondary Soil Mechanics Method, determines the soil force through a simple elastic-plastic relationship between soil pressure and depth of soil penetration. On an optional basis, a crushable footpad attenuation system located on the bottom of the footpads may be included with either of these soil mechanics routines. The footpad attenuation system is discussed in Section 5.1.5. Symbols employed in the soil mechanics routines are defined in Figure 5-7. The soil forces are determined in the subroutine SOIL.

The surface of each footpad is represented by a number of concentric conical and/or cylindrical segments as shown in Figure 5-8. It is assumed that the footpad is always aligned with the plane of the landing surface. Upon entering the soil mechanics routine, the footpad velocities and soil penetration indicated in Figure 5-8 are available. The three components of the soil force in the Footpad Coordinate System are returned from the soil mechanics routine.

5.1.4.1 <u>Primary Soil Mechanics</u> - The footpad-soil interaction method developed in Reference (11) for the LM shaped footpad utilized principles which are fundamental to the interaction phenomenon occurring during soil penetration. Applicability of this basic method to a different footpad shape was demonstrated by the good agreement obtained between telemetered Surveyor lunar impact data and predicted landing dynamics as reported in Reference (12).

SYMBOL	DEFINITION	UNITS *
A	FOOTPAD AREA PROJECTED ON LANDING SURFACE	L ²
А _р	FOOTPAD AREA PROJECTED ON PLANE PARALLEL TO PLANE DEFINED BY SURFACE NORMAL AND VELOCITY V_{ap}	L ²
Α _Θ	FOOTPAD AREA PROJECTED ON PLANE NORMAL TO VELOCITY Vap	L ²
C _{ms}	SOIL DYNAMIC MECHANICAL STRENGTH COEFFICIENT	
Cd	SOIL DRAG COEFFICIENT	
d	DEPTH OF SOIL PENETRATION	L
dr/dd	CHANGE IN FOOTPAD RADIUS WITH RESPECT TO SOIL PENETRATION	
D _r	RELATIVE DENSITY OF SOIL (NO COMPACTION, $D_r = 0$; MAXIMUM COMPACTION, $D_r = 1$)	
F _{ap}	SOIL FORCE PARALLEL TO FOOTPAD VELOCITY	F
F _{np}	SOIL FORCE NORMAL TO FOOTPAD VELOCITY	F
F _x , F _y , F _z	COMPONENT'S OF SOIL FORCE IN FOOTPAD COORDINATE SYSTEM	F
g	LOCAL ACCELERATION OF GRAVITY	l/T ²
g _e	EARTH ACCELERATION OF GRAVITY	L/T ²
r	FOOTPAD RADIUS AT FOOTPAD-LANDING SURFACE INTERSECTION	L
r _m	MAXIMUM FOOTPAD RADIUS	L
R_1, R_2, R_3, R_4	PARAMETERS DEFINING FOOTPAD SHAPE (SEE FIGURE 5-8)	L
s ₁ , s ₂ , s ₃ , s ₄	PARAMETERS DEFINING FOOTPAD SHAPE (SEE FIGURE 5-8)	L
V _{ap}	TOTAL VELOCITY OF FOOTPAD	L/T
V _x , V _y , V _z	COMPONENTS OF FOOTPAD VELOCITY IN FOOTPAD COORDINATE SYSTEM	L/T
Ŷ	SOIL UNIT WEIGHT	F/L ³
η	WEDGE SHAPE FACTOR FOR MOVING SOIL MASS	
Θ	ANGLE DEFINING DIRECTION OF VELOCITY V _{ap} RELATIVE TO SURFACE NORMAL	А
λ	RATIO OF AXIAL TO NORMAL SOIL FORCE	
ρ	BULK MASS DENSITY OF SOIL	FT ² /L ⁴
Φ	INTERNAL FRICTION ANGLE OF SOIL	А

* UNITS:

L – LENGTH

F – FORCE

T – TIME

A – ANGLE

FIGURE 5-7 SYMBOLS FOR SOIL MECHANICS ROUTINE





In Reference (11) a theory of soil elasto-plastic deformation is used to define the force between the footpad and the deformed soil surface. The soil mass, displaced by the moving footpad, is considered as a degree of freedom independent of the lander system. A spring, representing the soil elasticity, is placed between the footpad and the soil mass. Additional external forces applied to the soil mass represent a momentum transfer force and a force due to the soil strength.

It was shown in the Surveyor simulation that sufficient accuracy can be obtained by neglecting soil elasticity and assuming that the moving soil mass is attached rigidly to the footpad. This simplification results in the removal of the soil mass differential equation from the analysis. Therefore, the soil force acting on the footpad is considered to be the sum of a soil strength term, a soil drag term, and a term approximating the effect of the changing soil mass.

The Primary Soil Mechanics Method employs the empirical relationships discussed in Reference (12). Forces acting on each footpad consist of an axial force, F_{ap} , parallel to the velocity vector of the footpad and a force, F_{pp} , normal to the velocity vector.

The axial force is the sum of the forces due to the soil strength, soil drag, and effect of the changing soil mass and is expressed as

$$F_{ap} = C_{ms} \rho g dA_{\theta} + C_{d} \rho A_{\theta} V_{ap}^{2} + 3\eta \rho r^{2} \frac{dr}{dd} \left(\frac{A_{\theta}}{A}\right)^{3/2} V_{ap}^{2} \cos\theta \qquad (5-30)$$

Coefficients C_{ms} and C_d , representing the soil dynamic mechanical strength and drag coefficients respectively, are empirical factors determined from test and discussed further in later paragraphs. The wedge shape factor for moving soil mass, n, is a function of angle of internal friction of the soil, and is also discussed further in the section on empirical relationships. The term $(A_{\theta}/A)^{3/2}$ accounts for the effect of the angle θ . Change in footpad radius with respect to depth of the footpad, dr/dd, is defined by footpad geometry at the current depth of soil penetration.

The force acting on a footpad normal to the velocity vector is

$$F_{np} = \lambda F_{ap}$$
(5-31)

This force is always directed out of the soil and is in the plane defined by

the surface normal and the velocity vector. The quantity λ is discussed in the paragraphs defining the empirical relationships. Figure 5-9 presents a diagramatic representation of these two soil forces.



FIGURE 5-9 SOIL FORCES PRIMARY SOIL MECHANICS METHOD

Empirical relationships are determined in terms of the soil properties from impact and drag tests conducted during the LM study. Properties of twelve soils are described in Figure 5-10. The most significant of these properties are: unit weight, γ (bulk mass density, ρ , times g_{e}); relative density, D_{r} ; the angle of internal friction, ϕ ; and, to a lesser degree, the

elastic modulus of the soil, E. Based on results reported in Reference (12), the properties $\gamma, \; D_{_{\ensuremath{\mathbf{r}}}}$ and φ are adequate to describe the soil for landing dynamics and are used in the present study. In the soil mechanics subroutine, the empirical relationships for C $_{ms}$, C $_{d}$, λ , and η used for the Surveyor footpad, are employed during the determination of the soil forces acting on each footpad. These relationships are presented in Figure 5-11.

NO.	BENDIX DESIGNATION	DESCRIPTION	RELATIVE DENSITY D _r	UNIT * WEIGHT Ŷ dynes cm ³	FRICTION ANGLE ර deg.	EL ASTIC MODULUS E dynes ⁽ cm ²
1	RS LOOSE	RED CRUSHED VOLCANIC SCORIA (NARROWLY GRADED)	0	6 50	40	379 x 10 ⁶
2	PS LOOSE	WHITE CRUSHED PUMICE (NARROWLY GRADED)	0	374	43	310 × 10 ⁶
3	RS INTER	RED CRUSHED VOLCANIC SCORIA (NARROWLY GRADED)	.45	721	44.5	593 x 10 ⁶
4	RS DENSE	RED CRUSHED VOLCANIC SCORIA(NARROWLY GRADED)	.80	795	47.3	752 x 10 ⁶
5	RSM-a LOOSE	MIXTURE OF RS AND CRUSHED MARBLE (MS) (NARROWLY GRADED)	0	914	37	538 × 10 ⁶
6	RC2 LOOSE	RED CRUSHED VOLCANIC SCORIA(BROADLY GRADED)	0	965	43	552 × 10 ⁶
7	SS LOOSE	WHITE SILICA SAND (WEDRON 40 40 NARROWLY GRADED)	0	1488	29	690 × 10 ⁶
8	SS INTER,	WHITE SILICA SAND (WEDRON 40 40- NARROWLY GRADED)	.53	- 1634	36.8	1241 × 10 ⁶
9	SS DENSE	WHITE SILICA SAND (WEDRON 40 40- NARROWLY GRADED)	.69	1681	39	1655 x 10 ⁶
10	LSM INTER.	MIXTURE OF RC AND AIR-FLOATED CLAY	. 50	1217	42	276 × 10 ⁶
11	LSM DENSE	MIXTURE OF RC AND AIR-FLOATED CLAY	.70	1288	42	414 × 10 ^{.6}
12	RSM-b DENSE	MIXTURE OF RS AND CRUSHED MARBLE (MS) (NARROWLY GRADED)	.75	1335	48	1380 × 10 ⁶

*TO OBTAIN BULK MASS DENSITY OF SOIL, DIVIDE THIS QUANTITY BY EARTH ACCELERATION OF GRAVITY. THIS INFORMATION OBTAINED FROM REFERENCE (11).

FIGURE 5-10 PROPERTIES OF SOILS

WEDGE SHAPE FACTOR FOR MOVING SOIL MASS:

$$\eta = \frac{\pi}{3} \frac{1 + \mathrm{TAN} \, \phi/2}{1 - \mathrm{TAN} \, \phi/2}$$

SOIL DYNAMIC MECHANICAL STRENGTH COEFFICIENT:

$$C_{ms} = 29e^{1.4 D_r} TAN \phi$$

SOIL DRAG COEFFICIENT:

 $\label{eq:cd} C_d \; = \; 0.8 + (\frac{g}{g_e}) \; (4 + 80 \; D_r) \; (\frac{r}{r_m})^2 \; f \; (\Theta \;) \; \text{TAN } \phi \; \; \text{FOR } D_r < 0.5$ $C_{d} = 0.8 + 4 \, (\frac{g}{g_{e}}) \, e^{\frac{4.83 \ D_{r}}{r_{m}}} \, (\frac{r}{r_{m}})^{2} \, f \, (\theta) \, \text{TAN} \, \phi \ \text{FOR } D_{r} \geq 0.5$ WHERE: ~

$$f(\theta) = 1 - \frac{2\theta}{\pi}$$
 FOR $0 \le \theta < 45^{\circ}$

 $f(\theta) = 0$ FOR $\theta \ge 45^{\circ}$

RATIO OF AXIAL TO NORMAL SOIL FORCE:

$$\lambda = 0.25 \quad (\frac{A_p}{A_{\theta}}) (1 - e^{-50\theta}) (1 + SIN\theta) \quad FOR \quad 0 \le \theta < 90^{\circ}$$

$$\lambda = 0.5 \quad (\frac{A_p}{A_{\theta}}) \quad FOR \quad 90^{\circ} \le \theta < 135^{\circ}$$

$$\lambda = 0 \quad FOR \quad \theta > 135^{\circ}$$

FIGURE 5-11 EMPIRICAL SOIL RELATIONSHIPS

5.1.4.2 Secondary Soil Mechanics - An alternate soil mechanics routine is available in the Landing Loads and Motions Program. This method determines the pressure acting on a footpad in terms of depth of penetration of the footpad. The pressure-penetration relationship is defined as shown in Figure 5-12. Initially the soil pressure increases linearly from zero at zero penetration to a selected pressure at a specified cutoff depth. Beyond this depth, the pressure remains constant. The normal soil force is the product of the pressure determined from the relationship shown in Figure 5-12 and the area of the footpad projected on the landing surface.



THE SOIL FORCE NORMAL TO THE LANDING SURFACE OBTAINED BY THE SECONDARY SOIL MECHANICS METHOD IS GIVEN AS $F_x = A d K_p \text{ OR } F_x = A P_{max}$ WHICHEVER IS LESS.

FIGURE 5-12 NORMAL SOIL FORCE SECONDARY SOIL MECHANICS METHOD

During the integration of the footpad's equations of motion, the normal soil force is compared to the force required to bring the footpad to rest during the next integration time step. This critical soil force is determined by

$$F_{cr} = m_{i} \left(g \cos \zeta - \frac{V_{x}}{\Delta t} \right) - S_{x}$$
 (5-32)

where

- F_{cr} = critical soil force.
- m_i = mass of footpad.
- g = local acceleration of gravity.
- ζ = ground slope.
- V_x = velocity of footpad into landing surface.
- Δt = integration step size.

 S_x = component of landing gear strut loads normal to landing surface. The normal soil force applied to the footpad is never allowed to be greater than the magnitude of the force determine by Equation (5-32).

A force, acting in the plane of the landing surface, is obtained by multiplying the normal force by the selected value for the coefficient of friction. This force is applied in a direction opposite to the footpad's velocity in the plane of the landing surface. The components of this inplane soil force are expressed as

$$F_{y} = -\frac{v_{y}}{\sqrt{v_{y}^{2} + v_{z}^{2}}} \quad \mu F_{x}$$

$$F_{z} = -\frac{v_{z}}{\sqrt{v_{y}^{2} + v_{z}^{2}}} \quad \mu F_{x}$$
(5-33)

where

V and V = inplane footpad velocities as shown in Figure 5-12.
F and F = components of inplane soil force.

$$\mu$$
 = coefficient of friction.
F = normal soil force.

It should be noted that the force, F_x , employed in Equation (5-33), is the magnitude of the normal soil force obtained after the force resulting from the relationships of Figure 5-12 is compared with the critical soil force given by Equation (5-32).

5.1.5 <u>Footpad Attenuation System</u> - An additional attenuation system may be located on the bottom of the footpads to limit the landing loads of the footpads. Provisions have been made for including a crushable material on any or all of the conical segments used to represent the footpad shape. The amount of attenuation material crushing for each footpad is determined by a balance between the attenuator force and the soil force. The footpad attenuation system may be included with either the Primary or Secondary Soil Mechanics Methods. Forces associated with the footpad attenuation system are determined in the soil mechanics subroutine.

The footpad attenuation system may have three crush levels and the pressure-stroke relationship for the attenuator material is shown in Figure 5-13. It is assumed that the attenuator is crushed in a direction normal to the plane of the landing surface. The depth of soil penetration and amount of attenuator crushing, are determined by comparing the crush pressure with the soil pressure. With a soft soil and a stiff footpad attenuation material, most of the deformation will take place in the soil. For a very hard soil, a majority of the deformation will occur in the attenuator material. Intermediate values result in deformation of both the soil and attenuator materials.

At the end of a time interval, the attenuator thickness of each footpad which experiences crushing is adjusted to reflect this deformation. This is done by subtracting the crush distance from the coordinate locating the bottom of the footpad. Thus, if the lander rebounds, the crushed shape of the footpad attenuator is retained for the next impact. When the attenuator on a footpad is completely crushed the footpad attenuation portion of the analysis is bypassed, and the soil forces are applied directly to the footpad.

5.1.6 Footpad Sliding Motion - The two forces determined by the soil mechanics in the plane of the landing surface (F_y and F_z) are components of the inplane soil force F_f which is in a direction opposite to the sliding velocity of the footpad. This force tends to retard the sliding motion of the footpad along the landing surface.



FIGURE 5-13 FOOTPAD ATTENUATION SYSTEM

Referring to Figure 5-14, the equation of motion for the footpad in the plane of the landing surface and in the direction of the sliding motion is

$${}^{\mathbf{m}}\mathbf{i} \frac{\mathrm{d}\mathbf{V}}{\mathrm{d}\mathbf{t}} = -\mathbf{F}_{\mathbf{f}} - \mathbf{S}_{\mathbf{h}} \ (\mathbf{\bar{f}} \cdot \mathbf{\bar{h}}) \tag{5-34}$$

Therefore, the velocity of the footpad at the end of an integration step is

$$\mathbf{v}_{\mathbf{s}}\Big|_{\mathbf{t}_{n}^{+}\Delta \mathbf{t}} = -\frac{\mathbf{F}_{\mathbf{f}} + \mathbf{S}_{\mathbf{h}}(\mathbf{\bar{f}} \cdot \mathbf{\bar{h}})}{\mathbf{m}_{\mathbf{i}}} \Delta \mathbf{t} + \mathbf{v}_{\mathbf{s}}\Big|_{\mathbf{t}_{n}}$$
(5-35)

The critical value of F_f , which will just bring the footpad to rest is obtained from Equation (5-35) by setting the velocity at the end of the integration step $(V_s | t_n + \Delta t)$ equal to zero. Thus,



V_S = SLIDING VELOCITY OF FOOTPAD

S_h = COMPONENT OF LANDING GEAR STRUT LOAD PARALLEL TO LANDING SURFACE

 \overline{h} = UNIT VECTOR IN DIRECTION OF S_h

F_f = COMPONENT OF SOIL FORCE IN PLANE OF LANDING SURFACE

 \overline{f} = UNIT VECTOR IN DIRECTION OF F_{f}

m; = MASS OF FOOTPAD

 $\Delta t = INTEGRATION STEP SIZE$

WHEN THE FORCES F_f AND S_h OPPOSE EACH OTHER, THE FOLLOWING HOLDS

 $(\overline{f} \cdot \overline{h}) < 0$

WHILE

 $(\overline{f} \cdot \overline{h}) > 0$

WHEN THESE FORCES ACT IN THE SAME DIRECTION.

FIGURE 5-14 EVALUATION OF IN-PLANE SOIL FORCE MAGNITUDE

$$F_{fc} = \frac{m_i V_s}{\Delta t} \bigg|_{t_n} - S_h (\bar{f} \cdot \bar{h})$$
(5-36)

Equation (5-36), in conjunction with Equation (5-37), is employed to maintain the magnitude of $F_{\rm f}$ at a value equal to or less than that which is required to bring the footpad to rest during the next integration step. When the relationship of Equation (5-37) holds, the magnitude of $F_{\rm f}$ determined by

the soil mechanics routine is not changed.

$$F_{f} \leq F_{fc}$$
 (5-37)

If the condition of Equation (5-37) is not met, the magnitude of F_f is recalculated by Equation (5-38) when F_f and $S_h(\overline{f} \cdot \overline{h})$ are in opposite directions and

$$F_{f} = \frac{m_{i}V_{s}}{\Delta t} \bigg|_{t_{n}} - S_{h}(\overline{f} \cdot \overline{h})$$
(5-38)

Equation (5-39) governs the magnitude of F_f when $S_h(\bar{f}\cdot\bar{h})$ and F_f are in the same direction.

$$F_{f} = \frac{m_{i} V_{s}}{\Delta t} \Big|_{t_{n}}$$
(5-39)

5.1.7 Lander Stability - To determine the stability of a legged lander configuration, the "plane of lander motion," as shown in Figure 5-15, is defined. This plane is defined by the gravity vector \overline{g} , and the translational velocity $\dot{\overline{R}}$, of the lander's center of gravity. The lander is considered unstable when the gravity vector passes outside of the area bounded by the lander footpads. If all the lander footpads are on the same side of the plane of lander motion, the lander is said to be experiencing yaw instability.

The more common case is when two footpads are astride the plane of lander motion, as shown in Figure 5-15. In this case, the vector \overline{L} , extending from the lander center of gravity to the intersection point of a line between these two footpads and the plane of lander motion, is obtained. The stability angle is then defined as

$$S = \cos^{-1} \left\{ \frac{(\overline{L} \cdot \overline{g})}{|\overline{L}| |\overline{g}|} \right\}$$
(5-40)

As long as S is positive, the lander is considered to be stable, When S passes through zero the lander is said to be experiencing pitch instability.

To aid in the evaluation of the lander stability, the pitching velocity is also determined. This quantity is the component of the lander's total angular velocity in the plane of lander motion.



FIGURE 5-15 LANDER STABILITY ANGLE

This stability determination is made in the subroutine STABLE. It is an optional routine of the Landing Loads and Motions Program and is performed if the input indicator JCKSAB is set equal to 1. If the lander becomes unstable, the run is terminated with a printed message stating which type of instability was experienced. 5.2 <u>Program Description</u> - The Landing Loads and Motions Program is best described by defining functions of the program subroutines and examining program organization, as presented in a flow diagram. A listing of the program is given in Appendix I. All programming is in FORTRAN 2.0 for machine computation on CDC 6600 computers.

5.2.1 <u>Subroutines</u> - The Landing Loads and Motions Program is divided into three OVERLAY segments. Each segment consists of an executive subprogram and a number of subroutines as shown in Figure 5-16. This organization has been used to minimize the core storage requirements of the program. Several subroutines have multiple entry points, as indicated in Figure 5-16. The function of each subroutine, depending on the point of entry, is defined in Figure 5-16.

OVERLAY(LLMPT5,0,0) consists of the executive subprogram LLMP. LLMP calls the other two overlays in the proper order and contains all of the COMMON blocks.

READIT is the executive subprogram in OVERLAY(LLMPT5,1,0). This segment of the program reads and prints the input data and initializes all the routines before integration of the equations of motion.

LLMPEX is the executive subprogram in OVERLAY(LLMPT5,2,0). It controls the solution of the equations of motion. Subroutines in this portion of the program obtain the forces in the landing gear struts and determine the soil forces. These forces are summed on the center body and contacting footpads, resulting accelerations determined, equations of motion integrated, and time history quantities printed. At the completion of a time history, program control is returned to OVERLAY(LLMPT5,0,0) for the possible consideration of an additional data set.

Two numerical integration methods are incorporated in the subroutine RKCUT. These consist of a constant step Runge-Kutta method and a variable step Runge-Kutta method. A description of these integration techniques is presented in Appendix F.

SUBPROGRAM NAME	SUBROUTINE NAME	ENTRY Point	OVERLAY LOCATION	SUBPROGRAM OR SUBROUTINE OPERATIONS
LLMP		LLMP	OVERLAY (LLMPT5,0,0)	MAIN EXECUTIVE SUBPROGRAM. CONTROLS THE EXECUTION OF THE TOTAL PROGRAM.
READIT		READIT	OVERLAY (LLMPT5,1,0)	EXECUTIVE SUBPROGRAM IN OVERLAY (LLMPT5,1,0). ALSO READS THE INPUT DATA AND DETERMINES IF THE AMOUNT OF DATA IS CONSISTENT WITH THE IN- PUT CONTROL INDICATORS. IF NOT, THE PROGRAM IS TERMINATED. INITIALIZES THE PROGRAM.
	DATAOT	DATAOT		PRINTS INPUT DATA.
LLMPEX	ETPAD			EXECUTIVE SUBPROGRAM IN OVERLAY (LLMPT5,2,0). CONTROLS INTEGRATION OF EQUATIONS OF MOTION.
,		FITAD		GEAR STRUTS AND SOIL FORCES.
	GEOM	GEOM		DETERMINES DIRECTION COSINE MATRIX AND TIME DERIVATIVES OF EULER ANGLES.
	STRUT	STRUT		DETERMINES LOAD IN LANDING GEAR STRUTS.
	SOIL	SOIL		DETERMINES SOIL FORCES.
	INITUP			NUMERICAL INTEGRATION INITIALIZATION SUBROUTINE.
		LOC		DEFINES AND STORES CUTOFF VARIABLES.
		INUPD		SETS UP STORAGE FOR INTEGRATED VARIABLES.
	RKCUT		OVERLAY (LLMPT5,2,0)	RUNGE-KUTTA NUMERICAL INTEGRATION SUB- ROUTINE.
		SETUP		INITIALIZATION OF RKCUT ROUTINE.
		INTEG		INTEGRATION OF EQUATIONS OF MOTION.
		UPDATE		UPDATES INTEGRATION VARIABLES AND MODIFIES INTEGRATION INTERVAL.
		CUT		CHECKS CUTOFF LIMITS.
	OUTPUT	OUTPUT		PRINTS TIME HISTORY INFORMATION.
		SUMMRY		PRINTS SUMMARY INFORMATION AT THE END OF A DATA CASE.
	STABLE	STABLE		CHECKS STABILITY OF LANDER.
	MATINV	MATINV		MATRIX INVERSION SUBROUTINE.
	GMPRD	GMPRD		MATRIX MULTIPLICATION SUBROUTINE.

FIGURE 5–16 LANDING LOADS AND MOTIONS PROGRAM SUBROUTINES AND SUBPROGRAMS

5.2.2 <u>Flow Diagram</u> - A flow diagram showing the general operation of the Landing Loads and Motions Program is presented in Figure 5-17. The three OVERLAY segments are shown in addition to the various subprograms and subroutines which are located in each OVERLAY. This diagram is not intended to be a comprehensive programming chart, but shows the general flow of the program logic and indicates the order of operations within each subroutine. A complete listing of the Landing Loads and Motions Program is given in Appendix I.

5.3 <u>Program Operation</u> - Successful operation of the Landing Loads and Motions Program depends on proper input of data and correct interpretation of output data. These considerations are discussed in the following paragraphs. Examples of required input data and resulting output data for a typical landing condition are given in Appendix D.

5.3.1 <u>Input Data</u> - Required as input data is information describing the geometric and inertia properties of the specific lander to be studied; initial lander attitudes; linear and rotational velocities; surface conditions such as ground slope and soil properties; and the indicators needed to control the program's operation. This section defines the format of the input data cards and contains instructions for properly supplying input data to the program.

Figure 5-18 shows the required format for the input data cards. Columns 6 through 9 contain a card number, which must be right justified. Input data is placed in floating point form in columns 11-20, 21-30, 31-40, 41-50, and 51-60. Data input in a F format need not be right justified, but data in an E format must be right justified. In either case, the data must be contained entirely within the field of 10 columns provided. Columns 61 and on may be used for sequence numbers, identification statements, or comments. Following the last card of a data set, a card with NEXT in columns 1 through 4, must appear. Multiple cases may be run by stacking the data sets. Note that the input indicator IDSETN(Card 8) must be equal to 1 if another data set follows the current data set. The data cards for any additional data sets follow the first data set, and each of these are terminated with a NEXT in columns 1 through 4. A card with STOP in columns 1 through 4 signals the end of all the data sets.



FIGURE 5-17 FLOW DIAGRAM LANDING LOADS AND MOTIONS PROGRAM



FIGURE 5-17 FLOW DIAGRAM LANDING LOADS AND MOTIONS PROGRAM (Continued)

There are a number of indicators and counters which check the input data as it is read to ensure that the correct amount of information has been input. If the number of data cards is incorrect, the run will be terminated, and error messages printed to indicate where the data error occurred. All of the indicators governing these input options are discussed in Figure 5-19.

Data cards for the first data set must contain all of the information required to completely initialize the first case. Only optional data, consistent with the input control indicators, may be left out of the first data set. For following data sets, only the information which is to be different (or in addition to that of the preceding case) needs to be changed on the appropriate cards.

There is no specific order in which the low number data cards, Cards 1 through 42, must appear in a data set. However, the order of the higher numbered cards (Cards 100 through 1700) is of importance. All of these cards with higher numbers define variables which are subscripted in the Landing Loads and Motions Program. Therefore, all the cards with the same card number must be input in the order in which the user requires this information stored in the program.

All input parameters and their associated data card numbers are defined in Figure 5-20. Most of these parameters are adequately explained in this figure, and Figure 5-19, but a number require additional comments.

There is no specific system of units associated with the input information, except for the angular quantities which must be expressed in degrees. All other parameters may be expressed in any consistent set of units, either English or Metric (inches or centimeters, pounds or dynes).

Care must be exercised to ensure that lander velocity, position, and attitude are correctly initialized. All lander initial conditions are referenced to the Gravity Coordinate System, Figure 5-1. The three components of velocity VELX, VELY, and VELZ are parallel to the axes of the Gravity Coordinate System. Note that a negative VELX is directed into the landing surface.

The desired ground slope, relating the Surface Coordinate System to the Gravity Coordinate System, must be established. This is accomplished by rotating the Surface Coordinate System about the Gravity Coordinate System Y_g axis an amount ζ equal to the ground slope. A positive slope corresponds to 'a rotation in the positive sense about Y_g .

The lander's initial angular orientation is obtained by three successive rotations about the three lander axes. The first is a rotation, ANGX, of the Lander Coordinate System about the Gravity Coordinate System X_g axis. This is followed by a rotation, ANGY, about the displaced position of the Lander

Coordinate System Y axis. Final angular orientation is obtained by the rotation, ANGZ, about the direction of the Z axis resulting from the first two rotations. When these three rotations are zero, the initial attitude of the Lander Coordinate System is aligned with the Gravity Coordinate System.

The input quantity CUTERR (Card 7) defines a tolerance band, above the landing surface, which governs the placement of a footpad's equations of motion into the integration routine. When a footpad enters this band, the inertia effects of the footpad are removed from the center body equations of motion, initial conditions for the footpad are determined by center body motions, and integration of footpad equations of motion begun. This tolerance parameter has meaning only when INLEG (Card 2) is 0.

When including the effects of an elastic center body structure, the indicator MODEIN (Card 42) governs the number of center body modes which are input. NMODES (Card 3) governs the number of modes actually included in the analysis, and may be less than or equal to MODEIN.

A discussion of the input data required for a typical landing condition is presented in Sections 5.3.3 and 6.2.2 and the computer output for this case is given in Appendix D.

5.3.2 <u>Output Data</u> - At specified times during the integration of the equations of motion, various time varying quantities defining the position, velocities, and accelerations of the lander center body and footpads are printed. At the completion of a run, a summary page listing the maximum strokes in the landing gear struts, the reason for termination of the run, and

-	_	_						_	_	_	_					_			_		_	_	-	_		_	-	
8																												6
9																												ß
- 00						<u> </u>	<u> </u>	<u> </u>	<u> </u>			<u>+</u>						<u>├</u>	-						-	-	-	8
5						<u> </u>	ļ	ļ	ļ			ļ			ļ			I		ļ	<u> </u>					1-		5
5																								2	S	2	2	5
3												[·	1											\sim	-	-	\sim	56
52							 	<u> </u>																S	5	5	5	55
															<u> </u>										-	-	-	4
Ś								L				L			ļ			Ļ						N	2	2	<u> </u>	5
3																								ပ		C	H	3
2						6		1																	٤.	a		22
-						-		<u> </u>				<u> </u>												<u> </u>	۵.	0	0	-
5							ļ				<u> </u>	ļ												_	-	<u> </u>	<u> </u>	5
50																												3
9												T																49
80								t										<u> </u>						(-			œ
				<u> </u>		-		<u> </u>				 						<u> </u>							-			2
4																		L						4	4	4	4	4
\$								Z																\sim	\sim	\sim	\sim	18
5		R					1	F				1					4							S	S	S	S	\$5
T		0					t	<u> </u>											-					V	-	=	-	4
4						2												2						-	2	-	=	4
4		d						S									Ω	\sim						ပ		U U	F	4
42		F				M		D									A	S						F	F	D	D	42
		-				111	t										~	6						д	6	0	$\overline{\mathbf{O}}$	-
						<u> </u>	<u> </u>	<u> </u>			<u> </u>							<u>⊢</u>								–	<u> </u>	1
¥				ļ	ļ	L	I	 	ļ	L	L	I							L	· .								4
ŝ						1														-								ရှိ
8																			3	3		()		((8
5						<u> </u>		<u> </u>									· · · ·							~	-			~
<u></u>		<u> </u>		_		 						ļ																3
ñ				0	2		2	8									(×	S)	-	\sim))		36
58		G		Z	N		8	A	ш				I	N	N		3	-	C	2		Δ.	S	S	S	S	S	35
Ŧ		ш		ш	LL		ы	5	>	N		N		N	7			3	T	4		_1	(7)	N	Ň	М	N	Ŧ
<u> </u>		<u> </u>				12			-									<u> </u>	-	<u> </u>			1					<u> </u>
ŝ		-		<u> </u>		4		×	-	<u> </u>								\sim		-			μ	0		0	-	ŝ
32		Ν		z	Z	E	\square	U U	2	Z	N	ш		8	8		A	S	\vdash			0	R	Ľ.	ш	0		33
E		1		-		ш	CO	-	G	A	W	N		CO	CO		R	S	A	A		S	D	Р	Р	S	S	31
0								 																				0
<u> </u>						 	<u> </u>					ļ			<u> </u>			I									·	3
Ň						L					L																	š
28				1			1												2	2				-	\sim			8
5																		_				2		2	2	2	2	5
9					~																							5
2					ш		L			L	<u> </u>							<u> </u>	<u> </u>	0,				\sim			_	2
25		Z	L	>	×			n				[~	N		2	\frown	ပ	2		d		S	S	S	S	25
24		С	n	Г	Ľ.,	N	H	0	>	٢		٢		Y	X		(2	H	Р		L	G	W	M	M	M	24
ŝ		Г	0			-	-	1	-	(5									F	-		-	ш	C	F	C	-	5
2							-															_			<u> </u>		-	~
2		<u>a</u>	0	Z	Z	N	2	0	8	Z	∽	<u> </u>		8			A	S	-			0		<u> </u>	ш			5
 ⊼		-	Z	-	-	T	- I	-	വ	A	≥	>	ĺ	с С	с I		≃	S	A			S	-	₽.	4	ပ	ပ	2
2							1				1																	0
								<u> </u>			-		<u> </u>						<u> </u>	-								2
Ξ						<u> </u>								1				ļ										Ξ
8																				-		((\frown	\frown		12
1														[$\overline{}$		1		1		-	-	5
5	~	<u> </u>	5		~	<u> </u>	t	 	<u> </u>	 	t	<u>†</u>	100	t —	-	5		t	5	100					.			ø
Ē						 	<u> </u>	-			 	 	1.		<u> </u>				<u>x</u>			-		\rightarrow	\vdash		-	Ē
12	Z	\sim	ш	\times	×		E	U U			L		S	\times	<u> </u>	S		1	0	2	ш	<u> </u>	9	S	S	S	S	12
				ш	LL.	×	2	2	A	×		\times	A	×	×	A	\sim		I	۵.	۵.	-	ш	M	Σ	S	Σ	14
Z.	ш	<				-	-	C	F	(5			=	_	-	M	0		-		7			0		U U	-	m
3 14	SЕ	M N	0								1.			1	-			1	<u>⊦</u>	Ļ	<u>-</u> ۱				L .			
2 13 14	A S E	A M	0	٩I	al	1	-	<u> </u>			<			l m						L	L	(-	.,	L L L L L			
12 13 14	ASE	I M A	0 W	a n	N D	M	1 N	ш	ш	z	×	ш	8	8	8	_		<u> </u>		-	Т	0	0	ш	ш.,			=
11 12 13 14	C A S E	TIMA	0 W N	I N D	I N D	HMA	1 / 1	Ľ	ZЕ	AN	W X	ш >	<u>в</u> С	C B	ш С	ш.,	8	S	L A	AT	NT	S 0	N 0	ΡF	ц Ч	C C	<u>-</u> 	11 12
10 11 12 13 14	C A S E	TIMA	0 W N	I N D	a n i	HMA	1 / 1	L	ZE	AN	X M	ш >	8 0	C B	В С	ш.,	8	S	A 7	AT	NT	S 0	N O	ΡF	ц 4	с С	U U	11 11
9 10 11 12 13 14		2 T 1 M A	0 W N S		2 1 N D	H M H	1 N I	S N F	9 Z E	N A N	X M 1	ы С	C B	C B	C B		R	S S	L A 6	AT	I N T	i S 0	0 N 0	t PF	E F	C C		10 11 12
9 10 11 12 13 14	1 C A S E	2 T I M A	3 N M O	4 I N D	5 1 N D	6 H M /	7 I V I	8 N F	9 Z E	0 A N	1 W X	2 V E	3 C B 3	4 C B	5 C B	6 F I	7 R /	8 S	1 A 1	D A T	1 N T	2 S 0	3 N 0	4 P F	5 P F	0 C C	7 C I	9 10 11 12
8 9 10 11 12 13 14	1 C A S E	2 T I M A	3 N M O	4 I N D	2 1 N D	6 H M /		8 N	9 Z E	1 0 A N	1 1 W X	1 2 V E	13 CB	1 4 C B	15 CB	1 6 F I	1 7 R /	1 8 S 5	1 9 A 1	2 0 A T	2 1 N T	2 2 S 0	2 3 N 0	2 4 P F	2 5 P F	2 6 C D	2 7 C I	8 9 10 11 12
7 8 9 10 11 12 13 14	1 C A S E	2 T 1 M A	3 N M O	4 1 N D	5 1 N D	6 H M /		8 N F	9 Z E	1 0 A N	1 1 W X	1 2 V E	1 3 C B	1 4 C B	15 CB	16 F	1 7 R	1 8 S	1 9 A 1	2 0 A T	2 1 N T	2 2 S 0	2 3 N 0	2 4 P F	2 5 P F	2 6 C D	2 7 C I	7 8 9 10 11 1
5 7 8 9 10 11 12 13 14		2 T I M A	3 N M O	4 1 N D	5 1 N D	6 H M /		8 N F	9 Z E	1 0 A N	1 1 W X	1 2 V E	1 3 C B	1 4 C B	1 5 C B	16 F	17 R	1 8 S	1 9 A 1	2 0 A T	2 1 N T	2 2 S 0	2 3 N 0	2 4 P F	25 PF	2 6 C D	2 7 C I	1 7 8 9 10 11 1
6 7 8 9 10 11 12 13 14	1 C A S E	2 T I M A	3 N M O	4 1 N D	5 1 N D	6 H M /		8 8	9 Z E	1 0 A N	1 1 W X	1 2 V E	1 3 C B	1 4 C B	1 5 C B	1 6 F	1 7 R	1 8 S	1 9 A 1	2 0 A T	2 1 N T	2 2 S 0	2 3 N 0	24 PF	25 PF	2 6 C D	2 7 C I	6 7 8 9 10 11 1
5 6 7 8 9 10 11 12 13 14	1 C A S E	2 T I M A	3 N M O	4 1 N D	2 1 N D	H W H		8 N F	3 Z E	1 0 A N	1 1 W X	1 2 V E	1 3 C B	1 4 C B	1 5 C B	1 0 E	1 7 R	1 8 S	1 9 A 1	2 0 A T	2 1 N T	2 2 8 0	2 3 N 0	24 PF	2 5 P F	2 6 C D	2 7 C I	5 6 7 8 9 10 11 12
4 5 6 7 8 9 10 11 12 13 14	1 CASE		3 N M O	4 1 N D	5 1 N D	9 H W H		8 8	9 Z E	1 0 A N	1 1 W X	1 2 V E	1 3 C B	1 4 C B	1 5 C B	1 0 E	1 7 R	1 8 S	1 9 A 7	2 0 A T	2 1 N T	2 2 8 0	2 3 N 0	2 4 P F	2 5 P F	2 6 C D	2 7 C I	4 5 6 7 8 9 10 11 12
3 4 5 6 7 8 9 10 11 12 13 14	1 CASE	2 T I M A	3 N M O		5 1 N D	4 W H 9		8 8	9 Z E	1 0 A N	1 1 W X	1 2 V E	1 3 C B	1 4 C B	1 5 C B	1 0 E	1 7 R	1 8 S	1 9 A 7	2 0 A T	2 1 N T	2 2 8 0	2 3 N 0	2 4 P F	2 5 P F	2 6 C D	2 7 C	3 4 5 6 7 8 9 10 11 12
3 4 5 6 7 8 9 10 11 12 13 14		2 T I M A	3 N M O		5 1 N D	H W H 9		8 N	9 Z E	1 0 A N	1 1 W X	1 2 V E	1 3 C B	1 4 C B	1 5 C B	1 0 E	1 7 R	1 8 S	1 9 A 1	2 0 A T	2 1 N T	2 2 8 0	2 3 N 0	2 4 P F	2 5 P F	2 6 C D	2 7 C	3 4 5 6 7 8 9 10 11 12
2 3 4 5 6 7 8 9 10 11 12 13 14		2 T I M A	3 N M O		2 I N D	H H 9		8 8	9 Z E		1 1 W X	1 2 V E	1 3 C B	1 4 C B	1 5 C B	1 0 E	1 7 R	1 8 S	1 9 A	2 0 A T	2 1 N T		2 3 N 0	2 4 PF	2 5 P F		2 7 C	2 3 4 5 6 7 8 9 10 11 13
1 2 3 4 5 6 7 8 9 10 11 12 13 14			3 N M O			/ W H 9		8 N E	3 Z 6		1 1 W X	1 2 V E	1 3 C B	1 4 C B			1 7 R	1 8 S	1 9 A	2 0 A T	2 1 N T		2 3 N 0	2 4 PF	2 5 P F			1 2 3 4 5 6 7 8 9 10 11 13

FIGURE 5-18 INPUT DATA FORMAT LANDING LOADS AND MOTIONS PROGRAM

59 60 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 57 58 59 60 5) S 5 5 S 2 58 $\overline{}$ _ _ _ 5 S 2 Ω. S ŝ S • 57 33 34 35 36 37 38 39 40 41 42 43 44 45 46 47 48 49 50 51 52 53 54 55 56 TDS S S S S S S S FCD C M Z 2 ۵ 0 ٥ 0 X N × PDSY > -C F C F S S S S . د م D C Ч 2 ш Ъ 0 Ω 8 S Ξ M ٩ S S A Ь. C S S ۵. ۵. ٩ 4) 4) 4 4) 4 4 4 4 4 4 4 4 .) S M S (\neg -S S _ S TDS X Ś S S S D 5 _ F C D о С T D z 1 Δ Ω PDSΥ Ŧ T × > N × ပ C S F F n S S S) R ш ш 8 R D D R R К M Ν W D S ۵. ۵. Ъ S S Ad S C S S S ۵. ۵. ŝ 3 ~ $\overline{}$ 3 ŝ ŝ $\widehat{=}$ 36 37 . . 3 ŝ 3 ŝ ŝ 3 Ξ. ŝ \neg --S Ś _ ^ TWS B DS 33 34 35 S W C D S S S 3 S Ω 8 ____ Ś S C M PDSV Ω Δ × С 0 ပ Ω -C C × >--N S F DT N ပ ပ RT ∍ 9 Ś S Z --Σ Ś Ś Ś S R × Z N Σ Ο 32 2 R A ш Ű. Δ £ R A 0 0 Ξ æ 32 N N W Ъ ۵. ۵. C N N ۵., ۵. 31 S S S G ۵. ٩ C S S S 9 21 22 23 24 25 26 27 28 29 30 31 30 29 2 2 12[) 28 C D S (2) DS(2) 2) 2 2 2 S (2) F T D S (2 _ . . 2 _ 27 2 \sim τ 26 S ÷ Ś S S S S 8 8 Δ G × 25 Ś S ≥ M υT D S X D S Y C D × S E E E Ο C 9 Х Z 24 Ξ Z LĻ, > × ·----S S Ś DT C Σ ш S ш S 23 ပ × H) <u>ب</u> ۵ ۲ 0 Ц. Σ Z R 0 Z Σ X × · ----22 ≃ ¢ **---**Ľ. R 0 ≻ ٩. Uυ 7 × ۵. ٩ Р U L S S S C 7 ۵. ٩ ٩. S Ś > 0 S 2 19 20 2 2 12 13 14 15 16 17 18 ----I -----CDS(1) DS(1)) -1 ---------. -S S S Z X 0 U T (1 Ś S S ŝ Ś S Ś ----TW N S S 0 Q ٥ 8 S ഫ S W 2 × ۵ ۵ a 0 ---C ш × Х C ပ > Σ × > N Ê F 02 <u>0</u> ۵. بد R – XN ပ Σ Ś PFT S S S ш R ပ ш S Ś S C ⊢ -0 ۲. 0 ပ R × Q W 0 8 ပ R 0 8 X M × X R ۵. S S LL_ ۵. ن ပ S S S _ Ŀ 2 × × × G X ۵. ۵. ۵. ۵. ٩ 9 10 11 S 00 ъ 0 ----2 3 4 S و 1 œ S 0 ---Ņ 0 0 0 0 0 0 0 C 0 0 0 0 2 0 2 ŝ ŝ ŝ ŝ 3 ŝ m 3 3 3 4 -**Z**3. 0 0 0 0 0 0 ò 0 0 0 0 00 2 8 ŝ 2 -7 ----4 5 9 7 80 S 0 ----e ----S 3 4 2 ----

FIGURE 5-18 INPUT DATA FORMAT LANDING LOADS AND MOTIONS PROGRAM (Continued)

	_			_				_	-					_											_		 _
8		\sim	-																								 3
6	_	5	ŝ	5		_																		+	-+	-+	 6
<u></u>																									+	+	
ŝ	<u>1</u>			~																		_					 ŝ
5	•																										5
(g)	_		-																								9
5				N.	-																			+			 5
5		~	~																								 5
ι.Υ	N	-	H																								5
ន	S		D	\supset																					1		33
2	0	0	0	0	-																					-	 2
	-	-	-																						+		 <u> </u>
5		<u> </u>	4	ш																							 2
ន		\sim	-																			1					ន
9		4	4	4																							 6
o																										+	 -
4	-																									+	 4
4	-																					_					4
\$	-1	\sim		-						- 1																	\$
5		×	~	N																							5
È			-																						+	\rightarrow	 -
1	14		-									<u> </u>	ļ	ļ			L										 4
4	S	>	D																								4
3		0	0	0																					T		\$
	a	۵.	6	d																							
님									<u> </u>			<u> </u>															
¥													L														 ¥
ရာ	\sim	3	ŝ	3								l.	l													1	ŝ
8	3	-										_															8
1	-												<u> </u>													-+	 1
ŝ																											 <u> </u>
36		\sim	$\overline{}$	\sim	-																	_					ŝ
35	(X	Υ	Ζ)							_															35
4	N		L		N								1														 4
	-	<u> </u>	-	-																							 ~
ñ	2	2	<u> </u>	Р	9																						 ŝ
32		0	0		C)								[[- (R
=	۵.	Р	Р	Р	Р																						
6		-																							+		 6
9			\square						<u> </u>			<u> </u>		 	<u> </u>												 ŝ
ឌ		2	2	2																						. 1	23
82	2	,	•	•									T	I													28
	-	1	-	-									1				(· · · · ·									 5
6								·																		+	
~ ~	_	(\sim	\sim								<u> </u>	ļ	ļ			L	ļ									 2
25	\sim	X	~	N	\sim																						25
7	N	Т	-	F	7										-												2
13		-	-	5	(5	<u> </u>						<u>+</u>	t	-			<u> </u>	t									 6
~		-			<u> </u>	<u> </u>						 	{	(<u> </u>			-	I									 2
22		0	0	0	0							L		<u> </u>			Ì										 5
21	٩	4	D	₽	a																						ដ
2														Γ										r 1			2
6	_		<u> </u>		<u> </u>		<u> </u>		<u> </u>			†	+	<u> </u>	t	-		t	<u> </u>								 嵩
Ē				ļ	ļ	<u> </u>	ļ	 	┣──	J	J		1	<u> </u>	<u> </u>	<u> </u>	<u> </u>	<u> </u>		ļ		 	┣───	<u> </u>			
18		-				L	L		L			ļ	_	I	L	L	ļ	L	L			ļ	L		(2
11	-	—	-		\sim						1							1								I	
9	_		\sim	し		<u> </u>	t	<u>†</u>	<u> </u>	· · · · ·	1	1	1	t	1	<u> </u>	1	t	1				t				 3
E		-	-	NI		┣	•	<u> </u>	<u> </u>	t		1	+	+	<u> </u>	<u> </u>	+	†	 	<u> </u>		1	 	<u> </u>			 5
Ē		L <u>^</u>	~	IN	Ļ	L	ļ	 	ļ	 	 		1		 	 	 	 	 	I			· .				 E.
	M	Η	1-	1	\times								\bot		L	L			L.			<u> </u>					 1
3	S	D	n		G		I			1		[1	1		Ι			1								 2
2	~	Ċ					t	t	t	1	t	t	+	1	t	1	1	†	t			1					 2
E			ΗĽ	10	-		-	 	 	├	<u> </u>	+	+	+		 	 	 	 	 		 		\vdash	\vdash		
Ξ	للم		<u> </u>		L Q		I				L	L	1	1	<u> </u>	ļ	 	I	I	ļ	l		L				 E
2								1						1				1									12
	0	0	10	10			ļ	1	1	<u> </u>	1	1-	1	1	1	1	1	1	1	1		1	1	1	[]		
Ĩ-	_	<u>ا</u>	+=	+_	1	┣	ŧ		t	 	ł	+ -		+	<u>+</u>		 	 	t	f	f		1	1 1			 t-I
~	0					┣	ļ	I	<u> </u>	Ļ	Ļ		+	+	 	 	.	 	1	ļ		L	ļ	ļ			 l "
~	ŝ	4	2	و ا	1	l		1	l I		I I	1		1	1		L	L	1		L						너
2					-	<u> </u>		Ι	Γ_		Γ	1	1	1	T	Τ	Ι		1		1	1	Γ	Γ			5
F			ţ	+	 	┣	+	t	+	<u> </u>	 	+	+	+	+	+	+	+	+	+	 	t	t	+	h		 +
ľ.		ļ	ļ	ļ	ļ	┣				 	 	+	1		+		 	 		 	 		 		 	├ ──-	 12
4		L					۵.,				L				1		1	1	1								
-			<u> </u>			×	0				T							Γ	[I	[1			3
		†	t—	<u> </u>	t	T _W		1	t	<u> </u>	t —	\mathbf{t}	1	1	1	1	1	<u>† </u>	+	t —	†	1	1	†	t		 1
		ļ	 		<u> </u>		Ē	_	+	 	+	╉───	+	+	+	+	╆	+	+	{—	 	+	+	ł	 		 +-4
-		-	-	-		-		-															-	-	-		

FIGURE 5-18 INPUT DATA FORMAT LANDING LOADS AND MOTIONS PROGRAM (Continued)

CARD 2: IPTCNT INDICATOR WHICH DEFINES NUMBER OF INTEGRATION STEPS BETWEEN PRINT TIMES. INLEG = 0 FOOTPAD EQUATIONS OF MOTION INTEGRATED ONLY AFTER ACTUAL IMPACT WITH LANDING SURFACE. INLEG = 1 FOOTPAD EQUATIONS OF MOTION INTEGRATED AT ALL TIMES. IFPPRT > 0 NUMBER OF INTEGRATION TIME STEPS TO BE PRINTED FOLLOWING IMPACT OF A FOOTPAD. IFPPRT = 0 IPTCNT (CARD 2) CONTROLS PRINT INTERVAL AT ALL TIMES. CARD 3: NMODES = 0RIGID CENTER BODY ASSUMED. NUMBER OF CENTER BODY MODES INCLUDED IN ANALYSIS (5 MAXIMUM). NMODES > 0 NOOUT = 0NO SECONDARY ACCELERATION OUTPUT POINTS INCLUDED. NOOUT > 0 NUMBER OF SECONDARY ACCELERATION OUTPUT POINTS (10 MAXIMUM). CARD 8: NFORC = 0 NO SECONDARY TIME HISTORY OUTPUT OBTAINED. SECONDARY TIME HISTORY PLACED ON MAGNETIC TAPE CORRESPONDING TO TIMES OF NFORC =1 PRINTED OUTPUT. NO SECONDARY INTEGRATION VARIABLE OUTPUT. 10U0UT = 0 IQUOUT =1 SECONDARY INTEGRATION VARIABLE OUTPUT OBTAINED WITH NORMAL OUTPUT. THIS OPTION IS OVERRIDDEN WHEN NMODES =0 SINCE THE NORMAL OUTPUT LISTS ALL THE INTEGRATED VARIABLES IN THIS CASE. JCKSAB = 0NO LANDER STABILITY CHECK. JCKSAB = 1 LANDER STABILITY CHECK PERFORMED. IDSETN =0 INDICATES ANOTHER DATA CASE DOES NOT FOLLOW THE CURRENT DATA. INDICATES ANOTHER DATA CASE DOES FOLLOW THE CURRENT DATA. IDSETN = 1CARD 21: NTYPE = 0PRIMARY SOIL MECHANICS ROUTINE. SECONDARY SOIL MECHANICS ROUTINE. NTYPE =1 **CARD 23**: NOL EG INDICATES NUMBER OF LEGS (5 MAXIMUM). ILEG = 0 INVERTED TRIPOD GEAR. ILEG =1CANTILEVER GEAR. **CARD 42:** MODEIN = 0NO MODAL DATA INPUT. MODEIN > 0NUMBER OF CENTER BODY MODES TO BE INPUT (5 MAXIMUM). NUMBER OF INPUT CARDS 100 MUST EQUAL NOOUT (CARD 3). CARD 100: NUMBER OF INPUT CARDS 200 MUST EQUAL NOLEG (CARD 23). CARD 200: NUMBER OF INPUT CARDS 300 MUST EQUAL NOLEG (CARD 23). CARD 300: CARD 400: NUMBER OF INPUT CARDS 400 MUST EQUAL TWO TIMES NOLEG (CARD 23). NUMBER OF INPUT CARDS 500 MUST EQUAL MODEIN (CARD 42). CARD 500: NUMBER OF INPUT CARDS 600 MUST EQUAL MODEIN (CARD 42). CARD 600: NUMBER OF INPUT CARDS 700 MUST EQUAL MODEIN (CARD 42). CARD 700: NUMBER OF INPUT CARDS 800 MUST EQUAL NOLEG (CARD 23). CARD 800: NUMBER OF INPUT CARDS 900 MUST EQUAL NOLEG (CARD 23). CARD 900: CARD 1000: NUMBER OF INPUT CARDS 1000 MUST EQUAL NOLEG (CARD 23). NUMBER OF INPUT CARDS 1100 MUST EQUAL TWO TIMES NOLEG (CARD 23). CARD 1100: NUMBER OF INPUT CARDS 1200 MUST EQUAL TWO TIMES NOLEG (CARD 23). CARD 1200: NUMBER OF INPUT CARDS 1300 MUST EQUAL TWO TIMES NOLEG (CARD 23). CARD 1300: NUMBER OF INPUT CARDS 1400 MUST EQUAL NOOUT (CARD 3). CARD 1400: NUMBER OF INPUT CARDS 1500 MUST EQUAL NOOUT (CARD 3). CARD 1500: NUMBER OF INPUT CARDS 1600 MUST EQUAL NOOUT (CARD 3). CARD 1600: NUMBER OF INPUT CARDS 1700 MUST EQUAL MODEIN (CARD 42). CARD 1700:

> FIGURE 5-19 INPUT DATA CONTROL INDICATORS LANDING LOADS AND MOTIONS PROGRAM

CARD NO.	INPUT VARIABLE	COORDINATE SYSTEM*	VARIABLE DEFINITION
1	CASENO		CASE NUMBER.
· 2	TIMAX		MAXIMUM RUN TIME.
2	IPTCNT		TIME HISTORY PRINT CONTROL, NUMBER OF At's BETWEEN PRINTS.
2	INLEG		FOOTPAD INTEGRATION CONTROL INDICATOR.
			= 0 FOOTPAD FOULTIONS INTEGRATED ONLY AFTER FOOTPAD IMPACT
			= 1 FOOTPAD FOUATIONS INTEGRATED AT ALL TIMES
2	IFPPRT		PRINT CONTROL INDICATOR FOR FOOTPAD IMPACT
			= 0 CONTROL OF PRINT INTERVAL ALWAYS GOVERNED BY IPTCNT
			> 0 NUMBER INTEGRATION TIME STEPS TO BE PRINTED FOLLOWING
			FOOTPAD IMPACT
3	NMODES		NUMBER OF ELASTIC MODES INCLUDED IN ANALYSIS.
3	NOOUT		NUMBER OF SECONDARY ACCELERATION OUTPUT POINTS.
4	INDFXD		INDICATOR TO SUPPRESS CENTER BODY Xs DEGREE OF FREEDOM.
			= 0 ALLOW DEGREE OF FREEDOM
			★ 0 SUPPRESS DEGREE OF FREEDOM
4	INDFYD		INDICATOR TO SUPPRESS CENTER BODY YS DEGREE OF FREEDOM (SEE
			INDFXD).
4	INDFZD		INDICATOR TO SUPPRESS CENTER BODY Z _S DEGREE OF FREEDOM (SEE
			INDFXD).
5	INDFXR		INDICATOR TO SUPPRESS CENTER BODY ROTATION ABOUT X AXIS (SEE
			INDFXD).
5	INDFYR		INDICATOR TO SUPPRESS CENTER BODY ROTATION ABOUT Y AXIS (SEE
			INDFXD).
5	INDFZR		INDICATOR TO SUPPRESS CENTER BODY ROTATION ABOUT Z AXIS (SEE
			INDFXD).
6	HMAX		MAXIMUM INTEGRATION TIME INTERVAL.
6	HMIN		MINIMUM INTEGRATION TIME INTERVAL (VARIABLE STEP RUNGE-KUTTA).
6	EMAX		WAXIMUM INTEGRATION ACCURACY (VARIABLE STEP RUNGE-KUTTA).
6	EMIN		WINIWUW IN FEGRATION ACCURACY (VARIABLE STEP RUNGE-KUTTA).
Ь	IP		QUANTITY USED TO SET INITIAL INTEGRATION STEP SIZE.
,			$\Delta I = \Delta I_{\text{max}} / (2^{-1}).$
· /			
			IVARIA - O VARIADLE STEF INTEGRATION. IVARIA - 1 CONSTANT STEP INTEGRATION
7	імтн		DUNMY VARIARIE SET FOLIAL TO 0
7	CUTERR		TOLERANCE ON CONTROL OF FOOTPAD IMPACT
Ŕ	NEORC		SECONDARY TIME HISTORY OUTPUT INDICATOR.
, v			NEORC = 1 SECONDARY OUTPUT ON TAPE 3.
			NFORC = 0 NO SECONDARY OUTPUT.
8	юлопт		INTEGRATED VARIABLE OUTPUT INDICATOR.
Ĭ			IQUOUT = 1 PRINT ALL INTEGRATED VARIABLES.
1			IQUOUT = 0 NO INTEGRATED VARIABLE OUTPUT.
8	JCKSAB		STABILITY CHECK INDICATOR.
			JCKSAB = 0 - DO NOT CHECK STABILITY.
•			JCKSAB = 1 - CHECK LANDER STABILITY.
8	IDSETN		MULTIPLE DATA CASE INDICATOR.
			IDSETN = 1 - ANOTHER DATA SET FOLLOWING CURRENT DATA SET.
			IDSETN = 0 - NO ADDITIONAL DATA S⊫TS.

* NOTE: THE FOLLOWING ABBREVIATIONS ARE USED TO DEFINE THE COORDINATE SYSTEMS.

SCS - SURFACE COORDINATE SYSTEM.

LCS - LANDER COORDINATE SYSTEM.

GCS - GRAVITY COORDINATE SYSTEM

FIGURE 5-20 INPUT DATA LANDING LOADS AND MOTIONS PROGRAM

CARD NO.	INPUT VARIABLE	COORDINATE SYSTEM*	VARIABLE DEFINITION
9	ZETA	GCS	GROUND SLOPE.
9	GRAV		ACCELERATION OF GRAVITY ON PLANET.
9	GRAVE		ACCELERATION OF GRAVITY ON EARTH.
10	ANGX	LCS	INITIAL ANGULAR ROTATION OF LCS ABOUT Xg.
10	ANGY	LCS	ANGULAR ROTATION OF LCS ABOUT DIRECTION OF Y FOLLOWING ANGX.
10	ANGZ	LCS	ANGULAR ROTATION OF LCS ABOUT DIRECTION OF Z FOLLOWING ANGX AND ANGY.
11	WX	LCS	INITIAL LANDER ANGULAR VELOCITY ABOUT X AXIS.
11	WY	LCS	INITIAL LANDER ANGULAR VELOCITY ABOUT Y AXIS.
11	WZ	LCS	INITIAL LANDER ANGULAR VELOCITY ABOUT Z AXIS.
12	VELX	GCS	INITIAL LANDER VELOCITY ALONG X _g axis.
12	VELY	GCS	INITIAL LANDER VELOCITY ALONG Y _g AXIS.
12	VELZ	GUS	INITIAL LANDER VELOCITY ALONG Z _g AXIS.
13	CBMASS		CENTER BODY MASS.
14	CBIXX		CENTER BODY MASS MOMENT OF INERTIA – I _{XX} .
14			CENTER BODY MASS MOMENT OF INERTIA - Iyy.
14	CBIZZ		CENTER BODY MASS MOMENT OF INERTIA - Izz.
15			CENTER BODY PRODUCT OF INERTIA - 1xy.
15			
15	EDMASS		CENTER BUDY PRUDUCT OF INERTIA - IVZ.
10	PAD(I)		
19	SS(1)		RADIUS UF TIM FUUTPAD SEGMENT (SEE FIGURE 5-8).
10	33(1)		(SEE FIGURE 5-8).
19	ATTHCK(I)		THICKNESS OF ITH SEGMENT OF FOOTPAD ATTENUATION MATERIAL RELATIVE TO BOTTOM OF FOOTPAD (SEE FIGURE 5-13).
20	ATTPRS(I)		CRUSH PRESSURE OF ITH SEGMENT OF FOOTPAD ATTENUATION MATERIAL (SEE FIGURE 5-13)
21	NTYPE		SOIL MECHANICS INDICATOR.
		1	NTYPE = 0 - PRIMARY SOIL MECHANICS.
			NTYPE = 1 - SECONDARY SOIL MECHANICS.
22	SOILP(1)		FOR NTYPE = 0 - SOIL INTERNAL FRICTION ANGLE.
			FOR NTYPE = 1 - COEFFICIENT OF FRICTION.
22	SUILP(2)		FOR NIYPE = U - SUIL UNIT WEIGHT.
00	COLL D(2)		FOR NITYPE = 1 - SLUPE OF ELASTIC SUIL PENETRATION RELATIONSHIP.
22	SUILP(3)		FOR NITPE = $0 - 5011$ RELATIVE DENSITY.
22	NOLEC		FUR NITE - 1 - MAXIMUM SUL PRESSURE.
23	NULEG		
23	ILLU		ILEG = 0 - INVERTED TRIPOD.
			ILEG = 1 - CANTILEVER.
23	DRAGST		DISTANCE FROM CENTER BODY ATTACH POINT TO DRAG STRUT ATTACH
		1	POINT ALONG AXIS OF MAIN STRUT FOR CANTILEVER GEAR.
24	PFCMS(I)		MAIN STRUTTTH COMPRESSION PLASTIC LOAD LEVEL (SEE FIGURE 4-18)
25			MAIN STRUTTIH TENSION PLASTIC LOAD LEVEL (SEE FIGURE 4-18).

*NOTE: THE FOLLOWING ABBREVIATIONS ARE USED TO DEFINE THE COORDINATE SYSTEMS.

SCS - SURFACE COORDINATE SYSTEM.

LCS - LANDER COORDINATE SYSTEM.

GCS - GRAVITY COORDINATE SYSTEM.

FIGURE 5-20 INPUT DATA LANDING LOADS AND MOTIONS PROGRAM (Continued)

CARD NO.	INPUT Variable	COORDINATE SYSTEM*	VARIABLE DEFINITION
26	CDCMs(I)		
20 27			MAIN STRUTTH TENSION DI ASTIC STRUKE LIMIT (SEE FIGURE 4~10).
21	SPCMS(I)		MAIN STRUTTIN TENSION FLASTIC STRUKE LIMIT (SEE FIGURE 4-10).
20			MAIN STRUTTIN TENSION SPRING DATE (SEE FIGURE 4-10).
23	SCMXMS		MANYMIIM ALL OWARLE MAIN STRIIT COMPRESSION STROKE
30	STMXMS		MAXIMUM ALLOWADLE MAIN STRUT COMPRESSION STRUKE.
30	SRUCMS		MAIN STRUT COMPRESSION UNI OADING SPRING RATE
30	SRIITMS	1	MAIN STRUT TENSION UNI DADING SPRING RATE
31	IRETMS		MAIN STRUT THIS ON ONE ONDIG INDICATOR
51			1RETMS = $0 - 100 0$ AD ALONG SRUCMS OR SRUTMS.
			IRETMS = $1 - IINI OADING SPRING RATE OBTAINED FROM LOAD-STROKE$
		-	CURVE.
32	FRICMS		MAIN STRUT SLIDING FRICTION FORCE.
32	COEFMS		COEFFICIENT OF MAIN STRUT VELOCITY DEPENDENT FRICTION.
32	GAMMS		POWER OF MAIN STRUT VELOCITY DEPENDENT FRICTION.
32	AEI1		BENDING STIFFNESS (EI) OF UPPER SECTION OF CANTILEVER GEAR MAIN
			STRUT. (SEE FIGURE 5-5)
32	AEI2		BENDING STIFFNESS (EI) OF LOWER SECTION OF CANTILEVER GEAR MAIN
			STRUT (SEE FIGURE 55).
33	PFCDS(I)		DRAG STRUT ITH COMPRESSION PLASTIC LOAD LEVEL (SEE FIG. 4-18)
34	PFTDS(I)		DRAG STRUT ITH TENSION PLASTIC LOAD LEVEL (SEE FIG. 4-18)
35	CDCDS(I)		DRAG STRUT ITH COMPRESSION PLASTIC STROKE LIMIT (SEE FIGURE 4-18).
36	CDTDS(I)		DRAG STRUT ITH TENSION PLASTIC STROKE LIMIT (SEE FIGURE 4-18).
37	SRCDS(I)		DRAG STRUT ITH COMPRESSION SPRING RATE (SEE FIGURE 4-18).
38	SRTDS(I)		DRAG STRUT ITH TENSION SPRING RATE (SEE FIGURE 4-18)
39	SCMXDS	9	MAXIMUM ALLOWABLE DRAG STRUT COMPRESSION STROKE.
39	STMXDS		MAXIMUM ALLOWABLE DRAG STRUT TENSION STROKE.
39	SRUCDS		DRAG STRUT COMPRESSION UNLOADING SPRING RATE.
39	SRUTDS		DRAG STRUT TENSION UNLOADING SPRING RATE.
40	IRETDS		DRAG STRUT UNLOADING INDICATOR.
			IRETDS = $0 - UNLOAD$ ALONG SRUCDS OR SRUTDS.
			IRETDS = 1 - UNLOADING SPRING RATE OBTAINED FROM LOAD-STROKE
			CURVE.
41	FRICDS		DRAG STRUT SLIDING FRICTION FORCE.
41	COEFDS		COEFFICIENT OF DRAG STRUT VELOCITY DEPENDENT FRICTION.
41	GAMDS		POWER OF DRAG STRUT VELOCITY DEPENDENT FRICTION.
42	MODEIN		NUMBER OF ELASTIC MODES INPUT.
100	XOUT(I)	LCS	X COORDINATE OF ITH ACCELERATION OUTPUT POINT.
100	YOUT(I)	LCS	Y COORDINATE OF ITH ACCELERATION OUTPUT POINT.
100	ZOUT(I)	LCS	Z COORDINATE OF ITH ACCELERATION OUTPUT POINT.

*NOTE: THE FOLLOWING ABBREVIATIONS ARE USED TO DEFINE THE COORDINATE SYSTEMS.

SCS - SURFACE COORDINATE SYSTEM.

LCS - LANDER COORDINATE SYSTEM.

GCS - GRAVITY COORDINATE SYSTEM

FIGURE 5-20 INPUT DATA LANDING LOADS AND MOTIONS PROGRAM (Continued)

CARD NO.	INPUT VARIABLE	COORDINATE System*	
200	XFP(I)	LCS	X COORDINATE OF I TH FOOTPAD.
200	YFP(I)	LCS	Y COORDINATE OF ITH FOOTPAD.
200	ZFP(I)	LCS	Z COORDINATE OF I TH FOOTPAD.
300	XMSCB(I)	LCS	X COORDINATE OF CENTER BODY END OF I TH MAIN STRUT.
300	YMSCB(I)	LCS	Y COORDINATE OF CENTER BODY END OF ITH MAIN STRUT.
300	ZMSCB(I)	LCS	Z COORDINATE OF CENTER BODY END OF ITH MAIN STRUT.
400	XDSCB(I)	LCS	X COORDINATE OF CENTER BODY END OF I TH DRAG STRUT.
400	YDSCB(I)	LCS	Y COORDINATE OF CENTER BODY END OF I TH DRAG STRUT.
400	ZDSCB(I)	LCS	Z COORDINATE OF CENTER BODY END OF I TH DRAG STRUT.
500	GM(I)		GENERALIZED MASS OF ITH ELASTIC MODE. (EQUATION (5-13)
500	OMEGA(I)		FREQUENCY OF ITH ELASTIC MODE (Hz).
600	WNX(I)		GENERALIZED INERTIA PROPERTY Nyn FOR ITH ELASTIC MODE
			(EQUATION (5-13)).
600	WNY(I)		GENERALIZED INERTIA PROPERTY Nun FOR ITH ELASTIC MODE
			(EQUATION (5–13)).
600	WNZ(I)		GENERALIZED INERTIA PROPERTY N _{zn} FOR ITH ELASTIC MODE
			(EQUATION (5–13)).
700	PX(I)		GENERALIZED INERTIA PROPERTY P _{xn} for ith elastic mode
			(EQUATION (5-13)).
700	PY(I)		GENERALIZED INERTIA PROPERTY P _{vn} for ith elastic mode
			(EQUATION (5–13)).
700	PZ(I)		GENERALIZED INERTIA PROPERTY P _{zn} For ith elastic mode
			(EQUATION (5–13)).
800	PMSX(I,J)	LCS	X MODE SHAPE FOR ITH MAIN STRUT POINT IN J TH MODE.
900	PMSY(I,J)	LCS	Y MODE SHAPE FOR I TH MAIN STRUT POINT IN J TH MODE.
1000	PMSZ(I,J)	LCS	Z MODE SHAPE FOR ITH MAIN STRUT POINT IN JTH MODE.
1100	PDSX(I,J)	LCS	X MODE SHAPE FOR ITH DRAG STRUT POINT IN JTH MODE.
1200	PDSY(I,J)	LCS	Y MODE SHAPE FOR ITH DRAG STRUT POINT IN JTH MODE.
1300	PDSZ(I,J)	LCS	Z MODE SHAPE FOR I TH DRAG STRUT POINT IN J TH MODE.
1400	POUTX(I,J)	LCS	X MODE SHAPE FOR ITH ACCELERATION POINT IN JTH MODE.
1500	POUTY(I,J)	LCS	Y MODE SHAPE FOR ITH ACCELERATION POINT IN JTH MODE.
1600	POUTZ(I,J)	LCS	Z MODE SHAPE FOR ITH ACCELERATION POINT IN JTH MODE.
1700	PCGX(I)	LCS	X MODE SHAPE AT CENTER OF GRAVITY IN I TH MODE.
1700	PCGY(I)	LCS	Y MODE SHAPE AT CENTER OF GRAVITY IN I TH MODE.
1700	PCGZ(I)	LCS	Z MODE SHAPE AT CENTER OF GRAVITY IN I TH MODE.

*NOTE: THE FOLLOWING ABBREVIATIONS ARE USED TO DEFINE THE COORDINATE SYSTEMS.

SCS - SURFACE COORDINATE SYSTEM.

LCS - LANDER COORDINATE SYSTEM.

GCS - GRAVITY COORDINATE SYSTEM.

FIGURE 5-20 INPUT DATA LANDING LOADS AND MOTIONS PROGRAM (Continued)

case run time are presented. In addition, optional output data, as requested through input indicators, may be obtained. These consist of lander stability angle and pitching velocity, all of the individual integrated quantities when considering a flexible center body, and accelerations at points other than the center body center of gravity. In this latter case, the coordinates of the points at which the accelerations are desired are included as input data. The individual integrated quantities are of interest when considering a flexible structure because the normal output presents the combination of the rigid body and elastic motions. Examples of the output data obtained from the Landing Loads and Motions Program are presented in Appendix D.

The units of the output data are consistent with the system of units used in the input data. Only the angular quantities are output in the specific units of degrees.

An additional output option is available in the Landing Loads and Motions Program. This option allows the output of the landing gear strut forces and all of the time history variables required to determine the inertia loading throughout the center body structure. With this information, the internal loads of the center body structural members may be determined with the Center Body Option of the Structural Analysis Program. This data is output at points in time corresponding to the times when printed output is generated by the program. This option results when the input indicator NFORC (Card 8) is equal to 1 and the data is placed on magnetic tape for future use. The Center Body Landing Loads Program, which is used to retrieve this data from the tape and generate input information for the Structural Analysis Program, is discussed in Appendix E.

5.3.3 <u>Example of Program Operation</u> - A listing of the input data, and the resulting printed output obtained from the Landing Loads and Motions Program is presented in Appendix D. The output pages in this appendix present the input data, the beginning of the time history, and the last pages of the time history including the summary information. This data set corresponds to a computer run demonstrating correlation with the third drop test conducted at NASA Langley Research Center on the Task Order Three lander. For this example case, the fixed step, Runge-Kutta integration routine was used with output printed every 10 integration steps. An integration time step of 0.0001 sec with a total real time of 0.1 sec was requested. In addition to the normal printed output, the center body time histories and landing gear loads were output on magnetic tape for later use with the Structural Analysis Program. The retrieval of this information from the tape is discussed in Appendix E. Flexible center body input data was determined with the Structural Analysis Program, as discussed in Sections 4.3.1.3 and 6.2.1.

The lander had three inverted tripod gears with crushable main struts and elastic drag struts. The Secondary Soil Mechanics routine was used in

conjunction with a footpad attenuation system. Since the landing case considered was planar in nature, the unnecessary rigid body degrees of freedom were suppressed to conserve computer run time. The equations of motion for all the footpads were integrated during the complete run.

This run, including a flexible center body structure represented with three free-free modes, and the real time of 0.1 sec required 53 sec of CDC 6600 CP time. Computer runs for this length of real time required approximately 34 CP sec to run a case with a rigid center body.

Plots showing various time histories obtained from this run are given in Section 6.2.2. Also shown is a comparison of the results of this run with those obtained with a rigid center body.

6. DEMONSTRATION OF PROGRAM CAPABILITIES

Examples of the capabilities of the various developed computer programs are presented in the following sections. The Task Order Three lander configuration, shown in Figure 1-1, was chosen for many of these analyses since experimental data was available with which to compare the computer results. In general, the correlation between experimental data and the analytical studies was quite good. In addition to the planar landings of the Task Order Three Lander, results for spatial landings on different types of soil are discussed.

6.1 Large Displacement Analysis of Landing Gears - The Landing Gear Option of the Structural Analysis Program was employed to analyze typical inverted tripod and cantilever landing gear configurations. An analysis of the Task Order Three inverted tripod gear to determine drag strut loads versus stroke normal to a landing surface is presented. Predicted drag strut loads are correlated with loads obtained in the sixth drop test on the Task Order Three landing gear performed at NASA Langley Research Center. In addition, drag strut loads as a function of stroke are predicted for an arbitrarily oriented friction plane to indicate the influence on these loads of a nonsymmetrically loaded gear.

Analysis of the one-sixth scale LM cantilever gear to determine maximum energy absorption capability for various orientations of the landing surface is discussed in Section 6.1.2. Friction coefficient of the surface was varied to show the way in which friction affects energy absorption.

The fixed finite element idealization of inverted tripod and cantilever landing gears, employed in the Structural Analysis Program Landing Gear Option, was shown in Figure 4-17.

6.1.1 <u>Inverted Tripod Gear</u> - The inverted tripod gear of the Task Order Three Lander was analyzed to determine drag strut loads as a function of gear stroke. A listing of input information and output data for this problem is presented in Appendix C. Only selected pages of computer output are included due to the volume of output. Header card (see Figure 4-36) for this data case does not appear in the output listing although it was the first card of the data set.

Main strut honeycomb properties used for the gear are given in Figure 6-1. The Surface Coordinate System and Lander Coordinate System were assumed to be coincident, with the X_L and X_S axes normal to (and pointing out of) the landing surface (see Figure 4-20). Hence, the Euler angles ψ , θ , and ϕ are equal to 0. This condition corresponds to a "straight-in" drop of a lander such that the drag struts of a gear are loaded symmetrically. Behavior of the gear was investigated in 60 steps for an applied displacement of 15.24 cm normal to the landing surface. Friction coefficient of the surface was assumed to be .8. Footpad honeycomb crush forces of 1.102 x 10⁹ dynes and 4.18 x 10⁹ dynes and corresponding crush strokes of 1.27 cm and 4.45 cm were assumed.



FIGURE 6-1 LOAD - STROKE CURVE FOR MAIN STRUT OF INVERTED TRIPOD GEAR - TASK ORDER THREE LANDER

A correlation between drag strut loads obtained in the computer run (steps 1, 2, 9, 10, 59 and 60 shown in Appendix C) and loads obtained for this gear during the sixth leg drop test performed at NASA Langley Research Center is depicted in Figure 6-2. As illustrated, at the onset of stroking the experimental drag strut loads appear to build up instantaneously. This is due to large footpad inertia forces acting during initial footpad impact; thus, correlation of drag strut loads at the onset of stroking is poor since the program assumes that the struts are initially unloaded. Footpad inertia forces are negligible after a normal (X_S) stroke of about 1 cm and from this point on correlation is reasonably good. In the figure, only one curve is shown for predicted drag strut loads since the drag struts were loaded equally because of symmetry. This example indicates that the static analysis employed in the Landing Gear Option can be used to obtain an estimate of dynamic drag strut loads.



FIGURE 6–2 DRAG STRUT LOAD CORRELATION – INVERTED TRIPOD GEAR
For a nonsymmetrically loaded gear, the drag strut loads may be considerably higher than for the symmetrically loaded case. To emphasize this fact, the inverted tripod gear described above was analyzed with the program using the same input data with the exception that the landing surface was oriented such that $\psi = 25^{\circ}$ and $\theta = \phi = 0$. This condition can be visualized by referring to Figure 4-20 and imagining that the landing surface is rotated 25° about the Z_S direction. For this run, drag strut loads as a function of stroke normal to the rotated landing surface are shown in Figure 6-3. As shown, one drag strut is in tension while the other is in compression. The maximum drag strut load of 4.52 x 10⁹ dynes is considerably higher than the drag strut loads shown in Figure 6-2 for the symmetrically loaded gear.



FIGURE 6-3 ANTICIPATED DRAG STRUT LOADS FOR INVERTED TRIPOD GEAR

6.1.2 <u>Cantilever Gear</u> - The one-sixth scale LM cantilever gear was analyzed to determine maximum energy absorption capability as a function of landing surface orientation and friction coefficient.

Honeycomb crush properties assumed for the main strut and drag struts of the gear are shown in Figures 6-4 and 6-5. Various orientations of the landing surface (see Figure 4-20) were considered, each of which provided symmetrical loading of the drag struts. These orientations are described by the Euler angle combinations $\psi = 0^{\circ}$, $\phi = 0^{\circ}$, and θ varied from 0° to 40°. The maximum value of 40° is the sum of the maximum ground slope relative to the local horizontal (30°) and the maximum pitch-up of the lander (10°) specified in typical design constraints, Section 3.1. Behavior of the gear was investigated by applying a displacement of 30.48 cm, normal to the landing surface, in 100 steps. This large value of normal displacement was selected to insure that the gear bottomed out before all steps were taken. For each coefficient of friction considered, a series of computer runs, were made in which θ was varied from 0° to 40°. Friction coefficients of 0.2, 0.8, and 1.0 were considered.

Maximum energy absorbed by the gear in a direction normal to the landing surface (X_S direction) is presented in Figure 6-6. Each point in the figure represents the total energy absorbed by the gear before the main strut bottomed out. Curves for friction coefficients of 0.8 and 1.0 are the same because in each case the friction was sufficient to prevent the footpad from sliding. For the friction coefficient of 0.2, the curve includes the effect of footpad sliding on the landing surface. As evidenced in this figure, energy absorption capability of the gear is minimum for values of θ between 20° and 30°.

6.2 <u>Analysis of Task Order Three Lander</u> - A complete analysis of a typical legged lander configuration is discussed in the following sections. This study consisted of performing a modal analysis on the center body structure employing the Center Body Option of the Structural Analysis Program. This modal data was then incorporated in the Landing Loads and Motions Program to account for the effects of a flexible center body on the landing response of the vehicle. Output from the landing program was used, in conjunction with the Center Body Landing Loads Program, to determine the distribution of inertia, gravity, and landing gear strut loads throughout the center body. These loads were then input to the Center Body Option of the Structural Analysis Program to determine internal member load distributions.

The legged lander considered was the 3/8 mass version of the Task Order Three lander. Figure 6-7 shows the structural configuration of this vehicle. Included in following analysis of this lander are examples of correlation between predicted results obtained with the Landing Loads and Motions Program and the results of the third drop test conducted at NASA Langley Research Center.



FIGURE 6-4 LOAD-STROKE CURVE FOR MAIN STRUT OF ONE-SIXTH SCALE LM CANTILEVER GEAR



FIGURE 6-5 LOAD-STROKE CURVE FOR DRAG STRUT OF ONE-SIXTH SCALE LM CANTILEVER GEAR



FIGURE 6-6 MAXIMUM ENERGY ABSORPTION FOR ONE-SIXTH SCALE LM CANTILEVER GEAR



FIGURE 6-7 GEOMETRY OF TASK ORDER THREE LANDER

6.2.1 <u>Modal Analysis of Center Body Structure</u> - The finite element idealization of the center body structure is shown in Figure 6-8. The member stiffness properties for each pair of bars representing a side beam (for example, members 10 and 11) and a radial beam (members 48 and 49) are such that the total stiffness of these deep beams is maintained. The moment carrying capabilities of members 46, 47, 48, 49, 50, and 51 about the lander X axis, where these members attach to the side beams, were removed. This was done to simulate the high flexibility associated with these attach points.

This idealization resulted in a stiffness matrix containing 33 joints with 198 degrees of freedom. Before obtaining the structure's frequencies and mode shapes, the stiffness matrix was reduced to 99 degrees of freedom by removing the three rotational degrees of freedom at each joint. Thus, the input mass matrix contained translational masses associated with all of the joints shown in Figure 6-8.

Twenty elastic modes were requested and the program output for this modal analysis is shown in Appendix B. Three modes, with frequencies of 34.6 Hz, 48.2 Hz, and 101 Hz were obtained whose modal deformation patterns were predominately in the lander Y-Z plane and are the type of modes that would be excited by a landing such as the third drop test of the Task Order Three lander. These are the modes which were included in the data for the Landing Loads and Motions Program, Section 6.2.2, when considering the effects of a flexible center body structure. Figures 6-9 through 6-11 show the mode shapes associated with these three frequencies.

6.2.2 <u>Correlation with Third Drop Test</u> - The predicted landing response for the initial conditions corresponding to NASA Langley's third drop test were obtained considering both a rigid and flexible center body structure. The initial conditions for this drop test condition are shown in Figure 6-12. A portion of the output from the Landing Loads and Motions Program for the flexible body is presented in Appendix D.

The main strut load-stroke relationship shown in Figure 6-1 was used for these correlation runs. A sketch of the actual footpad geometry, including footpad attenuation, and the footpad idealization employed in the landing analyses are shown in Figure 6-13. The Secondary Soil Mechanics routine was

employed with the following soil parameters:

Soil elasticity constant = 1.4×10^7 dynes/cm³ Coefficient of friction = 0.9

The distribution of center body mass and inertia and footpad masses used for the 3/8 mass Task Order Three lander in these runs is summarized in Figure 6-14. This breakdown was obtained by requiring that the combination of the center body mass moments of inertia and the footpad masses, multiplied by the square of their respective moment arms, approximated the moments of inertia of the total 3/8 mass Task Order Three lander. In addition, the center of gravity location for the total lander was maintained with this mass distribution.

Predicted center of gravity acceleration time histories and loads in the outside drag strut of leg 2 for a rigid center body are compared with the experimental data in Figures 6-15 and 6-16. The high frequency oscillations present in both the predicted load and acceleration results are the spring-mass effect of the footpad mass in conjunction with the landing gear struts. In addition, oscillation in the drag strut load near the end of the time history (when the footpad is off the ground) is the gear-mass system experiencing free vibration.

Comparison between measured and predicted center of gravity accelerations, main strut loads, main strut strokes, and drag strut loads when center body flexibility was included, is shown in Figures 6-17 through 6-22. Correlation between analytical and experimental results is good in this case. The inclusion of center body flexibility results in a less rapid buildup of center of gravity accelerations upon initial impact, especially in the Z direction. In addition, the inclusion of structural flexibility can readily be seen by the low frequency oscillation present in the center of gravity Z acceleration.

An interesting point is indicated in Figure 6-22. It can be seen that there is an initial oscillatory load in this drag strut before the footpad impacts the landing surface. This is the result of the inertia loading of the footpad mass on the end of the elastic landing gear struts. This inertia loading effect of the noncontacting footpad is also indicated in Figure 6-18 by the buildup of main strut load in leg 1 before footpad impact.



FIGURE 6-8 CENTER BODY STRUCTURAL IDEALIZATION



FIGURE 6-9 CENTER BODY MODE SHAPE ASSOCIATED WITH 34.6 H_z mode



FIGURE 6-10 CENTER BODY MODE SHAPE ASSOCIATED WITH 48.2 H_z MODE



FIGURE 6-11 CENTER BODY MODE SHAPE ASSOCIATED WITH 101 $\rm H_{Z}$ Mode



FIGURE 6-12 INITIAL CONDITIONS FOR DROP TEST THREE



IDEALIZED FOOTPAD

FIGURE 6-13 FOOTPAD IDEALIZATION

ITEM	MASS (grams)	X * (CM)	Y (CNI)	Z (CM)	I _{xx} (gram-cm ²)	^I yy (gram-cm ²)	I _{zz} (gram–cm ²)
CENTER BODY	180,500	67.97	0	0	8.031 x 10 ⁸	4.992 x 10 ⁸	4 .96 8 x 10 ⁸
FOOTPAD 1	3,497	11.43	0	144.78	-	-	-
FOOTPAD 2	3,497	11.43	-125.37	- 72.39	-		-
FOOTPAD 3	3,497	11.43	125.37	- 72.39			-

* RELATIVE TO BOTTOM OF FOOTPAD

FIGURE 6-14 3/8 MASS TASK ORDER THREE LANDER MASS BREAKDOWN





FIGURE 6-16 OUTSIDE DRAG STRUT LOAD WITH RIGID CENTER BODY













4 9-3

5.

FIGURE 6-22 LEG 1 DRAG STRUT LOAD WITH FLEXIBLE CENTER BODY

6.2.3 <u>Center Body Landing Loads</u> - Observing the center of gravity acceleration time histories shown in Figure 6-17, it is seen that there are several points of apparent high acceleration of the center body structure. The time history data stored on magnetic tape during this run was retrieved with the Center Body Landing Loads Program and total load distributions on the center body were determined at several of these points during the landing. The load distribution at the landing time of 0.011 seconds was selected as critical because it yielded the largest loads in the -Z direction at the center body center of gravity. Appendix E presents the load distribution obtained with the Center Body Landing Loads Program for this time.

6.2.4 <u>Center Body Internal Load Distribution</u> - The Center Body Option of the Structural Analysis Program was employed to determine joint displacements and internal load distributions in the center body structure for landing loads defined in Section 6.2.3 at a time of 0.011 seconds. The center body idealization shown in Figure 6-8 was employed. A listing of input information and output data for this problem is presented in Appendix A. Header card (see Figure 4-32) for this data case does not appear in the output listing although it was the first card of the data set.

In the center body finite element idealization, attach points for legs 2 and 3 were assumed to be joints 12, 15, 17, 22, 25, and 27. This is an approximation since the physical attach points for the drag struts are located at the ends of clevis fittings which are adjacent to the above joints. The center body was supported at joints 12, 15, 17, 22, 25 and 27 which were assumed to be pinned with zero translational displacements and zero applied moments. The center body landing loads distribution at 0.011 seconds is given in Appendix E. Due to the assumed drag strut locations in the center body idealization, unbalances in these loads are indicated, especially moment about the lander Y axis. To insure a static balance, the negative of these unbalance loads were applied at the center of gravity (joint 33) in addition to the three force components at this joint. For computer input, nominal values were selected by default for all data option indicators except INDITR and TMAX. INDITR (maximum number of iterations allowed) was specified as 100.

Included in computer output are displacements and external loads for

all joints and internal loads in all bar elements. The displacements at the center of gravity and the internal bending moments about the local Z axis for the radial beam (elements 46 through 51) are of particular significance. Results of this analysis are compared with the results for a redesigned center body structure in Figure 6-28, Section 6.3.

In order to emphasize the importance of proper idealization of the structure when determining displacements and internal loads, the loading condition defined above was applied to a refined idealization of the Task Order Three Lander Center Body as discussed in Appendix G. Displacements and internal loads obtained when employing the refined idealization are different than those obtained when employing the idealization in Figure 6-8 and probably are more accurate. For the purpose of demonstrating program capabilities, the idealization shown in Figure 6-8 is entirely adequate and will be employed in the following sections.

It must be kept in mind that a more refined idealization requires more computer run time. For example, run time for the idealization in Figure 6-8 was 46 CP seconds while that for the refined idealization in Appendix G was 106 CP seconds. A compromise between idealization refinement and computer run time must be made.

6.3 <u>Analysis of Modified Task Order Three Lander</u> - The analysis of the Task Order Three lander, Section 6.2, indicated considerable flexibility of the center body structure in the lander Y-Z plane. This inplane flexibility is indicated in the measured Z center of gravity acceleration, Figure 6-15, obtained during the NASA drop test program.

A modification of the Task Order Three lander center body structure to increase the stiffness and improve the load carrying capability in the Y-Z plane was studied. Three configurations, incorporating the addition of a number of tension straps, were considered in arriving at the Modified Task Order Three lander. These consisted of 9-strap, 12-strap, and 18-strap configurations. The 12-strap configuration is shown in Figure 6-23. The 9-strap configuration was the same as the 12-strap configuration with the upper straps 63, 64, and 65 removed. The 18-strap configuration was like the 12-strap with six additional straps added to the upper portion of the center body. These upper straps were located in a manner similar to the lower straps 57 through 62.



FIGURE 6-23 CENTER BODY IDEALIZATION INCLUDING TENSION STRAPS

A representative loading condition was applied to a finite element idealization of these three strap configurations employing the Center Body Option of the Structural Analysis Program. For this study, the basic center body structure was idealized as shown in Figure 6-8. Each set of tension straps was then added to this idealization, as indicated in Figure 6-23 for the 12-strap configuration. In each case, all straps were assumed to have a cross-sectional area of 1.29 square centimeters and a modulus of elasticity of 2.068 x 10^{12} dynes/cm².

Figure 6-24 shows the loading condition which was employed to evaluate each of the tension strap configurations. This condition was found to be the critical one when the most severe center of gravity accelerations noted during the analysis of the Task Order Three lander in Section 6.2 were combined for one and two leg landings. Employing the criterion of reducing both the radial beam bending moments about the local Z axes and inplane displacements, the comparison shown in Figure 6-25 was made for the three tension strap configurations. Based on the results shown in this figure, the 12-strap configuration was chosen as the Modified Task Order Three lander. This configuration resulted in a 3.1 reduction factor for radial beam bending moment and a 5.45 multiplication factor for center body inplane stiffness when compared to the Task Order Three lander structure. The 18-strap configuration was not selected because the slight improvement in the parameters of interest was not sufficient to justify six additional straps. The weight of the twelve tension straps was removed from the ballast weights in the idealization of the modified Task Order Three configuration to maintain the same total lander weight.



LOADS APPLIED AT CENTER OF GRAVITY WERE DETERMINED BY ASSUMING:

+ X ACCELERATION OF 50 g's + Z ACCELERATION OF 36 g's

AUCELERATION OF JUES

WEIGHT OF HUB AND RADIAL BEAMS = 1.196×10^8 DYNES, CONCENTRATED AT CENTER OF GRAVITY

THEREFORE, LOADS SHOWN ARE:

 $F_X = -m\ddot{X} = -5.983 \times 10^9$ DYNES $F_Z = -m\ddot{Z} = -4.315 \times 10^9$ DYNES $M_Y = MOMENTS$ SUMMED ABOUT CONTACTING FOOTPAD = -1.148 × 10¹² CM-DYNES

FIGURE 6-24 LOADING CONDITION FOR EVALUATION OF MODIFIED TASK ORDER THREE LANDER CENTER BODY STRUCTURE

. STRAP CONFIGURATION	MAXIMUM RADIAL BEAM TRANSVERSE BENDING MOMENT* (CMDYNES ULT.)	CENTER BODY INPLANE STIFFNESS** (DYNES/CM)		
NONE	8.203 x 1010	1.601 x 10 ⁹		
9	3.378 x 10 ¹⁰	5.366 x 10 ⁹		
12	2.649 x 10 ¹⁰	8.748 x 10 ⁹		
18	1.625 x 10 ¹⁰	9.159 x 10 ⁹		

* BENDING MOMENTS ARE ABOUT LOCAL Z AXIS OF ELEMENT AT "Q" END

** CENTER BODY INPLANE STIFFNESS WAS OBTAINED BY APPLYING A UNIT LOAD AT THE CENTER OF GRAVITY IN THE -Z GLOBAL DIRECTION. THE STIFFNESS FACTOR IS THIS LOAD DIVIDED BY THE RESULTING CENTER OF GRAVITY GLOBAL Z DISPLACEMENT

FIGURE 6–25 COMPARISON OF VARIOUS TENSION STRAP CONFIGURATIONS

Employing the finite element idealization shown in Figures 6-8 and 6-23, a modal analysis was performed on the Modified Task Order Three lander center body. Three modes, with frequencies of 75.2 Hz, 149 Hz, and 163 Hz, were obtained whose modal deformations were predominantly in the lander Y-Z plane. These modal frequencies are much higher than the comparable frequencies obtained for the Task Order Three Lander, Section 6.2.1, thus illustrating the stiffening effect of the straps.

The drop test three landing conditions were used to evaluate the Modified Task Order Three lander center body structure. Using the above three modes to represent the center body flexibility effects, the accelerations and drag strut load shown in Figures 6-26 and 6-27 were predicted with the Landing Loads and Motions Program. These quantities are superimposed on the predicted results obtained for the original Task Order Three lander center body. The oscillatory nature of the Z acceleration is due to the large excitation of the 75.2 Hz mode for the initial conditions considered.





FIGURE 6-27 OUTSIDE DRAG STRUT LOAD WITH MODIFIED CENTER BODY STRUCTURE

The distribution of landing loads throughout the center body structure was obtained for a number of points during the landing time history. At a real time of 0.011 seconds, the center body experienced high loads and this point was chosen for analysis with the Center Body Option of the Structural Analysis Program. Employing the center body finite element idealization shown in Figures 6-8 and 6-23, internal member loads and joint displacements at this time were determined. A comparison of these loads and displacements with the loads and displacements obtained for the Task Order Three lander, Section 6.2.4, are presented in Figure 6-28. As can be seen, the radial beam bending moments in the Modified Task Order Three center body are lower than those obtained for the Task Order Three Center body. In addition, the global Z displacement of the center body center of gravity is less for the Modified Task Order Three center body. This trend of lower member loads and reduced displacements is a result of the additional center body load paths and increased stiffness provided by the tension straps.

6.4 <u>Landing Analysis of Full-Mass Task Order Three Lander</u> - Several additional studies for the Task Order Three lander were performed. For all of the following discussions, the structural configuration, landing gear properties, and footpad idealization were the same as discussed in Section 6.2. The mass properties were that of a full mass Task Order Three lander and a local acceleration of gravity of 375 cm/sec² was employed.

A lander, horizontal with respect to the local acceleration of gravity, with a vertical velocity of -610 cm/sec along the gravity vector, and landing on a 10° slope, was considered. The initial orientation of the lander was such that a single footpad impacted the landing surface. For comparison, three short runs were made with these initial conditions and the following soil properties.

SS Dense Soil (Figure 5-10):

Primary soil mechanics routine Relative soil density = .69 Soil unit weight = 1681 dynes/cm³ Soil internal friction angle = 39°

```
SS Loose Soil (Figure 5-10):
    Primary soil mechanics routine
    Relative soil density = 0
    Soil unit weight = 1488 dynes/cm<sup>3</sup>
    Soil internal friction angle = 29°
Secondary Soil:
    Secondary soil mechanics routine
    Soil elasticity constant = 144 x 10<sup>7</sup> dynes/cm<sup>3</sup>
    Coefficient of friction = 0.9
```

The Secondary Soil is the same soil idealization as that employed during the correlation studies conducted for the third drop of the Task Order Three lander discussed in Section 6.2.2.

	JOINT 33 DISPLACEMENT (CM)			
	Х	Ŷ	Z	
TASK ORDER THREE LANDER (SECTION 6.2.4)	-9 . 577 x 10 ⁻²	-1.322 x 10 ⁻⁴	-4.289 x 10 ⁻¹	
MODIFIED TASK ORDER THREE LANDER	-4 . 794 x 10 ⁻²	–1.555 x 10 ^{–4}	-7.834 x 10 ⁻²	

	RADIAL BEAM BENDING MOMENT ABOUT LOCAL Z AXIS AT ''Q'' END (CM - DYNE)					
	ELEMENT 46	ELEMENT 47	ELEMENT 48	ELEMENT 49	ELEMENT 50	ELEMENT 51
TASK ORDER THREE LANDER (SECTION 6.2.4)	8.652 x 10 ⁹	8.022 x 10 ⁹	-3.648 x 10 ⁹	-8.028 x 10 ⁹	5.625 x 10 ⁵	-3.108 x 10 ⁶
MODIFIED TASK ORDER THREE LANDER	1.863 x 10 ⁹	1.573 x 10 ⁹	-1.869 x 10 ⁹	-1 . 611 x 10 ⁹	1.818 x 10 ⁷	3.716 x 10 ⁶

FIGURE 6-28 COMPARISON OF COMPUTER RESULTS FOR ORIGINAL AND MODIFIED TASK ORDER THREE LANDER

Figure 6-29 presents a comparison between the resulting main strut time histories for the three soils considered. As can be seen, the largest main strut load and stroking motion resulted when the Secondary Soil was used. This is reasonable since these soil properties were used to idealize the very rigid landing surface of the Task Order Three lander drop test program. The main strut stroke was much less for the SS Loose Soil than for the other soils.

A second set of initial conditions for the full mass lander were as follows:

Ground slope = 10°

Initial rotation about $X_{gaxis} = 20^{\circ}$

Vertical velocity = -610 cm/sec

For this run, the SS Dense Soil employed in the previous case was used.

Figures 6-30 and 6-31 present the accelerations and main strut loads resulting from this spatial landing. The spatial nature of this landing is obvious from these time histories. The main struts in each leg load up sequentially as their respective footpads impact the landing surface. These loads then drop off as the lander's initial kinetic energy is absorbed by the crushable struts.





FIGURE 6-30 FULL-MASS TASK ORDER THREE LANDER CENTER OF GRAVITY ACCELERATION TIME HISTORIES


THIS PAGE INTENTIONALLY LEFT BLANK

7. CONCLUSIONS

Two methods of analysis and associated computer programs were developed for the investigation of legged landers. One program, the Landing Loads and Motions Program, is used to predict landing loads and spatial motions of a lander. The second, the Structural Analysis Program, determines the internal load distributions in the lander center body structure or frequencies and mode shapes for a free-free center body structure. Energy absorption characteristics of an individual landing gear are also determined with this program. Each program contains options and capabilities which were shown to be important in the analysis of legged lander configurations. For example, in the Landing Loads and Motions Program, options to include the effect of center body flexibility and a comprehensive soil mechanics routine were included. An option to investigate large displacement stroking behavior of a landing gear was incorporated in the Structural Analysis Program.

In addition to the limited analyses made within the constraints of Task Order Five, comparisons of experimental and analytical results for ten additional unpublished tests conducted at the NASA Langley Research Center show that the programs are working satisfactorily. However, the effects of friction between the footpads and the landing surface are very difficult to simulate analytically because of the uncertainties involved in defining these effects during dynamic model testing. The limited analyses presented in this report do serve to demonstrate primary program capabilities. As a result of these analyses, the following conclusions are possible:

(1) Center body flexibility significantly affects landing loads and motions. This was illustrated during the landing studies conducted on the Task Order Three lander, Section 6.2.2. For a rigid center body, the drag strut loads (Figure 6-16) were higher than with a flexible center body (Figure 6-20). In addition, the center of gravity accelerations in the rigid center body (Figure 6-15) built up more rapidly than was indicated in either the test data or in the analysis with flexibility included (Figure 6-17). Very good correlation with test data resulted when center body flexibility was included.

- (2) Soil properties significantly affect landing gear strut behavior. In Figure 6-29, it was shown that much higher landing gear strut strokes and loads result when landing on hard soil. When landing in loose sand, strut stroke increased gradually with time with resulting lower strut loads. Overall lander motions are highly dependent upon soil properties. Therefore, it is important that programs for studying landing motions, include the capability for properly representing soil characteristics.
- (3) Care must be exercised in the finite element idealization of the center body structure. As discussed in Section 6.2, the predicted internal load distributions were different for two different idealizations of the same structure. However, it was also shown that the first idealization presented in Section 6.2 resulted in very good center body flexibility data for use with the Landing Loads and Motions Program. In general, a more sophisticated idealization is required for determination of internal loads than for generation of center body flexibility information for the landing program.
- (4) Landing gear orientation and landing surface friction greatly influence the large displacement stroking behavior of a landing gear configuration. Comparison between Figures 6-2 and 6-3 indicate that the drag strut loads in an inverted tripod gear are much different for two different gear orientations. As shown in Figure 6-6, the energy absorption capabilities of a landing gear are dependent on the coefficient of friction employed and the orientation of the gear with respect to the landing surface.
- (5) Increasing center body stiffness does not necessarily reduce peak center of gravity accelerations. As indicated in Figure 6-26, the peak predicted accelerations for the original Task Order Three center body and the Modified Task Order Three center body were of the same magnitude. The increased stiffness of the modified structure resulted in a higher frequency oscillation of the Z acceleration. However, in this case, the individual member loads were reduced due to the additional load paths available in the structure.

8. REFERENCES

- Otto, O. R., Melliere, R. A., and Lopatin, A.: "Design and Fabrication of a Full-Size Landing Impact Test Model of the Mars Legged Lander Configuration," NASA CR-111771, September 1970.
- Otto, O. R., Dorr, D. J., Laurenson, R. M., Burton, D. J., and Moore, R. L.: "Analyses and Limited Evaluations of Payload and Landing System Structures for the Survivable Soft Landing of Instrument Payloads," NASA CR-66914, April 1970.
- Serpanos, J. E.: "A User-Coded Matrix Generator for the Displacement Method - Format II Second Version of Fortran Matrix Abstraction Technique," Air Force Flight Dynamics Laboratory Technical Report 66-207, Volume IV, March 1967.
- 4. Varga, R. S.: <u>Matrix Iterative Analysis</u>, Prentice Hall, Inc., Englewood Cliffs, N. J., 1962.
- Hestenes, M. and Stiefel, E.: "Method of Conjungate Gradients for Solving Linear Systems," Report 1659, Nation Bureau of Standards, 1952.
- 6. Hurty, Walter C. and Robinstein, Moshe F.: <u>Dynamics of Structures</u>, Prentice-Hall, INC., Englewood Cliffs, N. J., 1964, pp 43-45.
- 7. Young, Ronald and McCallum, Helen: "The Determination of Frequencies and Modes of Undamped Structures Using Finite-Element Stiffness and Consistent-Mass Matrices," Aeroelastic and Structures Research Laboratory, Massachusetts Institute of Technology, Cambridge, Massachusetts, June 1965.
- 8. Martin, H. C.: "On the Derivation of Stiffness Matrices for the Analysis of Large Deflection and Stability Problems," Proceedings of the Conference on Matrix Methods in Structural Mechanics, TR-66-80, 1966, Wright-Patterson AFB, Ohio.
- 9. Thomson, William T.: <u>Vibration Theory and Applications</u>, Prentice-Hall Inc., Englewood Cliffs, N. J., 1965, pp 294-299.
- Bisplinghoff, Raymond L., Ashley, Holt, and Lafman, Robert L.: <u>Aeroelasticity</u>, Addison-Wesley Publishing Company, Inc., Cambridge, Massachusetts, 1955, pp 632-635.

- 11. "Summary Report Lunar Module (LM) Soil Mechanics Study," Report #AM-68-4, Bendix Energy Controls Division, South Bend, Indiana, November 1968.
- 12. Anderson, R. G.: "Application of LM Soil-Footpad Interaction Model to Surveyor Landing Dynamics Simulation," Paper 68K11, Bendix Energy Controls Division, South Bend, Indiana, November 1968.

APPENDIX A

EXAMPLE OF INPUT AND OUTPUT

CENTER BODY OPTION

STRUCTURAL ANALYSIS PROGRAM

	MASTER	STRUCTURAL ANALYSIS PROGRAM LEGGEJ LANDER 1 Agreement, comfract nas1-0137, task order nunger five McDonnell dojglas Astronauti's company, east	
STRUCTURAL ANALYSES	DATA - CARJ SODE		
	PLANK - 0 1.	COMMENTS NDDAL PDINT DEFINITIONS	
	N M.	REFERENCE POINTS NODAL POINT RESIRATINT DEFINITIONS	
	ייים דייים	FORCE VESTORS MOMENT VESTORS	
	2 4 8	SHEAR PANEL GEFINITIONS Formated-data terminator	
* * * * * * *	* * * * * * * *	· · · · · · · · · · · · · · · · · · ·	
JCINI INFO * * * * * * * * *	RMATION CARDS * * * * * * *	* * * * * * * * * * * * * * *	a angen ou anna anna an a
1 1 13.5692 31. 1 2 -28.6573 31.	7246 85.7758 7246 85.7758	-1 0	
1 3 13.5692 0.	85.7758 85.7758	1 P	and a second
	113-536	- 19 v	
	•7240 80.1758	G F	
1 8 13.5890 -60 1 9 -17.0074 -60	•9345 35•1790		
1 10 -28.0679 -50 1 11 13.589, -00	-9340 35.1790 -1445 -15 4170	27	
1 12 -28.(570 -90	•1445 -15•4173 -1 -1 •1445 +15•4173 -1 -1	-1 11	
1 13 13,5830 -74 1 14 -28.6573 -74	•2950 -42•8752 •295 -42•8752	12 13	
1 15 29,4834 -98 1 16 13,5830 -58	.3234 -56.7590 -1 -1 42 -70 358		
1 17 -28.6673 -58	•42 -70.358 -1 -1	-1	
1 10 -17.9030 0. 1 10 -17.9070 0.	-70.358	15 16	
1 21 -28.(67 0. 1 21 13.589 58.	42 -70.358 42 -70.358	17 18	
1 22 -28.667 59.	42 -70.358 -1 -1	-1	
1 23 13.089 74. 1 24 -28.057 74.	295 -42.8752 295 -42.8752	19 20	
1 25 29.4634 98. 1 26 13.589 9J.	3234 -56,769 -1 -1 1446 -15,4173	-1 21	
1 27 -28.067 93. 1 28 13.389 59.	1446 -15.4178 -1 -1 9346 35.179	-1 22	
1 29 -17.9373 54.	9346 25.179	23	
1 3L -28.667 53. 1 71 13.589 8.	9346 35.179 [.	24 25	i
1 32 -17.907 D. 1 33 6.	• =	25 27 1	
	* * * * * * * * *	* * * * * * * * * * * * *	
FORCE ANU * * * * * * *	MOME (I LIMLI CARUS * * * * * * * * * * *	* * * * * * * * * * * * * * * * * *	
* * * * * * * * * * * *	* * * * * * * * * * * * * * * * * * *	* * * * * * * * * * * * * * * *	ļ
* * * * * * * *	FORC: VECTOR CARDS * * * * * * * * * *	******	
4 1 1.8C1F+07-7.22 4 2-1.095F+u8-1.27 4 3 1.737F+97 2.36	TE+23 +.612E+07 3E+03 3.736E+08 4E+05 2.559E+07		
4 4 1.133E+07-1.54	9E+45 t++03E+C7		

	And and a set of the s						And a province of the second				and a subjectivity to and many or a real rate of a definition of the set of t	₩								A MANANA A A MANANA A																				-		
														* * * *	* * * * *	*	•		2.624E11	2.62JE11	2.620E11	2.620E11 2.620F11	2.62JE11	2.620E11	2.62JE11 2.62JE11	2.620E11	2.620E11	2.620E11	2.620E11 2.620E11	2.620E11 2.620F11	2.620E11	2.620E11 2.620E11	2.620E11	2.620E11	2.620E11	2.620E11 2.620E11	2.620E11	2.620E11	2.620E11	2.620E11	2.620E11	2.620E11
														*	* * *	*	• • •	*	6.820511	6.826E11	6.825E11	6.826E11 6.826F11	6.826E11	6.826E11	6.826E11 6.826F11	6.826E11	6.826E11 -	6.826E11	6.826E11 6.826E11	6.826E11 6.826E11	6.826E11	6.820E11 6.826F11	6.825E11	6.826E11	6.826E11	6.826E11 6.826E11	6.826E11	6.826E11	6.826E11	6.826E11	0.020511 6.826511	6.826E11
														***	* * * *	* * * * *	• • •	* * * *	1.5775	3.3299	6.4557	1.0281	2.1644	3.3299	1.2445	1.2445	2.7324	1.2445	1.5775	2.1644	6.4557	1.0281 .6243	2.1644	3.3299	1.2445	1+2445	2.1977	1.2445	1.5775	2.1644	5. 4557	1.0281
								:						* * * * *	* * *	* *	•	*	158.13	210.43	4.87	2•22 3.68	274.90	730.43	168.10	2.0104	109-46-	4011.2	4010.2 168.10	274.90	4.87	2,22 7,68	274.91	700.43	4310.2	4013.2	27 . 46	4010.2	158.10	274.91	1 u u • 4 0	2.22
5.892E+07 4.572E+07	3.742E+09 2.383E+06	7.265E+05 1.003E+07	2.372E+07	1.50JE+07 3.324F+07	1.254E+05	1.4796+08	1.052E+07	1.6335+05	1. 3686+87 3. 3676+87	2.393E+07	7.221E+06	9.952E+06	1.29JE+08	VECTOR C	* * * *	2.753E+C5	CARDS	*	105.70	435.21	113.95	55.53 6.98	111.43	435.21	106.70 29.58	29.58	255.28	29+58	29.55 106.70	111.43	110.95	55.53 6.08	111.43	+35.21	29.53	29.58 265 28	253.57	29.58	106.70	111.43	1111.95	55.53
.319E+65 .510E+05	.265E+04 .733E+17-1	-738E+U7	• 670E+0/	.191E+07	.351E+05	.153E+0	- 8665+84-	•1235+up	.1995+U/ .716E+Jo	• 960E+07	- 140E+07	. 224E+07 . 454E+05-		ED MOME IT	75 36 36 36	-327E+03-	DRMATION	*:	11.7742	24.8322	14.1123	7.6613 4.8387	16.129	24.8322	11.7742	9.2774	20+729	10.3140 9.2774	9.2774 11.7742	16.129	14.1123	7.6613	16.129	24.8322	9.2774	9.2774	16.3742	9.2774	11.7742	16-1290	14.1123	7.6613
-16 2	- 180-	90. 90. 90.		- 12	11.69	-9-1- 	2	1 20-		2-204	-90-1	+ 10 6 - 3		CIFI	*	5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		*	m	0.2	-	-1 ^	i s:		~ M	t. (49	2 E	4 10	6 0 c	13	12	3 80	20	: 12	3 U	95	M :	+ +	81		17
31.0F.	186F. 122E1	3954	0.55	ш. 1971 1971	10.1	17E	575	1415	16 E	1010	186E	335	1991 1991	*	*	8.	946		~ •	t n	÷	о и	n (n)	~	rα	1	ol (34	2121	2	14	ហ ដ មា ម	19	17	- 60 - 19	20	5.5	55	22	22	+. +. + 10 v	S ²
τ.9 1.8	⊃6. 	-1-8	- 2 - 2 - 2	40 G 6 M	9.7-		0 6 1 1	6-1 6-1	ະ ທີ່ ທີ່	- 2 -	9 0 • N	1.1	10 U	*	*	1 1 1 1	•	н *		4 0	M)	ьw	• ••>	t.	ი თ. -	~	au (т - со	0 न न न		15	20 - F	1 M)	44	19 19	117	3 6 T	1.8	12	10	л й И М	23
u u	r ø	σċ	1		- - - - -	41 4 6 4		810	л Э н Ni	21	u ið V N	й E Л Ю	200			-			-+ c	410	4	5 1	1	6 0 (0 -	11	4	4 - 1 - 1	4 4 0 0	11	19	26	52	200	1" in V iv	26	- 80 - 64	52	4 F)	010	0 4 0 M	35
\$ 3.	* *	33	÷		÷+	A 18	* .*		* *		+.+	A 16	+ + + +			10			. .	^ . ^		· • • •					· • ·	.						10.14	n . n	10.1	o .o	.0.	0 0		o .o	

	And a contract of the second at the second a	, san da						····································									and a second a second a second a											
.6243 6.826E11 2.620E11 0 2.1644 5.826E11 2.620E11	3 3,3299 6.826E11 2.620E11	7 1.5775 6.826E11 2.620E11	2 1.2445 6.826E11 2.620E11 2 1.2445 6.826F11 2.620F11	6 2.7804 6.826E11 2.620E11	2.1977 6.826E11 2.620E11	Z 1.2445 6.826E11 2.620E11 2 1.2445 6.826E11 2.620E11	2 1.0822 6.826E11 2.620E11	2 1.0822 6.826E11 2.620E11	Z 1.8522 6.826E11 Z.620E11 Z 1.0822 6.826E11 Z.620E11	2 1.3822 6.826E11 2.620E11	2	3 5551.7 5.825511 2.620511 3 5551.7 5.826511 2.620511	* * * * * * * * * * * * * * * * * * *						* * * * * * * * * * * * * * *				·				e de la composition de	an and a grant of the second se
1387 5.98 3.59 129 111.43 774.0	8322 435.21 700.	77+2 106.70 168.0	774 29.58 4010. 774 29.58 4010.	7290 255.28 109.4	3742 253.57. 27.46	774 29.58 4010. 774 29.58 4010.	645 47.23 2004	1645 4Z+28 2004	1645 47.28 2004.	645 47.28 2004	1645 47.28 2004	7741 2776.3 2776.	* * * * * * * * * * * * * *	SIIND	FORCE - DYNES	LENGTH - CM		ATO3 CARD	*			a for figure and the second						
6 36 24 25 22 4.8 6 37 23 25 28 4.6	6 38 24 27 3ú 24.	6 39 26 27 24 11.	6 40 26 23 23 9.2 6 41 27 30 24 9.2	6 42 28 29 1 20°	6 43 29 30 2 16.	6 44 1 23 5 9.27 6 45 2 30 4 9.2	6 46 28 31 8 1 8.0	6 47 29 32 9 4 5 0	6 49 9 32 19 1 8 0	6 50 18 31 3 1 8 1 8 0		6 53 32 33 4 53.	ANALYSIS OF TAS		-Annana -	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		DATA TESHIN	* * * * *	1	we we compare the second				a statut da fan af fan it fan it fan af f			

STRUCTURAL ANALYSIS CONTROL DATA

GENERAL DATA

INDRKT INDSFG INDSFL INDWKT INDWKT		0 4 4 0 0	 1 IMPLIES READ ALL MATAIGES FROM TAPE 0 IMPLIES WRITE GLOBAL ARA WATRIGES 1 IMPLIES WRITE LOCAL ARA WATRIGES 1 IMPLIES SAUE ALL MATAIGES ON TAPE 1 IMPLIES PRINT TOTAL STIEFNESS MATRIX
ISFDIM	H	12904	, MAX. STORAGE FOR STIFFNESS MATRIX

-

- And the state of the statement

ļ

i

DISPLACEMENT/ROTATION SOLUTION DATA.

	j		
ERRIOL	H	1.6036-04	ITERATION SOLUTION TOLERANCE
INDISL	H		, I IMPLIES AN INITIAL SOLUTION IN SOLVEC
RIDITR	н	2000	MAX. SOLUTION ITERATION CYCLES
STAUNI	n	ſ	, 1 IMPLIES CONSIDER PLASTICITY
INDRLX	ų	2	ITERATIVE SOLUTION NETHOD
TANDAKT	ŋ	0	, 1 INPLIES SAVE SOLVEC ON TAPE
MINRST	'n	9	, MIN. ALLOWABLE RESTRAINTS
RELAXF	H	1.00	RELAXATION FACTOR
TMAX	Ħ	190.010.	. ITERATION CP TERMINALION IIME

a sector data and sector and

		IC TOTAL POINT DI GLOBAL	SPLACEMENTS AND COORDINATE SYSTI	ROTATION S Em		
NOUAL POINT Number,	DISPLACEMENT X	DISPLACEMENT Y	D ISPLACEMENT Z	R0141 I0M X	RATION Y	ROTATION Z
5 2				·		
Ţ	-2.394u57E-91	-1.234398E-05	9.954863E-02	-2.290412E-33	-2.493685E-33	-3.99,52.2-25
2	-2.596587E-01	-1.103752E-03	6.875395E-04	-7.508595E-04	-2.517113E-J3	-2.2.7545±-15.
M.	-2.5947685-01	2.344819E-04	1.379575E-L1	1.36538 üE-05	-3.0357842-33	2.3]5]+1E-05
	-2.396246E-01	-7.071421E-04	1.293253E-D2	-1.330.39.3E-U5	-2.949074E-13	
ب م		7. 5 34 44 / L - L - L	1.8/40165-01	6.550032E-U6	-3.1009995-03	1.41/395E-35
اھ	-2.3795155-01		9.8754885-12	2.309811E-03	-2.453771E-03	8.514476E . 15
2	-2.582635E-01	-3.037241E-04	1.50+430E-u3	7.26947 4E-34	-2.4769366-J3	6.76337~E-05
60 -	-1.3031345-01	2.133496E-01	-2.923892E-32	1.953775E-ū3	-2.134581E-13	7 <u>*253446E+04</u>
σ	-1.30LC64E-01	1.997026E-01	-9.144344E-02	4.562935E-u4	-1.929153E-33	6.75428°E-04
- 10	-1-299991E-01	1.848225E-01	-1.073696E-01	-1-54639.9E-04	-1.4460.955-03	1.831387E-03
11	1.195158E-03	1.044167E-01	3.4235835-02	-3.897971E-J3	-1.287173E-33	2.293985E-ú3
- 12	.0.			-5.67438 QE-06	-1.318790E-03	2.132221E-J3
13	4.343227E-03	2.070061E-02	-1.512224E-02	-1.524434E-33	6.796597E-35	6.J13718E-05
	3.J46273E-03		2+885272E-63	1+93654.0E-04	-1.1063792-04	1.356412E-04
15	0	0.	•	-6.927687E-04	4.233830 5-64	-1.651343E-04
	-2.15337E-13	-6.087533E-03	-3.239054E-02	-1.848661E-03	5.487169E-04	-2.619314E-05
17	0.	•0	0.	-7.723632E-04	2.783482E-04	1.1.8644E-05
	-2.139731E-03	1.646172E-05	4.034278E-01.	-8.232547E-09	-1.3781525-03	-1.181016E-08
19	-2.121140E-03	4.047092E-06	-4.402838E-61	-2.117497E-38	-1.246994E-03	3.966774E-07
	-1.739390E-03	. 1.752866E-06	-4.351520E-61	-1.870346E-38	1.379476E-03	1.526218E-08
21	-2.3721266-03	6.124341E-03	-3.239479E-02	1.848546E-03	5.493697E-04	2.617138E-35
22	0.	0.	0.	7.724276E-04	-2.79ù772E-û4	-1.135609E-05
23	4.365225E-03	-2.072861E-02	-1.508775E-02	1.53188 2E-03	6.648816E-05	-6.310193E-05
24	3.1616162-03	-5.449777E-03	2.886710E-03	-1.93665.6E-04	-1.126262E-04	-1.901353E+04_
25	Ľ.	•	•	6.86955 0E-J4	4.229063E+04	1.697941E-04
26	1.188954E-03	-1.048444E-01	3.456437E-02	3.914264E-03	-1.295317E-U3	-2.314838E-03
27	•0	0.	•0	5.674096E-06	-1.326562E-03	-2.2J417/E-J3
28	-1.307529E-01	-2.196456E-01	-2.894178E-02	-1-975831E-113	-2.150572E-u3.	-7-172035E-04
29	-1.3C7460E-01	-2.031335E-01	-9.156317E-U2	-4.437621E-04	-1.944682E-33	-6.726332E-04
30	-1.347380E=01	-1.853135E-01	-1.076b10E-(1	1.81348 9E-04	-1.464383E-03	-1.884615E-D3
31	-9.576417E-02	-4.738775E-05	-4.13109E-01	1.323384E-06	-1.174504E-03	6.175342E-06
32	-9.575433E-02	-2.432919E-04	-4.4 39126E-01	1+25665 3E-06	-1.2042295-03	6.149842E-06
33	-9.576677E-02	-1.322386E-04	-4.288944E-01	1.289626E-06	-1.154116E-03	6.222322E-06
1068 ITERATIONS WERE	REQUIRED TO RE	ACH A MAXIMUM R	ELATIVE DIFFERE	NCE OF 9.653931	64E-05	

Ì

-

ļ	YOMENT		•0			• 0		• • • • • • • • • • • • • • • • • • • •	0.	0.0			· · ·	• •	• •		0. 0.			•••	0, -2.75384 EE406								Second a construction account		
	MQMENT Y		.0	•		•••			•••	•••				• •	•••	•	•••				0. 2.957030E+09										
ENTS	MQME NT X		0.	.0.					•••	.0							.0	•0			0. -9.18800 JE+05										
T FORCËS AND MOME Coordinate syster	FORCE Z	-	4.612303E+07	3.735090E+08	4-403000E+47	5.892000E+07 4.572000E+07	3.74200E+08	7.265000E+06	1.003000E+v7 2.372000E+C7	3 ~508397E+08	3.324000E+67	1.25+0006+05	1.0018386+09 - <u>1.4730006</u> +08	-7.277000E+07 -4.552000E±07	1.633000E+06 4 004.665400	1.5280006+07		2.53430696+67	-2.225000£+06 7.221400£+06	9.352000E+06	-1.290000E+09 -1.290000E+08 -	6.293171E+03		. 00021							
NODAL POINT	FORCE Y		-7.2270JJE+03	-1.223005108 2.0840.05105	-1-54900E+05	2.31803JE+05 4.5109JJE+05	1.265006E+08	3.738036E+07	3.670030E+07	-6.856081E+08 2.191000F+07	1.74300E+06	-1-351030E+06	3. /5/ 355E+U8 -1.153014E+05	3.229830E+04 4.8668036+04	1.123003E+06	-2.1990305+07	-1.71600JE+06 8.099014E+08	-3-66000JE+07 6.8517415408	-3.7400JJE+07	-3.224003E+05	<u>-3.646403E+05</u> 3.004030E+04	9.765102E+04	CENT	.0230	8						
	FORCE X		1. 301000E+07	-1 <u>-1980085458</u> 1.7370705407	1.1396 00E+07	4.910000E+06 1.373030E+07	+1.J86600E+08	-1.37500E+06	-2.505000E+06	-5.24348E+08. -5.350000E+07	-3.>31006+07 5.400605408		/•3214256408 -1•476006408	-6.288000E+07 -4.557000E+07	-9.341000E+07 7 438630640#	-5.3250 J0E+07	-3.516000E+07 5.509640E+08	-2.5140005402 -5.7120725408	-3.31300£426 -2.386030£406	-2.1330.02+06	-3.2760305+08 -3.5600305+07	6. 1056445+03	UTION ERRORS IN PER	.00023			į			• • • • • • • • • • • •	
	NODAL POINT NUMPER		4	01 M) J	n o	2	6 0 1	11	12 13	44	91	16	19 20	21	546	24 29	26 27	- 6 0 6- 1 0 1 0	30	0 M M M	TOTAL	RFLATIVE SOL					:			

	1																	
	TVENT 7			-6 6×7698E+J3 7.125+8.E+J3	<u>-4.32577+E+16</u> -4 186206E 35	-4.207454E+36 -4.332236E+04	2.6.329/E+16 1.372715E+35	-6.5094545419 4.444.933E+19	-7.123227E+13 6.7-0442E+33	-6.319781E+09 -5.712544£ 13	-4.702638E+03	-6.724742E+09 -7.334145E+09	4.031278E+09 1.82u832E+09	-1.824670č+03 4.618367E+09	4.668035E+09 6.232525E+39	7.368758E+39 7.471765E+39	5.876119E+09 5.876119E+09	-5.844829E+08 1.080452E+10
	MOMENT	2+11918. 59	<u>5.9137243+48</u> -1.027754≦+38	7.3619377+08 -6.2116912+68	6 <u>*336732E+</u> 9£ 1.599789 <u>_</u> +67	1.202607±+u7 3.406229ё+й6	<u>8.932963.t.6</u> -3.61126vE+36	1.531335+C8 -5.775895E+08	<u>6.2768505408</u> -7.191026E+08	-2.0970876+09 -2.2247055+09		2.5636036+09 7.731100E+08	-3.4617885408 -6.5385862+08	4.783456E+09 9.488049E+06	5.4770766+09 5.119799±+08	3.8453036+09 6.9088185+07	<u>-3,9160825+08</u> 5,517508E+06	5.338310E+09 6.u11774E+09
i	MOMENT X	1.527523:+07 -1.527523:+07	-9.153672±+06	-1.187904±+07 1.187904±+07	<u>+1.094831∑+06</u> 1.094831≦+06	9.459827c+04 -9.459&27c+04_	<u>=3.600683±+04</u> 3.600683 <u>5</u> +04	9.867145±406 -9.867145±406	<u>1.298391E+07</u> -1.298391E+07	-1.570495c+07 1.570495c+07	<u>3*960516E+06</u> -3.9605162+06	1.1691242+07 -1.169124E+07	<u>-3.463495E+07</u> 3.463495E+07	-3.462681c+07 3.462681c+07	9.971527±+06 -9.971527c+06	1.810623£+06 -1.810623£+06	<u>3.8618815+07</u> -3.861881E+07	-4.672166E+07 4.672166E+07
SF	FORC: Z	-1.04%4515+08 1.04%4515+08	-1.384467E+07 1.384467E+07	-3.625722E+06 3.625722E+06	<u>5, 5033185+05</u> 5, 563318E+05	-4. 698598E+05 . 4. 698598E+05	<u>3-266836≟+04</u> 9-266836E+04	1.331067E+07 -1.331067E+07	2.8815985+06 -2.8815985+06	1. 037496E+08 -1. 037496E+08	-1. 662653E+07 1. 662653E+07	-5. 706150£+07 5. 706156£+07	3. 1751256+07 -3. 1751256+07	-4.717858E+08 4.717858E+08	-1.025100E+08 1.025100E+08	-6.70J072E+07 6.70J072E+07	-9. 268548E+06 -9. 268548E+06	-3.580061E+08 3.580061E+08
DORDINATE SYSTEM	FORC- Y	-2.879997E+U7 2.879997E+U7	-6.558738E+07	1.379957E+07 -1.379957E+07	+2+090930E+05	-1.328559E+05 1.328559E+05	4.288699E+04 -4.288699E+04	-6.709352E+07 6.709352E+07	-1.206587E+07 1.206587E+07	-2.888498E+07 2.888498E+07	-1.567732E+08 1.567732E+08	-2.406391E+08 2.406391E+08	1.858049E+08 -1.858049E+08	2.749702E+08 -2.749702E+08	9.056839E+07 -9.056839E+07	2.540180E+08 -2.540180E+08	2.755679E+08 -2.755679E+08	3.223619E+08 -3.223619E+08
LOCAL CC	FORCE X	-4.382243E+07 4.882243E+07	8.583158E+07 -8.583158E+07	2.119086E+08 -2.119086E+08	-3.418606E407 3.418606E407	-8.918537E+07 8.918537E+07	<u>4.326431E+07</u> -4.326431E+07	8.344166E+J7 -8.344166E+Q7	<u>2.123405E+08</u> -2.123405E+08	-4.862856E+07 4.862856E+07	-1.594680E+08 1.594680E+08	-1.873301E+08 1.873301E+08	<u>3,133247E+06</u> -3,133247E+06	7.997026E+06 -7.997026E+06	-2.588178E+08 2.588178E+08	6.288825E+07 -6.288825E+07	-2.305926E+08 2.305926E+08	-3.288813E+08 3.288813E+08
		POINT P	POINT P	POINT P	POINT Q	POINT P Point D	POINT P	POINT POINT Q	POINT P	POINT P Point Q	POINT P	POINT P Point Q	POINT P	POINT P Point Q	POINT P	POINT P	POINT P	POINT P Point Q
	14852S R ,	(M	28	30	-	7	2	28	30	ю	M	£	Ħ	12	13	14		C
	N TNIC	N	2	4	4	5	5	e :	7	~	B	10	σ	10	11	12	12	11
	1 TUDAL	' न	+	~	'n	<u>به</u> ب	4	M	t.	9	ę	2	8	6	8	13	11	11
	3AP NUMBEP	1	7	M	4	ß	bb	7	80	, •	1 n .	Ħ	12	13	14	15	16	17

.

: . .

.

; ;

POINT BUT MUT MUT MUT MUT MUT MUT MUT MUT MUT M						BAR FO	RCES AND MOMENT JORDINATE SYSTE	S HS		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	AR NRED	VOUAL	PJINT N Q	U 1BER	\$	<u>EORCE</u> X	FORCE	FORCE Z	MOMENT X	MOMENT Z
1.1 1.4 <th></th> <th></th> <th>I</th> <th></th> <th></th> <th></th> <th>£</th> <th></th> <th></th> <th>and the second second</th>			I				£			and the second
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	18	12	14	10	POINT P	-1.201164E+J8 1.201164E+08	-3.559466E+08 3.559466E+08	-4. 093429£+08 4. 093429£+08	-6.442613E+07 6.442613E+07	5.8430055+69 -7.511183E+09 7.1534915+09 -3.774610E+09
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	۰. ۵.	13	14	11	POINT P POINT Q	-2*999368E+08 2*999368E+08	-2.693242E+08 2.693242E+08	-3. 803306E+05 3. 809306E+05	6.976098E+07 -6.976098E+07	-7.9313175+46 -5.443194E+49. 2.379936E+07 -5.775776E+09
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	20	13	15	11	POINT P	-1.115395E+09 1.115395E+09	6.056762E+07 -6.056762E+07	-1.953897E+06 1.952897E+06	-2.0607565+06 2.0607565+05	4.8898212+07 1.938506E+09 1.359683E+07 -1.976132E+06
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ç 1	† †	15	12	POINT P	3.491764E+06 -3.491764E+06	<u>-3.041003E+05</u> 3.041003E+05	-2.818253E+05 2.818253E+05	2.361462E+06 -2.361462E+06	<u>3.152152E+07</u> -1.9208295+07 -1.35130+2+07 -2.235554E+05
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	52	£1	16	18	POINT P Point Q	-5.390495E+08 5.39 <u>0495</u> E+08	-6.193668E+08 6.193668E+08	-1.637794E+08 1.637794E+08	-5.632838E+16 5.632838E+16	3.9324395+08 -9.0518056+09 4.8048366409 -1.0505856413
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	53	t T		20	D INIO4	1.209494 <u>5+08</u> -1.209494E+08	-3,369852E+08 3,369852E+08	<u>-1. 4854825+08</u> 1. 4854825+08	+9.52 <u>6940E+06</u> 9.526940E+06	- <u>1,3333765+09</u> 3.6 <u>943695+09</u> 6.10864225+19
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	54	16	17	41	POINT P Point Q	4.004135E+08 -4.004135E+08	+1.238612E+08 1.238612E+08	-2.397152E+08 2.397152E+08	1.067887E+07 -1.067887E+07	4.531354E+09 -2.233049E+09 5.454224E+09 -2.956520E+09
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	25	16	19	13	POINT P	-6.616721E+08	3.853367E+08 -3.853367E+08	1. 3671735+88 -1. 3671735+88	1.075441E+07 -1.075441E+07	-5.220281E+09 1.061675E-10 -2.766746E+09 1.189462E+13
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	26	17	. 20	14	A THIOP	-1.900115E+05 1.900115E+05	5.013956E+08 -5.013956E+08	2.3314115+08 -2.331411E+08	-6.145715E+86 6.145715E+86	-6.2912946+09 1.4378826+13 -7.3286096+09 1.4912706+13
20 17 POINT P 4.199635E+08 -3.932058E+05 -9.5627135+08 1.4006795+02 9.703436E+09 4.469666E+06 29 5627135+08 -1.4006795+02 9.5627135E+08 -1.370277E+08 -1.259434±07 4.469566E+05 29 21 23 23 237576+08 -1.370277E+08 -1.370277E+08 -1.275726E+09 -1.491271E+11 30 20 22 24 POINT P -6.620967E+08 -3.853362E+08 -1.370277E+08 -1.075806E+07 5.279435E+09 -1.491271E+11 31 21 22 24 POINT P -1.900115E+05 5.013946E+08 -3.33344E+09 -1.491647E+01 -1.49127E+11 31 21 22 24 POINT P -1.900115E+05 5.013946E+08 -2.40086E+06 7.40586E+06 -1.49127E+07 -1.49127E+09 -1.49127E+01 31 21 22 24 10 3.39346E+08 -1.4093826E+08 -1.491364+09 -1.491271E+09 -1.491271E+09 -1.491271E+09 -1.491271E+09 -1.491271E+09 -1.491272E+09 -1.491242E+09 -1.491242E+09 -1.491242E+09 -1.491242E+09 -1.491242E+	27	4	19	16	D TNIOG	8,352166E+06 -8,352166E+06	-4.252396E+05 4.252396E+05	-1. 301373E+08 1. 301373E+08	-2.993428E+02 2.993428E+02	2,369 <u>5465+19 -8,9566956+16</u> 1,738260E+09 -4,+36650E+16
21 23 PDINT P -6.020967E+08 -3.953362E+08 -1.370277E+08 -1.075806E+07 5.2775726E+99 -1.404947141 30 20 22 24 PDINT P 1.900115E+05 -5.013946E+08 -2.334346E+08 -5.334346E+06 7.3377376+09 -1.4430977E+11 31 21 22 24 PDINT P -1.900115E+05 5.013946E+08 -2.334346E+08 -5.334346E+06 7.3377376+09 -1.4432776E+11 31 21 22 24 PDINT P -1.900115E+05 5.013946E+08 -2.33434E+09 -1.457276+06 7.445296E+16 7.4377376E+11 31 21 22 24 PDINT P -5.39436E+08 -1.637294E+08 -1.4912716+17 -4.451222E+09 -1.4432976E+18 -1.4432976E+18 -1.4432946E+08 -1.4067716E+07 -4.451722E+09 -1.4432976E+18 -1.47521429 -1.4432976E+19 -1.4369274E+09 -1.437272E+09 -1.443894214E+09 -1.4432976E+18 -1.476248E+08 -1.067710E+07 -4.451722E+09 -2.494641E+09 -1.467710E+07 -4.451722E+09 -2.946441E+09 -1.46322445E+08 -1.476248E+08 -1.067710E+07 -1.4532145E+09 -1.49641E+09	28	19	20	17	POINT P POINT Q	4.199635E+08 -4.199635E+88	-3,992058E+05 3,992058E+05	-9.562713E+08 9.562713E+08	1.400679E+02 -1.400679E+02	9.7034365+09 4.4698665+06 1.2280345+07 -8.525796E+06
20 20 21 22 24 POINT 1.900115E+05 5.013946E+08 2.334346E+08 5.141646E+06 7.337378E+09 -1.437876E+11 31 21 22 24 POINT 0 -1.900115E+05 5.013946E+08 2.334346E+08 5.343446E+06 6.141646E+06 6.239944E+09 -1.437876E+11 31 21 22 24 POINT 0 -3.997941E+08 1.233825E+08 2.343346E+06 -1.067710E+07 -4.45333310E+19 -1.43784E+09 31 21 23 18 POINT 0 -3.997941E+08 1.233825F+08 2.400862E+08 -1.067710E+07 -4.4533340E+19 -2.493442E+09 -1.437842E+09 -1.436922E+109 -5.9464842E+09 -1.437842E+09 -1.436922E+109 -1.436922E+109 -5.9464842E+09 -1.436922E+09 -5.4067662E+109 -5.4067662E+109 -5.4067662E+109 -5.405766+09 -5.405766+09 -5.405766+09 -5.405766+09 -5.405766+09 -5.405766+09 -5.40576660E+09<	53	18	21	53	POINT P Point 0	-6+620967E+08 6+620967E+08	=3.853362E+08 3.853362E+08	-1.370277E+08 1.370277E+08	-1.075806E+07 1.075806E+07	2.7757265409 -1.189457E+13 5.229435E+09 -1.061677E+13
21 22 24 POINT Q -3.397341E+08 -1.233825E+08 2.400362E+08 -1.067710E+07 -5.461722E+09 -2.193442E+09 22 21 23 18 POINT Q -5.394566E+06 1.233825E+08 -2.400362E+08 1.067710E+07 -5.461722E+09 -2.946481E+09 27 23 18 POINT Q -5.394566E+06 6.205329E+08 1.652994E+08 5.696638E+06 -4.799573E+09 1.066623E+10 27 23 23 24 20 POINT Q -5.394566E+08 -1.460332E+08 -4.460332E+09 -4.439573E+09 -4.4302745E+09 -2.432745E+09 -2.432745E+09 -2.432745E+09 -2.432745E+09 -4.430226E+09 -3.432745E+09 -4.430226E+09 -3.432745E+09 -4.430226E+09 -3.432745E+09 -4.430226E+09 -5.437760E+09 -3.432745E+09 -4.430226E+109 -3.432745E+09 -3.432745E+09 -3.432745E+09 -4.430226E+109 -5.437760E+09 -3.432745E+09 -4.430226E+109 -5.437760E+09 -5.4	30	50 20	52	54	POINT P Point D	1.900115E+05 -1.900115E+05	-5.013946E+08 5.013946E+08	-2. 334346E+08 2. 334340E+08	6.141646E+06 -6.141646E+06	7.337373E+09 -1.491271E+19 6.299841E+09 -1.437876E+13
21 23 18 POINT P -5.394566E+06 6.205329F+08 1.653294E+08 5.696638E+06 -4.799573E+09 1.060623E+10 33 22 24 20 POINT P 1.20329E+18 1.476248E+08 -1.6596638E+06 -3.432745E+19 9.1080431E+19 33 22 24 20 POINT P 1.20329E+18 1.476248E+08 -1.476248E+08 -3.432745E+19 1.438929E+19 33 22 24 20 POINT Q -1.203299E+08 3.3569142E+108 1.476248E+08 -9.648216E+106 -3.432776E+19 -3.459226E+13 34 23 24 22 POINT Q -1.203299E+08 -3.569142E+108 -1.476248E+08 -9.648216E+106 -5.437760E+103 -3.696216E+103 34 23 24 22 POINT Q -3.114755E+103 -2.690715E+108 +4.460332E+05 7.006383E+07 -6.637976E+103 -5.776692E+03 34 23 24 22 POINT Q -3.114755E+103 -2.6500715E+103 +4.460332E+05 7.006383E+07 -5.651794E+07 5.776692E+03	31	21	22	24	ð INIOd á ÍNIOd	3.297941E+08 -3.997941E+08	-1.233825E+08 1.233825E+08	2* 403 862E+08 -2* 403 862E+08	±1.067719E+07 1.067710E+07	-4.53 <u>9310±+09 -2.193142</u> E+ <u>09</u> -5.461722E+09 -2.946481E+09
33 22 24 20 POINT Q 1.2082595410 3.3691426408 1.4762486408 9.6482166406 1.4175116409 3.6962066403 7 23 24 22 POINT Q -3.0147556408 2.6907156408 -4.4603326405 -7.0063836407 -6.6379766403 5.4377606403 34 23 24 22 POINT P -3.0147556408 2.6907156408 -4.46033226405 7.0063836407 -6.6379766403 5.4377606403	32	21	23	18	POINT P Point Q	-5.394360E+08 5.394360E+08	6.205329E+08 -6.205329E+08	1.632994E+08 -1.532994E+08	5.696638E+06 -5.696638E+06	-4.799573E+09 1.060623E+10 -3.832745E+09 9.088431E+09
34 23 24 22 POINT P -3.014755E+08 2.690715E+08 -4.463332E+05 -7.006383E+07 -6.637976E+16 5.437768E+09 POINT 0 3.114755E+10A -2.690715E+0A 4.461332E+05 7.006383E+07 5.771682E+09	₽£.	22	74	. 20	POINT P	<u>-1.208259E+08</u> .	-3 <u>*369142E+08</u> -3 * 369142E+08	1. 476248E+08 -1. 476248E+08	. <u>9+648216E+06</u> . -9,648216E+06	-6.102072E+09 1.438929E+13 1.417511E+09 -3.696206E+03
	34	53	24	22	POINT P Point Q	-3.014755E+08 3.014755E+08	2.690715E+08 -2.690715E+08	-4.463332E+05 4.463332E+05	-7.006383E+07 7.006383E+07	-6.637976E+16 5.43776JE+09 2.521794E+07 5.771682E+09
							•		> **	
			:	;		:				

	MOMENT Z	-1.962748E+09 1.122493E+06	1.936631E+07 3.699830E+05	-1.086276E+10 5.716933E+19	3.7767735+09 7.512544E+09	5.612745E+39 5.883348E+03	-6.096623E+38 -4.681298E+09	-7.473612E+03 -7.352184E+43	-4.053108E+09 -1.837760E+09	1.841662E+09 -4.649778E+19	<u>-4.499341E+19</u> -4.716782E+19	-6.672427E+09 -7.316778E+09	n. 8.652084E+09	0. 8.022325E+19	ū. -8.647720E+09	0. -8.028355E+09	0. 5.625294E+D5	0. -3.108322E+06
	MOME NT Y	4.96J297E+D7 1.361378E+G7	3.195032E+07 -1.362806E+07	-6.J28253E+09 -5.333381E+09	<u>-7.171846E+09</u> -5.830631E+09	4.168584E+08 2.269907E+07	<u>-5.413518E+08</u> -5.516229E+09	-1.020316E+08 -3.389114E+09	-3.556837E+08 -6.623720E+08	4.767884E+09 1.602149E+07	-2,4388886+09 1.448235E+09	-2.589172E+09 -7.606170E+08	<u>3,4981946+08</u> 5.7339386+09	-4.105504E+09 4.369464E+09	3.401349E+08 5.718269E+09	-4.129586E+09 4.349241E+09	-2.3820276+09 -1.030164E+10	-1.144171E+10 -1.310480E+10
	MOMENT X	2.133977E+06 -2.133977E+06	<u>-2,3455576+06</u> 2,345557E+06	4.690425E+07 -4.690425E+07	-6.470065E+07	-3.878545E+07 3.878545E+07	. <u>-1.006008E+07</u> 1.006008E+07	-1.929124E+06 1.929124E+06	<u>3.5434926+07</u> -3.5434926+07	3.542690E+07 -3.542690E+07	<u>-4.2310395+06</u> 4.231039E+06	-1.193982E+07 1.193982E+07	<u>4.863881E+06</u> -4.863881E+06	3.949656E+06 -3.949656E+06	<u>-4.789116E+06</u> 4.789116E+06	-3.880420E+06 3.880420E+06	-2.493369E+04 2.493369E+04	-2。318474E+04 2+318474E+04
ŧS	FORCE Z	-1.976272E+06 1.976272E+06	-2. 867 356E+05 2. 867 356E+05	3. 583704E+08 +3. 583704E+08	4.101261E+08 -4.101261E+08	-1. 055208E+07 1. 055208E+07	1.036847E+08 -1.036847E+08	6. 831450E+07 -6. 831450E+07	3, 2323335407 -3, 2323335407	-4.702663E+08 4.702663E+08	1.695516E+07 -1.695516E+07	5.733671E+07 -5.733671E+07	<u>-8. 645563E+07</u> 8. 646563E+07	-3. 751545E+06 3. 751545E+06	-8.611530E+07 8.611530E+07	-3.121859E+06 3.121859E+06	1. 8027335+08 -1. 8027335+08	3. 483801E+08 -3.488801E+08
DORDINATE SYSTEM	FORCE Y	-6.132402E+07 6.132402E+07	3.079259E+05 -3.079259E+05	-3.246212E+08 3.246212E+08	-3.560894E+08 -3.560894E+08	2.759769E+08 -2.759769E+08	-9.056280E+07 9.056280E+07	-2.537659E+08 2.537659E+08	-1.873529E+08 1.873529E+08	-2.763893E+08 2.763893E+08	-1-577479E+08 1-577479E+08	-2.394464E+08 2.394464E+08	1.229681E+08 -1.229681E+08	1.140176E+08 -1.140176E+08	-1.229060E+08 1.229060E+08	-1.141033E+08 1.141033E+08	7.995244E+03 -7.995244E+03	-4.417866E+04 4.417866E+04
LOCAL CO	FORCE X	-1.119495E+09 1.113495E+09	4.173026E+06 -4.173026E+06	-3.298882E+08 3.298882E+08	-1.1999275+08 1.199927E+08	-2.293956E+08 2.293956E+08	-2.599978E+08 2.599978E+08	6.362893E+07 -6.362893E+07	<u>3+096110E+06</u> -3-096110E+06	8.825360E+06 -8.825368E+06	<u>-1.590604E+08</u> 1.590604E+08	-1.879883E+08 1.879883E+08	-1.481512E+08 1.481512E+08	-4.738507E+08 4.738507E+08	<u>-1.477460E+08</u> 1.477460E+08	-4.747608E+08 4.747608E+08	7.529094E+38 -7.529094E+08	7.533622E+08 =7+533622E+08
		POINT P Point 0	POINT P Point 0	POINT P Point 0	POINT P POINT Q	POINT P POINT 0	POINT P Point 0	POINT P Point D	POINT P POINT 0	POINT P Point D	POINT P POINT Q	POINT P Point D	POINT P Point Q	POINT P POINT Q	POINT P POINT Q	POINT P Point Q	POINT P Point Q	POINT P POINT Q
	IBERS R	21	22	59 59	30	77	23	24	-	N	m	t	8	σ	18	19	8	თ
	N INICA	25	25	56		27	28	0 E	29	30	28	30	31	32	11	32	Ĩ	32
	400AL	23	24	5	- 2+	26	26	27	28	29	+	~	28	29	B	σ	18	19
	BAR NUMBEP	35	36	37	36	6£	U1	41	42	1 1 1	44	45	4Đ	47	4.8	49	50	51

ł

	T MOMENT NOHENT		E+U6 -1.362884E+U7 -4.58408UE+U3 E+U6 -5.254138E+U5 .1.102479E+Q3	E+06 -1.756919E+06 -1.854543E+19 E+06 -2.228071E+06 -1.854543E+19		and a second				And a starting and a start of a start of a start of a start of an and a start of a start			Announce of Announce of the state of announce of the state of the state of the state of						AND AND A CANADA AND AND AND AND AND AND AND AND A CANADA AND AND AND AND AND AND AND AND AN					
	MOMEN		-3.613494	-2.678310					t					:										
S MS	FORCE Z		1. 041596E+06 -1. 041596E+06				4			•		·					a we are seen as	-		 	A AMERICAN A A A A A A A A A A A A A A A A A A			
RCES AND MOMENT Jordinàte syste	EORCE		-2.562079E+08 2.562479E+08																				17. AND AND AND AND AND AND AND AND	
BAR FO	FORCE X		-7.695605E+06 7.695605E+06	2.796800E+07 -2.796800E+07	an ann an an an an ann an ann an ann an	 And is an and in the state of the state					to any our community where			•	An and the second s	;			· w menu ti	 		"And a set of the set		
			POINT P	POINT Q			And a second	-		i 		-						-						
	U 18ERS R ,		, M	t						÷ ÷						-								
	POINT N		33	33	!	-						1	-			-								
	NOJAL	;	51	35										:			-	-		•				
	BAR NUMAE?		25	53																				

----ł ł į ÷ 1 Ì • 1 1 t į ٠İ i TOTAL ł 1.10 1.02 39.96 į ł ł 1 -----TIME OUT 2.465 • 1.363 42.421 i Ì -----1.363 TIME IN 245-2.465 ł -----CPU TIME USAGE TABLE (SEC) -- DISPLACEMENT, ROTALION, EORCE, AND MOMENT SOLUTION OVERLAY Ì • ۷ STRUCTURAL STIFFNESS MATKIX GENERATION OVERLAY INPUT AND INITIALIZATION DVERLAY to the second seco ----1

.

APPENDIX B

EXAMPLE OF INPUT AND OUTPUT

MODAL ANALYSIS ROUTINE

STRUCTURAL ANALYSIS PROGRAM

STRUCTURAL ANALYSIS PROGRAM --- LEGGED LANDER Master Agreement, contract Nas1-8137, task order Numger Five McDonnell Douglas Astronautics Company, east

STRUCTURAL ANAL	YSIS DATA -	CARD CODE		
a		BLANK - O 1		COMMENTS NODAL PDINT DEFINITIONS
		u m		NODAL POTNI PESTRATNI DEFINITIONS
: 		4		FORCE VECTORS
		u u		MOMENT VECTORS RAR DEFINITIONS
-) N 80		SHEAR PANEL DEFINITIONS Formaled-data terhinator
13.5898	31.7246	85.7758		
1 2 -28.0670	31.7246	85.7758		
.1.3 13.5890	•	85.7758		
1 5 29.4894		113.538		
1 6 13.5890	-31.7245	85.7758		
-1728-0670 f 8 13.5890	-51.7246	85.7758 35.1790		
1 9.11- 9 1	-60.9345	35.1790		
1 10 -28.0670	-60.9345	35.1790		
1 11 15.5890 4 42 -28 nc70	-90-1446	-15-4175		
1 13 13.5690	-74.2950	-42.8752		
1 14 -28.0670	-74.2950	-42.8752		
1 15 29.4894	-98.3234	-56.7690		
1 16 13.5890 1 17 -28.0670	- 58.42 - 58.42	-70.358 -70.358		
1 18 13.5890	0.	-70.358		
0206-21- 61 1		-70,35.5		
1 20 -28.067 1 21 13.589	58.42	-70.358 -70.358		
1 22 -28.067	58.42	-70.358		
1 23 13.589	74.295	-42.8752		
1 24 -28.067	74.295	-42.8752		· ·
1 26 13.589	90.1446	-15.4178		
1 27 -28.067	30.1446	-15,4178		
1 28 13,589 1 29 -17.9070	50.9346 50.0346	35.179 36.170		
1 30 -28.067	50.9346	35.179		
1 31 13.569	-	0 -		
1 32 -17.907	•			
6 1 1 2 3	11.774?	106.70	168.10	1.5775 6.926E11 2.620E11
6 2 1 3 28	16-1290	211.43	274.90	2.1644 6.826E11 2.620E11
6 3 2 4 30	24.8322	435.21	700.43	3.3299 6.926E11 2.620E11
	14.1129	110.95 66 67		0.495/ 0.826£11 2.62UE11 • • • • • • • • • • • • • • • • • • •
 	1.8387	5.98 6.98	3.68	1.44501 0.45504511 2.6207611 .47431 6.425611 2.6207611
6 7 3 6 28	16.1290	111.43	274.90	2.1644 6.826E11 2.620E11
6 8 4 7 30	24-8322	435.21	700-43	3.3299 6.826F11 2.620F11
6 4 6 7 3 7 7 7 3	11.7742	106.70	166.10	1,5775 6,826411 2,620641
6 11 7 10 4 5	4222-6	29.53 53	4010.2	1.2449 5.326tll 2.620tll 1.7445 f.R76fl 2.620f1
6 12 8 9 11	20.729	255.28	109.46	2.7804 6.826F11 2.620E11
6 13 9 10 12	16.3742	253.57	27.46	2.1977 6.825611 2.620611
CT TI Q 11 Q .	A-12.4	04+62	4010.5	1.2445 0.846EII 2.6cucii

			-									A MARINE			province a second se		p statutions (summaries Van) state the																			j			H
620E11	620E11	62UEII	620E11	620E11	620E11	620F11	620E11	620E11	620E11	620E11	620E11	620E11	620E11 620F11	620E11	620E11	620E11	620E11	620E11	620E11	620E11	620E11	620E11	bZUE11	620F11	620E11	620E11	620E11	620E11	620E11	52022	620E11 620511	620E11	620E11						
1 2.	~ •			1 2.	N.0 +1+		1	~	1 2.	12	1 2.1			1 2	1 2.	12.	1 2	- - -	2	- - 		N (N 0 1 +	••	1 2.	1 2.4	1.2	1 5	~			1 2	1 2.						
326E1	326E1	11020	326E1	326E1	326E1	12651	926E1	326E1	326E1	326E1	326E1	326E1	326E1 326E1	326E1	826E1	326E1	326E1	326E1	326E1	326E1	326E1	13920	17071 1021	176F1	326E1	326E1	326E1	326E1	126E1	11020	13926 13926	326E1	826E1						
5 6.6	ف د		0 0 0	1 6.5			5.0	5 6	5 6.8	4 6. 1	7 6.1	ی م		9	9 6.1	7 6.5	1 6.1	ů	4 6-1	9 9			0 4 0 4		5	5	2 6.8	2 2	0 0		0 0 N 0	9	7 6.1						
1.244	1.577		6.455	1.028	- 6243 2 464	BOE E	1.577	1.244	1.244	2.780	2.197	1.244	1.577	2.164	3, 329	6.455	1.026	.6243	2.164	3.329	1-577	1.544	1.244	2.197	1.244	1.244	1.032	1.082	1.032	200-1	1.082	5551.	5551						
0.2	5.0		°+•	2	ۍ و		.10	0.2	2.0	-46	46	0. 2	10.2	06	.43	~	Ņ	8	96	54.	212		2.4	0 4 9 9 9 9	0.2	0.2	4.2	4 . 2	N I		N 0	9.3	5.3		-				
401	169		2.4	2.2	3.6	002	168	401	401	109	27.	104	168	274	700	4.8	2.2	3.6	274	200	150				401	401	200	200	202			277	277						
5.8	2.		. 62	53	38	10.1	1	58	58	9.28	3.57	58	5.70	1.43	5.21	1.95	53	38	- 43	5.21	0.10	50	200		54	58	28	28	5			6.3	5.3						
29,	1		11	55		- M - 1	106	29,	29,	52	52	6 I	2 0 F	H	5	11(55	ů.	-	4		5.0	2 10	10	29.	29,	47	1	1]!		27	27						
774	2422	1 C 4 1	1129	613	387	8322	7742	774	774	7290	3742	774	7742	1299	8322	1129	613	367	1290	8322	2422	21	7 200	3742	774	774	645	645	645	542	5 4 5 4 7 4	7741	7741						
9.2			- - - -	7.6	4 • 8 • 4	3	Ŧ	9.2	9.2	20.	16.	8°6	5 T	16.	24.	14.	7.6	6.4	16.	24		N 6			5.6	9°5	8.9	3.0	6) (6) (6) (6) (6) (6) (6) (6) (58	58.						
44	ۍ ۱	•••	- - -	11	17 1 2	50	4 6	13	14	16	17	<u>.</u>	t t 0 0	13	20	22	21	25	28	2	11	5	÷	• •	m	4	•	σ		51	x 0	• • ••	4						
12		2	1 4 1 4	15	1 5 5 7	17	5	13	20	19	20	51	222	53	24	24	25	52	26	27	2		200		53	ĐĐ	31	22	÷ i	22	56	i M	Ê						
10	;;	1;	1	14	4 M) - - -	16	16	17	18	1 0	4 1	212	51	22	30	3	1 te	23	5	2	5	2	; ď	-	~	8 N	6 0	6 0 (0 1 1	1	25						
15	÷		4 7	20	5.5	18	1	22	26	27	28	5	9. F	32	Р Э	46	ŝ	36	5		,		4 G	4 P.	4	45	46	÷		3	2 5	5	50						
9	ie u	C 4	o vo	£	س م	o vo	o o	e ع	9	و	¢	.	ഗഗ	c	9 . '	v	G	so i	۵	vo i	• م	.	0 u	s uc	ى	ġ	¢	ا	w 1	۰ ام	שים.	o j	v	60					

.

16

.

STRUCTURAL ANALYSIS CONTROL DATA

GFNERAL DATA

				2.4905+03 6.0995+03	1.7367+03	2.1275+03 6.0996+03	1.736E+03	2.1277+03	6.0995403 1.736E+03	-	19 .	80 P	21	6./	117	139	158		a success and a success of a success		-	
				3.549E+03 6.093E+03	1.736E+03	3.549E+03 6.099F+03	1.7366+03	3.543E+03	6.099E+03 1.736E+03	3.579E+03	15	37	0 7 7	2 C C C C C C C C C C C C C C C C C C C	116	135	157	195				
				3.549E+93 3.789E+03	1.736F+03	3.549E+03 3.789F+03	1.7366+03	3.5495+03	3.799E+03 1.736E+03	3.578E+03	14	in L	22	3 4	115	134	153	194				
				3.549E+03 3.789F+03	1.379E+03	3,549E+03 3,789E+03	2,346E+03	3.549E+03	3.789E+03 1.879E+03	3,578E+03	13	32		5.0	111	133	152	193				
				6.099E+03 3.789E+03	1.879E+03	6.099E+03 3.789E+03	2.346E+03	6.099E+03	3.799E+03 1.879E+03	3.270E+04	σ	E C	90	500	110	129	151	189				
				6.099E+83 3.483E+83	1.879E+03	6,099E+03 3,483E+03	2,346E+03	6.099E+03	3.483E+03 1.879E+03	3.270E+04	8	27	44	20 P	109	128	147	188				
TAPE CES ES APE MATRIX X		يبا		6.099E+03 3.483E+03	3.533E+03	6.099E+03 3.483F+03	3.812E+03	6.099E+03	3.483E+03 3.533E+03	3.270E+04	2	26	ر ا	19	105	127	146	1.67				
TRIGES FROM L BAR MATRI BAR MATRIC BAR MATRIC TRIGES ON T STIFFNESS FNESS MATRI		SIS Data on tap	M DY MOJES	DUCE SYSTEM 3.789E+03 3.483E+03	3.533E+03	3,7996+03 3,4436+03	3.5126+03	3.799E+03	3.533E+03	3.487E+04 CED SYSTEM	Ð	25	++	59 9	401	123	145	183				
READ ALL MA WRITE GLOBA WRITF LOGAL Save All Ma Print Total Ge for Stif		MODE ANALY Writf Mode	EDUCE SYSTE ON-RIGID 30	R OF THE RE 3.7395+03 2.4905+03	3.533E+03	3.789E+03	3.812E+03	3.7895+03	2.127E+03 3.533E+03	3.4.876+04 IN THE REDU	~	12	43	29	101	122	141	192				
1 IMPLIES D IMPLIES 0 IMPLIES 1 IMPLIES 1 IMPLIES 1 IMPLIES 1 IMPLIES		, PUN NOPMAL 1 IMPLIES	ORDER OF R PEQUIRED N	. MASS VECTO 3.7995+03 2.4905+03	6. 199E+03	3.789E+03	6.099E+03	3.789E+03	2.127E+03 6.199E+03	3.487E+04		202	55	61	00	121	140	181				
0 4 4 0 0 3	YSTS DATA		66 50	HF DIAGONAL						AINS THE RC												
INDRKT = INDSFG = INDSFL = INDMKT = INDMKT = INDMTS = ISPDIM =12°	MCDAL ANALY	= VMNUNI = VMNUNI	IPEDTO = NFIGVL =	AMASS IS TI						IPWKR CONT												

. . .

	GENE	RALIZED INERTIA PR	OPERTIES			
VAPIARLES	*1	2	M	4	ß	
XNN	8.40331315E+04	1.12127882E+05	2.96850587±+04	3.421797135+04	6.06333976E+04	
4N M	5.43594868E+04	2.66459996E+04	5.37421£16E+04	6.16372988E+04	7.68004646E+94	
4N.4	1.97045967E+04	8.55226510£+04	5.417350505+04	6.14136396E+04	7.68967378F+34	
Xa	2.43423465E+01	-3.53679722E-01	4.676975215+03	-2.57396930E+02	-1.07756323E+92	
٨٥	-2.42335120E+06	-1.66506291E+02	-1.109361845+05	1.90101918E+03	4.44512439E+01	
۲a	2.483514705+06	1.70546160E+02	1.003041565+05	-1.55043815€+03	2.11936905E+02	
5M DIAGONAL	8.40437575E+04	1.12143266E+05	6.8A004331E+04	7 • 86644549E+04	1. 07190300E+n5	
·						
					N	
		- - -				
		.				
						i

		- 05			- 11 - 12 - 12	-91 -01 -01	-04 -01 -05	-04 -01 -05	-04 -01 -04	- 01 - 01	-01 -01 -01	-03 -01 -01	03	-03
	ις.	3×94729713E4		2	-3.79674797E- 4.87913279E- 5.92113921E-	-8,796747975 -6,596205965 -7.691056915	1.94579946E 4.88943089F -6.5691056 <u>9E</u> -	1.95392954E- -6.59620596E- 7.00313008E-	2-36205962E- 4-83523035E- -1.25474255E-	8.796747976- 4.87913279E- -5.82113821E-	8,80216802E- -6,59621596E- 7,69647696E-	-3,82972629E- -1,31327913E- -2,28988899E-	-3,82931030F- 9,75609756E- 1,637940385-	-3.82710027E- 1.73004130E-
	ŧ	3.06529126E+02		4	-9.87226277E-01 6.06508516E-01 5.04744526E-01	-9.872262772-01 -5.37469586E-01 -5.32420925E-01	<u>6.399026765-03</u> 6.08272506E-03 -5.10705596E-03	6.41727494E-03 -6.38686131E-01 4.75304136E-03	1.33029197E-02 6.96472019E-01 -9.11900487E-03	<u>1.00000005+00</u> 5.06569343E-01 -5.14111922E-01	1.000000000000000000000000000000000000	<u>5.745133826-01</u> 6.770072996-02 -2.145985406-01	5.749391736-01 -6.301703166-02 1.325425796-01	5.74391727E-01 -1.25729927E-01
ADIANS/SEC)	M	3.86055553E+02	R COMPLETE SYSTEM	m	-3,985756395-01 9.666011795-03 2.97151275-01	-3.990667985-01 -9.609528495-01 -2.95186640E-01	-3.862966605-01 8.558271125-03 3.15078535E-01	-3.872799785-01 -9.993123775-03 -2.924852655-01	-9.11640472E-01 9.79616896E-03 5.621316316-01	-3,705795685-01 7.421414545-03 2.526620835-01	-3.71070727E-01 -3.35216110E-03 -2.79960707E-01	3.22691552E-01 2.01424361E-01 1.64783890E-01	3.229371325-01 -1.201129675-01 -7.310903735-02	3,226915525-01 -2,35584479E-01
AL FREQUENCIES (R	N	2.17377127E+D2	NG MODE SHAPES FOI	2	-2.44301291E-03 -8.04528241E-03 -5.86504753E-01	-2+42282525E-03 -1.26662378E-02 -5.89846423E-01	3497825501£-07 -7.93330387E-03 6.630315955-05	3.62913769E-07 -1.26567891E-02 6.16871752E-05	-3.12289330E-86 5.50956353E-01 6.84475132E-05	2+44374196E-63 -8+84524847E-03 5+86624459E-01	2.42355799E-03 -1.26661782E-02 5.89166364E-01	<u>1.40958166E-03</u> -2.62964036E-01 7.46027047E-01	<u>1.33559793E-03.</u> -2.82951430E-01 7.45567299E-01	<u>1.37163774E-03</u> -2.76640912E-01
NATUR		2.17351330E+D2	CORRESPONDI		-7 <u>-6458248995-84</u> -3,096532695-04 8.73360464E-01	-7 <u>*58548871E-04</u> -4*30481914E-84 8*72368550E-81	-2*59211498E+84 4*31750705E-07 9*67557094E-01	-7.77723368E-84 5.36324826E-87 8.97323094E-81	-5.36389826E-02 -2.44923047E-05 1.0000000£400	-7.64617519E-84 3.18526176E-84 8.73388045E-01	-7.58582668E-84 4.31476741E-04 8.72316169E-81	-1.59738033£-63 6.42685210E-01 4.99994558E-81	-1.56867247E-03 6.42781317E-01 4.99568899E-01	-1.57812693E-03 6.39983011E-01
				ELEMENT REFERENCE	X- TRANS. Y- TRANS. Z- TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. V-TRANS.
				NODAL POINT NUMPER		~~~	ભ ભાગ	***	տտտ	ى تە ت	~ ~ ~	c s «	σσσ	10 10

+++

-8.87262873E-01 -7.53923539E-01 1.35013550E-01

1.28437956E-01 1.28832117E-01 -2.60888078E-01

9.99754420E-01 7.29371316E-01 -1.46782908E-01

1.51243315E-03 4.94111533E-01 3.80239355E-01

-2.95658276E-03 7.68073416E-01 4.32949065E-01

X-TRANS. Y-TRANS. Z-TRANS.

111

-8.87804878E-01

1.20012165E-01

1.00000000<u>0E+00</u>

1.49663707E-03

-2+94442421E-03

X-TRANS.

•						And the second se		and the second																							+ 	
	5	1.00000000E+00 -1.88617896E-01	-3.51327913E-05 -2.45203252E-01	4.29319699E-01	-2.621138215-05	3.363423635-91 -5.74525745E-01	1.58590786E-J3	-2.41133211E-01 4.24769648E-01	9.88346893E~01	2.64878049E-01 7.23035230E-01	8 887/58835-01	-3.35392954E~01	-9.59391599E-01	1.81192412E-05	2,68726287E-01	- 3. 4906399/E-Ub	1.91734417E-05 -1.98563695E+01	4.78102981E-06	1.80487805E-05	-3.39241192E-01 5.31544715E-06	- 4. 887468975+01	2.64878049E-01	TN_3NC3CCNC3 + /_	-9.383468835+01 -3.364471645-01	9.59991599E-01	-3.69351491E-05	-2.45094851E-01 -4.29364439E-01		-4.64281843E-05 3.32791328f-01	5.74525745E-01	-1,75609756E-03	
	4	-1.42274939E-01 2.76094391E-01	-3.61951314E-01 -1.04987835E-01	-7.98114355E-01	-3.62895377E-01	4.07846715E-01	-7.58515915E-01	-2.84124088E+01 -5.48965937E-01	-3.41240876E-01	-2.99635036E-01 -5.10765423E-01		3.029905356-01	5.34549878E-01	3.92700730E-03	-2.99939173E-01	2.555578852E-US	3.933090026-03 1.55778599E-01	-2.09975669E-03	9.92700730E-03	3.03406326E-01 -4.11922141E-03	3.60705596F-01	-2.993309006-01	10-300001007°C	9.61313869E-01 7.02676309E-01	-5.53527981E-01	3.55413625E-01	-9.27007299E-02 4.09124088E-01		3.56447689E-01 7.14720195F-02	-4.19829684E-01	7.45133820E-01	•
R COMPLETE SYSTEM	F	-7.573673975-01 1.637524565-01	1.933939105-01 3.7966661225-01	-3.49705705E-01	1.939341355-01	-3.833447046-41 3.796660122E-01	4.00540275E-01	5.181728985-01 -3.4996956655-01	-6.15667976E-01	<u>6.47839c005-03</u> -5.63113049E-01	-6 16/333075-01	-0.134263315-01 -5.333938212-03	5.972495095-01	-6.19597250E-01	-4.204322205-03	-I.79543222E+UI	-6.200894095-01 2.185412575-03	1.316552065-01	-6.195972505-01	4.255589391E-03 2.58104126E-01		-1.48747544E-02	TU#367T67604+6-	-5,915011795-01 1.782868666-02	5.820235765-01	2.034872305-01	-3.823673975-01 -2.384096445-01		2.04075f215-01 3.86051f915-01	3.681237725-01	4.21660118E-01	· · · · · · · · · · · · · · · · · · ·
ING MODE SHAPES FOR	N	4.90792256E-01 3.00832976E-01	-2.44133329E-05 8.28495119E-01	4.92840928E-01	4.93961797E-04	r.659553826E-01 4.59153826E-01	5.38364987E-D2	9.95880245-01 2.687343235-01	-5-43841413E-04	1.0000000E+00 5.91606768E-01	10 10 10 10 10 10 10 10 10 10 10 10 10 1	-7.36410594E-44 9.98522579E-01	5.92989649E-01	1.22584663E-07	9.98224417E-01	-4.21099122E-U5	1.17732579E-07 9.97295519F-01	-4.20564779E-05	1.20228086E-07	9.97606258E-01 -4.15482996E-05	5 441405005-04	9.999999779E-01	-9-916U//3UE-UI	5.383165345-04 0.000022666601	-5.92990972E-01	2.43401510E-05	8.25437206E-01 -4.92808538F-01		-4.93978346E-04 7.658984846-01	-4.59123979E-01	-5.38327712E-02	
CURRES=0ND]	F	7.64954128E-01 4.29695377E-01	1.10464596E-04 4.21694583E-01	2.37541799E-01	3.70019353E-04	3.9584376UE-UI 2.15227415E-01	2.69010444E-02	2.26490270E-01 6.07667741E-01	3.68916742E-03	1.45328090E-03 -5.48291723E-03	70 - 100 - 100 - F	3.009/7191995-03 9.56003308E-04	-1.19730161E-02	3.14894165E-03	-4.45210559E-05	-5.13274776E-01	3.075295005-03 -4.4473586AF-05	-5.13914779E-01	3.11015996E-03	-4_445153155-05 -6_070078745-01	3 - 6881 7076E-03	-1.542478995-03	->.430110405-03	3.66976496E-03 	-1.19202667E-02	1.10512526E-04	-4.21768507E-01 2.3758580AF-01		3.70146751E-04 -7.07012060E-01	2,152692165-01	2.69063978E-02	
	EL EMENT REFERENCE	Y-TRANS. Z-TPANS.	X-TRANS. Y-TRANS.	Z-TRANS.	X-TRANS.	Z-TRANS.	X-TRANS.	Y-TRANS. Z-TRANS.	X-TRANS.	Y-TRANS. Z-TRANS.		Y-TRANS.	Z-TRANS.	X-TRANS.	Y-TRANS.	Z- TRANS.	X-TRANS.	Z-TRANS.	X-TRANS.	Y-TRANS. Z-TRANS.	V- TBANC	Y-TRANS.	-1 KANS.	X-TRANS.	Z-TRANS.	X-TRANS.	Y-TRANS. 7-TRANS.		X-TRANS.	Z-TPANS.	X-TRANS.	
	NOPAL POINT NUMBER	12 12	13 51	13	14	1 t	15	15 15	16	16 16	:	17	17	18	80 (15	0`0 **	19	20	20 20	10	212	12	22	22	23	50 M	2	24	t đ	25	

CORRESPONDING MODE SHAPES FOR COMPLETE SYSTEM

NUMBER	ELEMENT	4	2	8	4	5
25	V-TRANS.	-2.265731675-01	9.95848202F-01	-5.2578585856-01	-2.67274939E-01	-2.409756105-01
25	Z-TRANS.	6.37692423E-01	-2.68652123E-01	-3.33497C532-01	5.59354015E-01	-4.24769648E-01
26	X-TRANS.	~2.95637765E-03	-1.51277633F-D3	9.96070727E-01	-1.52554745E-01	3.87262873E-01
26	Y-TRANS.	-7.50117493E-01	4.94007192E-01	-7.25687623E-01	1.52311436E-01	-7.53329539E-01
26	Z-TRANS.	4.32975376E-01	-3.00180144E-01	-1.39415521E-01	2.65450122E-01	-1.35013550E-01
27	X-TRANS.	-2.94421518E-03	-1.49698350E-03	9.963163065-01	-1.52128954E-01	9.87262873E-01
27	Y-TRANS.	-7.54397375E-01	4.90687333E-01	7.529469555-01	-1.66605839E-01	1.00000000E+00
27	Z-TRANS.	4.29722061E-01	-3.00773716E-01	1.559430265-01	-2.81204380E-01	1.98672097E-01
28	X-TRANS.	-1.59723590E+03	-1.40936794E-03	3.06483301E-01	-5.84489051E-01	3.93550136E-03
28	Y-TRANS.	-5,42559981E-01	-2.83052187E-01	-1.99453725E-01	7.420924576-02	-1.31327913F-01
28	Z-TRANS.	5.00061108E-01	-7.45958561F-01	1.707760315-01	2.09184915E-01	2.28780488E-01
62	X-TRANS.	-1.56059758E+03	-1.33537968E-03	3.06729890E-01	-5.84854015E-01	3.93712737E-03
56	Y-TRANS.	-6.42756056E-01	-2.83039644E-01	1.192956785-01	-6.69099757E-02	9.75067751E-02
62	Z-TRANS.	4.99627302E-01	-7.45498597E-01	-7.68172888E-02	-1.30048662E-01	-1.63739837E-01
30	X-TRANS.	-1.57804869E-03	-1.37142237E-03	3.064833015-01	-5.84306569E-01	3.93387534E-03
30	Y-TRANS.	-5.39958309E-01	-2.76728754E-01	2,319744605-01	-1.33272506E-01	1.72953930E-01
30	Z-TRANS.	5.01671159E-01	-7.42081213E-01	-1.50834971E-01	-2.51946472E-01	-2.93224932E-01
31	X-TRANS.	-2.03274563E-05	1.16369554E-07	3.48722986E-03	-3.94160584E-06	4.751761525-05
31	Y-TRANS.	3.19899868E-05	-7.17412968E-01	2.529460555-03	1.80352798E-01	9°40379404E-04
31	Z-TRANS.	-6.16432244E-D1	-4.23244294E-05	-1.673870332-01	2.67153285E-03	-4.68997290E-06
32	X-TRANS.	-2.02422691E-05	1.16366734E-07	3.487229865-03	-3.939172755-06	4.75067751E-05
32	Y-TRANS.	3.200535885-05	-7.180272195-01	-1.777259335-03	-1.26703163E-01	3.00379404E-03
32	Z-TRANS.	-6.17769949E-01	-4.23227614E-05	1.18492141E-01	-1.88907786E-03	5.52345528E-06
33	X-TRANS.	-2.02913627E-05	1.16372161E-07	3.487229A6E-03	-3.94038929E-06	4.751761525-05
22	Y-TRANS.	3.199559946-05	-Z.17555508E-01	6.59341C61E-04	4.77554745E-02	1.83035230E-03
23	Z-TRANS.	-6.16989651E-01	-4.23223351E-05	-4.39096267E-02	7.01946472E-04	-2.996747975-07

-

-

111

226

ļ

BLES		.	P	6	10
×	4.95876587E+04	6.16583024E+04	5.83885038E+04	1.07689532E+04	5,77606280E+03
	2.61634448E+04	2.30648816E+04	3.69414879E+84	8.14582536E+83	1.80811712E+03
. 2	2.63276932E+04	3.91984612E+04	2.17851152E+04	34 38 05 24 38 E+ 03	4.37789279E+03
	1.91219536E+01	-1.60387525E+01	1.83574812E+02	-2.78692195E+02	1.22624238E+01
	-2.56508463E+01	4.46301049E+01	-2.33262968E+06	3.88844061E+05	-9.50505703E+02
2	1.03682271E+01	-2.07531475E+01	2.32055734E+86	-3.80791838E+05	8.95655019E+02
H DIAGONAL	5.18393983E+04	6*20608226E+04	5.84775534E+04	1.11472515E+84	5+98103636E+03

Notice for the states and states	and a second		<u>5</u>	4	40	£	10	
CONRESPONDING CONNECTION A	-		5.42350223E±02	6.33°68051E+02	6.34553229E+02	3.57333355+02	<u>9. 57550540E+02</u>	
NUMBER F. FAMS J. STATASTER J. MAY STAR. (2) J. MAY MAY STAR. (2) J. MAY MAY STAR. (2) <thj. may="" may<="" th=""><th></th><th></th><th>CORRESPONDI</th><th>ING MODE SHAPES FOR</th><th>SOMPLETE SYSTEM</th><th></th><th></th><th></th></thj.>			CORRESPONDI	ING MODE SHAPES FOR	SOMPLETE SYSTEM			
L FTANK 1.5777377-01 7.51/3615-02 3.133842.08-01 4.03751.8-13 7.12311095-01 Z FTANK -1.50013015-01 5.4005016-01 1.13041318-01 1.10341316-0 1.10413142-01 Z FTANK -1.5050000-01 1.3000000-01 1.3000000-01 1.13051438-01 1.14141916-05 Z FTANK -1.50515600-01 1.4000000-01 1.300000-01 1.13051438-01 1.13041406-01 Z FTANK -1.50515600-01 1.4000000-01 1.300000-01 1.13054038-00 1.1304000-05 Z FTANK -1.50515600-01 1.4000000-01 1.2000000-01 1.13054038-01 1.13054038-01 Z FTANK -1.50515600-01 1.4000000-01 1.2000000-01 1.13054038-01 1.13054038-01 Z FTANK -1.50575600-01 1.4000000-01 1.2000000-01 1.13054038-01 1.13054038-01 Z FTANK -1.10000000-01 1.20000000-01 1.120200000-01 1.13054028-01 1.13054028-01 Z FTANK -1.1000000000 1.140000000-01 <th>NODAL POINT NUMBER</th> <th>EL ENENT REFERENCE</th> <td>2</td> <td>6</td> <td>ŝ</td> <td>G</td> <td>10</td> <td></td>	NODAL POINT NUMBER	EL ENENT REFERENCE	2	6	ŝ	G	10	
I C: TRMS -1: 0: 0: 0: 0: 0: 0: 0: 0: 0: 0: 0: 0: 0:	٩							
Interpretation Construction Constructio		Y-TPANS	1.578793775-01 -4.367380115-01	-7,51736915E-02	3.30368268E-03	3.98575513E-03	7.023316895-32	
Z F. TRMS 1.5826M12E-01 -7.4/155411E-02 -1.43554312F-01 -2.4125997E-02 Z F. TRMS -5.2599926E-01 0.10051015E-01		Z-IRANS.	-9.68871595E-01	6.68828161E-01	2.2.09690359E-01	-3.88582737E-02	-1.564339345-01 -7.38414838E-02	
Z FTMMS. -C.959500E-01 -0.9050437E-01 -0.9050437E-01 -0.9050437E-01 -0.9050437E-01 -0.9050437E-01 -0.40151407E-05 -0.4015147E-05 -0.4015147E-01 -0.4015147E-01 -0.4015147E-01 -0.4015147E-01 -0.4015147E-01 -0.4015147E-01 -0.4015147E-01 -0.4015147E-01 -0.40151477E-01 -0.40151477E-01 -0.40151477E-01 -0.40151477E-01 -0.40151477E-01 -0.40151477E-01 -0.40151477E-01 -0.40151477E-01 -0.40151477E-01 -0.401514777E-01 -0.40151477E-01	2	X-TRANS.	1.58268492E-01	-7.47105141E-02	3.193589136-03	9.15589956E-03	7.24122997E-02	
R FTANS	~ ~	Y-TRANS. 7-TRANS.	-2.95428016E~01 _6 286002225_01	9.096804386-01 6 103309875-01	1.92470170E-03	-5.25525315E-04	-4.64703335E-02	
3 FTRNKS				TR-J/GCNCCAT+0	Lu-36+00+100 at	70-300472007-02-	3•71 384 14 4E - N3	
3 Y.TAMS. -1.073307165-01 1.03034075-06 1.0502055-05 -5.11375055-05 -1.1040536-01 4 Y.TAMS. -3.077830175-05 1.307495116-05 1.307495116-05 1.307495126-01 -1.3075055-06 -1.3075055-06 -1.3075055-06 -1.307565526-01 -1.307565526-01 -1.307565526-01 -1.307565526-01 -1.307565526-01 -1.3075656-01 -1.3075666-01 -1.30555526-02 -1.30555526-02 -1.30555526-02 -1.30555526-02 -1.30555526-02 -1.30555526-02 -1.30555526-02 -1.30555526-02 -1.3055666-04 -1.3057656-06 -1.35565526-02 -1.356566126-04 -1.3057656-06 -1.357676-05 -1.3567676-06 -1.3576756-01 -1.3576756-01 -1.3576756-01 -1.35677576-05 -1.35676766-01 -1.3567676-01 -1.35676766-01 -1.35676766-01 -1.35676766-01 -1.35676766-01 -1.35676766-01 -1.35676766-01 -1.35767666-01 -1.35676766-01 -1.35676766-01 -1.35767676-02 -1.377776-02 -1.377776-02 -1.37776766-02 -1.377767666-02 -1.377776-02 -1.377776-02 -1.377767666-02 -1.377767666-02 -1.377776-02 -1.3777767666-02 -1.377776766	F	X-TRANS.	-3+05155642E+05	1,37795276E-05	3.96034898E-03	1.02342698E-02	-3.13469034E-05	
4 KTRMS -1.07720101-05 1.0372001E-05 1.0372001E-05 -1.07720101E-05 -1.07720101E-05 4 ZTRMS -2.055552E-05 2.050552E-01 1.0375001E-01 -2.055525E-01 -2.0555251E-05 -2.0555251E-05 -2.0555251E-05 -2.0555525E-01 -2.0555251E-05 -2.0555251E-05 -2.0555251E-05 -2.0555251E-05 -2.0555251E-05 -2.0555251E-05 -2.0555251E-05 -2.0555251E-05 -2.0557301E-05 -2.0557301E-05 -2.057301E-05 -2.057301E-	10 10	Y-TRANS. Z-TRANS.	-4.95330739E-01 1.08949416F-05	1.00000099E+00 -9.12922649E+06	1.76920282E-05	-5.44795660E-04 -8.71387603E-04	-1.18406236E-01	
4 F. F. RMIS - 7.1782401E-05 1,39706057E-05 1,49752055F-05 1,49752055F-05 1,4075205E-05 2,17520305E-02 2,13731395E-05 2,13731395E-05 2,13731395E-05 2,13731395E-05 2,13731395E-05 2,117301305E-05 2,117301305E-01 2,117301305E-01 2,117301305E-01 2,117301305E-01 2,117301305E-01 2,117301305E-01 2,117301305E-01 2,117301305E-01 2,1174175015E-01 2,117417505E-02 2,117417505E-02 2,111177205E-02 2,1117			10 10 4 L C C C C C C C C C C C C C C C C C C	20452550775 40	7A-37AA502047			
\mathbf{r}	-	X-TRANS.	-3,077521015-05	1.39740621E-05	3.05265527E-03	1.12728980E-02	-3.43915115E-05	
F KIRNS -1.10001015-05 2.137351F-15 -1.11001016F-05 2.10102706F-01 -0.01001006+00 F KIRNS -1.10011015F-05 1.150101015F-01 1.1601010016-01 -0.010010016-01 F KIRNS -1.1707317E-05 1.150101016F-01 1.160101016F-01 -0.010010016-01 F KIRNS -1.1707317E-01 -1.5173931E-01 -1.5173931E-01 -1.00010016-01 F KIRNS -1.17087116-01 -1.100101016-01 -1.00010016-01 -1.00010016-01 F KIRNS -1.17087116-01 -1.100101016-01 -1.00010016-01 -1.001010016-01 F KIRNS -1.100101016-01 -1.100101016-01 -1.0010101016-01 -1.001010016-01 T KIRNS -1.100101016-01 -1.100101016-01 -1.00101001010-01 -1.0010100101010 T KIRNS -1.140127217E-01 -1.1409230520-01 -1.1409230520-01 -1.1409230520-01 F KIRNS -1.140127217E-01 -1.140927017E-02 -1.140927017E-02 -1.140927017E-02 P F KIRNS -1.	8-4	T-TRANS.	-2.955252926-01	9.10606762E-01 2 250495575-06	1.45842489E-05	-2.13721030E-04	-4.65369511E-02	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	r I	4-1 MAN30	-1. TUDUTT3E-U5	2+39445965= 00	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1		2.U1082720E-04	
5 7 - TRMS. 3.66616775-01 6.373320365-01 1.400004555-05 4.600320365-01 3.1003000067-00 6 VTRMS. -1.7573775-01 6.3733775-01 3.53755056-01 4.600330065-02 4.60030005705 7 VTRMS. -1.75737755-01 7.517951057-01 7.537550767-01 7.51795105707 7.51795105707 7 VTRMS. -1.570377576-01 7.5179530566-01 7.5179570577 3.153053056-02 -1.623335557770 7 VTRMS. -1.900373756-01 7.497503197-01 2.1095392257-03 3.130657475705 -1.647733355-02 7 VTRMS. -1.900273757-01 -0.400730357-01 7.497503526-02 -0.401733356-02 -0.401733356-02 8 VTRMS. -1.90027375-01 -0.40073026-01 -0.40073026-02 -0.401737576-02 -0.40173356-02 9 VTRMS. -1.400272775-01 -0.40073026-01 -0.40073026-02 -0.401737576-02 -0.40173356-02 9 VTRMS. -1.400277775-01 -0.400730760-02 -0.400730260-02 -0.400730260-02 9 VTRMS. -1.	2	X-TRANS.	-4.866731525-05	2.40435337E-05	-1.71021818E-02	2.81262047E-02	-9.172802085-05	
C. Trans. C. M. 1.579317E - 01 T. 1.9450195E - 01 C. 1.9476317E - 01 T. 1.947643195E - 01 C. 1.94763317E - 01 C. 1.9477570E - 01 C. 1.94777570E - 01 C. 1.94777570E - 01 C. 1.94777570E - 01 C. 1.9477570E - 01 C. 1.9477570E - 01 C. 1.947550E - 02 C. 1.933955757 - 01 C. 1.9339555757 C. 1.9475631546 - 01 C. 1.9339555757 C. 1.947550E - 02 C. 1.94927217E - 01 C. 1.94927217E - 01 C. 1.94927217E - 02 C. 1.94927297E - 02 C. 1.94927297E - 02 C. 1.94927297E - 02 <thc. -="" 01<="" 1.94977777e="" th=""> C. 1.949777727E</thc.>		V-TRANS.	3.886196775-01	6.37332038E-01	1.16000495E-05	4.60302705E-03	1.00030000E+00	
6 \mathbf{X} - TRANS. -1.57079377E-017.51780915E-01-3.15315530E-03-7.0334777E-036 \mathbf{V} - TRANS. -1.57079371E-01-3.0563715F-01-3.153315530E-03-7.0334777E-027 \mathbf{Y} - TRANS. -1.56357575E-015.6505715F-01-3.153315530E-03-7.2735526-127 \mathbf{Y} - TRANS. -1.56357575E-017.475603195F-01-1.109555322E-039.466733556-027 \mathbf{Y} - TRANS. -1.59357575E-017.475603195F-01-1.109555322E-039.46673355F-027 \mathbf{Y} - TRANS. -2.9942015F-01-1.10930975E-01-1.109355322E-039.46673355F-027 \mathbf{Y} - TRANS. -1.49027237F-01-5.053917305E-02-1.209555326E-039.466739556-029 \mathbf{Y} - TRANS. -1.49027237F-01-5.93173726E-02-1.5935634E-02-7.24956056E-029 \mathbf{Y} - TRANS. -1.49027337F-035.56739315F-02 2.93290726E-01 -1.59956376F-02-1.5193955676-029 \mathbf{Y} - TRANS. -1.49027337F-035.56739315F-02 2.93907366E-02 -1.519376576576-029 \mathbf{Y} - TRANS. -1.49027337F-035.56739315F-02-1.5939126F-02-1.5173776F-029 \mathbf{Y} - TRANS. -1.49027337F-035.56739315F-02-1.6939126F-02-1.51737276F-029 \mathbf{Y} - TRANS. -1.49027377F-035.56739316F-02-1.5939126F-02-1.51937266F-029 \mathbf{Y} - TRANS. 2.18AN956575F-01-2.93908366F-02-1.5939126F-02-1.51937266F-029 \mathbf{Y} - TRANS. 2.164035657-01-2.93907366F	2	- CHARTI-2	2*17996109E-U5	-1.54284391E-05	_ 1.95224654E-01	-1.01472008E-01	2.67345171E-04	
6 T-TRMNS: -4.95039115-01 9.650371555-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.95039155-01 -4.9505755-01 -6.10339915-01 -4.95055555-01 -6.10330915-01 -4.95055555-01 -6.10330915-01 -4.950555556-03 -4.0505545-03 -4.050556-03 -4.0505565-02 -4.05055656-03 -4.05055656-02 -4.05055656-02 -4.050556676-02 -4.050556676-02 -4.050556676 -6.0512667575-02 -6.103709727275-01 -6.1037097670 -6.1037075672 -6.104077576702 </td <th>Ŷ</th> <th>X-TRANS.</th> <td>-1.57879377E-01</td> <td>7.51736915E-02</td> <td>3.30847024E-03</td> <td>8.33875340E-03</td> <td>-7.0334777E-02</td> <td></td>	Ŷ	X-TRANS.	-1.57879377E-01	7.51736915E-02	3.30847024E-03	8.33875340E-03	-7.0334777E-02	
6 Z-TRANS. 8.58071595C-01 6.6012016E-01 2.09555747E-01 -3.01749306E-02 7.0147550E-12 7 X-TRANS. -1:50357596-01 7.475603197E-01 -1.09555637E-03 9.00572427E-03 -7.647560316E-02 7 X-TRANS. -2:95420165-01 7.475603197E-01 -1.09555637E-03 9.00572427E-03 -4.647333355-02 7 X-TRANS. -2:9542016571 -0.1033307E-01 -1.030723572-03 9.00572457E-02 -5.13495656E-01 8 XTRANS. -1.49027237E-01 -2.85272716E-02 -5.954661716-02 -5.34956926E-02 8 XTRANS. 2:54005037E-01 -2.8529720E-02 -4.53471624E-02 -5.34956562-02 9 X-TRANS. 2:54005037E-01 -3.9392022 -5.34956562-02 -5.56739216-02 -5.349569562-02 9 X-TRANS. 2:14455427 -3.9316246-02 -4.54377844E-02 -5.3445545562-02 -5.3445545562-02 9 X-TRANS. 2:144542576-01 -5.99773299-02 -4.24327994E-02 -4.1495556562-02 -5.3445426702 -5.3445545567-02 9 X-TRANS.<	Q	Y-TRANS.	-4.950389115-01	9.96294531E-01	-3.15315530E-03	3.53888208E-04	-1.22433954E-01	
7X: TRANS1:56365795917.475663195-013.1995363262-039.406574256-03-7.245560565027Y: TRNS2:95230165-019.405555326-039.406574256-05-4.4733356-026K: TRNS1:490272375-01-1:490372376-01-1:490272376-02-5.3934056356-026K: TRNS1:490272375-01-5.3520013906-01-4.3377056401-4.35770564016K: TRNS1:490272375-01-5.393013906-02-4.524764376-02-5.39344164-029X: TRNS1:490272375-01-5.393013906-02-4.3377056401-4.33770564019X: TRNS1:490272375-01-5.39391266-02-4.1294656176-02-5.39344166-029X: TRNS1:49027375-01-5.99377356-02-4.432726401-4.129660616-029X: TRNS.2:194241256-01-5.9990826602-4.129460206-029X: TRNS.2:194241256-01-5.99777326-01-4.394952996-02-4.129600206-029X: TRNS.2:194241256-01-5.991773266-02-4.43272946-02-5.30341266-029X: TRNS.2:19441256-01-5.991773266-02-4.43272946-02-5.30341266-0210X: TRNS.2:19441256-01-1.972302246-01-4.39495295666-0210X: TRNS.2:09139326576-01-4.923515366-02-4.12960206-0210Y: TRNS.2:091393266-02-4.9437255656666176-02-5.30344466-0211Y: TRNS.1.55593355576-01-5.9977722912246-02-4.24372556666176-0211Y: TRNS.	9	Z-TRANS.	8.588715955-01	-6.68328161E-01	2.09665747E-01	-3.81748906E-02	7.401472505-32	
7V-TRANS. $-2.954280165-01$ $9.09650365-03$ $9.0667242365-05$ $-4.64773355-05$ 7Z-TRANS. $5.2859922255-01$ $-6.10930977-01$ $1.8022039565-01$ $-9.105366476-02$ $-5.184062365-03$ 6X-TRANS. $-1.490272375-01$ $-6.10930977-02$ $2.95227205-02$ $-4.5824754375-02$ $-5.64754375-02$ $-5.6479325-02$ 6X-TRANS. $-1.490272375-01$ $-5.313726401-01$ $-4.35477082647-01$ $-4.35477082647-02$ $-5.93441646-03$ $-1.314854927-02$ 9X-TRANS. $-1.490272375-01$ $-2.5930037666-01$ $4.377026407-02$ $-2.9394912616-02$ $-5.137666569595-02$ 9X-TRANS. $-1.490272372-01$ $-2.939432316-02$ $-5.93941646-02$ $-5.4377946-02$ $-5.31357945-02$ 9X-TRANS. $-1.490272372-01$ $-2.9379326-02$ $-4.24377945-02$ $-3.13555526-02$ $-4.24377945-02$ $-3.13555526-02$ 9X-TRANS. $-1.400372376-01$ $-2.9974326-01$ -5.937106202 $-3.13555526-02$ $-4.2437708576-02$ $-3.13555526-02$ 9X-TRANS. $-1.40037656-01$ $-2.991416-02$ $-5.997172821266-02$ $-4.24372945-02$ $-7.135555526-02$ 10X-TRANS. $2.000000306-01$ $-3.8657133856-01$ $-3.657065666136-02$ $-3.135555626-02$ 11Y-TRANS. $1.400376956-01$ $-3.665666136-02$ $-4.243729946-02$ $-4.110870516-02$ 11Y-TRANS. $1.55933556-01$ $-3.66573386-02$ $-4.52372106-02$ $-6.33694246-02$ 11Y-TRANS. $1.559349476-02$ $-3.$	7	X-TRANS.	-1.58365759E-01	7.47568319E-D2	3.19839282E-03	3.48953742E-03	-7.245560856402	
7Z-TRANS.5.265932226-01-6.10330976-011.430203936-01-0.105365366-036 $K - TRANS.$ -1.490272375-035.562505375-022.932222024.524754375-02-5.184054926-020 $V - TRANS.$ 2.540856056-014.377026406-022.932927205-024.524754375-02-5.184054926-029 $V - TRANS.$ 2.540856056-014.377026406-022.2919902365-024.5527606126656-039 $X - TRANS.$ -1.4490272375-012.997494216-01-5.991942266-024.129060206-039 $X - TRANS.$ -1.4490272375-015.567392316-022.5919902376-024.12906006-039 $X - TRANS.$ -1.4490272375-01-2.997494216-01-5.991708276-02-4.43272946-02-5.44653695026-039 $X - TRANS.$ -1.4490272375-01-5.997494216-01-5.991708276-02-4.129360206-039 $X - TRANS.$ -1.4490272375-01-5.997494216-01-5.991708276-02-4.129365026-029 $X - TRANS.$ -1.4490272375-01-5.997494216-01-5.997106266-5.9465365136-0310 $X - TRANS.$ -1.4490255366-013.365215366-02-4.22477436602-5.364565156-0210 $Y - TRANS.$ 1.455536566-01-2.997494216-01-5.997106266-5.9365956106-0310 $X - TRANS.$ 1.455536566-01-2.9371366-02-4.2245595666011 $Y - TRANS.$ 1.5555336566-01-3.441773766-02-4.1291050670211 $Y - TRANS.$ 1.555639561013.565539566-01-3.59315556-0112 $X $	~	Y-TRANS.	-2.95428016E-01	3.09680403E-01	-1.89555632E-03	9.88672423E-05	-4.64733355-92	
6 $N = TRANS$ -1.49027237E-035.527605317E-022.95202720E-024.57474345437E-025.457436435E-026 $V = TRANS$ 2.55406563E-014.377702640E-01-4.354710246-02-5.45366177E-02-5.46346435E-029 $X = TRANS$ -1.449027237E-012.93101390E-01-4.354710246-02-5.95666177E-02-5.46536957E-029 $X = TRANS$ -1.449027237E-015.56739231E-022.9399028E-024.572794E-02-5.46653695E-029 $X = TRANS$ -1.449027237E-015.5673931E-022.9990028E-024.537369456-02-4.12906506-039 $X = TRANS$ -1.449027237E-01-5.95479339E-02-4.594793926-02-4.24475965626-029 $X = TRANS$ -1.449027237E-01-5.99773299-02-4.244759996E-02-3.135556566-0210 $X = TRANS$ -1.44993269557E-01-2.491431227E-01-5.993773299E-02-4.24905996E-02-3.135556566-0210 $X = TRANS$ -1.4000000026-013.4651366-02-3.48471742E-02-5.48475956566-02-4.110870566-0211 $Y = TRANS$ 1.0000000026-013.65531586-02-3.503177426E-01-3.914975776-02-3.1107576-0211 $Y = TRANS$ 1.0000000026-013.65631586-02-3.5731264E-02-3.510175776-0212 $X = TRANS$ 1.0000000026+013.65631586-02-3.510175776-0213 $X = TRANS$ 1.555543566-01-3.5102757602-4.23940596666-0214 $Y = TRANS$ 1.0000000026+013.656315366-02-3.510275467215 $X = TRANS$ <th>7</th> <th>Z-TRANS.</th> <td>5.28599222E-01</td> <td>-6.10930997E-01</td> <td>1.55020593E-01</td> <td>-8.18536634E-02</td> <td>-5.18405236E-03</td> <td></td>	7	Z-TRANS.	5.28599222E-01	-6.10930997E-01	1.55020593E-01	-8.18536634E-02	-5.18405236E-03	
0 V-TRANS. 2.554065613E-01 -2.633001330E-01 -4.3577024E-02 -6.93341614E-03 -1.51405452E-02 9 X-TRANS. -1.449027237E-01 5.56739231E-02 2.51873272E-01 -2.55686177E-02 -5.067120635E-02 9 X-TRANS. -1.449027237E-03 5.56739231E-02 2.593930620E-02 -4.129062020E-03 9 X-TRANS. 2.11403255626-01 5.5673931E-02 2.593739420E-02 -4.53405995E-02 -5.846530526-02 9 X-TRANS. 2.10310051E-01 -5.9977329E-02 -4.23405995E-02 -5.846530526-02 9 X-TRANS. 2.10310051E-01 -5.393126-02 2.43471742E-02 -5.8405556526-02 10 X-TRANS. 2.4041431225E-01 -5.937108276-02 -5.845551366-02 -5.8465556526-02 10 X-TRANS. 2.4000000026-01 -5.94471742E-02 -5.8405556526-02 -5.845556526-02 10 X-TRANS. 1.0000000026-01 -5.930156156-02 -5.84655565626-02 -5.8455565626-02 11 X-TRANS. 1.55555555672 -5.930156156-02 -5.930556566 -5.855555656	9	X-TRANS.	-1.49027237E-93	5.62760537E-02	2.95292720E-02	4.52475437E-02	-5.838025125-02	
B Z-TRANS. 4.37702640E-01 4.37702640E-02 2.21873272E-01 -2.55606177E-02 5.06712603E-03 9 X-TRANS. -1.49027237E-03 5.5673931E-02 2.9999828E-02 4.53339126E-02 5.4663695E-02 9 X-TRANS. 2.13374126E-01 -2.998749421E-01 -5.999773295-02 4.53339126E-02 5.4663695E-02 9 X-TRANS. 2.13970651E-01 -2.998749421E-01 -5.99773295-02 4.5337796220 4.12996020E-02 9 X-TRANS. 2.14902753710827E-01 -5.937695050 4.12996020E-02 -3.13555556502 10 X-TRANS. -1.400326956-03 5.559591941E-02 2.4531495126E-02 -4.2943595050-02 -3.135555566-02 10 X-TRANS. -1.400326956-03 5.559591941E-02 -5.03369426-02 -3.13555056-02 10 X-TRANS. 1.000000000000000000000000000000000000	•0	Y-TRANS.	2.54085603E-01	-2.83001390E-01	-4.35471824E-02	-6.93341614E-D3	-1.314854926-02	
9 X=TRANS. -1.49027237E-03 5.56739211E-02 2.9930828E-02 4.53239120E-02 5.94663695E-01 5.99377329E-02 4.53239120E-02 5.94663695E-02 4.12906020E-03 4.12906020E-03 4.129366020E-03 4.129366020E-03 4.129366020E-03 4.129366020E-02 7.135565056-02 7.1355656566-02 7.1355656566-02 7.1355656566-02 7.1355656566-02 7.1355656566-02 7.1355656566-02 7.1365556566-02 7.1365556566-02 7.1365556566-02 7.1365556566-02 7.1365556566-02 7.1365565666-02 7.1365565666-02 7.136556566-02 7.1365565666-02 7.13655656666-02 7.13655656666-02 7.136556566666-02 7.136556566666-02 7.136556566666-02 7.136556566666-02 7.136556566666-02 7.136556566666-02 7.136556566666-02 7.136556566666-02 7.1365565666666-02 7.136556566666666666666666666666666666666	8	Z-TRANS.		4.37702640E-02	2.21873272E-01	-2.55686177E-02	-5.06712863E-03	
9 V-TRANS. 2:134741255-01 -2:93749421E-01 -5:99773299-02 -4.24327294E-02 4.12900002012 9 Z-TRANS. 3.68190651E-01 6.354793395-02 -4.24327294E-02 4.1290050206-03 10 X-TRANS. -1.460326955-01 6.354793395-02 -4.24327294E-02 -3.13555656276-02 10 Y-TRANS. -1.460326955-01 5.59591941E-02 2.46314742E-02 -5.430391126E-02 7.65565613E-01 10 Y-TRANS. 2.00000030E-01 -2.4914312257-01 -3.48471742E-02 -5.303991126E-02 7.64833265E-02 11 Y-TRANS. 1.55553565-01 3.46477536E-01 -3.49471742E-02 -5.303991126E-02 -4.11087051E-02 11 Y-TRANS. 1.55553565-01 3.46575356-01 -3.46471742E-02 -3.49471742E-02 -3.51175972956E-02 11 Y-TRANS. 1.55553565-01 3.46555556-01 -3.511755702 -3.51175772912 -4.11087051E-02 11 Y-TRANS. 1.0000000005+00 1.900000005+00 1.309153976-01 -9.511175776-02 11 Y-TRANS. 1.000000006+00	œ	X-TRANS.	-1.490272375-93	5.567392315-02	2.41488285-82	4.5323812AF-02	-6. 8466 369 0E-03	
9 Z-TRANS. 3.68100661E-01 6.35479335E-02 Z.16770827E-01 4.354059696-02 3.13555555E-02 10 X-TRANS. -1.460326955-03 5.59591941E-02 Z.16770827E-01 -4.354710827E-02 -5.1357105-02 -5.13555555E-02 10 Y-TRANS. -1.460326955-03 5.59591941E-02 Z.167717421E-02 -5.0339112165-02 -5.1355565556-02 10 Y-TRANS. 2.000000365-01 3.464717421E-02 -5.3339112165-02 -5.1355656566 11 X-TRANS. 1.555933565-01 3.4655174302-02 1.972302246-01 -4.410870515-02 11 X-TRANS. 1.555933555-01 3.766332656-01 -3.665733565-02 -4.410870515-02 11 Y-TRANS. 1.0000000102+01 3.766095415-01 1.000000000+00 1.309153976-01 -9.51017575-02 11 Y-TRANS. 1.0000000102+01 -3.6605762365-01 -3.650576201 -3.51017575-02 11 Y-TRANS. 1.0000000102+01 -3.650572355-01 -3.51017575-02 -3.51017575-02 11 Y-TRANS. 1.5555395616-01 -3.573355395-01	6	Y-TRANS.	2.134241255-01	-2.98749421E-N1	-5.997773296-02	-4.243272946-02		
10 X-TRANS. -1.460326956-03 5.59591941E-02 2.93019615E-02 4.52157110E-02 5.63563613E-03 10 Y-TRANS. 2.00000030E-01 -2.491431225-01 -3.46471742E-02 -5.33391126E-02 7.66565613E-03 18 Z-TRANS. 3.4044552556-01 3.46571742E-02 -5.33391126E-02 7.66565613E-02 18 Z-TRANS. 3.444552556-01 3.46571742E-02 -5.376462692E-02 -3.76483326E-02 11 X-TRANS. 1.55553365E-01 3.70313924E-02 1.97230224E-01 -4.41087051E-02 11 Y-TRANS. 1.00000000000 3.7660954416-01 1.0000000000 -4.41087051E-02 11 Y-TRANS. 1.00000000000 3.7660954416-01 1.300153975001 -3.510177575-02 12 X-TRANS. 1.000000000000 -3.606762396-01 -3.97229132E-01 -3.501429196-02 12 X-TRANS. 1.55544776-01 3.6573355396-01 -3.5057026-01 -3.506762396-01	6	Z-TRANS.	3.68190661E-01	6.35479339E-02	2.16770827E-01	-4.394059896-02	-3,135556526-02	
ID Y-TRANS. 2.00000030E-01 -2.43143122E-01 -3.48471742E-02 -5.30391126E-02 7.6656513E-03 11 Z-TRANS. 3.44455253E-01 3.36521538E-02 1.97230224E-01 -4.9215956E-02 7.6656513E-03 11 Y-TRANS. 1.55553365E-01 3.36521538E-02 5.5717430E-02 7.12242699E-02 -4.11087051E-02 11 Y-TRANS. 1.000000000000101 3.76639541F-01 1.00000000E+00 1.30915397E-01 -9.5101757F-02 11 Y-TRANS. 1.000000000E+00 3.76639541F-01 1.0000000E+00 1.30915397E-01 -9.5101757F-02 11 Y-TRANS. 1.00000000E+00 3.76639541F-01 -3.97229132E-01 -9.5101757F-02 12 Z-TRANS. 1.00000000E+00 3.76505356F-02 -3.6057239152E-01 -3.975397E-01 -9.5101757F-02 12 X-TRANS. 1.555549776-03 -3.60572391254E-02 5.53635539F-02 -4.1200F-02 -4.1200F-02 12 X-TRANS. 1.555544776-03 -3.6057239F-01 -3.67531254E-02 -4.230915397E-01 -4.22957044726-02	10	X-TRANS.	-1.488326455-03	5.585919416~02	2. 4381 961 6F - N2	4.5235711AF-02	60-37678-3 <u>-</u>	
I Z=TRANS. J.44455253E-01 J.86521538E-02 J.97230224E-01 4.9215956E-02 J.8000000102 11 X-TRANS. 1.55053366E-01 3.70313924E-02 5.57177430E-02 7.784269566E-02 -4.11087051E-02 11 Y-TRANS. 1.000000000E+00 3.766039541E-01 4.00000000E+00 1.30915397E-01 -9.51017757E-02 11 Y-TRANS. 1.000000000E+00 1.00000000E+00 1.30915397E-01 -9.51017757E-02 11 Y-TRANS. 5.25630934E-03 3.66676239E-01 -3.97229132E-01 -9.51017757E-02 11 Z-TRANS. 5.25630934E-03 3.66676239E-01 -3.97229132E-01 -9.51017757E-02 12 Z-TRANS. 5.25630934E-03 -3.660576239E-01 -3.97229132E-01 -9.96719100E-02 2.30142919E-02 12 X-TRANS. 1.55544777E-01 3.67531254E-02 5.833355332E-01 -9.236704200 -1.223670420	10	Y-TRANS.						
11 X-TRANS. 1.55059366E-01 3.70319824E-02 5.5717430E-02 7.16242699E-02 -4.11087051E-02 11 Y-TRANS. 1.000000006+00 3.76609541F-01 1.0000000E+00 1.50915397E-01 -9.51017757E-02 11 Y-TRANS. 1.000000006+00 3.76609541F-01 1.0000000E+00 1.30915397E-01 -9.51017757E-02 11 Z-TRANS. 5.2553019345-03 -3.606762396E-01 -3.97229132E-01 -9.94671900E-02 2.30142919E-02 12 X-TRANS. 1.555447476-01 3.65731254E-02 5.53635539E-02 9.024992465E-02 -4.22967042E-02	18	Z-TRANS.	3.44455253E-01	3.86521538E-02	1.97230224E-01	-4.92159568E-02	-3.76483326E-02	
11 Y-TRANS. 1.00000000E+00 3.76639541F-01 1.0000000E+00 1.30915397E-01 -9.5101757F-02 14 Z-TRANS. 6.256319345-03 -3.60676239E-01 -3.97229132E-01 -9.46719100E-02 2.30142919E-02 12 X-TRANS. 1.55544747E-01 3.67531264E-02 5.53835539E-02 0.02499246E-02 -4.22367042E-02	11	X-TRANS.	1.55059366E-01	3.70319324E-02	5.5717430E-02	7.78242699E-02	-4.110870515-02	
<u>11</u>	11	Y- TRANS.	1.0000000E+00	3.76639541F-01	1.0000000E+00	1.30915397E-01	-9.51017757E-02	
<u>12 X-TRANS.</u> 1.555447476-11 3.675312546-12 5 .538355396-02 0.024932466-02 - +.22367042E-02	11	Z-TRANS.	6.25681934E~03	-3.50575239E-01	-3,97229132E-01	-9.46719100E-02	2.30142919E-02	
	12	X-TRANS.	1.555447476-31	3.67531254E-02	5.53835539E-02	8.02493248E-02	-4.224670425-02	
						it for the second s		

NUMPEK	REFERENCE	ę				
12	Y-TRANS.	5.94455253°-01	3.55959134E-01	9.20564106E-01	-1.19203609E-02	-5.49155479E-02
12	Z-TRANS.	5.72470817E-02	-3.29087541F-01	-3.66027726E-01	-7.17139134E-02	-4.08055435E-03
13	X-TRANS.	-1.412451365-34	4.12976332E-03	-1.68176293E-03	-4.72652042E-03	-6.94278909E-03
	Y-TRANS.	2.535992222-01	1.110 <u>69940E-01</u>	5.02326428E-01	3.06201260E-02	-7.78258987E-02
	Z-TRANS.	-4.2470817101	-5.13221852F-01	-6.86190634E-01	-1.43645745E-01	2.41792984E-02
4 4 4 4 4	X-TRANS. Y-TRANS. Z-TRANS.	-1.132295725-04 1.535019465-01 -2.551556427-01	3.310729966403 1.03238550E-01 -4.77072719E-01	+1.22229892E-03 4.61443449E-01 +6.30515758E-01	-5.30397539E-03 -8.44241498E-03 -8.44241498E-03	-7.50541360E-03 -5.82070160E-02 -6.14994842E-03
15	X-TRANS.	1.795914+02-03	-1.58591941E-02	9.21717522E-03	-1.57745907E-02	-1.86487657E-02
15	Y-TRANS.	-1.85019455E-01	8.99027327E-03	3.56353244E-01	-6.38023822E-01	1.96491988E-01
15	Z-TRANS.	3.357003995-01	-3.71375637E-01	-4.180240815-01	9.95694987E-01	-4.69034214E-01
16	X+TRANS.	-1.554474717-01	-3.011579+3E-02	-5.84868727E-02	-8.59405731E-02	2.94716327E-02
16	Y-TRANS.	-4.955252927-01	-1.98100973E-01	2.40549834E-02	-2.02368779E-02	-1.92204417E-92
16	Z-TRANS.	-9.570078315-01	-6.99934692E-01	-9.57101972E-01	-1.79074881E-01	5.97228237E-92
17	X- TRANS.	-1.559338527-01	-2.99914729E-02	-5.80408587E-02	-8.85280879E-02	3.04330879E-02
17	Y- TRANS.	-3.881322965-01	-1.80361278F+01	1.84914966E-02	-2.85071427E-03	-5.85101776E-02
17	Z- TRANS.	-5.679988335-01	-6.42427050E-01	-8.84723679E-01	-6.68077149E-02	-6.57360546E-03
18	X-TRANS.	2.15561479E-06	-5.73876795E-07	-5.85860327E-02	-9.03905843E-02	2.24426159E-04
18	Y-TPANS.	-4.986391325-01	-2.00231599E-01	-3.82976947E-06	-9.37610551E-05	-2.15461239E-02
18	Z+TRANS.	2.38326848E-06	-6.61980500E-06	2.96924994E-01	-1.37816007E-02	3.56389047E-05
19	X-TRANS.	2.152723745-06	-5.85456233E-07	-5.79281858E-02	-9.05586917E-02	2.249159386-04
19	Y-TRANS.	-4.170233465-11	-1.96521538F-01	-3.32514242E-06	-2.25986877E-04	-5.002165446-02
19	Z-TRANS.	-2.029192986-06	-4.61273370E-06	3.21403990E-01	3.00674612E-02	-7.557384156-05
50	X-TRANS.	2.18190651E-06	-6.007410\$4E-07	-5.81319913E-02	-9.03760179E-02	2.24426159E-04
50	Y-TRANS.	-3.99383259E-11	-1.91472904E-01	-3.14431007E-06	-2.60621126E-04	-5.73841490E-02
50	Z-TRANS.	-3.37937743E-06	-2.67021759E-06	2.58411405E-01	3.96860012E-02	-1.00822867E-04
21	X+ TRANS.	1.55447471E-01	3.01157943E-02	-5.84846136E-02	-8.62211075E-02	-2.90385448E-02
21	Y- TRANS.	-4.95525232E-01	-1.98054655E-01	-2.40625495E-02	2.00695544E-02	-1.93243828E-02
21	Z- TRANS.	9.570038912-01	6.99934692E-01	-9.57078269E-01	-1.79626173E-01	-5.88133391E-02
22	X-TRANS.	1.559378526-01	2.99814729E-02	-5.80386650E-02	-8.88171507E-02	-2.99913382E-02
22	Y-TRANS.	-3.881322965-01	-1.90361278E-01	-1.84977631E-02	2.31959200E-03	-5.85101776E-02
22	Z-TRANS.	5.478988335-01	6.42427050E-01	-8.84702136E-01	-6.67465833E-02	6.90342139E-03
53 53 53 53	X-TRANS. Y-TRANS. Z-TRANS.	1.51653696E-J4 2.53501946E-01 4.2470A171E-01	-4.13385927E-03 1.11116259£-01 5.19221852E-01	-1.68038201E-03 -5.02323190E-01 -6.86173203E-01	-4.67521381E-03 -3.13447850E-02 -1.43860149E-01	6.86877436E-03 -7.76959723E-02 -2.34560416E-02
24	X-TRANS.	1.235408565-04	-3.31542381E-03	-1.22135597E-03	-5,24381560E-03	7.53572975E-03
24	Y-TRANS.	1.53501946F-01	1.03288560F-01	-4.61440358E-01	7,91528249E-03	-5.92503248E-02
24	Z-TRANS.	2.551556425-01	4.77072719E-01	-6.30499676E-01	-6,99132120E-02	6.48765699E-03
25	X-TRANS.	-1.371595335-03	1.58499305E-02	9.21360491E-03	-1.55675458E-O2	1.87786921E-02

		20	· · · · · · · · · · · · · · · · · · ·						
10	1.93297137E-01 4.64270247E-01	4.072325685-02 -9 <u>.445214385-02</u> -2.25533805 5- 02	4.183926335-02 -5.495845665-02 4.42515851E-03	5.81203934E-02 -1.31783653E-02 5.19272412E-03	5.82070160E-02 3.9255989 <u>8E-03</u> 3.15634474E-92	5.81203984E-J2 7.42312689E-03 3.78922001E-02	-1.00553014E-05 -8.69207449E-03 2.99313382E-05	-1.00476397E-05 2.443971806E-122 -9.18536163E-05	-1.00563014E-05 5.54352534E-03 -1.81333911E-05
6	6.39774711E-01 1.0000000E+00	7.81922355E-02 -1.31694716E-01 -9.08826046E-02	8.06288004E-02 1.14199909E-02 -7.16743190E-02	4.577906065-02 6.814088775-03 -2.552293865-02	4.58561990E-02 4.24694784E-02 -4.36534810E-02	4.57671396E-02 5.09075206E-02 -4.88707766E-02	7.57794193E-04 -3.95901430E-05 -1.16737598E-02	7.56797938E-04 1.11323679E-04 3.29086297E-02	7.57342991E-04 2.53600763E-05 7.51246695E-03
8	-3.56 349144E-01 -4.18012486E-01	5.57178102E-02 -9.99988441E-01 -3.97216669E-01	5,53835922E-02 -9,20552490E-01 -3,66016652E-01	2.95267844E-02 4.35373765E-02 2.21072733E-01	2.91956148E-D2 5.99674374E-D2 2.16768914E-D1	2。92994887E-02 3.48385496E-02 1.97228862E-01	2.840432085-04 -5.768815665-06 3.369082085-01	2.83846018E-04 -6.46203099E-06 3.65463079E-01	2.84039778E-04 -6.06227729E-06 3.48896117E-01
2	9.00990037E-03 3.71421955E-01	-3.70912459E-02 3.76655959E-01 3.60676239E-01	-3.67577592E-02 3.55859194f-01 3.29133459F-01	-5.62763537E-02 -2.430013905-01 -4.37841593E-02	-5,56739231E-02 -2, <u>98749421E-01</u> -6,35479383E-02	-5.59591941E-02 -2.49143122E-01 -3.855215395-02	4.645669295-06 -3.49894205E-01 -7.22556739E+06	4.645663295406 -3.91333951E-01 -5.51644280E-06	4.64566329E-06 -3.63131079E-01 -6.48449356E-05
و	-1.950194555-01 -3.3570038955+01	-1.559543565-01 1.00000005+00 -6.252918295-03	-1.554474715-01 6.944552532-01 -5.7255803325-02	1.466926075-03 2.540846036-01 -4.399872585-01	1.466976076-03 2.134241255-01 -3.681906515-01	1.464980545-03 2.00300005-01 -3.444552535-01	-9.225680935-06 1.785019465-04 2.29474798E-05	-3.222762655-95 3.747041715-04 -1.9134241255-05	-3.226653705-06 5.215926075-04 4.820038915-07
EL EMENT REFERENCE	Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TPANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.
NODAL POINT Numper	25	26 26 26	27 27 27	8 C C C	62 56	300 300 300	F. M. P.	32 32 32	33 33 33

ł

CORRESPONDING MODE SHAPES FOR COMPLETE SYSTEM

	11	21	13	19	15
	1.31532959E+04	1.36318685E+04	8.20837363E+03	1.38481395E+04	1.10906860E+04
	6.61479808E+03	1.27906844E+04	1.00846576E+04	9.12324109E+03	1.24273351E+05
	6.69924065E+03	1.26940061E+04	5.38717422E+03	1.69286570E+04	1.24311118E+05
	-4.99032160E+00	-2.17158100E+05	-1.84386186E+03	-3.75078907E+01	3+65542814E+05
	1.32640724E+01	8.45696133E+05	-1.07233413E+05	-1.04437073E+03	-8.44577469E+0 5
	3.74334725E+01	8.48944696E+05	9.53436229E+04	8.58899507E+02	-8.45047303E+05
ONAL	1.32336673E+04	1.95582795E+04	1.18401027E+04	1.99500187E+04	1.29837578E+05

GENERALIZED INERTIA PROPERTIES

11 12 13 13 CORRESPONDING 1.059590.054.01 1.733369626.01 1.3 CORRESPONDING 1.059590.054.01 1.1059590.054.01 1.3 CORRESPONDING 1.059590.054.01 1.1059590.054.01 1.3 CORRESPONDING 1.05199940.02 1.05199940.01 1.3 1 XTRANS 5.200464.02 5.401779466.01 1.1394523356.01 1 XTRANS 5.20149456.02 1.139455395.01 1.3 2 YTRANS 5.20149456.01 1.139455395.01 1.2 2 YTRANS 5.20149456.02 2.44529456.01 1.2 1.2 2 YTRANS 1.01330456.01 1.13774966.01 1.2 1.2 2 YTRANS 1.0139454516.01 1.13774966.01 1.2 1.2 2 YTRANS 1.0139445156.01 2.44579456.01 1.2 1.2 3 YTRANS 1.013945156.01 1.13774796.01 1.2 1.2 2 YTRANS 1.1374446.06 2.445959604.46	12	11		L
Lorrespondent Lorrespo			5 7	
CORRESPONDTIG HODE SHAPES F.R. CONRESPONDTIG HODE SHAPES F.R. 11 12 13 <th>7E+83 1.05969809E+03</th> <th>1. 20336962E+03</th> <th>1.20433575E+03</th> <th>1.37674831E+03</th>	7E+83 1.05969809E+03	1. 20336962E+03	1.20433575E+03	1.37674831E+03
NUMER LENKIT 1 12 13 14	RESPONDING MODE SHAPES FOR	COMPLETE SYSTEM		
1 X - TRANS. 3,6138946E - 02 -1.40545528E - 03 1 1.4057555E - 01 1 1 X - TRANS. 3,20046545E - 01 9,1657666 - 03 7,40525155E - 01 1 2 X - TRANS. 3,20046545E - 01 -1.454551646 - 01 -1.45456 - 01 -7 2 X - TRANS. -9,4999906 - 02 -1.45455126 - 01 -0 -1.4162365 - 01 -7 3 X - TRANS. -9,4999906 - 02 -1.45455126 - 01 -0 -1.4162365 - 01 -0 3 X - TRANS. -1.0195365 - 05 -1.425455282455 - 05 -1.425455282455 - 01 -0 -0 -0 -0 -1.4166 - 01 -1.425455282455 - 01 -0 0	12	13	14	15 .
1 Z - TRANS. -1.005/1080-012 9.4167/6000-02 1.4057/152705-01 -7 2 X - TRANS. -9.41999005-02 -1.455705-01 -7 2 X - TRANS. -9.41999005-02 -1.455705-01 -7 2 X - TRANS. -9.44999005-02 -1.4550175-01 -7 3 X - TRANS. -9.4999005-06 -1.45057175-01 -7 3 X - TRANS. -9.351705-01 -1.453013776-01 -4.4999915746-01 -4.4999195746-01 3 X - TRANS. -1.0571355-05 5.4230131276-01 2.4474556-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.49991576-01 -4.499195276-01 -4.49919566-01 -4.499195276-01 -4.4991956-01 -4.4991956-01 -4.4991976-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-01 -4.49919566-0	+6E-02 -7.00685250E-03	2.05755578E-01	-7.69559033E-02	-6.93589744E-01
Z Y-FRMS. 5.08105-02 -1.357576-61 -7 Z TRNNS. -9.4999900E-02 -1.35757095-03 -2.1455917-03 -1.357575104 6 Z TRNNS. -9.4999900E-02 -1.357570556-01 -1.35757556-01 6 Z TRNNS. -1.07511435-01 2.40705765-01 -1.4757556-01 6 Z TRNNS. -1.03511436-01 2.40705765-01 -1.4757556-01 6 Z -7.7RNS. -1.03511456-01 2.40705765-01 -1.4757556-01 6 Z 7 2.7581035556-05 -2.41640352566-05 -2.416404566-01 -1.4757556-01 6 Z 7 2.778NS -7.531035516-01 2.43752666-01 -1.4575756-01 -2.4575556-01 6 Z 7 2.401350516-02 -2.433752666 -1.4375756601 -1.457755601 -2.457555666 -1.437526666-01 -1.45775756666 -1.4375266666 -1.4375266666 -1.4375266666 -1.4375266666 -1.4375266666 -1.4375266666 -1.44747695666 -1.447474595975666 -1.4375266666	00E-01 9.16776860E-03 14E-02 6.18377885E-02	7.452251524-03 -1.39452333E-01	6.22569938E-U2 1.87292556E-U1	-1.04/43590E-02 -3.81709402E-02
7 7	36E-02 -9.41951775E-03	2.05755578E-01	-7.75248933E-02	-7.05555556E-01
3 X-TRANS. -6.33711666E-05 -1.037515-01 2.407455375E-01 4 7 7-TRANS. -1.025561445E-01 2.4070576E-06 -2.31455375E-01 4 4 X-TRANS. -7.5520354E-05 -2.31455266E-01 -1 4 X-TRANS. -7.5520354E-05 -2.16103592E-02 2.44376266E-01 -1 5 X-TRANS. -7.55203542102 7.35130074E-06 -2.314376266E-01 -1 5 X-TRANS. -5.051304465E-05 -7.43335556-01 -6.5350045E-01 -5 5 X-TRANS. -5.051304465E-05 -7.43335556-01 -6.5350045E-01 -5 5 X-TRANS. 1.00010006+00 -7.43335556-01 -6.5350045E-01 -5 6 X-TRANS. -1.0001006+01 -3.824340565E-01 -5 -6.5350045E-01 -5 7 X-TRANS. -1.0001006+01 -3.824340565E-01 -5 -6.633960466E-01 -6 7 X-TRANS. -1.0001006+01 -3.824340565E-01 -6 -6.13346656E-01 -6 7 X-TRANS. -1.00010106+10 -3.824341056E-01 -6 -6	38E-82 -1.35767899E+03 59E-01 -1.52475824E-82	-2.14629817E-03 -1.70765720E-01	-1.32527264E-01 3.36842105E-02	-7.69658120E-04 -3.87863248E-02
7 7 7 7 7 7 5 3 36573575-01 5 4 4 5 3 355755755-01 6 6 -2 3 35575556666 1 1 4 5 3 3557555666 1 1 4 5 3 355755666 -2 1 4 5 3 355755666 -2 1	58E-06 -1.A39A1379F-03	2.40745436E-01	2.22901849F-03	-7.9444444E-01
4 X-TRANS. -7.55200354 ± 0.6 -2.16003592 ± 0.6 -3.14376266 ± 0.1 -1 4 Y-TRANS. -9.51345135 ± 0.6 -7.5313074 ± 0.6 6.9350041 ± 0.4 -1 5 X-TRANS. -9.51345135 ± 0.6 -7.5313074 ± 0.6 6.9350041 ± 0.4 -1 5 X-TRANS. -5.02115342 ± 0.6 -7.33953515 ± 0.6 -2.5253549766 ± 0.6 -9.575600406 ± 0.4 -9.575600406 ± 0.4 -9.575600406 ± 0.4 -9.575600406 ± 0.6 -9.5756005 ± 0.6 -9.575600406 ± 0.6 -9.575600406 ± 0.6 -9.575600406 ± 0.6 -9.575600406 ± 0.6 -9.5756004206 ± 0.6 -9.57520526 ± 0.6 -9.526576 ± 0.6 -9.526576 ± 0.6 -9.526572055 ± 0.6 -9.526572055 ± 0.6	+3E-01 2.68708578E-06 55E-05 5.42301287E-01	-2.91455375E-04 4.59305274E-01	6.19250830E-02 4.22664770E-03	-1.67735043E-04 -5.01709402E-01
4 7 -TRANS, $-9.58345135E-02$ $7.33130074E-05$ $6.95360441E-04$ -1 7 7 -TRANS, $-5.6973651392E-05$ $-5.152535497E-01$ -5 5 $7.73130074E-05$ $6.95360041E-04$ -1 7 7 -TRANS, $-5.02315345E-05$ $-5.02315345E-01$ -5 5 7 -TRANS, $-3.61030000E+00$ $-3.824345126E-02$ $-5.15336456E-01$ -9 6 7 -TRANS, $-3.609552392E-02$ $-5.0234565E-01$ -9 6 7 -TRANS, $-3.6009416E-02$ $-2.1000000000000000000000000000000000000$	545-86 -12.468836005.60	2.143762686_01	1 . 086740415-0 1	-7.786394796-01
4 Z-TRANS: -5.9597895289 -7.839535516-02 -5.555354977-01 -7 5 X-TRANS: -5.02115342E-05 -5.03346656E-01 -5 5 X-TRANS: -5.02115342E-05 -5.03346656E-01 -5 5 X-TRANS: -1.0000000E+00 -3.928341035-05 -5.786004066-04 9 5 X-TRANS: -1.06791360E-01 -3.9283325-05 -5.786004066-04 9 6 X-TRANS: -1.06791360E-01 -3.9283325-05 -5.786004066-04 9 6 X-TRANS: -1.06791360E-01 -3.9283325-05 -5.786004965-01 -9 7 X-TRANS: -1.06791360E-01 -3.928426926-03 -6.049949266-01 -3 7 X-TRANS: -3.04989900E-02 1.40059265-03 -5.074393566-01 -3 7 X-TRANS: -3.0449595012 1.4521769146-02 -3.44219526-01 -3 7 X-TRANS: -3.0449595012 1.4521769146-02 -3.4224481056-03 -3.542189506-01 -3 8 X-TRANS: -3.6429690006-012 1.4521769146-02 -3.422447617221092 -7 -9	35E-02 7.3513824F=06	6.95360041E-04	-1.320531065-01	1.115A1197F-04
F X-TRANS. -5.02115342E-05 -6.9255394E-61 -5.73346056E-01 -5 F X-TRANS. 1.0000000E+00 -3.934405E-05 -5.73600406E-04 9 F X-TRANS. -3.61389446E-02 -7.0000000E+00 -3.934405E-01 9 6 X-TRANS. -3.61389446E-02 -7.000000E+03 -3.000000E+03 -5.73600406E-04 9 7 X-TRANS. -1.06791350E-01 -9.16139026E-02 -5.01496915E-01 -9 7 X-TRANS. -3.6439446E-02 -7.40092026E-03 -8.00795137E-03 -6.0795137E-03 -6.0795137E-03 -6.0795137E-03 -6.0795137E-03 -7.722210372E-03 -7.722210956-01 -7 7 X-TRANS. -1.413494592E-01 1.4527176914E-02 2.434406E-02 -7.420305656-01 -7 8 X-TRANS. -1.4147405926-01 1.4527176912 -1.43305656-01 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 <	39E-06 ~7.83963551E-02	-2.52535497E-01	-2.32954007E-03	-1.38162393E-02
5 Y TRANS. 1.0000000E400 -3.9243403E-05 -5.75600406E-04 9 6 X - TRANS. -3.61399446E-02 -7.40036232E-03 -5.75600406E-04 9 6 X - TRANS. -3.61399446E-02 -7.40366232E-03 -5.059535E-01 -9 7 X - TRANS. -1.06791360E-01 -9.165146-02 -7.403666232E-03 -6.0595137E-01 -1.37423935E-01 -1.3742893536E-01 -1.37428939556-01 -1.37428939556-01 -1.37428939556-01 -1.37428939556-01 -1.352631095E-01 -1.352631095E-01 -1.33526910406E-02 -1.33526910406E-02 -1.33526910406E-02 -1.33526910406E-02 -1.33526910406E-02 -1.33526910406E-02 -1.35276-01 -1.33526910406E-02 -1.352766	+2E-05 -6.99552894E-81	-5.78346856E-01	-5.31531532E-03	-4.14487179E-01
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	00E+00 ~3.92434103E-05	-5.78600406E-04	9.77714557E-02	7.73931624E-05
6 X-TRANS. -3.61389446E-02 -7.80358232E-83 2.01466154E-01 -9.165136256-03 -6.067951352-03 -6.067951352-03 -6.067951352-03 -6.067951352-03 -6.067951352-03 -6.067951352-03 -6.067951352-03 -6.067951352-03 -6.057951352-03 -6.05795352-03 -6.05795352-03 -6.05795352-03 -1.072395526-01 -1.072395526-01 -1.0723953956-01 -7.022109525-02 -7 7 X-TRANS. -9.4540959526-02 1.415705146-02 3.2314510505-02 -7 -9 -9 -9 -9 -9 -9 -7 -9 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 -7 <t< td=""><td><u> 365-05 3.00000000000000000000000000000000000</u></td><td>1.00000000E+00</td><td>9.20815552E-03</td><td>-7.60683761E-01</td></t<>	<u> 365-05 3.00000000000000000000000000000000000</u>	1.00000000E+00	9.20815552E-03	-7.60683761E-01
6 V-TRANS. -1.06791360E-01 -9.16219826E-03 -8.06795132E-03 -6.06795132E-03 -6.06795132E-03 -1.37423935E-01 -1.374239355-01 -1.374239355-01 -1.374239355-01 -1.374239355-01 -1.374239355-01 -1.374239355-01 -1.3742393556-01 -1.3742393556-01 -1.3742393556-01 -1.3742393556-01 -1.3742393556-01 -1.374239556-01 -1.374239556-01 -1.374239556-01 -1.374239556-01 -1.374239556-01 -1.374239556-01 -1.374239556-01 -1.374239556-01 -1.37423956-01 -1.35263766-01 -9.4747975165-01 -1.1339039556-02 -1.1339039556-02 -1.1339039556-02 -1.1339039556-02 -1.1339039556-02 -1.135223276-01 -1.133903956-02 -1.1339032956-02 -1.135223276-01 -1.13490156-02 -1.1339032956-02 -1.1339032956-02 -1.1390329456-01 -1.22210956-01 -1.22210956-01 -1.22210956-01 -1.22210956-	+6E-02 -7.80958232E-83	2.04868154E-01	8.07491702E-02	-6.93589744E-01
7 X-TRANS. -3.84324204E-0? -9.4939900EE02 1.372355E-03 2.0499429E-01 -3.5208925E-03 -1. 7 Y-TRANS. -9.4999900EE02 1.3723557670E-03 3.552089526E-01 -3. 7 Z-TRANS. -9.4999900EE02 1.37235576E-03 3.552089526E-01 -3. 8 X-TRANS. -8.15408595E-05 2.0451364E-02 2.045146106E-02 -7 -9. 8 Y-TRANS. -6.64217354E-02 2.04517364E-02 3.29107505E-02 -7 -9 8 Y-TRANS. -1.447405926-01 -1.490597944E-03 -1.352697636E-01 -9 8 Y-TRANS. -1.447405926-01 -1.490597944E-03 -1.352697636E-02 -7 9 X-TRANS. -1.447405926-01 -1.33903956-02 -6.326572016-02 -7 9 X-TRANS. -9.472277896-02 2.04316026E-02 3.2207413796E-02 -7 9 X-TRANS. -9.472277896E-02 2.043164026E-02 3.2207413796E-02 -7 9 X-TRANS. -1.630132736E-02 3.2307413796E-02 -7 -9 -7 10 X-TRANS.	o Ut−U1 -9.16219826E-U3 L4E-U2 6.18606\$20E-02	-8.06/95132E-03 -1.37423935E-01	6.21147463E-02 -1.89853011E-01	1.01367521E-02 -3.78846154E-02
7 Y-ITANS. -9.49699000E-02 1.37737670E-03 3.542069255-03 -1 7 Z-IRANS. -9.49699000E-02 1.37737670E-03 3.542069255-03 -1 8 X-IRANS. -8.154085955-05 2.42175916 3.542069256-03 -3 -3 8 Y-IRANS. -8.154085955-05 2.48146105 3.291475055-02 -7 8 Y-IRANS. -6.642173245-02 2.4821437465-02 3.291457656-02 -7 9 X-IRANS. -1.447405926-01 1.4896393956-02 3.237443766-02 -7 9 X-IRANS. -9.240926305-05 2.483443746-03 -1.43526876366-02 -7 9 Y-IRANS. -9.2409265306-02 2.48443746-03 -1.43526876366-02 -1 9 Y-IRANS. -9.465036526-01 1.4389033565-02 -1.432573266-02 -1 -1 10 X-IRANS. -9.4650365226-01 5.484437466-02 -1.435233265-02 -1 -1 -1 -1 -1<0104066-02	04F-02 -9.422176365-03	2 . A4 994 97 9F - A1	A.131A1603E-02	-7.059829065-01
7 Z-TRANS. -1.013404595E-01 -1.52776944-02 -1.670365396E-01 -3 8 X-TRANS. -6.64217324E-02 2.075176944-03 -1.670365396E-01 -3 8 Y-TRANS. -6.64217324E-02 2.075176946-03 -7.122210955-01 -9 8 Y-TRANS. -6.64217324E-02 2.075176403 -1.352697654E-01 -9 9 X-TRANS. -9.24092630E-05 2.075176403 -1.352697654E-01 -9 9 X-TRANS. -9.24092630E-05 2.0133993955-02 -6.32657201E-02 -7 9 X-TRANS. -9.472277895-02 1.133993955-02 -6.32657201E-02 -7 9 X-TRANS. -9.472277895-02 1.133993955-02 -6.32657201E-02 -7 9 X-TRANS. -1.6301572671E-01 1.133993955-02 -6.32657201E-02 -7 10 X-TRANS. -1.01024271E-01 2.26428055E-02 -1.03023455E-01 -1 10 X-TRANS. -1.01024271E-01 -1.2230032455E-01 -1 -1 -7 10 X-TRANS. -1.01024271E-01 -1.2230136466-02 -1.030324556-01 <	00E-02 1.37237678F+#3	3.542089255-03	-1.324798485-01	9-94017094E-04
R X-TRANS. -6.15408595E-05 2.02140105E-02 3.23107505E-02 -7 B Y-TRANS. -6.642173245-02 2.07517316E-03 -7.12221095E-02 -7 B Z-TRANS. -6.642173245-02 2.07517316E-03 -7.12221095E-02 -7 B Z-TRANS. -9.2409265-01 -1.14740592E-01 -1.13399395E-02 -7 -9 9 X-TRANS. -9.24092630E-05 2.8144370E-02 -3.2297443795-02 -7 -9 9 X-TRANS. -9.47227789E-02 1.133993955E-02 -0.33657701E-02 -7 9 X-TRANS. -9.47227789E-02 1.13399395E-02 -6.336657201E-02 -7 9 X-TRANS. -1.63016522E-01 -6.10411602E-83 -1.019232321E-01 -1 10 X-TRANS. -1.01024271E-01 2.05420053E-02 -7.06793035E-02 -7.05373321E-01 -1 10 X-TRANS. -1.01024271E-01 2.05420053E-02 -1.03033245E-01 -1 -1 -1 10 X-TRANS. -1.01024271E-01 2.05420	59E-011.52176914E-82	-1.70385396E-01	-3.68326221E-02	-3.89529915E-02
B V T TRANS. -6.64217324E-02 2.07517310E-03 -7.12221095E-02 -7 B Z - TRANS. -4.44740592E-01 -1.403059794E-03 -7.12221095E-02 -7 9 X+ TRANS. -9.4727789E-02 1.41399395E-02 -3.29744379E-02 -7 9 Y T RANS. -9.47227789E-02 1.41399395E-02 -6.32657201E-02 -7 9 Y T RANS. -9.47227789E-02 1.41399395E-02 -6.32657201E-02 -7 9 Y T RANS. -1.653016522E-01 -6.110411602E-02 -6.32667201E-02 -7 10 X T RANS. -1.01024271E-01 2.05420055E-02 -7.06793075E-02 -7 10 X T RANS. -1.01024271E-01 2.054206055E-02 -1.03303245E-02 -1 10 X T RANS. -1.738055436E-02 -1.255601826E-02 -1.033032455E-01 -1 11 X T RANS. -1.73805552E-02 -6.035650102E-02 -1.033032455E-01 -1 11 X T RANS. -1.738056522E-01 -2.5626567236E-02 -1.554792009E-02 -1 -1 <td>35E-05 2.02148106E-02</td> <td>3.29107505E-02</td> <td>-7.42057847E-02</td> <td>-5.01282051E-01</td>	35E-05 2.02148106E-02	3.29107505E-02	-7.42057847E-02	-5.01282051E-01
8 Z - TRANS. -1.14740592E-01 -1.40059794E-03 -1.35266763E-01 -9 9 X - TRANS. -9.47227789E-05 1.13399395E-02 -3.29744379E-02 -7 9 X - TRANS. -9.47227789E-02 1.13399395E-02 -6.32657201E-02 -7 9 X - TRANS. -9.47227789E-02 1.13399395E-02 -6.32657201E-02 -7 9 X - TRANS. -1.653016522E-01 -6.10411602E-02 -7 -0.32657201E-02 -7 10 X - TRANS. -1.01024271E-01 2.26420165E-02 -1.01923321E-01 -1 10 X - TRANS. -1.01024271E-01 2.26420165E-02 -1.0303245E-02 -1.0303245E-01 -1 10 X - TRANS. -1.01024271E-01 2.26420165E-02 -1.0303245E-01 -1 10 X - TRANS. -1.730365436E-02 -1.23014826E-02 -1.03679105E-02 -1 11 X - TRANS. -1.73036552E-02 -6.31467512E-01 -1 -1 11 Y - TRANS. -1.3636552E-02 -1.2562457236E-02 -1.05656127E-02 -1 11 Y - TRANS. -1.1.73836562E-02 <td< td=""><td>24E-02 2.87517316E-03</td><td>-7.12221095E-02</td><td>-7.08866761E-02</td><td>-1.86282051E-01</td></td<>	24E-02 2.87517316E-03	-7.12221095E-02	-7.08866761E-02	-1.86282051E-01
9 X-TRANS. -9.24032530E-05 2.83143776-02 3.29744379E-02 -7 9 Y-TRANS. -9.47227789E-02 1.13390395E-02 -6.33657201E-02 -3 9 X-TRANS. -9.47227789E-02 1.13390395E-02 -6.33657201E-02 -3 9 Z-TRANS. -1.653016522E-01 -6.101411602E-02 -6.33657201E-02 -7 10 X-TRANS. -1.01024271E-01 2.05420063E-02 -7.06793105E-02 -7 10 X-TRANS. -1.01024271E-01 2.05420063E-02 -1.0303245E-01 -1 10 X-TRANS. -1.749721657E-01 -2.056420053E-02 -1.0303245E-01 -1 10 X-TRANS. -1.730365436E-02 -4.556426528-02 -1.056456-01 -1 11 X-TRANS. -1.730365528-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.314655178-02 -6.3146555178-02 -6.3146555178-02 -6.3146555178-02 -6.3146555178-02 -6.3146555178-0	326-01 -1-00597946-03	-1.35268763E-01	-9.65860597E-02	1,03632479E-01
9 V - TRANS. -9.472277895-02 1.133983955-02 -8.326572015-02 -3 9 Z - TRANS. -1.630165225-01 -6.1014116025-93 -1.195233275-01 -1 10 X - TRANS. -1.630165225-01 -6.1014116025-93 -1.195233275-01 -1 10 X - TRANS. -1.010242715-01 2.054200555-92 -1.0507321656-02 -1 10 Y - TRANS. -1.010242715 -1.25540182655-82 -1.030324556-02 -1 10 X - TRANS. -1.7303543656-01 -1.25540182655-82 -1.030324556-01 -1 11 X - TRANS. -1.7303554365-02 -6.314655127-01 -1 -1 11 Y - TRANS. -1.333355-02 -5.6545572355-02 -6.314655175-01 -1 11 Z - TRANS. -1.3303565255-02 -5.65656572355-02 -6.314655175-02 -6.314655175-02 -6.314655175-02 -6.314655175-02 -6.314655175-02 -6.314655175-02 -6.314655175-02 -6.314655175-02 -6.314655175-02 -6.314655175-02 -6.3146555175-02 -6.3146555175-02 -6.3146555175-02 <td< td=""><td>30E-05. 2.83143770E-02</td><td>3.29741379E-02</td><td>-7.43480322E-02</td><td>-5.01202051E-01</td></td<>	30E-05. 2.83143770E-02	3.29741379E-02	-7.43480322E-02	-5.01202051E-01
10 X-TRANS, -9,10303273E-05 1.99105162E-02 3.20600406E-02 -7 10 Y-TRANS, -1.01024271E-01 2.26503655E-02 -1.0303245E-01 -1 10 Y-TRANS, -1.74972167E-01 2.2650318265E-02 -1.0303245E-01 -1 10 Z-TRANS, -1.74972167E-01 -1.255011826E-02 -1.0303245E-01 -1 11 X-TRANS, 3.59385426E-02 -8.558342738E-03 -1.654792089E-01 -1 11 Y-TRANS, -1.73838552E-02 -5.658651002E-02 -6.33465512F-01 -3 11 Z-TRANS, -1.3838552E-02 -5.658651002E-02 -6.33465517F-02 -6	39E+02 1.13393395E-02 22E+01	-8.32657201E-02 -1.19523327E-01	-3.29397819E-02 -1.11948791E-01	1.74017094E-01 -1.01196581E-01
10 X-TRANS, -1,1630(J2(3E-05) <u>1,99105162E-02</u> -7,60793103E-02 -1 10 Y-TRANS, -1,01024271E-01 2,26428063E-02 -7,60793103E-02 -1 10 Z-TRANS, -1,74972167E-01 _1.255011826E-02 -7,6079303245E-01 -1 11 X-TRANS, 3,59385438E-02 -8,55834238E-03 -1,5547927E-01 -3 11 Y-TRANS, -1,73836552E-02 -5,655651002E-02 -6,31465517E-02 -6 11 Z-TRANS, -1,340013365-01 -2,262645723E-02 -6,31465517E-02 -6				
10 T-TRANS, -1.010C4671E-01 Z.26928063E-02 -1.03003245E-01 -1. 10 Z-TRANS, -1.74972167E-01 _1.25801826E-02 -1.03003245E-01 -1. 11 X-TRANS, 3.59385438E-02 -8.583561002E-03 -1.654792089E-01 -3. 11 Y-TRANS, -1.73838552E-02 -5.88561002E-02 -1.65517E-02 -6. 11 Z-TRANS, -1.340013365-01 _2.262645723E-02 _6.3465517E-02 -6.	3E-05 1.98105162E-82	3.28600406E-02	-7.40635372E-02	-5.20085470E-01
<u>11 X-TRANS。 3.59385438E-D2 -A.55834238F-B3 -1.554792089E-D1 -1</u> 11 <u>Y-TRANS1.73836552E-D2 -5.83561002E-92 -1.85851927E-D1 -3</u> 11 Z-TRANS1.34 <u>001336E-D1 -2.26245723E-D2 -6.31465517E-D2 -6</u>	1E-U1 2.26428063E-02 37E-Q1 -1.25801826E-02	-1.68/93103E-02 -1.03803245E-01	-1.9464201UE+U2 -1.05832148E-01	4.39743594E-UI -2.53584274E-U1
11 Y-TRANS, -1.738365525-02 -5.835610026-82 -1.858519 276-01 -3 11 Z-TRANS, -1.8340013365-01 -2.262457236-02 -6.314655175-02 -6	38E-02 -8.55834238E~83.	-1.54792089E-01	-1.93409199E-01	-7°04700855E-01
	525-02 -5.835610025-82 365-01 -2.262457235-92	-1.85851927E-01 -6.31465517E-02	-3.79943101E-02 -6.82313893E-02	2.47094017E-02 1.02136752E-02
12 X-TRANS. 3.820975286-02 -1.098709495-02 -1.551724146-01 -1	285-82 -1.098709695-82	-1°55172414E-01	-1.93172119E-01	-7,16239316E-01

	1465-02 715-02	148F-01 36E-01 44E+01	84E-01 52E-02 03E-03	441E-01 221E-01 11E-01	946-01 086-02 716-02	10.65 + 01 7 + E - 02 10 4 E - 02	151E-01 159E-05 44E-01	51E-01 22E-05 17E-01	206-01 1226-05 1086-01	194E-01 27E-02 96E-02	066-01 29 <u>6-02</u> 34e-02	986-01 366-01 796-01	81E-01 112E-02 163E-03	
15	3.961538 1.609401	-7.979632 4.358974 2.594444	-7.811965 1.321367 1.984188	-4.060256 6.713675 3.861111	-6.940170 -2.526923 3.409401	-7.059829 3.293589 1.442735	-5.012820 1.939743 -2.14444	-5.012820 1.672222 2.020940	-5.196581 1.419658 5.076923	-6.940170 -2.523504 3.407264	-7.059829 -3.290598 1.441830	-7.982905 -4.353974 2.594871	-7.811965 -1.319658 1.982478	
14	-1.88335704E-01 -3.10621147E-02	-2.71977240E-01 4.66714093E-01 2.33475581E-01	-2.419630166-01 -2.752963496-01 -9.511142726-02	5.52441916E-01 1.0000000E+00 5.31531532E-01	-2.72024656E-01 -2.01327643E-01 -1.42437174E-01	-2.72403983E-01 -2.06495970E-01 -4.86960645E-02	-6.07396871E-04 -2.28022760E-01 -1.08345187E-04	-6.08345187E-04 -2.15079236E-01 2.25462304E-04	-6.05974395E-04 -1.90753912E-01 2.95448080E-04	2.71123755E-01 -2.01801802E-01 1.43669986E-01	2.71503082E-01 -2.06306306E-01 4.66237340E-02	2.697486965-01 4.634423906-01 -2.306780462-01	2.39971550E-01 -2.74110953E-01 8.25509720E-02	
13	-4.623478705-02 -1.289032455-01	-1.17761156E-01 1.76343213E-01 1.50228195E-01	-1.046272822-01 -6.209164302-02 -1.390720086-01	2.87145030E-01 3.97433148E-01 3.021044625-01	-4.65770731£-02 2.57606491£-02 6.78625761£-02	-4.61713996E-02 -9.22163243E-03 -1.10509F355-01	-6.449036512-02 1.24987223E-03 -1.121450302-02	-6.46298174E-02 <u>1.181414915-03</u> 2.392241385-02	-6.43128803E-02 1.048935092E-03 3.099645033-02	-4.95436105E-02 -2.35420892E-02 6.62905E80E-02	-4.915060855-02 1.149097223-02 -1.11029412E-01	-1.20727688E-01 -1.814142075-01 1.527636925-01	-1.072641995-01 6.509884445-02 -1.399594325-01	
12	1.46042893E-02 7.07418957E-03	-3.23411878E-03 -4.67521348E-01 -2.69383109E-01	-2,28994163E-02 6,09398153E-02 3,53851146E-02	-6.98073290E-01 -8.59692968E-01 -5.01926913E-01	-7.89860927E-03 -4.86491779E-02 -3.79502994E-02	-1.03025551E-02 1.28702593E-02 9.17938117E-03	2.01194091E-02 -1.27625974E-05 2.19526879E-03	2.02243393E-02 -2.02600972E-05 1.28509142E-02	1.89144732E-02 -2.18969178E-05 2.58707462E-02	-7.90704244E-03 4.86257905E-02 -3.79634224E-02	-1.03113189E-02 -1.29148134E-02 9.17992402E-03	-3.23172245E-03 4.67508497E-01 -2.69400992E-01	-2.28970946E-02 -6.09575510E-02 3.53700628E-02	
. 11	-1.07370296E-01 -1.74192A305-01	1.14940993E-03 4 <u>.</u> 14072592E- <u>0</u> 2 -9.133823205- <u>0</u> 2	1.041416175-03 4.76957998E-92 -9.468047215-92	-2.11957248E-03 -5.0501020E-01 9.61055444E-01	-3.400133602-02 1.2313595-01 -5.299447966-92	627254516-02 2.025606776-31 4.771765755-03	4.284123805-96 1.314629265-01 1.342240045-06	4.30416399E-06 1.97775515-01 -5.085504345-07	4.275217105-06 2.002994685-01 -1.005566595-06	3.40236028E-02 1.23135159E-01 5.29949796E-02	3.627254515-02 2.025506775-01 -4.76953908E-03	-1.14117123E-03 4.81957594E-02 9.12936995E-02	-1.037195485-03 4.769539085-02 9.470273985-02	
EL CMENT Reference	Y-TZANS. Z-TRANS.	X-TRANS. Y-TPANS. Z-TRANS.	X- TRANS. Y- TRANS. Z- TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X- TRANS. Y- TPANS. 7- TPANS.	X-TRANS. Y-TRANS. Z-TRANS.	X- TRANS. Y- TRANS. Z- TRANS.	X-TRANS. Y-TPANS. Z-TRANS.	X- TRANS. Y- TRANS. Z- TRANS.	X-TRANS. Y-TPANS. Z-TRANS.	X-TRANS. Y-TPANS. Z-TRANS.	X- TRANS. Y- TRANS. Z- TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	
VOPAL POINT Numper	12 12	5 F F F	4 4 4 4 4 4 4 4	4 4 4 7 7 7 7 7	4 4 4 0 4 4 0 4 4	17 17 17	4 4 4 6	4 4 7 0-0-0	20 20 20 20	27 17 17 17	22 25 25	8 8 8 8 8 8 8 8 8 8	25 25 25 25 25 25 25 25 25 25 25 25 25 2	

~

•

			12	15	14	15
55	Y-TRANS.	-5.05010020E-01	8.598892225~01	-4°08341785E-01	3.92413466E-01	-6.71794872E-01
25	Z-TRANS.	-8.51055444E-01	-5.01726703E-01	3.07809331E-01	-5.25841631E-01	3.86196581E-01
26	X-TRANS.	-3.59385438E-02	-8.545526736-83	-1.568204876-01	1.90516833E-01	-7.04790855E-01
26	Y-TRANS.	-1.73792029E-02	5. 83538940F-82	1.86232252E-01	-3.45566619E-02	-2.46495726E-02
26	Z-TRANS.	1.34001336E-01	~2.26494866E-82	-6.38691684E-02	6.70934092E-02	1.01752137E-02
27	X-TRANS.	-3.82097528E-02	-1.09738861E-02	-1.57200811E-01	1.902797536-01	-7.16666667E-01
27	Y-TRANS.	-1.073702965-01	-1.45940848E-02	4.83265720E-02	-1.87482219E-01	-3.96410256E-02
27	Z-TRANS.	1.7419283DE-01	7.042985105-03	-1.29056795E-01	2.86818397E-02	1.61153846E-02
28	X-TRANS.	7.98485861E-05	2.02226978E-02	3.20486815E-02	7.48221906E-02	-5.01282051E~01
28	Y-TRANS.	-6.64217324E-02	-2.855281546-83	7.19827586E-02	-6.95590327E-02	1.86282051E-01
28	Z-TRANS.	1.14740592E-01	-1.11890559E~03	-1.36282961E-01	9.40730204E-02	1.03547009E-01
53	X-TRANS.	8.06947228E-05	2.03222706E-02	3.21120690E-02	7.49644381E-02	-5.01262051E-01
5	Y-TRANS.	-9.472277895-02	-1.13217027E-02	8.36080122E-02	-3.140350885-02	-1.73374359E-01
62	Z-TRANS.	1.63916522E-01	-6.12949240E-83	-1.20740365E-01	1.09720247E-01	-1.01153846E-01
30	X-TRANS.	8.00712536E-05	1.90183871E-02	3.19979716E-02	7.46799431E-02	-5.20085470E-01
10	Y-TRANS.	-1.01024271E-01	-2.26237657E-02	7.90821501E-02	-1.80132764E-02	-4.39743598E-01
30	Z-TRANS.	1.74972167E-01	-1.25266307E-82	-1.04969574E-01	1.03888099E-01	-2.53461538E-01
31	X-TRANS.	2.11578713E-06	1.86762597E-01	-1.25849391E-03	3.64248459E-06	9.99572650E-01
31	Y-TRANS.	-2.64374393E-04	-8.236050535-87	6.21450304E-05	-1.01944049E-02	-1.99102564E-05
31	Z-TRANS.	7.78445780E-07	-1.06872538E-04	-7.53296146E-03	-7.20246562E-05	-6。19230769E-94
32	X-TPANS.	2.1111111E-06	1.06599437E-81	-1.25722617E-03	3.63300142E-06	9.97863248E-01
32	V-TRANS.	-2.23335560E-04	1.468933525~26	-2,02205882E-04	3 . 55286866E-02	2。09%&4444E-05
32	Z-TRANS.	-9.13159653E-07	-5.893921045~05	2.68509128E-02	2.50497866E-04	-2.32820513E-04
33	X-TRÂNS.	2.11511913E-06	1.06774819E-01	-1.25912779E-03	3.64201043E-06	1.0000000E+00
33	V-TRANS.	-2.464929965-04	1-59698708E-07	-5.15973631E-05	9.47368421E-03	-2.33376068E-06
33	Z-TRANS.	5.06568693E-08	-8.62542252E-05	7.25912779E-03	6.67140825E-05	-4.62393162E+04

	GENE	RALIZED INERTIA PRO	OPERTIES			•
VARIARLES	16	17	13	19	20	
XNM	4.496378475404	4.415632555+04	1.87547312E+04	6.83417642E+04	1.44983817E+04	
ANM	4 . 66476114£+94	1.70675464E+94	2 .1714 3558E+D4	4.25923677E+04	1.13153543E+04	
ZNM	1.819748945+04	4.61491339E+04	?.11 374296E+04	4.465967515+04	1.89149873E+04	
Xd	4.237511465+93	6.96745451E+00	4.01127R06E+05	2.68242553E+02	-4.51132124E+01	
۶	1.344997555+05	1.45051495E+02	1.47309395496+06	5.01191735E+02	3+30674779E+02	
Ζd	-1.35196947E+05	-4.29227309E+02	1.484645385+06	4.73750979E+02		
GM DIAGONAL	5.48594422E+04	5.36365053E+04	3.082846785+04	7.77964035E+04	2.23643617E+04	
		:				
	-					1 -
						•
						- 1 mm
						1260
						,
		-				

•

				annos - v											1
ť		and and the second						Ĩ							
20	2.81151036E+03		20	-4.20851831E-03 8.33442845E-02 -1.97707338E-01	-6.00741363£-03 -1.26498103E-01 2.57593388E-01	-1.29207063E-03 9.24469595E-02 -3.79430692E-03	-1.59667883E-03 -1.26909561E-01 8.96870468E-04	-5.75094300E-03 -3.29607951E-02 -5.75172124E-04	4.75880806E-03 9.36901363E-02 1.82706920E-01	6,276250425-03 -1,266667015-01 -2,53123921E-01	1.13397953E-01 1.64550163E-01 1.45054896E-01	1。13420480E-01 -1。53409617E-01 -8。62286243E-02	1.07443263E-01 -2.17452401E-01 -1.21782372E-01	5.61630399E-02 3.17480244E-01 1.29362204E-01	3.29874573E=02
19	2+69850389E+03		19	3.96046852E-01 9.49973598E-01 -3.350175705-01	4.058077115-01 -7.289921435-01 2.581747195-01	-2.11298194E-04 9.41922839E-01 3.34797462E-06	-2.10073087E-04 -7.16203026E-01 2.77203394E-04	1.04294778E-04 -1.40287945E-01 -6.11517914E-05	-3.96046852E-01 3.49975598E-01 3.336017570E-01	-4.06051733E-01 -7.28892143E-01 -2.57686676E-01	-2.1639824E-03 3.63836018E-01 6.32991703E-01	-2.096144465-03 -1.45534407E-01 -2.56222549E-01	-1.972181552-03 -2.51830161E-01 -4.44363104E-01	3.91998497E-01 -2.0275744325-01 1.00000005400	4.02635432E-01
18	2.67732900E+03	COMPLETE SYSTEM	18	-1.37659033E-01 3.17981240E-02 1.65055131E-01	-1.31906£162-01 3.10347752E-02 3.59796438E-01	-9.059524172-02 -7.717557252-05 8.744637902-01	-3.43002545E-02 6.61238238E-05 8.49872774E-01	<u>9.525021206-0</u> 1. 1.4003727 <u>5-05</u> 2.088210355-01	-1.37574215E-01 -3.195090582-02 1.64970314E-01	-1.31721798E-01 -3.08990670E+02 3.598812555-01	-1.879528952-01. -1.697201025-02 -2.432569975-02	-1.885436185-01 6.211135935-02 -3.703996435-02	-1.890535242-01 1.512238565-01 -8.292620376-02	-1.322307046-01 -2.150127236-01 -9.338422396-02	-1.244274915-01
17	2.49077254E+03	NG MODE SHAPES FOR	17	3.64965145F-01 1.0000000E+00 3.50798291E-01	3.69 <u>2376395-01</u> -6.70 <u>339555E-01</u> -3.78007646E-01	-3.35997302E-04 9.916797445-01 -5.0505915-04	-3.732957515-04 -6.55273219E-01 -6.12772656E-04	-3. <u>62716438E-04</u> -6.79109512E-02 -6.35484536E+04	-3.65414836F-01 1.0000000E+00 -3.49449057 <u>E</u> +01	-3.69912300E-01 -6.70339535E-01 3.77194163E-01	-5.84663818E-01 -3.59680694E-02 2.83112210E-01	-5.87912008E-01 2.100292335-01 -1.42503562E-01	-5.7420233E-01 2.45109952E-01 -2.23903755E-01	-5.92334034E-02 6.999033056-01 -1.594333266-01	-5.69822332F-02
16	2.48526099E+03	CORRESPONDI	16	-1.34243176°-01 -4.940445555-03 3.825475505-01	-1.394676592-01 -5.519437555-03 -2.124205135-01	-1.64267990E-01 -1.929611255-33 -1.740291225-01	-1.455748555-01 1.220016545-03 -2.840363945-01	-5.69561621E-02 1.?69544372-04 -3.01157992E-01	+1.729197695-01 1.249733225-03 3.433709585-01	-1.78544251°-01 9.01406121°-03 -2.141439215-01	_ 3.405293535-01 2.757650955-01 2.2339040535-01	3.423490495-31 -1.382134005-31 -5.364264225-02	3.329950375-J1 -2.164598945-91 -5.332506205-94	3"9.00661709-01 -5.406120761-01 5.893767037-01	3,963313485-01
			EL EMENT REFERENCE	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-FRANS. Y-TRANS. Z-TRANS.	X+TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X- TRANS. Y- TRANS. Z- TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X+TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TPANS.	X-TRANS.
			NCDAL POINT NUMEEP	ल स न	~~~~	(M (M (M	444	ພິມ	ם ם ש	~~~	ac ac ac	םי סי סי	10 110	11 11 11	12

NATURAL FREQUENCIES (RADIANS/SEC)

		CORRESPOND	ING MODE SHAPES FOR	COMPLETE SYSTEM										
AL POINT UMBER	EL EMENT REFERENCE	16	17	8	19	20								
12 12	Y-TRANS. Z-TRANS.	5.00413565E-01 -3.87675765E-01	-5.04384979E-01 1.00382290E-01	-3.08142494E-01 -1.45122986E-01	1.76891166E-01 -7.73060029E-01	-2.666399345-01 -8.446189895-02								
13	X-TRANS.	9.52026468E-02	1.35956825E-01	-6.39525021E-02	1.75988287E-02	-2.30168236E-01								
13 13	Y-TRANS. Z-TRANS.	-5.14971050E-01 6.93879239E-01	1.18641791E-01 -4.87069935E-01	-7.64631043E-01 -4.49109415E-01	-5.00244021E-01 8.17471938E-01	5.75548488E-01 3.12674496E-01								
14	X-TRAWS.	8.57733664E-02	1.18956600E-01	-5.32739610E-03	1.96046852E-D2	-2.81114746E-01								
4	Z-TRANS.	1.65922250E-01 -5.68155500E-01	-3.757589396-01 1.58646278E-01	-6.85496183E-01 -3.96183206E-01	3.90678380E-01 -6.31771596E-01	-1.60159369E-01 -3.95028898E-02								
45	V-TDANC	4.76923077F=02	3.49478A755-09	4 . 0000000F400	4.70473402F-02	-9.946350775-04								
15	Y-TRANS.	-1.07692308E-01	-2,36567612E-01	-1.74809160E-01	7.18643241E-02	6.20680651E-02								
15	Z-TRANS.	-1.29693962E-01	-1.01506634E-01	-9.78795589E-02	-1.26256711E-01	6。34445246E-02								
16	X-TRANS.	-2.31349222E-01	2.90532944E-01	-1.32145886E-01	-3.90678380E-01	2.48743661E-02								
16 16	Z-TRANS.	4.502057825-03 1.00000000E+00	3.81380706E-01 -3.42478075E-01	-1.54792197E-01 -1.35368957E-01	-8.1625153UE-U1 6.42996584E-01	4.39813983E-01 2.67413432E-01								
17	X-TRANS.	-2.314309355-01 -0 ccne76765-02	2.97054194E-01 -+ 60732404E-01	-1.24427481E+01 -2 786260666-01	-3.98487067E-01 - 705514805-01	-1.84897134E-03 -1. 648694686								
17	Z-TRANS.	-7.39867659E-01	2.79739150E-01	-1.94147583E-01	-5.08540752E-01	-2.32528226E-81								
18	X-TRANS.	-6.808105875-01	-1.17247543F-03	-1.922815956-01	-1.33504148E-04	-5.82122825F-B4								
18	Y-TRANS.	-6.69148056E-04	3.83654284E-01	1.37404580E-04	-7.80624695E-01	3.87141387E-01								
18	Z-TRANS.	-2.54673284E-01	-4.41421193E-04	-5.91772689E-03	-2.01854563E-05	-3.91578220E-04								
19	X-TRANS.	-6.85277089E-01	-1.18034630E-03	-1.93129771E-01	-1.34187408E-04	-5.83825987E-04								
19	V-TRANS.	7.8378825555-05	-3.914998885-02 c 476697606-04	-4.145038175-05 7 766764485-03	3.24792582E-01 E 405757505-05	<u>-2.938914895-01</u> 5 013731035-01								
13	- CMAA1-7	Th-Benthousned	+n-3n6696011.6	1.0100101405	CB-366000664.C	7. U&C3C40CC=U\$								
20	X-TRAMS.	-6.67080232E-01	-1.14908927E-03	-1.92875318E-01	-1.32918497E-04	~5°48897888E-04								
20	Z-TRANS.		-1.402/4342E-U1 6.59545761E-04	-/.952155319E-U2 1.78456319E-D1	1.10566130E-04	7.03801662E-04								
12	X-TRANS.	-2.30355666E-U1 -6 976/04465-03	-2.914324265-01 7 817807055-01	-1.52319922E-Ul 4 EEAA46EAE-A1	3.905/8380E-01 -8.46264870E-04	-2.511917016-02 / T03400006-04								
21	Z-TRANS.	9,99172870E-01	3.46076006E-01	-1.351145046-01	-6.42996584E-01	-2.68534213E-81								
22	X-TRANS.	-2.30355666E-01	-2.97728806E-01	-1.24597116E-01	3.96243045E-01	9.25852346E-04								
22	7-TRANS.	-7.388751035-02	-1.694400/2E-01 -2.824375985-01	-1.9430319E-U1	5.08796730F-01	-4. D3/35402E-01								
;														
23	X-TRANS.	9.56989247E-02 E 414747775564	-1.35597032E-01	-6.39694656E-02 7 64.8046705-01	-1.76134700E-02 -/.98611967E-01	2.31516706E-01 c 7e3e6045.01								
53	Z-TRANS.	6.92307692E-01	4.89543512E-01	-4.48854962E-01	-8.17715959E-01	-3.14947678E-01								
24	X-TRANS.	8.61869313E-02	-1.18664268E-01	-5.34690416E-03	-1.95510005E-02	2.62733806E-01								
24	Y-IRANS.	-1.645988425-01	-3.76433551E-01	6.85326548E-01	3.91166423E-01	-1.60644901E-01								
24	Z-TRANS.	-5.67659222E-01	-1.60535192E-01	-3.96352841E-01	6.31527574E-01	4°09593996E-02								
1														†*
--------------------	-----------------------	------------------------------------	--	--	--	---	---	--	--	---	--	--	---	----
•								-						
f	20	\$.26397200E-02 -6.34806169E-02	-5.62708952E-02 3.20687781E-01 -1.31776111F-01	-3.29417725E-02 -2.70017200E-01 8.74784442E-02	-1.12707562E-01 1.65939992E-01 1.48249785E-01	-1.12739901E-01 -1.54212732E-01 9.86485577E-02	+1.06795661E-01 -2.18565039E-01 1.25215297E-01	-6.58652820E-06 1.19738962E-01 -5.76414898E-04	-6.60557687E-06 -1.19215442E-01 5.69651449E-04	-6.630862195~06 1.784094655-02 -8.77356378E-05			-	
	19	7.20351391E-02 1.26159102E-01	-3.92142509E-01 -2.02513421E-01 -1.0000000E+00	-4.02879453E-01 1.75915569E-01 7.73060029E-01	1.944607135-03 3.639360185-01 -6.329917035-01	1.92020498E-03 - <u>1.45532016E-01</u> 2.55466569E-01	1.79331381E-03 -2.51930161E-01 4.44363104E-01	1.89507077E-05 -2.23230844E-03 -2.36603221E-05	1.90900390E-05 1.89482674E-03 2.06295754E-03	1.91020010E-05 -4.63884822E-04 -4.74621767E-06				
COMPLETE SYSTEM	18	1.74724343E-01 -9.787955895-02	-1.321458862-01 2.151823586-01 -9.304495346-02	-1.24342663E-01 3.0797275856E-01 -1.453774395	-1.87971077E-01 1.68532E55E-02 -2.42409E21E-02	-1.884648012-01 -6.201865.995-02 -3.70653096E-02	-1.888888895-01 -1.51060221E-01 -8.29685175E-02	3.499575915-02 -4.89652248E-05 -3.15097540E-03	3,54199473E-02 5,0 <u>3138253E-05</u> 2,52926209E-03	3,53519932E-02 -6,63189143E-06 -7,40033927E-04				
NG MODE SHAPES FOR	17	-2.33139071E-01 1.01056492E-01	6.06701147E-02 5.91702271E-01 1.619297315-01		5.35788172F-01 -9.69574934F-02 -2.82212728E-01	5.88936352F-01 2.10501462E-01 1.42702946E-01	5.74544637E-01 2.46008545E-01 2.23891258E-01	1.55947930E-06 -3.54396222E-01 -6.15695975E-04	1.50326053E-06 3.62041826E-01 6.23791320E-04	1.54171351F-06 -4.84821228E-02 -8.64178038E-02				
CORRESPONDI	16	1.085194385-01 -1.300248145-01	3.79900744F-01 5.38130697E-01 6.99330025F-01	3.46765922-01 -4.985938795-01 -3.873449135-01	3.784615385-01 -2.754342435-01 2.249966395-01	3.402412248-01 1.374649935-01 5.312655995-92	3.320099265-01 2.156327545-01 -1.445822995-03	-4.219362285-04 6.17535153E+34 -3.579990905-01	-4.154673285-04 -6.31430355-04 3.535235735-01	+4。209436729-34 9.429453276-05 -4.993382966402			•	
	EL EMENT REFERENCE	Y-TRANS. Z-TRANS.	K-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X+TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-TRANS. Z-TRANS.	X-TRANS. Y-T <u>r</u> ans. Z-Trans.			·	
	NODAL POINT NUMBER	25 25	26 26 26	27 75 75	82 82 82 N N N	ନ ଜ ଅ ଅ	0 0 0 N 0 0	71 21 21	25 25	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8				

	TIME IN	TIME OUT	TOTAL
INPUT AND INITIALIZATION OVERLAY	• 300	1.229	.93
STRUCTURAL STIFFNESS MATRIX GENERATION OVERLAY	1.229	2.128	
NORMAL MODE ANALYSIS OVERLAY	2.128	116.605	114.40
		k	

THIS PAGE INTENTIONALLY LEFT BLANK

APPENDIX C

EXAMPLE OF INPUT AND OUTPUT

LANDING GEAR OPTION

STRUCTURAL ANALYSIS PROGRAM

CARD GODE BLANK - 0 BLANK	- CARD CODE	BLANK - D COMMENT CARDS	I GEAR AND LOAD CARD 2 FRICTION CARD	3 APPLIEU ULSPLACEMENT CARU	4 STRUT MATERIAL CARDS	5 SULVILUM PARAMELEN LANU 6 Nodal Point Cards	P FUOLPAD CARD	6 NAIFALAL FARANLEN CANOS 8 MAIFALAL FARANLEN CANOS 8 MAIFALAL CADISH CANOS	10 COMPRESSION CONTREMISE CARUS	II TENSION CRUSH DISTANCE CARDS	12 DERINGTION DIREGENTE DIREGE	14 COMPRESSION FLASTIC FORCE CARUS	15 IEMSLUM PLASTIC FUNCE CANUS 16 UATA TERHIMATOR CARD	******************	20 CARD 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	4 4 4 4 4 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6						*** * * * * * * * * * * * * * * * * * *	ANETER CARD	2 1 5.65518 8 2 2 4 2 4 2 4 2 4 2 4 2 4 2 4 2 4 4 5 4 5	Carros 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	8. 23.5 8. 6. 3. 10.58					AMETER CARDS 	7.30E11 4.57	******************
---	-------------	-------------------------	---	-----------------------------	------------------------	--	----------------	---	---------------------------------	---------------------------------	--------------------------------	------------------------------------	---	--------------------	--	---	--	--	--	--	--	---	-------------	--	---	---------------------------	--	--	--	--	------------------	--------------	--------------------

ſ	COMPRESSION LRUSH UISTANGE CARDS
	2 8.36 11.35 14.15 27.94 1.620
	TENSION CRUSH DISTANCE CARDS
	2 1.420
	COMPRESSION SPRING RATE CARDS
	2 2482.ED6 848.ED6 1023.E06 1198.ED6 2.98E10
r '	TENSION SPRING RATE CARDS
	2 2.82£10
	COMPRESSION PLASTIC FORCE CARDS
	2 1.07E09 1.54E09 1.92E09 3.34E09 1.E30
r •	TENSICN PLASTIC FORCE CARDS
	2 . 1.630
	DATA TEKHIMATOR CARD

	MOMENTS	NATE STSTER 2	•	•••	•	SREAK Noment		• • •	LEVELS = 2	IS STEP COORDINATE SYSTEM	7	
	EXTERNAL	RFACE COORDI			••	SHEAR FORCE		••••	UUTPAD CRUSH	UN ENERGY TH. Lander	×	. •n
		33 ×	0.	.0			NODE 4	NUDE 4	UNCRUSHED FI	NE ATTENUALT	7	• •
ALYS 15	DES	E SYSTEM Z	. Ľ.			SHEAK NOMENT			NU NEEK OF	ACE CUCREINAT	-	• n
S FUR GEAR A	EXTERNAL FOR	CE COORDINATI		•••	- 0	SHEAR FORCE		•••		SURF	×	°°
CONDT I TOWN		SUKFA		••	•		NUDE 1 I	NODE 2 1		NATÉ SYSTEN	7	-
UT I TNT	CNS	STEN 4	-	9560UL+01 0	1013021	AXIAL TIFFNESS				KBEU Anŭer Coorui		• •
And	TE VEFINITI	UCRDINATE SY	0.006+01 0.	000E+41 0.	•	KIAL S		• •		ENERUY ABSO	-	•0
	LOORDINA	SURFACE CL	2.35	08E+01 0. -2.35	001+300	IAL JAL				UINATE SYSTE		•0
		AL ZI	•	6.080	-2.515	TUKAL AX	ů.	• •		URFACE CUOKL		• 0
		FOIL	1	N M	4	STRUC	-	NM		3	ſ	•

2
2
4
a
4
4
-
a
3
-
a
4
3
Ξ
-
3
3
a
4
2
Ē
1

.

NTS NTS	SYSTEM	Z	•••	•••	Shéar hoheat		ELS = 2 1EP	Y ZINATE SYSTEM	•0		NTS SYSTEH			SHEAR	0.	0.	ELS = 1	RDINATE SYSTEM Y Z	•
EXTERNAL HORE	KFACE COORDINATE	h	••	•••	SHLAR FORCE	•••	UN PAD CRUSH LEVI	LANDER COOL	0. 0.				••••	SHEAR			GUTPAD CRUSH LEVI ON ENERGY THIS S	LANDER COO	1.400E+09 0.
	70	×	-8- 0.			NODE 4 NODE 4	NCRUSHED F	- SYSTEM			ns	*			NODE 4	NODE 4	NCRUSHED F	SYSTER Z	ч.
	SYSTEM	Ζ	1.64U51E+U8 1.08227E+U8	1.04051E+06 4.90229E+08	S HE AR H UNENT	• • •	· NUMBER OF T	E CUORDINATE Y	• •	SALISFACTIC	SYS ILM	7	3.23397E+08 3.38504E+08 5.23397E+08 4.85797E+08	SHEAK		•	NUMBER OF L	E COORDINATE	9 0.
ERNAL FORCE	COURDINATE	λ.	57100E+07	57100E+07	SHE AR F CRCE			SURFAC	7 0.	STEP 2 TOLERANCE	ERNAL FORCE	_	*8900E+08	SHEAR	1 OKUL			SURF AC	8 1.400L+0
EXT	SURFACE	X	-8.U2U75E+U7 -7.	-8.UZU75E+07 7. 6.20405E+08 0.		NOUE 1 0. NUDE 2 0. NOUE 3 0.		V ZINATE SYSTEM	-2,9946+0	INCREMENTAL	SURFACE	×	-1.561361408 -1. -9.193451408 0. -1.2501361408 1.		NODE I 0.	NODE 2 0. NODE 3 0.	-	ULNATE SYSTEM	-1.1836+0
SNDITT	SYSTEM	7	U. 1.95800±+01	U. 5.69206E+U1	AXIAL STIFFNESS	5.49657E+10 2.482U0E+10 5.49657E+10	ILSORBED	LANDER COUR	.8796+07 0.	20 ITERATI	ITTONS SYSTEM	7	0. 1.95800E+01 0. 5.103406+01	AXIAL	5.49688E+10	2.48200E+09 5.49688E+10	IBSORBEU	LANDER CUOR X	.714E+09 U.
JRDINATE DEFIN	AUE CCORDINATE	.	2.35UUUE+U1 0.	-2.35000E+UI 4.	AXIAL STROKE	=3.\$9645E=03 -1.97336E=01 -3.\$9645E=03	TOTAL ENERGY A	SYSIER Z	-2.4546+07 7		ACE CUORUINATE	L.	2.35000E+01 0. -2.35000E+01	AXIAL	51KUNE -7.U7240E-03	-3.94715E-01 -7.0724UE-U3	TOTAL ENERGY A	SYSTEM 2	-1.163E+08 1
	SURFA	×	0. 6.08000 ±+01	0. -2.48960E+01	AXIAL FORCE	-1.97682E+08 -4.89787E+08 -1.97682E+08		E CUOKDINATE Y	7 0.		SURFA		U. 6.08040E+01 U. 44420E+01	AXIAL	-3.83761E+08	-9.79684£+08 -3.88761E+08		E COORDINATE Y	9 0.
	NODAL	FOINT	4 ~	n 4	STRUCTURAL ELEMENT	40.0		SURFAC X	7.879E+0		NUDAL	FUINT	4 9 19 1	TRUCTURAL	LLENEN!	210		SURFAC	1.714E+0

	SURFA	KUINATE DEFIN. CF CODEDINATE	ITTUNS SVSTEM	SURFA	EXTERNAL FORCI	SYSTEM		- EXTERNAL MO Réace coordina	NENTS TE SYSTEM	
INIO	×		7			Z	×		7	
10	U. 6. U 80 08 E + 01	2.55000E+01 0.	0. 1.95800c+u1	-1.40797E+U8 -9.98270E+06	-1.44713E+08 0.	3.19360E+08 3.45173E+06		•0	•••	
1 1	U. 2.28640±+01	-2.35000E+01	u. 5.18605±+01	-1.407976+09 1.27986E+09	1.44/13E+00	-1.J2369E+09		•••	•••	
ICTURAL MENT	AXIAL Force	AXIAL Stroke	AXIAL STIFFNESS		SHE AR	SHLAR		SHL AK	SHEAR 	
4 11 1	3.778312+08 1.07008£+09 3.77831E+08	-6.673582-03 -1.76881E+00 -6.67358E-03	<mark></mark>	NUDE 1 NUDE 2 NODE 3			NODE 4		•0	
	Ē	OTAL ENERGY A	USORBEN			TWUNBER UF UI FUUTPAD	NCKUSHED F	UCTPAU CRUSH L UN ENERGY THIS	EVELS = 1 -	
SURFACE X	COORDINATE Y	SYSTEN	LANDLK LUOR	Y Z Z	SURFAI	CE COORDINATE Y	SYSTER Z		.008014815~3 Y	ITSTEH Z
031E+09	.0	-9.7466+08 4.	.U31E+U9 D.	342.5-	iE+06 0.		0.		.	0.
DAL	SURFA	RDINATE DEFIN. Ce cuordinate	IT LONS SYSTEM	SURFA	EXTERNAL FORCI	ES SYSTEM	20	RFACE COURNINA	MENTS ILE SYSTEM	
INT	×			X			×	A	2	
4003	0. 6.08088E+U1 0. 2.26188E+01	2.35000E+01 0. 0.	0. 1.95840E+01 0. 5.19723E+01	-1.369882+08 -9.974202+08 -1.369882+08 1.271402+09	<u>-1.42380E+0</u> 6 0. 1.42380 <u>E+0</u> 8 0.	3.14886E+08 5.87351E+08 3.14886E+08 3.14886E+08 -1.01/12E+09		••••	•••••	
CTURAL BEBT	AXIAL	AXIAL Stroke	AXIAL		SHEAR 	SHE AR MOMENT		SHEAR FORCE	SHEAR	
	3.71741E+08 1.07808E+08 3.71741E+08	-6.76279E-03 -1.96548E+00 -6.76279E-03	5.49685E+10 0. 5.49685E+10	NODE 1 Node 2 Node 3			NODE 4 NODE 4 NODE 4	• • • • • • • •	•••	
SURFACE	T CORDINATE	OTAL ENERGY AU System 2	SORBED LANDER COOR X	ULNATE SYSTER	I SURFAI		MCRUSHEU F ATTENUAIL SYSTEM	COTPAU CRUSH T UN ENERGY THIS 	LEVELS = 1 SIEP CORDINATE S Y	NSTEH
3556409	c	-1-0445464	.3556+09 0.	-1-086	1E+09 0.	0.	•	0.	0.	

X 000E+01 648E+01 XIAL Orce		SYSTEM	SURFA	CE CCORDINATE	es System	SUr	EATERNAL KFACE COURDI	INATE SYSTEM	
00 E + 01 48 E + 01 IA L RG E	7	7	×	٨	1	X	λ	1	
48E+01 IAL RCE	2.35000E+01 0.	0. 1.9580UE+01	-5.29597E+07 -1.71052E+09	-1.224476+08 0.	2.90537E+08 8.72077E+06	 	•••		
48E+01 IAL RCE	-2.350UUE+UI	u .	-5.29597E+U7	1.224475+06	2.90537E+08	•0	• n	.0	1
IAL	.0	5.57597±+61	1.81644E+09	••	-1.45315E+09	0.	•	••	
RCE	AXIAL	AXIAL		SHEAR	SHEAR		SHEAR	SHEAR	
	STROKE	STIFFNESS		FORCE	HUNENT		FORCE	HOMEN	
UZE+U8	-5.81619E-U3	5.496//E+1U	NUDE 1	u.	0.	NODE 4	0.	•0	
00 E + 0 9	-1.17898E+01	0. 5.496776410	NODE 2 MODE 2		0.	NODE 4	0.	-0. 	
	TOTAL ENERGY A	ASOURED				NCRUSHED FU	COLPAD CRUSH UN ENERGY TH	I LEVELS = 1 IIS STFP	
DINATE	SYSTEM	LANDER COD	RUINATE SYSTEM	SURFA	CE COURDINATE	SYSTEM	LANDER	CUURDINATE S	TSTER
~	2	×	7 4	×	•	7	×	•	7
		*****	*****	****	****	***			
		12 LIEKAL	LONS REQUIRED	FOR TOLERANCE	SAT ISFACTION				
SURF	ORDINATE DEFINACE CORDINATE	SYSTEM	SURFA	EXTERNAL FORC	ES SYSIEM	1 0 5	RFACE COORUI	MUMENTS NATE SYSTEM	
-		7	×	ł	7	×		7	
00E+01	2.3500UE+U1 0.	0. 1.95800E+01	-5.08/76E+07 -1.70880E+09	-1.20648E+08 0.	2.80503E+06 8.75438E+68	••			
00+3 80	-2.35000E+U1 0.	0. 5.58455E+41	-5.08/76E+07 1.81056E+09	1.20648E+08	2.86503E+08 -1.44844E+09	•••	•••	•0	
CIAL '	AXIAL	AXIAL	- - -	SHEAR	SHEAR		SHEAR	SHEAR	
RCE	STROKE	STIFFNESS		FORCE	HOHENT		FORCE	HONENI	i - -
006E+08 006E+08	-5.73075E-03 -1.19952E+01	5.49676E+10 0.	NODE 1 NODE 2	••		NODE 4 NODE 4	ч. С.	•••	
JU6E+U8	-5.73075E-03	5.49676E+10	NODE 3	• 0		NODE 4		•0	
	TOTAL ENERGY A	BSORBEU			NUMBER OF UI FOOTPAD	NCRUSHED_FU	COTPAD CRUSH	r LEVELS = 1 IIS SIEP	
EDI NATE Y	SYSTEM . Z	LANDER CUDI	RDINATE SYSTEM Y z	SURF A X	CE COORLINATE Y	SYSTEM	LANDER X	COORDINATE S	YSTEH Z
•	5 440E+04		046.4-	E+09 0.			•0	.0	

INCREMENTAL STEP

1 ۱ . . ł ł -----, ì l i -----1 ----1..... ł 1 ł Ì ; ; ; ł i TCTAL 5.80 1 13. ļ 1 50.050 TIME CUT -----728.00 i ţ l 56.025 : 50.030 ī TIME IN ! -----CELL IINE USAGE TABLE (SEL) I 1 1 -----;;; 1 į i 1 ţ ł i ļ ļ ļ Ì ł i ļ ---i ļ 1 i ---i ļ ····· i 1 INFUT AND INITIALIZATION GVERLAY -----GEAK INFLT AND ANALYSES CVERLAY 1 1 ł 1.1 : 1 ; ţ ţ • ı ł i ----į

APPENDIX D

EXAMPLE OF INPUT AND OUTPUT

LANDING LOADS AND MOTIONS PROGRAM

	ILM CARDS READ THIS RUN - 1 21 400 4	Z 2 2 4 2 2 00 300 300 30 21 10 11 12 12 13 14 19 10 17 10 19 20 400 400 400 400 400 400 54 35 35 37 30 35 25 27 26 40 40 40 400 50 90 90 90 90 600	
	1 CASE NC = 306 1 CASE NC = 306 2 NEGRES = 1000C = 0 4 INDFXD = 0 INDFYD = 1 INDFZ 5 INDFXR = 1 INDFYL = 0 INDFZ 6 HAXF = 1 INDFZ = 0 UNDFZ 7 IVACF = 1 INTH = 0 CUTER = 8 NFGRC = 1 IQUOL = 0 JCKSAB	0]RLE6 =]]FFFRT =] 0 = 0 2 = 1 2 = 1	i I I
1 CASE NC = 306 2 TIMAR # IJUUGE-DI IFTCAT = 10 INLEG = 1 IFFFRT = 0 3 NAGUES = 3 NOGUT = 0 4 INDFXD = 0 INDFYD = 1 5 INDFXR = 1 INDFYR = 0 5 INDFXR = 1 INDFCPR = 1 7 LVARF = 1 INTH = 0 CUERR = 160% - 4 7 LVARF = 1 INTH = 0 CUERR = 0. CMIN = 0. IF = 0 7 LVARF = 1 INTH = 0 CUERR = 0.05EIN = 0.	* * INTIAL CONLITIONS * 9 2eta = 0.	* - D7E+UC 6KAVE = 9.037E+U2 DTE+U2 6KAVE = 9.037E+U2 MZ = 0. VELZ = 0.	
1 UASE NG = 366 1 UASE NG = 366 2 TIMMAX IJUUGE-UN IFICKT = JU INCEG = NIFFKT = 0 2 TIMMAX = JUUGFUG = 0 4 INUFXG = 1 INOFVG = 1 INCFZD = 0 5 INOFXR = 1 INOFVG = 1 INOFZR = 1 7 IVAKT = 1 INTH = 0 CUIERR = 1.006C-04 7 IVAKT = 1 INTH = 0 CUIERR = 1.006C-04 8 MFCKC = 1 IQUOUT = 0 JOSETK = 0 7 IVAKT = 1 INTH = 0 CUIERR = 1.005ETK = 0 8 MFCKC = 1 IQUOUT = 0 JOSETK = 0 8 MFCKC = 1 IQUOUT = 0 JOSETK = 0 1 INTH = 0. MECK = 5.837c+02 1 AND = 0. MECK = 5.837c+02 1 AND = 0. WZ = 0. WZ = 0. 1 AND = 0. VECZ = 0. 1 AND = 0. VECZ = 0. 1 AND = 0. WZ = 0.	* * CENTER BOLY GAIA * * 13 CEMASS = 1.8056+05 14 CEIXY = 3.US1E+05 15 CEIXY = 0.	592E+08 CbIZZ = 4.508c+168 CbIYz = 0.	
$1 \text{ UASE NC} = 306$ $1 \text{ UASE NC} = 306$ $2 \text{ THMR X I UUVE-UT IFTUKT} = 10 \text{ IMEK X = 1 UUVE-UT IFTUKT} = 1$ $2 \text{ INDFXR = 1 INDFVU = 1 INEFVU = 0$ $5 \text{ INDFXR = 1 INDFVK = 0}$ $5 \text{ INDFXR = 1 INDFVK = 0}$ $7 \text{ INDFK = 1 INDFVK = 0}$ $7 \text{ IVAF = 1 INTH = 0 \text{ UNEFK = 1}$ $6 \text{ MEDUC = 1 INDETK = 0}$ $7 \text{ IVAF = 1 INTH = 0 \text{ UNEFK = 1}$ 9 ZETA = 0 $4 \text{ INTHA COMPITIONS * * 0}$ 9 ZETA = 0 $4 \text{ INTHA COMPITIONS * * 0}$ 9 ZETA = 0 $4 \text{ INTER COMPITIONS * * 0}$ 9 ZETA = 0 $4 \text{ INTER COMPITIONS * 0}$ $4 \text{ INTER COMPITIONS * 0}$ $4 \text{ INTER COMPITIONS * 0}$ 5 ZETA = 0 $4 \text{ INTER COMPITIONS * 0}$ $4 \text{ INTER COMPITIONS * 0}$ $4 \text{ INTER COMPITIONS * 0}$ 5 ZETA = 0 $4 \text{ INTER COMPITIONS * 0 \text{ INTER COMPILIENCE 0}$ $4 \text{ INTER COMPILIONS * 0}$ $4 INTER COMPILIONS * 0$	NU SECULARY ULTUTIS ** F GUTPAL DATA * * Ib FFMASS 3.4976403 If #AD(I) = 1.2564401 If #AD(I) = 2.21864405 If #AD(I) = 2.2286405 If #AD(I) = 2.2286405	E+01 1.005E+01 1.005±401 970±400 5.870±400 170E+06 5.170E+00 170E+06 5.170E+00	
1 USE NG = 36 1 USE NG = 36 2 NOUNS = 1 NOUY = 1 UTKED = 1 TFFRT = 0 2 NOUNS = 1 NOUY = 1 UTKED = 0 5 NOEKS = 1 NOUY = 0 NOFTS = 1 5 NOFK = 1 NOUY = 0 NOFTS = 1 7 TARF = 1.000 - 0 - 0.05XA = 1.000 - 0.0 7 TARF = 1.000 - 0 - 0.05XA = 1.000 - 0.0 7 TARF = 1.000 - 0 - 0.05XA = 0 - 0.05Th = 0.0 7 TARF = 1.000 - 0 - 0.05XA = 0 - 0.05Th = 0.0 8 22TH = 0.0 - 0.05XA = 0 - 0.05Th = 0.05Th = 0.0 1 TARF = 1.000 - 0 - 0.05XA = 0 - 0.05Th = 0.05Th = 0.0 1 TARF = 1.000 - 0 - 0.05XA = 0 - 0.05Th = 0.05TH	+ + SCIL UATA + +		

						and the second se	2.5 12 T ± 0.7		• ALLE - C.								0.0.1.1.1.L												tion of a constant of a constant of the second of the seco		مىرىنىيى مىلىكى مىلىكى مىلىكى مەردىكىكى مەردىكەن مەردىكەن مەردىكە مەردىكە مەردىكە مەردىكە مەردىكە مەردىكە مەردى مەردىكى مەردىكى					a company of Names of A of the contract		
		с+ uź Е+01	ctul S.Santeria	5 • Jù li E + L 4	6, 300E+10	90+389I.T	SKCTPS =		AC14 - 4	E+01	t+01	c+u1	2.750L+11	Z./50E+11 5.ul/0F+60	5.000L+t0	5.500±+10					:																	
	= 1.4481+02 = 7.2391402		ZMSCE = -5.677	5 + 0 - 0	5.000E+00	- 1.148E+d9 2.500_+45	ns = 1.420c+10		201506 = 5.358	ZUSCE = 9.558	1421-9- = 3207	2056 = -2.065	2.23GE+11	2.2305+11 4.0505+50	4.000±+00	5.5u6c+10								Z = 1.971E+04	Z = 2:1716+04	1.0036+00 1.0036+05	2.3Z1.+66	1145-115		0646401. 	0046-U.L	3056-01 1886-01 1985-01		04JE-02	50 3E-0 2	24.4E−U2 804E−U2		206E-01
	26P 46+02 26P	4 <u>6+02 2</u> 79 0. 9.8326+U1	9.632E+U1 1.42.65.434	5.uuue+u9	4.Ju0E+u0	1.0236+U9 2.58664-J9	EFUO SKUCI		2.356E+U1	2.3502437 4.3345461	t.965L+U1	9.5142+01	0.5056741 1.0566411	1.650E+11 5.0006+00	3.U00L+00	5.506E+10	5-500C+10 E+01 SRUCI	- 7MC -				459L+01	10+3010 10+3010	365+04 MN. 745+04 MN.	946+04 MN	E+10 P2 =	$E+06F_z = 2$		t-61 8. 4.	E-01 5.			E-U1 0.	E-01 0.	- <u>-</u>	E+00 - 5.	• • • •	E-61 5.
	н Ц. Н 1.25	= 1,23 MSLB = - MSCB = -	MSC2 = 12+00	0c+19	01+00	0 <u>5+08</u> -	= 2.000		DSCa =		05CE = -	DSCE =	0c+11	02411' 0-+60	0E+00	0+10		; ' c	•			6A = 5.	6A = 1.			= -2.423 = -108	= -2.353	600°*	4 .605 0.	5.182	201.6-	130°C 167°S 167°S	-3.800	102.01	-6.000	1.JOU	•••	-7.574
17A * *	. = J UKA65T .648E+01 YFP .648E+01 YFP	3.6488401 YFP 2.9496+01 Y 2.9496+01 Y	2.949£+[1 Y 106+00 1.52	00-2 2.00	10E+00 5.00	52E+09 C+48 :0F+09 2-50	E+01 STMXMS	0 1 1 0 0	-3.139E+61 Y	-3:139E+01 T	-3.134E+L1 Y	-5.139E+u1 Y	0E+10 1.10	50£+10 1.10	10E+00 2.00	10E+10 5.50	16+00 STMXDS			─── ─ + + ∀-	1	3.405E+04 OME	0.800E+04 UME	8-4042+04 WN 2.6685+04 WN	5.6315+04 WN	2.4566401 PY 1.6776403 PY	1.852E+02 PY	2.69UE-U2	2.690E-02 0.	2.265E-U1	-C.CO2E-UL	4.000576-01 6.0776-01 6.776-01	0.	• • •	•			7.6501-01
LÉG UA	3 ILE P = +5 P = -5	SCB = 5	508 = 1.04	5.00	2. 0 2. 1 1 1 1	2 4 5 1 4 1 4 1 4 1 4 1 4	2.794		SCB =		1925 808	5CB =	- 270 - 2.56	- 90.91 	1.00	5.51	5.000		5	MODAL	ų	999 111 21	55 11 H 11 H			× × 11 11	1 = X	H XSH	H XSM		HSY =			= XSO	D3X =	LSX =		- 101
*		ながなって	× "	וו ו	1 II 0 10	ի ։ Խ <i>ս</i>	i S I	= SH	יא די	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	5 ¥	5 ×	, , ,	ji ii An u	i i nin	וו אנצי	- S -	H SQ	2	+	1 th		h m	40	 		~ 4		нц 10-	. uz 1 1 (N 1	,		- u 	~ ~	F			- u - (*)
	NCLE 1. *	и н ј н н јн	1 #	F 1 F	CLTM	SREF	Stad	14E 1) +	ju u ∳4 ⊨	•)) • (ן נו נו	PF CL:		CD1-0.	SECL	SCMXI	TKE T	ن ۱۱۰۰		terit.) - 	4 H	∦ • ∔• -	• • • •	и и н нч	"	0 	.∦ II 	1 31 1 4 4 7 1	" -• f	н н н н н н н н н		н н Н П	Ļ	н н 	и и Н (-	1 II 11
1	2002	510 3)(5	300	25	27	82	-02	1	100	004		400	2 7 2 7 4	1 IS 10	200	37	55	- 4 #	;		64	100	005	009		004	102	900 900	009	005	nn5	10000	1100	1100		0011	1200	1264

1.850±~U2 -9.206£-01 1.8880£-02 1.8880£-51 1.8886£-51	-3.06UE-U1 -6.047E-U1 -5.060E-U1 -5.047E-U1 -5.47L-01 PCGZ = -4.050E-U1 PCGZ = -4.050E-U2	FL62 = 3.489E-U.	
0. -7.650E-01 7.574E-01 8.724E-01 =2.807E-01 8.724E-01 -2.807E-01	4:297E-01 1:638E-01 -1:200E-02 5:900E-01 4:297E-01 1:638E-01 -1:200E-02 5:900E-01 0. PCCY 0. 3:500E-03 PCGY 0.	n. 	
1200 1 = 4 FUSY = 1200 1 = 5 FUSY = 1200 1 = 6 FUSY = 1300 1 = 2 FUSZ =	1300 1 = 2 F0SZ = 1300 1 = 4 F0SZ = 1300 1 = 6 F0SZ = 1300 1 = 6 F0SZ = 1700 1 = 2 F0GX =	1/00 1 = 3 PUGK =	

	ACE COCREINATE	SYSTERJ					
Å	?						
U	30°0 0.0						
. INITIAL C.G. VELOCITY	00°0 0°	:					
INITAL ANGULAR VELOCITY	JO . C. 00	1					
						•	
			* * *				
TIME = 0. STEF 5426 = 1	.800001E-04					,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	:
CENTER BOUT MOTIONS	:	a e					
TRANSLATION - SURFACE COCRDINATE SYSTEM	ROT FTION	,		VELOCITY ARE AC	 betekATI6N		
UISP VEL VEL ACCEL X 7.924±+01 -1.146±+03 -9.807±+02 X 0. 0. 0.	EULER 1	ANGLES PHI THETA FSI	9.5506+50	LANLER CCCRUINA X L Z U	15 SYSILM	i İ	
FUOTPAD POTICNS - SURFACE COORUINATE SYSTEM		-	and of the second second second				
FOCTPAD 1 015F VCL 015F VCL 1405F01 -1.1405F01 1405F01 -1.1405F01 1405F01 -1.1405F01 -1.1405F10 -1.	AULEL 3 =9.83754U2 3 =9.84724U2 3 =9.84724U2	Dist L. 1.254c+Uc 1.254c+Uc		аност 1.5215 • • • • • • • • • • • • • • • • • • •		а. В. В. В. В. В. В. В. В. В. В. В. В. В.	
" ACCELERATIONS IN EARTH G*S = LANUER CUCRUINATE > X γ "" CENTER CF GRAVITY =9.663E=U1 """, "-1.4	YSIEM 2 50e-01					- - 	
- " LANEIKG GEAR STRUT INFORMATION"' SIF Leg crush - Lading Str	CKE		01KUK	بر ا	- LúÂIJ	0 I KLAE	
L C C HAIN STRUI U. U. C A AIN STRUI U. C A AIN STRUI U. U. U.	LKAG ST	1 1 2	۰. د د د:	טאאים בואטר 2 החשה בואטר 4 טאאים בואטר 4	••• •••		
	*	*	* * *				
TIME = 1.003006-33 STEP SIZE = 1	.000000£-04	4 		e incine e contraco			
CENTER BODY HOTIONS							
TRANSLATION - SURFACE COGRDINATE SYSTEM	HUTATION			VELUCALY AND AU	ÚZLERATIUN		
X 7.810±401 -1.142±403 1.166±404	EULER /	INGLES PHI	•	KANDER ULURUINA X 6	TE SYSTEM u.		
Z -1.730L-04 -7.044L-01 -1.7792+03		PS1	9 • 5 t a : + : 5	7 -1 c-81+88	- 3. 909L + cc		
FOCTFAL MOTIONS - SURFACE CCCRLINATE SYSTEM					7		
ULSP VEL	AÜLLL	ÚISH	VLL	ALGEL LISP	VLL	AÜLEL	

									:									
+-L 0.04614.4 +L1 +1.06014.5 -11 +1.00014.5	: : I	01 RCKF - 1-107	-1.0766-46							ACCFL 51.054.4	FC1"=2.852224J5		51 xCKc 24 J4J6-10 15.3916-10 15.3916-10		-			
5616+u2 0.00464 6086+u1 -3.04564 6086+u1 -3.04564		LUAU Kut Z -6.5512+07	KUT 4 -1.4292+19 KUT 6 -1.4252+09				NG AGGALERATION	KUINAIL SYSTEM Kuinail System Leuu -1.4586444		ŭISP 2. Jele 1015P 3.Jo7€3	.2126+01 -6.1226 .2126+01 -6.1226		kuī 2 -1.11664U Ruī 4 -2.9656+69 Kuī 6 -2.9656+09					NU ACCELEKALIJUN Kuinate System 0 1-2.0511+64 3.
441,405 -6 444,405 -6	:	o URAU J	E DKAG LI				161 DCT (Y - 2		-n Z	1	5UE+US −0 00≟+US -6	,	B UKAC UI DRAC UI DRAC UI	1				ANDER ULU X 16. ULU Y 2. UC
25+131 -14 25+131 -14 25+131 -14 29		⇒TKCKÉ -1.18rt-lu	-4.45EE-12 -6.450E-10	*			-	• + ⁻ C • + ⁻ C	•	Ч ЕТ	16+U2 -5,36 16+Ū2 5,86		51 KCKE -6.03 + 4-03 -6.439E-03	4 				• + a u r +
+L+0£ +0.12 +E+02 0.12		LÚAU -6.531.tuř	-1.051	*			ł	E ATAAT S	- ISd	2 ~	+ L+02 -1.1 9 + L+02 -1.1 9		LCAU -1.1161-0 -3.541E+0 -3.541E+0	;		1		5 Р НГ 3 ТНЕТА 9 PST 9
+84 6. +84 -1.c55 +84 1.255		AG STALT 1	AG STRUT S AG STRUT S	*	14		NOT	JLER ANGLES		10 n hea	F04 -1.251		NG STRUL 1 NG STRUT 3 NG STRUT 3	:			ION	חרבע שאפרבי
-03 -1+ 1448- -03 -1+0478 -03 -1+0478	SYSTEM Z 735c-u1	ŘСКЕ 14/Е−Ũ5 LR	17E-05 LK	*	1.000000c-		ROTAT	Ę		ACCE1 46CE1 -03 -1.504č	1517.5- 50	SYSIEM Z Z756-U1	RCKE 141E-03 LK 892E-01 LR	: : :	1.000JBBE-		ROTAT	
11 -1.149t+ 11 -1.625t+ 11 -1.023t+	UORUINATE	140 ST 25+05 -6.3	112+08 -6.5 112+08 -6.5	*	F 512E =		rstem		8E+D3	STSTEN X 1 -1.15064	10 - 8.020E+	UCRUINATE	140 51 06+06 3.8 865+08 -2.4	4			VSTEM	.CEL .32+04 .72+03
4.6296+0 1.0526+0 1.0526+0	+ LANUER C X D3E+J1 D.	ricn Lo Strut -2.07	518UI -1.61 518UI -1.61		30E-93 STè		CKUINATE S	56+03 3.47 66+03 3.47	14.2- CU+35	CCRUINATE	9.606E+1	- LANUER (2 326+01 0.	FION		00E-03 SYE	:	OCKDINATE S	54 01 11 11 11 11 11 11 11 11 11 11 11 11
	EARTH 6*5	JI INFCRMAT	-UL MAIN -UL MAIN	*	= 2,00001	SNC	SURFACE LE	- 10+3 - 10+3 - 10+3	-03 -3.392	Sukrace		EANIH 6*5	UT INFCKMAT For Main S For Main S	4		SNO	SURFALE CO	ve 101 -1.101 0.
FOCTPAU 1 FOCTPAU 2 FOCTPAU 3	CULLERATIONS IN CENTER CF GRAV	ANELNG GEAR STRL Leg crush 1.	2 6.632E- 5 6.632E-	***	11HE	NIEK BODY MOTLO	TRANSLATICN -	Y 0.	1168.1- 2	DUTPAC NUTIONS -	FOCTPAC 2 FOCTPAC 3	CCELEFATICNS IN Center CF GKAV	NULING GEPK STRU Let		1146	ENTER BOUY NOTIO	TRANSLATION -	D15F 7,585E 7 0, 7 0, 7 - 7,924E
	A	` ت		F		5				u i		Ŧ			•	5		

SLATION - SURFACE COCRUINATE SYSTEM ROTATION DISP VEL ACCELERATION X 4.730E+01 3.743E+03 Y 0. LANDER CUCKULIATE SYSTEM Y 0. 0. Y 0. 0. Z 1.015E+01 -1.125E+03 EULER ANGLES PHI 0. 1.01104 Y 0. 0. Z 1.015E+01 -1.125E+03 EULER ANGLES PHI 0. 1.01104 Y 0. 0. Z 1.015E+01 2.025L+02 ANTIONS SURFACE COORUINATE SYSTEM 0. NOTIONS SURFACE COORUINATE SYSTEM Y AD 0. 0. 0. AD 0. 0. 1.05P Y 0. 0. 0.
DISP VELOCITY AND ACCELEMATION X 4.730E+01 3.743E+01 -1.125E+03 EULEK ANGLES PH1 0. LANDEK LUCHUINTE SYSTEM Y 0. 0. 0. 0. 0. 0. 0. Y 0. 0. 0. 0. 0. 0. 0. 0. Z 1.0155+01 2.0055+02 0.0055+03 EULEK ANGLES PH1 0. 2 0.
X 4.730E+01 3.743E+01 -1.125E+03 EULER ANGLES PH1 0. X 0. 0. 0. 0. 0. 2 0. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2. 2.
HOTIONS - SURFACE COORUINATE SYSTEM X PAD 1 015P VEL ACCEL D15P VEL VEL 015P VEL 015P VEL 015P VEL 0159 VEL 0159 VEL 0139 VEL 0139 VEL 0139 VEL 0130 VEL 014 0130 VEL 014 0130 VEL 0130 VEL 014 0130 VEL 014 0130 VEL 0130 VEL 0130 VEL 0130 VEL 0130 VEL 014
X X 2 4 DISP VEL ACCEL DISP VEL VEL USP 155E+00 2:912E-04 -5.825E+00 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0.035E+02 9:938E+
FRU 2 00.3905444 00.2012444 11.44205443 11.2972414 1402544 10.417544 -0.7725741 1.5902462 3.09024 PAD 3 8.3986400 6.75/E4ud -1.4562403 1.2972402 -1.03124400 1.4172404 -6.5756401 1.5902462 3.09024
ATICNS IN EAKTH 6*S - LANDER COOKULNATE SYSTEM
ER CF GRAVITY -9.464é-01 0. 6.2376+00
GEAR STRUT INFORMATION
CRUSH LOAD STROKE STROKE STROKE STROKE
2.6534£400 MAIN STRUT 01.6702401 LKAG STRUT 1 -2.00244400 -2.6445£55 DKAG STRUT 2 -2.0472406 -3.649£555 -2.33355400 MAIN STRUT 01.13955401 LKAG STRUT 3 1.6045408 2.9176±03 DKAG STRUT 4 -1.4665408 -2.7025405 -3.3395400 MAIN STRUT 01.14545401 LKAG STRUT 3 1.66444408 2.9176570 UKAG STRUT 6 -1.4665408 -2.7025405
BGDY MOTIONS
SLATION - SURFACE COCRDIMATE SYSTEM ROTATICN
DISP VEL AGCEL VELOCLERATION 7 4.734E+01 3.015E+01 -1.530E+03 -1.530E+03 7 0. 0. 0. -1.530E+03 7 0. 0. 0. -1.530E+03 2 9.944E+00 1.540EK -1.540EK -1.540EK
HOLLONS + SURFACE COORDINATE SYSTEM
PAD 1 VEL ACCEL DISF VEL ACCEL UISF VEL ACCEL DISF ACCEL ACCEL UISF ACCEL ACCEL
ATICNS IN EARTH 6*S - LANUER CUURDINATE SYSTEM
ER CF GRAVITY -9-2992-01 U. 1.543E+01
GEAR STRUT INFORMATION

-01 SILF SILE = 1.000u0uL-04 11 SYSTEF -RUTATION 11 -1.00uu0uL-04 -1.00uu1 12 2.000E04-04 -1.00uu0uL-04 12 2.000E04-04 -1.00uu1 12 2.000E0404 -1.00uu1 12 2.000E0404 -1.00uu1 13 2.000E040 -1.00uu1 14 2.000E040 -1.00uu1 15 2.000E040 -0.00uu1 14 2.000Uu0 -0.00uu1 14 2.000Uu0 2.000Uu0 15 2.000Uu0 2.000Uu0 10 2.000Uu0 2.000Uu0 11 0. 1.000Uu0 11 0. 0. 11 0. 0. 11 0. 0.		VELGCIIT ANU AGGELENATIUN Lanek Luoruinate System X 0. v. 2 0. 0. 0. 0.	ALCEL UISP 2 0 ACCEL UISP 2 5.C08E+04 -0.557E+01 1.597E+02 -7.725L+04 5.U08E+04 -6.557E+01 1.597E+02 -7.725L+04		KE LUAU STRCKE 28-03 UKAU JIKUT 2 -5.7562+U7 -1.4662+U3 66-03 URAU STRUT 4 4.3276+D3 7.8606-D3 65-U3 URAG STRUT 6 4.3276+D3 7.8682+U3					
01 STCF S1.06 1.000 11 -1.004+65 -1.004 12 -2.6026+04 -1.01 12 -2.6026+04 -1.01 12 -5.6026+01 -3.504 12 -5.5036+01 -3.504 12 -5.5036+01 -3.504 12 -5.5036+01 -3.504 14 -5.5036+01 -3.504 14 -1.11 -3.504 14 -1.11 -3.504 11 0 -1.11334+01 11 0 -1.11334+01 11 0 -1.11334+01 11 0 -1.11334+01 11 0 -1.11334+01 11 0 -1.11334+01 11 0 -1.11334+01 11 0 -1.11334+01 11 0 -1.11334+01 11 0 -1.11334+01 12 5.5006-02 3.01 13 5.5006-02 3.01 14 0 -1.11334+01 15 0 -1.11334+01 16 0 -1.11334+01 17 0 -1.11334+01 16 0 -1.1144	101L-04	CUTATION LULEK ANGLES PHI U. PSI U.	AUCEL DISP vel 5276403 - 0.5P vel 6376403 -1.2966402 -1.4476401 -	10	LUAU SIRUI 1 -5.756E+J7 -1.040 Lurag Sirui 1 -5.756E+J7 -1.045 Lurag Sirui 3 -4.091L+V0 -7.439 Lurag Sirui 5 -4.051E+J8 -7.459	and and a second second second second second second second second second second second second second second se	20KL 566-92	000 t 0 000 t 0 335 - 02	J5E-U2 5E-02 15E-02 55c-02	
	01 STEF SIZE = 1.000	INATE SYSTER - F Acuel 1 -1.904e+65 2 2.802E+04	JIATE SYSTEM D1SF 157E+00 9.852E-01 -3. 445E+00 2.559E+01 2.	ANDER CUDRLINATE STSTET Y Z J1 0. 2.924E+1	LOAU SIRCKE 1 01.7795+0 1 01.183+40 1 01.1835+40		UI 1.350E-UZ 7.11 UI 0.	UT U. TENSION TIME TENSION 2.2506-02 4.55 02 5.2606-02 4.51	UZ 5.490E-UZ 3.01 02 3.710E-UZ 3.01 02 5.710E-UZ 3.01 02 3.710E-02 3.01	FOK 58.524 CF SECONDS

APPENDIX E

OPERATING INSTRUCTIONS CENTER BODY LANDING LOADS PROGRAM E.1 <u>Introduction</u> - The Center Body Landing Loads Program retrieves and lists, for specific points in time, the information placed on magnetic tape by the Landing Loads and Motions Program. Information continuously stored on tape consists of landing gear strut forces acting on the center body; angular positions, velocities, and accelerations and the translational accelerations of the center body center of gravity; and the displacements, velocities, and accelerations of the normal modes employed in representing the flexible center body structure. The Center Body Landing Loads Program uses the information on this tape, in addition to other required input data, to obtain the distribution of landing gear strut loads and inertia loads throughout the center body structure for the selected points in time. This load information may then be used in conjunction with the Center Body Option of the Structural Analysis Program to determine the internal member load distribution for the time point of interest.

E.2 <u>Input Data</u> - A number of input data cards are required by the Center Body Landing Loads Program. These properly initialize the program, define the mass properties of the center body, and define the points in time during the landing time history at which the loads are to be determined. The required format for these cards is shown in Figure E-1 while the input variables are defined in Figure E-2. Also indicated in Figure E-2 is whether the various input quantities must be input as integer or floating point numbers. All integer numbers and floating point numbers input with an E format must be right justified. Floating point numbers input with an F format may appear anywhere inside their allotted field on the data card.

In addition to the required card input, two magnetic tapes are required. The mode shape data and joint coordinate information is required on TAPE4. This tape is generated when performing the modal analysis on the center body structure with the Structural Analysis Program. Secondly, the time history data obtained from the Landing Loads and Motions Program must be available on TAPE3. Since the time history data on this tape corresponds to points in time when normal printed output was obtained from the Landing Loads and Motions Program, only center body loads at these points in time may be obtained with the Center Body Landing Loads Program.

COLUMN:						e.			
1 5	6 10	11 11	5 16 20	21 25	5 26 30	31 35	36 40	41 45	46 50
NOJTS	NOLEG	NOMODE	NOTIME	NOSETS					
MODE(1)	MODE(2)	MODE(3)	MODE(4)	MODE(5)					
(1)SMTOL	JOTMS(2)	JOTMS(3)	JOTINS(4)	JOTMS(5)					
JOTDS(1)	JOTDS(2)	JOTDS(3)	JOTDS(4)	JOTDS(5)	JOTDS(6)	(L)SQTOL	JOTDS(8)	(6)SUTOL	JOTDS(10)
ZETA		GRAV							
TYM(I)		TYM(I+1)		TYM(1+2)		TYM(I+3)		TYM(1+4)	
AMASS(I)		AMASS(1+1)		, AMASS(I+2)		AMASS(1+3)		AMASS(1+4)	

2
~
3
0
Ř
ц.,
S
6
_
C 5
ž
Ξ
Z
A
أحجر
>
Ď
0
$\boldsymbol{\omega}$
2
ш
2
5
<u> </u>
. 1
2
õ
L.
Ø
F
4
Ω
L
5
IPUJ
INPUI
LUUN
-1 INPUT
E-1 INPUT
E-1 INPUT
RE E-1 INPUT
URE E-1 INPUT
GURE E-1 INPUT
FIGURE E-1 INPUT

INPUT VARIABLE	TYPE NUMBER*	VARIABLE DEFINITION
NOJTS	I	NUMBER OF JOINTS IN CENTER BODY STRUCTURAL IDEALIZATION (74 MAXIMUM)
NOLEG	I	NUMBER OF LEGS (5 MAXIMUM)
NOMODE	I	NUMBER OF MODES INCLUDED IN ANALYSIS (5 MAXIMUM)
NOTIME	I	NUMBER OF TIME POINTS TO BE READ FROM SECONDARY TIME HISTORY TAPE (TAPE3) OBTAINED FROM LANDING LOADS AND MOTIONS PROGRAM (20 MAXIMUM)
NOSETS	I	NUMBER OF NORMAL MODE SETS APPEARING ON TAPE (TAPE4) OBTAINED DURING NORMAL MODE ANALYSIS WITH THE STRUCTURAL ANALYSIS PROGRAM
MODE(I)	I	MODE NUMBER OF ITH NORMAL MODE OBTAINED FROM THE STRUCTURAL ANALYSIS PROGRAM (5 MAXIMUM)
JOTMS(1)	I	JOINT NUMBER LOCATION OF I TH MAIN STRUT (5 MAXIMUM)
JOTDS(I)	I	JOINT NUMBER LOCATION OF ITH DRAG STRUT (10 MAXIMUM)
ZETA	F	GROUND SLOPE
GRAV	F	LOCAL ACCELERATION OF GRAVITY
TYM(I)	F	TIMES, IN ASCENDING ORDER, AT WHICH CENTER BODY LOADS ARE TO BE DETERMINED (20 MAXIMUM, 5 TO A DATA CARD)
AMASS(I)	F	MASS ASSOCIATED WITH ITH STRUCTURAL JOINT. (74 MAXIMUM, 5 TO A DATA CARD)

*I – INTEGER NUMBER; F – FLOATING POINT NUMBER

FIGURE E-2 INPUT DATA - CENTER BODY LANDING LOADS PROGRAM

The use of the input indicators NOSETS and MODE(I) require some additional comments. The modal analysis output obtained from the Structural Analysis Program is placed on magnetic tape in data sets each containing information for five modes. The first set contains the lowest five elastic modes, the second set contains the next five modes, etc. The input indicator NOSETS defines the number of these modal data sets that appear on TAPE4. The modes in the first data set are numbered 1 through 5, those in the second data set 6 through 10, etc. MODE(I) defines the mode number of the Ith mode which is required by the Center Body Landing Loads Program.

E.3 <u>Output Data</u> - For a selected point in time the landing loads at each of the joints used in the center body structural idealization are printed. These loads are the combination of the inertia loads, gravity loads, and landing gear strut loads. The components of the landing loads are given in the Lander Coordinate System and thus may be directly input to the Center Body Option of the Structural Analysis Program. In addition, the sum of forces and moments about the center of gravity are given for information purposes.

E.4 <u>Program Operation</u> - An example of the output from the Center Body Landing Loads Program is given on the following pages. This information was required during the analysis of the Task Order Three lander, Section 6.2.3, at a time of 0.011 seconds. This time was selected for determining internal loads distribution in the lander center body using the Center Body Option of the Structural Analysis Program as discussed in Section 6.2.4.

A listing of the Center Body Landing Loads Program follows the example output.

Ŧ

			· · · _ · _ ·			A many structure of a																		and more and a d	PA (s) "Symbol of an an an an an an an an an an an an an			a manual of the second of the							÷	ar air air	
		a and a second and a second and a second and a second and a second and a second and a second a second a second	· • • • • • • • • • • • • • • • • • • •						11 M M M																												
2 8.724E-61 8.724E-01	8.9755-01 1.000-440	8.730E-U1	5.000E-01	5.416c-01	4.2255-41 4.2976-01	2.3756-01	6.077E-U1	-5.483c-83 -1.197,-82	-6.1356-01	-6.076±-01	-5.4366-03 -1.105r-12	2.376c-01	2.153e-01 6.077E-01	4.530E-61	5.001E-U1	4-99601	5.017c-01 -6.164c-01	-6.1785-01	-6.1/UE-U1	1	2	2.9726-01 -2.9526-01	3.151E-01	-2.36215-U1	2-8275-01 -	1.548E-01	-7.311c-02 -1.4376-41	-1-4666-01	-3.497E-U1	<u>3.7975-61</u> -3.490c-01	-5.631E-01	9.5/26-W1 -1.795£-01	1.617E-U1 2 5815-01	-5.484E-01	5-8206-01 -3-3846-01	3.681-01	-3.356-01 -1.3946-u1
Y -3.097E-14 -4.304E-04	5.368E-07	3.105E-34		6.40CE-01	7.656±-01	4.217E+01 3.028E-31	2.265E-01	1.4532-03 9.666F-04	-4.452E-05	<u>+4.4486-05</u> -4.4486-05	-1.942E-U3 -1.056F-J3	-4.218E-01	-3.939E-ŭ1 -2.266E-ŭ1	-7.601E-u1	-6.427E-01	-6-426E-01	-6.400E-01 3.199E-05	3.2016-15 - 2001-05	<u>3.2445-45</u>	:	X	9.60665-J3 -9.6166-D3	8,568E-03	-0.330£-U3 9.796£-U3	7.421E-03	-0.326E-U3 2.014E-01	-1.2016-01 -2.3665-01	7.2946-01		<u>-3.8386-01</u> 5.1826-01	6.478E-13	-5.304E-U3 -4.204E-03	2.165E-U3 / 2066-03	-1.487E-32	1=3055-U2 -3-8265-01	3.861E-U1	-5.258E-01 -7.257E-01
× -7.58456-14 -7.55856-04	-7.777E-04 -5.363E-02	-7.6466-04	-1.597E-03	-1.578E-03	-2.9446-03	1.105E-04	2.690±+02	3.688E+03 3.670F-03	3.1496-05	3.110E-05	3.6705-03 3.6705-03	1.1056-04	3.701c-04 2.651c-02	-2.956£-03	-1.597e-03	-1.561E-03	-1.578E-03 -2.033E-05	-2.024E-05	-2.1236-15	н З	×	-3.986c-01 -3.9916-01	-3.863£-01	-3.8/36-01 -8.116E-01	-3.7466-01 -3.746-01	-3.227E-01	3.229E-01	9.9985-01	1.934E-01	1.939E-01 4.005E-01	-6.157E-01	-6.194E-01 -6.196E-01	-6.201E-01	-5.9186-01	-5.916č-01	2.041E-01	4.217E-01 9.961E-ů1
JCINI 2	2 - 2 - 2	10	8	10	11	13	15	16	18	50	21	38	24	26 27	28	62	30	35	كت	MODE.	INIC	- 1 N	Μ.	4.0	д ,	- 89	6 C		12	15	16	18	91	21	22	54	5 9 7 7

u1 11 12 12 12 12 12				
1.559 1.708 1.708 1.708 1.508 1.508 1.1.508 1.1.85 1.1.85 1.1.85 1.1.85 1.1.85 1.1.85 1.1.85 1.1.85 1.1.1.85 1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	2.0976- 2.0976- 1.8806- 1.9886- 1.9886-		- a 494	- J.J.D.D
- 7.5256-01 -1.995 E-01 1.1055 E-01 2.525 E-01 -1.777 E-03 -1.777 E-03 -6.649 E-04	X 3.1666-03 1.9256-03 1.7656-05 1.1666-05 1.1666-05			
9.963E-01 3.065E-01 3.065E-01 3.065E-01 5.487E-03 3.487E-03 3.487E-03	8 8 8 9 1 1 1 1 1 1 1 1 1 1 1 1 1	3	55555555555555555555555555555555555555	2 • 84 0 E
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		0 × 9 2 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	74532983649824383	3

					e e un " o un e mener					4.4 Advecting of a summary representation of the second se Second second r>second second sec				14 MILES - 441 MILES - 441-0						Summer - S		• vages • was •						
	u. 250L = c.lorL+U. 250L = c.lorL+U. 1.20L = c.lorL+U. 2.202+u1 Fol = L2.179E+U0 k.2 = 01.94E+U2 k.2 = 0.	1,1172+60 600 = /.3366+41 600 = -3.054644 2,52 <u>36=01 600 = 1.2270+01066</u> = - <u>0.030404</u> -2.0046=02 000 = 1.630e+01 6006 = 1.650e+04		-1.679c+U7 FMSY = 1. <u>-1.679c+U8 FMSY = 1.</u> 9.546c+U8 FMSY = -0.65(c+J0 FN2Z = 2.120 9.546c+U8 FMSY = -0.65(c+J0 FN2Z = 2.120c)		-1.371£+48 FUSY = -1.201£+48 FUSZ = 2.7272+18 -1.371£+48 FUSY = .14281£+448.FUSZ = .24267-348	1.694e+48 FUSY = 0.372e+02 FUSZ = 4.059c+10 -1.602E+48 FUSY = 4.944e+10 FUSZ = -+.62e+10 -2.050200 FUSY = 73333550 FUSZ = -+.62e+10	100111400 1001	LCAUS UISTRIAUTION	17	4 = 612E = 417 3 = 736E = 408	<u>-5+559c+07</u> 4.403E+07	5.892E+07 4.\$72E+07	3a742±+08 385±+06		2.3/2E+U/ 5.363E+D8	1.50467447 3.6246467 2.5546467	1 - 244 - 16 1 - 244 - 16	14.479614-08 11.479614-08		1.60.52±4.05 4.387±407	1.528E+47 3.307E+47	2 <u>4555</u> 408 2.3335407	t.000C.01 b.366E+08	-2.225E+U6 7 034E+06	f.ecterus 9.952E+16	-1.27DE+09 -1.04DE+09	-1.2906+08
STORY CATA	ан предстатии и пре	л л л 90 90 90	KUT LUAES	FMSX = FMSX = FMSX =	RUT LUAUS	FDSX = FESX =	FDSX H NSX H N N	Fusk Fusk	SOLY LANLING		-7.2276+03 -1.2736+08	-2.0846+05 -1.549c+35	2.316E+05 4.51LE+05	1.265E+18. 7.705E+07	3.22(E+U7	3.4366+18	<pre>< 191Etur 1.743E+06 2 0715408</pre>	-1.351E+06	-4.405£408 -1.153£405 - 0.000	2. 6666 + 0 4 4. 8666 + 0 4	4.9686+08	-2.139E+02 -1.716E+06	-3.973E+08	-3.4365+08	-7.698E+07	-3.2246+07	6.454E+05 -3.646E+05	2.0416+04
ITH ENTL	<pre> King (1) King (1)</pre>	4006 = 1 4006 = 2 4006 = 3	MAIN ST	H 10 100	DRAG ST	N	н н н н н н	م بن ۱۱۱۱	CENTER 1	×	1.801E+07 -1.058E+68		4. <u>910E+06</u> 1.873E+07	-1.086E+ u8 -2.922E+06	-1.038E+06	-2.582E+06		-1.000E+08	- <u>5</u> .1346+96 -1.0176+66	-4.657E+07	<u>-3.0846408</u>	~5 .825E+07 -3.516E+07	9-404E+06	1.280E+06	-3.313E+06	-2.0335406	-3 <u>.454</u> E+06 -3.276E+08	-3.562E+07
4 2 2 1 1										LNTOP	40	t (N	3 9	8	6 07	121	7 # P	16	11		22	25	25 25	57	80 G Cl	30	32	55.

ĺ

LANDING LOACS SUMMED AT C.G.	FX = -2.2775+04	FT = +9.8036F4U3 FZ = -2.316E4U4	TX = 9.1885+U5 TY = -2.9575+D9	12 = 2.758L+Ub	

•		PROGRAM CBLL(INPUT,OUTPUT,TAPE5=INPUT,TAPE6=OUTPUT,TAPE3,TAPE4)	CBLL	10
С			CBLL	20
С		CENTER BODY LANDING LOADS PROGRAM	CBLL	30
С		MASTER AGREEMENT, CONTRACT NAS1-8137, TASK ORDER NUMBER FIVE	CBLL	40
C		MCDONNELL DOUGLAS ASTRONAUTICS COMPANY - EAST	CBLL	50
С			CBLL	60
С		THIS PROGRAM COMBINES THE EFFECTS OF THE LANDING GEAR STRUT LOADS	CBLL	70
С		AND INERTIA LOADS AND OUTPUTS THESE FOR THE STRUCTURAL ANALYSIS	CBLL	80
С		PROGRAM	CBLL	90
С			CBLL	100
С		THE FOLLOWING TAPE ARE REQUIRED -	CBLL	110
С		1. TAPE3 - TIME HISTORY INFORMATION FROM THE LANDING LOADS	CBLL	120
С		AND MOTIONS PROGRAM	CBLL	130
C		2. TAPE4 - MODAL DATA FROM THE STRUCTURAL ANALYSIS PROGRAM	CBLL	140
С			CBLL	150
		DIMENSION DUMMY(75),XJT(80),YJT(80),ZJT(80),PHIX(80,30),	CBLL	160
	÷	<pre>* PHIY(80,30),PHIZ(80,30),AMASS(80),XPHI(80,5),</pre>	CBLL	170
	÷	<pre>* YPHI(80,5),ZPHI(80,5),MODE(5),JOTMS(5),JOTDS(10),</pre>	CBLL	180
	÷	<pre>* TYM(20),GC(5),GCD(5),GCDD(5),FMSX(5),FMSY(5),FMSZ(5),</pre>	CBLL	190
	ł	<pre>* FDSX(10),FDSY(10),FDSZ(10),DC(3,3),FX(80),FY(80),FZ(80)</pre>	, CBLL	200
	÷	<pre>* FXJT(80),FYJT(80),FZJT(80),</pre>	CBLL	210
	ł	* SAX(80),SAY(80),SAZ(80)	CBLL	220
С			CBLL	230
С		PROGRAM INITIALIZATION DATA	CBLL	240
С			CBLL	250
		READ(5,500)NOJTS,NOLEG,NOMODE,NOTIME,NOSETS	CBLL	260
		READ(5,500)(MODE(I), I=1, NOMODE)	CBLL	270
		READ(5,500)(JOTMS(I),I=1,NOLEG)	CBLL	280
		NODS=2*NOLEG	CBLL	290
		READ(5,500)(JOTDS(I),I=1,NODS)	CBLL	300
		READ(5,501)ZETA,GRAV	CBLL	310
		READ(5,501)(TYM(I),I=1,NOTIME)	CBLL	320
		READ(5,501)(AMASS(I), I=1, NOJTS)	CBLL	330
c			CBLL	340
č		PRINT PROGRAM INITIALIZATION DATA	CBLL	350
č			CBLL	360
		WRITE(6,600)	CBLL	370
		WRITE(6,601)NOJTS, NOLEG, NOMODE, NOTIME, NOSETS	CBLL	380
		WRITE $(6 \cdot 602)$ (MODE (I) $\cdot I = 1 \cdot NOMODE$)	CBLL	390
		WRITE($6,603$)(JOTMS(I), I=1,NOLEG)	CBLL	400
		WRITE(6,604)(JOTDS(I), I=1, NODS)	CBLL	410
		WRITE(6,605)ZEIA,GRAV	CBLL	420
		WRITE($6 \cdot 607$)(TYM(I) \cdot I=1 \cdot NOTIME)	CBLL	430
		WRITE($6 \cdot 606$)(AMASS(I) $\cdot I = 1 \cdot NOJIS$)	CBLL	440
c			CBLL	450
ĉ		READ MODAL DATA FROM TAPE4	CBLI	460
ĉ			CBLL	470
C		REWIND 4	CBLI	480
			CBLI	490
			CBLL	500
		13=0	CBLI	510
			CBLI	520
		15=0	CBLL	530
	150	CONTINUE	CBLL	540
		READ(4,400)MMM, (DUMMY(I), I=1,5)	CBLL	550

```
155 CONTINUE
                                                                                 CBLL 560
      IF (MMM.EQ.21)GO TO 150
                                                                                 CBLL 570
                                                                                 CBLL 580
      IF (MMM.EQ.121)GO TO 150
                                                                                 CBLL 590
      IF (MMM.EQ.221)GO TO 160
                                                                                 CBLL 600
      IF(MMM.EQ.321)GO TO 165
                                                                                 CBLL 610
      IF (MMM.EQ.421)GO TO 156
                                                                                 CBLL 620
      IF (MMM. EQ. 521) GO TO 150
                                                                                 CBLL 630
      IF (MMM. EQ. 721) GO TO 150
                                                                                 CBLL 640
      WRITE(6,404)
                                                                                 CBLL 650
      STOP
  156 CONTINUE
                                                                                 CBLL 660
                                                                                 CBLL 670
      I3=0
                                                                                 CBLL 680
      I4 = I4 + 5
                                                                                 CBLL 690
      I5 = I5 + 1
                                                                                 CBLL 700
      IF(I5.EQ.NOSETS)GO TO 200
                                                                                 CBLL 710
CBLL 720
      CALL SKFILE(4,1)
      GO TO 150
                                                                                 CBLL 730
С
Ċ
                                                                                 CBLL 740
      JOINT COORDINATES
Ċ
                                                                                 CBLL 750
                                                                                 CBLL 760
  160 CONTINUE
                                                                                 CBLL 770
CBLL 780
CBLL 790
      IF(I5.NE.0)GO TO 150
      I = I + 1
      XJT(I1) = DU(4MY(3))
                                                                                 CBLL 800
      YJT(I1)=DUMMY(1)
                                                                                 CBLL 810
      ZJT(I1) = DUMMY(2)
      GO TO 150
                                                                                 CBLL 820
С
                                                                                 CBLL 830
                                                                                 CBLL 840
С
      MODE SHAPES
C
                                                                                 CBLL 850
                                                                                 CBLL 860
  165 CONTINUE
                                                                                 CBLL 870
      12 = 12 + 1
      GO TO(166,168,170)I2
                                                                                 CBLL 880
                                                                                 CBLL 890
  166 I3=I3+1
                                                                                 CBLL 900
      DO 167 I=1,5
                                                                                 CBLL 910
  167 PHIX(I3,I4+I)=DUMMY(I)
                                                                                 CBLL 920
      GO TO 150
  168 DO 169 I=1,5
                                                                                 CBLL 930
  169 PHIY(I3,I4+I)=DUMMY(I)
                                                                                 CBLL 940
                                                                                 CBLL 950
      GO TO 150
                                                                                 CBLL 960
  170 DO 171 I=1,5
                                                                                 CBLL 970
  171 PHIZ(I3,I4+I)=DUMMY(I)
                                                                                 CBLL 980
      12=0 .
                                                                                 CBLL 990
      GO TO 150
С
                                                                                 CBLL1000
С
      SORT MODE SHAPE DATA
                                                                                 CBLL1010
С
                                                                                 CBLL1020
  200 CONTINUE
                                                                                 CBLL1030
      NNN=5*NOSETS
                                                                                 CBLL1040
      DO 215 I=1,NNN
                                                                                  CBLL1050
                                                                                  CBLL1060
      K = 10
      IF(I.EQ.MODE(1))K=1
                                                                                  CBLL1070
       IF(I.EQ.MODE(2))K=2
                                                                                  CBLL1080
                                                                                  CBLL1090
      IF(I \cdot EQ \cdot MODE(3))K=3
      IF(I.EQ.MODE(4))K=4
                                                                                  CBLL1100
                                                                                  CBLL1110
       IF(I \cdot EQ \cdot MODE(5))K=5
```

ł

```
CBLL1120
      IF(K.EQ.10)GO TO 215
      DO 215 J=1,NOJTS
                                                                            CBLL1130
                                                                            CBLL1140
      XPHI(J,K) = PHIX(J,I)
                                                                            CBLL1150
      YPHI(J,K)=PHIY(J,I)
                                                                            CBLL1160
      ZPHI(J,K)=PHIZ(J,I)
  215 CONTINUE
                                                                            CBLL1170
                                                                            CBLL1180
C
č
c
                                                                            CBLL1190
      PRINT DATA READ FROM TAPE4
                                                                            CBLL1200
                                                                            CBLL1210
      WRITE(6,401)
      DO 220 I=1,NOJTS
                                                                            CBLL1220
  220 WRITE(6,402)I,XJT(I),YJT(I),ZJT(I)
                                                                            CBLL1230
      WRITE(6,403)
                                                                            CBLL1240
                                                                            CBLL1250
      DO 225 J=1,NOMODE
      WRITE(6,407)MODE(J)
                                                                            CBLL1260
                                                                            CBLL1270
      DO 225 I=1,NOJTS
                                                                            CBLL1280
      WRITE(6,402)I,XPHI(I,J),YPHI(I,J),ZPHI(I,J)
                                                                            CBLL1290
  225 CONTINUE
C
C
                                                                            CBLL1300
      DETERMINE C. G. LOCATION
                                                                            CBLL1310
С
                                                                            CBLL1320
                                                                            CBLL1330
      TOTM=0.0
                                                                            CBLL1340
      XTOT=0.0
                                                                            CBLL1350
      YTOT=0.0
                                                                            CBLL1360
      ZTOT=0.0
      DO 230 J=1,NOJTS
                                                                            CBLL1370
                                                                            CBLL1380
      TOTM=TOTM+AMASS(J)
                                                                            CBLL1390
      XTOT=XTOT+AMASS(J)*XJT(J)
                                                                            CBLL1400
      YTOT=YTOT+AMASS(J)*YJT(J)
  230 ZTOT=ZTOT+AMASS(J)*ZJT(J)
                                                                            CBLL1410
      XCG=XTOT/TOTM
                                                                            CBLL1420
                                                                            CBLL1430
      YCG=YTOT/TOTM
      ZCG=ZTOT/TOTM
                                                                            CBLL1440
                                                                            CBLL1450
      DO 235 J=1,NOJTS
      XJT(J) = XJT(J) - XCG
                                                                            CBLL1460
                                                                            CBLL1470
      YJT(J)=YJT(J)-YCG
                                                                            CBLL1480
  235 ZJT(J)=ZJT(J)-ZCG
Ċ
                                                                            CBLL1490
С
      READ TIME HISTORY DATA FROM TAPE3
                                                                            CBLL1500
C
                                                                            CBLL1510
      REWIND 3
                                                                            CBLL1520
      NOREAD=14+9*NOLEG+3*NOMODE
                                                                            CBLL1530
                                                                             CBLL1540
      I_{J}=0
  700 CONTINUE
                                                                             CBLL1550
                                                                             CBLL1560
      IJ=IJ+1
                                                                            CBLL1570
      IF(IJ.GT.NOTIME)STOP
  710 READ(3)(DUMMY(I), I=1, NOREAD)
                                                                             CBLL1580
      IF(ABS(DUMMY(1)-TYM(IJ)).GT.(DUMMY(2)/10.)) GO TO 710
                                                                             CBLL1590
С
                                                                             CBLL1600
С
      TIME POINT HAS BEEN FOUND
                                                                             CBLL1610
С
                                                                             CBLL1620
                                                                             CBLL1630
      TYM(IJ)=DUMMY(1)
                                                                             CBLL1640
      XSDD = DUMMY(3)
      YSDD = DUMMY(4)
                                                                             CBLL1650
                                                                             CBLL1660
      ZSDD = DUMMY(5)
      PHI =DUMMY(6)
                                                                             CBLL1670
```

<pre>WX =DUMMY(10) CEL WZ =DUMMY(11) CEL WZ =DUMMY(12) CEL WTD =DUMMY(12) CEL WTD =DUMMY(13) CEL UZD =DUMMY(14) CEL UZD =DUMMY(14) CEL CEL UI =1+1+3*(1-1) CEL GC(1)=DUMMY(11+1) CEL GC(1)=DUMMY(11+1) CEL GC(1)=DUMMY(11+1) CEL II =14+3*NONODE CEL DO 702 I=1.NOLEG CEL II =11+3*(I-1) CEL FMSX(I)=DUMMY(11+1) CEL FMSX(I)=DUMMY(11+1) CEL FMSX(I)=DUMMY(11+1) CEL FMSX(I)=DUMMY(11+1) CEL FMSX(I)=DUMMY(11+2) CEL DO 703 I=1.NODS CEL II =11+1+3*NONODE 3*NOLEG CEL DO 703 I=1.NODS CEL FDSY(I)=DUMMY(11+1) CEL FDSY(I)=DUMMY(11+2) CEL FDSY(I)=DUM</pre>			THTA =DUMMY(7) PSI =DUMMY(8)	CBL CBL
<pre>WY =DUMMY(10) WZ =DUMMY(11) WZD =DUMMY(12) WZD =DUMMY(12) WZD =DUMMY(13) CEL WZD =DUMMY(14) (I=1=4 DO 701 I=1.NOMODE I =I=1+1+3*(I=1) GC(I)=DUMMY(11+1) CEL GC(I)=DUMMY(11+1) CEL GC(I)=DUMMY(11+2) CEL DC 702 I=1.NOLEG I =I=1+1+3*(I=1) FMSY(I)=DUMMY(11+2) CEL FMSY(I)=DUMMY(11+2) CEL CO 703 I=1.NODS CEL DC 703 I=1.NODS CEL DC 704 I=1.NOMODE CEL WRITE(6+612) WRITE(6+612) WRITE(6+612) WRITE(6+612) WRITE(6+612) WRITE(6+612) WRITE(6+612) WRITE(6+612) CEL CO 705 I=1.NOLEG CEL CEL CEL CEL CEL CEL CEL CEL CEL CEL</pre>			WX =DUMMY(9)	CBL
<pre>WZ =DUMMY(11) WZ =DUMMY(12) WZ =DUMMY(12) WYD =DUMMY(13) CBL WYD =DUMMY(13) CBL WYD =DUMMY(14) CBL U =DUMMY(14) CBL D =DUMMY(14) CBL CBL D =DUMMY(11+1) CBL CGL(1)=DUMMY(11+1) CBL GGL(1)=DUMMY(11+1) CBL GGL(1)=DUMMY(11+1) CBL D =DUMMY(11+1) CBL D =DUMMY(11+1) CBL FMSX(1)=DUMMY(11+1) CBL FMSX(1)=DUMMY(11+1) CBL T =DSX(1)=DUMMY(11+1) CBL T =DSX(1)=DUMMY(11+1) CBL FDSX(1)=DUMMY(11+2) CBL FDSX(1)=FDSX(1)=FDSY(1)=FDSZ(1) CBL FDSX(1)=FDSX(1)=FDSZ(1) CBL FDSX(2)=SINMA+COSPSI CBL FDSX(FDSX(FDSSI CBL FDSX(FDSX(FDSX) CBL FDSX(</pre>			WY =DUMMY(10)	CBL
<pre>WXD =DUMMY(12) WXD =DUMMY(13) WZD =DUMMY(14) (I=14 D0 701 l=1,NOMODE (I=1+1+3*(I=1) GC(I)=DUMMY(11) GC(I)=DUMMY(11) GC(I)=DUMMY(11) GC(I)=DUMMY(11) GC(I)=DUMMY(11) GC(I)=DUMMY(11) GC(I)=DUMMY(11) GC(I)=DUMMY(11) (GC(I)=DUMMY(11)) CC(I)=DUMMY(11) CC(I)=DUMMY(11) CC(I)=DUMMY(11) CC(I)=DUMMY(11) CC(I)=DUMMY(11) GC(I)=DUMMY(I)=DUMMY(I)=DUMMY(I)=DUMMY(I) GC(I)=DUMMY(I)=</pre>			WZ =DUMMY(11)	CBL
<pre>WYD =DUMMY(13) WZD =DUMMY(14) GEL II=14 GCD =DUMMY(14) GEL DO 701 I=1*NOMODE GCD [I]=DUMMY(I1) GEL GC(I)=DUMMY(I1+1) GEL GC(I)=DUMMY(I1+1) GEL GC(I)=DUMMY(I1+2) GEL DO 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 702 I=1*NOLEG GCD 703 I=1*NODS GCD 704 I=1*NONDS GCD 705 I=1*NOLEG GCD 705 I=1*NONDE GCD 70</pre>			WXD =DUMMY(12)	CBL
<pre>W2D = DUMMY(14) II=14 CBL D0 701 l=1,NOMODE (CBL II=11+1+3*(1-1) CBL GC(I)=DUMMY(11+1) CBL GC(I)=DUMMY(11+2) (CBL T1=14+3*NOMODE D0 702 l=1,NOLEG II=11+1+3*(1-1) CBL FMSY(1)=DUMMY(11+1) CBL FMSY(1)=DUMMY(11+2) (CBL D0 703 l=1,NODS (CBL D0 703 l=1,NODS (CBL D0 703 l=1,NODS (CBL T1E(6+614)) CBL WRITE(6+619) CBL WRITE(6+610)TYM(1J)+XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, CBL WRITE(6+619) CBL T04 WRITE(6+611)I,GC(I),GCD(I),GCDD(I) WRITE(6+611)I,GC(I),GCDD(I) CBL T05 VRIIE(6+612)I,FMSY(I),FMSZ(I) CBL T06 WRITE(6+612)I,FMSY(I),FMSZ(I) CBL T06 WRITE(6+612)I,FDSY(I),FMSZ(I) CBL T06 WRITE(6+612)I,FDSY(I),FMSZ(I) CBL CC CALCULATE DIRECTION COSINE MATRIX CBL COSPHI=COS(PHI) CIALCOS(FHI) CBL COSPHI=COS(PHI) CBL COSPHI=SINPHI COSINH*SINPSI CBL CC CALCULATE DIRECTION COSINE MATRIX CBL CC CALCULATE DIRECTION COSINE MATRIX CBL CC CC CC CC CC CC CC CC CC CC CC CC CC</pre>			WYD =DUMMY(13)	CBL
<pre>11 = 14 D0 701 1 = 1,NOMODE CBL I1 = 11 + 13 + (1-1) CBL GC (1) = DUMMY (11) CBL GC (1) = DUMMY (11) CBL I1 = 14 + 3 * NOMODE CBL I1 = 14 + 3 * NOMODE CBL D0 702 1 = 1,NOLEG CBL I1 = 11 + 13 * (1-1) CBL FMSX (1) = DUMMY (11) CBL FMSX (1) = DUMMY (11+1) CBL I1 = 14 + 3 * NOMODE + 3 * NOLEG CBL D0 703 1 = 1,NODS CBL D0 703 1 = 1,NODS CBL I1 = 11 + 1 + 3 * (1-1) CBL FDSX (1) = DUMMY (11) CBL I1 = 11 + 1 + 3 * (1-1) CBL FDSX (1) = DUMMY (11) CBL I1 = 11 + 1 + 3 * (1-1) CBL FDSX (1) = DUMMY (11) CBL I1 = 11 + 1 + 3 * (1-1) CBL FDSX (1) = DUMMY (11) + 2) CBL WR ITE (6,610) TWM (11) + 2) CBL WR ITE (6,610) TWM (11) + 2) CBL WR ITE (6,610) TWM (11) + 2) CBL WR ITE (6,610) TWM (11) + 2) CBL WR ITE (6,610) TWM (11) + 2) CBL WR ITE (6,610) TWM (11) + 2) CBL WR ITE (6,610) TWM (11) + 2) CBL WR ITE (6,610) TWM (11) + 2) CBL WR ITE (6,610) TWM (11) + 2) CBL CBL D0 704 I = 1,NODS CBD (1),GCDD (1) CBL D0 705 I = 1,NOLEG CBL T05 WR ITE (6,612) I + FMSX (1) + FMSZ (1) CBL T06 WR ITE (6,612) I + FMSX (1) + FMSZ (1) CBL T06 WR ITE (6,613) I + FDSX (1) + FMSZ (1) CBL CC CALCULATE DIRECTION COSINE MATRIX CBL COSPH1 = COS(PH1) CBL SINPH = SIN(PH1) CBL COSPH1 = COS(PH1) CBL CALCULATE DIRECTION COSINE MATRIX CBL COSPH1 = COS(PH1) CBL D0 705 I = 1,NOPSI CBL COSPH1 = COSPH1 * SINPSI CBL D0 C(1, 2) = SINPH * A + B COSPH1 * SINPSI CBL D0 C(1, 2) = SINPH * A + B CC (2, 2) = SINTHA * CCOSPH1 * COSPSI CBL DC (2, 2) = SINTHA * CCOSPH1 * COSPSI CBL DC (2, 2) = SINTHA * CCOSPSI CBL DC (2, 2</pre>			WZD =DUMMY(14)	
D0 101111111100000 I1 = 11 + 1 + 31 (1 - 1) GC (1) = DUMMY (11 + 1) GC (1) = DUMMY (11 + 2) I = 1 + 4 + 38 NOMODE D0 702 I = 1, NOLEG D0 702 I = 1, NOLEG I1 = 1 + 1 + 34 (1 - 1) FMSY (1) = DUMMY (11 + 2) CEL I = 1 + 4 + 38 NOMODE + 38 NOLEG D0 703 I = 1, NODS I = 1 + 1 + 34 (1 - 1) FD SX (1) = DUMMY (11 + 1) CEL FD SX (1) = DUMMY (11 + 1) GEL WR ITE (6, 612) WR ITE (6, 612) WR ITE (6, 612) WR ITE (6, 612) TYM (1 + 1) CEL WR ITE (6, 612) TYM (1 + 1) CEL WR ITE (6, 612) TYM (1 + 1), SCDD, YSDD, ZSDD, PHI, THTA, PSI, WX, WY, WZ, WR ITE (6, 612) WR ITE (6, 612) I, SCDD (1) WR ITE (6, 612) I, SCDD (1), SCDD (1) CEL WR ITE (6, 612) I, FMSY (1), FMSY (1), FMSZ (1) WR ITE (6, 612) I, FMSX (1), FMSY (1), FMSZ (1) CEL WR ITE (6, 612) I, FMSX (1), FMSY (1), FMSZ (1) CEL C CALCULATE DIRECTION COSINE MATRIX C C CALCULATE DIRECTION COSINE MATRIX C C CALCULATE DIRECTION COSINE MATRIX C C C CALCULATE DIRECTION COSINE MATRIX C C CALCULATE DIRECTION COSINE MATRIX C C C C C C C C C C C C C C C C C C C			11=14 DO 701 I=1.NONODE	
<pre>GC(1)=DUMMY(11) GC(1)=DUMMY(11+2) GCD(1)=DUMMY(11+2) GCD(1)=DUMMY(11+2) GCD(1)=DUMMY(11+2) GCD(1)=DUMMY(11+1) GCD(1)=DUMMY(11+1) GCD(1)=DUMMY(11+1) GCD(1)=DUMMY(11+1) GCD(1)=DUMMY(11+2) GCD(1)=DUMMY(11+2) GCD(1)=DUMMY(11+2) GCD(1)=DUMMY(11+1) GCD(1)=DUMMY(11+1) GCD(1)=DUMMY(11+1) GCD(1)=DUMMY(11+1) GCD(1)=DUMMY(11+1) GCD(1)=DUMMY(11+2) WRITE(6,614) WRITE(6,614) GCD(1)=DUMMY(1),SCDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, WRITE(6,614) GCD(1)=GCD(1),GCD(1),GCD(1) GCD(1)=GCD(1),GCD(1),GCD(1) GCD(1)=GCD(1),FMSZ(1),FMSZ(1) GCD(1)=GCD(1),FMSZ(1),FMSZ(1) GCD(1)=GCD(1),FMSZ(1),FMSZ(1) GCD(1)=GCD(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1) GCD(1)=GCD(1),GCD(1),GCD(1) GCD(1)=GCD(1),GCD(1),GCD(1) GCD(1)=GCD(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),GCD(1),GCD(1),GCD(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),GCD(1),GCD(1),GCD(1) GCD(1)=GCD(1),GCD(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),GCD(1),GCD(1),GCD(1) GCD(1)=GCD(1),GCD(1),GCD(1),GCD(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),FDSZ(1),FDSZ(1),FDSZ(1) GCD(1)=GCD(1),GCD(1),GCD(1),GCD(1),GCD(1) GCD(1)=GCD(1),G</pre>			1 = 1 + 1 + 2 + (1 - 1)	CBL
CCONTRACTOR AND A CONTRACT AND A CON			GC(I) = D(IMMY(I))	CBL
701 GCDD(1)=DUMMY(11+2) GEL II=14+3*NOMODE GEL D0 702 I=1.NOLEG GEL II=11+13*(I-1) GEL FMSY(I)=DUMMY(11) GEL FMSY(I)=DUMMY(11+1) GEL T01 GCD0(1)=DUMMY(11+2) GEL II=14+3*NOMODE+3*NOLEG GEL D0 703 I=1,NODS GEL II=11+13*(I-1) GEL FDSY(I)=DUMMY(11+2) GEL FDSY(I)=DUMMY(11+2) GEL WRITE(6+619) GEL WRITE(6+610)TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, GEL WRITE(6+610)TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, GEL WRITE(6+610)TYM(IJ),SCDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, GEL WRITE(6+610)TYM(IJ),SCDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, GEL WRITE(6+610)TYM(IJ),SCDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, GEL T04 WRITE(6+610)TYM(IJ),FMSY(I),FMS2(I) GEL T05 I=1,NOLEG GEL T06 WRITE(6+616) GEL T176 WRITE(6+616) GEL T076 WRITE(6+616) GEL T08 WRITE(6+613)I,FDSY(I),FDSY(I),FDSZ(I) GEL C CALCULATE DIRECTION COSINE MATRIX <t< td=""><td></td><td></td><td>GCD(I)=D(IMMY(I)+1)</td><td>CBL</td></t<>			GCD(I)=D(IMMY(I)+1)	CBL
II=14+3#NOMODE CBL D0 702 I=1,NOLEG CBL II=11+143#(I-1) CBL FMSX(I)=DUMMY(I1) CBL FMSX(I)=DUMMY(I1+1) CBL 702 FMS2(I)=DUMMY(I1+2) CBL II=14+3*NOMDE+3*NOLEG CBL D0 703 I=1,NODS CBL I1=11+13*X(I-1) CBL FDSY(I)=DUMMY(I1+2) CBL FDSY(I)=DUMMY(I1+1) CBL 703 FDS2(I)=DUMMY(I1+2) CBL wRITE(6+619) CBL wRITE(6+619) CBL wRITE(6+610)TYM(I),XSDD,YSDD,ZSDD,PHI,THTA+PSI,WX,WY,WZ, CBL * WXD,WYD,WZD CBL D0 704 I=1,NOMDE CBL YMITE(6+610)TYM(I),SCDD(I),GCDD(I) CBL WRITE(6+611)I+GC(I),FDSY(I),FDSZ(I) CBL YMITE(6+616) CBL D0 706 I=1,NOLEG CBL YO6 WRITE(6+616) CBL D0 706 I=1,NOLEG CBL YMITE(6+616) CBL D0 706 I=1,NOLEG CBL YMITE(6+616) CBL C CALCULATE DIRECTION COSINE MATRIX C		701	GCDD(I) = DUMMY(II+2)	CBL
DO 702 I=1,NOLEG CBL I1=II+13*(I-1) CBL FMSX(I)=DUMMY(I1) CBL FMSY(I)=DUMMY(I1+1) CBL 702 FMSY(I)=DUMMY(I1+2) CBL I1=14+3*NOMODE+3*NOLEG CBL DO 703 I=1,NODS CBL FDSX(I)=DUMMY(I1) CBL FDSX(I)=DUMMY(I1+1) CBL 703 FDSZ(I)=DUMMY(I1+2) CBL WRITE(6,619) CBL WRITE(6,610) TVM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, CBL WRITE(6,610) TVM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, CBL WRITE(6,610) TVM(IJ),SDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, CBL WRITE(6,611) 1,GC(I),GCD(I),GCDD(I) CBL DO 704 I=1,NOMODE CBL 705 WRITE(6,612) I,FMSX(I),FMSY(I),FMSZ(I) CBL 00 705 I=1,NOLEG CBL 706 WRITE(6,613) I,FDSX(I),FMSY(I),FMSZ(I) CBL CC CALCULATE DIRECTION COSINE MATRIX CBB C CALCULATE DIRECTION COSINE MATRIX CBB C COSPHI=CS(PHI) CBINE MATRIX CBB C CALCULATE DIRECTION COSINE MATRIX CBB C COSPHI=CS(PHI) CBINE MATRIX CBB C CALCULATE DIRECTION COSINE MATRIX CBB C COSPHI=CS(PHI) CBINE MATRIX CBB C COSPHI=SINPSI CBI C CALCULATE DIRECTION COSINE MATRIX CBB C COSPHI=SINPSI CBI C CALCULATE DIRECTION COSINE MATRIX CBB C COSPHI=SINPSI CBI C C CALCULATE DIRECTION COSINE MATRIX CBB C COSPHI=SINPSI CBI C C CALCULATE DIRECTION COSINE MATRIX CBB C C CALCULATE DIRECTION COSINE MATRIX CBB C C C C CALCULATE DIRECTION COSINE MATRIX CBB C C C C C C CALCULATE DIRECTION COSINE MATRIX CBB C C C C C C C C C C C C C C C C C C C			II=14+3*NOMODE	CBL
<pre>I1=II-I+3*(I-1) FMSX(1)=DUMMY(II) FMSY(I)=DUMMY(II+1) CBL FMSY(I)=DUMMY(II+1) CBL T02 FMSZ(I)=DUMMY(II+1) CBL D0 703 I=1,NODS I1=I++3*(I-1) CBL FDSY(I)=DUMMY(II) FDSY(I)=DUMMY(II) FDSY(I)=DUMMY(II+1) CBL WRITE(6+619) WRITE(6+619) WRITE(6+619) WRITE(6+614) WRITE(6+619) CBL T04 WRITE(6+610)TYM(IJ)+XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, BD D 704 I=1,NOMODE CBL T04 WRITE(6+610)TYM(IJ)+XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, BD T05 WRITE(6+611)I+GC(I)+GCD(I) CBL T05 WRITE(6+612)I+FMSY(I)+FMSY(I)+FMSZ(I) CC CC CC CCCCCCCCCCCCCCCCCCCCCCCCCCC</pre>			DO 702 I=1,NOLEG	CBL
<pre>FMSX(1)=DUMMY(I1) FMSY(1)=DUMMY(I1+1) CBL FMSY(1)=DUMMY(I1+1) CBL TO2 FMSZ(1)=DUMMY(I1+2) CBL TI=1++3*NOMODE+3*NOLEG CTOT TO TO TO TO TO TO TO TO TO TO TO TO</pre>			I 1 = I I + 1 + 3 * (I - 1)	CBL
<pre>FMSY(1)=DUMMY(11+1) 702 FMSZ(1)=DUMMY(11+2) 11=14+3*NOMDE4-3*NOLEG CRL 11=14+3*NOMODE4-3*NOLEG C703 I=1,NODS CRL PDSY(1)=DUMMY(11) FDSY(1)=DUMMY(11) FDSY(1)=DUMMY(11)+1) CRL FDSY(1)=DUMMY(11+2) CRL WRITE(6,619) WRITE(6,619) WRITE(6,610) TYM(1J)*XSDD*YSDD*ZSDD*PHI*THTA*PSI*WX*WY*WZ* CRL WRITE(6,610) TYM(1J)*XSDD*YSDD*ZSDD*PHI*THTA*PSI*WX*WY*WZ* CRL D704 WRITE(6,610) TYM(1J)*CSDD*PSDD*ZSDD*PHI*THTA*PSI*WX*WY*WZ* CRL T04 WRITE(6,610) TYM(1J)*CSDD*PSDD*ZSDD*PHI*THTA*PSI*WX*WY*WZ* CRL T05 WRITE(6,611)*GCCD(1)*GCDD(1) CRL T05 URITE(6,612)*FMSX(1)*FMSY(1)*FMSZ(1) CRL T06 WRITE(6,613)*FMSX(1)*FMSY(1)*FMSZ(1) CRL CCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCCC</pre>			FMSX(I)=DUMMY(I1)	CBL
702 FMSZ(1)=DUMMY(11+2) CRL II=14+3*NOMODE3*NOLEG CRL D0 703 I=1,NODS II=II+1+3*(I-1) CRL FDSX(1)=DUMMY(I1) CRL FDSY(I)=DUMMY(I1) CRL FDSY(I)=DUMMY(I1+1) CRL T03 FDSZ(I)=DUMMY(I1+2) CRL WRITE(6,619) CRL WRITE(6,610)TYM(IJ)*XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX*WY*WZ, CRL WNTTE(6,610)TYM(IJ)*XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX*WY*WZ, CRL WRITE(6,610)TYM(IJ)*XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX*WY*WZ, CRL WRITE(6,610)TYM(IJ)*XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX*WY*WZ, CRL D0 704 I=1,NOMODE CRL D0 704 I=1,NONDE CRL D0 705 I=1,NOLEG CRL D0 705 I=1,NOLEG CRL C0 CALCULATE DIRECTION COSINE MATRIX CRL C1 CALCULATE DIRECTION COSINE MATRIX			FMSY(I)=DUMMY(I1+1)	CBL
<pre>II=14+3*NOMODE+3*NOLEG CB DO 703 I=1,NODS CB II=II+1+3*(I=1) CB FDSX(I)=DUMMY(I1) CBL FDSX(I)=DUMMY(I1+1) CBL WRITE(6,619) CBL WRITE(6,619) CBL WRITE(6,610) TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, CBL WRITE(6,610) TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, CBL WRITE(6,610) IYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, CBL WRITE(6,610) IYM(IJ),SCDD(I),GCDD(I) CBL DO 704 I=1,NOMODE CBL 704 WRITE(6,611) IsGC(I),GCDD(I),GCDD(I) CBL DO 705 I=1,NOLEG CBL 705 WRITE(6,612) I,FMSX(I),FMSY(I),FMSZ(I) CBL DO 706 I=1,NODS CBL 706 WRITE(6,613) I,FDSX(I),FDSY(I),FDSZ(I) CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C COSPHI=COS(PHI) CBL SINPHI=SIN(PHI) CBL SINPHI=SIN(PHI) CBL SINPHI=SIN(PHI) CBL SINPHI=SIN(PHI) CBL SINPHI=SIN(PHI) CBL SINPHI=SIN(PHI) CBL SINPHI=SIN(PHI) CBL DC(1,1)=COSTHA*COSPHI CBL DC(1,2)=SINPHI*A+B C CSHA+COSCHI*A+CC DC(2,2)=SINTHA*C+COSPHI*COSPSI CBL DC(2,2)=SINTHA*C+COSPHI*COSPSI CBL</pre>		702	FMSZ(I)=DUMMY(I1+2)	CBL
D0 703 1=1,NODS I1=I1+13*(I=1) FDSX(I)=DUMMY(I1) FDSY(I)=DUMMY(I1+1) 703 FDSX(I)=DUMMY(I1+1) 703 FDSX(I)=DUMMY(I1+2) WRITE(6,619) WRITE(6,619) WRITE(6,614) WRITE(6,610) TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, * WXD,WYD,WZD D0 704 I=1,NOMODE 704 WRITE(6,611)I,GC(I),GCD(I),GCDD(I) WRITE(6,620) D0 705 I=1,NOLEG 705 WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) C C C CALCULATE DIRECTION COSINE MATRIX C C CALCULATE DIRECTION COSINE MATRIX C C CALCULATE DIRECTION COSINE MATRIX C C CALCULATE DIRECTION COSINE MATRIX C C COSPHI=COS(PH1) SINPHI=SIN(PH1) C CALCULATE DIRECTION COSINE MATRIX C C COSPHI=COS(PH1) SINTHA=SIN(ITHA) C CALCULATE DIRECTION COSINE MATRIX C C CALCULATE DIRECTION COSINE MATRIX C C C C CALCULATE DIRECTION COSINE MATRIX C C C C C C C C C C C C C C C C C C C			II=14+3*NOMODE+3*NOLEG	CBL
11=11+1+3*(1=1) CBL FDSX(1)=DUMMY(11) CBL FDSY(1)=DUMMY(11+1) CBL 703 FDS2(1)=DUMMY(11+2) CBL WRITE(6+619) CBL WRITE(6+614) CBL WRITE(6+610) CBL WRITE(6+611) CBL D0 704 11=1,NOMODE CBL 704 WRITE(6+611) D0 705 WRITE(6+611) GC(1)+GCD(1)+GCD(1) WRITE(6+620) CBL D0 705 WRITE(6+612) FMSX(1)+FMSX(1)+FMSZ(1) WRITE(6+612) GBL D0 705 WRITE(6+613) I+FDSX(1)+FMSZ(1) WRITE(6+613) I+FDSX(1)+FMSZ(1) C CALCULATE C CALCULATE C CALCULATE C CALCULATE SINFHA=SIN(PH1) CBL COSPHI=COS(PH1) CBL COSPHI=COS(FSI) CBL SINFHA=SIN(PSI) CBL COSTHA=COS(THTA) CBL SINFHA=SIN(PSI) CBL			DO 703 I=1,NODS	CBL
FDSX(I)=DUMMY(II+1) CBL 703 FDS2(I)=DUMMY(II+1) CBL WRITE(6,619) CBL WRITE(6,614) CBL WRITE(6,614) CBL WRITE(6,614) CBL WRITE(6,614) CBL WRITE(6,614) CBL WRITE(6,614) CBL D0 704 I=1,NOMODE CBL 704 WRITE(6,620) CBL D0 705 I=1,NOLEG CBL 705 WRITE(6,612)I;FMSX(I),FMSY(I),FMSZ(I) CBL D0 706 I=1,NODS CGL 706 WRITE(6,613)I;FDSX(I),FDSY(I),FDSZ(I) CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C COSPHI=COS(PHI) CBL SINPHI=SIN(PHI) CBL CBL COSPHI=COS(PHI) CBL CBL DC(1,1)=COSTHA*COSPHI CBL CBL DC(1,2)=SINPHI*APSI CBL CBL DC(1,2)=SINPHI*APSI CBL CBL DC(1,2)=SINPHI*A+C CBL CBL DC(1,2)=SINPHA*APSI CBL CBL DC			1 = 1 + 1 + 3 + (1 - 1)	
TO3 FD32(1)=DUMMY(11+2) CBL WRITE(6,619) CBL WRITE(6,614) CBL WRITE(6,610) TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, CBL * WXD,WYD,WZD CBL D0 704 I=1,NOMODE CBL 705 WRITE(6,611)I,GC(I),GCD(I),GCDD(I) CBL WRITE(6,620) CBL D0 705 I=1,NOLEG CBL 706 WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) CBL WRITE(6,616) CBL D0 706 I=1,NODS CBL 706 WRITE(6,613)I,FDSX(I),FDSY(I),FDSZ(I) CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C COSPHI=COS(PHI) CBL SINPHI=SIN(PHI) CBL CBL COSPSI=COS(PFI) CBL CBL SINPH=SIN(PSI) CBL CBL DC(1,1)=COSTHA*COSPHI CBL CBL DC(1,2)=SINPHI*A=B CBL CBL C(1,3)=COSPHI*A+C CBL CBL DC(1,2)=SINPHI*A=B CBL CBL DC(1,2)=SINPHA*A=B CBL CBL				
WRITE(6,619) CBL WRITE(6,619) CBL WRITE(6,610) CBL WRITE(6,610) CBL WRITE(6,611) CBL D0 704 I=1,NOMODE CBL 704 WRITE(6,611) GCD(I),GCD(I) WRITE(6,612) CBL WRITE(6,612) CBL 0 705 I=1,NOLEG CBL 705 WRITE(6,612) CBL WRITE(6,613) CBL 0 706 I=1,NODS CBL 706 WRITE(6,613) CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C COSPHI=COS(PHI) CBL SINPHI=SIN(PHI) CBL CBL COSPSI=COS(PHI) CBL CBL SINPHI=SIN(PSI) CBL CBL DC(1,1)=COSTHA*COSPHI CBL CBL DC(1,2)=SINPHI*A=B CBL CBL C CSINPHI*SINPSI CBL DC(1,2)=SINPHI*A=B CBL CBL DC(1,3)=COSPHI*A+C CBL CBL DC(1,2)=SINTHA*C+COSPHI*COSPSI CBL		703	FDST(I) = DUMMY(I) + 2)	CBL
<pre>WRITE(6,614) // CBL WRITE(6,610) TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ,</pre>		105	WRITE(6.619)	CBL
<pre>WRITE(6,610)TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, WRITE(6,610)TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ, CBL D0 704 I=1,N0MODE 704 WRITE(6,611)I,GC(I),GCD(I),GCDD(I) WRITE(6,620) D0 705 I=1,N0LEG 705 WRITE(6,612)I,FMSY(I),FMSY(I),FMSZ(I) WRITE(6,613)I,FDSX(I),FMSY(I),FMSZ(I) C C C CALCULATE DIRECTION COSINE MATRIX C C CALCULATE DIRECTION COSINE MATRIX C C C C CALCULATE DIRECTION COSINE MATRIX C C C C CALCULATE DIRECTION COSINE MATRIX C C C C C C C C C C C C C C C C C C C</pre>			WRITE(6.614)	CBL
<pre>* WXD,WYD,WZD CBL D0 704 I=1,NOMODE CBL T04 WRITE(6,611)1,GCD(I),GCD(I),GCDD(I) WRITE(6,620) CBL WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) CBL T05 WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) CBL T06 WRITE(6,613)I,FDSX(I),FDSY(I),FDSZ(I) CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C C C CALCULATE DIRECTION COSINE MATRIX CBL C C C C CALCULATE DIRECTION COSINE MATRIX CBL C C C C C C C C C C C C C C C C C C C</pre>			WRITE(6,610)TYM(IJ),XSDD,YSDD,ZSDD,PHI,THTA,PSI,WX,WY,WZ,	CBL
DO 704 I=1,NOMODE CBL 704 WRITE(6,611)I,GCCD(I),GCDD(I) CBL WRITE(6,620) CBL WRITE(6,620) CBL DO 705 I=1,NOLEG CBL 705 WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) CBL WRITE(6,613)I,FDSX(I),FDSY(I),FDSZ(I) CBL CC CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C COSPHI=COS(PHI) CBL COSPHI=COS(PHI) CBL COSPHI=COS(PHI) CBL COSPSI=COS(PSI) CBL COSPSI=COS(PSI) CBL COSPSI=COS(PSI) CBL COSPSI=COS(PSI) CBL COSPSI=COS(PSI CBL COSPHI*SIN(PSI) CBL DC(1,2)=SIN(PSI CBL CC(1,3)=COSPHI*A+C DC(2,2)=SINTHA*COSPHI*COSPSI CBL DC(2,2)=SINTHA*E-SINPHI*COSPSI CBL DC(2,2)=SINTHA*E-SINPHI*COSPSI CBL		÷	₩ WXD,WYD,WZD	CBL
704 WRITE(6,611)I,GC(I),GCD(I),GCD(I) CBL WRITE(6,620) CBL D0 705 I=1,NOLEG CBL 705 WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) CBL WRITE(6,613)I,FDSX(I),FMSY(I),FMSZ(I) CBL D0 706 I=1,NODS 706 WRITE(6,613)I,FDSX(I),FDSY(I),FDSZ(I) CBL C CALCULATE DIRECTION COSINE MATRIX CBL COSPHI=COS(PHI) CBL CBL COSPHI=COS(PHI) COSTHA=COS(FNI) CBL CBL COSPSI CBL DC(1,1)=COSTHA*COSPHI CBL CBL			DO 704 I=1,NOMODE	· CBL
WRITE(6,620) CBL DO 705 I=1,NOLEG CBL 705 WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) CBL WRITE(6,616) CBL DO 706 I=1,NODS CBL 706 WRITE(6,613)I,FDSX(I),FDSY(I),FDSZ(I) CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C COSPHI=COS(PHI) CBL SINPHI=SIN(PHI) CBL CBL COSPHI=COS(THTA) CBL CBL SINPHI=SIN(PHI) CBL CBL COSPSI=COS(PSI) CBL CBL SINPA=SIN(THTA) CBL CBL COSPSI=COS(PSI) CBL CBL SINPSI=SIN(PSI) CBL CBL DC(1,1)=COSTHA*COSPHI CBL CBL DC(1,2)=SINPHI*A-B CBL CBL C=SINPHI*SINPSI CBL CBL DC(1,3)=COSPHI*A+C CBL CBL DC(2,2)=SINTHA*C+COSPHI*COSPSI CBL CBL DC(2,2)=SINTHA*C+COSPSI CBL CBL DC(2,2)=SINTHA*B CBL CBL		704	WRITE(6,611)I,GC(I),GCD(I),GCDD(I)	CBL
D0 705 I=1,NOLEG CBL 705 WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) CBL WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) CBL D0 706 I=1,NODS CBL 706 WRITE(6,613)I,FDSX(I),FDSY(I),FDSZ(I) CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CASINPALACOS(PHI) CBL CBL C(1,1)=COSTHA*COSPHI			WRITE(6,620)	CBL
705 WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I) CBL WRITE(6,616) CBL D0 706 IEI,NODS 706 WRITE(6,613)I,FDSX(I),FDSY(I),FDSZ(I) CBL C CALCULATE DIRECTION COSINE MATRIX CBL C CALCULATE DIRECTION COSINE MATRIX CBL C COSPHI=COS(PHI) CBL SINPHI=SIN(PHI) CBL CBL COSTHA=COS(THTA) CBL CBL COSPSI=COS(PSI) CBL CBL SINPSI=SIN(PSI) CBL CBL COSPHI=COS(PHIA CBL CBL COSPSI=COS(PSI) CBL CBL SINPSI=SIN(PSI) CBL CBL DC(1,1)=COSTHA*COSPHI CBL CBL A=SINTHA*COSPSI CBL CBL DC(1,2)=SINPHI*A-B CBL CBL C(1,2)=SINPHI*A-B CBL CBL DC(1,2)=SINTHA*C+COSPHI*COSPSI CBL CBL DC(2,2)=SINTHA*SINPSI CBL CBL DC(2,2)=SINTHA*B-SINPHI*COSPSI CBL CBL DC(2,3)=SINTHA*B-SINPHI*COSPSI CBL CBL			DO 705 I=1,NOLEG	CBL
WRITE(6,616) CBL DO 706 I=1,NODS CBL 706 WRITE(6,613)I,FDSX(I),FDSY(I),FDSZ(I) CBL C CBL C CALCULATE DIRECTION COSINE MATRIX CBL COSPHI=COS(PHI) CBL CBL COSPSI=COS(PSI) CBL CBL COSPSIECOS(PSI) DC(1,1)=COSTHA*COSPHI CBL CBL CCL DC(1,2)=SINPHI*A+C CBL CBL CCL DC(1,2)=SINPHI*A+C CBL CBL CCL DC(1,2)=SINPHA*SINPSI CBL		705	WRITE(6,612)I,FMSX(I),FMSY(I),FMSZ(I)	CBL
DO706T=1;NODSCBL706WRITE(6;613)I;FDSX(I);FDSY(I);FDSZ(I)CBLCCALCULATE DIRECTION COSINE MATRIXCBLCCOSPHI=COS(PHI)CBLCCOSPHI=SIN(PHI)CBLCOSTHA=COS(THTA)CBLCOSTHA=COS(THTA)CBLCOSPSI=COS(PSI)CBLCOSPSI=COS(PSI)CBLCOSPSI=COS(PSI)CBLCOSPSI=SIN(PSI)CBLDC(1,1)=COSTHA*COSPHICBLA=SINTHA*COSPSICBLDC(1,2)=SINPHI*A-BCBLC=SINPHI*SINPSICBLDC(1,3)=COSPHI*A+CCBLDC(2,1)=COSTHA*SINPSICBLDC(2,2)=SINTHA*C+COSPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBL			WRITE(6,616)	CBL
ConstructionConstructionConstructionConstructionCCALCULATE DIRECTION COSINE MATRIXCBLCCOSPHI=COS(PHI)CBLCCOSPHI=SIN(PHI)CBLCOSTHA=COS(THTA)CBLCOSTHA=COS(THTA)CBLCOSPSI=COS(PSI)CBLCOSPSI=COS(PSI)CBLCOSPSI=COS(PSI)CBLCOSPSI=COS(PSI)CBLCOSPSI=COS(PSI)CBLCOSPSI=SIN(PSI)CBLCOSPHI=SIN(PSI)CBLC(1,1)=COSTHA*COSPHICBLDC(1,2)=SINPHI*A-BCBLC=SINPHI*SINPSICBLDC(1,2)=SINPHI*A-BCBLDC(1,3)=COSPHI*A+CCBLDC(2,1)=COSTHA*SINPSICBLDC(2,2)=SINTHA*C+COSPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*C+COSPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSICBLCC2,2)=SINTHA*B-SINPHI*COSPSIC		701	DU 706 1=19NUDS WRITE (4.412NI, EDSY(I), EDSY(I), EDS7(I)	CBL
C CALCULATE DIRECTION COSINE MATRIX CBL COSPHI=COS(PHI) CBL COSPHI=SIN(PHI) CBL SINPHI=SIN(PHI) CBL COSTHA=COS(THTA) CBL COSTHA=SIN(THTA) CBL COSPSI=COS(PSI) CBL SINPSI=SIN(PSI) CBL DC(1,1)=COSTHA*COSPHI CBL A=SINTHA*COSPSI CBL DC(1,2)=SINPHI*A=B CBL C=SINPHI*SINPSI CBL C=SINPHI*SINPSI CBL C=SINPHI*A=SINPSI CBL DC(2,1)=COSTHA*SINPSI CBL DC(2,3)=SINTHA*C+COSPHI*COSPSI CBL	c	100	WRITE(0)015/19FU3X(1/9FU3)(1/9FU32(1/	CBL
CCBLCOSPHI=COS(PHI)CBLSINPHI=SIN(PHI)CBLCOSTHA=COS(THTA)CBLCOSPSI=COS(PSI)CBLCOSPSI=COS(PSI)CBLCOSPSI=COS(PSI)CBLDC(1,1)=COSTHA*COSPHICBLB=COSPHI*SINPSICBLDC(1,2)=SINPHI*A-BCBLC=SINPHI*SINPSICBLDC(1,3)=COSPHI*A+CCBLDC(2,1)=COSTHA*SINPSICBLDC(2,2)=SINTHA*C+COSPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*C+CCBLCBLCC2,3)=SINTHA*C+CCBLCBLCC2,3)=SINTHA*B-SINPHI*COSPSICBLCBLCC2,3)=SINTHA*C+CCBLCBLCC2,3)=SINTHA*C+CCBLCBLCC2,3)=SINTHA*C+CCC2,3)CBLCC2,3)CC2,3)CBL <t< td=""><td>c</td><td></td><td>CALCULATE DIRECTION COSINE MATRIX</td><td>CBL</td></t<>	c		CALCULATE DIRECTION COSINE MATRIX	CBL
COSPHI=COS(PHI) CBL SINPHI=SIN(PHI) CBL COSTHA=COS(THTA) CBL SINTHA=SIN(THTA) CBL COSPSI=COS(PSI) CBL COSPSI=COS(PSI) CBL SINPSI=SIN(PSI) CBL DC(1,1)=COSTHA*COSPHI CBL A=SINTHA*COSPSI CBL DC(1,2)=SINPHI*A-B CBL C=SINPHI*SINPSI CBL DC(1,3)=COSPHI*A+C CBL DC(2,1)=COSTHA*SINPSI CBL DC(2,3)=SINTHA*C+COSPHI*COSPSI CBL DC(2,3)=SINTHA*B-SINPHI*COSPSI CBL	ĉ		CALCOLATE DIRECTION COSTNE MATRIX	CBL
SINPHISSIN(PHI) CBL COSTHA=COS(THTA) CBL SINTHA=SIN(THTA) CBL COSPSI=COS(PSI) CBL SINPSI=SIN(PSI) CBL DC(1,1)=COSTHA*COSPHI CBL A=SINTHA*COSPSI CBL B=COSPHI*SINPSI CBL DC(1,2)=SINPHI*A-B CBL C=SINPHI*SINPSI CBL DC(1,3)=COSPHI*A+C CBL DC(2,1)=COSTHA*SINPSI CBL DC(2,3)=SINTHA*C+COSPHI*COSPSI CBL DC(2,3)=SINTHA*B-SINPHI*COSPSI CBL	C		COSPHI=COS(PHI)	CBL
COSTHA=COS(THTA)CBLSINTHA=SIN(THTA)CBLCOSPSI=COS(PSI)CBLCOSPSI=SIN(PSI)CBLDC(1,1)=COSTHA*COSPHICBLA=SINTHA*COSPSICBLB=COSPHI*SINPSICBLDC(1,2)=SINPHI*A=BCBLC=SINPHI*SINPSICBLDC(1,3)=COSPHI*A+CCBLDC(2,1)=COSTHA*SINPSICBLDC(2,2)=SINTHA*C+COSPHI*COSPSICBLDC(2,3)=SINTHA*B=SINPHI*COSPSICBLDC(2,3)=SINTHA*B=SINPHI*COSPSICBLDC(2,3)=SINTHA*B=SINPHI*COSPSICBLDC(2,3)=SINTHA*B=SINPHI*COSPSICBLCBLC(2,3)=SINTHA*B=SINPHI*COSPSICBLCBLC(2,3)=SINTHA*B=SINPHI*COSPSICBLCBLC(2,3)=SINTHA*B=SINPHI*COSPSICBLCBLC(2,3)=SINTHA*B=SINPHI*COSPSICBLCBLC(2,3)=SINTHA*B=SINPHI*COSPSICBLCBLC(2,3)=SINTHA*B=SINPHI*COSPSICBLCBLC(2,3)=SINTHA*B=SINPHI*COSPSICBLCBLC(2,3)=SINTHA*B=SINPHI*COSPSICBLCBLC(2,3)=SINTHA*B=SINPHI*COSPSICBL			SINPHI=SIN(PHI)	CBL
SINTHA=SIN(THTA)CBLCOSPSI=COS(PSI)CRLSINPSI=SIN(PSI)CBLDC(1,1)=COSTHA*COSPHICBLA=SINTHA*COSPSICBLB=COSPHI*SINPSICBLDC(1,2)=SINPHI*A-BCBLC=SINPHI*SINPSICBLDC(1,3)=COSPHI*A+CCBLDC(2,1)=COSTHA*SINPSICBLDC(2,2)=SINTHA*C+COSPHI*COSPSICBLDC(2,3)=SINTHA*B-SINPACBLDC(2,3)=S			COSTHA=COS(THTA)	CBL
COSPSI=COS(PSI) CRL SINPSI=SIN(PSI) CBL DC(1,1)=COSTHA*COSPHI CBL A=SINTHA*COSPSI CBL B=COSPHI*SINPSI CBL DC(1,2)=SINPHI*A-B CBL C=SINPHI*SINPSI CBL DC(1,3)=COSPHI*A+C CBL DC(2,1)=COSTHA*SINPSI CBL DC(2,2)=SINTHA*C+COSPHI*COSPSI CBL DC(2,3)=SINTHA*B=SINPHI*COSPSI CBL DC(2,3)=SINTHA*B=SINPHI*COSPSI CBL			SINTHA=SIN(THTA)	CBL
SINPSI=SIN(PSI) CBI DC(1,1)=COSTHA*COSPHI CBI A=SINTHA*COSPSI CBI B=COSPHI*SINPSI CBI DC(1,2)=SINPHI*A-B CBI C=SINPHI*SINPSI CBI DC(1,3)=COSPHI*A+C CBI DC(2,1)=COSTHA*SINPSI CBI DC(2,2)=SINTHA*C+COSPHI*COSPSI CBI DC(2,3)=SINTHA*B=SINPHI*COSPSI CBI DC(2,3)=SINTHA*B=SINPHI*COSPSI CBI			COSPSI=COS(PSI)	CBL
DC(1,1)=COSTHA*COSPHI CBI A=SINTHA*COSPSI CBI B=COSPHI*SINPSI CBI DC(1,2)=SINPHI*A-B CBI C=SINPHI*SINPSI CBI DC(1,3)=COSPHI*A+C CBI DC(2,1)=COSTHA*SINPSI CBI DC(2,2)=SINTHA*C+COSPHI*COSPSI CBI DC(2,3)=SINTHA*B-SINPHI*COSPSI CBI			SINPSI=SIN(PSI)	CBI
A=SINTHA*COSPSICBIB=COSPHI*SINPSICBIDC(1,2)=SINPHI*A=BCBIC=SINPHI*SINPSICBIDC(1,3)=COSPHI*A+CCBIDC(2,1)=COSTHA*SINPSICBIDC(2,2)=SINTHA*C+COSPHI*COSPSICBIDC(2,3)=SINTHA*B=SINPHI*COSPSICBIDC(2,3)=SINTHA*B=SINPHI*COSPSICBIDC(2,3)=SINTHA*B=SINPHI*COSPSICBI			DC(1,1)=COSTHA*COSPHI	CBL
B=COSPHI*SINPSICBIDC(1,2)=SINPHI*A=BCBIC=SINPHI*SINPSICBIDC(1,3)=COSPHI*A+CCBIDC(2,1)=COSTHA*SINPSICBIDC(2,2)=SINTHA*C+COSPHI*COSPSICBIDC(2,3)=SINTHA*B=SINPHI*COSPSICBIDC(2,3)=SINTHA*B=SINPHI*COSPSICBI			A=SINTHA*COSPSI	CBI
DC(1;2)=SINPHI*A=BCBIC=SINPHI*SINPSICBIDC(1;3)=COSPHI*A+CCBIDC(2;1)=COSTHA*SINPSICBIDC(2;2)=SINTHA*C+COSPHI*COSPSICBIDC(2;3)=SINTHA*B=SINPHI*COSPSICBIDC(2;3)=SINTHA*B=SINPHI*COSPSICBI				CBI
DC(1,3)=COSPHI*A+CCBIDC(2,1)=COSTHA*SINPSICBIDC(2,2)=SINTHA*C+COSPHI*COSPSICBIDC(2,3)=SINTHA*B-SINPHI*COSPSICBI			UC(1)Z)=SINPHI*A=D C=CINDHI*CINDCI	
DC(2,1)=COSTHA*SINPSI CBI DC(2,2)=SINTHA*C+COSPHI*COSPSI CBI DC(2,3)=SINTHA*B-SINPHI*COSPSI CBI			C = 0 INFIT = 0 SINE SI	
DC(2,2)=SINTHA*C+COSPHI*COSPSI CBI DC(2,3)=SINTHA*B-SINPHI*COSPSI CBI			DC(2.1)=COSTHA*SINPSI	CBI
DC(2,3)=SINTHA*B-SINPHI*COSPSI CB			DC(2.2)=SINTHA*C+COSPHI*COSPSI	CBI
			DC(2,3)=SINTHA*B-SINPHI*COSPSI	CBI

L1680 L1690 L1700 L1710 L1720 L1730 L1740 L1750 L1760 L1770 L1780 L1790 L1800 L1810 L1820 L1830 L1840 L1850 L1860 L1870 L1880 L1890 L1900 L1910 L1920 L1930 L1940 L1950 L1960 L1970 L1980 L1990 LL2000 LL2010 LL2020 LL2030 LL2040 LL2050 LL2060 LL2070 LL2080 LL2090 LL2100 LL2110 LL2120 LL2130 LL2140 LL2150 LL2160 LL2170 LL2180 LL2190 LL2200 LL2210 LL2220 LL2230

```
DC(3,1)=-SINTHA
                                                                          CBLL2240
    DC(3,2)=SINPHI*COSTHA
                                                                          CBLL2250
   DC(3,3)=COSPHI*COSTHA
                                                                          CBLL2260
   ZETA=-ZETA
                                                                          CBLL2270
                                                                          CBLL2280
   ACCELERATION COMPONENTS
                                                                          CBLL2290
                                                                          CBLL2300
   GX = -GRAV * COS(ZETA)
                                                                          CBLL2310
   GZ=GRAV*SIN(ZETA)
                                                                          CBLL2320
    GRAVX=DC(1,1)*GX+DC(3,1)*GZ
                                                                          CBLL2330
    GRAVY=DC(1,2)*GX+DC(3,2)*GZ
                                                                          CBLL2340
    GRAVZ=DC(1,3)*GX+DC(3,3)*GZ
                                                                          CBLL2350
    XCGA=DC(1,1)*XSDD+DC(2,1)*YSDD+DC(3,1)*ZSDD
                                                                          CBLL2360
    YCGA=DC(1,2)*XSDD+DC(2,2)*YSDD+DC(3,2)*ZSDD
                                                                          CBLL2370
    ZCGA=DC(1,3)*XSDD+DC(2,3)*YSDD+DC(3,3)*ZSDD
                                                                          CBLL2380
    WXX = -(WY * * 2 + WZ * * 2)
                                                                          CBLL2390
                                                                          CBLL2400
    WXY=WX*WY
    WXZ=WX*WZ
                                                                          CBLL2410
    WYY = -(WX * * 2 + WZ * * 2)
                                                                          CBLL2420
    WYZ=WY*WZ
                                                                          CBLL2430
    WZZ=-(WX**2+WY**2)
                                                                          CBLL2440
    XSUM=0.
                                                                          CBLL2450
                                                                          CBLL2460
    YSUM=0.
    ZSUM=0.
                                                                          CBLL2470
                                                                          CBLL2480
    TXSUM=0.
    TYSUM=0.
                                                                          CBLL2490
    TZSUM=0.
                                                                          CBLL2500
                                                                          CBLL2510
    DETERMINE JOINT ACCELERATIONS
                                                                          CBLL2520
                                                                          CBLL2530
    DO 750 J=1,NOJTS
                                                                          CBLL2540
    ELX=XJT(J)
                                                                          CBLL2550
    ELY=YJT(J)
                                                                          CBLL2560
    ELZ=ZJT(J)
                                                                          CBLL2570
    ELDX=0.0
                                                                          CBLL2580
    ELDY=0.0
                                                                          CBLL2590
    ELDZ=0.0
                                                                          CBLL2600
    ELDDX=0.0
                                                                          CBLL2610
    ELDDY=0.0
                                                                          CBLL2620
    ELDDZ=0.0
                                                                          CBLL2630
    DO 720 I=1,NOMODE
                                                                          CBLL2640
    ELX=ELX+XPHI(J,I)*GC(I)
                                                                          CBLL2650
    ELY=ELY+YPHI(J,I)*GC(I)
                                                                          CBLL2660
    ELZ=ELZ+ZPHI(J,I)*GC(I)
                                                                          CBLL2670
    ELDX=ELDX+XPHI(J,I)*GCD(I)
                                                                          CBLL2680
    ELDY=ELDY+YPHI(J,I)*GCD(I)
                                                                          CBLL2690
    ELDZ=ELDZ+ZPHI(J,I)*GCD(I)
                                                                          CBLL2700
                                                                          CBLL2710
    ELDDX=ELDDX+XPHI(J,I)*GCDD(I)
    ELDDY=ELDDY+YPHI(J,I)*GCDD(I)
                                                                          CBLL2720
    ELDDZ=ELDDZ+ZPHI(J,I)*GCDD(I)
                                                                          CBLL2730
720 CONTINUE
                                                                          CBLL2740
                                                                          CBLL2750
    SAX(J)=ELX
    SAY(J)=ELY
                                                                          CBLL2760
                                                                          CBLL2770
    SAZ(J) = ELZ
            XCGA+ELDDX-WZD*ELY+WYD*ELZ+2.*(-WZ*ELDY+WY*ELDZ)
                                                                          CBLL2780
    ACLX =
                                                                          CBLL2790
   ¥
            +WXX*ELX+WXY*ELY+WXZ*ELZ
```

С

С

С

```
YCGA+ELDDY+WZD*ELX-WXD*ELZ+2.*(WZ*ELDX-WX*ELDZ)
      ACLY =
                                                                             CBLL2800
    ٠¥
               +WXY*ELX+WYY*ELY+WYZ*ELZ
                                                                             CBLL2810
      ACLZ =
               ZCGA+ELDDZ-WYD*ELX+WXD*ELY+2.*(-WY*ELDX+WX*ELDY)
                                                                             CBLL2820
     ¥
               +WXZ*ELX+WYZ*ELY+WZZ*ELZ
                                                                             CBLL2830
      FX(J) = -ACLX * AMASS(J)
                                                                             CBLL2840
      FY(J) = -ACLY * AMASS(J)
                                                                             CBLL2850
      FZ(J) = -ACLZ*AMASS(J)
                                                                             CBLL2860
  750 CONTINUE
                                                                             CBLL2870
С
                                                                             CBLL2880
С
      SUM FORCES AT EACH JOINT
                                                                             CBLL2890
C
                                                                             CBLL2900
      WRITE(6,615)
                                                                             CBLL2910
      DO 755 J=1,NOJTS
                                                                             CBLL2920
      FXJT(J)=0.
                                                                             CBLL2930
                                                                             CBLL2940
      FYJT(J)=0.
  755 FZJT(J)=0.
                                                                             CBLL2950
      DO 760 I=1.NOLEG
                                                                             CBLL2960
      J=JOTMS(I)
                                                                             CBLL2970
      FXJT(J)=FMSX(I)
                                                                             CBLL2980
      FYJT(J) = FMSY(I)
                                                                             CBLL2990
  760 FZJT(J)=FMSZ(I)
                                                                             CBLL3000
      DO 765 I=1,NODS
                                                                             CBLL3010
      J = JOTDS(I)
                                                                             CBLL3020
      FXJT(J)=FDSX(I)
                                                                             CBLL3030
                                                                             CBLL3040
      FYJT(J) = FDSY(I)
  765 FZJT(J) = FDSZ(I)
                                                                             CBLL3050
      DO 780 J=1,NOJTS
                                                                             CBLL3060
      FXJT(J)=FXJT(J)+FX(J)+AMASS(J)*GRAVX
                                                                             CBLL3070
      FYJT(J)=FYJT(J)+FY(J)+AMASS(J)*GRAVY
                                                                             CBLL3080
      FZJT(J) = FZJT(J) + FZ(J) + AMASS(J) * GRAVZ
                                                                             CBLL3090
C
                                                                             CBLL3100
С
      SUM FORCES AND MOMENTS AT C. G.
                                                                             CBLL3110
С
                                                                             CBLL3120
      XSUM=XSUM+FXJT(J)
                                                                             CBLL3130
      YSUM=YSUM+FYJT(J)
                                                                             CBLL3140
      ZSUM=ZSUM+FZJT(J)
                                                                             CBLL3150
      TXSUM=TXSUM+SAY(J)*FZJT(J)-SAZ(J)*FYJT(J)
                                                                             CBLL3160
      TYSUM=TYSUM+SAZ(J)*FXJT(J)-SAX(J)*FZJT(J)
                                                                             CBLL3170
      TZSUM=TZSUM+SAX(J)*FYJT(J)-SAY(J)*FXJT(J)
                                                                             CBLL3180
      WRITE(6,402)J,FXJT(J),FYJT(J),FZJT(J)
                                                                             CBLL3190
  780 CONTINUE
                                                                             CBLL 3200
С
                                                                             CBLL3210
С
      SUMMARY INFORMATION
                                                                             CBLL3220
С
                                                                             CBLL3230
      WRITE(6,617)XSUM, YSUM, ZSUM, TXSUM, TYSUM, TZSUM
                                                                             CBLL3240
      GO TO 700
                                                                             CBLL3250
С
                                                                             CBLL3260
Ĉ
      FORMAT STATEMENTS
                                                                             CBLL3270
С
                                                                             CBLL3280
  400 FORMAT(5X,13,2X,7E10.3)
                                                                             CBLL3290
  401 FORMAT(/15X+17HJOINT COORDINATES//4X+5HJOINT+7X+1HX+11X+1HY+11X+
                                                                             CBLL3300
     ¥
             1HZ)
                                                                             CBLL3310
  402 FORMAT(6X, I2, 1X, 3(2X, E10.3))
                                                                             CBLL3320
  403 FORMAT(//,15X,11HMODE SHAPES/)
                                                                             CBLL3330
  404 FORMAT(//10X,45HINCORRECT INFORMATION READ FROM TAPE4 - STOP )
                                                                             CBLL3340
  407 FORMAT(/6X, 7HMODE = , I2, //, 4X, 5HJOINT, 7X, 1HX, 11X, 1HY, 11X, 1HZ)
                                                                             CBLL3350
```

i

```
500 FORMAT(1515)
                                                                       CBLL3360
501 FORMAT(5E10.3)
                                                                       CBLL3370
600 FORMAT(1H1,51X,33HCENTER BODY LANDING LOADS PROGRAM
                                                                       CBLL3380
   ¥
                                            /42X,53HMASTER AGREEMENT, CCBLL3390
   *ONTRACT NAS1-8137, TASK ORDER FIVE/46X,45HMCDONNELL DOUGLAS ASTRONCBLL3400
  *AUTICS COMPANY - EAST,////)
                                                                       CBLL3410
601 FORMAT(10X, 9HNOJTS = ,I10/10X,9HNOLEG = ,I10/10X,9HNOMODE = ,
                                                                       CBLL3420
           I10/10X,9HNOTIME = ,I10/10X,9HNOSETS = ,I10)
                                                                       CBLL3430
602 FORMAT(10X,9HMODE
                       = ,5I10)
                                                                       CBLL3440
603 \text{ FORMAT(lox,9HJOTMS} = ,5110)
                                                                       CBLL3450
604 FORMAT(lox,9HJOTDS = ,10I10)
                                                                       CBLL3460
605 FORMAT(10X,9HZETA
                        = ,E10.3/10X,9HGRAV
                                                                       CBLL3470
                                              = .E10.3)
                        = ,10E10.3,/(19X,10E10.3))
                                                                       CBL13480
606 FORMAT(10X,9HMASS
                        = ,10E10.3/29X,10E10.3)
607 FORMAT(10X,9HTYM
                                                                       CBLL3490
610 FORMAT(15X ,7HTIME = ,E10.3/10X,7HXSDD = ,E10.3,8H YSDD = ,
                                                                       CBLL3500
   ¥
           E10.3,8H ZSDD = ,E10.3/10X,7HPHI = ,E10.3,8H THTA = ,
                                                                       CBLL3510
   ×
            E10.3,8H PSI = ,E10.3/10X,7HWX
                                             = ,E10.3,8H WY = ,
                                                                       CBLL3520
           E10.3,8H WZ = ,E10.3/10X,7HWXD = ,E10.3,8H WYD = ,
   ¥
                                                                       CBLL3530
           E10.3,8H WZD = ,E10.3)
                                                                       CBLL3540
   ¥
611 FORMAT(10x,7HMODE = ,15,5x,8H GC
                                     = ,E10.3,8H GCD = ,E10.3,
                                                                       CBLL3550
          8H GCDD = +E10.3
   ¥
                                                                       CBLL3560
                     = ,I5,5X,8H FMSX = ,E10.3,8H FMSY = ,E10.3,
                                                                       CBLL3570
612 FORMAT(10X,7HI
          8H FMSZ = .E10.3
                                                                       CBLL3580
   <del>x</del>
613 FORMAT(10X,7HI
                      = ,I5,5X,8H FDSX = ,E10.3,8H FDSY = ,E10.3,
                                                                       CBLL3590
          8H FDSZ = .E10.3
                                                                       CBLL3600
  ×
614 FORMAT(/,15X,17HTIME HISTORY DATA,/)
                                                                       CBLL3610
615 FORMAT(//15X,38HCENTER BODY LANDING LOADS DISTRIBUTION//4X,5HJOINTCBLL3620
          ,7X,1HX,11X,1HY,11X,1HZ)
                                                                       CBLL3630
616 FORMAT(/,15X,16HDRAG STRUT LOADS,/)
                                                                       CBLL3640
617 FORMAT(//,15X,28HLANDING LOADS SUMMED AT C.G.//,6X,5HFX = ,
                                                                       CBLL3650
          E10.3/6X,5HFY = ,E10.3/6X,5HFZ = ,E10.3/6X,5HTX = ,
                                                                       CBLL3660
   ¥
   ¥
          E10.3/6X,5HTY = .E10.3/6X,5HTZ = .E10.3
                                                                       CBLL3670
619 FORMAT(1H1)
                                                                       CBLL3680
620 FORMAT(/,15X,16HMAIN STRUT LOADS,/)
                                                                       CBLL3690
    FND
                                                                       CBLL3700
```
THIS PAGE INTENTIONALLY LEFT BLANK

APPENDIX F

NUMERICAL INTEGRATION ROUTINE LANDING LOADS AND MOTIONS PROGRAM

F.1 <u>Introduction</u> - The methods and procedures employed during the numerical integration of the equations of motion in the Landing Loads and Motions Program are described in the following paragraphs. These operations are performed in the subroutines RKCUT and INITUP. The sequence of program calls for these subroutines are shown in Figure 5-17 and subroutine listings are given in Appendix I.

F.2 <u>Method</u> - The second-order equations of motion in the Landing Loads and Motions Program are reduced to an equivalent set of simultaneous firstorder equations. Each of these first order equations takes the general form of

$$\dot{y} = f(y,t)$$
 where $y(t_0) = y_0$ (F-1)

At any step n in the integration of these equations with respect to time t, the quantities t_n , y_n , and $\dot{y}_n = f(y_n, t_n)$ are available. To compute y_{n+1} and thus $\dot{y}_{n+1} = f(y_{n+1}, t_{n+1})$, where $t_{n+1} = t_n + \Delta t$, and Δt (program variable HZ) is the integration interval, the following Runge-Kutta formulas are used:

$$y_{A} = y_{n} + \frac{\Delta t}{2} \dot{y}_{n}, \qquad \dot{q}_{n} = \dot{y}_{n}, \qquad \dot{y}_{A} = f(y_{A}, t_{n} + \frac{\Delta t}{2})$$

$$y_{B} = y_{n} + \frac{\Delta t}{2} \dot{y}_{A}, \qquad \dot{q}_{A} = \dot{q}_{n} + 2\dot{y}_{A}, \qquad \dot{y}_{B} = f(y_{B}, t_{n} + \frac{\Delta t}{2})$$

$$y_{C} = y_{n} + \Delta t \dot{y}_{B}, \qquad \dot{q}_{B} = \dot{q}_{A} + 2\dot{y}_{B}, \qquad \dot{y}_{C} = f(y_{C}, t_{n} + \Delta t)$$

$$y_{n+1} = y_{n} + \Delta t (\dot{q}_{B} + \dot{y}_{C}), \qquad \dot{q}_{C} = \dot{y}_{C}, \qquad \dot{y}_{n+1} = f(y_{n+1}, t_{n} + \Delta t)$$
(F-2)

References (F-1) and (F-2) present an explanation of these formulas and a discussion of the Runge-Kutta method.

These formulas are applied to the system of equations using either a fixed or variable integration interval, Δt . When a variable Δt is used, the routine computes a truncation error indicator at each step of the integration, based on the quantity

$$E_{n+1} = Max \{ Min [|\Delta t(\dot{y}_{n+1}^{i} - \dot{y}_{C}^{i}) |, | \frac{\Delta t(\dot{y}_{n+1}^{i} - \dot{y}_{C}^{i})}{y_{n+1}^{i}} |] \}$$
(F-3)

where i = 1, 2, ..., N and N is the total number of integrated variables, and the superscript i denotes the ith integrated variable. The notation of Equation (F-3), indicates that for each ith integrated variable, the minimum of the two quantities

$$\Delta t (\dot{y}_{n+1}^{i} - \dot{y}_{C}^{i}) \mid \text{and} \mid \frac{\Delta t (\dot{y}_{n+1}^{i} - \dot{y}_{C}^{i})}{\overset{i}{y_{n+1}^{i}}}$$

is saved. The truncation error indicator, E_{n+1} , is then set equal to the maximum of these minimums. The value of E_{n+1} is compared with the two constants E_{max} (EMAX) and E_{min} (EMIN) which are the input values for the maximum and minimum allowable truncation errors, respectively. This comparison results in a modification of Δt as follows:

- 1. If $E_{n+1} < E_{min}$ for two consecutive integration steps, Δt is doubled and the integration continues with the new Δt . However, if Δt equals Δt_{max} (HMAX), Δt is not doubled.
- 2. If $E_{\min} \leq E_{n+1} \leq E_{\max}$, Δt is left unchanged.
- 3. If $E_{n+1} \ge E_{max}$, Δt is halved and the values at step n, which are saved in COMMON, are used to continue the integration. However, if Δt equals Δt_{min} (HMIN), Δt is not halved.

This procedure is continuously applied to the center body equations of motion, Equation (5-23). The equations of motion for the footpads, Equation (5-4), are continuously integrated by the above procedure when the "con-tacting footpad" option of the program is employed. If the "noncontacting footpad" option is specified, a footpad's equations of motion are integrated only when the footpad is determined to be in contact with the landing surface. For a footpad approaching the landing surface, contact is defined as the

condition when the bottom of the footpad is found to be within a tolerance band above and below the landing surface, as shown in Figure F-1(a). If this footpad is found to have penetrated below this tolerance band as shown in Figure F-1(b), a linear interpolation procedure is employed to estimate at what time the footpad would have entered the tolerance band. The integration is backed up and then continued from this point in time with the footpad considered contacting.

F.3 <u>Subroutine Calls</u> - There are six calls to the integration routine. A brief description of each of the entries into the subroutines is given below.

Subroutine	Entry Point	Operation
INITUP	LOC	Sets up a list of cutoff variables and their cutoff values.
INITUP	INUPD	Sets up a list of all integrated variables.
RKCUT	SETUP	Initializes numerical integration procedure for type of integration requested.
RK CUT	INTEG	Performs all integration and com- putes truncation error indicator.
RKCUT	UPDATE	Updates integration variables and modifies integration interval when required.
RKCUT	CUT	Checks cutoff limits for introduction of footpad equations into integration routine.



FIGURE F-1 INTEGRATION CONTROL TOLERANCE FOR FOOTPAD EQUATIONS OF MOTION

F-4 References

- F-1 Hamming, R. W., <u>Numerical Methods for Scientists and Engineers</u>, McGraw-Hill Book Company, New York, 1962, pp. 212-213.
- F-2 Conte, S. D., <u>Elementary Numerical Analysis</u>, McGraw-Hill Book Company, New York, 1965, pp. 204-258.

THIS PAGE INTENTIONALLY LEFT BLANK

APPENDIX G

CENTER BODY INTERNAL LOAD

DISTRIBUTION FOR TASK ORDER THREE

LANDER USING REFINED IDEALIZATION

A refined idealization of the Task Order Three lander center body, shown in Figure G-1, was employed to determine displacements and internal loads for the loading condition defined in Section 6.2.4. This idealization, which employs 72 joints and 108 bar elements, includes a more refined model of the side beams and radial beams. The center hub was modeled with nine elements to more nearly represent a cylinder. In addition, clevis fittings, which connect the drag struts of a gear to the center body, were idealized with elements such as 4, 11, 29, and 36. Accordingly, joints 22 and 27 (and 40 and 45 which are hidden) are the actual center body attach points for gears 2 and 3 and were assumed to be pinned supports. Main strut attach points (joints 25 and 43) for gears 2 and 3 were also assumed to be pinned supports.

Loads that were applied to the joints in the idealization of Figure 6-8 were applied to coincident joints in Figure G-1. For example, loads that were applied to joint 6 in Figure 6-8 were applied to joint 10 in the refined idealization. Loads that were applied to joints 31, 32 and 33 (on the center hub) in Figure 6-8 were distributed to joints 67 through 72, some of which are hidden in Figure G-1. These loads were distributed to the refined idealization joint locations in a statically equivalent manner.

A comparison of selected displacements and internal loads obtained with the 33 joint idealization and the refined idealization is given in Figure G-2.



FIGURE G-1 REFINED CENTER BODY STRUCTURAL IDEALIZATION

		ELEMENT INTERNAL LOADS AT "P" END				
	FORCE	FORCE	FORCE	MOMENT	MOMENT	MOMENT
	X	Y	Z	X	Y	Z
	(DYNES)	(DYNES)	(DYNES)	(CM-DYNES)	(CM-DYNES)	(CM-DYNES)
ELEMENT 14 (FIGURE 6-8)	-2.588 x 10 ⁸	9.057 x 10 ⁷	-1.025 x 10 ⁸	9.972 x 10 ⁶	5.477 x 10 ⁹	4.688 x 10 ⁹
ELEMENT 21 (FIGURE G-1)	-2.140 x 10 ⁸	1.230 x 10 ⁸	-1.778 x 10 ⁸	1.088 x 10 ⁷	3.830 x 10 ⁹	8.550 x 10 ⁹
ELEMENT 32 (FIGURE 6~8)	-5.394 x 10 ⁸	6.205 x 10 ⁸	1.633 x 10 ⁸	5.697 x 10 ⁶	4.800 x 10 ⁹	1.061 x 10 ¹⁰
ELEMENT 52 (FIGURE G-1)	-1.115 x 10 ⁹	1.163 x 10 ⁹	1.822 x 10 ⁸	5.800 x 10 ⁷	6.190 x 10 ⁹	1.756 x 10 ¹⁰

Г

FIGURE G-2 COMPARISON OF COMPUTER RESULTS FOR 33 AND 72 JOINT IDEALIZATIONS OF TASK ORDER THREE LANDER CENTER BODY

*FIGURE 6-8 ILLUSTRATES THE 33 JOINT IDEALIZATION. **FIGURE G-1 ILLUSTRATES THE REFINED (72 JOINT) IDEALIZATION.

	JOINT	DISPLACEMEN	rs (CM)
	Х	Ŷ	Z
JOINT 1 (FIGURE 6-8)*	-2.594 x 10 ⁻¹	-1.284 x 10 ⁻⁵	9.955 x 10 ⁻²
JOINT 1 (FIGURE G-1)**	-5.325 x 10 ⁻¹	-1.877 x 10 ⁻⁴	1.775 x 10 ⁻¹
JOINT 8 (FIGURE 68)	-1.342 x 10 ⁻¹	2.195 x 10 ⁻¹	-2.636 x 10 ⁻²
JOINT 14 (FIGURE G-1)	-2.365×10^{-1}	4.810×10^{-1}	-1.007 x 10 ⁻¹

APPENDIX H

PROGRAM LISTING

STRUCTURAL ANALYSIS PROGRAM

		OVERLAY (SASLP, 0, 0)	0000	10
		·		
		PROGRAM SAPTS (INDUT.OUTPUT.TAPES=INDUT.TAPE6=OUTPUT.	MATN	10
	1	TAPE1, TAPE2, TAPE9)	MAIN	20
	-	COMMON COM(30)	MAIN	30
		EQUIVALENCE (COM(17), INDRKT)	MAIN	40
		EQUIVALENCE (COM(24), INDNMA)	MAIN	50
_		EQUIVALENCE (COM(30), INDPGM)	MAIN	60
Ĉ		THE DIMENSION OF COMMIN CAN BE INCREASED OF DECREASED	ΜΑΙΝ	80
c		TO CONTROL THE STORAGE AVAILABLE FOR THE STIFFNESS	MAIN	90
c		MATRIX (IF ISEDIM IS ALSO CHANGED).	MAIN	100
č			MAIN	110
С		COMMON / CMAIN / COMAIN(14455)	MAIN	120
		COMMON / CMAIN / COMAIN(16955)	MAIN	130
	5	CONTINUE	MAIN	140
С		INITIALIZE WRSTRK ROUTINE	MAIN	150
~		CALL WRSTRI(I,J,DUM)	MAIN	170
c		DATA AND VARIABLE SET-UP	MAIN	180
c		DATA AND VARIABLE SET OF	MAIN	190
•		CALL SECOND (TIMEO)	MAIN	200
		CALL OVERLAY (5LSASLP,1,0,6HRECALL)	MAIN	210
		CALL SECOND (TIME1)	MAIN	220
		IF (INDPGM .EQ. O) GO TO 8	MAIN	230
Ç			MAIN	240
Ċ		GEAR ANALYSIS INPUT AND PROCESSING ROUTINES	MAIN	220
C		CALL OVERIAN (SUGEARRISSO SHRECALL)	MAIN	270
		CALL SECOND(TIME2)	MAIN	280
		GO TO 20	MAIN	290
	8	CONTINUE	MAIN	300
C			MAIN	310
C		CENTER BODY OPTION ROUTINES	MAIN	315
Ć		LOCAL STIFFNESS AND TRANSFORMATION MATRICES AND STRUCTURAL	MAIN	320
C		STIFFNESS MATRIX GENERATION	MAIN	320
C		TE (INDERT, ED. O) CALL OVERLAY (51 SASLE.2.0.6HEECALL)	MAIN	350
		CALL SECOND (TIME2)	MATN	360
С		******	MAIN	370
С		* OVERLAYS THREE AND FOUR ARE MUTUALLY EXCLUSIVE *	MAIN	380
С		***************************************	MAIN	390
		IF (INDNMA.NE.O) GO TO 10	MAIN	400
ç		DICHLACEMENT DOTATION FORCE AND NOVENT COLUTION	MAIN	410
Ċ		DISPLACEMENT, ROTATION, FORCE, AND MOMENT SOLUTION		420
C		CALL OVERLAY (51 SASLP. 3.0.6HRECALL)	MAIN	440
		CALL SECOND (TIME3)	MAIN	450
		GO TO 20	MAIN	460
С			MAIN	470
С		NORMAL MODE ANALYSIS SECTION	MAIN	480
C			MAIN	490
10		CALL OVERLAY (5LSASLP,4,0,6HRECALL)		500
20		CALL SECOND (TIME4)	MAIN	520
<u>د</u> ب		CONTAINCE		

C C C C C C

C C C

	TO1=TIME1-TIME0 MAIN	530
	T12=TIME2-TIME1 MAIN	540
	WRITE(6,40)TIME0,TIME1,TO1 MAIN	550
	IF (INDPGM •EQ• 0) GO TO 25 MAIN	560
	WRITE(6,44)TIME1,TIME2,T12 MAIN	570
	GO TO 5 MAIN	580
	25 CONTINUE MAIN	590
	WRITE(6,42)TIME1,TIME2,T12 MAIN	600
	IF (INDNMA.EQ.O) GO TO 30 MAIN	610
	T23OR4=TIME4-TIME2 MAIN	620
	WRITE (6,50) TIME2,TIME4,T23OR4 MAIN	630
	GO TO 5 MAIN	640
30	CONTINUE	650
	T230R4=TIME3-TIME2 MAIN	660
	WRITE (6,60) TIME2, TIME3, T230R4 MAIN	670
_	GO TO 5 MAIN	680
C	MAIN	690
40	FORMAT (IHI38X,26HCPU TIME USAGE TABLE (SEC)///63X32H TIME IN MAIN	700
	ITIME OUT TOTAL, /33HOINPUT AND INITIALIZATION OVERLAY, 26X2(2XFMAIN	710
	2 10•3),1XF10•2 /) MAIN	720
	42 FORMAT (47HOSTRUCTURAL STIFFNESS MATRIX GENERATION OVERLAY, MAIN	730
	$\frac{1}{1} \frac{14 \times F10.3}{10} \frac{2 \times F10.3}{10} \frac{1 \times F10.2}{10} \frac{1}{10} \frac{10}{10} \frac{10}$	1 740
	44 FORMAT (32HOGEAR INPUT AND ANALYSIS OVERLAY, 27X2(2XF10+3), MAIN	1 750
- 0	I IXFI0.2) MAIN	1 760
50	FORMAT (29HONORMAL MODE ANALYSIS OVERLAY, 32XF10.3,2XF10.3,1XF10.2)MAIN	1 770
00	FORMAT (39HODISPLACEMENT) ROTATION, FORCE, AND MOMENT SOLUTION OVEMAL	1 700
		1 190
	ENU MAIN	1 800

		SUBROUTINE ERPNT1 (ALPHA,LNGTH,ICODE)	ERPN	10
С	•	PRINT ERROR MESSAGES	ERPN	20
С		ICODE = 0, NON-FATAL ERROR, CONTINUE	ERPN	30
С		ICODE = 1, FATAL ERROR, STOP	ERPN	40
С		ICODE =-1, FATAL ERROR, SET FATAL ERROR INDICATOR AND	ERPN	50
С		CONTINUE	ERPN	60
		DATA IFATAL / O /	ERPN	70
		DIMENSION ALPHA(13)	ERPN	80
		IF(ICODE•NE•O) GO TO 70	ERPN	82
		WRITE(6,50) (ALPHA(I),I=1,LNGTH)	ERPN	84
		GO TO 30	ERPN	86
	70	CONTINUE	ERPN	88
		WRITE (6,40) (ALPHA(I),I=1,LNGTH)	ERPN	90
		IF (ICODE) 10,30,20	ERPN	100
10		IFATAL=1	ERPN	110
		GO TO 30	ERPN	120
20		STOP	ERPN	130
30		RETURN	ERPN	140
С		CHECK ENTRY FOR FATAL INDICATOR	ERPN	150
		ENTRY ERPNT2	ERPN	160
		IF (IFATAL•EQ•1) STOP	ERPN	170
		RETURN	ERPN	180
С			ERPN	190
	40	FORMAT(20H ***ERROR*** /13A10//)	ERPN	200
	50	FORMAT(20H ***WARNING*** /13A10//)	ERPN	205
		END	ERPN	210

с	SUBROUTINE WRSTRK (IROW, JCOL, STFNZ)	WRST	10 20
C C	IROW = ROW NO. JCOL = COLUMN NO. STFNZ = VARIABLE	WRST	30
C	DIMENSION JCOLD(6), STFNZD(6), COL(6) DATA COL / 6*4HCOL. /	WRST WRST WRST	60 70
C C C	THE NON ZERO ELEMENTS OF THE STRUCTURAL STIFFNESS MATRIX ARE PASSED ONE AT A TIME AND PRINTED BY ROWS SIX	WRST WRST WRST	80 90 100
c	VALUES PER LINE	WRST	110
c	IF (IRSAVE.EQ.IROW) GO TO 10	WRST	120
c	IF NEW ROW, PRINT ANY VARIABLES OF THE LAST ROW BEING HELD IN	WRST	140
c	A LINE AND START A NEW LINE		150
10	INDRNS=1	WRST	160
	GO TO 20	WRST	170
	IF (ICOUNT•NE•6) GO TO 40	WRST	180
C C	IF SIX VARIABLES HAVE BEEN STORED IN A LINE, PRINT IT AND	WRST WRST WRST	190 200 210
20	CONTINUE IF (INDLIN•NE•1) GO TO 50	WRST	220
	WRITE (6,80) IRSAVE, (COL(I), JCOLD(I), STENZD(I), I=1, ICOUNT)	WRST	240
	LINET=LINET+2	WRST	250
	INDLIN=2	WRST	260
30	ICOUNT=0	WRST	270
	IF (INDRNS+EQ+0) GO TO 40	WRST	280
	INDRNS=0	WRST	290
4.0	INDLIN=1	WRST	300
	IRSAVE=IROW	WRST	310
40	JCOLD(ICOUNT)=JCOL $STFNZD(ICOUNT)=STFNZ$ $IF (LINET.GI.56) GO TO 60$	WRST WRST WRST	330 340 350
50	RETURN	WRST	360
	WRITE (6,90) (COL(I),JCOLD(I),STFNZD(I),I=1,ICOUNT)	WRST	370
	LINET=LINET+1	WRST	380
60	GO TO 30	WRST	390
	IF (IROW+LT+O) RETURN	WRST	400
	WRITE (6+70)	WRST	410
c	LINET=3	WRST	420
	RETURN	WRST	430
C C	INITIALIZE THIS ROUTINE	WRST WRST WRST	440 450 460
	ENTRY WRSTRI	WRST	470
	LINET = 59	WRST	480
	ICOUNT = 0	WRST	490
	INDLIN = 1	WRST	500
c	INDRNS = 0	WRST	510
	IRSAVE = 1	WRST	520
	RETURN	WRST	530
C	FORMAT 100 IS USED TO PRINT THE FIRST LINE FOR EACH ROW	WRST	540 550

C C	FORMAT 200 IS USED FOR ALL OTHER LINES (INDLIN .NE. 1)	WRST 56 WRST 57 WRST 58	0 20 30
70	FORMAT (28H1STRUCTURAL STIFFNESS MATRIX//)	WRST 59	0
80	FORMAT (5HOROW I3.6(1XA4.)I4.)H= $E10.3$)	WRST 60	0
90	FORMAT (8X,6(1XA4,14,1H=E10.3))	WRST 61	. Ó
	END	WRST 62	20

OVERLA, (SASLP, 1, 0)	OV10 10
PROGRAM INITAL COMMON COM(30) EQUIVALENCE (COM(30), INDPGM)	INIT 10 INIT 20 INIT 30 INIT 40
READ DATA CARDS AND SORT INTO PROPER ORDER	INIT 50 INIT 60
CALL DATSET IF(INDPGM-EQ-1)RETURN	INIT 70 INIT 80 INIT 90
READ VARIABLE INPUT VALUES ARRANGED BY (DATSET)	INIT 100
CALL RDDATA RETURN END	INIT 120 INIT 130 INIT 140

SUBROUTINE DATSET Dats 10 DIMENSION CARD(8), IDATA(7) Dats 20 COMMON COM(30) Dats 30 EQUIVALENCE (COM(1), NJOINT) Dats 40 EQUIVALENCE (COM(2), NFORCE) Dats 50 EQUIVALENCE (COM(3), NNOMNT) Dats 60 EQUIVALENCE (COM(3), NNOMNT) Dats 70 EQUIVALENCE (COM(30), NNOMNT) Dats 70 EQUIVALENCE (COM(30), NNPNT) Dats 70 EQUIVALENCE (COM(30), NNPNT) Dats 70 EQUIVALENCE (COM(30), NNPNT) Dats 70 EQUIVALENCE (COM(30), INDPGM) Dats 110 DIMENSION ERJ(5) DATS 110 DIMENSION ERJ(5), ERF(5), ERM(5), ERB(5) DATS 120 DIMENSION ERJ(5) DATS 120 DATA ERS / 10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 130 DATS 120 DATA ERJ / 10H THE LIMIT, 10HS ARE NOT, 10H NUMBERED , DATS 120 DATS 120 DATA ERJ / 10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 210 DATS 120 DATA ERJ / 10H THE DATCH, 10HY TALLY / DATS 220 DATA ERS / 10H THE ORCE, 10HYTIALLY / DATS 220 DATA ERS / 10H THE ORCE, 10HYTIALLY / DATS 220 DAT				
DIMENSION CARD(3), IDATA(7) DATS 20 COMMON COM(30) DATS 30 EQUIVALENCE (COM(1), NORT) DATS 40 EQUIVALENCE (COM(2), NRORCE) DATS 50 EQUIVALENCE (COM(3), NRORNT) DATS 70 EQUIVALENCE (COM(3), NAPAR) DATS 70 EQUIVALENCE (COM(2), NJPHT) DATS 70 EQUIVALENCE (COM(20), NLIMIT) DATS 100 EQUIVALENCE (COM(30), INDPGM) DATS 110 DIMENSION ERL(5) DATS 110 DIMERSION ERL(5) DATS 110 DIMERSION ERL(5) DATS 110 DATA ERS / 104 THE SHEAR, 104 PANELS AR, 104E NOT NUMB, DATS 120 DATA ERS / 104 THE NOTALL, 1041Y / DATS 130 DATA ERS / 104 THE NOTAL, 1041Y / DATS 130 DATA ERS / 104 THE NOTAL, 1041Y / DATS 230 DATA ERS / 104 THE PORCE, 104NTALLY / DATS 230 DATA ERS / 104 THE		SUBROUTINE DATSET	DATS	10
COMMON COM(30) DATS 30 EQUIVALENCE (COM(1), NJOINT) DATS 40 EQUIVALENCE (COM(2), NFORCE) DATS 50 EQUIVALENCE (COM(2), NNOMNT) DATS 70 EQUIVALENCE (COM(2), NNOMNT) DATS 70 EQUIVALENCE (COM(2), NLIMIT) DATS 70 DIMENSION ERL(5) DATS 710 DIMENSION ERL(5) DATS 710 DIMENSION ERL(5) DATS 710 DATS 710 DATS 710		DIMENSION CARD(8), IDATA(7)	DATS	20
EQUIVALENCE (COM(1), NJOINT) DATS 40 EQUIVALENCE (COM(2), NFORCE) DATS 50 EQUIVALENCE (COM(3), NFORMT) DATS 60 EQUIVALENCE (COM(3), NADANT) DATS 70 EQUIVALENCE (COM(2), NAPAT) DATS 70 EQUIVALENCE (COM(2), NAPAT) DATS 100 EQUIVALENCE (COM(2), NAPAT) DATS 100 DIMENSION ERL(5) DATS 120 DIMENSION ERL(5) DATS 120 DIMENSION ERL(5) DATS 110 DATA ERSH / 10H THE SHEAR. 10H PANELS AR. 10HE NOT NUMB, DATS 120 DATA ERSH / 10H THE LIMIT, 10HS ARE NOT, 10H NUMBERED , DATS 160 DATA ERL / 10H THE DSEQUE, 10HNTIALLY. / DATS 120 DATA ERL / 10H THE NOSAL, 10H POINTS AR, 10HE NOT NUMB, DATS 120 DATA ERL / 10H THE NOSAL, 10H POINTS AR, 10HE NOT NUMB, DATS 120 DATA ERL / 10H THE ROSEQUE, 10HNTIALLY / DATS 210 DATA ERL / 10H THE ROSEQUE, 10HNTIALLY / DATS 220 DATA ERL / 10H THE ROSEQUE, 10HNTIALLY / DATS 220 DATA ERL / 10H THE MOVEN, 10HT VECTORS , 10HARE NOT NUM, DATS 220 DATA ERL / 10H THE MOVEN, 10HTALLY / DATS 220 DATA ERL / 10H THE DATA, 10HCASE		COMMON COM(30)	DATS	30
EQUIVALENCE (COM(2), NEORCE) DATS 50 EQUIVALENCE (COM(4), NEAR) DATS 70 EQUIVALENCE (COM(4), NEAR) DATS 70 EQUIVALENCE (COM(20), NUPRT) DATS 80 EQUIVALENCE (COM(20), NUPRT) DATS 100 EQUIVALENCE (COM(20), NUPRT) DATS 100 EQUIVALENCE (COM(20), NUPRGM) DATS 100 DIMENSION ERL(5) DATS 110 DIMENSION ERSH(5) DATS 120 DATA ERSH / 10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 150 DATS 120 DATA ERSH / 10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 150 DATS 120 DATA ERSH / 10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 120 DATA 117, 10HS 200 DATA ERL / 10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 120 DATS 120 DATA ERL / 10H THE SHEAR, 10H POINTS AR, 10HE NOT NUMB, DATS 120 DATS 120 DATA ERL / 10H THE CARCE, 10H VECTORS A, 10HRE NOT NUM, DATS 220 DATA ERK / 10H THE DATA (10HNTIALLY / DATS 230 DATA ERK / 10H THE SHEAR, 10HUENTIALLY / DATS 230 DATA ERK / 10H THE DATA (10HOUENTIALLY / DATS 240 DATA ERK / 10H THE CARCE, 10H VECTORS A, 10HARE NOT NU, DATS 240 DATS 240 DATA ERK / 10H THE DATA (10HUENTIALLY / DATS 240 DATS 240 DATA E		EQUIVALENCE (COM(1), NJOINT)	DATS	40
EQUIVALENCE (COM (3), NMONNT) DATS 60 EQUIVALENCE (COM (5), NJPAT) DATS 70 EQUIVALENCE (COM (2), NJPAT) DATS 80 EQUIVALENCE (COM (2), NJPAT) DATS 10 EQUIVALENCE (COM (2), NJPAT) DATS 10 EQUIVALENCE (COM (2), NJPAT) DATS 100 EQUIVALENCE (COM (2), NJPAT) DATS 120 DIMENSION ERJ(5) DATS 120 DIMENSION ERJ(5) DATS 120 DATA ERSH /10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 150 DATS 120 DATA ERSH /10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 160 DATS 120 DATA ERSH /10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 120 DATS 120 DATA ERSH /10H THE NDAL, 10H VEY / DATS 120 DATA ERSH /10H THE NDAL, 10H VEY / DATS 200 DATA ERSH / 10H THE NDAL, 10H VEY (/ DATS 200 DATA ERSH / 10H THE NDAL, 10H VEY (/ DATS 200 DATA ERSH / 10H THE NDALE, 10H VEY (/ DATS 220 DATA ERSH / 10H THE PORCE, 10H VEY (/ DATS 220 DATA ERSH / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT NUM, DATS 220 DATS 220 DATA ERSH / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N, DATS 220		EQUIVALENCE (COM(2), NFORCE)	DATS	50
EQUIVALENCE (COM (4), NJPAT) DATS 70 EQUIVALENCE (COM (2), NJPAT) DATS 80 EQUIVALENCE (COM (2), NLIMIT) DATS 100 EQUIVALENCE (COM (3), INDPGM) DATS 110 DIMENSION ERL(5) DATS 110 DIMENSION ERL(5) DATS 110 DATA ERSH /10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 150 DATA ERSH /10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 150 DATA ERL / 10H THE LIMIT, 10HS ARE NOT, 10H NUMBERED , DATS 150 DATA ERL / 10H THE LIMIT, 10HS ARE NOT, 10H NUMBERED , DATS 160 DATA ERL / 10H THE CORE, 10H VECTORS AR, 10HE NOT NUMB, DATS 210 DATA ERL / 10H THE CORE, 10H VECTORS A, 10HRE NOT NUM, DATS 220 DATA ERK / 10H THE KORE, 10H VECTORS A, 10HRE NOT NUM, DATS 220 DATA ERK / 10H THE KORE, 10H VECTORS A, 10HRE NOT NUM, DATS 220 DATA ERK / 10H THE KORE, 10H VECTORS A, 10HARE NOT NU, DATS 220 DATA ERK / 10H THE KORE, 10H VECTORS A, 10HARE NOT NU, DATS 220 DATA ERK / 10H THE KORE, 10H VECTORS A, 10HARE NOT NU, DATS 220 DATA ERK / 10H THE KORE, 10HUETIALLY / DATS 220 DATA ERK / 10H THE KORE, 10HUETIALLY / DATS 220 DATA ERK / 10H THE KORE, 10HUETIALLY / DATS 220 <td></td> <td>EQUIVALENCE (COM(3), NMOMNT)</td> <td>DATS</td> <td>60</td>		EQUIVALENCE (COM(3), NMOMNT)	DATS	60
EQUIVALENCE (COM(5), NLTMT)DATS 80EQUIVALENCE (COM(20), NLTMIT)DATS 100EQUIVALENCE (COM(30), TADPGM)DATS 110DIMENSION ERL(5)DATS 110DIMENSION ERL(5)DATS 110DIMENSION ERL(5)DATS 110DATA ERSH / IOH THE SHEAR, 10H PANELS AR, 10HE NOT NUMB,DATS 150DATA ERSH / IOH THE LIMIT, 10HS ARE NOT, 10H NUMBERED ,DATS 160DATA ERSH / IOH THE LIMIT, 10HS ARE NOT, 10H NUMBERED ,DATS 170DATA ERSH / IOH THE LIMIT, 10HS ARE NOT, 10H NUMBERED ,DATS 180DATA ERSH / IOH THE NOTAL, 10H POINTS AR, 10HE NOT NUMB,DATS 120DATA ERSH / IOH THE NOTAL, 10H VECTORS A, 10HRE NOT NUM,DATS 220DATA ERSH / IOH THE FORCE, 10H VECTORS A, 10HRE NOT NUM,DATS 220DATA ERSH / IOH THE BAR D, 10HT VECTORS A, 10HRE NOT NUM,DATS 220DATA ERSH / 10H THE MOMEN, 10HT VECTORS A, 10HRE NOT NUM,DATS 220DATA ERSH / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT NU,DATS 240DATA GERR / 10HGEAR/DATS 250110HMBERED SEQ, 10HUENTIALLY /DATS 250DATA ECHDC / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A,DATS 250DATA ECHDC / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A,DATS 330NJOINT=0DATS 340DATS 340NJOINT=0DATS 340NHORKT=0DATS 340MAR=0DATS 340NHORKT=0DATS 340NHORKT=0DATS 340MOMNT=0DATS 340MAR = 0DATS 420MAR = 0DATS 420MAR = 0DATS 420MAR =		EQUIVALENCE (COM(4), NBAR)	DATS	70
EQUIVALENCE (COM (20), NNPPAN) DATS 100 EQUIVALENCE (COM (30), INDPGM) DATS 110 DIMENSION ERL(5) DATS 110 DIMENSION ERL(5) DATS 120 DIMENSION ERSH(5) DATS 150 DATA ERSH / 10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 150 DATA ERL / 10H THE LIMIT, 10HS ARE NOT, 10H NUMBERED , DATS 160 DATA ERL / 10H THE LIMIT, 10HS ARE NOT, 10H NUMBERED , DATS 170 1 10HERED SEQUE, 10HNTIALLY / DATS 180 DATA ERL / 10H THE MODAL, 10H POINTS AR, 10HE NOT NUMB, DATS 210 DATA ERL / 10H THE MORE, 10HNTIALLY / DATS 220 DATA ERF / 10H THE MORE, 10HVECTORS A, 10HRE NOT NUM, DATS 220 DATA ERF / 10H THE BAR D. 10HEFINITIONS, 10H ARE NOT NUM, DATS 220 DATA ERM / 10H THE MORE, 10HUENTIALLY / DATS 220 DATA ERM / 10H THE BAR D. 10HEFINITIONS, 10H ARE NOT NUM, DATS 220 DATA ERM / 10H THE BAR D. 10HEFINITIONS, 10H ARE NOT NUM, DATS 220 DATA ERM / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A, DATS 220 DATA STRUCT / 10HSTRUCTURE / DATS 330 NOINT=0 DATS 330 NAT ERD / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A, DATS 330 NAT ERD / 10H		EQUIVALENCE (COM(5), NJPNT)	DATS	80
EQUIVALENCE (COM (29), NSHPAN) DATS 100 EQUIVALENCE (COM (29), NSHPAN) DATS 110 DIMENSION ERL(5) DATS 120 DIMENSION ERSH(5) DATS 130 DATA ERSH /10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 150 DATA ERSH /10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 160 DATA ERSH /10H THE SHEAR, 10H POINTS AR, 10HE NOT NUMB, DATS 160 DATA ERSH / 10H THE NODAL, 10H POINTS AR, 10HE NOT NUMB, DATS 120 DATA ERS / 10H THE FORCE, 10H VECTORS A, 10HRE NOT NUMB, DATS 220 DATA ERF / 10H THE MOMEN, 10HT VECTORS , 10HARE NOT NUM, DATS 230 DATA ERM / 10H THE MOMEN, 10HT VECTORS , 10HARE NOT NU, DATS 230 1 10HBERED SEQUE, 10HUENTIALLY / DATS 230 1 10HBERED SEQUE, 10HUENTIALLY / DATS 230 1 10HBERED SEQUE, 10HUENTIALLY / DATS 230 1 10HBERED SEGUE, 10HUENTIALLY / DATS 230 1 10HUENTARUEY DATS 230 DATA ERD / 10H THE DATA , 10HCASE DOES , 10HNOT H		EQUIVALENCE (COM (20), NLIMIT)	DATS	90
EUDIVALENCE (CONGO), INPEGNDATS 110DIMENSION ERSISDATS 120DIMENSION ERSH(5)DATS 120DATA ERSH /IOH THE SREAR, IOH PANELS AR, IOHE NOT NUMB,DATS 140DATA ERSH /IOH THE SREAR, IOH PANELS AR, IOHE NOT NUMB,DATS 160DATA ERL / IOH THE SREAR, IOH PANELS AR, IOHE NOT NUMB,DATS 160DATA ERL / IOH THE LIMIT, IOHS ARE NOT, IOH NUMBERED,DATS 1701IOHERED SEQUE, IOHNTIALLY /DATS 1700DATA ERJ / IOH THE MODAL, IOH POINTS AR, IOHE NOT NUMB,DATS 1800DATA ERJ / IOH THE MODAL, IOH POINTS AR, IOHE NOT NUMB,DATS 2100DATA ERF / IOH THE FORCE, IOHNTIALLY /DATS 2200DATA ERG / IOH THE MODAL, IOH VECTORS A, IOHARE NOT NUM,DATS 2301IOHBRERED SEQUE, IOHUENTIALLY /DATS 240DATA ERG / IOH THE DATA D, IOHEFINITIONS, IOH ARE NOT NU,DATS 2501IOHUMBERED SEG, IOHUENTIALLY /DATS 260DATA ERG / IOH THE DATA, IOHCASE DOES, IOHNOT HAVE A,DATS 260DATA STRUCT / IOHSTRUCTURE /DATS 320DATA STRUCT / IOHSTRUCTURE /DATS 330NFORCEE0IOHN ACCEPTAR, IOHLE HEADER, IOHCARD,DATS 330NFORCEE0DATS 330NFORCEE0DATS 330NJOINT=0DATS 340NJOINT=0DATS 430MJOINT = 0DATS 430MARAR = 0D		EQUIVALENCE (COM (29), NSHPAN)	DATS	100
DIMENSION ERL(3), ERF(5), ERM(5), ERB(5) DIMENSION ERSH(5) DATS 120 DATA ERSH /10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATS 150 DATA ERSH /10H THE SHEAR, 10H PANELS AR, 10HE NOT NUMB, DATA ERSH /10H THE LIMIT, 10HS ARE NOT, 10H NUMBERED, DATA ERJ / 10H THE LIMIT, 10HS ARE NOT, 10H NUMBERED, DATA ERJ / 10H THE LIMIT, 10HS ARE NOT NUM, DATS 120 DATA ERJ / 10H THE SEQUE, 10HNTIALLY / DATS 210 DATA ERF / 10H THE SEQUE, 10HNTIALLY / DATS 220 DATA ERF / 10H THE SEQUE, 10HNTIALLY / DATS 220 DATA ERF / 10H THE PARCE, 10H VECTORS A, 10HRE NOT NUM, DATS 220 DATA ERF / 10H THE BAR D, 10HENTIALLY / DATS 220 DATA ERF / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT NU, DATS 220 DATA ERF / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N, DATS 220 DATA GEAR / 10HGEAR / DATA ERF / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A, DATS 220 DATA EDHDC (DATA EDHDC / 10HN ACCEPTAR, 10HLE HEADER , 10HCARD , NJOINT=0 NLIMIT=0 NJOINT=0 NJOI		EQUIVALENCE (COM(30), INDPOM)	DATS	120
DIMENSION ERSI(5), ERT 3), ERT		DIMENSION ERI())	DATS	120
DIAMONINE ENDING ENDING AND AND AND AND AND AND AND AND AND AND		DIMENSION EDGUES	DATS	140
110HERED SEQUE, 10HNTTALLY.DATA ISL110HERED SEQUE, 10HNTTALLY.DATA ISL110HSEQUENTIAL, 10HSARE NOT, 10H NUMBERED ,DATS 180110HSEQUENTIAL, 10H YDATS 180110HERED SEQUE, 10H POINTS AR, 10HE NOT NUMB,DATS 180110HERED SEQUE, 10HNTTALLYDATS 210110HERED SEQU, 10HENTIALLYDATS 210110HERED SEQU, 10HENTIALLYDATS 230110HBERED SEQ, 10HUENTIALLYDATS 230110HBERED SEQ, 10HUENTIALLYDATS 240DATA ERF / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT NU,DATS 240DATA ERF / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N,DATS 250110HUMBERED SE, 10HQUENTIALLYDATS 260DATA SEAR / 10HGEAR/DATS 260DATA SEAR / 10HGEAR/DATS 320110HIN ACCEPTAB, 10HLE HEADER , 10HNOT HAVE A,DATS 310NJOINT=0DATS 340DATS 350NLIMIT=0DATS 350DATS 350NBAR=0DATS 380DATS 380REWIND 1DATS 380DATS 420MJONT = 0DATS 430DATS 430MJONT = 0DATS 430MJONT = 0DATS 430MJPAT = 0DATS 430		DATA ERSH /10H THE SHEAR, 10H DANELS AR, 10HE NOT NUMB.		150
DATA ERL / 10H THE LIMIT, 10HS ARE NOT, 10H NUMBERED , DATS 170 1 10HSEQUENTIAL, 10HLY / DATS 180 DATA ERJ / 10H THE NODAL, 10H POINTS AR, 10HE NOT NUMB, DATS 190 DATA ERJ / 10H THE HODAL, 10H POINTS AR, 10HE NOT NUMB, DATS 210 DATA ERF / 10H THE FORCE, 10H VECTORS A, 10HRE NOT NUM, DATS 220 DATA ERM / 10H THE MOMEN, 10HT VECTORS A, 10HRE NOT NU, DATS 220 DATA ERE / 10H THE MOMEN, 10HT VECTORS A, 10HRE NOT NU, DATS 240 DATA ERE / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N, DATS 250 1 10HMBERED SEQ 10HQUENTIALLY / DATS 260 DATA GEAR / 10HGEAR / ADATS 250 DATA STRUCT / 10HSTRUCTURE / DATS 320 NJOINT=0 NACCEPTAB, 10HLE HEADER , 10HCARD, DATS 320 NJOINT=0 NATS 320 NJOINT=0 DATS 320 NJOINT=0 NATS 320 NJOINT=0 DATS 320 NJOINT = 0 DATS 320 NJOINT = 0 DATS 320 NJOINT = 1 DATS 320 NJOINT = 0 DATS 320 NJOINT = 1 DA		1 10HERED SEQUE 10HERILLY.	DATS	160
110HSEQUENTIAL, 10HLY20		DATA ERL / 10H THE LIMIT, 10HS, ARE NOT, 10H NUMBERED .	DATS	170
DATA ERJ / 10H THE NODAL, 10H POINTS AR, 10HE NOT NUMB, 1 OHERED SEQUE; 10H NTIALLY / DATS 210 DATA ERF / 10H THE FORCE, 10H VECTORS A, 10HRE NOT NUM, DATS 210 1 10HBERED SEQU, 10H ENTIALLY / DATS 230 1 10HMBERED SEQ, 10HUENTIALLY / DATS 230 1 10HMBERED SEQ, 10HUENTIALLY / DATS 240 DATA ERB / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT NU, DATS 260 DIMENSION EDHDC16) DATA GEAR / 10HGEAR / DATS 260 DATA GEAR / 10HGEAR / DATS 270 DATA END / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A, DATS 320 NJOINT=0 NACCEPTAR, 10HLE HEADER , 10HCARD / DATS 320 NLIMIT=0 NATS 300 NFORCE=0 NATS 300 MJOINT = 0 DATS 300 MJOINT = 0 DATS 300 MJOINT = 0 DATS 300 MJOINT = 0 DATS 300 MFORCE = 0 DATS 420 DATS 420 DATS 420 DATS 420 DATS 420 DATS 420 DATS 420 DATS 510 DATS			DATS	180
110HERED SEQUE, 10HNTIALLY/DATS 200DATA ERF / 10H THE FORCE, 10H VECTORS A, 10HRE NOT NUM,DATS 210110HBERED SEQU, 10HENTIALLY /DATS 220DATA ERM / 10H THE MOMEN, 10HT VECTORS , 10HARE NOT NU,DATS 230110HMBERED SEQ, 10HUENTIALLY /DATS 250110HMBERED SE, 10HQUENTIALLY /DATS 250110HUMBERED SE, 10HQUENTIALLY /DATS 250110HUMBERED SE, 10HQUENTIALLY /DATS 270DATA ERE / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N,DATS 270DATA GEAR / 10HGEAR/DATS 280DATA EDHDC (6)DATS 280DATA EDHDC / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A,DATS 310NJOINT=0NACCEPTAB, 10HLE HEADER , 10HCARD./NJOINT=0DATS 350NBAR=0DATS 360NSHPAN = 0DATS 380REWIND 1DATS 390MJOINT = 0DATS 420MJOINT = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420 <t< td=""><td></td><td>DATA ERJ / 10H THE NODAL, 10H POINTS AR, 10HE NOT NUMB,</td><td>DATS</td><td>190</td></t<>		DATA ERJ / 10H THE NODAL, 10H POINTS AR, 10HE NOT NUMB,	DATS	190
DATA ERF / 10H THE FORCE, 10H VECTORS A, 10HRE NOT NUM, 1 10HBERED SEQU, 10HENTIALLY / DATS 220 DATA ERM / 10H THE MOMEN, 10HT VECTORS , 10HARE NOT NU, DATS 220 1 10HMBERED SEG, 10HUENTIALLY / DATS 240 DATA ERE / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N, DATS 250 1 10HUMBERED SE, 10HQUENTIALLY / DATS 260 DATA GEAR / 10HGEAR / DATS 270 DATA GEAR / 10HGEAR / DATS 270 DATA GEAR / 10HGEAR / DATS 320 1 10HN ACCEPTAB, 10HLE HEADER , 10HNOT HAVE A, DATS 300 1 10HN ACCEPTAB, 10HLE HEADER , 10HCARD, / DATS 310 NJOINT=0 NLIMIT=0 DATS 320 NLIMIT=0 NATS 320 NBAR=0 NJPNT = 0 DATS 320 MJOINT = 0 DATS 420 MFORCE = 0 DATS 420 DATS 420 DATS 520 DATS 540 DATS 540		1 10HERED SEQUE, 10HNTIALLY /	DATS	200
110HBERED SEOU, 10HENTIALLY /DATS 220DATA ERM / 10H THE MOMEN, 10HT VECTORS , 10HARE NOT NU,DATS 230110HMBERED SEG, 10HUENTIALLY /DATS 240DATA ER5 / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N,DATS 250110HUMBERED SE, 10HQUENTIALLY /DATS 260DIMENSION EDHOC(6)DATS 280DATA STRUCT / 10HSTRUCTURE /DATS 280DATA EDHDC / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A,DATS 300110HN ACCEPTAB, 10HLE HEADER , 10HCARD./ DATS 320NJOINT=0DATS 320NKIMIT=0DATS 320NBAR=0DATS 320NJOINT = 0DATS 320MJOINT = 0DATS 410MFORCE = 0DATS 420MFORCF = 0DATS 420<		DATA ERF / 10H THE FORCE, 10H VECTORS A, 10HRE NOT NUM,	DATS	210
DATA ERM / 10H THE MOMEN, 10HT VECTORS , 10HARE NOT NU, DATS 230 1 10HMBERED SEQ, 10HUENTIALLY / DATS 240 DATA ERB / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N, DATS 250 1 10HUMBERED SE, 10HQUENTIALLY / DATS 260 DIMENSION EDHDC(6) DATA GEAR / 10HGEAR / DATS 260 DATA GEAR / 10HGEAR / DATS 260 DATA EDHDC / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A, DATS 300 1 10HN ACCEPTAB, 10HLE HEADER , 10HCARD. / DATS 310 NJOINT=0 NATS 320 NJONT=0 NATS 320 NSHPAN = 0 DATS 320 MJPNT = 1 DATS 320 MJPNT = 0 DATS 320 MJPNT =		1 10HBERED SEQU, 10HENTIALLY /	DATS	220
110HMBERED SEQ, 10HUENTIALLY /DATS 240DATA ERE / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N,DATS 250110HUMBERED SE, 10HQUENTIALLY/DATS 260DIMENSION EDHDC(6)DATS 270DATA GEAR / 10HGEAR/DATS 280DATA STRUCT / 10HSTRUCTURE /DATS 200DATA EDHDC / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A,DATS 310110HN ACCEPTAB, 10HLE HEADER , 10HCARD./NJOINT=0DATS 320NLIMIT=0DATS 320NBAR=0DATS 350NBAR=0DATS 370NSHPAN = 0DATS 380REWIND 1DATS 410MLIMIT = 0DATS 420MFORCE = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORCF = 0DATS 420MFORT = 0DATS 420IFRIST = 0DATS 420IFADCD = 0DATS 420IF (FOF, 5) 12, 14DATS 52012 WRITE(6, 240)DATS 520240 FORMAT (39H1END OF JORFND OF DATA SET ON UNIT 5)DATS 530STOPDATS 540DATS 54014 CONTINUEDATS 550DATS 540DATS 540DATS 540DATS 540DATS 540DATS 540 <t< td=""><td></td><td>DATA ERM / 10H THE MOMEN, 10HT VECTORS , 10HARE NOT NU,</td><td>DATS</td><td>230</td></t<>		DATA ERM / 10H THE MOMEN, 10HT VECTORS , 10HARE NOT NU,	DATS	230
DATA ERS / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N, DATS 250 1 10HUMBERED SE, 10HQUENTIALLY/ DATS 260 DIMENSION EDHOC(6) DATS 270 DATA GEAR / 10HGEAR / DATS 270 DATA STRUCT / 10HSTRUCTURE / DATS 200 1 10HN ACCEPTAB, 10HCASE DOES, 10HNOT HAVE A, DATS 300 1 10HN ACCEPTAB, 10HLE HEADER, 10HCARD, / DATS 310 NJOINT=0 DATS 320 NLIMIT=0 DATS 320 NEORCE=0 DATS 340 NBAR=0 DATS 350 NSHPAN = 0 DATS 370 NSHPAN = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 10 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 420 MJOINT = 0 DATS 420 MJOINT = 10 DATS 420 MJOINT = 0 DATS 420 MJOINT = 10 DATS 420 DATS 4		1 10HMBERED SEQ, 10HUENTIALLY /	DATS	240
1 10HUMBERED SE, 10HQUENTIALLY/ DATS 260 DIMENSION EDHDC(6) DATS 270 DATA GEAR / 10HGEAR / DATS 280 DATA STRUCT / 10HSTRUCTURE / DATS 300 1 10HN ACCEPTAB, 10HCASE DOES, 10HNOT HAVE A, DATS 310 NJOINT=0 DATS 320 NLIMIT=0 DATS 320 NFORCE=0 DATS 350 NBAR=0 DATS 350 NSHPAN = 0 DATS 320 REWIND 1 DATS 320 MFORCE = 0 DATS 320 MJOINT = 0 DATS 370 MJOINT = 0 DATS 380 MFORCE = 0 DATS 320 MJOINT = 0 DATS 320 MFORCE = 0 DATS 420 MFORCE = 0 DATS 440 IFRIST = 0 DATS 400 IFRAST = 0 DATS 400 IFRAS (5,2		DATA ERE / 10H THE BAR D, 10HEFINITIONS, 10H ARE NOT N,	DATS	250
DIMENSION EDHDC(6) DATA GEAR / 10HGEAR / DATA STRUCT / 10HSTRUCTURE / DATA EDHDC / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A, DATS 300 1 10HN ACCEPTAB, 10HLE HEADER , 10HCARD. / DATS 310 NJOINT=0 DATS 320 NLIMIT=0 DATS 320 NGONE=0 DATS 340 NMOMNT=0 DATS 350 NBAR=0 DATS 350 NBAR=0 DATS 350 NGNFPAN = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 420 MJOINT = 0 DATS 420 MJOINT = 0 DATS 420 MFORCE = 0 DATS 420 NTS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 NTS 420		I IOHUMBERED SE, IOHQUENTIALLY/	DATS	260
DATA GEAR / 10HGEAR / DATS 280 DATA STRUCT / 10HSTRUCTURE / DATS 290 DATA EDHDC / 10H THE DATA , 10HCASE DOES , 10HNOT HAVE A, DATS 300 1 10HN ACCEPTAB, 10HLE HEADER , 10HCARD. / DATS 310 NJOINT=0 DATS 320 NFORCE=0 DATS 340 NMOMNT=0 DATS 350 NBAR=0 DATS 350 NBAR=0 DATS 370 NSHPAN = 0 DATS 370 MJOINT = 0 DATS 370 MJOINT = 0 DATS 410 MLIMIT = C DATS 410 MLIMIT = C DATS 410 MLIMIT = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 DATS 420 DATS 420 DATS 420 DATS 420 DATS 420 DATS 500 DATS 500 DA		DIMENSION EDHDC(6)	DATS	270
DATA STRUCT / IGHSTRUCTORE / DATA EDHDC / 10H THE DATA ; 10HCASE DOES ; 10HNOT HAVE A; DATS 300 NINT=0 NJOINT=0 NKIMIT=0 NKIMIT=0 NBAR=0 NBAR=0 NBAR=0 NJPNT=0 NGRCE=0 NJPNT=0 NGRCE = 0 NGRCE = 0 NJOINT = 0 NGRCE = 0 MJOINT = 0 MJOI		DATA GEAR / 10HGEAR /	DATS	280
DATA EDROC 7TOH THE DATA , TOHCASE DOES , TOHNOT HAVE A;DATS 310110HN ACCEPTAB, TOHLE HEADER , 10HCARD./DATS 310NJOINT=0DATS 320DATS 340NMOMNT=0DATS 350NBAR=0DATS 360NJPNT=0DATS 370NSHPAN = 0DATS 370MJOINT = 0DATS 370MJOINT = 0DATS 410MLIMIT = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 420MFORCE = 0DATS 500MFORCE = 0DATS 500IFRIST = 0DATS 50010READ (5,200) CARDDATS 50011WRITE(6, 240)DATS 510240FORMAT (39HIEND OF JORFND OF DATA SET ON UNIT 5)DATS 530STOPDATS 550DATS 55014CONTINUEDATS 550NTS 540DATS 550NTS 540DATS 550NTS 540DATS 550NTS 540DATS 550STOPDATS 550NTS 540DATS 550 <td></td> <td>DATA STRUCT / IOHSTRUCTURE /</td> <td>DATS</td> <td>290</td>		DATA STRUCT / IOHSTRUCTURE /	DATS	290
NJOINT=0 DATS 320 NLIMIT=0 DATS 330 NFORCE=0 DATS 340 NMOMNT=0 DATS 350 NBAR=0 DATS 360 NJPNT=0 DATS 370 NSHPAN = 0 DATS 390 MJOINT = 0 DATS 390 MJOINT = 0 DATS 400 MJPNT = 0 DATS 410 MLIMIT = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MMOMNT = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MBAR = 0 DATS 420 MBAR = 0 DATS 440 IFRIST = 0 DATS 450 IFRADCD = 0 DATS 450 IFRAT = 0 DATS 450 IFRAT = 0 DATS 500 IFRAT = 0 DATS 510 DATS 500 DATS 510 DATS 500 DATS 510 DATS 520 DATS 520 </td <td></td> <td>1 10H HE DATA , IOHCASE DOES , IOHNOT HAVE A,</td> <td>DATS</td> <td>310</td>		1 10H HE DATA , IOHCASE DOES , IOHNOT HAVE A,	DATS	310
NOTITIED DATS 320 NEMITIED DATS 330 NFORCE=0 DATS 350 NBAR=0 DATS 350 NBAR=0 DATS 370 NSHPAN = 0 DATS 370 REWIND 1 DATS 390 MJOINT = 0 DATS 390 MJOINT = 0 DATS 410 MLMIT = 0 DATS 420 MFORCE = 0 DATS 420 MFORCF = 0 DATS 420 MFORCF = 0 DATS 420 MFORCF = 0 DATS 420 MFORCF = 0 DATS 420 MFORCF = 0 DATS 420 MFORCF = 0 DATS 420 MSHPAN = 0 DATS 420 MSHPAN = 0 DATS 440 MBAR = 0 DATS 450 MSHPAN = 0 DATS 450 IFRIST = 0 DATS 450 IFRADCD = 0 DATS 450 IFRADCD = 0 DATS 510 IF (FOF, 5) 12, 14 DATS 510 12 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JORFND OF DATA SET ON UNIT 5) DATS 530 STOP DATS 540 14 CONTINUE DATS 550		NIOINT=0	DATS	320
NETRICE=0 DATS 340 NMOMNT=0 DATS 350 NBAR=0 DATS 360 NJPNT=0 DATS 370 NSHPAN = 0 DATS 380 REWIND 1 DATS 390 MJOINT = 0 DATS 400 MJOINT = 0 DATS 410 MLIMIT = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MBAR = 0 DATS 450 MSHPAN = 0 DATS 450 IFRIST = 0 DATS 450 IFRST = 0 DATS 500 IF (FOF, 5) 12, 14 DATS 500 I2 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JOB===FND OF DATA SET ON UNIT 5) DATS 520 240 FORMAT (39HIEND OF JOB===FND OF DATA SET ON UNIT 5) DATS 540 DATS 540 DATS 550 12 WRITE(6, 240) DATS 550 STOP DATS 540 14 CONTINUE DATS 550 <			DATS	330
NMOMNT=0 DATS 350 NBAR=0 DATS 360 NJPNT=0 DATS 370 NSHPAN = 0 DATS 380 REWIND 1 DATS 390 MJOINT = 0 DATS 400 MJPNT = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORCE = 0 DATS 420 MFORT = 0 DATS 420 MBAR = 0 DATS 420 MBAR = 0 DATS 420 MSHPAN = 0 DATS 450 IFRIST = 0 DATS 450 IFRST = 0 DATS 450 IF (FOF, 5) 12, 14 DATS 500 IF (FOF, 5) 12, 14 DATS 510 I2 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JORFND OF DATA SET ON UNIT 5) DATS 520 ACONTINUE DATS 550 DATS 520 DATS 520		NEOREE	DATS	340
NBAR=0 DATS 360 NJPNT=0 DATS 370 NSHPAN = 0 DATS 380 REWIND 1 DATS 390 MJOINT = 0 DATS 400 MJPNT = 0 DATS 410 MLIMIT = 0 DATS 420 MFORCE = 0 DATS 420 MMOMNT = 0 DATS 420 MBAR = 0 DATS 440 MBAR = 0 DATS 450 MSHPAN = 0 DATS 450 IFRIST = 0 DATS 450 IFRST = 0 DATS 460 IFRST = 0 DATS 450 DATS 450 DATS 450 DATS 500 DATS 500 IF (FOF, 5) 12, 14 DATS 510 DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP DATS 540 ACONTINUE DATS 530		MMONNT=0	DATS	350
NJPNT=0 DATS 370 NSHPAN = 0 DATS 380 REWIND 1 DATS 390 MJOINT = 0 DATS 400 MJPNT = 0 DATS 400 MLIMIT = 0 DATS 420 MFORCE = 0 DATS 420 MMOMNT = 0 DATS 420 MBAR = 0 DATS 420 MSHPAN = 0 DATS 440 IFRIST = 0 DATS 450 IFRST = 0 DATS 450 IF (FOF, 5) 12, 14 DATS 500 I2 WRITE(6, 240) DATS 510 240 FORMAT (39H1END OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP DATS 550 14 CONTINUE DATS 550		NBAR=0	DATS	360
NSHPAN = 0 DATS 380 REWIND 1 DATS 390 MJOINT = 0 DATS 400 MJPNT = 0 DATS 410 MLIMIT = 0 DATS 420 MFORCE = 0 DATS 420 MMOMNT = 0 DATS 430 MMOMNT = 0 DATS 440 MBAR = 0 DATS 450 MSHPAN = 0 DATS 450 IFRIST = 0 DATS 460 IFRST = 0 DATS 480 IFRST = 0 DATS 490 10 READ (5,200) CARD DATS 500 IF (FOF, 5) 12, 14 DATS 510 12 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP DATS 550 14 CONTINUE DATS 550		NJPNT=0	DATS	370
REWIND 1 DATS 390 MJOINT = 0 DATS 400 MJPNT = 0 DATS 410 MLIMIT = 0 DATS 420 MFORCE = 0 DATS 420 MMOMNT = 0 DATS 430 MBAR = 0 DATS 440 MBAR = 0 DATS 450 MSHPAN = 0 DATS 450 IFRIST = 0 DATS 460 IFRST = 0 DATS 470 IFRST = 0 DATS 480 IFRST = 0 DATS 480 IFRST = 0 DATS 500 IF (FOF, 5) 12, 14 DATS 500 I2 WRITE(6, 240) DATS 510 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 ATS 540 DATS 540		NSHPAN ≈ 0	DATS	380
MJOINT = 0 DATS 400 MJPNT = 0 DATS 410 MLIMIT = 0 DATS 420 MFORCE = 0 DATS 430 MMOMNT = 0 DATS 440 MBAR = 0 DATS 450 MSHPAN = 0 DATS 450 IFRIST = 0 DATS 460 IFRST = 0 DATS 470 IFRST = 0 DATS 480 IFRST = 0 DATS 480 IFRST = 0 DATS 500 IV READ (5:200) CARD DATS 500 IF (FOF, 5) 12; 14 DATS 510 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP DATS 540 14 CONTINUE DATS 540		REWIND 1	DATS	390
MJPNT = 0 DATS 410 MLIMIT = 0 DATS 420 MFORCE = 0 DATS 430 MMOMNT = 0 DATS 440 MBAR = 0 DATS 450 MSHPAN = 0 DATS 450 MSHPAN = 0 DATS 460 IFRIST = 0 DATS 460 IFRIST = 0 DATS 470 IBADCD = 0 DATS 480 IFRST = 0 DATS 480 IFRST = 0 DATS 500 IV READ (5:200) CARD DATS 500 IF (FOF, 5) 12; 14 DATS 510 DATS 520 DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP DATS 540 14 CONTINUE DATS 540		MJOINT = 0	DATS	400
MLIMIT = 0 DATS 420 MFORCE = 0 DATS 430 MMOMNT = 0 DATS 440 MBAR = 0 DATS 450 MSHPAN = 0 DATS 450 IFRIST = 0 DATS 460 IFRST = 0 DATS 470 IFRST = 0 DATS 480 IFRST = 0 DATS 480 IFRST = 0 DATS 490 DATS 500 DATS 500 IF (FOF, 5) 12, 14 DATS 510 12 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 14 CONTINUE DATS 550		MJPNT = O	DATS	410
MFORCE = 0 DATS 430 MMOMNT = 0 DATS 440 MBAR = 0 DATS 450 MSHPAN = 0 DATS 450 IFRIST = 0 DATS 470 IBADCD = 0 DATS 480 IFRST = 0 DATS 480 IFRST = 0 DATS 490 10 READ (5:200) CARD DATS 500 IF (FOF, 5) 12; 14 DATS 510 12 WRITE(6; 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 540 14 CONTINUE DATS 550		MLIMIT = 0	DATS	420
MMOMNT = 0 DATS 440 MBAR = 0 DATS 450 MSHPAN = 0 DATS 460 IFRIST = 0 DATS 470 IBADCD = 0 DATS 480 IFRST = 0 DATS 480 IFRST = 0 DATS 480 IFRST = 0 DATS 500 IV READ (5:200) CARD DATS 500 IF (FOF, 5) 12; 14 DATS 510 12 WRITE(6; 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP DATS 540 14 CONTINUE DATS 540		MFORCE = 0	DATS	430
MBAR = 0 DATS 450 MSHPAN = 0 DATS 460 IFRIST = 0 DATS 470 IBADCD = 0 DATS 480 IFRST = 0 DATS 480 IFRST = 0 DATS 500 10 READ (5,200) CARD DATS 500 IF (FOF, 5) 12, 14 DATS 510 12 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP 14 CONTINUE DATS 550		MMOMNT = 0	DATS	440
MSHPAN = 0 DATS 460 IFRIST = 0 DATS 470 IBADCD = 0 DATS 480 IFRST = 0 DATS 490 10 READ (5,200) CARD DATS 500 IF (FOF, 5) 12, 14 DATS 510 12 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 540 14 CONTINUE DATS 550		MBAR = 0	DATS	450
IFRIST = 0 DATS 470 IBADCD = 0 DATS 480 IFRST = 0 DATS 490 10 READ (5,200) CARD DATS 500 IF (FOF, 5) 12, 14 DATS 510 12 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP 14 CONTINUE		MSHPAN = 0	DATS	460
1BADED = 0 DATS 480 IFRST = 0 DATS 490 10 READ (5,200) CARD DATS 500 IF (FOF, 5) 12, 14 DATS 510 12 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP 14 CONTINUE		IFRIST = 0	DATS	4/0
10 READ (5,200) CARD DATS 500 10 READ (5,200) CARD DATS 510 17 WRITE(6, 240) DATS 520 240 FORMAT (39H1END OF JOBFND OF DATA SET ON UNIT 5.) DATS 520 240 FORMAT (39H1END OF JOBFND OF DATA SET ON UNIT 5.) DATS 530 240 FORMAT (39H1END OF JOBFND OF DATA SET ON UNIT 5.) DATS 530 240 FORMAT (39H1END OF JOBFND OF DATA SET ON UNIT 5.) DATS 530 240 FORMAT (39H1END OF JOBFND OF DATA SET ON UNIT 5.) DATS 540 240 FORMAT (39H1END OF JOBFND OF DATA SET ON UNIT 5.) DATS 540 240 FORMAT (39H1END OF JOBFND OF DATA SET ON UNIT 5.) DATS 540			DAIS	480
IO READ (5,200) CARD DATS 500 IF (FOF, 5) 12, 14 DATS 510 12 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP 14 CONTINUE	10	$\frac{1}{1}$		490
12 WRITE(6, 240) DATS 520 240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5.) DATS 530 STOP DATS 540 14 CONTINUE DATS 550	τu	KEAU (09200) CAKU IE (EOE, 5) 12, 14		500
240 FORMAT (39HIEND OF JOBFND OF DATA SET ON UNIT 5) DATS 530 STOP 14 CONTINUE	17	WRITE(6. 240)	DATS	520
STOP 14 CONTINUE	240	FORMAT (39HIEND OF JORFND OF DATA SET ON UNIT 5)	DATS	520
14 CONTINUE DATS 550	2 4 .'	CTOD	DATE	540
	14	CONTINUÉ	DATS	550

```
IF ( IFRST .NE. 0 ) GO TO 17
                                                                         DATS 560
     WRITE (6,190)
                                                                         DATS 570
      IFRST = 1
                                                                         DATS 580
   17 CONTINUE
                                                                         DATS 590
      IF ( IFRIST .NE. 0 ) GO TO 16
                                                                         DATS 600
С
                                                                         DATS 610
Ċ
          CHECK FOR DATA SET HEADER CARD (GEAR) OR (STRUCTURE)
                                                                         DATS 620
С
               LEFT JUSTIFIED IN COL. 1-10
                                                                         DATS 630
С
                                                                         DATS 640
      INDPGM = 1
                                                                         DATS 650
      IF ( CARD(1) . EQ. GEAR ) GO TO 13
                                                                         DATS 660
      INDPGM = 0
                                                                         DATS 670
      IF ( CARD(1) •EQ• STRUCT) GO TO 15
                                                                         DATS 680
      IF ( IBADCD .EQ. 0 ) CALL ERPNT1( EDHDC, 6, 0 )
                                                                         DATS 690
      IBADCD = 1
                                                                         DATS 700
      WRITE(6,230) CARD
                                                                         DATS 710
      GO TO 10
                                                                         DATS 720
   13 CONTINUE
                                                                         DATS 730
      IF ( IBADCD .NE. 0 ) WRITE(6,190)
                                                                         DATS 740
      RETURN
                                                                         DATS 750
   15 CONTINUE
                                                                         DATS 760
      IF ( IBADCD .NE. 0 ) WRITE(6,190)
                                                                         DATS 770
      WRITE(6,192)
                                                                         DATS 780
                                                                         DATS 790
      IFRIST = 1
      GO TO 10
                                                                         DATS 800
   16 CONTINUE
                                                                         DATS 810
      WRITE (6,230) CARD
                                                                         DATS 820
С
                                                                         DATS 830
С
               CONVERT CODE AND SUBSCRIPT TO INTEGER
                                                                         DATS 840
С
                                                                         DATS 850
      DECODE (5,210,CARD(1)) ICODE, INDEX
                                                                         DATS 860
С
                                                                         DATS 870
С
          ANY CODE NO. . GT. 7 IS CONSIDERED AN END OF RECORD
                                                                         DATS 880
      IF (ICODE.GT.7) GO TO 110
                                                                         DATS 890
                                                                         DATS 900
C
          IF COLUMN ONE IS BLANK OR ZERO CONSIDER IT A COMMENT
С
                                                                         DATS 910
      IF (ICODE.EQ.0) GO TO 10
                                                                         DATS 920
С
                                                                         DATS 930
      WRITE(1,300) ICODE
                                                                         DATS 940
      WRITE(1,200) CARD
                                                                         DATS 950
      GO TO (20,40,50,60,70,80,90), ICODE
                                                                         DATS 960
20
                                                                         DATS 970
      NJOINT=NJOINT+1
      MJOINT = MAXO( MJOINT, INDEX )
                                                                         DATS 980
      GO TO 10
                                                                         DATS 990
40
      NJPNT=NJPNT+1
                                                                         DATS1000
      MJPNT = MAXO(MJPNT, INDEX)
                                                                         DATS1010
      GO TO 10
                                                                         DATS1020
50
      NLIMIT=NLIMIT+1
                                                                         DATS1030
      MLIMIT = MAXO( MLIMIT, INDEX )
                                                                         DATS1040
      GO TO 10
                                                                          DATS1050
60
      NFORCE=NFORCE+1
                                                                          DATS1060
      MFORCE = MAXO( MFORCE, INDEX )
                                                                         DATS1070
      GO TO 10
                                                                          DATS1080
70
      NMOMNT=NMOMNT+1
                                                                         DATS1090
      MMOMNT = MAXO( MMOMNT, INDEX )
                                                                         DATS1100
      GO TO 10
                                                                          DATS1110
```

```
80
      NBAR=NBAR+1
                                                                            DATS1120
      MBAR
            = MAXO( MBAR , INDEX )
                                                                            DATS1130
      GO TO 10
                                                                            DATS1140
   90 CONTINUE
                                                                            DATS1150
      MSHPAN = MAXO(MSHPAN, INDEX)
                                                                            DATS1160
      NSHPAN = NSHPAN + 1
                                                                            DATS1170
      GO TO 10
                                                                            DATS1180
  110 CONTINUE
                                                                            DATS1190
С
          CHECK TO SEE IF INPUT CARDS ARE SEQUENCED PROPERLY
                                                                            DATS1200
С
                    (THIS WILL NOT CATCH ALL KEY PUNCH ERRORS)
                                                                            DATS1210
      NJPNT = NJPNT + NJOINT
                                                                            DATS1220
      MJPNT = MAXO(MJPNT, MJOINT)
                                                                            DATS1230
            MJPNT •NE• NJPNT ) CALL ERPNT1( ERJ, 5, -1 )
      IF (
                                                                            DATS1240
            MLIMIT •NE• NLIMIT ) CALL ERPNT1( ERL, 5, -1 )
      1F
         (
                                                                            DATS1250
            MFORCE •NE• NFORCE ) CALL ERPNT1( ERF, 5, -1 )
MMOMNT •NE• NMOMNT ) CALL ERPNT1( ERM, 5, -1 )
      IF
         1
                                                                            DATS1260
      IF
                                                                            DATS1270
         (
                    •NE• NBAR ) CALL ERPNT1( ERB, 5, -1 )
      IF (
                                                                            DATS1280
            MBAR
      IF ( MSHPAN .NE. NSHPAN ) CALL ERPNT1( ERSH, 5, -1 )
                                                                            DATS1290
                                                                            DATS1300
C
Ĉ
           IF A FATAL ERROR HAS OCCURRED, STOP
                                                                            DATS1310
С
                                                                            DATS1320
      CALL ERPNT2
                                                                            DATS1330
С
                                                                            DATS1340
      END FILE 1
                                                                            DATS1350
      RETURN
                                                                            DATS1360
C
                                                                            DATS1370
  190 FORMAT ( 1H1, 43X 45HSTRUCTURAL ANALYSIS PROGRAM --- LEGGED LANDERDATS1380
     1/37X60HMASTER AGREEMENT, CONTRACT NAS1-8137, TASK ORDER NUMBER FIVDATS1390
     2E/45X44HMCDONNELL DOUGLAS ASTRONAUTICS COMPANY, EAST
                                                                            DATS1400
     3 /// )
                                                                            DATS1410
  192 FORMAT
                                                                            DATS1420
                 1
     337H STRUCTURAL ANALYSIS DATA - CARD CODE, //28X
                                                            27HBLANK - 0
                                                                            DATS1430
             COMMENTS ,/32X1H1,13X,23HNODAL POINT DEFINITIONS,/32X1H2,13XDATS1440
     4
     5,16HREFERENCE POINTS,/32X1H3,13X,33HNODAL POINT RESTRAINT DEFINITIDATS1450
     60NS,/32X1H4,13X,13HFORCE VECTORS,/32X1H5,13X,15HMOMENT VECTORS ,/3DATS1460
     7 2X1H6, 13X, 15HBAR DEFINITIONS, / 32X1H7,13X
                                                                            DATS1470
     8 23HSHEAR PANEL DEFINITIONS,/32X1H8,13X,24HFORMATED-DATA TERMINATDATS1480
     90R / )
                                                                            DATS1490
200
      FORMAT (8A10)
                                                                            DATS1500
  210 FORMAT ( 11, 14 )
                                                                            DATS1510
220
      FORMAT (1XI4)
                                                                            DAT$1520
      FORMAT (1X8A10)
230
                                                                            DATS1530
  300 FORMAT( I4 )
                                                                            DATS1540
      END
                                                                            DATS1550
```

```
RDDA
 SUBROUTINE RDDATA
                                                                            10
 COMMON
          NJOINT, NFORCE, NMOMNT, NBAR , NJPNT, ITINJO, IBAR
                                                                     RDDA
                                                                            20
          NROW , INDSFL, INDSFG, INDISL, INDITR, ERRTOL, RELAXF
                                                                      RDDA
                                                                            30
1
2
        , INDRLX, INDWKT, INDRKT, INDPLS, MINRST, NLIMIT, NRWTOR
                                                                     RDDA
                                                                            40
                                                                            50
3
        , IREDTO, NEIGVL, INDNMA, INDWNM, TMAX , ISFDIM, INDWTS
                                                                     RDDA
4
        , NSHPAN, INDPGM
                                                                     RDDA
                                                                            60
     TOTAL LENGTH WITH STRSTF(0) .EQ. 4049
                                                                      RDDA
                                                                            70
 COMMON / CMAIN /
                                                                      RDDA
                                                                            80
1
     RJXYZ (100,3), IJRFM ( 74,6), IJALFM( 74,2), FRCVCT( 74,3),
                                                                      RDDA
                                                                            90
     RMTVCT( 74,3), IBARP (
                              130), IBARQ (
                                                                      RDDA 100
2
                                             130), IBARR (
                                                             130),
3
     KS
              130), BAREA (
                              130), BARIN (
                                              130), BARIT (
                                                                      RDDA 110
           (
                                                             130),
                                                                      RDDA 120
4
     BARJ
              130), BARYM (
                              130), BARSM (
                                              130), RKN
                                                             130),
           {
                                                          (
5
     RKT
              130), SOLVEC(
                              444), RLIMTU(
                                               88), RLIMTL(
                                                              88),
                                                                      RDDA 130
           (
     DPRTIT(
                                                                      RDDA 140
               -88), IROWL (
                              445), STRSTF(1)
6
 COMMON/SP/IAPAN(30), IBPAN(30), ICPAN(30), IDPAN(30),
                                                                     RDDA 150
                                                                      RDDA 160
1
           VSPAN(30), ESPAN(30), TSPAN(30)
 DIMENSION STIFML(12,12), TRANSM(3,3), AMASS(102), IRWKP(103)
                                                                      RDDA 170
 EQUIVALENCE ( KS(1), STIFML(1,1)), ( BARIT(1), TRANSM(1,1))
                                                                      RDDA 180
 EQUIVALENCE ( RLIMTU(1), AMASS(1)), ( RLIMTL(15), IRWKP(1 ))
                                                                      RDDA 190
 DIMENSION ERORD(4), ERRWKP(5)
                                                                      RDDA 200
 DIMENSION ERLT(5)
                                                                      RDDA 210
 DIMENSION ERRED(4)
                                                                      RDDA 220
                                                                      RDDA 230
 DIMENSION ERFROW(5)
 DIMENSION ERLN(5)
                                                                      RDDA 240
 DIMENSION ERPQ(4)
                                                                      RDDA 250
 DIMENSION ERBN(4), ERJNL(4)
                                                                      RDDA 260
 DIMENSION ERJN( 5), ERJR( 7), ERFR( 6), ERMT(6)
                                                                      RDDA 270
                                                                      RDDA 280
 DIMENSION EREGVL(8)
 DATA EREGVL/ 10H AN EXCESS, 10HIVE NUMBER, 10H OF EIGENV,
                                                                      RDDA 290
                                                                      RDDA 300
              10HALUES HAVE, 10H BEEN REQU, 10HESTED BY U,
1
2
              10HSE OF NEIG, 10HVL.
                                                         1
                                                                      RDDA 310
 DATA ERRED / 40H IREDTO CAN NOT BE GREATER THAN NROW
                                                                      RDDA 320
 DATA ERORD /40H IREDTO CAN NOT BE LARGER THAN 102
                                                                      RDDA 330
 DATA ERRWKP/50H THE IRWKP ARRAY IS NOT FILLED IN ASCENDING ORDER /RDDA 340
 DATA ERPQ / 35H A ZERO LENGTH BAR IS NOT PERMITTED
                                                                      RDDA 350
                                                        1
 DATA ERBN/ 10H THERE MUS, 10HT BE AT LE, 10HAST DNE BA,
                                                                      RDDA 360
1
                                                                      RDDA 370
            10HR
 DATA ERJNL/10H THERE MUS, 10HT BE AT LE, 10HAST TWO NO,
                                                                      RDDA 380
            10HDAL POINTS/
                                                                      RDDA 390
1
 DATA ERLN /50H A RESTRAINT INDICATOR IS OUT OF BOUNDS
                                                                     /RDDA 400
 DATA ERFROW /45H INPUT DATA DISAGREES WITH MATRICES ON TAPE9
                                                                     /RDDA 410
 DATA ERLT /50H LOWER REACTION LIMIT CONFLICTS WITH UPPER LIMIT
                                                                     /RDDA 420
 DATA ERJN /10H A BAR DEF, 10HINITION US, 10HES AN UNDE, 10HFINED NODARDDA 430
                                                                      RDDA 440
           ,10HL POINT
1
                         - 7
 DATA ERJR /10H A BAR DEF, 10HINITION US, 10HES AN UNDE, 10HFINED POINRDDA 450
            ,10HT TO DEFIN,10HE THE BEND,10HING PLANES /
                                                                      RDDA 460
1
 DATA ERFR /10H A NODAL P,10HOINT IS AS,10HSOCIATED W,10HITH AN UNDRDDA 470
            ,10HEFINED FOR,10HCE VECTOR /
                                                                      RDDA 480
1
 DATA ERMT /10H A NODAL P,10HOINT IS AS,10HSOCIATED W,10HITH AN UNDRDDA 490
            ,10HEFINED MOM,10HENT VECTOR/
                                                                      RDDA 500
1
 DIMENSION ERSP3(3), ERSP4(4), ERSP5(5), ERSP6(4), ERSP7(4),
                                                                      RDDA 510
     ERSP8(4), ERSP9(4)
                                                                      RDDA 520
 DATA ERSP3 / 10H ONLY 130 , 10HBARS ARE A, 10HLLOWED.
                                                                      RDDA 530
 DATA ERSP4 / 10H ONLY 74 , 10HBAR NODE P, 10HOINTS ARE ,
                                                                      RDDA 540
                                                                      RDDA 550
              10HALLOWED.
1
                             1
```

С

RDDA 560 RDDA 570 DATA ERSP9 / 10H ONLY 30 , 10HSHEAR PANE, 10HLS ARE ALL, 1 10HOWED. RDDA 580 26 , 10HREFERENCE , 10H NODE POIN, DATA ERSP5 / 10H ONLY RDDA 590 10HTS ARE ALL, 10HOWED. 1 1 RDDA 600 DATA ERSP6 / 10H ONLY 88 , 10HLIMIT CARD, 10HS ARE ALLO, RDDA 610 10HWED. 1 1 74 , 10HFORCE VECT, 10HORS ARE AL, DATA ERSP7 / 10H ONLY RDDA 620 RDDA 630 10HLOWED. 1 1 DATA ERSP8 / 10H ONLY 74 , 10HMOMENT VEC, 10HTORS ARE A, RDDA 640 RDDA 650 10HLLOWED. 1 1 NAMELIST / INDATA / INDSFL, INDSFG, IND1SL, INDITR, ERRTOL, SOLVEC RDDA 660 , RELAXF, INDRLX RDDA 670 1 RDDA 680 , INDRKT, INDWKT, INDPLS, MINRST 2 , INDNMA, IRWKP, AMASS, IREDTO, INDWNM, TMAX RDDA 690 3 RDDA 700 4 , ISFDIM, INDWTS ,NEIGVL RDDA 710 RDDA 720 RDDA 730 CHECK INPUT DATA AGAINST VARIABLE DIMENSION LIMITS RDDA 740 IF (NJOINT .GT. 74) CALL ERPNT1(ERSP4, 4, -1) RDDA 750 IF (NJPNT .GT. 100) CALL ERPNT1(ERSP5, 5, -1) IF (NLIMIT .GT. 88) CALL ERPNT1(ERSP6, 4, -1) RDDA 760 IF (NFORCE .GT. IF (NMOMNT .GT. 74) CALL ERPNT1(ERSP7, 4, -1) RDDA 770 IF (NMOMNT •GT• 74) CALL ERPNT1(ERSP8, 4, -1) IF (NSHPAN •GT• 30) CALL ERPNT1(ERSP9, 4, -1) RDDA 780 RDDA 790 IF (NBAR .GT. 130) CALL ERPNT1(ERSP3, 3, -1) RDDA 800 CALL ERPNT2 RDDA 810 REWIND 1 **RDDA 820** 1 READ(1,135) ICODE **RDDA 830** 135 FORMAT (I4) **RDDA 840** RDDA 850 IF (EOF, 1) 58, 4 **4 CONTINUE** RDDA 860 RDDA 870 GO TO (21, 22, 23, 24, 25, 26, 27), ICODE 21 CONTINUE RDDA 880 READ CODE 1 DATA **RDDA 890** C READ(1,140) INDX, (RJXYZ(INDX,J), J=1,3), (IJRFM(INDX,J), J=1,6), RDDA 900 RDDA 910 (IJALFM(INDX,J),J=1,2) 1 GO TO 1 RDDA 920 22 CONTINUE **RDDA 930** С RDDA 940 READ CODE 2 DATA READ(1,140) INDX, (RJXYZ(INDX,J), J=1,3) RDDA 950 GO TO 1 **RDDA 960 RDDA 970** 23 CONTINUE С CODE 3 DATA **RDDA 980** READ READ(1,140) INDX, RLIMTU(INDX), RLIMTL(INDX), DPRTIT(INDX) RDDA 990 GO TO 1 RDDA1000 24 CONTINUE RDDA1010 С READ CODE 4 DATA RDDA1020 READ(1,140) INDX, (FRCVCT(INDX,J),J=1,3) RDDA1030 GO TO 1 RDDA1040 25 CONTINUE RDDA1050 C READ CODE 5 DATA RDDA1060 READ(1,140) INDX, (RMTVCT(INDX,J),J=1,3) RDDA1070 GO TO 1 RDDA1080 26 CONTINUE RDDA1090 С READ CODE 6 DATA RDDA1100 READ(1,160) INDX, IBARP(INDX), IBARQ(INDX), IBARR(INDX), RDDA1110

ţ.

C C

(INDX), BAREA(INDX), BARIN(INDX), RDDA1120 1 KS ž BARIT(INDX), BARJ (INDX), BARYM(INDX), RDDA1130 . 3 BARSM(INDX), RKN (INDX), RKT (INDX) RDDA1140 GO TO 1 RDDA1150 С READ CODE 7 DATA **RDDA1160** RDDA1170 27 CONTINUE READ(1,150) INDX, IAPAN(INDX), IBPAN(INDX), ICPAN(INDX), IDPAN(INDX), VSPAN(INDX), ESPAN(INDX), TSPAN(INDX) RDDA1180 RDDA1190 1 GO TO 1 RDDA1200 58 CONTINUE RDDA1210 RDDA1220 С Ċ CHECK BAR DEFINITION RDDA1230 С RDDA1240 DO 60 I=1,NBAR RDDA1250 IF (IBARP(I).GT.NJOINT.OR.IBARP(I).LT.1) CALL ERPNT1 (ERJN,5,-1) RDDA1260 IF (IBARQ(I).GT.NJOINT.OR.IBARQ(I).LT.1) CALL ERPNT1 (ERJN,5,-1) RDDA1270 (IBARP(I).EQ.IBARQ(I)) CALL ERPNT1 (ERPQ,4,-1) RDDA1280 IF IF (IBARR(I).GT.NJPNT) CALL ERPNT1 (ERJR,7,-1) RDDA1290 60 C C RDDA1300 CHECK JOINT DEFINITIONS FOR MISSING FORCE, MOMENT VECTORS, RDDA1310 C AND RESTRAINT INFORMATION RDDA1320 RDDA1330 DO 80 I=1,NJOINT RDDA1340 DO 70 J=1,6 K=IJRFM(I,J) RDDA1350 IF (IABS(K).GT.NLIMIT) CALL ERPNT1 (ERLN,5,-1) RDDA1360 RDDA1370 IF (K.LE.0) GO TO 70 IF (RLIMTU(K).LE.RLIMTL(K)) CALL ERPNT1 (ERLT,5,-1) RDDA1380 RDDA1390 70 CONTINUE IF (IJALFM(I,1).GT.NFORCE.OR.IJALFM(I,1).LT.0) CALL ERPNT1 (ERFR.6RDDA1400 1, -1RDDA1410 IF (IJALFM(I,2).GT.NMOMNT.OR.IJALFM(I,2).LT.0) CALL ERPNT1 (ERMT,6RDDA1420 RDDA1430 1,-1) 80 CONTINUE RDDA1440 RDDA1450 ¢ RDDA1460 ¢ CALL SHEAR PANEL SUBROUTINE RDDA1470 ¢ PDDA1480 CALL SHRPAN IF (NBAR .GT. 130) CALL ERPNT1(ERSP3, 3, -1) RDDA1490 RDDA1500 CALL ERPNT2 C C RDDA1510 RDDA1520 INITIALIZE DATA С RDDA1530 RDDA1540 NROW=6*NJOINT RDDA1550 INDITR=NROW*3 RDDA1560 INDPLS=0 RDDA1570 INDWKT=0 RDDA1580 INDRKT=0 RDDA1590 MINRST=6 RDDA1600 IND1SL=0 RDDA1610 ERRTOL=.0001 RDDA1620 INDSFG=1 RDDA1630 INDSFL=1 RDDA1640 INDWTS = 0RDDA1650 INDRLX = 2RDDA1660 $RELAXF = 1_{\circ}$ ISFDIM = 12904RDDA1670

85	DO 85 I = 1, NROW SOLVEC(I) = 0.0	RDDA1680 RDDA1690
	INDNMA=0	RDDA1700
	IREDTO=NROW	RDDA1710
	INDWNM=0	RDDA1720
	NEIGVL = 5	RDDA1730
	TMAX=9999•	RDDA1740
С		RDDA1750
č	READ INDICATORS AND CONTROL DATA IN BY NAMELIST	RDDA1760
С		RDDA1770
	READ (5,INDATA)	RDDA1780
	WRITE(6,210) INDRKT, INDSFG, INDSFL, INDWKT, INDWTS, ISFDIM	RDDA1790
	IF $(INDNMA \bullet EQ \bullet 0)$ GO TO 100	RDDA1800
С	CHECK NORMAL MODE ANALYSIS DATA	RDDA1810
С	NEIGVL MUST BE A MULTIPLE OF 5	RDDA1820
	I = NEIGVL/5	RDDA1830
	IF (I $*5 \cdot NE \cdot NE \cdot IGVL$) NEIGVL = $5*(I+1)$	RDDA1840
	WRITE(6,230) INDNMA, INDWNM, IREDTO, NEIGVL	RDDA1850
	WRITE(6,240) (AMASS(I), I=1, IREDTO)	RDDA1860
	NEIGVL = NEIGVL + 6	RDDA1870
	IF (NEIGVL .GT.IREDTO) CALL ERPNT1(EREGVL, 8, -1)	RDDA1880
	IF (IREDT0.GT.102) CALL ERPNT1 (ERORD.41)	RDDA1890
	IF (IREDTO EQ. NROW) GO TO 100	RDDA1900
	WRITE(6,250) (IRWKP(I),I=1,IREDTO)	RDDA1910
	IF (IREDTO.GT.NROW) CALL ERPNT1 (ERRED.41)	RDDA1920
	J=IREDTO-1	RDDA1930
	DO 90 I=1,J	RDDA1940
90	IF (IRWKP(I).GE.IRWKP(I+1)) CALL ERPNT1 (FRRWKP.51)	RDDA1950
100	CONTINUE	RDDA1960
	IE (INDRKT-FR-0) GO TO 130	RDDA1970
c		RDDA1980
č	READ STRUCTURAL STIFFNESS MATRIX DATA AND LOCAL STIFFNESS	RDDA1990
С	AND TRANSFORMATION MATRICES	RDDA2000
	REWIND 9	RDDA2010
	READ (9) NROW1	RDDA2020
	IF (NROW•NE•NROW1) CALL ERPNT1 (ERFROW•5•1)	RDDA2030
	NROW1=NROW1+1	RDDA2040
	READ (9) (IROWL(I),I=1,NROW1)	RDDA2050
	J=1	RDDA2060
	DO 110 I=1.NROW	RDDA2070
	I I = I ROWL (I+1) - 1	RDDA2080
	READ (9) $(STRSTF(J), J=JJ, II)$	RDDA2090
110	JJ=II+1	RDDA2100
	READ (9) STRSTF(JJ)	RDDA2110
	IF (INDNMA • NE • 0) GO TO 130	RDDA2120
с		RDDA2130
С	FILE 2 IS NOT NEEDED IF OVERLAY 3 IS NOT CALLED	RDDA2140
С		RDDA2150
	REWIND 2	RDDA2160
	DO 120 I=1,NBAR	RDDA2170
	READ (9) (TRANSM(J,1), J=1,9)	RDDA2180
	WRITE (2) (TRANSM(J,1), J=1,9)	RDDA2190
	READ (9) (STIFML(J,1), J=1,144)	RDDA2200
120	WRITE (2) (STIFML(J,1),J=1,144)	RDDA2210
•	IF (INDISL.NE.O) READ (9) (SOLVEC(I), I=1, NROW)	RDDA2220
130	IF (NJOINT+LT+2) CALL ERPNT1 (ERJNL+4+-1)	RDDA2230

```
IF (NBAR.LT.1) CALL ERPNT1 (ERBN,4,-1)
                                                                           RDDA2240
      IF ( INDNMA .NE. 0 ) GO TO 138
                                                                           RDDA2250
      ERRTOL = ABS(ERRTOL)
                                                                           RDDA2260
      WRITE(6,220) ERRTOL, INDISL, INDITR, INDPLS, INDRLX, INDWKT,
                                                                           RDDA2270
                    MINRST, RELAXF, TMAX
                                                                           RDDA2280
     1
      IF ( INDISL .NE. 0 ) WRITE(6,260) (SOLVEC(I),I=1,NROW )
                                                                           RDDA2290
  138 CONTINUE
                                                                           RDDA2300
      CALL ERPNT2
                                                                           RDDA2310
      RETURN
                                                                           RDDA2320
C
                                                                           RDDA2330
  140 FORMAT ( 1XI4, 3E10.3,6I3,2I4 )
                                                                           RDDA2340
  150 FORMAT ( 1XI4, 4I3, 3E10.2 )
                                                                           RDDA2350
  160 FORMAT (1XI4, 3I3, A3, E8.3, 5E9.3, 2F5.2 )
                                                                           RDDA2360
  210 FORMAT (33H1STRUCTURAL ANALYSIS CONTROL DATA //
                                                                           RDDA2370
              13HOGENERAL DATA
                                   11
                                                                           RDDA2380
        9H INDRKT =,15,7X,39H, 1 IMPLIES READ ALL MATRICES FROM TAPE /
                                                                           RDDA2390
        9H INDSFG =, I5, 7X 37H, 0 IMPLIES WRITE GLOBAL BAR MATRICES /
     ×
                                                                           RDDA2400
        9H INDSFL =, 15,7X 36H, 0 IMPLIES WRITE LOCAL BAR MATRICES /
     ¥
                                                                           RDDA2410
        9H INDWKT =, 15, 7X, 37H, 1 IMPLIES SAVE ALL MATRICES ON TAPE
                                                                           RDDA2420
     ×
                                                                           RDDA2430
     ¥
        9H INDWTS =, 15, 7X, 40H, 1 IMPLIES PRINT TOTAL STIFFNESS MATRIX /
     ¥
        9H ISFDIM ≠,17,5X,35H, MAX, STORAGE FOR STIFFNESS MATRIX /
                                                                           RDDA2440
                                                                           RDDA2450
  220 FORMAT (36HODISPLACEMENT/ROTATION SOLUTION DATA //
                                                                            RDDA2460
        9H ERRTOL =,E10.3,2X30H, ITERATION SOLUTION TOLERANCE
                                                                           PDDA2470
        9H INDISL =,15,7X,41H, 1 IMPLIES AN INITIAL SOLUTION IN SOLVEC /RDDA2480
     ¥
     ×
        9H INDITR =, 17, 5X 32H, MAX. SOLUTION ITERATION CYCLES
                                                                            RDDA2490
                                                                      1
        9H INDPLS =, I5, 7X, 31H, 1 IMPLIES CONSIDER PLASTICITY
                                                                            RDDA2500
     ¥
                                                                    1
        9H INDRLX =, 15,7X,28H, ITERATIVE SOLUTION METHOD /
                                                                            RDDA2510
        9H INDWKT =, 15, 7X, 31H, 1 IMPLIES SAVE SOLVEC ON TAPE /
                                                                            RDDA2520
     ¥
        9H MINRST =, I5, 7X, 27H, MIN. ALLOWABLE RESTRAINTS
                                                                            RDDA2530
        9H RELAXF =, F7.2, 5X20H, RELAXATION FACTOR
                                                                            RDDA2540
     ¥
     ¥
        9H TMAX
                  =,F10.3, 2X31H, ITERATION CP TERMINATION TIME
                                                                            RDDA2550
                                                                            RDDA2560
     ×
         )
  230 FORMAT (20H0MODAL ANALYSIS DATA
                                                                            RDDA2570
                                                  1 1
        9H INDNMA =, 15, 7X, 26H, RUN NORMAL MODE ANALYSIS
                                                                            RDDA2580
        9H INDWNM =, I5, 7X, 35H, 1 IMPLIES WRITE MODE DATA ON TAPE /
     ¥
                                                                            RDDA2590
        9H IREDTO =, I5, 7X, 25H, ORDER OF REDUCED SYSTEM /
     ¥
                                                                            RDDA2600
        9H NEIGVL =, 15,7X,31H, REQUIRED NON-RIGID BODY MODES /
                                                                            RDDA2610
     ¥-
                                                                            RDDA2620
         )
  240 FORMAT
             (56H AMASS IS THE DIAGONAL MASS VECTOR OF THE REDUCED SYSTRDDA2630
     *EM / (21X10E11.3))
                                                                            RDDA2640
  250 FORMAT ( 54H IRWKP CONTAINS THE ROWS TO KEEP IN THE REDUCED SYSTEMRDDA2650
        / (21×10I11))
                                                                            RDDA2660
  260 FORMAT (30H SOLVEC IS THE SOLUTION VECTOR / (21X10E11.3)
                                                                            RDDA2670
                                                                     )
      END
                                                                            RDDA2680
```

```
SHRP
                                                                              10
  SUBROUTINE SHRPAN
  COMMON COM(30)
                                                                        SHRP
                                                                              20
  EQUIVALENCE ( COM( 1), NJOINT )
EQUIVALENCE ( COM( 4), NBAR )
                                                                        SHRP
                                                                              30
                                                                        SHRP
                                                                              40
  EQUIVALENCE ( COM( 29), NSHPAN )
                                                                        SHRP
                                                                              50
                                                                        SHRP
                                                                              60
  COMMON / CMAIN /
       RJXYZ (100,3), IJRFM ( 74,6), IJALFM( 74,2), FRCVCT( 74,3),
                                                                        SHRP
                                                                              70
 1
       RMTVCT( 74,3), IBARP ( 130), IBARQ ( 130), IBARR (
                                                                        SHRP
                                                               130).
                                                                              80
 2
             ( 130), BAREA ( 130), BARIN ( 130), BARIT (
                                                                130),
                                                                        SHRP
                                                                              90
 3
       KS
                                                                        SHRP 100
       BARJ
             ( 130), BARYM ( 130), BARSM (
                                               130), RKN (
                                                                130),
                                                                        SHRP 110
            ( 130), SOLVEC( 444), RLIMTU(
                                                 88), RLIMTL(
                                                                 88),
       RKT
  5
       DPRTIT( 88), IROWL ( 445), STRSTF(1)
                                                                        SHRP 120
  COMMON/SP/IAPAN(30), IBPAN(30), ICPAN(30), IDPAN(30),
VSPAN(30), ESPAN(30), TSPAN(30)
                                                                        SHRP 130
                                                                        SHRP 140
  1
                                                                        SHRP 150
   DIMENSION
               VABC(12), AB(3), BC(3), CD(3), DA(3)
                                                                        SHRP 160
   EQUIVALENCE(VABC(1),AB(1)),(VABC(4),BC(1)),(VABC(7),CD(1)),
                                                                        SHRP 170
             (VABC(10),DA(1))
 1
  DIMENSION ERSP1(6), ERSP2(5)
                                                                        SHRP 180
  DATA ERSP2 / 10H A SHEAR P, 10HANEL USES , 10HAN UNDEFIN,
                                                                        SHRP 190
                10HED NODAL P, 10HOINT.
                                                                        SHRP 200
                                            1
  1
  DATA ERSP1 / 10H SHEAR PAN, 10HEL CORNER , 10HPOINTS DO ,
                                                                        SHRP 210
                                                                        SHRP 220
SHRP 230
                10HNOT DEFINE, 10H A RECTANG, 10HLE.
  1
  DATA IBLK3S /10H
                                                                        SHRP 240
SHRP 250
   IF ( NSHPAN .EQ. 0 ) RETURN
       CHECK SHEAR PANEL NODE POINT DEFINITIONS
  DO 10 I = 1, NSHPAN
                                                                        SHRP 260
                                                                        SHRP 270
   IF ( IAPAN(I) .GT. NJOINT ) CALL ERPNT1( ERSP2, 5, -1 )
                                                                        SHRP 280
   IF ( IBPAN(I) .GT. NJOINT ) CALL ERPNT1( ERSP2, 5, -1 )
   IF ( ICPAN(I) .GT. NJOINT ) CALL ERPNT1( ERSP2, 5, -1 )
                                                                        SHRP 290
                                                                        SHRP 300
   IF ( IDPAN(I) .GT. NJOINT ) CALL ERPNT1( ERSP2, 5, -1 )
                                                                        SHRP 310
10 CONTINUE
   CALL ERPNT2
                                                                        SHRP 320
   DO 200 ISP = 1, NSHPAN
                                                                        SHRP 330
                                                                        SHRP 340
   IA = IAPAN(ISP)
                                                                        SHRP 350
   IB = IBPAN(ISP )
                                                                        SHRP 360
   IC = ICPAN(ISP )
                                                                        SHRP 370
   ID = IDPAN(ISP)
       CALCULATE THE VECTORS AB, BC, CD, AND DA TO CHECK
                                                                        SHRP 380
                   CONDITIONS OF THE SHEAR PANEL
                                                                        SHRP 390
                                                                        SHRP 400
   AL = 0.0
                                                                        SHRP 410
  BL = 0.0
   CL = 0.0
                                                                         SHRP 420
   DO 20 I = 1, 3
                                                                         SHRP 430
   AB(I) = RJXYZ(IB,I) - RJXYZ(IA,I)
                                                                         SHRP 440
   BC(I) = RJXYZ(IC,I) - RJXYZ(IB,I)
                                                                         SHRP 450
                                                                         SHRP 460
   CD(I) = RJXYZ(ID,I) - RJXYZ(IC,I)
                                                                         SHRP 470
   DA(I) = RJXYZ(IA,I) - RJXYZ(ID,I)
                                                                         SHRP 480
   AL = AL + AB(I) **2
   BL = BL + BC(I) * * 2
                                                                         SHRP 490
                                                                         SHRP 500
   CL
      = CL + CD(I) * * 2
                                                                         SHRP 510
20 CONTINUE
  AL = SQRT(AL)
                                                                         SHRP 520
                                                                         SHRP 530
  BL
      = SQRT( BL )
   C1
      = SQRT( CL )
                                                                         SHRP 540
                                                                         SHRP 550
```

1

ę

300

С

С

С

С

C C IS THIS SHEAR PANEL A RECTANGLE SHRP 560 SHRP 570 č CHECK CORNER ANGLES (90) SHRP 580 DO 40 I = 1, 7, 3SHRP 590 CALL DOTX(ALL, VABC(I), VABC(I+3)) SHRP 600 IF (ABS(ALL) .LT. 1.E-6) GO TO 40 SHRP 610 CALL ERPNT1(ERSP1, 6, -1) SHRP 620 GO TO 200 SHRP 630 40 CONTINUE SHRP 640 C CHECK PLANE CONDITIONS SHRP 650 SHRP 660 CALL CROSSX(DA, AB, BC) CALL CROSSX(AB, BC, CD) SHRP 670 CALL DOTX (BC(1), AB, AB) CALL DOTX (BC(2), DA, DA) SHRP 680 SHRP 690 SHRP 700 BC(1) = SQRT(BC(1))BC(2) = SQRT(BC(2))SHRP 710 DO 60 I = 1, 3 SHRP 720 BC(3) = ABS(DA(I)/BC(2) - AB(I)/BC(1))SHRP 730 CD(1) = .5*BC(3) + 1.E-5SHRP 740 IF (BC(3)/ CD(1) •LE• •0001) GO TO 60 SHRP 750 CALL ERPNT1 (ERSP1, 6, -1) SHRP 760 GO TO 200 SHRP 770 60 CONTINUE SHRP 780 IFRIST = 0SHRP 790 NBAR = NBAR + 1SHRP 800 IBARP(NBAR) = IASHRP 810 IBARQ(NBAR) = ICSHRP 820 IBARR(NBAR) =-ID SHRP 830 80 CONTINUE SHRP 840 AREA = (AL**2+BL**2)**1.5 * TSPAN(ISP)/(4.*AL*BL*(1.+VSPAN(ISP)))SHRP 850 (NBAR) = IBLK3SKS SHRP 860 BAREA(NBAR) = AREASHRP 870 BARIN(NBAR) = 0.0SHRP 880 BARIT(NBAR) = 0.0SHRP 890 BARJ (NBAR) = 0.0SHRP 900 BARSM(NBAR) = 0.0SHRP 910 BARYM(NBAR) = ESPAN(ISP)SHRP 920 IF (IFRIST .NE. 0) GO TO 200 SHRP 930 IFRIST = 1SHRP 940 SHRP 950 NBAR = NBAR + 1SHRP 960 IBARP(NBAR) = IDIBARQ(NBAR) = IBSHRP 970 IBARR(NBAR) =-IA SHRP 980 AL = CLSHRP 990 GO TO 80 SHRP1000 200 CONTINUE SHRP1010 RETURN SHRP1020 END SHRP1030

	SUBROUTINE CROSSX(S, A, B) DIMENSION A(1), B(1), S(1)	CROS CROS	10 20
С	S .EQ. THE VECTOR OR CROSS PRODUCT OF VECTORS A AND B	CROS	30
	S(1) = A(2) * B(3) - A(3) * P(2)	CROS	40
	S(2) = -A(1) * B(3) + A(3) * B(1)	CROS	50
	S(3) = A(1) * B(2) - A(2) * B(1)	CROS	60
	RETURN	CROS	70
	ENTRY DOTX	CROS	80
С	S(1) •EQ• THE DOT OR INNER PRODUCT OF TWO VECTORS A AND B	CROS	90
	S(1) = A(1) * B(1) + A(2) * B(2) + A(3) * B(3)	cros	100
	RETURN	CROS	110
	END	CROS	120

OVERLAY (SASLP, 2, 0)

С

С

С

С

С

С

С

С

C C

С

10

20

С

С

30

C

303

STIF 10 PROGRAM STIFF STIF COMMON COM(30) 20 EQUIVALENCE (COM (24), INDNMA) STIF 30 EQUIVALENCE (COM(4), NBAR) STIF 40 EQUIVALENCE (COM (7), IBAR) STIF 50 EQUIVALENCE (COM(9), INDSFL) STIF 60 EQUIVALENCE (COM(10), INDSFG) STIF 70 COMMON / CMAIN / STIF 80 RJXYZ (100,3), IJRFM (74,6), IJALFM(74,2), FRCVCT(74,3), 1 STIF 90 RMTVCT(74,3), IBARP (130), IBARQ (130), IBARR (STIF 130), 100 2 3 KS 130), BAREA (130), BARIN (130), BARIT (130), STIF 110 (130), BARYM (130), BARSM (130), RKN STIF 120 4 BARJ (130), (STIF 130 RKT 130), SOLVEC(444), RLIMTU(88), RLIMTL(88), 5 (DPRTIT(88), IROWL (445), STRSTF(1) **STIF 140** 6 DIMENSION ISTSTF(1) STIF 150 STIF 160 EQUIVALENCE (STRSTF(1), ISTSTF(1)) **STIF 170** STIF 180 STIF 190 SET UP STRUCTURAL STIFFNESS MATRIX STIF 200 CALL SETSTE STIF 210 STIE 220 BUILD STRUCTURAL STIFFNESS MATRIX BY SUMMING BAR STIFFNESS STIF 230 MATRICES IF (INDNMA .NE. 0) GO TO 6 STIF 240 STIF 250 CALL WRSTD1 STIF 260 6 CONTINUE DO 30 IBAR=1,NBAR STIF 270 STIF 280 COMPUTE STIFFNESS MATRIX AND TRANSFORMATION MATRIX FOR BAR STIF 290 NUMBER IBAR STIF. 300 CALL STFTRN STIF 310 SAVE LOCAL STIFFNESS M AND TRANSFORMATION M ON FILE TAPE? STIF 320 STIF 330 STIF 340 IF (INDNMA .NE. 0) GO TO 8 STIE 350 CALL WRSTDK STIF 360 8 CONTINUE **STIF 370** IF (INDSFL.NE.0) GO TO 10 STIF 380 CALL WRBDAT STIF 390 CONTINUE TRANSFORM BAR STIFFNESS MATRIX TO GLOBAL COORDINATE SYSTEM STIF 400 STIF 410 CALL TRASMK IF (INDSFG.NE.O) GO TO 20 STIE 420 STIF 430 CALL BRSTRA STIE 440 CONTINUE PLACE TRANSFORMED BAP STIFFMESS MATRIX IN STRUCTURAL STIE 450 STIF 460 STIFFNESS MATRIX STIF 470 CALL STORSM STIF 480 CONTINUE STIF 490 RETURN STIE 500 END

OV20 10

SUBROUTINE SETSTE SETS 10 COMMON COM(30) SETS 20 EQUIVALENCE (COM(4), NBAR) SETS 30 EQUIVALENCE (COM (1), NJOINT) SETS 40 EQUIVALENCE (COM(8), NROW) SETS 50 EQUIVALENCE (COM (27), ISFDIM) SETS 60 SETS 70 COMMON / CMAIN / RJXYZ (100,3), IJRFM (74,6), IJALFM(74,2), FRCVCT(74,3), SETS 80 1 130), IBARQ (130), BARIN (2 RMTVCT(74,3), IBARP (130), IBARR (130), SETS 90 130), BARIT (130), BAREA (**SETS 100** 3 KS 130), ſ 130), BARYM (130), BARSM (130), RKN (130), SETS 110 4 BARJ (130), SOLVEC(444), RLIMTU(88), RLIMTL(88), SETS 120 RKT 5 (88), IROWL (445), STRSTF(1) SETS 130 6 DPRTITE DIMENSION ILSET(74,74), ISTSTE(1) **SETS 140** EQUIVALENCE (STRSTF(1), ISTSTF(1), ILSET(1,1)) **SETS 150 SETS 160** DIMENSION ILSETN(74) DIMENSION NZPRW(74) **SETS 170** NULL=NJOINT/2 SETS 180 SETS 190 COMPILE AN ARRAY OF RELATED NODAL POINTS TO SET-UP STIFFNESS MSETS 200 **SETS 210** SETS 220 DO 5 I = 1, 74 SETS 230 5 NZPRW(I) = 0SETS 240 DO 10 I=1,NULL SETS 250 ILSET(I,1)=I SETS 260 10 NZPRW(I) = 1SETS 270 DO 20 I=1.NBAR SFTS 280 IP = IBARP(I)**SETS 290 SETS 300** IQ=IBARQ(I) **SETS 310** NRP=NZPRW(IP)+1 **SETS 320** NRQ=NZPRW(IQ)+1NZPRW(IP)=NRP **SETS 330** NZPRW(IQ) = NRQ**SETS 340** ILSET(IP,NRP)=IQ **SETS 350** 20 ILSET(IQ,NRQ)=IP SETS 360 SETS 370 NULL=NULL+1 SETS 380 DO 30 I=NULL,NJOINT SETS 390 NRP=NZPRW(I)+1 SETS 400 NZPRW(I)=NRP **SETS 410** 30 SETS 420 ILSET(I, NRP)=I **SETS 430** CALCULATE THE TOTAL NO. OF STORAGE WORDS NEEDED FOR THE SETS 440 STIFFNESS MATRIX AND COMPARE WITH WHAT IS AVAILABLE SETS 450 SETS 460 NULL = 0SETS 470 DO 32 I = 1, NJOINT SETS 480 32 NULL = NULL + NZPRW(I) SETS 490 NULL = NULL * 42 + NJOINT*6 + 2 SETS 500 IF (NULL .LE. ISFDIM) GO TO 34 SETS 510 WRITE(6, 200) NULL, ISFDIM SETS 520 STOP SETS 530 34 CONTINUE SETS 540 SETS 550

С С С

C

С

С

С

С

С

304

С

6		
C	- · · · · · · · · · · · · · · · · · · ·	SETS 560
C	SORT ABOVE ARRAY BY NODAL POINT	SETS 570
С		SETS 580
	DO 90 I=1,NJOINT	SETS 590
	NRQ=NZPRW(I)	SETS 600
	IE (NR0-E0.1) GO TO 90	SETS 610
60		
40		SE15 620
	DU = 80 J = 20 NRQ	SETS 630
- 4	IF (ILSE((1,J-1)-ILSE((1,J)) 80,50,70	SETS 640
50	NRQ=NRQ-1	SETS 650
	NZPRW(I)=NRQ	SETS 660
	DO 60 L=J,NRQ	SETS 670
60	ILSET(I,L)=ILSET(I,L+1)	SETS 680
	GO TO 40	SETS 690
70	NULL=ILSET(I,J)	SETS 700
	$ILSET(I_{j}J)=ILSET(I_{j}J-1)$	SETS 710
	ILSET(I,J-1)=NULL	SETS 720
	NRP=1	SETS 730
80	CONTINUE	SETS 740
0.7	IE (NPP-NE-0) GO TO 40	SETS 750
00		SETS 750
~	CONTINUL	SETS 700
Ċ		SEIS 770
C	WRITE ARRAY ON FILE I	SETS 780
C		SETS 790
	REWIND	SEIS 800
	DO 100 I=1,NJOINT	SETS 810
	NRQ=NZPRW(I)	SETS 820
100	WRITE (1) (ILSET(I,J),J=1,NRQ)	SETS 830
С		SETS 840
С	BUILD STIFFNESS MATRIX STORAGE ARRAY	SETS 850
С		SETS 860
	REWIND 1	SETS 870
	NU[1] = 1	SETS 880
	NROW=0	SETS 890
		SETS OND
		SETS 010
		SETS 910
		SEIS 920
	READ (I) (ILSEIN(J), J=I, NRP)	SEIS 930
	NFIRST=NULL+1	SE15 940
	NROW=NROW+1	SETS 950
С		SETS 960
	L = NRQ + NULL - 1	SETS 970
	DO 110 $J = NULL$, L	SETS 980
110	STRSTF(J)=0.0	SETS 990
	ISTSTF(NULL)=~NROW	SETS1000
	IROWL (NROW) = NULL	SETS1010
	NULLENFIRST	SETS1020
C		SET \$1030
	DO 120 J=1.NRP	SETSIAAA
	ISTSTF(NULL) = (6*(ILSETN(J)-1)+1)	SETS1040
120	N[j] = N[j] + 7	SETSION
120		SE131000
C	NERGI-NULL-1	SE131070
		SE131000
		SEIS1090
	NKOW=NKOW+1	SETS1100
	ISTSIF(NULL)=-NROW	SETS1110

	IROWL(I NULL=NU	NROW)=NULL JLL+1			SETS1120 SETS1130
	DO 130	J=NFIRST,NLAST			SETS1140
	STRSTF	(NULL)=STRSTF(J)			SETS1150
130	NULL=N	JLL+1			SETS1160
140	CONTINU	JE			SETS1170
	ISTSTF	(NULL)=-1			SETS1180
	IROWL(NROW+1)=NULL			SETS1190
	RETURN				SETS1200
200	FORMAT	(16H0**** ERROR ****, I6, 22H WORDS OF STORAGE ARE	,		SETS1210
	1	39HREQUIRED FOR THE STIFFNESS MATRIX (ONLY, 16	,	/	SET51220
	2 22X	45H WORDS OF STORAGE ARE AVAILABLE WITH PRESENT ,			SET51230
	3	12HDIMENSIONING)			SETS1240
	END				SETS1250

SUBROUTINE STETRN	STFT	10
COMMON COM(30)	STFT	20
EQUIVALENCE (COM(6), ITINJO)	STFT	30
EQUIVALENCE (COM(7), IBAR)	STFT	40
DIMENSION SIDEM(144)	STFT	50
EQUIVALENCE (STIFML, S1DEM)	STFT	60
COMMON / CMAIN /	STET	70
1 RJXYZ (100,3), IJRFM (74,6), IJALFM(74,2), FRCVCT(74,3),	STET	80
2 RMTVCT(74,3), IBARP (130), IBARQ (130), IBARR (130),	STFT	90
3 KS (130), BAREA (130), BARIN (130), BARIT (130),	STET	100
4 BARJ (130), BARYM (130), BARSM (130), RKN (130),	STFT	110
5 RKT (130), SOLVEC(444), RLIMTU(88), RLIMTL(88),	STFT	120
6 DPRTIT(88), IROWL (445), STRSTF(1)	STFT	130
DIMENSION REARR(1)	STFT	140
EQUIVALENCE (IBARR(1), REARR(1))	STFT	150
COMMON / OVER2 / STIFML(12,12), TRANSM(3,3)	STFT	160
DATA I3BLK / 3H /	STFT	170
ITINJC=1	STFT	180
IF (ABS(BARIT(IBAR))+ABS(BARIN(IBAR))+ABS(BARJ(IBAR))•FO•O) ITINJ	OSTFT	190
1=0	STFT	200
IP=IBARP(IBAR)	STFT	210
IQ=IBARO(IBAR)	STFT	220
IR =IABS(IBARR(IBAR))	STFT	230
	STET	240
COMPUTE UNIT VECTORS ALONG LOCAL AXIS FOR GLOBAL COORDINATE	STFT	250
TRANSFORMATION MATRIX	STFT	260
X=RJXYZ(IQ,1)-RJXYZ(IP,1)	STFT	270
Y=RJXYZ(IQ,2)-RJXYZ(IP,2)	STFT	280
Z=RJXYZ(IQ,3)-RJXYZ(IP,3)	STET	290
BARLGT=SQRT(X*X+Y*Y+Z*Z)	STET	300
IF (IBARR(IBAR) •LT• 0) REARR(IBAR) = -BARLGT	SIFI	310
TRANSM(1,1)=X/BARLGT		320
TRANSM(1)2)=Y/BARLGT		220
RANSM(1,3) = 2/BARLGI	0151	250
	STET	240
	STET	370
	CTET	380
	CTET	200
	STET	400
	STET	410
A = ROATZ(IR) I = ROATZ(IP) I	STET	420
T = R[X + Z] = R[X + Z] = R[X + Z] + Z[X + Z]	STET	430
$\Sigma = (\Sigma \wedge T \in \Sigma \setminus T \in \Sigma \setminus T \in \Sigma \cap S \cap T \in \Sigma \cap S \cap S \cap S \cap S \cap S \cap S \cap S \cap S \cap S \cap$	STET	440
DDP = SORT (X*X+Y*Y+7*7-DPD*50PD)	STET	450
TRANSM(2, +) = (X - TRANSM(1, +) * DPD) / DDR	STET	460
TRANSM(2+2) = (Y - TRANSM(1+2)*DPD)/DDR	STET	470
TRANSM(2,3) = (7 - TRANSM(1,3) * DPD) / DDR	STET	480
TRANSM(3,1) = TRANSM(1,2) * TRANSM(2,3) - TRANSM(2,2) * TRANSM(1,3)	STET	490
TRANSM(3,2)=-TRANSM(1,1)*TRANSM(2,3)+TRANSM(2,1)*TRANSM(1,3)	STFT	500
TRANSM(3,3)=TRANSM(1,1)*TRANSM(2,2)-TRANSM(2,1)*TRANSM(1,2)	STET	510
	STFT	520
COMPUTE STIFFNESS MATRIX	STFT	530
	STET	540
CONTINUE	STFT	550

C C C

10

20

C C C 30



C C	ZERO RIGHT HALF OF STIFFNESS MATRIX	STFT 560 STFT 570 STET 580
C	DO 40 I=1.12 DO 40 J=I.12	STFT 590 STFT 590 STFT 600
40 C	$STIFML(I \cdot J) = 0 \cdot 0$	STFT 610 STFT 620
C C	FILL LEFT HALF OF STIFFNESS MATRIX AND THE DIAGONAL	STFT 630 STFT 640
	S1DEM(1)≈BARYM(IBAR)*BARFA(IBAR)/BARLGT S1DEM(37)=-S1DEM(1)	STFT 650 STFT 660
	SIDEM(40)=SIDEM(I) IF (ITINJ0•EQ•0) GO TO 160	STFT 670 STFT 680
	S1DEM(14)=12•*BARYM(IEAR)*BARIN(IBAR)/BARLGT**3 S1DEM(27)=12•*BARYM(IBAR)*BARIT(IBAR)/BARLGT**3	STFT 690 STFT 700
	S1DEM(50)=-S1DEM(14) S1DEM(53)=S1DEM(14)	STFT 710 STFT 720
	S1DEM(63) =- S1DEM(27)	STFT 730
	SIDEM(00)=SIDEM(21) SIDEM(79)=BARSM(IBAR)*PARJ(IBAR)/RARLGT	STFT 750
	SIDEM(87)=-6.*BARYM(IBAR)*BARIT(IBAR)/BARLGT**2 SIDEM(90)=-SIDEM(87)	STFT 760 STFT 770
	SIDEM(92)=4.*BARYM(IBAR)*BARIT(IBAR)/BARLGT SIDEM(98)=6.*BARYM(IBAR)*BARIN(IBAR)/BARLGT**2	STFT 780 STFT 790
	S1DEM(101)=-S1DEM(98) S1DEM(105)=4.*BARYM(IBAR)*BARIN(IBAR)/BARLGT	STFT 800 STFT 810
	S1DEM(115)=-S1DEM(79) S1DEM(118)=S1DEM(79)	STFT 820 STFT 830
	S1DEM(123)=S1DEM(87) S1DEM(126)=-S1DEM(87)	STFT 840 STFT 850
	S1DEM(128)=2•*BARYM(IBAR)*BARIT(IBAR)/BARLGT S1DEM(131)=S1DEM(92)	STFT 860 STFT 870
	SIDEM(134)=SIDEM(98)	STFT 880
	SIDEM(137)=-SIDEM(98) SIDEM(141)=2.*BARYM(IBAR)*BARIN(IBAR)/PARLGT	STFT 890 STFT 900
c	SIDEM(144)=SIDEM(105) IF (KS(IBAR).EQ.I3ELK) GO TO 160 DECODE KS(IBAR)	STET 910 STET 920 STET 930
C	IR = 1 IMPLIES SHEAR STRAIN IS INCLUDED	STFT 940
C	IP = 1 FREE SP• $IP = 2$ FREE SQ IQ = 1 FREE 6P• $IQ = 2$ FREE 6Q	STFT 950 STFT 960
	DECODE (3,180,KS(IBAR))IR,IP,IQ X=BARSM(IBAR)*BAREA(IBAR)/BARLGT	STFT 970 STFT 980
	IF (RKN(IBAR)•EQ•O•O) GO TO 50 AN=STIFML(2•2)*RKN(IBAR)/X	STFT 990 STFT1000
50	GO TO 60 AN=0•0	STFT1010 STFT1020
6 0	CONTINUE LE (RKT(LBAR).EQ.0.0) GO TO 70	STFT1030 STFT1040
	AT=STIFML(3,3)*RKT(IBAR)/X	STFT1050 STFT1060
70	AT=0.0	STFT1070
80	IF (IR+EQ+1) GO TO 120	STFT1080 STFT1090
с	IF (IP+EQ+0) GO TO 120	STFT1100 STFT1110

C C C 40 C C C

С С С С С

С

C	Y MOMENT FREE	STFT1120
C ,		STFT1130
	X=1./(4.+AT)	STFT1140
	STIFML(3,3)=X*STIFML(3,3)	STFT1150
	STIFML(3,6)=X*STIFML(3,6)	STFT1160
	STIFML(6,6)=X*STIFML(6,6)	STFT1170
	STIFML(8,11)=0.0	STFT1180
	IF (IP.FQ.2) GO TO 90	STET1190
	I=11	STET1200
		ST. 71200
		51111210
0.0		STFT1220
90		SIF11230
		S1F11240
100	CONTINUE	STFT1250
	Y=X+X	STFT1260
	STIFML(3,I)=Y*STIFML(3,I)	STFT1270
	STIFML(6,I)=Y*STIFML(6,I)	STFT1280
	Y = Y + X	STFT1290
	STIFML(I,I)=Y*STIFML(6,I)	STFT1300
	STIFML(3,J)=0.0	STFT1310
	STIFML(6,J)=0.0	STFT1320
	STIFML(J,J)=0.0	STFT1330
110	IF (IQ•NE•0) GO TO 130	STFT1340
С		STET1350
С	Z MOMENT NOT FREE	STFT1360
С		STFT1370
	IF (RKN(IBAR) • EQ • 0 • 0) GO TO 160	STFT1380
	$X = 1 \cdot / (1 \cdot + AN)$	STET1390
	STIFMI $(2 \cdot 2) = X \times STIFMI (2 \cdot 2)$	STET1400
	$STIFMI(2.5) = X \times STIFMI(2.5)$	STET1410
	STIFML(2.9) = X*STIFML(2.9)	STFT1420
	STIFM! (2.12)-Y*STIFM! (2.12)	STET1/30
	STIFM! (5.5)=X*STIFM! (5.5)	STET1440
	STIFM! (5.9)=X*STIFM! (5.9)	STET1450
	STIEM! (5,12)-Y*STIEM! (5,12)	STET1460
	$Y = 1 = 2 / (4 \times (1 + 1 + 1 + 1))$	STET1470
	<pre><= 16 _ Def (4e*(1e*1e/A(N)))</pre>	STET1480
	STIEML(0,30)-(1,-2,(0,2)) +1 (ANN)/2011EML(0,30)	CTET1400
	STIFME(9912)=(10=30/(200(10+10/00)//00)10ME(9912) STIFME(12 10)=V*STIFME(12 10)	STITI470
	STIFML(12912)=X*STIFML(12912)	31111900
~	GO TO 160	STET1510
Ć		STFT1520
Ć	Y MOMENT NOT FREE	SIFI1530
C		SIFT1540
120	IF $(RKT(IBAR) \bullet EQ \bullet 0 \bullet 0)$ GO TO IIO	SIF11550
	X=1•/(1•+A)	STF11560
	STIFML(3,3)=X*STIFML(3,3)	STFT1570
	STIFML(3,6)=X*STIFML(3,6)	STFT1580
	STIFML(3,8)=X*STIFML(3,8)	STFT1590
	STIFML(3,11)=X*STIFML(3,11)	STFT1600
	STIFML(6,6)=X*STIFML(6,6)	STFT1610
	STIFML(6,8)=X*STIFML(6,8)	STFT1620
	STIFML(6,11)=X*STIFML(6,11)	STFT1630
	X=13./(4.*(1.+1./AT))	STFT1640
	STIFML(8,8)=X*STIFML(8,8)	STFT1650
	STIFML(8,11)=(13./(2.*(1.+1./AT)))*STIFML(8,11)	STFT1660
	STIFML(11,11)=X*STIFML(11,11)	STFT1670
_	GC TO 110	STFT1680
-----	---	----------------------
C		SIF11690
C	Z MOMENT FREE	STF11700
C		STF11/10 STF11/20
130	$X = 1 \bullet / \{4 \bullet + AN\}$	STFT1720
	SIIFML(2) = X + SIIFML(2)	STITT50
	S = IFML(2,5) = X*S = IFML(2,5)	STET1750
	SIIFML(5,5)=X*SIIFML(5,5)	STET1760
	STIFML(9,12)=0.0	STF11700
	IF (IQ+EQ+2) GO IO 140	SIFI170
	I = 12	SIF11700
	J=9	STF11790
	GO TO 150	STF11800
140	I = 9	STF11810
	J=12	STFT1820
150	CONTINUE	STFT1830
	Y=X+X	STFT1840
	STIFML(2,I)=Y*STIFML(2,I)	STFT1850
	STIFML(5,I)=Y*STIFML(5,I)	STFT1860
	Y=Y+X	STFT1870
	STIFML(I,I)=Y*STIFML(12,12)	STFT1880
	STIFML(2,J)=0.0	STFT1890
	STIFML(5,J)=0.0	STFT1900
	STIFML(J,J)=0.)	STFT1910
С		STFT1920
Ċ	FILL LEFT HALF OF STIFFNESS MATRIX	STFT1930
ĉ		STFT1940
160	CONTINUE	STFT1950
100	$DC = 170 I = 1 \cdot 11$	STFT1960
		STFT1970
	$D_{0} = 170 = 11 \cdot 12$	STFT1980
170	STIEML $(J \bullet I) = STIEML (I \bullet J)$	SŤĚT1990
110	DETUDN	STFT2000
c		STFT2010
100	FORMAT (311)	STFT2020
100		STET2030
	END	011 12/200

```
STOR

    SUBROUTINE STORSM

                                                                                   10
      COMMON COM(30)
                                                                             STOR
                                                                                   20
      EQUIVALENCE ( COM( 6), ITINJO )
                                                                             STOR
                                                                                   30
      EQUIVALENCE ( COM( 7), IBAR
                                      ١
                                                                             STOR
                                                                                   40
                                                                                   50
                                                                             STOR
      COMMON / CMAIN /
          RJXYZ (100,3), IJRFM ( 74,6), IJALFM( 74,2), FRCVCT( 74,3),
                                                                             STOR
                                                                                   60
     1
          RMTVCT( 74,3), IBARP (
                                   130), IBARQ (
                                                    130), IBARR (
                                                                             STOR
                                                                    130),
                                                                                   70
     2
                                                    130), BARIT (
                   130), BAREA (
                                    130), BARIN (
     3
          KS
                                                                    130),
                                                                             STOR
                                                                                   80
                 (
     4
          BARJ
                1
                    130), BARYM (
                                   130), BARSM (
                                                    130), RKN (
                                                                    130),
                                                                             STOR
                                                                                   90
                   130), SOLVEC(
                                                     88), RLIMTL(
                                                                             STOR 100
          RKT
                                   444), RLIMTU(
     5
                                                                     88),
                 (
                     88), IROWL ( 445), STRSTF(1)
                                                                             STOR 110
     6
          DPRTIT(
      DIMENSION ISTSTF(1)
                                                                             STOR 120
      EQUIVALENCE ( STRSTF(1), ISTSTF(1))
                                                                             STOR 130
      COMMON / OVER2 / STIFML(12,12), TRANSM(3,3)
                                                                             STOR 140
                                                                             STOR 150
      DIMENSION IRC13(6), IRC24(6)
      DATA IRC13 / 1, 2, 3, 7, 8, 9 /
                                                                             STOR 160
      DATA IRC24 / 4, 5, 6, 10, 11, 12 /
                                                                             STOR 170
                                                                             STOR 180
      ISP = (IBARP(IBAR) - 1) * 6 + 1
      ISQ=(IBARQ(IBAR)-1)*6+1
                                                                             STOR 190
                                                                             STOR 200
C
          FIND INCREMENT FROM START OF ROW P TO COLUMNS P AND Q
С
                                                                             STOR 210
                                                                             STOR 220
Ĉ
                                     IPP AND IPQ
                                CALL
      J1=0
                                                                             STOR 230
      II=IROWL(ISP)
                                                                             STOR 240
                                                                             STOR 250
      I 1 = I I + 1
10
      IF (ISTSTF(I1).NE.JSP) GO TO 20
                                                                             STOR 260
                                                                             STOR 270
      IPP=I1-II
                                                                             STOR 280
      J1 = J1 + 1
      GO TO 30
                                                                             STOR 290
20
      IF (ISTSTF(I1).NE.ISQ) GO TO 30
                                                                             STOR 300
                                                                             STOR 310
      IPQ=I1-II
      J1=J1+1
                                                                             STOR 320
30
                                                                             STOR 330
      11 = 11 + 7
      IF (J1.NE.2) GO TO 10
                                                                             STOR 340
                                                                             STOR 350
С
                                                                             STOR 360
С
          ADD TO THE 6 P ROWS
                                                                             STOR 370
С
      IF (ITINJO.NE.O) GO TO 40
                                                                             STOR 380
      I1=II+IPP+1
                                                                             STOR 390
      STRSTF(I1)=STRSTF(I1)+STIFML(1,1)
                                                                             STOR 400
                                                                             STOR 410
      I = I I + I PQ + 1
      STRSTF(I1)=STRSTF(I1)+STIFML(1,4)
                                                                             STOR 420
                                                                             STOR 430
      GO TO 70
40
                                                                             STOR 440
      CONTINUE
      DO 60 I=1,6
                                                                             STOR 450
                                                                             STOR 460
      IP=II+IPP
                                                                             STOR 470
      IQ = II + IPQ
                                                                             STOR 480
      I1=IRC13(I)
      DO 50 J=1,6
                                                                              STOR 490
                                                                              STOR 500
      J1=IRC13(J)
                                                                              STOR 510
      JJ=IP+J
      STRSTF(JJ)=STRSTF(JJ)+STIFML(I1,J1)
                                                                              STOR
                                                                                   520
      J1 = IRC24(J)
                                                                              STOR 530
                                                                              STOR 540
      JJ=IQ+J
                                                                              STOR 550
50
      STRSTF(JJ)=STRSTF(JJ)+STIFML(I1,J1)
```

60	J1=ISP+I II=IROWL(J1)	STOR STOR	560 570
C		STOR	580
С	FIND INCREMENT FROM START OF ROW Q TO COLUMNS P AND Q	STOR	590
С	CALL IPP	STOR	600
70	CONTINUE	STOR	610
	J1=0	STOR	620
	II=IROWL(ISQ)	STOR	630
	I 1 = I I + 1	STOR	640
80	IF (ISTSTF(I1).NE.ISQ) GO TO 90	STOR	650
	I QQ=I 1-I I	STOR	660
	11=11+1	STOR	670
	GO TO 100	STOR	680
90	IF (ISTSTF(I1).NE.ISP) GO TO 100	STOR	690
	IQP=II-II	STOR	700
	J1=J1+1	STOR	710
100	$I_1 = I_1 + 7$	STOR	720
	IF (J1•NE•2) GO TO 80	STOR	730
С		STOR	740
С	ADD TO THE 6 Q ROWS	STOR	750
С		STOR	760
	IF (ITINJO.NE.O) GO TO 110	STOR	770
	I = I I + I QQ + 1	STOR	780
	STRSTF(I1)=STRSTF(I1)+STIFML(4,1)	STOR	790
	I = I I + I QP + 1	STOR	800
	STRSTF(I1)=STRSTF(I1)+STIFML(4,4)	STOR	810
	RETURN	STOR	820
110	CONTINUE	STOR	830
	DO 130 I=1,6	STOR	840
	IO = II + IOO	STOR	850
	IP=II+IQP	STOR	860
	I = IPC24(I)	STOR	870
	DQ 120 J=1,6	STOR	880
	J1=IRC13(J)	STOR	890
	L+AI=TC	STOR	900
	<pre>strstf(JJ)=strstf(JJ)+stifmL(I1,J1)</pre>	STOR	910
	J1 = IRC24(J)	STOR	920
	JJ=IQ+J	STOR	930
120	STRSTF(JJ)=STRSTF(JJ)+STIFML(I1,J1)	STOR	940
	J1=ISQ+I	STOR	950
130	II=IROWL(J1)	STOR	960
	RETURN	STOR	970
	END	STOR	980

	SUBROUTINE TRASMK	TRAS	10
	COMMON COM(30)	TRAS	20
	EQUIVALENCE (COM(6), ITINJO)	TRAS	30
	COMMON / OVER2 / STIFML(12,12), TRANSM(3,3)	TRAS	40
	DIMENSION A(12), IJ(4), IJC(12)	TRAS	50
	DIMENSION IRC(12)	TRAS	60
	EQUIVALENCE(AXF, STIFML(1,1))	TRAS	70
	DATA IJ / 0,3,6,9 /	TRAS	80
	DATA IJC / 3*0, 3*3, 3*6, 3*9 /	TRAS	90
	DATA IRC / 1,2,3,1,2,3,1,2,3,1,2,3 /	TRAS	100
		TRAS	110
С		TRAS	120
С	IF THERE IS BENDING GO TO 8	TRAS	130
С		TRAS	140
	IF (ITINJO•NE•O) GO TO 20	TRAS	150
	DO 10 I=1,3	TRAS	160
	DO 10 J=1,3	TRAS	170
	AHOLD=AXF*TRANSM(1,J)*TRANSM(1,I)	TRAS	180
	STIFML(I,J)=AHOLD	TRAS	190
	STIFML(I,J+3)=-AHOLD	TRAS	200
	STIFML(I+3,J)=-AHOLD	TRAS	210
10	STIFML(I+3,J+3)=AHOLD	TRAS	220
	RETURN	TRAS	230
20	CONTINUE	TRAS	240
С		TRAS	250
С	MULTIPLY TRANSFORMATION MATRIX TIMES STIFFNESS MATRIX	TRAS	260
С	STORE RESULT IN STIFFNESS MATRIX	TRAS	270
С		TRAS	280
	DO 40 K=1,12	TRAS	290
	ICR=0	TRAS	300
	DO 30 IC=1,4	TRAS	310
	DO 30 I=1,3	TRAS	320
	ICR=ICR+1	TRAS	330
	A (I CR) = 0 • 0	TRAS	340
	DO 30 J=1,3	TRAS	350
	JJ=J+IJ(IC)	TRAS	360
	IF (STIFML(K,JJ).EQ.O.) GO TO 30	TRAS	370
	A(ICR)=A(ICR)+STIFML(K,JJ)*TRANSM(J,I)	TRAS	380
30	CONTINUE	TRAS	390
	DO 40 J=1,12	TRAS	400
40	STIFML(K,J)=A(J)	TRAS	410
C		TRAS	420
C	,	TRAS	430
C	MULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THE	TRAS	440
C	TRANSFORMATION MATRIX	TRAS	450
C		TRAS	460
	DO 60 K=1,12	TPAS	470
	DO 50 ICR=K,12	TRAS	480
	$A(ICR)=U_{\bullet}O$	TRAS	490
		TRAS	500
		TDAC	510
F O	JJ=J+IJC(ICK) A/ICD)=A/ICD)=CTIEML(./)#TDANCM/ .T)	TRAS	520
50	$\frac{1}{2} = \frac{1}{2}$	TDAC	- 22U - 540
60		TDAC	540
00	DO TO T=1.11	TPAC	550
		11/40	200

C C C

IC=I+1 DO 70 J=IC,12	TRAS 570 TRAS 580
STIFML(I,J)=STIFML(J,I)	TRAS 590
	TRAS 600
AS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFFNESS	TRAS 610
MATRIX TRANSFORMED TO THE GLOBAL COORDINATE SYSTEM	TRAS 620
	TRAS 630
RETURN	TRAS 640
END	TRAS 650

70 C C C C

	SUBROUTINE WRBDAT	WRBD	10
	COMMON COM(30)	WRBD	20
	EQUIVALENCE (COM(7), IBAR)	WRBD	30
	COMMON / OVER2 / STIFML(12,12), TRANSM(3,3)	WRBD	40
	COMMON / CMAIN /	WRBD	50
	1 RJXYZ (100,3), IJRFM (74,6), IJALFM(74,2), FRCVCT(74,3),	WRBD	60
	2 RMTVCT(74,3), IBARP (130), IBARQ (130), IBARR (130),	WRBD	70
	3 KS (130), BAREA (130), BARIN (130), BARIT (130),	WRBD	30
	4 BARJ (130), BARYM (130), BARSM (130), RKN (130),	WRBD	90
	5 RKT (130), SOLVEC(444), RLIMTU(88), RLIMTL(88),	WRBD	100
	6 DPRTIT(88), IROWL (445), STRSTF(1)	WRBD	110
C		WRBD	120
C	IF INDSEL .NE. 0, PRINI LOCAL STIFFNESS MAIRIX AND	WRBD	130
C	TRANSFORMATION MATRIX FOR BAR(I)	WRBD	140
С		WRBD	150
	WRITE (6,10) IBAR, IBARP(IBAR), IBARQ(IBAR), ((S 1FML(1,J), J=1,12), I	=/vRBD	160
	11912) $((1RANSM(19J)9J=193)91=193)$	WRBD	170
~	RETURN	WRBD	180
C	TE INDEED NE A DOINT THE TRANSFORMED LOCAL STREETERS NATOR	WRBD	190
C	IF INDSEG .NE. U, PRINT THE TRANSFORMED LOCAL STIFFNESS MATRI	XWRBD	200
C		WROD	210
	ENTRE DASTRA WDITE (A :20) ((STIEM:(I, 1), (=1,12), I=1,12)	WRBD	220
	DETIIDN	WRED	240
C		WRRD	250
10	FORMAT (13H) BAR NUMBER 14/.25HO POINT P IS NODAL POINT 14/.25H	PWRBD	260
10	IOINT & IS NODAL POINT 14/11840 STIEFNESS MATRIX//12(12F11.3+/)+2	3WRBD	270
	2H0 TRANSFORMATION MATRIX//-3(3F1]-3-/))	WRBD	280
20	FORMAT (30H0 TRANSFORMED STIFFNESS MATRIX//,,12(12E11.3,/))	WRBD	290
-	END	WRBD	300

	SUBROUTINE WRSTDK	WRSD	10
	COMMON / OVER2 / STIFML(12,12), TRANSM(3,3)	WRSD	20
		WRSD	30
	WRITE LOCAL STIFFNESS AND ASSOCIATED TRANSFORMATION	WRSD	40
	MATRICES ON FILE 2 FOR LATER USE IN FMBARS (AND) IF	WRSD	50
	INDWKT •NE• 0, IN PANDTK)	WRSD	60
		WRSD	70
	WRITE (2) ((TRANSM(I,J),J=1,3),I=1,3)	WRSD	80
	WRITE (2) ((STIFML(I,J),J=1,12),I=1,12)	WRSD	90
	RETURN	WRSD	100
	ENTRY WRSTD1	WRSD	110
-	REWIND 2	WRSD	120
	RETURN	WRSD	130
	END	WRSD	140

OVERLAY (SASLP, 3, 0)

OV30 10

PROGRAM FINIAL COMMON COM(30) EQUIVALENCE (COM(16), INDWKT) EQUIVALENCE (COM(8), NROW) COMMON / CMAIN / 1 RJXYZ (100,3), IJRFM (74,6), IJALFM(74,2), FRCVCT(74,3), 2 RMTVCT(74,3), IBARP (130), IBARQ (130), IBARR (130), 3 KS (130), BAREA (130), BARIN (130), BARIT (130), 4 BARJ (130), BAREA (130), BARIN (130), BARIT (130), 5 RKT (130), SOLVEC(444), RLIMTU(88), RLIMTL(88), 6 DPRTIT(88), IROWL (445), STRSTF(1) PANDTK, PRINTS THE NON ZERO ELEMENTS OF THE STRUCTURAL STIFFNESS MATRIX AND CONDENSES THE BLOCK STORAGE WHERE POSSIBLE CALL PANDTK	FINI 10 FINI 20 FINI 30 FINI 40 FINI 50 FINI 50 FINI 70 FINI 90 FINI 100 FINI 100 FINI 120 FINI 130 FINI 140 FINI 150 FINI 160 FINI 170
SOLVE CALL SOLVE IF (INDWKT.EQ.0) GO TO 10 WRITE (9) (SOLVEC(I),I=1,NROW) END FILE 9 CONTINUE	FINI 170 FINI 180 FINI 190 FINI 200 FINI 210 FINI 220 FINI 220 FINI 230 FINI 240
CALCULATE LOCAL FORCE-MOMENT VECTORS FOR EACH BAR CALL FMBARS RETURN END	FINI 250 FINI 260 FINI 27C FINI 280 FINI 290 FINI 300

```
SUBROUTINE PANDTK
                                                                     PAND
                                                                           10
COMMON COM(30)
                                                                     PAND
                                                                           20
EQUIVALENCE ( COM( 8), NROW
                                                                     PAND
                                                                           30
                                ١
EQUIVALENCE ( COM(16), INDWKT )
                                                                     PAND
                                                                           40
EQUIVALENCE ( COM( 4 ), NBAR
                                                                           50
                                                                     PAND
                                  3
EQUIVALENCE ( COM ( 28), INDWTS )
                                                                     PAND
                                                                            60
COMMON / CMAIN /
                                                                     PAND
                                                                           70
     RJXYZ (100,3), IJRFM ( 74,6), IJALFM( 74,2), FRCVCT( 74,3),
                                                                           80
                                                                     PAND
1
     RMTVCT( 74,3), IBARP ( 130), IBARQ (
                                            130), IBARR (
                                                                           90
                                                            130),
                                                                     PAND
2
3
              130), BAREA (
                            130), BARIN (
                                             130), BARIT (
                                                             130),
                                                                     PAND 100
     KS
           (
     BARJ
              130), BARYM (
                            130), BARSM (
                                             130), RKN
                                                                     PAND 110
4
           (
                                                        (
                                                             130),
              130), SOLVEC( 444), RLIMTU(
                                                                     PAND 120
5
     RKT
                                              88), RLIMTL(
                                                              88),
           (
                                                                     PAND 130
     DPRTIT(
               88), IROWL ( 445), STRSTF(1)
6
                                                                     PAND 140
DIMENSION STF(519), ISTF(519)
DIMENSION ISTSTF(1), STFDIA(444), STIFML(12,12), TRANSM(3,3)
                                                                      PAND 150
                                                                      PAND 160
EQUIVALENCE ( STRSTF(1), ISTSTF(1))
EQUIVALENCE ( BAREA (1), STEDIA(1))
                                                                     PAND 170
                                                                     PAND 180
EQUIVALENCE ( RJXYZ(1,1),STIFML(1,1))
EQUIVALENCE ( RJXYZ(50,2),TRANSM(1,1) )
                                                                      PAND 190
COMMON / OVER3 / COM3(519)
                                                                      PAND 200
EQUIVALENCE ( COM3( 1 ), STF, ISTF )
                                                                     PAND 210
                                                                      PAND 220
LDIA = 0
LDIAS = 0
                                                                      PAND 230
 IF (INDWKT.EQ.0) GO TO 10
                                                                      PAND 240
                                                                      PAND
                                                                          250
     WRITE THE NO. OF ROWS AND THEIR LOCATION IN STIFFNESS MATRIX
                                                                     PAND 260
                                                                      PAND 270
               ON FILE 9
                                                                      PAND 280
                                                                      PAND 290
REWIND 9
                                                                      PAND 300
 WRITE (9) NROW
                                                                      PAND 310
 J = NROW + 1
                                                                      PAND 320
WRITE (9) (IROWL(I), I=1,J)
                                                                      PAND 330
 CONTINUE
                                                                      PAND 340
     PRINT NON ZERO ROW ELEMENTS AND DECIDE WHETHER THE ROW
                                                                      PAND 350
          SHOULD BE STORED IN BLOCK OR ELEMENT FORMAT
                                                                      PAND 360
                                                                      PAND 370
 LOCSTF=1
                                                                      PAND 380
 DO 110 I=1.NROW
                                                                      PAND 390
 TSLIMN7=0
                                                                      PAND 400
 ISUMGP=0
                                                                      PAND 410
 LOCSTF=LOCSTF+1
                                                                      PAND 420
 IF (ISTSTF(LOCSTF).LT.0) GO TO 50
                                                                      PAND 430
                                                                      PAND 440
 IC=ISTSTF(LOCSTF)-1
 ISUMGP=1SUMGP+1
                                                                      PAND 450
                                                                      PAND 460
 DO 40 J=1,6
 LOCSTF=LOCSTF+1
                                                                      PAND 470
 IF (ABS(STRSTF(LOCSTF)).LT.1.E-9) GO TO 40
                                                                      PAND 480
                                                                      PAND 490
 ICC=IC+J
 IF ( INDWTS .NE. O ) CALL WRSTRK ( I, ICC, STRSTF(LOCSTF))
                                                                      PAND 500
 IF (I.NE.ICC) GO TO 30
                                                                      PAND 510
 LDIA=LOCSTF
                                                                      PAND 520
 GO TO 40
                                                                      PAND 530
                                                                      PAND 540
 ISUMNZ=ISUMNZ+1
                                                                      PAND 550
 CONTINUE
```

```
C
C
C
C
C
C
```

C

20

30

50.	GO TO 20 CONTINUE	PAND 560 PAND 570
	IF INDWKT •NE• 0, WRITE STIFFNESS MATRIX IN BLOCK FORMAT ON FILE 9 TO SAVE FOR FUTURE RUNS OF THE PROGRAM	PAND 580 PAND 590 PAND 600 PAND 610 PAND 620
-	LCSTF1=IROWL(I) IF (INDWKT.EQ.O) GO TO 60 NULL=LOCSTF-1	PAND 630 PAND 640 PAND 650
60 C	CONTINUE	PAND 660 PAND 670 PAND 680
c c c	REMOVE THE DIAGONAL ELEMENTS OF THE STRUCTURAL STIFFNESS MATRIX, AND SAVE THEM IN THE STEDIA ARRAY	PAND 690 PAND 700 PAND 710
	IF (LDIA •EQ• LDIAS) GO TO 70 LDIAS = LDIA STFDIA(I)=STRSTF(LDIA) STRSTF(LDIA)=0.0 GO TO 80	PAND 720 PAND 730 PAND 740 PAND 750 PAND 750
70 80	$STFDIA(I)=0.0$ $CONTINUE$ $SUMNZ = ISUMNZ$ $SUMGP = ISUMGP$ $IE (SUMNZ (SUMGP = GI_{0}, 3.0.) GO TO 110$	PAND 750 PAND 770 PAND 780 PAND 790 PAND 800 PAND 810
	IF HALF OR MORE OF THE ROW ELEMENTS ARE ZERO, REMOVE THEM AND PLACE THAT ROW IN ELEMENT FORMAT	PAND 820 PAND 830 PAND 840 PAND 850
	NULL=1 DO 90 J=1.ISUMGP LCSTF1=LCSTF1+1 IC=ISTSTF(LCSTF1)-1 DO 90 K=1.6 LCSTF1=LCSTF1+1 US (0.000 (CTS1)) +1 1 5 0) CO TO 00	PAND 860 PAND 870 PAND 880 PAND 890 PAND 900 PAND 910
	NULL=NULL+1 ISTF(NULL)=IC+K NULL=NULL+1 CTE(NULL)=IC+K	PAND 920 PAND 930 PAND 940 PAND 950
90	CONTINUE ISTF(1)=-NULL/2 LCSTF1=IROWL(I) DO 100 J=1,NULL LCSTF1=LCSTF1+1	PAND 980 PAND 970 PAND 980 PAND 990 PAND1000 PAND1010
100 110 C	STRSTF(LCSTF1)=STF(J) CONTINUE	PAND1020 PAND1030 PAND1040
C C	PRINI LAST LINE IN WRSTRK IF (INDWTS .NE. O) CALL WRSTRK(-1,-1,1)	PAND1050 PAND1060 PAND1070
	IF INDWKT •NE• 0, THEN THE LOCAL STIFFNESS AND TRANSFORMATION MATRICES MUST BE SAVED ON FILE 9	PAND1080 PAND1090 PAND1100 PAND1110

<pre>IF (INDWKT.EQ.0) GO TO 130 REWIND 2 DO 120 I=1;NBAR READ (2) (TRANSM(J,1),J=1,9) WRITE (9) (TRANSM(J,1),J=1,9) READ (2) (STIFML(J,1),J=1,144) 120 WRITE (9) (STIFML(J,1),J=1,144) 130 RETURN END</pre>	PAND1120 PAND1130 PAND1140 PAND1150 PAND1160 PAND1160 PAND1170 PAND1180 PAND1200
---	--

```
SOLV
SUBROUTINE SOLVE
                                                                              10
                                                                        SOLV
COMMON COM(30)
                                                                              20
EQUIVALENCE ( COM( 1), NJOINT )
                                                                        SOLV
                                                                              30
EQUIVALENCE ( COM( 8), NROW
                                 )
                                                                        SOLV
                                                                              40
EQUIVALENCE ( COM(11), INDISL )
                                                                        SOLV
                                                                              50
EQUIVALENCE ( COM(12), INDITR )
                                                                        SOLV
                                                                              60
EQUIVALENCE ( COM(13), ERRTOL )
                                                                        SOLV
                                                                              70
EQUIVALENCE ( COM(14), RELAXF )
                                                                        SOLV
                                                                              80
EQUIVALENCE ( COM(15), INDRLX )
                                                                        SOLV
                                                                              90
EQUIVALENCE ( COM(18), INDPLS )
                                                                        SOLV 100
EQUIVALENCE ( COM(19), MINRST )
                                                                        SOLV 110
EQUIVALENCE ( COM(26), TMAX )
                                                                        SOLV 120
COMMON / CMAIN /
                                                                        SOLV 130
     RJXYZ (100,3), IJRFM ( 74,6), IJALFM( 74,2), FRCVCT( 74,3),
                                                                        SOLV 140
1
     RMTVCT( 74,3), IBARP ( 130), IBARQ ( 130), IBARR (
KS ( 130), BAREA ( 130), BARIN ( 13C), BARIT (
2
                                                                        SOLV 150
                                                               1301.
                                                                        SOLV 160
3
                                                               130),
               130), BARYM ( 130), BARSM ( 130), RKN (
                                                                        SOLV 170
4
     BARJ
                                                               130),
           {
               130), SOLVEC( 444), RLIMTU( 88), RLIMTL(
r,
     RKT
                                                                        SOLV 180
            1
                                                                88).
                88), IROWL ( 445), STRSTF(1)
6
     DPRTIT(
                                                                        SOLV 190
DIMENSION SOLVC2(444), ISTSTF(1), STFDIA(444)
                                                                        SOLV 200
 DIMENSION RCG(444), APCG(444)
                                                                        SOLV 210
 EQUIVALENCE ( RJXYZ(1,1), NONROW(1) )
                                                                        SOLV 220
 EQUIVALENCE (
                 FRCVCT(1,1), SOLVC2(1) )
                                                                        SOLV 230
EQUIVALENCE ( STRSTF(1), ISTSTF(1))
EQUIVALENCE ( BAREA (1), STFDIA(1))
                                                                        SOLV 240
                                                                        SOLV 250
EQUIVALENCE ( BARJ(50), RCG(1) )
                                                                        SOLV 260
 COMMON / OVER3 / FORCMG(1)
                                                                        SOLV 270
 DIMENSION ALP(6), NONROW(300)
                                                                        SOLV 280
 DIMENSION ERRFMN(6)
                                                                        SOLV 290
 DIMENSION ERMRST(5)
                                                                        SOLV 300
 DIMENSION ERDIA(5)
                                                                        SOLV 310
DIMENSION ERRL(6)
                                                                        SOLV 320
 DATA ERRL /50H THE VARIABLE ASSOCIATED WITH THIS ROW HAS A CONST, SOLV 330
                                                                        SOLV 340
             10HANT VALUE
1
                               1
                                                                        SOLV 350
 DATA ERDIA /10H THE DIAGO, 10HNAL ELEMEN, 10HT FOR THIS,
              10H ROW CAN N. 10HOT BE ZERO /
                                                                        SOLV 360
1
 DATA ERMRST /10H THE NUMBE, 10HR OF RESTR, 10HAINTS IS L,
                                                                        SOLV 370
               10HESS THAN M, 10HINRST
                                                                        SOLV 380
1
                                            1
 DATA ALP /10HX--FORCE , 10HY--FORCE , 10HZ--FORCE
                                                                        SOLV 390
            10HX--MOMENT , 10HY--MOMENT , 10HZ--MOMENT /
                                                                        SOLV 400
1
DATA ERREMN / 10H TOO MANY , 10HROWS ARE B, 10HEING REMOV,
                                                                        SOLV 410
                10HED FOR THE, 10H SIZE OF N, 10HONROW
                                                                       /SOLV 420
1
 WRITE (6,420)
                                                                        SOLV 430
                                                                        SOLV 440
      SET-UP MINIMUM ERROR FOR THE AITKEN DELTA SQUARED PROCESS
                                                                        SOLV 450
                                                                        SOLV 460
 IF (INDRLX.EQ.0) ERRTL2=AMIN1(ERRTOL*ERRTOL,1.E-10)
                                                                        SOLV 470
                                                                        SOLV 480
     CALCULATE FORCE AND MOMENT VECTOR
                                                                        SOLV 490
                                                                        SOLV 500
 DO 10 I=1,300
                                                                        SOLV 510
 NONROW(I)=0
                                                                        SOLV 520
 DO 20 I=1,NROW
                                                                        SOLV 530
                                                                        SOLV 540
FORCMG(I)=0.0
 NULL=0
                                                                        SOLV 550
```

	DO 80 I=1,NJOINT IC=IJALFM(I,1) IF (IC.NE.0) GO TO 30 NULL=NULL+3	SOLV 560 SOLV 570 SOLV 580 SOLV 590
20	GO TO 50	SOLV 600
50	NULL=NULL+1	SOLV 620
40 50	FORCMG(NULL)=FRCVCT(IC,J)	SOLV 630
50	IC=IJALFM(1,2) IF (IC+NE+0) GO TO 60	SOLV 640
	NULL=NULL+3	SOLV 660
60	GO TO 80 DO 70 1=1-3	SOLV 670
00	NULL=NULL+1	SOLV 690
70	FORCMG(NULL)=RMTVCT(IC,J)	SOLV 700
80 C	CONTINUE	SOLV 710 SOLV 720
č	STORE ROW NUMBERS NOT REQUIRED IN THE SOLUTION IN NONROW	SOLV 730
C	INDISL = 0 IMPLIES NO INITIAL SOLUTION IS AVAILABLE	SOLV 740
C	INDISL = 0	SOLV 750
	NULL=0	SOLV 770
	J=0 I=0	SOLV 780
	DO 130 K=1.NJOINT	SOLV 800
	DO 130 L=1.6	SOLV 810
	I=I+1 COLVC2/IN-SCINEC(IN	SOLV 820
	IF (IJRFM(K)L).EQ.0) GO TO 100	SOLV 890
	NULL=NULL+1	SOLV 850
	IF (NULL•GT•300) CALL ERPNT1 (ERRFMN•6•1)	SOLV 860
	IC=IABS(IJRFM(K,L))	SOLV 880
	IF (DPRTIT(IC).EQ0) GO TO 90	SOLV 890
	INDISL = 1	SOLV 900
	SOLVLC(I)=DPRTIT(IC)	SOLV 920
	GO TO 130	SOLV 930
90	SOLVEC(I)=0.0	SOLV 940
	GO TO 130	SOLV 960
100	IF (STFDIA(I).EQ.0.0) GO TO 110	SOLV 970
	IF (INDRLX •EQ• 2) GO TO 130	SOLV 980
	$\begin{array}{c} \text{GO TO } 120 \end{array}$	SOLV 990
110	CONTINUE	SOLV1010
		SOLV1020
	GO TO 130	SOLV1050
120	CONTINUE	SOLV1050
	SOLVEC(I)=FORCMG(I)/STEDIA(I)	SOLV1060
130	CONTINUF	SOLV1080
* * * *	IF (J.NE.O) STOP	SOLV1090
<i>c</i>	NNONR W=NULL	SOLV1100
C		SULVIIIO

C C	ITERATION LOOP	SOLV1120 SOLV1130
140	IF (NNONRW.LT.MINRST) CALL ERPNT1 (ERMRST,5,1) IF (INDRLX .NE. 2) GO TO 1010 IF (INDISL .EQ. 1) INDISL = 1 FVMAX = 1.E-4	SOLV1140 SOLV1150 SOLV1160 SOLV1170 SOLV1180
	DO 1000 I = 1, NROW IF (FVMAX •LT• ABS(FORCMG(I))) FVMAX = ABS(FORCMG(I)) APCG(I) = 0.0	SOLV1190 SOLV1200
	RCG(I) = FORCMG(I) IF (INDISL •NE• 0) GO TO 1000 SOLVC2(I) = 0•0	SOLV1220 SOLV1230 SOLV1240
1000	SOLVEC(I) = FORCMG(I) CONTINUE	SOLV1250 SOLV1260
1010	I TRNO=0 I UPDAT=4 I NDURD=0	SOLV1270 SOLV1280 SOLV1290
145	INDOPT=0 GO TO 150 CONTINUE	SOLV1310 SOLV1320 SOLV1320
1015	DO 1015 I = 1, NROW $RCG(I) = FORCMG(I) - APCG(I)$ $SOLVEC(I) = PCG(I)$	SOLV1350 SOLV1350
150	ERRMAX=0.0 $PAP = 0.0$	SOLV1300 SOLV1370 SOLV1380
	NULL=1 ITRNO=ITRNO+1	SOLV1390 SOLV1400 SOLV1410
	IF (I.NE.NONROW(NULL)) GO TO 160 NULL=NULL+1	SOLV1420 SOLV1430 SOLV1440
160	SAVE=0.0 LOCSTF=IROWL(I)+1 IF (ISTSTF(LOCSTF).LT.0) GO TO 190	SOLV1450 SOLV1460 SOLV1470 SOLV1480
C C C	ROW IN BLOCK FORMAT, ONE COLUMN INDICATOR FOR 6 NON-ZERO	SOLV1490 SOLV1500 SOLV1510
170	ICOL=ISTSTF(LOCSTF)-1 DO 180 J=1,6 ICOL=ICOL+1	SOLV1520 SOLV1530 SOLV1540
180	SAVE=SAVE+STRSTF(LOCSTF)*SOLVEC(ICOL) LOCSTF=LOCSTF+1 IF (ISTSTF(LOCSTF)+LT+0) GO TO 210	SOLV1550 SOLV1560 SOLV1570 SOLV1580
C C	GO TO 170 ROW IN ELEMENT FORMAT, ONE COLUMN INDICATOR FOR EACH NON-ZERO	SOLV1590 SOLV1600 SOLV1610
C 190	NONELE=-ISTSTF(LOCSTF) DO 200 J=1,NONELE	SOLV1620 SOLV1630 SOLV1640
	LOCSTF=LOCSTF+1 ICOL=ISTSTF(LOCSTF) LOCSTF=LOCSTF+1	SOLV1650 SOLV1660 SOLV1670

200 SAVE=SAVE+STRSTF(LOCSTF)*SOLVEC(ICOL) SOLV1680 210 CONTINUE SOLV1690 IF (INDRLX .NE. 2) GO TO 1100 SOLV1700 APCG(I) = SAVE + STFDIA(I)*SOLVEC(I) SOLV1710 PAP = SOLVEC(I) * APCG(I) + PAPSOLV1720 RCG2 = RCG2 + RCG(I) * SOLVEC(I)SOLV1730 GO TO 270 SOLV1740 1100 CONTINUE SOLV1750 Ĉ SOLV1760 С SUBTRACT SAVE FROM FORCMG(I) AND DIVIDE BY THE DIAGONAL SOLV1770 ELEMENT (WHICH IS NOT CONSIDERED IN THE ABOVE CALCULATION) С SOLV1780 TO GET X(N+1) C SOLV1790 IF (INDRLX.EQ.0) GO TO 220 SOLV1800 Ċ SOLV1810 С USE OVERRELAXATION METHOD SOLV1820 С SOLV1830 SAVE=RELAXF*(FORCMG(I)-SAVE)/STFDIA(I)+(1.-RELAXF)*SOLVEC(I) SOLV1840 ERR1=ABS(SOLVEC(I)-SAVE) SOL V1850 GO TO 250 SOLV1860 SAVE=(FORCMG(I)-SAVE)/STFDIA(I) 220 SOLV1870 ERR1=ABS(SOLVEC(I)-SAVE) SOLV1880 IF (ITRNO.LT.IUPDAT) GO TO 250 SOLV1890 С SOLV1900 USE THE AITKEN DELTA SQUARED PROCESS TO IMPROVE THE С SOLV1910 С CONVERGENCE PROCESS SOLV1920 IF (ERR1/(ABS(SAVE)+1.E-14).LT.ERRTL2) GO TO 230 SOLV1930 ERR2=ABS(SOLVC2(I)-SOLVEC(I)) SOLV1940 SOLV1950 IF (ERR1.GE.ERR2) GO TO 230 SAVE1=SAVE-(SAVE-SOLVEC(I))**2/(SAVE-2.*SOLVEC(I)+SOLVC2(I)) SOLV1960 INDUPD=1 SOLV1970 ERE1=ABS(SOLVEC(I)-SAVE1) SOLV1980 SOLV(2(I)=SAVE1 SOLV1990 GO TO 240 SOLV2000 SOLV(2(I)=SAVE SOLV2010 230 SOLV2020 240 SOLVEC(I)=SAVE ERR=ERR1/(ABS(SOLVC2(I))+1.E-12) SOL V2030 GO TO 260 SOLV2040 CONTINUE 250 SULV2050 SOLVC2(I) = SOLVEC(I)SOLV2060 SOLVEC(I)=SAVE SOLV2070 ERR=ERR1/(ABS(SAVE)+1.E-12) SOLV2080 260 CONTINUE SOLV2090 SOLV2100 С IF (FRR.GT.ERRMAX) FRRMAX=FRR SOLV2110 SOLV2120 270 CONTINUE IF (INDRLX .NE. 2) GO TO 1270 SOLV2130 IF (ITRNO .NE. 1) GO TO 1220 SOLV2140 IF (INDISL NE. 0) GO TO 145 SOLV2150 1220 CONTINUE SOLV2160 IF (PAP \bullet EQ \bullet O \bullet) PAP = $1 \bullet$ E-9SOLV2170 ALPA = RCG2 / PAPSOLV2180 RCG2 = 0.0SOLV2190 DO 1240 J = 1, NROW SOLV2200 SOLVC2(J) = SOLVC2(J) + ALPA*SOLVEC(J) SOLV2210 RCG(J) = RCG(J) - ALPA* APCG(J)SOLV2220 ERRMAX = AMAX1(ABS(RCG(J)), ERRMAX)SOLV2230

```
1240 RCG2 = RCG2 + RCG(J)*APCG(J)
                                                                            SOLV2240
      ERRMAX = ERRMAX/ FVMAX
                                                                            SOLV2250
      BETA
            = -RCG2 / PAP
                                                                            SOLV2260
      DO 1260 J = 1, NROW
                                                                            SOLV2270
 1260 \text{ SOLVEC}(J) = \text{RCG}(J) + \text{BETA* SOLVEC}(J)
                                                                            SOLV2280
 1270 CONTINUE
                                                                            SOLV2290
      IF (INDUPD.EQ.0) GO TO 290
                                                                            SOLV2300
      DO 280 K=1.NROW
                                                                            SOLV2310
280
      SOLVEC(K)=SOLVC2(K)
                                                                            SOLV2320
      IUPDAT=IUPDAT+2
                                                                            SOLV2330
      INDUPD=0
                                                                            SOLV2340
290
      CONTINUE
                                                                            SOLV2350
С
                                                                            SOLV2360
C
          CHECK LOOP TERMINATION CONDITIONS
                                                                            SOLV2370
C
                                                                            SOLV2380
      IF (INDITR.EQ.ITRNO) GO TO 320
                                                                            SOLV2390
      IF (INDITR.LT.ITRNO) GO TO 310
                                                                            SOLV2400
      IF (INDOPT.GT.O) GO TO 320
                                                                            SOLV2410
      CALL SECOND (TIME)
                                                                            SOLV2420
      IF (TIME.GT.TMAX-5.) GO TO 300
                                                                            SOLV2430
      IF (ERRMAX.GT.ERRTOL) GO TO 150
                                                                            SOLV2440
      GO TO 320
                                                                            SOLV2450
      WRITE (6,440)
300
                                                                            SOLV2460
      INDPLS=0
                                                                            SOLV2470
      GO TO 320
                                                                            SOLV2480
310
      CONTINUE
                                                                            SOLV2490
      WRITE (6,450) INDITR, ERRMAX
                                                                            SOLV2500
      CONTINUE
320
                                                                            SOLV2510
                                                                            SOLV2520
      IF ( INDRLX .NE. 2 ) GO TO 1310
      DO 1300 J = 1, NROW
                                                                            SOLV2530
 1300 \text{ SOLVEC}(J) = \text{SOLVC2}(J)
                                                                            SOLV2540
      INDOPT = 2
                                                                            SOLV2550
 1310 CONTINUE
                                                                            SOLV2560
      CALL PNTFMV (SOLVEC, NJOINT, 1)
                                                                            SOLV2570
      WRITE (6,460) ITRNO, ERRMAX
                                                                            SOLV2580
                                                                            SOLV2590
      IUPDAT=IUPDAT+IUPDAT
      INDOPT=INDOPT+1
                                                                            SOLV2600
      IF (INDOPT.LT.2) GO TO 150
                                                                            SOLV2610
С
                                                                            SOLV2620
С
          FIND THE COMPLETE FORCE-MOMENT VECTOR
                                                                            SOLV2630
C
                                                                            SOLV2640
      DO 370 K=1,NNONRW
                                                                            SOLV2650
                                                                            SOLV2660
      SAVE=0.0
      I = NONROW(K)
                                                                            SOLV2670
      IF (SOLVEC(I).NE.0.0) SAVE=STFDIA(I)*SOLVEC(I)
                                                                            SOLV2680
      LOCSTF=IROWL(I)+1
                                                                            SOLV2690
      IF ( ISTSTF(LOCSTF) .EQ. 0 ) GO TO 370
                                                                            SOLV2700
      IF (ISTSTF(LOCSTF).LT.0) GO TO 350
                                                                            SOLV2710
      ICOL=ISTSTF(LOCSTF)-1
330
                                                                            SOLV2720
      DO 340 J=1,6
                                                                            SOLV2730
      ICOL=ICOL+1
                                                                            SOLV2740
      LOCSTF=LOCSTF+1
                                                                            SOLV2750
340
      SAVE=SAVE+STRSTF(LOCSTF)*SOLVEC(ICOL)
                                                                            SOLV2760
      LOCSTF=LOCSTF+1
                                                                            SOLV2770
      IF (ISTSTF(LOCSTF).LT.0) GO TO 370
                                                                            SOLV2780
      GO TO 330
                                                                            SOLV2790
```

350	NONELE=-ISTSTF(LOCSTF)	SOLV2800
	LOCSTF=LOCSTF+1	SOL V2820
	ICOL = ISISTE(LOCSTE)	SOL V2830
	LOCSTF=LOCSTF+1	SOLV2840
360	SAVE=SAVE+STRSTF(LOCSTF)*SOLVEC(ICOL)	SOLV2850
370	FORCMG(I)=SAVE	SOL V2860
	CALL PNTFMV (FORCMG+NJOINT+-1)	SOLV2870
С		SOLV2880
Ċ	INDRES = 0 IMPLIES NO PLASTICITY	SOL V2890
č		SOL V2900
-	IF (INDPLS.EQ.0) RETURN	SOLV2910
	NULL=0	SOLV2920
	INDBDL=NNONRW	SOLV2930
	LOCSTF=0	SOLV2940
	DO 410 I=1,NJOINT	SOLV2950
	DO 410 J=1,6	SOLV2960
	NULL=NULL+1	SOLV2970
	IF (IJRFM(I,J),EQ.0) GO TO 410	SOLV2980
	LOCSTF=LOCSTF+1	SOLV2990
	IF (IJRFM(I,J).LT.0) GO TO 410	SOLV3000
	ICOL=IJRFM(I,J)	SOLV3010
	IF (RLIMTU(ICOL)•GE•FORCMG(NULL)) GO TO 380	SOLV3020
	FORCMG(NULL)=RLIMTU(ICOL)	SOLV3030
	GO TO 390	SOLV3040
380	IF (REIMTL(ICOL)+LE+FORCMG(NULL)) GO TO 410	SOLV3050
	FORCMG(NULL)=RLIMTL(ICOL)	SOLV3060
390	NNONRW=NNONRW-1	SOLV3070
	IJRFM(I,J)=0	SOLV3080
	DO 400 K=LOCSTF,NNONRW	SOLV3090
400	NONROW(K)=NONROW(K+1)	SOLV3100
	LOCSTF=LOCSTF-1	SOLV3110
	IF (STFDIA(NULL)•EQ•O•O) CALL ERPNT1 (ERDIA•5•-1)	SOLV3120
	IF (SOLVEC(NULL)•NE•O•O) CALL ERPNT1 (ERRL,6,-1)	SOLV3130
	WRITE (6,470) ALP(J),I	SOLV3140
410	CONTINUE	SOLV3150
	CALL ERPNT2	SOLV3160
	IF (INDBDL•EQ•NNONRW) RETURN	SOLV3170
	INDISL = 1	SOLV3180
	GO TO 140	SOLV3190
C		SOLV3200
420	FORMAT (1H1)	SOLV3210
430	FORMAT (29H THE DIAGONAL ELEMENT FOR ROW, I4, 16H CAN NOT BE ZERO)	SOLV3220
440	FORMAT (66H SOLUTION ITERATIONS STOPPED BECAUSE JOB IS APPROACHIN	IGSOLV3230
	I TIME LIMIT)	SOLV3240
450	FORMAT (10HOMORE THANI5,28H ITERATIONS, MAXIMUM ERROR =E16.8)	SOLV3250
460	FORMAT (1X14,68H ITERATIONS WERE REQUIRED TO REACH A MAXIMUM RELA	TSOLV3260
	LIVE DIFFERENCE OF E14.7)	SOLV3270
470	FORMAT (5H THE Al0,37HRESTRAINT WAS VICLATED FOR JOINT NO. 14)	SOLV3280
	END	SOLV3290

	SUBROUTINE PNTFMV (ARRAY, NJGINT, IND)	PNTF	10
	COMMON COM(30)	PNTE	20
	EQUIVALENCE (COM (8), NROW)	PNIF	<i>2</i> 0
	DIMENSION ARRAY(I)	PNIC	40 60
		PNIF	50
	11=1	DNTE	20
			80
	$I = \{1, 0\} $	DNTE	90
c		PNTE	100
č	WRITE PAGE TITLE AT TOP OF NEXT PAGE	PNTE	110
ĉ		PNTF	120
-	LCOUNT=7	PNTF	130
	IF (IND.LT.0) GO TO 10	PNTF	140
	WRITE (6,50)	PNTF	150
	GO TO 20	PNTF	160
10	CONTINUE	PNTF	170
	WRITE (6,60)	PNTF	180
20	CONTINUE	PNTE	190
	WRITE (6,70) I, (ARRAY(K), K=11,12)	PNIF	200
	I = I I + 6		210
			220
20			240
50		PNTF	250
		PNTF	260
	XF=0.0	PNTF	270
	YE C O	PNTF	280
	ZF=0.0	PNTF	290
	DO 40 I=1, NJOINT	PNTF	300
	Κ=Κ+1	PNTF	310
	XF=XF+ARRAY(K)	PNTF	320
	YF=YF+ARRAY(K+1)	PNTF	330
	ZF=ZF+ARRAY(K+2)	PNTF	340
40	K=K+5	PNTE	350
	WRITE (6,80) XF,YF,ZF	PNIF	300
	SAV = 0		280
	DO 45 I = I, NROW		200
	45 SAV = AMAXI(SAV) ABS(ARRAY(I))	PNTE	400
	$AF = AF/SAV * 100 \bullet$	PNTE	410
	$TF = TF/SAV * 100 \bullet$	PNTF	420
	WPITE(6.90) XE-YE-ZE	PNTF	430
	RETURN	PNTF	440
C		PNTF	450
50	FORMAT (1H1,36X39HNODAL POINT DISPLACEMENTS AND ROTATIONS/44X24H	GLPNTF	460
	10BAL COORDINATE SYSTEM/1H0,4X11HNODAL POINT,5X3(2X12HDISPLACFMEN	T1PNTF	470
	2X),3(4X8HROTATION3X)/7X7HNUMBER,2(14X1HX,14X1HY,14X1HZ)///)	PNTF	480
60	FORMAT (1H1,40X30HNODAL POINT FORCES AND MOMENTS/44X24HGLOBAL CO	ORPNTF	490
	1DINATE SYSTEM/1H0,4X11HNODAL POINT,5X3(5X5HFORCE5X),3(5X6HMOMENT)	4XPNTF	500
	2)/7X7HNUMBER 2(14X1HX,14X1HY,14X1HZ)///)	PNTF	510
70	FORMAT (3X13)9X6(E15)6)		520
80	FURMAL (IHU)685HIULAL9885FID:00 AD FORMAT (IHU)685HIULAL9885FID:00 FORMAT (IHU)685 FORMAT (IHU)68		540
	YU FUKMAT (INU DADDHKELATIVE DULUTION ERRORD IN PLACENT /		550
		F 14 1 F	120

PNTF 560

4

END

```
SUBROUTINE FMBARS
                                                                                 FMBA
                                                                                       10
      COMMON COM(30)
                                                                                 FMBA
                                                                                        20
      EQUIVALENCE ( COM( 4), NBAR )
                                                                                        30
                                                                                 FMBA
      EQUIVALENCE ( COM ( 29 ), NSHPAN )
                                                                                 FMBA
                                                                                       40
      COMMON / CMAIN /
                                                                                 FMBA
                                                                                       50
           RJXYZ (100,3), IJRFM ( 74,6), IJALFM( 74,2), FRCVCT( 74,3),
                                                                                 FMBA
     1
                                                                                        60
           RMTVCT( 74,3), IBARP ( 130), IBARQ ( 130), IBARR (
KS ( 130), BAREA ( 130), BARIN ( 130), BARIT (
     2
                                                                                 FMBA 70
                                                                       130),
     3
                                                                        130),
                                                                                 FMBA
                                                                                       80
           BARJ ( 130), BARYM ( 130), BARSM ( 130), RKN ( 130),
RKT ( 130), SOLVEC( 444), RLIMTU( 88), RLIMTL( 88),
DPRTIT( 88), IROWL ( 445), STRSTF(1)
     4
                                                                                 FMBA
                                                                                       90
     5
                                                                                 FMBA 100
                                                                                 FMBA 110
     6
      DIMENSION RBARR(1)
                                                                                 FMBA 120
      EQUIVALENCE ( IBARR(1), RBARR(1) )
                                                                                 FMBA 130
      DIMENSION STIFML(12,12), TRANSM(3,3)
                                                                                 FMBA 140
      EQUIVALENCE ( RJXYZ(1,1), STIFML(1,1) )
                                                                                 FMBA 150
      EQUIVALENCE ( RJXYZ(50,2),TRANSM(1,1) )
                                                                                 FMBA 160
      DIMENSION IPR(6), DISPL(12), DISPL2(12), FMLSTF(12)
                                                                                 FMBA 170
      DATA IPR / 1, 2, 3, 7, 8, 9 /
                                                                                 EMBA 180
      LCOUNT = 70
                                                                                 FMBA 190
                                                                                 FMBA 200
      NPANEL = 0
      LCUNT2 = 70
                                                                                 FMBA 210
      I10R2 = 0
                                                                                 FMBA 220
С
           CALCULATE THE NO. OF NON SHEAR PANEL BARS
                                                                                 FMBA 230
      NBARI = NBAR - 2*NSHPAN
                                                                                 FMBA 240
      REWIND 2
                                                                                 FMBA 250
      DO 50 I=1,NBAR
                                                                                 FMBA 260
      IF ( I .LE. NBARI ) GO TO 8
                                                                                 FMBA 270
      IF ( LCUNT2 .LT. 56 ) GO TO 6
                                                                                 FMBA 280
      WRITE(6,80)
                                                                                 FMBA 290
      LCUNT2 = 7
                                                                                 FMBA 300
    6 CONTINUE
                                                                                 FMBA 310
                                                                                 FMBA 320
      BARLGT = ABS(RBARR(I))
      I10R2 = I10R2 + 1
                                                                                 FMBA 330
      GO TO 12
                                                                                 FMBA 340
    8 CONTINUE
                                                                                 FMBA 350
       IF (LCOUNT.LT.56) GO TO 10
                                                                                 FM3A 360
С
                                                                                 EMBA 370
С
           WRITE PAGE TITLE AT TOP OF NEXT PAGE
                                                                                 FMBA 380
С
                                                                                 FMBA 390
      LCOUNT=7
                                                                                 FMBA 400
      WRITE (6,60)
                                                                                 FMBA 410
10
      CONTINUE
                                                                                 FMBA 420
       IR=IBARR(I)
                                                                                 FMBA 430
   12 CONTINUE
                                                                                 FMBA 440
      IP = IEARP(I)
                                                                                 FMBA 450
       IQ=IBARQ(I)
                                                                                 FMBA 460
      READ (2) ((TRANSM(K,J),J=1,3),K=1,3)
                                                                                 FMBA 470
      READ (2) ((STIFML(K,J),J=1,12),K=1,12)
                                                                                 FMBA 480
C
C
                                                                                 FMBA 490
           SET UP DISPLACEMENT VECTOR
                                                                                 FMBA 500
С
                                                                                 FMBA 510
      JJ = (IO - 1) * 6
                                                                                 FMBA 520
      II = (IP - 1) * 6
                                                                                 FMBA 530
      DO 20 K=1,6
                                                                                 FMBA 540
      II = II + 1
                                                                                 FMBA 550
```

	I = IPR(K)	FMBA 560 FMBA 570
		FMBA 580
	I 1 = I 1 + 3	FMBA 590
20	DISPL(II) = SOLVEC(JJ)	FMBA 600
С		FMBA 610
С	TRANSFORM DISPLACEMENT VECTOR	FMBA 620
С		FMBA 630
	I I = -3	FMBA 640
	I ROW= 0	FMBA 650
	DO 30 K=1,4	FMBA 660
	II=II+3	FMBA 670
	DO_30_J=1,3	EMBA 680
	IROW=IROW+1	FMBA 690
	DISPL2(IROW)=0.0	FMBA 700
	DO 30 JJ=1,3	EMBA 720
20.		EMBA 720
30 r	DISPLZ(IROW)=DISPLZ(IROW)+IRANSM(3,33)*DISPL(II)	EMBA 740
ć	CALCULATE LOCAL FORCE-MOMENT VECTOR	EMBA 750
ĉ	CALCOLATE LOCAL FORCE HOMENT FLETOR	FMBA 760
C	$PO = 40 \ \text{K} = 1 \times 12$	EMBA 770
	FMLSTF(K) = 0	FMBA 780
	DO 40 J=1,12	FMBA 790
40	FMLSTF(K)=FMLSTF(K)+STIFML(K,J)*DISPL2(J)	FMBA 800
	IF (I •LE• NBARI) GO TO 48	FMBA 810
	SHFLOW = 2•*FMLSTF(1) / BARLGT	FMBA 820
	IF (IIOR2 •NE• 1) GO TO 42	FMBA 830
	IPS = IP	FMBA 840
	IQS = IQ	FMBA 850
	SHFSV = SHFLOW	FMDA 800
	GO TO 50	EMDA STU
	42 CONTINUE	EMBA 890
	NPANEL = NPANEL + I CHECK = (ABC(CHELOW) + ABS(CHECK)) * 5	EMBA 900
	SHESV - (ABSISTELUM) T ADSISTED V / / A · 2	EMBA 910
	SHESV = SIGN(SHESV) SHELOW / WRITE(4.00) NRANEL, IDS, IO, IOS, IP, SHESV	EMBA 920
	$\mathbf{T} = \mathbf{O}^{-1} \mathbf{T} \mathbf{O} \mathbf{P} \mathbf{O}^{-1} \mathbf{O} \mathbf{P} \mathbf{O}^{-1} \mathbf{O} \mathbf{O}^{-1} \mathbf{O} \mathbf{O}^{-1} $	EMBA 930
	1 GNZ = 1 GUNT2 + 2	FMBA 940
	GO TO 50	FMBA 950
	48 CONTINUE	FMBA 960
	WRITE (6,70) I, IP, IQ, IR, (FMLSTF(K), K=1,3), (FMLSTF(K), K=7,9), (FMLSTEBBA 970
	1F(K),K=4,6),(FMLSTF(K),K=10,12)	FMBA 980
	LCOUNT=LCOUNT+3	FMBA 990
50	CONTINUE	FMBA1000
	RETURN	FMSA1010
С		FMBA1020
60	FORMAT (1H1,49X22HBAR FORCES AND MOMENTS/49X24HLOCAL COORDINA	TE SYEMBA1030
	ISTEMS/1H0,29H BAR NODAL POINT NUMBERS,1IX3(5X5HFORCES))	3(5X6FMBA1040
	2HMOMENT4X)/30H NUMBER P Q R 94X2(14X1HX)14X1HY	,14X1FMDA1050
	382)///) FORMAT (2012 6012 6012 118 - DOINT D (4816.6/200-118 - 6	DOINT EMBALOGU
13	FURMAT (3A1390A1394A1394A139116 FUINT F 90L1340/29A9116 F 10 .4E15.67)	FMBA1080
	RO FORMAT(1H151X17HSHEAR PANEL FLOWS /	FMBA1090
	1 49X24HLOCAL COORDINATE SYSTEMS /	FMBA1100
	246HO PANEL NODAL POINT NUMBERS EQUIVALENT /	FMBA1110

346	SН	NUM	4B	ER	A	В	С		D		SHEAR FLOW,	///)	FMBA1120
90 FC	ORMA	Т	l	1НО,	15,	I8,	315,	7 X	E10.3	9	7XE10.3)			FMBA1130
E٨	١D													FMBA1140

.

OVERLAY (SASLP, 4, 0)

•

٨

	PROGRAM NLMDAL DIMENSION SM(444) DIMENSION AMASSV(102) DIMENSION AMASS(102), RJZ(1) DIMENSION AMASS(102), IRWKP(103) DIMENSION EGVALU(444), EGVECT(2220), WORK1(444), 1 WORK2(444), ADIA(444), BOFDIA(444) DIMENSION ROWNXT(444), IRWTOR(444), T(444) COMMON COM(30) EQUIVALENCE (COM(8), NROW) EQUIVALENCE (COM(21), NRWTOR) EQUIVALENCE (COM(22), IREDTO , MORDER) EQUIVALENCE (COM(23), NEIGVL)	NLMD 10 NLMD 20 NLMD 30 NLMD 50 NLMD 60 NLMD 70 NLMD 70 NLMD 80 NLMD 90 NLMD 100 NLMD 110 NLMD 120 NLMD 130
	COMMON / CMAIN /	NLMD 140
	2 RMTVCT(74,3), IBARP (130), IBARQ (130), IBARR (130),	NLMD 160
	3 KS (130), BAREA (130), BARIN (130), BARIT (130),	NLMD 170
	4 BARJ (130), BARYN (130), BARSM (130), RKN (130),	NLMD 180
	5 RKI (1307, SOLVECT 444), REIMIOU 607, REIMIEU 807, 6 DEPTIT(88), IROWI (4445), STRSTE(1)	NLMD 200
	EQUIVALENCE (RJXYZ(1,1),RJX (1)),(RJXYZ(1,2),RJY(1))	NLMD 210
	EQUIVALENCE (RJXYZ(1,3),RJZ(1)),(FRCVCT(1,1),SM(1), ADIA(1))	NLMD 220
	EQUIVALENCE (BAREA(1),ROWNXT(1)), (SOLVEC(1), IRWTOR(1))	NLMD 230
	FOUIVALENCE (REINTU(1), AMASS(1)), (REIMTL(15), IRWKP(1))	NLMD 240
	AQUIVALENCE (STRSTF(I), ISTSTF)	NLMD 250
	EQUIVALENCE (SIRSIF(5400), EGVALU)	NLMD 270
	FOULVALENCE (SIRSTE(7964), WORK1)	NEMD 280
	ENTITYALENCE (STRSTE(8408), MORK2)	NLMD 290
	EQUIVALENCE (STRSTF(9740), T)	NLMD 300
С	INITIALIZE THE PRINT ROUTINE	NLMD 310
	CALL PRNTO	NLMD 320
	DO 10 I=1.MORDER	NLMD 350
TO	AMASSV(1)=AMASS(1)	NLMD 340
		NLMD 344
	DO 11 I=1,MORDER	NLMD 346
11	SUM=SUM+AMASS(I)	NLMD 348
	SMMAX=SUM/ORDER	NLMD 350
	DC 12 I=1,MORDER	NLMD 352
12	AMASS(I)=AMASS(I)/SMMAX	NLMD 354
C	REDUCE STIFFNESS MATRIX	NEMD 350
	CALL REDUCE KOUNT=MORDER*MORDER	NLMD 360
	SUM=0.0	NLMD 362
	DO 13 I=1,MORDER	NLMD 364
	IDIAG=(I-1)*MORDER+I	NLMD 366
13	SUM=SUM+STRSTF(IDIAG)	NLMD 368
	SKMAX=SUM/ORDER	NLMD 370
1.6	DO 14 1=1,KOUNI	NLMU 3/2
14 C	51K51F(1)=51K51F(1)/ 5KMAX T	NIND 200
C	1	MERO 070

C			$MATRIX = ((M) * * - \bullet 5) * (K) * ((M) * * - \bullet 5)$		400
c		CALL	STFMAS (STRSTF,AMASS,MORDER)		420
C			HOUSEHOLDER METHOD TO OBTAIN MATRIX IN TRIPLE-DIAGONAL FORM		450 440 450
c		CALL	TRIDIA (MORDER,STRSTF,ADIA,BOFDIA)		460
C C			FIND THE NEIGVL + 6 SMALLEST EIGENVALUES		470 480 490
c		CALL	EIGVAL (MORDER, EGVALU, ADIA, BOFDIA, WORK1, WORK2, NEIGVL)		500 510
C C			FIND THE EIGENVECTORS FOR THE NEIGVL EIGENVALUES ABOVE		520 530
		RATI IGV1	O = SKMAX / SMMAX = 2	NLMD NLMD	535 540
		IGV2 REWI	= 6 ND 9	NLMD NLMD	550 560
	15	CONT IGV1	I NUE = IGV1 + 5	NLMD NLMD	570 580
		IGV2 J=1	= IGV2 + 5	NLMD NLMD	590 600
		DO 2 CALL	0 I=IGV1, IGV2 EIGVEC (MORDER, EGVALU(I), STRSTF, ADIA, BOFDIA, EGVECT(J), WORK1, W	NLMD NLMD	610 620
21	1	LORK2 EGVA) LU(I)=RATIO*EGVALU(I)	NLMD NLMD	630 635
20 C		J=J+	NROW	NLMD NLMD	640 650
C C			CALCULATE EIGENVECTOR FOR REDUCED MATRIX FORM	NLMD NLMD	660 670
		K=0 DO 5	0 M=1,5	NLMD NLMD	680 690
		SAVE DO 3	= 0.0 0 J=1.MORDER	NLMD NLMD	700 710
		I=K+ FGVF	J CT(I)=AMASS(J)*FGVFCT(I)	NLMD NLMD	720 730
30		IF (Cont	ABS(SAVE)•LT•APS(FGVECT(I))) SAVE=EGVECT(I) INUE	NLMD NLMD	740 750
		DO 4 I=K+	0 J=1•MORDER	NLMD NLMD	760 770
40 50		EGVE K=K+	CT(I)=EGVECT(I)/SAVF NROW	NLMD NLMD	780 790
С		CALL	. PRNT1 (EGVALU,EGVECT,AMASSV,IRWKP,NROW,MORDER,IRWTOR,WORK1)	NLMD NLMD	800 810
C C			TAKE ABOVE EIGENVECTORS AND USE THE TRANSFORMATION MATRICES TO GET FINAL EIGENVECTORS	NLMD NLMD	820 830
		IF (Call	<pre>IREDTO.EQ.NRCW) GO TO 60 . FNALEV (EGVECT.AMASS.MORDER.NEIGVL.NRCW.NRWTOR.T.MORK1.IRWKP.</pre>	NLMD NLMD	840 850
60]	LIRVT CONT	OR) INUE	NLMD MLMD	860 870
C C			PRINT	NLMD NLMD	880 890
C		CALL	PRNT2 (EGVALU, EGVECT, AMASSV, IRWKP, NROW, MORDER, IRWTOR, WORK1)	NLMD	900 910
		RETU	IRN	NLMD	920 930 940

.

	I DEN1 ENTRY	F PUNPACK PACK,UNPAC	к				РАСК РАСК	010 020
	VFD	42/4CPACK,	18/3				PACK	030
РАСК	DATA	0					PACK	040
	SAL	B2	***	DO NI			DACK	050
	582	XI	*	BZ = NL			DACK	070
	504	80		E4 = 0			DACK	010
	502	9		$B_{2} = 9$			PACK	000
	000 CP7	1		DO - OU P7 - 1			PACK	100
	501	BITOONE	-	$x_{5} = -0$	INITIALLI		PACK	110
WORDITS	177	¥7-¥7		$x_{7} = 0$			PACK	120
MONDITO	542	BITIONE		$X_2 = 0200000000$	600000000		PACK	130
	SA4	B1+B4		X4 = INPUTA(1)	+ 84)		PACK	140
	ZR	X4.FSTBIT		IF (X4 • EQ• 04	.0) GO TO FSTBIT		PACK	150
	BX7	X5		X7 = X5			PACK	160
	BX6	Χ4		X6 = X4			РАСК	170
	SA6	B3+65		OUTPUTA(1 + BS	5) = X6		PACK	180
	SB5	B5+1		B5 = B5 + 1			PACK	190
	ZR	FSTBIT		GO TO FSTBIT			PACK	200
LOOP	SA4	B1+B4		X4 = INPUTA(B)	1 + B4)		PACK	210
	ZR	X4,EQZERO		IF (X4 • EQ• 0	O) GO TO EQZERU)	PACK	220
	BX7	X7+X2		X = LOGICAL SU	JM X7 AND X2		PACK	250
	BX6	X4		X6 = X4			PACK	240
	SA6	83+85			35) = X6		PACK	250
507500	SES	85+1		$35 = 55 \pm 1$	VO 1 BIT PICHT		DACK	200
EQZERU	8XZ \$84	⊥ B4 ⊥1		BA = BA + 1	AZ I DII KIGH		PACK	280
LOIDII	504 E0	B4+1 B4+82+END(∼нк	IF (P4 • F0• B	2) GO TO ENDCHK		PACK	290
		B4,86,100	5	TE (B4 •NE• B	6) GO TO LOOP		PACK	300
	SA7	B3+B7		OUTPUTA(1 +	(B7) = X7		PACK	310
	SE7	B7+1		B7 = B7 + 1			PACK	320
	SB6	B6+60		B6 = B6 + 60			PACK	330
	7 R	WORDITS		GO TO WORDI			PACK	340
ENDCHK	SA7	B3+B7		OUTPUTA(1 +	B7) = X7		PACK	350
	SX6	R5		X6 = B5			PACK	360
	SA6	B3		OUTPUTA(1)	= X6		РАСК	370
	ZR	PACK		RETURN			PACK	380
	VFD	42/6CUNPA	СК ,18	/3			PACK	390
UNPACK	DATA	0					PACK	400
	SA2	B2	***	82 = NF			PACK	410
	SB2	X2	*				PACK	420
	SB4	9		B4 = 9	INITIALLY		PACK	430
	SH 5	0		B5 = 0	INITIALLY		PACK	440
	SA1	EI 	***				PACK	450
	SE6	XI	*	$B6 \neq LENGIH OF$				450
	587	0		B7 = 0			DACK	480
	582			X = 1			DACK	490
	186	AL=AL BA-BA-LOO	012		A) GO TO LOOP 12		PACK	500
WUKDJIS		049009LUU VE±01	r J 2.		+ X5 1		PACK	510
	584 CVE	X0701 X511		$X^{+} = X^{-} + 1$			PACK	520
	3A9 SB7	87+60		B7 = B7 + 60			PACK	530
	PI	X4.X4P1115		IF (X4 POSIT	IVE.) GO TO X4P	LUS	PACK	540
200101	SA3	B1+B4		X3 = INPUTA(1)	+ 84)		PACK	550

	BX7	X3	X7 = X3	PACK 560
	SA7	B3+B5	OUTPUTA(1 + B5) = X7	PACK 570
	SB4	B4+1	B4 = B4+1	PACK 580
	ZR	COMCOD1	GO TO COMCODI	PACK 590
X4PLUS	SA6	B3+B5	OUTPUTA($1 + 85$) = $X6 = 0.0$	PACK 600
COMCOD1	LX4	1	CIRCULAR SHIFT X4 1 BIT LEFT	PACK 610
	SB5	B5+1	B5 = B5+1	PACK 620
	EQ	B5,B2,UNPACK	IF (B5 •EQ• 32) RETURN	PACK 630
	EQ	B5,B7,WORDJTS	IF (B5 •EQ• B7) GO TO WORDJTS	PACK 640
	NE	B4,B6,LOOPJ1	IF (B4 •NE• B6) GO TO LOOPJ1	PACK 650
LOOPJ2	SA6	B3+B5	OUTPUTA(1 + $B5$) = $X6$ = 0.0	PACK 660
	SB5	85+1	B5 = B5 + 1	PACK 670
	NE	B5,B2,LOOPJ2	IF (B5 •NE• B2) GO TO LOOPJ2	PACK 680
	ZR	UNPACK	RETURN	PACK 690
BITOONE	DATA	0400000000000000	000000	PACK 700
BITIONE	DATA	020000000000000	000000	PACK 710
	END			PACK 720

.

```
REDU
                                                                                 10
     SUBROUTINE REDUCE
                                                                           REDU
                                                                                 20
     DIMENSION ISTSTF(1)
                                                                           REDU
                                                                                 30
     DIMENSION IOBUF(444)
                                                                           REDU
                                                                                 40
     COMMON / CMAIN /
          RJXYZ (100,3), IJRFM ( 74,6), IJALFM( 74,2), FRCVCT( 74,3),
                                                                           REDU
                                                                                 50
    1
          RMTVCT( 74,3), IBARP ( 130), IBARQ ( 130), IBARR (
                                                                           REDU
                                                                                 60
                                                                  130).
    2
                                                  130), BARIT (
                                                                  130),
                                                                           REDIT
                                                                                 70
                  130), BAREA (
                                 130), BARIN (
          KS
    3
                (
                                                  130), RKN
                                  130), BARSM (
                                                                           REDU
                   130), BARYM (
                                                               (
                                                                  130),
                                                                                 80
          BARJ
    4
                1
                    130), SOLVEC( 444), RLIMTU(
88), IROWL ( 445), STRSTF(1)
                                                                           REDU
                                                                                 90
                   130), SOLVEC(
                                                    88), RLIMTL(
                                                                   88) •
          RKT
     5
                (
                                                                           REDU 100
          DPRTIT(
     6
      DIMENSION IRWKP(103), SM(444), ROWNXT(444)
                                                                           REDU 110
                                                                           REDU 120
      DIMENSION T(444), IRCPST(444), IRWTOR(444)
                                                                           REDU 130
      EQUIVALENCE ( STRSTF(1), ISTSTF(1))
                                                                           REDU 140
      EQUIVALENCE ( FRCVCT(1,1), SM(1)), ( BAREA(1), ROWNXT(1))
      EQUIVALENCE ( SOLVEC(1) , IRWTOR(1)), ( RLIMTL(15), IRWKP(1))
                                                                           REDU 150
      EQUIVALENCE ( IROWL(1), IOBUF(1))
                                                                           REDU 160
                                                                           REDU 170
      EQUIVALENCE ( STRSTF(7964 ), IRCPST )
      EQUIVALENCE ( STRSTF(9740 ), T
                                                                           REDU 180
                                            )
                                                                           REDU 190
      INTEGER TAPEA, TAPEB
      COMMON COM(30)
                                                                           REDU 200
                                                                           REDU 210
      EQUIVALENCE ( COM ( 8), NROW )
      EQUIVALENCE ( COM ( 22), IREDTO)
                                                                           REDU 220
                                                                           REDU 230
      EQUIVALENCE ( COM ( 21), NRWTOR)
                                                                           REDU 240
      EQUIVALENCE ( COM ( 28), INDWTS )
                                                                           REDU 250
                                                                           REDU 255
      DIMENSION ERSD(13)
                                                                           REDU 260
      DIMENSION ERZD(12)
                 150H TOO MANY ROWS HAVE BEEN REDUCED FROM THE STIFFNESREDU 262
      DATA ERSD
     1,60HS MATRIX LEADING TO A VERY SMALL DIAGONAL ELEMENT IN A ROW B, REDU 264
                                                                           REDU 266
     220HEING REMOVED
                  /50H TOO MANY ROWS HAVE BEEN REDUCED FROM THE STIFFNESREDU 270
      DATA ERZD
     1,60HS MATRIX LEADING TO A ZERO DIAGONAL ELEMENT IN A ROW BEING R, REDU 280
                                                                           REDU 285
     210HEMOVED
                    1
                                                                           REDU 290
      TAPEA = 1
                                                                           REDU 300
      TAPEB = 2
                                                                           REDU 310
          SET UP IRWTOR WITH THE ROWS TO BE REDUCED IN DESCENDING ORDER
                                                                           REDU 320
                                                                           REDU 330
                                                                           REDU 340
      IRWKP(IREDTO+1)=0
                                                                           REDU 350
      1 = 1
                                                                           REDU 360
      IRWTOR(1)=0
      IF (IREDTO.EQ.NROW) GO TO 30
                                                                           REDU 370
                                                                           REDU 380
      NRWTOR=NROW-IREDTO
                                                                           REDU 390
      LAST=NRWTOR+1
                                                                           REDU 400
      DO 20 I=1.NROW
                                                                           REDU 410
      IF (I.EQ.IRWKP(J)) GO TO 10
                                                                            REDU 420
      LAST=LAST-1
                                                                            REDU 430
      IRWTOR(LAST)=I
                                                                            REDU 440
      GO TO 20
                                                                            REDU 450
10
      J = J + 1
                                                                            REDU 460
20
      CONTINUE
                                                                            REDU 470
30
      CONTINUE
                                                                            REDU 480
          WRITE STIFFNESS MATRIX ON TAPEA BY ROWS, WITH THE EXCEPTION
                                                                            REDU 490
                THAT THE FIRST ROW TO BE REMOVED IS STORED IN ROWNXT
                                                                            REDU 500
```

C

С

С

Ċ

336

С

С

С

С		REDU	510
	REWIND TAPEA	REDU	520
	L=1	REDU	530
	DO 160 I=1, NROW	REDU	540
	IF=1	REDU	550
40	L=L+1	REDU	560
	IF (ISTSTF(L)+LT+0) GO TO 110	REDU	570
	IC=ISTSTF(L)	REDU	580
	IF (IC.GT.IF) GO TO 70	REDU	590
50	IL=IC+5	REDU	600
	DO 60 J=IF,IL	REDU	610
	L=L+1	REDU	620
	IF (ABS(STRSTF(L)).LT.1.E-8) STRSTF(L)=0.0	REDU	630
	IF (STRSTF(L).EQ.0.0) GO TO 60	REDU	640
	IF (INDWTS .NE. 0) CALL WRSTRK (I. J. STRSTF(L)) REDU	650
60	SM(J)=STRSTF(L)	REDU	660
	IF=IL+1	REDU	670
	GO TO 40	REDU	680
70	II = IC-1	REDU	690
	IF (IF.EQ.O.OR.IL.EQ.O) GO TO 80	REDU	700
	IF (IABS(IF).LT.444.AND.IABS(IL).LT.444) GO TO 90	REDU	710
80	CONTINUE	REDU	720
	WRITE (6,340) IF,IL	REDU	730
	IF=L-200	REDU	740
	IL=L+200	REDU	750
	WRITE (6,350) (STRSTF(J), J=IF, IL)	REDU	760
	WRITE (6,360) (ISTSTF(J),J=IF,IL)	REDU	770
	STOP	REDU	780
90	CONTINUE	REDU	790
	DO 100 J=IF,IL	REDU	800
100	SM(J)=0.0	REDU	810
	IF=1C	REDU	1 820
	GO TO 50	REDU	1 830
110	CONTINUE	REDU	840
	IF (IF.GT.NROW) GO TO 130	REDU	850
	DO 120 J=IF,NROW	REDU	860
120	SM(J)=0.0	REDU	870
130	CONTINUE	REDU	880
	IF (I.NE.IRWTOR(1)) GO TO 150	REDU	1 890
	DO 140 J=1,NROW	REDU	900
140	ROWNXT(J)=SM(J)	REDU	910
	GO TO 160	REDU	1 920
С	PACK AND WRITE THE ROW IN BLOCK FORMAT	REDU	930
150	CALL PACK (SM(1), NROW, IOBUF(1))	REDU	J 940
	KWRIT = IOBUF(1)	REDU	J 950
	WRITE(TAPEA) KWRIT, (IOBUF(J),J=1,KWRIT)	REDU	J 960
160	CONTINUE	REDL	J 970
C	PRINT LAST LINE	REDU	1 980
	IF (INDWTS .NE. O) CALL WRSTRK(-1,-1,1.)	REDU	J 990
	DO 170 J=1,NROW	REDU	1000
170	IRCPST(J)≃J	REDU	1010
С		REDU	1020
с	SET NUMBER OF ACTIVE ROWS	REDU	1030
С		REDU	1040
	NACTRW=NROW	REDU	11050
	KREAD = 0	REDU	J1060

a

	KREAD1 = 1 IF (IREDTO.EQ.NROW) GO TO 320	REDU1070 REDU1080
С		REDU1090
с	START OF LOOP TO REDUCE STRUCTURAL STIFFNESS MATRIX	REDU1100
С		REDU1110
	ICOUNT=0	REDU1112
	SAVTOT=0.	REDU1114
	KWRIT1= 4402	REDU1120
	KWRIT = 2502	REDU1130
	REWIND 9	REDU1140
	IRWTOR(NRWTOR+1)=0	REDU1150
	DO 310 IRWR=1,NRWTOR	REDUIIGO
	REWIND TAPEA	REDUIT70
	REWIND TAPEB	REDUIISO
	IROW=IRWTOR(IRWR)	REDUII90
	IR1=IRCPST(IROW)	REDUI200
~	IF (IRWR.NE.NRWICK) GO TO ISO	REDUIZIO
.C	REDUCE KWRITI TO FIT LATER BUFFER LIMITS	REDUIZZO
	KWRI/1= 2830	REDUI250
~	IRZEU	REDUI240
Č	IRI CORRENT POSITION (ROW + COL. NO.) OF ROW BEING	
Ċ	LEAST CURRENT POSITION OF NEXT ROW TO BE REMOVED	REDU1200
c	TRZ CORRENT POSITION OF NEXT ROW TO BE REMOVED	REDU1280
C	60 10 190	REDU1290
180	IR2=IRWTOR(IRWR+1)	REDU1300
100		REDU1310
190	CONTINUE	REDU1320
170		REDU1330
c		REDU1340
č	BUILD TRANSFORMATION MATRIX OF ORDER (NACTRW)*(NACTRW-1)	REDU1350
č	SAVE ONLY LAST ROW, SINCE FIRST NACTRW-1 ROWS ARE THE	REDU1360
Ċ	IDENTITY MATRIX	REDU1370
	NACTR2=NACTRW-1	REDU1380
	SAVE=ROWNXT(IR1)	REDU1390
	ICOUNT=ICOUNT+1	REDU1392
	SAVTOT = SAVTOT + ABS(SAVE)	REDU1394
	AVSAVE=SAVTOT/ICOUNT	REDU1396
	IF(SAVE•EQ•0•) CALL ERPNT1(ERZD•12•1)	REDU1400
	IF(ABS(SAVE)/AVSAVE.LT.1.E-06) CALL ERPNT1(ERSD,13,0)	REDU1405
	DO 200 I=1,IRR	REDU1410
200	T(I)=-ROWNXT(I)/SAVE	REDU1420
	IF (IR1 •GT• NACTR2) GO TO 212	REDU1430
	DO 210 I=IR1,NACTR2	REDU1440
210	T(I)=-ROWNXT(I+1)/SAVE	REDU1450
212	CONTINUE	REDU1460
C	********	REDUI470
C	CAVE LAST DOW OF TRANSFORMATION MATRIX ON FILE UNIT O	REDUI400
C	SAVE LAST NUW OF TRANSFORMATION MATRIX ON FILE UNIT 9	REDUI 500
L L	WRITE(9) (T(K), $K-1$, NA(TR2)	REDU1510
c	W(X + U(Y) + U	REDU1520
		REDU1530
Ċ	START LOOP TO PROCESS ACTIVE ROW IN COORDINATE REDUCTION	REDU1540
č	OTHER FOR TO THOUSON HETTE NOW IN COMPLETE HEADERFOR	REDU1550
<u> </u>	DO 290 ICRNTR=1.NACTR2	REDU1560

.

C.	READ STIFFNESS MATRIX ROW, ICRNTR		REDU1570
•	IF (KREAD .GT. KREAD1) GO TO 215		REDU1590
	KREAD1 = 1		REDU1600
	READ (TAPEA) KREAD (STRSTF(I), I=1, KREAD)		REDU1610
215	CALL UNPACK (SIRSIF(KREADI), NACTRW, SM(1))		REDU1620
C	KREADI - KREADI + ISISIF(KREADI)		REDUI630
c			REDU1650
`	SAVE=SM(IR1)		REDU1660
	IF (SAVE.NE.O.) GO TO 230		REDU1670
	IF (IR1 .GT. NACTR2) GO TO 260		REDU1680
	DO 220 I=IR1,NACTR2		REDU1690
220	SM(1) = SM(1+1)		REDU1700
230	00 10 200 D0 200 I=1. IPP		REDULTIO
240	SM(I) = SM(I) + SAVF*T(I)		REDULTZO
210	IF (IR1 .GT. NACTR2) GO TO 260		REDU1740
	DO 250 I=IR1,NACTR2		REDU1750
250	SM(I) = SM(I+1) + SAVE * T(I)		REDU1760
260	CONTINUE		REDU1770
C			REDU1780
C	IF THIS ROW IS THE NEXT TO BE REMOVED, SAVE THE ROW	IN ROWNXT	REDU1790
C	LE (LCENTE NE LES) CO TO 290		REDU1800
	DO 270 I=1.NACTR2		REDUI810
270	ROWNXT(I) = SM(I)		REDU1830
2.0	IF (ICRNTR •NE• NACTR2) GO TO 290		REDU1840
	IF (KWRIT •NE• 2502) GO TO 285		REDU1850
	GO TO 290		REDU1860
C			REDU1870
C	WRITE THE REDUCED ROW ON TO TAPEB		REDU1880
Ċ			REDUI890
280	CALL PACK (SM(1), NACTR2, STRSTE(KWRIT))		REDUI900
200	KWRIT = KWRIT + ISTSTF(KWRIT)		REDU1920
	IF (ICRNTR .EQ. NACTR2) GO TO 285		REDU1930
	IF (KWRIT •LT•KWRIT1) GO TO 287		REDU1940
285	ISTSTF(2501) = KWRIT - 2502		REDU1950
	KWRIT = KWRIT - 1		REDU1960
	WRITE(TAPEB)(STRSTF(I), I= 2501, KWRIT)		REDU1970
207	KWRII = 2502		REDU1980
281	CONTINUE		REDUI990
290	CONTINUE		REDU2000
c			REDU2020
č	REVERSE READ/WRITE TAPE VALUES		REDU2030
С			REDU2040
	I = TAPEA		REDU2050
	TAPEA=TAPEB		REDU2060
	TAPEB=I		REDU2070
Ċ	******		REDU2080
c	UPDATE IRCEST ARRAY TO ADJUST FOR THE ROW + COLUMN	REMOVED	REDUZU90
č	GEATE THE OF ANAL TO ADDODT TON THE ROW T COLUMN		REDU2110
_	IRCPST(IROW)=-IRCPST(IROW)		REDU2120

REDU2130 K=IROW+1 REDU2140 DO 300 I=K,NROW REDU2150 IF (IRCPST(I).LT.0) GO TO 300 REDU2160 IRCPST(I)=IRCPST(I)-1 REDU2170 CONTINUE 300 REDU2180 С REDU2190 С UPDATE NO. OF ACTIVE ROWS REDU2200 С REDU2210 NACTRW=NACTR2 310 CONTINUE REDU2220 REDU2230 С REDU2240 COPY STIFFNESS MATRIX TO CORE С REDU2250 С REDU2260 320 CONTINUE **REDU2270** IRR=0 REDU2280 REWIND TAPEA REDU2290 DO 330 I=1,NACTRW С REDU2300 BREAK DOWN THE INPUT BLOCKS REDU2310 С **REDU2320** С IF (KREAD .GT. KREAD1) GO TO 325 REDU2330 REDU2340 KREAD1 = 1READ (TAPEA) KREAD, (IOBUF(I), I=1, KREAD) REDU2350 325 CALL UNPACK (IOBUF(KREAD1), NACTRW, SM(1)) REDU2360 REDU2370 KREAD1 = KREAD1 + IOBUF(KREAD1) REDU2380 DO 330 J=1,NACTRW REDU2390 IRR=IRR+1 REDU2400 330 STRSTF(IRR)=SM(J) С REDU2410 č REVERSE THE RECORD POSITION IN THE TRANSFORMATION **REDU2420** С MATRIX FILE AND PLACE ON UNIT 1 REDU2430 REDU2440 С IF(IREDTO.EQ.NROW) RETURN REDU2445 REDU2450 REWIND 1 NACTRW = IREDTO REDU2460 DO 335 I = 1, NRWTOR REDU2470 REDU2480 BACKSPACE 9 REDU2490 READ(9)(SM(L),L=1, NACTRW) REDU2500 WRITE(1)(SM(L),L=1, NACTRW) REDU2510 BACKSPACE 9 NACTRW = NACTRW + 1REDU2520 REDU2530 335 CONTINUE **REDU2540** С REDU2550 RETURN REDU2560 C FORMAT (7H IF IL ,2110) REDU2570 340 350 FORMAT (10E12.5) REDU2580 REDU2590 FORMAT (10112) 360 REDU2600 END

	SUBROUTINE STFMAS (STRSTF, AMASS, MORDER)	STFM 10
	DIMENSION STRSTF(MORDER,MORDER), AMASS(1)	STFM 20
С		STFM 30
C	COMPUTE (M)**-1/2 = ((M)**-1/2)T FOR DIAGONAL MATRIX	STFM 40
С		STFM 50
	DO 10 I=1,MORDER	STFM 60
10	AMASS(I)=1/SQRT(AMASS(I))	STEM 70
С		STFM 80
С	COMPUTE ((M)**-1/2)T * (K) * (M)**-1/2	STFM 90
С	STORE THE RESULT IN LOWER TRIANGLE INCLUDING DIAGONAL	STFM 100
С	FORMAT IN K	STFM 110
	DO 20 I=1.MORDER	STEM 120
	DO 20 J=I,MORDER	STFM 130
20	STRSTF(I,J)=STRSTF(I,J)*AMASS(J)	STEM 140
-	LOCSTF=0	STEM 150
	DO 30 I=1.MORDER	STEM 160
		STEM 170
		STEM 180
20		STEM 190
00		STEM 200
	RETORN	STEM 200
	END	STEM 210

a

	SUBROUTINE FNALEV (EGVECT, AMASS, MORDER, NEIGVL, NROW, NRWTOR, T, T2, I 1KP, IRWTOR)	RWFNAL	10
	DIMENSION T2(1), IRWKP(1), IRWTOR(1)	ENAL	30
	DIMENSION AMASS(1), EGVECT(1), $T(1)$	FNAL	40
С		FNAL	50
С	CALCULATE EIGENVECTORS FOR THE NON-REDUCED MATRIX FROM	FNAL	60
С	TRANSFORMATION MATRICES STORED IN FILE 9 BY SUBROUTINE	FNAL	70
С	REDUCE	FNAL	80
С		FNAL	90
	REWIND 1	FNAL	100
	NACTRW=MORDER	FNAL	110
	DO 90 I=1.NRWTOR	FNAL	120
	READ (1) $(T(L),L=1,NACTRW)$	FNAL	130
	IF (I.EQ.1) GO TO 40	FNAL	140
	I 4 = 1	FNAL	150
	I 1 = 1	FNAL	160
	I2=NRWTOR	FNAL	170
	I 3=MORDER+1	FNAL	180
	DO 30 J=1.NACTRW	FNAL	190
	IF (I4.EQ.I) GO TO 10	FNAL	200
	IF (IRWKP(II).GT.IRWTOR(I2)) GO TO 20	FNAL	210
10	T2(I1)=T(J)	FNAL	220
	I 1 = I 1 + 1	FNAL	230
	GO TO 30	FNAL	240
20	$T_2(I_3) = T(J)$	FNAL	250
	I 2 = I 2 - 1	FNAL	260
	I 3=I 3+1	FNAL	270
	I 4 = I 4 + 1	FNAL	280
30	CONTINUE	FNAL	290
	GO TC 60	FNAL	300
40	CONTINUE	FNAL	310
	DO 50 J=1•NACTRW	FNAL	320
50	$T_2(J) = T(J)$	FNAL	330
60	CONTINUE	FNAL	340
	κ = 0	FNAL	350
	DO 80 J=1,5	FNAL	360
	SAVE=0.0	FNAL	370
	DO 70 M=1,NACTRW	FNAL	380
	L=K+M	FNAL	390
7 0	SAVE=SAVE+T2(M)*EGVECT(L)	FNAL	400
	EGVECT(L+1)=SAVE	FNAL	410
- 80	K=K+NROW	FNAL	420
	NACTRW=NACTRW+1	FNAL	430
90	CONTINUE	FNAL	440
	RETURN	FNAL	450
	END	FNAL	460

	SUBROUTINE EEPNT (EGVALU, EGVECT, NEIGVL, NROW)	EEPN	10
	DIMENSION EGVALU(1), EGVECT(1)	EEPN	20
С		EEPN	30
	К1=1	EEPN	40
	K2=NROW	EEPN	50
	DO 10 I=1,NEIGVL	EEPN	60
	WRITE (6,20) I,EGVALU(I)	EEPN	70
С	IF (ILT. 7) GO TO 40	EEPN	80
	WRITE (6,30) (EGVECT(K),K=K1,K2)	EEPN	90
	K1=K1+NROW	EEPN	100
	K2=K2+NROW	EEPN	110
10	CONTINUE	EEPN	120
	RETURN	EEPN	130
С		EEPN	140
20	FORMAT (13HO EIGENVALUE(,I2,3H) =,E14.7)	EEPN	150
30	FORMAT (10X30H THE CORRESPONDING EIGENVECTOR/(10X10E12•4))	EEPN	160
	END	EEPN	170

	SUBROUTINE TRIDIA (NR,R,A,PQ)	TRID	10
С	TRI-DIAGONALIZES SYMMETRIC MATRIX BY HOUSEHOLDER METHOD	TRID	20
С	NR = ORDER OF MATRIX	TRID	30
С	R = 1-DIMENSIONAL ARRY OF NR/2*(NR+1) ELEMENTS	TRID	40
C	1. CONTAINS LOWER TRIANGULAR PART PLUS DIAGONAL OF MATRIX	TRID	50
C	2. AFTER TRIPLE-DIAGONALIZATION CONTAINS THE DIAGONAL PLUS		. 70
C	* VECTORS	TRID	80
C	A = I-DIMENSIONAL ARRAY OF NR ELEMENTS, CONTAINS DIAGONAL		90
C	ELEMENTS OF TRIPLE-DIAGONAL MATRIA	TRID	100
C	PQ - I-DIMENSIONAL ARRAY OF MR LELEMENTSY	TRID	110
c	2. CONTAINS ELEMENTS OF A VECTOR	TRID	120
č	3. CONTAINS THE OFF-DIAGONAL TERM OF TRIPLE-DIAGONAL MATRIX	TRID	130
ć		TRID	140
C	DIMENSION $R(1) \cdot PQ(1) \cdot A(1)$	TRID	150
		TRID	160
	NR1=NR-1	TRID	170
	DO 180 I=2;NR1	TRID	180
	IA=IA+I	TRID	190
С	CALCULATE ELEMENTS OF W VECTOR	TRID	200
	S=0•	TRID	210
	JAIA	TRID	220
	DO 10 J=I,NR		230
	S=S+R(JA)**2	TRID	240
	L+AL=AL		250
10	CONTINUE		200
	SS=SQRT(S)	TRID	280
	PG(I-I) = -SIGN(SS)R(IA)	TRID	290
20	IF (5) 209190920	TRID	300
30	A = -F(1, 1)/3 B(1A) = SOB(1, SOD*(1, OD+B(1A)*X))	TRID	310
20		TRID	320
40	$R(I\Delta) = 0$	TRID	330
	I = (R(IA)) = 60,200,60	TRID	340
60	x = 500 * x / R (IA)	TRID	350
70	JA=IA	TRID	360
	DO 80 J=I•NR1	TRID	370
		TRID	380
	R(JA) = X + R(JA)	TRID	390
80 C8	CONTINUE	TRID	400
С	CALCULATE ELEMENTS OF P VECTOR	IRID	410
	JAI=IA+1	TRID	420
	DO 130 J=I,NR		450
	JA=JAI		440
	KA = IA		450
		TRID	400
	DO(120 N-19)NR	TRID	480
	$F = (K - 1) = 0.100 \cdot 100$	TRID	490
00	11 (N=07 2091009100 18- 1813	TRID	500
20		TRID	510
100	JA=JA+K	TRID	520
110	KA=KA+K	TRID	530
120	CONTINUE	TRID	540
	JAI=JAI+J	TRID	550

130	CONTINUE	TRID 560
С	CALCULATE ELEMENTS OF Q VECTOR	TRID 570
	AK=0•	TRID 580
	AI=AL	TRID 590
	DO 140 J=I,NR	TRID 600
	AK=AK+R(JA)*PQ(J)	TRID 610
		TRID 620
140	CONTINUE	TRID 630
	AK=2•00*AK	TRID 640
	AI=AL	TRID 650
	DO 150 J=I,NR	TRID 660
	$PQ(J) = 2 \cdot 00 \cdot PQ(J) - AK \cdot R(JA)$	TRID 670
	L+AL=AL	TRID 680
150	CONTINUE	TRID 690
С	CALCULATE ELEMENTS OF NEW R MATRIX	TRID 700
	KK = I A	TRID 710
	JA=IA	TRID 720
	DO 170 J=I,NR	TRID 730
	KA=IA	TRID 740
	DO 160 K=I,J	TRID 750
	KK=KK+1	TRID 760
	R(KK)=R(KK)-PQ(K)*R(JA)-R(KA)*PQ(J)	TRID 770
	КА=КА+К	TRID 780
160	CONTINUE	TRID 790
	KK=KK+I-1	TRID 800
	L+AL=AL	TRID 810
170	CONTINUE	TRID 820
180	CONTINUE	TRID 830
	GO TO 210	TRID 840
190	X=0•	TRID 850
	GO TO 30	TRID 860
200	X=0.	TRID 870
	GO TO 70	TRID 880
C	SORT ALPHAS AND BETAS	TRID 890
210	0 = A I	TRID 900
	DO 220 I=1.NR	TRID 910
	IA=IA+I	TRID 920
	A(I) = R(IA)	TRID 930
220	CONTINUE	TRID 940
	PQ(NR) = R(1A-1)	TRID 950
	N=NR	TRID 960
230		TRID 970
	PQ(N) = PQ(N-1)	TRID 930
24.0	1r (2-N) 23092409240	IKID 990
24()		IRID1000
		TRIDIOIO
		TRID1020

·

345

٨

•
	SUBROUTINE EIGVAL (IP+F+A+B+F+W+MVAL)	FIVL	10
С	FINDS EIGENVALUES OF TRI-DIAGONAL MATRICES BY ORTEGA METHOD	FIVL	20
č	LP = ORDER OF MATRIX	ĒĪVL	зŏ
С	E = 1-DIMENSIONAL ARRAY OF LP ELEMENTS. THE ELGENVALUES ARE	FIVL	40
Ċ	STORED IN THIS ARRAY IN ASCENDING ORDER	EIVL	50
č	A = 1-DIMENSIONAL ARRAY OF LP FLEMENTS. CONTAINS DIAGONAL	EIVL	60
Ċ	FLEMENTS OF TRIPLE-DIAGONAL MATRIX	EIVL	70
c	B = 1-DIMENSIONAL ARRAY OF LP ELEMENTS. CONTAINS OFF-DIAGONA	LEIVL	80
c	FLEMENTS OF TRIPLE-DIAGONAL MATRIX	FTVI	90
ĉ	E AND W = 1-DIMENSION ARRAY OF LP FLEMENTS FACH USED AS	FTVI	100
c	WORK AREAS	FIVI	110
ĉ	MVAL = NUMBER OR EIGENVALUES TO BE CALCULATED.	FIVE	120
C	DIMENSION $F(1) \cdot A(3) \cdot F(1) \cdot F(1) \cdot W(1)$	EIVL	130
C	FIND UPPER AND LOWER BOUNDS AND NORMALIZE INPUT	FIVL	140
C	BDEARS(A(1))	FIVL	150
		FIVI	160
10	BD = AMAX1(BD + ABS(A(1)) + B(1) * * 2 • 0 • 00)	FIVL	170
-	BD=BD+1	FIVL	180
		EIVL	190
	$\Delta(I) = \Delta(I) / BD$	FIVI	200
	B(1) = B(1) ZBD	FIVL	210
		ETVI	220
20		FIVE	230
20		FIVI	240
30	$I = \frac{1}{2} \left(\frac{1}{2} \left(\frac{1}{2} \right) \left($	OFIVE	250
50		FIVI	260
4.0	Y-(W/K)+F(K))*-5	FIVE	270
40 C	EIND NUMBER OF EIGENVALUES. N. GREATER THAN OR EQUAL TO X	FIVI	280
C	S2=1.	FIVE	290
		FIVI	300
	F(1) = A(1) - Y	FIVE	310
		FIVE	320
50		FIVE	330
50		FIVE	340
			250
			360
6 7			270
00			280
			200
70			400
10			400
00			410
00	$\frac{1}{10} = \frac{1}{100} = \frac{1}{$		420
100	[(ABS(r(1-1))+ABS(r(1-2))-1)(0-15)] = 100,110,110		450
100			440
110	F (1 = 2) = F (1 = 2) * 1 • 0 E 1 2 E (1) = (/ (1) = 2) * E (1 = 1) = 2 (1) * * 2 * E (1 = 2)		490
110	$\Gamma(1) - (M(1) - A) A \Gamma(1 - 1) - O(1) A A Z A (1 - 2)$	ETVL	400
120	GU TU 140 5/11-///IN-VX8CIGN/1 00 C1)		470
120	$r(1) - (R(1) - K) \wedge S1 G((1 + 0)) + S1)$		400
100			
140	F(1) = (A(1) + A(N) + F(1 + 1) + S1 + S1 + S(1 + 1) + S(2) + S(510
140	32-31		520
			520
160			550
120	01-010NAD0101/9F(1//	EIVL EIVL	540
	121-21	EIVL	550

IF (IS2+IS1) 160,170,160	EIVL	560
N=N+1	EIVL	570
CONTINUE	EIVL	580
TRAP EIGENVALUES IN SMALLER AND SMALLER BOUNDS	EIVL	590
N=LP-N	EIVL	600
IF (N-K) 200,180,180	EIVL	610
DO 190 J=K,N	EIVL	620
X=(U)W	EIVL	630
N=N+1	EIVL	640
IF (LP-N) 30,210,210	EIVL	650
DO 220 J=N,LP	EIVL	660
IF (X-E(J)) 30,30,220	EIVL	670
E(J)=X	EIVL	680
GO TO 30	EIVL	690
CONTINUE	EIVL	700
RESTORE INPUT AND ORDER EIGENVALUES	EIVL	710
DO 240 I=1,LP	EIVL	720
A(I)=A(I)*BD	EIVL	730
B(I)=B(I)*BD	EIVL	740
DO 250 I=1,MVAL	EIVL	750
E(I)=(W(I)+E(I))*BD*•5	EIVL	760
RETURN	EIVL	770
END	EIVL	780
	<pre>IF (IS2+IS1) 160,170,160 N=N+1 CONTINUE TRAP EIGENVALUES IN SMALLER AND SMALLER BOUNDS N=LP=N IF (N=K) 200,180,180 DO 190 J=K,N W(J)=X N=N+1 IF (LP=N) 30,210,210 DO 220 J=N,LP IF (X=E(J)) 30,30,220 E(J)=X GO TO 30 CONTINUE RESTORE INPUT AND ORDER EIGENVALUES DO 240 I=1,LP A(I)=A(I)*BD B(I)=B(I)*BD B(I)=B(I)*BD DO 250 I=1,MVAL E(I)=(W(I)+E(I))*BD*.5 RETURN END</pre>	IF (IS2+IS1) 160,170,160 EIVL N=N+1 EIVL CONTINUE EIVL TRAP EIGENVALUES IN SMALLER AND SMALLER BOUNDS EIVL N=LP-N EIVL UF (N-K) 200,180,180 EIVL D0 190 J=K,N EIVL W(J)=X EIVL N=N+1 EIVL IF (X-E(J)) 30,210,210 EIVL D0 220 J=N,LP EIVL IF (X-E(J)) 30,30,220 EIVL GO TO 30 EIVL CONTINUE EIVL RESTORE INPUT AND ORDER EIGENVALUES EIVL D0 240 I=1,LP EIVL A(I)=A(I)*BD EIVL B(I)=B(I)*RD EIVL D(I)==((I)+RD EIVL E(I)==((I)+RD EIVL B(I)==((I)+RD EIVL RETURN EIVL RETURN EIVL END EIVL

	SUBROUTINE EIGVEC (NR,E,R,A,B,V,P,Q)	EIVC	10
С	GIVEN AN EIGENVALUE, FIND THE CCRRESPONDING EIGENVECTOR USING	EIVC	20
С	WILKINSON METHOD	EIVC	30
C.	NR = ORDER OF MATRIX	EIVC	40
C	R = 1-DIMENSION ARRAY OF NR/2*(NR+1) ELEMENTS CONTAINING	EIVC	50
С	THE W VECTOR	EIVC	60
С	E = EIGENVALUE	EIVC	70
С	A = 1-DIMENSION ARRAY OF NR ELEMENTS CONTAINING DIAGONAL	EIVC	80
Ç	TERMS OF TRIPLE-DIAGONAL MATRIX	EIVC	90
С	B = 1-DIMENSION ARRAY OF NR ELEMENTS CONTAINING OFF-DIAGONAL	EIVC	100
Ċ	TERMS OF TRIPLE-DIAGONAL MATRIX	EIVC	110
c	V = 1-DIMENSION ARRAY OF NR ELEMENTS CONTAINING THE	FIVC	120
č	FIGENVECTOR OF THE ORIGINAL SYMMETRIC MATRIX	EIVC	130
č	P AND G = 1-DIMENSIONAL ARRAYS OF NR ELEMENTS EACH USED AS	FIVC	140
c	WORK AREAS	FIVC	150
C	$DIMENSION = R(1) \cdot A(1) \cdot B(1) \cdot V(1) \cdot P(1) \cdot Q(1)$	EIVC	160
		FIVC	170
с	SET UP SIMULTANEOUS EQUATIONS LEE COMPUTE P. Q. R	FIVC	180
C		EIVC	190
		EIVC	200
		EIVC	210
	10 + 10 + 10 + 10 + 10 + 10 + 10 + 10 +	FIVC	220
10		EIVC	220
10			250
20		EIVC	240
30	P(1) = X	EIVC	250
	Q(1) = Y	EIVC	200
	V(1) = 0	EIVC	270
	X = A(1+1) - E - B(1+1) * Y X	EIVC	200
	IF (NRI-1) 40,70,40	EIVC	290
4)	Y = 3(1+2)	EIVC	300
	GO TO 70	EIVC	310
50	P(I)=B(I+1)	EIVC	320
	Q(I) = A(I+1) - E	EIVC	330
	Z=X/P(I)	EIVC	340
	X=Y-Z*Q(I)	EIVC	350
	IF (NR1-I) 60,70,60	EIVC	360
60	V(I) = B(I+2)	EIVC	370
	Y=-Z*V(I)	EIVC	380
70	CONTINUE	EIVC	390
С	SOLVE FOR EIGENVECTOR OF TRI-DIAGONAL MATRIX	EIVC	400
	IF (X) 90,80,90	EIVC	410
80	$V(NR) = 1 \cdot E10$	EIVC	420
	GO TO 100	EIVC	430
90	$V(NR) = 1 \cdot 0.07X$	EIVC	440
100	I = NR 1	FIVC	450
	$V(I) = (1 \cdot 00 - 0(I) * V(NR)) / P(I)$	FIVC	460
	Y=/(NR)**2+/(I)**2	FIVC	470
110		FIVC	480
110		FIVC	490
120		EIVC	500
T C (1	V - V - I / I / * * 0	EIVC	510
	$A = A + V (1)^{\alpha} Z$	EIVC	510
c		EIVC	520
			50
100		EIVC	540
	DO 140 I=1,NR	EIVC	550

	V(I) = V(I) / X	EIVC	560
140	CONTINUE	EIVC	570
С	COMPUTE EIGENVECTOR OF ORIGINAL MATRIX	EIVC	580
	I = NR	EIVC	590
150	I = I - 1	EIVC	600
	0 = L	EIVC	610
	κ = 1	EIVC	620
160	K=K+1	EIVC	630
	X+L=L	EIVC	640
	IF (K-I) 160,170,160	EIVC	650
170	Y=0.	EIVC	660
	L=AL	EIVC	670
	DO 180 K=I,NR	EIVC	680
	Y=Y+R(AL)V(K)	EIVC	690
	JA=JA+K	EIVC	700
180	CONTINUE	EIVC	710
	Y=2•00*Ý	EIVC	720
	L=AL	EIVC	730
	DO 190 K=I•NR	EIVC	740
	V(K) = V(K) - Y R(JA)	EIVC	750
	JA=JA+K	EIVC	760
190	CONTINUE	EIVC	770
	IF (I-2) 150,200,150	EIVC	780
200	RETURN	EIVC	790
	END	EIVC	800

.

349

	SUBROUTINE PRNT1 (EGVALU, EGVECT, AMASS, IRWKP, NROW, MORDER, IRWTOR, WO	RPRNT	10
	1KA)	PRNT	20
	DIMENSION WORKA(1)	PRNT	30
	DIMENSION EGVALU(1), EGVECT(NROW,5), AMASS(1), IRWKP(1), IRWTOR(1) PRNT	40
	COMMON COM(30)	PRNT	50
	EQUIVALENCE (COM(1), NJOINT)	PRNT	60
	EQUIVALENCE (COM(25) • INDWNM)	PRNT	70
	COMMON / CMAIN / COMAIN(1)	PRNT	80
		PRNT	90
	FOULVALENCE (COMAIN(101), RIX)	PRNT	100
	EQUIVALENCE (COMAIN(201), RJZ)	PRNT	110
	DIMENSION WAY (5), WAY (5), WAZ (5), $PX(5)$, $PY(5)$, $PZ(5)$, $GM(5)$	PRNT	120
	$A = R_{1}X(1) + R_{1}Y(1) + R_{1}Z(1)$	PRNT	130
		PRNT	140
		PRNT	150
	DATA ALPHA / 8HX-TRANS. 8HY-TRANS. 8HZ-TRANS.	PRNT	160
	1 AHX-ROTATA 8HY-ROTATA 8HZ-ROTATA /	PRNT	170
	DATA ITADE / 9 /	PRNT	180
	IF (IFRIST NE, 0) GO TO 4	PRNT	190
		PRNT	200
	WRITE $(6*280)$ (EGVALU(1)*1*1*6)	PRNT	210
c	SETUP ELGENVECTOR NO. SYMBOLS	PRNT	220
		PRNT	230
	2 I GVNO(1) = 1	PRNT	240
		PRNT	250
	MR = 6	PRNT	260
	GO TO 6	PRNT	270
	4 CONTINUE	PRNT	280
C	UPDATE EIGENVECTOR NO. SYMBOLS	PRNT	290
~		PRNT	300
	8 I G V N O (1) = I G V N O (1) + 5	PRNT	310
	6 CONTINUE	PRNT	320
		DRNT	330
		DRNT	340
		PRNT	350
10	$E_{0} = E_{0}	PRNT	360
10	WRITE $(6, 290)$ (IGVNO(I) I=1.5), (FGVALU(I) I=1.8 MR)	PRNT	370
	IF (NROW-FO-MORDER) GO TO 20	PRNT	380
	WRITE (6-300)	PRNT	390
	GO TO 40	PRNT	400
20	DO 30 I=1 NROW	PRNT	410
30		PRNT	420
50		PRNT	430
40	CONTINUE	PRNT	440
70		PRNT	450
		PRNT	460
c		PRNT	470
ĉ	ZERO OUT SUMMING FIELDS	PRNT	480
c		PRNT	490
C	DO 50 1=1.5	PRNT	500
		PRNT	510
		PRNT	520
		PRNT	530
		PRNT	540
	PY(I)=0.0	PRNT	550
			~~~

C

¢

50	PZ(I)=0.0 GM(I)=0.0	PRNT 560 PRNT 570 DBNT 580
C C	CALCULATE ABOVE VARIABLES	PRNT 590 PRNT 590 PRNT 600
C	N = (IRWKP(1)-1)/6+1	PRNT 610
	DO 140 I=1 MORDER	PRNT 620
	M=IRWKP(I)	PRNT 630
	K=M-(M-1)/6*6	PRNT 640
	M = (M-1)/6+1	PRNT 650
	IF (LCOUNT.LT.56) GO TO 60	PRNT 660
	WRITE (6,320)	PRNT 670
	IF (NROW • EQ • MORDER) GO TO 500	PRNT 675
	WRITE (6,300)	PRNT 680
	GO TO 501	PRNT 682
500	WRITE(6,400)	PRNT 684
501	CONTINUE	PRNI 686
	WRITE(6,310) (IGVNO(1),1=1,5)	PRNI 690
	LCOUNT=5	PRNI 700 DRNT 710
	N=M	PRNT 710
60		PRNT 730
60	$IE (N_{2}EO_{2}M) = GO_{2}TO_{2}TO_{2}$	PRNT 740
	WPITE (6.330)	PRNT 750
		PRNT 760
	N=M	PRNT 770
70	CONTINUE	PRNT 780
c		PRNT 790
Ċ	WRITE MODE SHAPES OF REDUCED SYSTEM	PRNT 800
Ċ		PRNT 810
	WRITE (6,340) M,ALPHA(K),(EGVECT(I,J),J=1,5)	PRNT 820
	LCOUNT=LCOUNT+1	PRNT 830
	IF (K.GT.3) GO TO 140	PRNT 840
	IF (K-2) 80,100,120	PRNT 850
C	X TRANSLATION MODE SHAPE ELEMENT	PRN1 860
80	DO 90 J=1,5	PRNT 870
	SAVE=EGVECT(1)J)*AMASS(1)	PRNI 600
	PX(J)=PX(J)+RJX(M)*SAVE	PRNT 000
		PRNT 910
00		PRNT 920
90	GO TO 140	PRNT 930
c	Y TRANSLATION MODE SHAPE FIEMENT	PRNT 940
100	DO 110 J=1.5	PRNT 950
100	SAVE=EGVECT(I,J)*AMASS(I)	PRNT 960
	PY(J) = PY(J) + RJY(M) * SAVE	PRNT 970
	SAVE=SAVE*EGVECT(I,J)	PRNT 980
	WNX(J) #WNX(J) + SAVE	PRNT 990
110	WNZ(J)=WNZ(J)+SAVE	PRNT1000
	GO TO 140	PRNT1010
С	Z TRANSLATION MODE SHAPE ELEMENT	PRNT1020
120	DO 130 J=1,5	PRNT1030
	SAVE=EGVECT(I,J)*AMASS(I)	RRNT1040
	PZ(J)=PZ(J)+RJZ(M)*SAVE	PRNT1050
	SAVE=SAVE*EGVECT(I,J)	PKN11050
	WNX(J)=WNX(J)+SAVE	PKNI1070

,

D0 150 K=1+MORDER PRNT1100 GM (1) = GM(1) + AMASS(K)*EGVECT(K,I)**2 PRNT1120 GM (1) = GM(1) + AMASS(K)*EGVECT(K,I)**2 PRNT1130 WRITE (6,390) WRITE (6,390) (IGVNO(1),I=1,5), (WNY(1),I=1,5),(WMZ(1),I=1,5),(PX(1),PRNT160 II=15),(PY(1),I=1,5),(PZ(1),I=1,5), (GM (I),I=1,5) PRNT170 RETURN C PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 C PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 G PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 G PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 C PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 G PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 G PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 G PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 G D G I = I MORDER AND.(INDWNM.EQ.0)) RETURN PRNT1220 G G G I = I MORDER AND.(INDWNM.EQ.0)) RETURN PRNT1250 SAVE1=MORDER PRNT2 G CALL P1A721 (121,WORKA,MORDER,ITAPE) PRNT1260 G CALL P1A721 (121,WORKA,MORDER,ITAPE) PRNT1300 LC=0 G CALL P1A721 (121,WORKA,MORDER,ITAPE) PRNT130 I F (INROW.EC.MORDER) GO TO 180 PRNT130 MRITE (6,400) WRITE (6,400) (IGVNO(1),I=1,5) (EGVALU(1),I=LR,MR) PRNT130 WRITE (6,400) WRITE (6,400) PRNT130 WRITE (6,400) (IGVNO(1),I=1,5) (EGVALU(1),I=LR,MR) PRNT130 MRITE (6,400) WRITE (6,400) PRNT130 MRITE (6,400) PRNT130 MRITE (6,400) FRNT130 MRITE (6,400) FRNT130 MRITE (6,400) FRNT130 MRITE (6,400) (IGVNO(1),I=1,5) (EGVALU(1),I=LR,MR) PRNT130 MRITE (6,400) FRNT130 MRITE (6,400) PRNT130 MRITE (6,400) FRNT130 MRITE (6,400) (IGVNO(1),I=1,5) (EGVALU(1),I=LR,MR) PRNT130 MRITE (6,400) FRNT130 MRITE (6,400) FRNT130 MRITE (6,400) (IGVNO(1),I=1,5) (FRNT140 MRITE (6,400) (IGVNO(1),I=1,5) (FRNT140 MRITE (6,100,INT PRNT130 MRITE (6,100,INT PRNT130 MRITE (6,100,INT PRNT130 MRITE (6,100,INT PRNT130 MRITE (FNNT140 MRITE (FNNT140 M	130 140	WNY(J)=WNY(J)+SAVE CONTINUE	PRNT1080 PRNT1090
D0 150 K=1,M0RDER pRNT1120 GM(1) = GM(1) + AMASS(K)*EGVECT(K,I)**2 PRNT1120 PRNT1140 WRITE (6,350) (IGVN0(I),I=1,5) WRITE (6,360) (IGVN0(I),I=1,5), (WN((I),I=1,5),(WNZ(I),I=1,5),(PX(I),PRNT1160 II=1,5),(PY(I),I=1,5),(PZ(I),I=1,5), (GM(I),I=1,5),(PX(I),PRNT1160 PRNT120 PRNT120 C PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 C PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT120 FNT120 FIT(NDWM.EC.0.MORDER].AND.(INDNM.EC0.0)) RETURN PRNT120 FNT120 FNT120 FNT120 FNT120 FNT120 FNT120 DO 160 I=.MORDER WRITE (ITAPE,380) SAVE.SAVE1 PRNT1260 SAVE=NJOINT SAVE.SAVE1 PRNT1260 C CONTINUE PRNT22 IF (INDNM.EC.MORDER, ITAPE) PRNT1260 WRITE (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I,I=1,NJOINT) PRNT1300 C CONTINUE PRNT126 FNT1300 FR WRITE (G+300) (IGVN0(I),I=1,5), (EGVALU(I),I=LR,MR) PRNT130 IF (NOW-ROED GO TO 180 PRNT130 WRITE (6+300) WRITE (G=10, IGVN0(I),I=1,5) (FGVALU(I),I=LR,MR) PRNT130 WRITE (6+300) WRITE(6,0) (IGVN0(I),I=1,5) (FGVALU(I),I=LR,MR) PRNT130 WRITE (6+300) WRITE(6+300) PRNT130 WRITE (6+300) FRNT130 WRITE (6+300) FRNT130 D 250 J=1,3 MORDER PRNT140 WRITE(6+300) (IGVN0(I),I=1,5) (FGVALU(I),I=LR,MR) PRNT130 WRITE(6+300) WRITE(6,0) (IGVN0(I),I=1,5) (FGVALU(I),I=LR,MR) PRNT130 WRITE(6+300) FRNT130 WRITE(6+300) (IGVN0(I),I=1,5) (FGVALU(I),I=LR,MR) PRNT130 WRITE(6+300) FRNT140 WRITE(6+300) (IGVN0(I),I=1,5) (FGVALU(I),I=LR,MR) PRNT130 D 250 J=1,3 MORDER PRNT140 WRITE(6+300) FRNT140 WRITE(6+300) FRNT140 WRIT		DO 150 I=1,5	PRNT1100
GM(1) = GM(1) + AMASS(K)*EGVECT(K,I)**2         PRNT1120           150         CONTINUE         PRNT1130           WRITE (6,350)         MRNT(1),1=1,5),(WNY(1),1=1,5),(WNZ(1),1=1,5),(PX(1),PRNT1150           WRITE (6,350)         MRNT(1),1=1,5),(WNY(1),1=1,5),(WNZ(1),1=1,5),(PX(1),PRNT1160           NRTE (6,360)         MRNT1200           II=-1,5),(PY(1),I=1,5),(PZ(1),I=1,5),(MNY(1),1=1,5),(WNZ(1),I=1,5),(PX(1),PRNT120           PRNT120         PRNT1200           PRNT120         PRNT1200           C         PRNT21           FIRTP PRT2         PRNT1200           FIRT (NROW-EQ.40) GO TO 170         PRNT1200           SAVE=NJOINT         PRNT1250           SAVE=NJOINT         PRNT1270           D LOGO I=1,MORDER         PRNT1270           PRNT1270         PRNT1270           D LOGO I=1,MORDER         PRNT1270           VRITE (ITAPE,390) SAVE,SAVE1         PRNT1270           D CAL PIA721 (121,WORKA,MORDER,ITAPE)         PRNT1320           VRITE (CALL PIA72, (121,WORKA,MORDER,ITAPE)         PRNT1320           VRITE (6+300) (GVNO(1),I=1,5), (EGVALU(1),I=I=R,MR)         PRNT1320           VRITE (6+300) (GVNO(1),I=1,5), (EGVALU(1),I=LR,MR)         PRNT1320           VRITE (6+300) (GVNO(1),I=1,5)         PRNT140           WRITE (6+300) (GVNO(1		DO 150 K=1,MORDER	PRNT1110
150 CONTINUE         PRNT1140           WRITE (6,350)         PRNT1140           WRITE (6,360) (IGVN0(I),I=1,5),(WNY(I),I=1,5),(WNZ(I),I=1,5),(PX(I),PRNT1160         PRNT1170           PRETURN         PRNT1200           C         PRINT MODE SHAPES FOR COMPLETE SYSTEM         PRNT1200           C         PRINT RETURN         PRNT1200           C         PRINT RODE SHAPES FOR COMPLETE SYSTEM         PRNT1200           C         PRINT RODE SHAPES FOR COMPLETE SYSTEM         PRNT12200           F (INROW-EG-MORDER).AND.(INDWNM-EG.0)) RETURN         PRNT1220           IF (INROW-EG-MORDER).AND.(INDWNM-EG.0)) RETURN         PRNT1240           SAVE1=MORDER         PRNT1260           WRITE (ITAPE,330) SAVE, SAVE1         PRNT1220           CALL PIA721 (121wORKA,MORDER, ITAPE)         PRNT1300           CALL PIA721 (121wORKA,MORDER, ITAPE)         PRNT1300           CALL PIA721 (121wORKA,MORDER, ITAPE)         PRNT1300           CONTINUE         IF (NROW-EG.MORDER) GO TO 180         PRNT1300           WRITE (6,400)         WRITE (6,400)         PRNT1300           WRITE (6,400) (IGVN0(I),I=I=15), (EGVALU(I),I=LR,MR)         PRNT1300           WRITE (6,400)         PRNT1400         PRNT1300           WRITE (6,400) (IGVN0(I),I=1+5), (EGVALU(I),I=LR,MR)         PRNT1400     <		GM(I) = GM(I) + AMASS(K)*EGVECT(K,I)**2	PRNT1120
WRITE (6,350)         PRN11150           WRITE (6,350) (IGVN0(I),I=1,5),(WNY(I),I=1,5),(WNZ(I),I=1,5),(PX(I),PRN11160           II=1,5),(PY(I),I=1,5),(PZ(I),I=1,5),(WNY(I),I=1,5),(WNZ(I),I=1,5),(PX(I),PRN11160           RETURN         PRN11100           RETURN         PRN11100           C         PRINT MODE SHAPES FOR COMPLETE SYSTEM         PRN11200           C         PRINT PRN12         PRN11200           IF (INROW-E0_MORDER).AND.(INDWNM.EQ.0)) RETURN         PRN11200           FF (INROW-E0_MORDER).AND.(INDWNM.EQ.0)) RETURN         PRN11200           SAVEI-MORDER         PRN11200           WRITE (ITAPE,380) SAVE,SAVE1         PRN11270           DO LOG I=1,HORDER         PRN11200           WARITE (ITAPE,390) (RJY(I),RJZ(I),RJZ(I),RJZ(I),I,I=1,NJOINT)         PRN11300           WRITE (FACM.MORDER) GO TO 180         PRN11300           WRITE (6+400)         PRN11300           WRITE (6+400)         PRN11300           WRITE (6+400)         PRN11300           WRITE (6+400)         PRN11300           VRITE (6+400)         PRN11300           WRITE (6+400)         PRN11300           WRITE (6+400)         PRN11300           VRITE (6+400)         PRN11300           VRITE (6+400)         PRN11300           VR	150	CONTINUE	PRNT1130
<pre>WRITE(6,360) (IGVNO(1),I=1,5) (WNY(1),I=1,5),(WNZ(1),I=1,5),(PX(I),PRNT1160 II=1,5),(PY(I),I=1,5),(PZ(I),I=1,5),(GM(I),I=1,5),(PX(I),PRNT1160 PRNT1200 PRNT1200 PRNT1200 C PRINT MODE SHAPES FOR COMPLETE SYSTEM PRNT1220 IF (INROW.EG.MORDER).AND.(INDWNM.EG.O)) RETURN PRNT1220 IF (INROW.EG.MORDER).AND.(INDWNM.EG.O)) RETURN PRNT1220 MRITE (ITAPE,380) SAVE,SAVE1 PRNT1250 SAVE1=MORDER PRNT1250 O 160 I=1.MORDER PRNT1250 WRITE (ITAPE,380) SAVE,SAVE1 PRNT1260 WRITE (ITAPE,380) SAVE,SAVE1 PRNT1270 D 0 160 I=1.MORDER PRNT1260 WRITE (ITAPE,390) (AUY(I),RJX(I),I,I=I=1,NJOINT) PRNT1310 LC=0 PRNT1320 FRNT1320 PRNT1320 WRITE (ITAPE,390) (IGVNO(I),I=1,5), (EGVALU(I),I=LR.MR) PRNT1350 WRITE (6,400) WRITE (6,400) PRNT1350 WRITE (6,400) PRNT1350 WRITE (6,400) PRNT1350 WRITE (6,400) PRNT1350 WRITE (6,400) PRNT1350 WRITE (6,400) PRNT1360 WRITE (6,400) PRNT1360 WRITE (6,400) PRNT1360 PRNT1360 D 0 ZOO I=1,NJOINT PRNT130 PRNT1400 N=NROW-MORDER PRNT4 D 0 ZOO I=1,NJOINT PRNT140 N=0 MENROW-MARDER PRNT4 D 0 ZOO I=1,NJOINT PRNT140 N=0 MENROW-MARDER PRNT4 D 0 ZOO I=1,NJOINT PRNT140 N=0 MENROW-MARDER PRNT4 D 0 IF (N.NE.IRWOP(K)) GO TO 200 PRNT140 N=NR1420 D 0 ZOO I=1,NJOINT PRNT140 PRNT140 N=NROW-MARDER PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRNT140 PRN</pre>		WRITE (6,350)	PRNT1140
WRITE (6,370) (WNX(1),I=1,5),(WNZ(1),I=1,5),(PX(1),PRNT1160           II=1,5),(PY(I),I=1,5),(PZ(I),I=1,5),(GM(I),I=1,5)           PRNT1170           RETURN           C           PRINT MODE SHAPES FOR COMPLETE SYSTEM           PRNT120           ENTRY PRNT2           IF (INDWM.EQ.MORDER).AND.(INDWNM.EQ.0)) RETURN           PRNT1220           IF (INDWM.EQ.MORDER).AND.(INDWNM.EQ.0)) RETURN           PRNT1220           OF (INDW.EQ.MORDER).AND.(INDWNM.EQ.0)) RETURN           PRNT1220           PRNT1220           DF (INDWM.EQ.0) GO TO 170           SAVE=NJOINT           SAVE=NJOINT           SAVE=NJOINT           SAVE=NJOINT           SAVE=NJOINT           CALL PIA721 (121).WORKA,MORDER, ITAPE)           WRITE (ITAPE,380) SAVE,SAVE1           DO 160 I=1.MORDER           WRITE (IARDE,390) (RJY(I),RJZ(I),RJX(I),I,I=1=NJOINT)           PRNT1300           WRITE (G.20, OROPER) GO TO 180           WRITE (G.4200) (IGVNO(I),I=1=1,5), (EGVALU(I),I=LR,MR)           WRITE(G.4200) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR)           VRNT1300           PRNT1300           PRNT1300           PRNT1300           PRNT1300           PRNT1420     <		WRITE(6,360) (IGVNO(I),I=1,5)	PRNT1150
11=1+5)*(PY(1)*,I=1+5)*(CA(1)*,I=1+5)*         PRN11100           RETURN         PRN11100           C         PRINT MODE         SHAPES FOR COMPLETE SYSTEM         PRN11200           C         PRINT MODE         SHAPES FOR COMPLETE SYSTEM         PRN11200           C         PRINT PRNT2         PRN11200         PRN11230           IF         (INDWN*E0*00 GO TO 170         PRN11250         PRN11250           SAVE1=MORDER         PRN11260         PRN11260           WRITE         (ITAPE \$380) SAVE, SAVE1         PRN11260           DO 160         I=1*,MORDER         PRN11260           MWRITE         (ITAPE \$390) (RAY(1)*,RJX(1)*,II*,I=1*,NJOINT)         PRN11300           LCOUTINUE         (ITAPE \$390) (RAY(1)*,RJX(1)*,II*,I=1*,NJOINT)         PRN11300           LCOUNT=11         IF         (INWOKE0*,GO TO 180         PRN11300           WRITE (6*,400)         WRITE (6*,400)         PRN11300         PRN11300           WRITE (6*,400)         (IGVNO(I)*,I=1*,5)         (EGVALU(I)*,I=LR*,MR)         PRN11300           LCOUNT=11         PRN1240         PRN11400         PRN11400           N=0         GO TO 180         PRN11400         PRN11400           N=0         IF<(IN*,KLRWP(K)) GO TO 200		WRITE (6,370) (WNX(I), I=1,5), (WNY(I), I=1,5), (WNZ(I), I=1,5), (PX(I)	PRNT1160
RETURN       PRN11100         C       PRIN1100         C       PRIN1200         C       PRIN1200         C       PRIN1200         ENTRY PRN12       PRN11200         IF (INDUMN.EQ.ONORDER).AND.(INDWNM.EQ.O)) RETURN       PRN11230         IF (INDUMN.EQ.O) GO TO 170       PRN11230         SAVE=NJOINT       PRN11260         SAVE=NJOINT       PRN11260         VEXT=NJORDER       PRN11280         WRITE (ITAPE.380) SAVE,SAVE1       PRN11280         DO 160 I=1.MORDER       PRN11280         WRITE (ITAPE.390) (RJY(I),RJZ(I),RJX(I),I ,I=1.NJOINT)       PRN1320         LC=0       PRN11300         VEXTE(6.4200) (IGVNO(I),I=1.5), (EGVALU(I),I=R,MR)       PRN1320         VEXTE(6.430) (IGVNO(I),I=1.5), (EGVALU(I),I=LR,MR)       PRN1300         WRITE(6.430) (IGVNO(I),I=1.5)       PRN1400         LCOUNT=11       PRN1400         N=0       CONTINUE       PRN1400         N=1       PRN1400       PRN1400         N=1       PRN1400       PRN1400         N=1       PRN1400       PRN1400         N=1       PRN1400       PRN1400         N=0       IF (N.NE.IRWEP(K)) GO TO 200       PRN1400         N=4		1I=1,5),(PY(I),I=1,5),(PZ(I),I=1,5), (GM( I ),I=1,5 )	PRNT1170
C         PRINI MODE SHAPES FOR COMPLETE SYSTEM         PRN11200           C         PRINT MODE SHAPES FOR COMPLETE SYSTEM         PRN1220           IF (INROW.EG.MORDER).AND.(INDWNM.EG.0)) RETURN         PRN1220           IF (INROW.EG.MORDER).AND.(INDWNM.EG.0)) RETURN         PRN1220           IF (INROW.EG.MORDER).AND.(INDWNM.EG.0)) RETURN         PRN1220           SAVE1=MORDER         PRN1270           WRITE (ITAPE,380) SAVE.SAVE1         PRN1270           D0 160 1=1.MORDER         PRN1270           MCAL P14721 (121.WORKA.MORDER, ITAPE)         PRN1300           VKALL P14721 (121.WORKA.MORDER, ITAPE)         PRN13100           CALL P14721 (121.WORKA.MORDER, ITAPE)         PRN1320           MCAONTINUE         PRN1320           IF (NROW.EG.MORDER) GO TO 180         PRN1330           WRITE (64.900)         VRIT1600           WRITE (64.910) (IGVNO(1).J=1.5), (EGVALU(1).J=LR.MR)         PRN1380           MCONTINUE         PRN1390           MCONTINUE         PRN1390           MCONTINUE         PRN1400           MCONTINUE         PRN1400           MCONTINUE         PRN1400           N=N+1         PRN1400           N=N+1         PRN1400           N=N+1         PRN1430           D0 250 J=1		RETURN	PRNT1180
C PRINT MODE SHAPES FOR COMPLETE SYSTEM PRN11210 PRN1210 ENTRY PRN12 IF (INDNUM.EQ.MORDER).AND.(INDWNM.EQ.0)) RETURN PRN11240 SAVE=NJOINT PRN11240 SAVE=NJOINT PRN11240 SAVE=NJOINT PRN11240 SAVE=NJOINT PRN11240 SAVE=NJOINT PRN11240 WRITE (ITAPE,380) SAVE,SAVE1 PRN11260 WRITE (ITAPE,380) SAVE,SAVE1 PRN11260 WRITE (ITAPE,390) (AVE,SAVE1 PRN11280 CALL PLA721 (121.WGRA,MORDER,ITAPE) PRN11310 UC CONTINUE IF (NROW.EQ.MORDER) GO TO 180 PRN11320 IF (NROW.EQ.MORDER) GO TO 180 PRN11350 WRITE (6.4200) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR) PRN11310 IF (NROW.EQ.MORDER) GO TO 180 PRN11350 WRITE (6.4200) (IGVNO(I),I=1,5) PRN11340 WRITE (6.4200) (IGVNO(I),I=1,5) PRN11340 VRITE (6.4200) (IGVNO(I),I=1,5) PRN11340 VRITE (6.4200) (IGVNO(I),I=1,5) PRN11410 PRN11390 M=NROW-MGRDER PRN11400 PRN1130 PRN11400 PRN11400 N=0 0 260 I=1,NJOINT PRN1140 N=N+1 N=0 0 IF (N.NE.IRWKP(K)) GO TO 200 PRN11400 NI=K K=41 PRN11400 PRN11400 NI=K K=41 PRN11400 PRN11400 NI=K K=41 PRN11400 PRN1150 PRN1150 NI=K(.NE.IRWTOR(M)) GO TO 210 PRN1150 PRN1150 NI=K(.NE.IRWTOR(M)) GO TO 210 PRN1150 PRN1150 NI=K(.NE.IRWTOR(M)) M=M=1 PRN1150 IF (N.GT.IRWKP(K)) K=K+1 IF (N.GT.IRWKP(K)) K=K+1 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN1150 PRN	C		PRNT1190
C PRNT1210 ENTRY PRNT2 PRNT2 IF ((NROW-EG.MORDER).AND.(INDWNM.EG.0)) RETURN PRNT1220 IF ((NROW-EG.MORDER).AND.(INDWNM.EG.0)) RETURN PRNT1220 SAVE1=MORDER PRNT1250 SAVE1=MORDER PRNT1250 O 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 160 I=1.MORDER PRNT1270 D 260 I=1.MORDER PRNT1370 IF (NROW-EG.MORDER) GO TO 180 WR ITE (6.910) (IGVNO(I).FI=1,5), (EGVALU(I).FI=LR.MR) PRNT1350 WR ITE (6.910) (IGVNO(I).FI=1,5) (EGVALU(I).FI=LR.MR) PRNT1350 MR ITE (6.910) (IGVNO(I).FI=1,5) PRNT1370 D 260 I=1.MJOINT PRNT1370 D 260 I=1.MJOINT PRNT1420 N=0 N=0 N=N+1 PRNT1450 N=1 F (N.NE.IRWKP(K)) GO TO 200 PRNT1450 NFNT1450 NFNT1450 D 1F (N.NE.IRWKP(K)) GO TO 210 PRNT1450 NFNT1450 D F (N.ST.IRWTOR(M)) GO TO 210 PRNT1510 N=NT1510 N=NT1550 IF (N.GT.IRWTOR(M)) M=M=1 G 0 TO 220 200 CONTINUE PRNT1550 IF (N.GT.IRWTOR(M)) M=M=1 G 0 TO 220 CONTINUE PRNT1550 IF (INDWMM.EG.0) GO TO 230 IF (INDWMM.EG.0) GO TO 230 IF (INDW.EG.MORDER) GO TO 250 IF (INDWMM.EG.0) GO TO 250 IF (INDW.EG.MORDER) GO TO 250 IF (INDWMM.EG.0) GO TO 250 IF (INDWMM.EG.0) GO TO 250 IF (INDWM.EG.0) GO TO 250 IF (INDWM.EG.0) GO TO 250 IF (INDW.EG.MORDER) GO TO 250 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1	С	PRINT MODE SHAPES FOR COMPLETE SYSTEM	PRNT1200
ENTRY PRN12         PRN1220           IF (INROW.EQ.MORDER).AND.(INDWNM.EQ.0)) RETURN         PRN1220           IF (INDWNM.EQ.0) GO TO 170         PRN12260           SAVE=NJOINT         PRN12260           SAVE1=MORDER         PRN12260           WRITE (ITAPE,380) SAVE.SAVE1         PRN12270           DO 160 [=1,MORDER         PRN12280           (AL P1A721 (121:WORKA,MORDER,ITAPE)         PRN13200           WRITE (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I,I=1,NJOINT)         PRN1320           LC=0         PRN1320           170         CONTINUE         PRN1320           170         CONTINUE         PRN1320           WRITE (6,200) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR)         PRN1330           WRITE (6,300) (IGVNO(I),I=1,5)         PRN1320           WRITE (6,300) (IGVNO(I),I=1,5)         PRN1320           M=NROW-MORDER         PRN1440           N=0         PRN1440           DO 260 I=1,NJOINT         PRN1440           N=N+1         PRN1440           N=N+1         PRN1440           N=N+1         PRN1440           N=1         PRN1440           N=0         PRN1440           DO 260 I=1,NJOINT         PRN1440           N=0         PRN1440	С		PRNT1210
<pre>IF ((NROW+EQ_MORDER).AND.(INDWNM.EQ.0)) RETURN PRN1240 IF (INDWNM.EQ.0) GO TO 170 PRN1240 SAVE=NJOINT PRN12260 WRITE (ITAPE,380) SAVE.SAVE1 PRN12270 DO 160 I=1,MORDER PRN12280 WRITE (ITAPE,390) (AUY(I),RJZ(I),I,I,I=1,NJOINT) PRN1280 WRITE (ITAPE,390) (RJY(I),RJZ(I),RJZ(I),I,I,I=1,NJOINT) PRN13100 WRITE (ITAPE,390) (RJY(I),RJZ(I),RJZ(I),I,I,I=1,NJOINT) PRN1330 IF (NROW-EQ.MORDER) GO TO 180 PRN1340 WRITE (6,290) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR) PRN1350 WRITE (6,310) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR) PRN1380 WRITE (6,400) PRN1340 WRITE (6,401) PRN1380 MANNOW-MORDER PRN1380 DO 260 I=1,NJOINT PRN1440 N=0 PRN1440 N=1 N=0 U2 50 J=1,3 PRN1440 NI=K (PRN1460 NI=K (PRN1460) PRN1450 MI=KCH PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=K (PRN1460) PRN1460 NI=RN0W-M+1 PRN1530 CONTINUE PRN1530 CONTINUE (ITAPE,410) (EGVECT(NI,L),L=1,5),LC PRN1530 NR1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1460 PRN1</pre>		ENTRY PRNT2	PRNT1220
<pre>IF (INDWNM.ECo.0) GO TO 170 SAVE=NJOINT FPRN1250 SAVE1=MORDER WRITE (ITAPE,380) SAVE,SAVE1 DI 100 [1=1,MORDER WRITE (ITAPE,390) SAVE,SAVE1 CALL PIA721 (121.WORKA,MORDER,ITAPE) WRITE (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I,I=1,NJOINT) LC=0 RNT1320 TF (NROW-EG.MORDER) GO TO 180 FRNT1330 IF (NROW-EG.MORDER) GO TO 180 WRITE (6,300) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR) WRITE (6,300) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR) WRITE (6,310) (IGVNO(I),I=1,5) COUNT=11 PRNT1380 MENROW-MORDER K=1 N=0 D 260 I=1,NJOINT DD 260 I=1,NJOINT N=K N=1 PRNT1420 PRNT1440 N=N+1 PRNT1450 NI=K K+1 GO TO 220 IF (N.NE.IRWKP(K)) GO TO 210 NI=K K=K+1 GO TO 220 CONTINUE IF (N.ST.IRWKP(K)) K=K+1 IF (N.ST.IRWKP(K)) K=K+1 IF (N.ST.IRWKP(K)) K=K+1 IF (N.ST.IRWTOR(M)) M=M-1 GO CONTINUE PRNT1520 CONTINUE IF (N.ST.IRWTOR(M)) M=M-1 FRNT1520 PRNT1520 P</pre>		IF ((NROW.EQ.MORDER).AND.(INDWNM.EQ.0)) RETURN	PRNT1230
SAVE=NJOINT         PRN1250           SAVEI=MORDER         PRN1260           WRITE (ITAPE,380) SAVE,SAVE1         PRN1280           Do 160 I=1.MORDER         PRN1280           PRN1280         PRN1280           CALL PIA721 (121.WORKA,MORDER,ITAPE)         PRN1320           WRITE (ITAPE,390) (RJY(I),RJX(I),I,I,I=1,NJOINT)         PRN1320           LC=0         PRN1320           IC=0         PRN1320           WRITE (6.20) (IGVN0(1),I=1,5)         (EGVALU(I),I=LR,MR)           WRITE (6.400)         PRN1360           WRITE (6.400)         PRN1370           LCOUNT=11         PRN1370           MRNUMEG.SOLOUNTER         PRN1380           MRNTHE (6.400)         PRN1370           WRITE (6.400)         PRN1380           WRITE (6.400)         PRN1370           MRNTHE (6.400)         PRN1380           MRNTHE (6.400)         PRN1370           MERCOUNT=11         PRN1380           MRNTHE (6.400)         PRN1380           WRITE (6.400)         PRN1380           WRITE (6.400)         PRN14140           MERCOUNTAL         PRN1380           PRNT1380         PRN14140           N=0         PRN114140           N=0 <td></td> <td>IF (INDWNM.EQ.0) GO TO 170</td> <td>PRNT1240</td>		IF (INDWNM.EQ.0) GO TO 170	PRNT1240
SAVE1=MORDER         PRNT1260           WRITE (ITAPE,380) SAVE,SAVE1         PRNT1270           D0 160 I=1.MORDER         PRNT1280           160 WORKA(I)=IRWKP(I)         PRNT1270           CALL PIA721 (121:WORKA,MORDER,ITAPE)         PRNT1300           WRITE (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I,I=1,NJOINT)         PRNT1310           LC=0         PRNT1320           170 CONTINUE         PRNT1320           WRITE (6,290) (IGVNO(1),I=1,5), (EGVALU(I),I=LR,MR)         PRNT1350           WRITE (6,400)         PRNT1360           M=NROW-MORDER         PRNT1360           K=1         PRNT1400           N=0         PRNT1400           D0 260 J=1.NJOINT         PRNT1450           D0 250 J=1.3         PRNT1450           N=K+1         PRNT1450           M=M-1         PRNT1450           M=M-1         PRNT1450           M=M-1         PRNT1500           M=M-1         PRNT1500           M=M-1         PRNT1500           M=M-1         PRNT1500           M=M		SAVE=NJOINT	PRNT1250
WRITE (ITAPE,380) SAVE,SAVE1         PRNT1270           D0 160 I=1.MORDER         PRNT1280           160         WORKA(I)=IRWKP(I)         PRNT1290           CALL PIA721 (121.MORKA,MORDER,ITAPE)         PRNT1310           WRITE (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I,I=1,NJOINT)         PRNT1310           LC=0         PRNT1320           170         CONTINUE         PRNT1320           IF (NROW-EG.MORDER) GO TO 180         PRNT1350           WRITE (6,200) (IGVN0(I),I=1,5), (EGVALU(I),I=LR,MR)         PRNT1350           WRITE (6,310) (IGVN0(I),I=1,5)         PRNT1360           WRITE (6,310) (IGVN0(I),I=1,5)         PRNT1370           LCOUNT=11         PRNT1380           M=NROW-MORDER         PRNT1400           K=1         PRNT1420           N=0         PRNT1420           DO 260 I=1,NJOINT         PRNT1420           DO 260 I=1,NJOINT         PRNT1420           DO 250 J=1,3         PRNT1420           N=K+1         PRNT1420           M=NROW-MARDER         PRNT1420           N=K         PRNT1420           DO 250 J=1,3         PRNT1420           N=K         PRNT1450           N=K         PRNT1450           M=M=1         PRNT1450 <t< td=""><td></td><td>SAVE1=MORDER</td><td>PRNT1260</td></t<>		SAVE1=MORDER	PRNT1260
DO         160         WORKA(I)=IRWKP(I)         PRNT1280           160         WORKA(I)=IRWKP(I)         PRNT1290         CALL P1A721 (121+WORKA,MORDER,ITAPE)         PRNT1300           WRITE (ITAPE,390) (RJY(I),RJZ(I),RJZ(I),I,I=I,NJOINT)         PRNT1310         PRNT1310           170         CONTINUE         PRNT1320           171         (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I,I=I,NJOINT)         PRNT1320           171         CONTINUE         PRNT1320           171         (IGVNO(I),I=1,5)         (EGVALU(I),I=LR,MR)         PRNT1350           WRITE (6,200) (IGVNO(I),I=1,5)         (EGVALU(I),I=LR,MR)         PRNT1350           WRITE (6,400)         WRIT1360         PRNT1370           WRITE (6,400)         WRIT1370         PRNT1350           WRITE (6,400)         WRIT131         PRNT1350           WRITE (6,400)         WRIT1370         PRNT1350           WRITE (6,400)         WRIT1370         PRNT1480           D0         PRNT14110         PRNT1420		WRITE (ITAPE,380) SAVE,SAVE1	PRNT1270
160       WORKA(I)=IRWKP(I)       PRNT1290         CALL P1A721 (121sWORKA,MORDER,ITAPE)       PRNT1300         WRITE (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I,I=I,NJOINT)       PRNT1310         LC=0       PRNT1320         170       CONTINUE       PRNT1310         WRITE (6,200) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR)       PRNT1350         WRITE(6,310) (IGVNO(I),I=1,5)       (EGVALU(I),I=LR,MR)         WRITE(6,310) (IGVNO(I),I=1,5)       PRNT1360         CONTINUE       PRNT1360         WRITE(6,310) (IGVNO(I),I=1,5)       PRNT1370         CONTINUE       PRNT1370         M=NROW-MORDER       PRNT1370         K=1       PRNT1420         N=0       PRNT1420         D0 260 I=1,NJOINT       PRNT1420         D0 250 J=1,3       PRNT1440         N=N+1       PRNT1450         I90 IF (N*NE*IRWKP(K)) GO TO 200       PRNT1450         N1=K       PRNT1450         WRNT1450       PRNT1450         M=M=1       PRNT1450         GO TO 220       PRNT1500         O TO 220       PRNT1500         O TO 190       PRNT150         IF (N*GT.IRWTOR(M)) M=M=1       PRNT150         M=M=1       PRNT150         GO TO		DO 160 I=1,MORDER	PRNT1280
CALL P1A721 (121:w0RKA,MORDER,ITAPE)       PRNT1300         WR ITE (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I,I=1,NJOINT)       PRNT1310         LC=0       PRNT1320         170 CONTINUE       PRNT1320         IF (NROW-E0-MORDER) GO TO 180       PRNT1350         WR ITE (6+290) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR)       PRNT1350         WR ITE (6+300) (IGVNO(I),I=1,5)       PRNT1370         LCOUNT=11       PRNT1370         180 CONTINUE       PRNT1400         M=NROW-MORDER       PRNT1400         K=1       PRNT1400         N=0       PRNT1420         D0 260 I=1,NJOINT       PRNT1430         D0 260 I=1,NJOINT       PRNT1430         N=N+1       PRNT1430         Y       PRNT1450         N=1+1       PRNT1450         Y       PRNT1450         N=1       PRNT150         N=1       PRNT150         PRNT150       PRNT150<	160	WORKA(I)=IRWKP(I)	PRNT1290
<pre>WRITE (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I,I=1,NJOINT) PRNT1310 LC=0 PRNT1320 PRNT1330 IF (NROW-EQ.MORDER) GO TO 180 PRNT1330 WRITE (6,400) WRITE (6,400) PRNT1350 WRITE (6,400) WRITE (6,400) PRNT1350 COUNT=11 PRNT1350 MENROW-MORDER PRNT1400 K=1 PRNT1410 N=0 DO 260 I=1,NJOINT PRNT1420 DO 260 I=1,NJOINT PRNT1420 DO 260 J=1,3 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1450 PRNT1510 N1=K PRWTOR(M)) GO TO 210 PRNT1450 PRNT1450 PRNT150 N1=RCW-M+1 PRNT150 PRNT150 PRNT150 PRNT150 PRNT150 PRNT150 PRNT150 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1550 PRNT1</pre>		CALL P1A721 (121,WORKA,MORDER,ITAPE)	PRNT1300
LC=0 PRNT1320 PRNT1320 PRNT1330 IF (NROW-EQ.MORDER) GO TO 180 PRNT1340 WRITE(6,290) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR) PRNT1350 WRITE(6,310) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR) PRNT1350 WRITE(6,310) (IGVNO(I),I=1,5) PRNT1350 COUNT=11 PRNT1450 LCOUNT=11 PRNT1450 M=NROW-MORDER PRNT1400 K=1 PRNT1400 K=1 PRNT1420 DO 250 J=1,3 PRNT1420 DO 250 J=1,3 PRNT1440 N=N+1 PRNT1440 N=N+1 PRNT1440 N=N+1 PRNT1450 GO TO 220 PRNT1450 N1=NROW-MH1 PRNT1450 N1=NROW-MH1 PRNT150 N1=NROW-MH1 PRNT150 IF (N+GT-IRWKP(K)) K=K+1 PRNT150 IF (N+GT-IRWKP(K)) K=K+1 PRNT1550 IF (N+GT-IRWKP(K)) (EGVECT(N1,L),L=1,5),LC PRNT1600 PRNT1600 PRNT1620 2352		WRITE (ITAPE,390) (RJY(I),RJZ(I),RJX(I),I ,I=1,NJOINT)	PRNT1310
170       CONTINUE       PRN11330         IF (NROW.EQ.MORDER) GO TO 180       PRN11340         WRITE (6,290) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR)       PRN11340         WRITE (6,400)       PRN11350         WRITE (6,400)       PRN11360         WRITE (6,400)       PRN11370         LCOUNT=11       PRN1380         180       CONTINUE         M=NROW-MORDER       PRN1400         K=1       PRN1420         D0 260 J=1,NJOINT       PRN1440         D0 250 J=1,3       PRN1440         N=N+1       PRN1440         N=K       PRN1440         N=K       PRN1450         O 0 220       PRN1440         N=K       PRN1450         N=K       PRN1450         N=K       PRN1450         D0 220       PRN1450         O 1 220       PRN1450         Q0 0 IF (N.NE.IRWTOR(M)) GO TO 210       PRN1520         N1=NROW=M+1       PRN1520         M=M=1       PRN1520         GO TO 220       PRN1520         210       CONTINUE       PRN1550         IF (N.GT.IRWF(K)) K=K+1       PRN1550         IF (N.GT.IRWF(K)) M=M-1       PRN1550         GO TO 190		LC=0	PRNT1320
IF (NRCW-EQ-MORDER) GO TO 180       PRN11340         WRITE(6,290) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR)       PRN11350         WRITE(6,310) (IGVNO(I),I=1,5)       PRN11350         WRITE(6,310) (IGVNO(I),I=1,5)       PRN11360         COUNT=11       PRN11380         M=NROW-MORDER       PRN11370         K=1       PRN11420         N=0       PRN11410         N=0       PRN11420         D0 260 I=1,NJOINT       PRN11420         D0 250 J=1,3       PRN11420         N=K       PRN11440         N=K+1       PRN11450         190 IF (N=NE=IRWKP(K)) GO TO 200       PRN11450         N=K       PRN11450         N=K       PRN11450         0 220       PRN11450         0 1F (N=NE=IRWTOR(M)) GO TO 210       PRN11500         N=M=1       PRN11500         M=M=1       PRN11500         M=M=1       PRN11520         GO TO 220       PRN11500         210 CONTINUE       PRN11500         IF (N=GT=IRWTOR(M)) M=M=1       PRN11500         GO TO 190       PRN11500         220 CONTINUE       PRN11500         IF (INDWIM=EQ=0) GO TO 230       PRN11500         IF (INDWIM=EQ=0) GO TO 230	170	CONTINUE	PRNT1330
<pre>WRITE(6,290) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR) PRNT1350 WRITE(6,310) (IGVNO(I),I=1,5) PRNT1370 LCOUNT=11 PRNT1370 LCOUNT=11 PRNT1370 M=NROW-MORDER PRNT1400 K=1 PRNT1400 K=1 PRNT1410 D0 260 I=1,NJOINT PRNT1420 D0 260 J=1,3 PRNT1430 D0 250 J=1,3 PRNT1430 N=N+1 PRNT1450 I90 IF (N.NE.IRWKP(K)) GO TO 200 PRNT1450 N1=K PRNT1450 N1=K PRNT1450 N1=K PRNT1450 N1=NROW-M+1 PRNT1450 N1=NROW-M+1 PRNT1510 M=M-1 PRNT1510 M=M-1 PRNT1520 GO TO 220 PRNT1530 IF (N.GT.IRWFOR(M)) M=M-1 PRNT1550 IF (N.GT.IRWFOR(M)) M=M-1 PRNT1550 IF (INDWNM.EQ.0) GO TO 230 LC=LC+1 PRNT1610 230 CONTINUE IF (INDWNM.EQ.0) GO TO 250</pre>		IF (NROW.EQ.MORDER) GO TO 180	PRNT1340
WRITE (6,400)       PRN1360         WRITE (6,310) (IGVNO(I),I=1,5)       PRN1370         LCOUNT=11       PRN1380         180       CONTINUE       PRN1380         M=NROW-MORDER       PRN1400         N=0       PRN1410         D0 260 I=1,NJOINT       PRN1420         D0 260 J=1,3       PRN1440         N=N+1       PRN1440         N=N+1       PRN1450         190       IF (N.NE.IRWKP(K)) GO TO 200       PRN1450         N=K       PRN1450       PRN1450         00 250 J=0       PRN1450       PRN1450         190       IF (N.NE.IRWKP(K)) GO TO 200       PRN1450         N=K       PRN1450       PRN1450         00 20       IF (N.NE.IRWTOR(M)) GO TO 210       PRN1500         N1=NROW-M+1       PRN1520       PRN1520         GO TO 220       PRN1520       PRN1520         210       CONTINUE       PRN1550         IF (N.GT.IRWKP(K)) K=K+1       PRN1550         IF (N.GT.IRWKP(K)) K=K+1       PRN1550         IF (N.GT.IRWKP(K)) M=M-1       PRN1550         GO TO 190       PRN1570         220       CONTINUE       PRN1570         IF (INDWNM.EQ.0) GO TO 230       PRN1570		WRITE(6,290) (IGVNO(I),I=1,5), (EGVALU(I),I=LR,MR)	PRNT1350
<pre>WRITE(6,310) (IGVNO(I),I=1,5) LCOUNT=11 PRNT1370 WRITE(6,310) (IGVNO(I),I=1,5) CONTINUE PRNT1380 M=NROW-MORDER PRNT1400 K=1 PRNT1400 K=1 PRNT1400 D0 260 I=1,NJOINT PRNT140 D0 250 J=1,3 PRNT1420 D0 250 J=1,3 PRNT1420 N=N+1 PRNT1450 PRNT1440 N1=K PRNT1450 K=K+1 PRNT1450 G0 T0 220 PRNT140 N1=NROW-M+1 PRNT1450 N1=NROW-M+1 PRNT1500 N1=NROW-M+1 PRNT1520 G0 T0 220 PRNT1500 N1=NROW-M+1 PRNT1520 G0 T0 220 PRNT1550 IF (N.GT.IRWKP(K)) K=K+1 PRNT1520 G0 T0 220 PRNT1550 IF (N.GT.IRWKP(K)) K=K+1 PRNT1550 IF (N.GT.IRWKP(K)) K=K+1 PRNT1550 IF (N.GT.IRWKP(K)) K=K+1 PRNT1550 IF (N.GT.IRWKP(K)) K=K+1 PRNT1550 IF (N.GT.IRWKP(K)) K=K+1 PRNT1550 IF (N.GT.IRWTOR(M)) M=M-1 PRNT1550 IF (INDWNM.EQ.0) G0 T0 230 LC=LC+1 PRNT1550 IF (INDWNM.EQ.0) G0 T0 230 PRNT1550 IF (INDWNM.EQ.0) G0 T0 250 PRNT1630 352</pre>		WRITE (6,400)	PRNT1360
LCOUNT=11       PRNT1380         180       CONTINUE       PRNT1390         M=NROW-MORDER       PRNT1400         k=1       PRNT1410         N=0       260 I=1,NJOINT       PRNT1410         D0 260 J=1,3       PRNT1420         N=N+1       PRNT1450         190       IF (N.NE.IRWKP(K)) GO TO 200       PRNT1450         N1=K       PRNT1450         K=K+1       PRNT1450         GO TO 220       PRNT1450         200       IF (N.NE.IRWTOR(M)) GO TO 210       PRNT1450         N1=NROW-M+1       PRNT1500         M=M-1       PRNT1520         GO TO 220       PRNT1520         210       CONTINUE       PRNT1520         IF (N.GT.IRWKP(K)) K=K+1       PRNT1520         IF (N.GT.IRWKP(K)) K=K+1       PRNT1520         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1550         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1550         IF (INDWNM.EQ.0) GO TO 230       PRNT1580         IF (INDWNM.EQ.0) GO TO 230       PRNT1580         IF (INDWNM.EQ.0) GO TO 230       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1620         230       CONTINUE       PRNT1620         IF (NROW.EQ.MORDER) GO TO 250		WRITE(6,310) (IGVNO(I),I=1,5)	PRNT1370
180       CONTINUE       PRNT1390         M=NROW-MORDER       PRNT1400         K=1       PRNT1410         N=0       PRNT1420         D0 260 I=1,NJOINT       PRNT1420         D0 250 J=1,3       PRNT1450         N=N+1       PRNT1450         190       IF (N*NE*IRWKP(K)) GO TO 200       PRNT1450         N1=K       PRNT1450         GO TO 220       PRNT1450         200       IF (N*NE*IRWTOR(M)) GO TO 210       PRNT1450         N1=NROW-M+1       PRNT1500         M=M-1       PRNT1520         GO TO 220       PRNT1520         210       CONTINUE       PRNT1520         IF (N*GT*IRWKP(K)) K=K+1       PRNT1530         IF (N*GT*IRWKP(K)) K=K+1       PRNT1530         IF (N*GT*IRWKP(K)) K=K+1       PRNT1550         IF (N*GT*IRWKP(K)) K=K+1       PRNT1550         IF (N*GT*IRWKP(K)) K=K+1       PRNT1550         IF (INDWNM*EQ*0) GO TO 230       PRNT1570         220       CONTINUE       PRNT1560         IF (INDWNM*EQ*0) GO TO 230       PRNT1560         IF (INDWNM*EQ*0) GO TO 230       PRNT1600         WRITE (ITAPE*410) (EGVECT(N1*L)*L=1*5)*LC       PRNT1620         IF (NROW*EQ*MORDER) GO TO 250		LCOUNT=11	PRNT1380
M=NROW-MORDER       PRNT1410         K=1       PRNT1410         N=0       PRNT1420         D0 260 I=1,NJOINT       PRNT1430         D0 250 J=1;3       PRNT1450         N=N+1       PRNT1450         190 IF (N*NE*IRWKP(K)) GO TO 200       PRNT1450         N1=K       PRNT1450         K=K+1       PRNT1450         GO TO 220       PRNT1470         N=NROW-M+1       PRNT1500         M=M=1       PRNT1520         GO TO 220       PRNT1520         210 CONTINUE       PRNT1520         IF (N*GT*IRWKP(K)) K=K+1       PRNT1520         IF (N*GT*IRWKP(K)) K=K+1       PRNT1520         IF (N*GT*IRWKP(K)) K=K+1       PRNT1520         IF (N*GT*IRWKP(K)) K=K+1       PRNT1530         IF (N*GT*IRWKP(K)) K=K+1       PRNT1550         IF (INDWIM*EQ*0) GO TO 230       PRNT1560         IF (INDWIM*EQ*0) GO TO 230       PRNT1580         IF (INDWIM*EQ*0) (EGVECT(N1*L)*L=1*5)*LC       PRNT1600         230 CONTINUE       PRNT1620         IF (NROW*EQ*MORDER) GO TO 250       PRNT1630         352       352	180	CONTINUE	PRNT1390
K=1 N=0       PRNT1410         D0 260 I=1,NJOINT       PRNT1420         D0 250 J=1,3       PRNT1450         N=N+1       PRNT1450         190 IF (N.NE.IRWKP(K)) GO TO 200       PRNT1460         N1=K       PRNT1450         K=K+1       PRNT1460         GO TO 220       PRNT1470         200 IF (N.NE.IRWTOR(M)) GO TO 210       PRNT1500         N1=NROW-M+1       PRNT1520         M=M-1       PRNT1520         GO TO 220       PRNT1520         J1=N (N.GT.IRWKOR(K)) K=K+1       PRNT1520         JF (N.GT.IRWFOR(M)) M=M-1       PRNT1520         GO TO 190       PRNT1550         Z20       CONTINUE         JF (INDWMM.EQ.0) GO TO 230       PRNT1570         LC=LC+1       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1610         Z30       CONTINUE       PRNT1620         JF (NROW-EQ.MORDER) GO TO 250       PRNT1630		M=NROW-MORDER	PRNT1400
N=0       PRNT1420         DO 260 I=1,NJOINT       PRNT14430         DO 250 J=1,3       PRNT14430         N=N+1       PRNT1450         190 IF (N.NE.IRWKP(K)) GO TO 200       PRNT1450         N1=K       PRNT1450         K=K+1       PRNT1450         GO TO 220       PRNT1470         200 IF (N.NE.IRWTOR(M)) GO TO 210       PRNT1500         N1=NROW-M+1       PRNT1520         M=M-1       PRNT1520         GO TO 220       PRNT1530         210 CONTINUE       PRNT1530         IF (N.GT.IRWKP(K)) K=K+1       PRNT1550         IF (N.GT.IRWKP(K)) K=K+1       PRNT1550         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1550         GO TO 190       PRNT1550         IF (INDWNM.EQ.0) GO TO 230       PRNT1550         LC=LC+1       PRNT1590         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1600         230 CONTINUE       PRNT1620         IF (NROW.EQ.MORDER) GO TO 250       PRNT1630         352       352		K=1	PRNT1410
D0 260 1=1,NJOINI       PRNT1430         D0 250 J=1,3       PRNT1440         N=N+1       PRNT1450         190 IF (N.NE.IRWKP(K)) GO TO 200       PRNT1450         N1=K       PRNT1450         K=K+1       PRNT1460         GO TO 220       PRNT1450         200 IF (N.NE.IRWTOR(M)) GO TO 210       PRNT1500         N1=NROW-M+1       PRNT1520         M=M-1       PRNT1520         GO TO 220       PRNT1520         IF (N.NET.IRWTOR(M)) MGO TO 210       PRNT1520         M=M-1       PRNT1520         GO TO 220       PRNT1520         IF (N.GT.IRWKP(K)) K=K+1       PRNT1550         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1550         GO TO 190       PRNT1570         220 CONTINUE       PRNT1570         IF (INDWMM.EQ.0) GO TO 230       PRNT1570         LC=LC+1       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1610         230 CONTINUE       PRNT1620         IF (NROW.EQ.MORDER) GO TO 250       PRNT1630         352       PR		N=0	PRNT1420
D0 250 J=1,3       PRNT1440         N=N+1       PRNT1450         190 IF (N.NE.IRWKP(K)) GO TO 200       PRNT1460         N1=K       PRNT1470         K=K+1       PRNT1480         GO TO 220       PRNT1490         200 IF (N.NE.IRWTOR(M)) GO TO 210       PRNT1510         M=M-1       PRNT1510         GO TO 220       PRNT1520         GO TO 220       PRNT1520         GO TO 220       PRNT1520         IF (N.GT.IRWKP(K)) K=K+1       PRNT1530         IF (N.GT.IRWKP(K)) K=K+1       PRNT1560         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1560         GO TO 190       PRNT1570         220 CONTINUE       PRNT1570         IF (INDWNM.EQ.0) GO TO 230       PRNT1590         LC=LC+1       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1620         230 CONTINUE       PRNT1620         IF (NROW.EQ.MORDER) GO TO 250       PRNT1630         352       A		DO 260 I=1,NJOINI	PRNT1430
N=N+1       PRNT1450         190       IF (N.NE.IRWKP(K)) GO TO 200       PRNT1460         N1=K       PRNT1480         GO TO 220       PRNT1480         200       IF (N.NE.IRWTOR(M)) GO TO 210       PRNT1490         N1=NROW-M+1       PRNT1500         M=M-1       PRNT1520         GO TO 220       PRNT1520         IF (N.NE.IRWTOR(M)) GO TO 210       PRNT1500         N1=NROW-M+1       PRNT1520         M=M-1       PRNT1520         GO TO 220       PRNT1520         IF (N.GT.IRWKP(K)) K=K+1       PRNT1550         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1550         GO TO 190       PRNT1570         220       CONTINUE       PRNT1560         IF (INDWNM.EQ.0) GO TO 230       PRNT1570         LC=LC+1       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1610         230       CONTINUE       PRNT1620         IF (NROW.EQ.MORDER) GO TO 250       PRNT1630		DO 250 J=1,3	PRNT1440
190       IF (N.NE.IRWKP(K)) GO TO 200       PRN11460         N1=K       PRN11470         GO TO 220       PRN11490         200       IF (N.NE.IRWTOR(M)) GO TO 210       PRN11490         N1=NROW-M+1       PRN11500         M=M-1       PRN11520         GO TO 220       PRN11520         210       CONTINUE       PRN1520         IF (N.GT.IRWKP(K)) K=K+1       PRN1500         IF (N.GT.IRWTOR(M)) M=M-1       PRN1560         GO TO 190       PRN1570         220       CONTINUE       PRN1560         IF (INDWNM.EQ.O) GO TO 230       PRN1570         LC=LC+1       PRN1580         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN1600         230       CONTINUE       PRN1600         IF (NROW.EQ.MORDER) GO TO 250       PRN1630		N=N+1	PRNT1450
N1=K       PRNT1470         K=K+1       PRNT1480         GO TO 220       PRNT1490         200 IF (N•NE•IRWTOR(M)) GO TO 210       PRNT1500         N1=NROW-M+1       PRNT1510         M=M-1       PRNT1520         GO TO 220       PRNT1530         210 CONTINUE       PRNT1520         IF (N•GT•IRWKP(K)) K=K+1       PRNT1550         IF (N•GT•IRWTOR(M)) M=M-1       PRNT1550         GO TO 190       PRNT1570         220 CONTINUE       PRNT1560         IF (INDWNM•EQ•0) GO TO 230       PRNT1580         IF (INDWNM•EQ•0) GO TO 230       PRNT1590         LC=LC+1       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1610         230 CONTINUE       PRNT1610         IF (NROW•EQ•MORDER) GO TO 250       PRNT1630	190	IF (NoNEOIRWKP(K)) GO TO 200	PRNT1460
K=K+1       PRNT1480         GO TO 220       PRNT1480         200       IF (N.NE.IRWTOR(M)) GO TO 210       PRNT1500         N1=NROW-M+1       PRNT1510         M=M-1       PRNT1520         GO TO 220       PRNT1520         210       CONTINUE       PRNT1530         210       CONTINUE       PRNT1540         IF (N.GT.IRWKP(K)) K=K+1       PRNT1550         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1560         GO TO 190       PRNT1570         220       CONTINUE       PRNT1580         IF (INDWNM.EQ.0) GO TO 230       PRNT1570         LC=LC+1       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1610         230       CONTINUE       PRNT1620         IF (NROW.EQ.MORDER) GO TO 250       PRNT1620		NI=K	PRNT1470
200       IF (N.NE.IRWTOR(M)) GO TO 210       PRN11490         200       IF (N.NE.IRWTOR(M)) GO TO 210       PRN11500         N1=NROW-M+1       PRN1510         M=M-1       PRN11510         GO TO 220       PRN11520         210       CONTINUE       PRN1530         210       CONTINUE       PRN1550         IF (N.GT.IRWKP(K)) K=K+1       PRN1550         IF (N.GT.IRWTOR(M)) M=M-1       PRN1560         GO TO 190       PRN1570         220       CONTINUE       PRN1570         IF (INDWNM.EQ.0) GO TO 230       PRN1580         LC=LC+1       PRN1590       PRN1590         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN1600         230       CONTINUE       PRN1600         IF (NROW.EQ.MORDER) GO TO 250       PRN1630		K=K+1	PRNT1480
200       IF (N.NE.IRWIOR(M)) GO TO 210       PRNT1500         N1=NROW-M+1       PRNT1510         M=M-1       PRNT1520         GO TO 220       PRNT1520         210       CONTINUE       PRNT1500         IF (N.GT.IRWKP(K)) K=K+1       PRNT1550         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1560         GO TO 190       PRNT1570         220       CONTINUE       PRNT1580         IF (INDWNM.EQ.0) GO TO 230       PRNT1590         LC=LC+1       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1610         230       CONTINUE       PRNT1620         IF (NROW-EQ.MORDER) GO TO 250       PRNT1630			PRN11490
N1=NROW-M+1       PRN1510         M=M-1       PRN1520         GO TO 220       PRN1530         210       CONTINUE       PRN1540         IF (N.GT.IRWKP(K)) K=K+1       PRN1550         IF (N.GT.IRWTOR(M)) M=M-1       PRN1560         GO TO 190       PRN1570         220       CONTINUE       PRN1580         IF (INDWNM.EQ.0) GO TO 230       PRN1590         LC=LC+1       PRN1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN1600         230       CONTINUE       PRN1610         230       CONTINUE       PRN1610         352       352       A	200	$IF (N \bullet NE \bullet IRWIOR(M)) GO IO 210$	PRNT1500
M=M-1       PRNT1520         GO TO 220       PRNT1530         210       CONTINUE       PRNT1540         IF (N.GT.IRWKP(K)) K=K+1       PRNT1550         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1560         GO TO 190       PRNT1570         220       CONTINUE       PRNT1580         IF (INDWNM.EQ.0) GO TO 230       PRNT1590         LC=LC+1       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1610         230       CONTINUE       PRNT1620         IF (NROW.EQ.MORDER) GO TO 250       PRNT1630		NI=NROW-M+I	PRNT1510
GO TO 220       PRN11530         210       CONTINUE       PRN11540         IF (N.GT.IRWKP(K)) K=K+1       PRN11550         IF (N.GT.IRWTOR(M)) M=M-1       PRN11560         GO TO 190       PRN11570         220       CONTINUE       PRN11580         IF (INDWNM.EQ.0) GO TO 230       PRN11590         LC=LC+1       PRN11600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN11610         230       CONTINUE       PRN11620         IF (NROW.EQ.MORDER) GO TO 250       PRN11630		M=M-1	PRN11520
210       CONTINUE       PRNT1540         IF (N.GT.IRWKP(K)) K=K+1       PRNT1550         IF (N.GT.IRWTOR(M)) M=M-1       PRNT1560         GO TO 190       PRNT1570         220       CONTINUE       PRNT1580         IF (INDWNM.EQ.0) GO TO 230       PRNT1590         LC=LC+1       PRNT1600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRNT1610         230       CONTINUE       PRNT1620         IF (NROW-EQ.MORDER) GO TO 250       PRNT1630	210	GUTU 220	PRNI1530
IF (N•G(•IRWRP(R))) K=R+1       PRN11550         IF (N•G(•IRWRP(R))) M=M=1       PRN11560         GO TO 190       PRN11570         220 CONTINUE       PRN11580         IF (INDWNM•EQ•0) GO TO 230       PRN11590         LC=LC+1       PRN11600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN11610         230 CONTINUE       PRN11620         IF (NROW•EQ•MORDER) GO TO 250       PRN11630	210		PRNI1540
IF (N•GI•IRWIOR(M)) M=M=1       PRN11560         GO TO 190       PRN11570         220 CONTINUE       PRN11580         IF (INDWNM•EQ•0) GO TO 230       PRN11590         LC=LC+1       PRN11600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN11610         230 CONTINUE       PRN11620         IF (NROW•EQ•MORDER) GO TO 250       PRN11630		$IF (N \circ O ) \circ IRWRP(N) ) N = N = 1$	PRNI1550
220       CONTINUE       PRN11570         220       CONTINUE       PRN11580         IF (INDWNM•EQ•0) GO TO 230       PRN11590         LC=LC+1       PRN11600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN11610         230       CONTINUE         IF (NROW•EQ•MORDER) GO TO 250       PRN11630		$\frac{1}{1} = \frac{1}{1} = \frac{1}$	PRN11500
220       CONTINUE       PRN11580         IF (INDWNM•EQ•0) GO TO 230       PRN11590         LC=LC+1       PRN11600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN11610         230       CONTINUE       PRN11620         IF (NROW•EQ•MORDER) GO TO 250       PRN11630	220		PRN11570
LC=LC+1       PRN11590         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN11610         230 CONTINUE       PRNT1610         IF (NROW-EQ-MORDER) GO TO 250       PRN11630	~~U	LE (INDWAM, EO. A) GO TO 220	PRN11000
LC=LC+1       PRN11600         WRITE (ITAPE,410) (EGVECT(N1,L),L=1,5),LC       PRN11610         230 CONTINUE       PRN11620         IF (NROW-EQ-MORDER) GO TO 250       PRN11630         352       352			PRN11590
230       CONTINUE       PRN1610         231       CONTINUE       PRN1620         232       IF (NROW-EQ-MORDER) GO TO 250       PRN1630		LUHLUTI WDITE (ITADE.AIA) /EGVECT(NI.L).L-1 50 LC	PRNI1600
IF (NROW+EQ+MORDER) GO TO 250         PRNT1630           352         352	230	WRITE TTARE 94107 TEOVECTINISE/SE4190/SEC	PRNI1610
352	250	TE (NROW-FO-MORDER) GO TO 250	PRNI1620
352			FIGHT000
	352	<b>A</b>	

.

	WRITE (6,340) I,ALPHA(J),(EGVECT(N1,L),L=1,5)	PRNT1640
	$IE (ICOUNT_I = ECOUNT_I = ECOUN$	DRNT1660
		DDNT1670
		DRNT1680
		DRNT1600
	WR1 E(0,310) (IGVNU(1),1=1,5)	PRN11070
		PRN11700
24.0		PRN11/10
240	TE ( ) NE 2N CO TO 250	PRN11720
	IF (JONE 03) GU 10 250	PRN11750
		PRN11740
250		PRN11750
290		PRN11760
24.0		PRN11770
260	CUNTINUE	PRNI1780
	IF (INDWNM+EQ+0) GO TO 270	PRNI1790
	LRI = LR - I	PRN11800
	WRITE (ITAPE, 420) (EGVALU(LRI+I), 6M(I), 1, 1=1,5)	PRN11810
	WRITE (ITAPE, 430) (WNX(I), WNY(I), WNZ(I), $PX(I)$ , $PY(I)$ , $PZ(I)$ , $I_{1}=1,5$ )	PRNT1820
	CALL PIA/21 (/21, AMASS, MORDER, 11APE)	PRNI1830
	END FILE ITAPE	PRNI1840
270	CONTINUE	PRNT1850
~	RETURN	PRNT1860
C c		PRNI1870
C		PRNT1880
C	INITIALIZE FOR EACH EIGENVALUE GROUP	PRNT1890
C		PRNT1900
	ENTRY PRNTO	PRNT1910
	IFRIST = 0	PRNT1920
	RETURN	PRNT1930
280	FORMAT (45H1EIGENVALUES ASSOCIATED WITH RIGID BODY MODES//(5XE14.7	PRNT1940
		PRNT1950
290	FORMAT (IHI, 41X33HNATURAL FREQUENCIES (RADIANS/SEC)//	PRNT1960
	1 20X5I18//2/X5E18.8//)	PRNT1970
300	FORMAT (36X44HCORRESPONDING MODE SHAPES FOR REDUCED SYSTEM)	PRN11980
310	FORMAT (24HO NODAL POINT ELEMENT/25H NUMBER REFERENCE,	PRN11990
	1 11X12, 41187)	PRNT2000
320	FORMAT (1H1)	PRNT2010
330	FORMAT (1H)	PRN12020
340	FORMAT (6XI3,8XA8,2X5E18.8)	PRNT2030
350	FORMAT (1H1,42X30HGENERALIZED INERTIA PROPERTIES/)	PRNT2040
360	D FORMAT ( 11H VARIABLES,9X5I18/)	PRNT2050
370	FORMAT (1H04X, 3HWNX, 19X5E18.8/1H04X, 3HWNY, 19X5E18.8/1H04X, 3HWNZ, 19	PRNT2060
	1X5E18.8/1H04X,3H PX,19X5E18.8/1H04X,3H PY,19X5E18.8/1H04X,3H PZ,19	PRNT2070
	2X5E18.8/1H05X,15HGM DIAGONAL ,6X5E18.8/(27X5E18.8))	PRNT2080
380	FORMAT (6X2H21,2X10H 5.0 ,2F10.3,37X3H 1)	PRNT2090
390	FORMAT (5X3H221,2X3E10.3,37X13)	PRNT2100
400	FORMAT (36X45HCORRESPONDING MODE SHAPES FOR COMPLETE SYSTEM)	PRNT2110
410	FORMAT (5X3H3212X,5E10.3,17XI3)	PRNT2120
420	FORMAT (5X3H4212X,2E10.3,47XI3)	PRNT2130
430	FORMAT (5X3H5212X,6E10.3,7XI3)	PRNT2140
	END	PRNT2150

	SUBROUTINE P1A721 (NO,ARRAY,LNGTH,I6)	P1A7	10
	DIMENSION ARRAY(1)	P1A7	20
	K=1	P1A7	30
	I = 1	P1A7	40
	DO 10 J=6,LNGTH,6	P1A7	50
	WRITE (I6,20) NO,(ARRAY(L),L=I,J),K	P1A7	60
	K=K+1	P1A7	70
	I = I + 6	P1A7	80
10	CONTINUE	PIA7	90
	IF (I.GT.LNGTH) RETURN	P1A7	100
	WRITE (I6,20) NO,(ARRAY(L),L=I,LNGTH),K	P1A7	110
	RETURN	P1A7	120
С		P1A7	130
20	FORMAT (5X,13,2X,6E10.3,7XI3)	P1A7	140
	END	P1A7	150

		CVERLAY ( GEARRT, 5,0)	0V50	10
	150	PROGRAM GEAREX COMMON / CMAIN / COMSAP(2000) EQUIVALENCE ( COMSAP(1), LGRTYP ) I/O UNIT 5 EQUALS INPUT UNIT I/O UNIT 6 EQUALS OUTPUT UNIT I/O UNIT 1 EQUALS WORK AREAV.R. ZERO OUT COMMON DO 150 I = 1, 2000 COMSAP(I) = 0.0	GEAR GEAR GEAR GEAR GEAR GEAR GEAR GEAR	10 20 30 40 50 60 70 80 90
C		CALL THE INPUT ROUTINE	GEAR	110
C C C		CALL OVERLAY ( 6LGEARRT,5,1,6HRECALL) IF( LGRTYP •EQ• 2) GO TO 200 CALL LANDING GEAR ANALYSIS ROUTINE FOR INVERTED TRIPOD GEAR	GEAR GEAR GEAR GEAR GEAR	120 130 140 150 160
C C	200	CALL OVERLAY ( 6LGEARRT,5,2,6HRECALL) GO TO 300 CONTINUE	GEAR GEAR GEAR GEAR GEAR	170 180 190 200 210
C		CALL LANDING GEAR ANALYSIS ROUTINE FOR CANTILEVER GEAR	GEAR GEAR	220 230
	300	CALL OVERLAY ( 6LGEARRT,5,3,6HRECALL) CONTINUE RETURN END	GEAR GEAR GEAR GEAR	240 250 260 270

355

4

,

SUBROUTINE OUTPUT	OUTP 10
COMMON / CMAIN /	00TP 20
I LIGRITY, LUIND, INITER, NPOINT, NREFP, NELEM, NOMD,	
2 IGIDIE, MMATI, IST , IMATI(4) , MNO , EDCRST(3),	0UTP 50
$ = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac$	
A INDELTAY CLASDITAY CLASAN(A) -	
2 DEINERT(4) AREA(4) STRET(4) S	00TP 70
7 SRUT( $4$ ) • SRUT( $4$ ) • CDC(5.4) • SRC(5.4) •	OUTP 90
8 CDI( $5,4$ ) · CCC( $5,4$ ) · SRT( $5,4$ ) · ·	OUTP 100
9 PEC(5,4) • PET(5,4) • NPN • XYZPOS(3,5)	OUTP 110
9 , DC(3,3) , PHI , THTA , PSI	OUTP 120
9 , CGYCDL(3), EGYCDL(3)	OUTP 130
COMMON / CMAIN /	OUTP 140
A NITERS, NDISPS, STLN2 ,	OUTP 150
B IDSPDR, IDSPD1, IDSPD2, SPDISP, COEFF, XYZDSP(3),	OUTP 160
C STLN(4) , STLNO(4) , STRP(4) ,	OUTP 170
D SPNGC(4) , SPNGT(4) , DISTC(4) ,	OUTP 180
E DISTT(4) , PFORC (4) , PFORT(4) ,	OUTP 190
F FOREVC(4) , FOREVT(4) , IPOSC(4) ,	OUTP 200
G IPOST(4) ,INDULC(4) , INDULT(4) ,	OUTP 210
$H = SIR(4) \qquad , AXIALF(4) \qquad , AXSIIF(4) \qquad , AXSIIF(4) \qquad , AXIALF(4) \qquad , AXIALF(4) \qquad , AXSIIF(4) \qquad , AXSIIF(4) \qquad , AXIALF(4) \qquad ,$	OUTP 220
I IPREV(4) , XYZDS (3) , SFORCE(30) ,	00TP 230
$J = CFORCE(30) \qquad \textbf{5}  \text{SUMFM4(6)} \qquad \textbf{5}  \text{R24XYZ(3)} \qquad \textbf{5}$	OUTP 240
K (M(30) ) FMVCLL(12/4) ) FMVCLG(12/4) )	OUTP 250
$L = 15 \Pi K \Gamma P (4), 25 \Pi K \Gamma P (4), 15 \Pi K \Pi P (4), 25 \Pi K \Pi P (4), CG I A I Z (2), M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 M = 100 $	OUTP 200
NATA IDENT. IDAGET / 99. O /	OUTP 280
LE ( LECNT ALTA 2 ) GO TO 700	OUTP 290
IP(NT = 0.0)	OUTP 300
IPAGCT = IPAGCT + 1	OUTP 310
WRITE(6,799)IPAGCT	OUTP 320
799 FORMAT ( 1H1 + 123X 4HPAGE + 14 )	OUTP 330
GO TO 570	OUTP 340
700 CONTINUE	OUTP 350
WRITE(6,760)	OUTP 360
760 FORMAT ( 1H0, 33X6(10H********) )	OUTP 370
570 CONTINUE	OUTP 380
IPCNT = IPCNT + 1	OUTP 390
NDISPS = NDISPS + 1	OUTP 400
WRITE(6,800)NDISPS, NITERS	OUTP 410
800 FORMATI ( 1H0,53X1/H INCREMENTAL STEP,15 /	001P 420
1 38X14,47H TTERATIONS REQUIRED FOR TOLERANCE SATISFACTION 7 1	001P 430
	OUTP 440
	OUTP 450
999 FORMAT ( 8H	OUTP 400
1 15HEXTERNAL FORCES 24X	OUTP 480
2 16HEXTERNAL MOMENTS / 8H NODAL	CUTP 490
2 4X 3( 7X25HSURFACE COORDINATE SYSTEM, 7X) /	OUTP 500
3 3X5HPOINT, 4X3(6X1HX,12X1HY,12X1HZ,6X), / )	OUTP 510
DO 210 I = 1, NPOINT	OUTP 520
IF ( I •EQ• 4 ) GO TO 200	OUTP 530
WRITE(6,1000)I,(XYZPOS(J,I),J=1,3),(FMVCTG(J,N),J=1,3),	OUTP 540
1 (FMVCTG(J,N),J=7,9)	OUTP 550

	N = N + 1	OUTP	560
200		OUTP	570
200	CONTINUE	OUTP	580
. 1 .	WRILE(6,1000)1,(XYZPOS(J,1),J=1,3),(SUMFM4(J),J=1,6)	OUTP	590
210	CONTINUE	OUTP	600
1000	FORMAI (4X12,4X9E13.5)	OUTP	610
		OUTP	620
	WRITE(6,1099)	OUTP	630
1099	FORMAT (20HOSTRUCTURAL AXIAL, 7X7H AXIAL, 7X5HAXIAL,	OUTP	640
÷	* 20X5HSHEAR,	OUTP	650
	1 8X5HSHEAR, 21X5HSHEAR, 8X5HSHEAR /	OUTP	660
	2 48H ELEMENT FORCE STROKE STIFFNESS, 18X	OUTP	670
	3 5HFORCE, 8X6HMOMENT, 20X5HFORCE, 8X6HMOMENT / )	OUTP	680
	DO 220 I = 1, NELEM	OUTP	690
C	ONE LOCAL SHEAR FORCE/MOMENT IS ZERO	OUTP	700
	SHARFP = YSHRFP(I) + ZSHRFP(I)	OUTP	710
	SHARMP = YSHRMP(I) + ZSHRMP(I)	OUTP	720
	SHARFQ = YSHRFQ(I) + ZSHRFQ(I)	OUTP	730
	SHARMQ = YSHRMQ(I) + ZSHRMQ(I)	OUTP	740
		OUTP	750
	$IF (I \bullet EQ \bullet 4) J = 5$	OUTP	760
220	WRITE(6,1100)I,AXIALF(I),STR(I) ,AXSTIF(I),J,SHARFP,SHARMP,	OUTP	770
	1 SHARFQ,SHARMQ	OUTP	780
1100	FORMAT ( 4XI2,4X3E13.5,3X4HNODE I2,3X2E13.5,3X7HNODE 4,3X2E13.5)	OUTP	790
	WRITE(6,1199) NFPCRL	OUTP	800
1199	FORMAT( 1H0,77X 42HNUMBER OF UNCRUSHED FOOTPAD CRUSH LEVELS =, I2 /	OUTP	810
	123X 21HTOTAL ENERGY ABSORBED,38X 36HFOOTPAD ATTENUATION ENERGY THI	IOUTP	820
	1S STEP /	OUTP	830
	21X, 2(4X25HSURFACE COORDINATE SYSTEM9X24HLANDER COORDINATE SYSTEM	OUTP	840
	3 •4X) / 1X4(5X1HX•10X1HY•10X1HZ•5X) )	OUTP	850
	WRITE(6,1200)(EGYXYZ(I),I=1,3),(EGYCDL(I),I=1,3),	OUTP	860
	1 (CGYXYZ(I),I=1,3),(CGYCDL(I),I=1,3)	OUTP	870
1200	FORMAT ( 1H0, 12E11.3 )	OUTP	880
	RETURN	OUTP	890
	END	OUTP	900

```
ENGY
                                                                            10
  SUBROUTINE ENERGY ( FORC2 )
                                                                       ENGY
  DIMENSION FORC2(3)
                                                                             20
                                                                       ENGY
  DIMENSION FORCL1(3), FORCL2(3), XYZDSL(3)
                                                                             30
  DIMENSION FORC1(3),
                                  IXY7(3)
                                                                       FNGY
                                                                             40
                                                                             50
                                                                       ENGY
  COMMON / CMAIN /
      LGRTYP, LDIND , INITER, NPOINT, NREFP , NELEM , NUMB
                                                                       ENGY
                                                                             60
 1
      NSTEP , NITER , TOL , ENGABS, IFPATN, NFPCRL, FPCRLD(3),
IGLDIR, NMATL , IST , IMATL(4) , MNO , FPCRST(3),
                                                                       ENGY
                                                                            70
 2
                                                                       ENGY
                                                                             80
 3
      INDPL(4)
                   , ELASBN(4) , ELASAX(4)
                                                                       ENGY
                                                                             90
 4
                                                                       ENGY 100
                     , AREA(4)
                                     , IRET(4)
 5
      BINERT(4)
                     scMax(4)
 6
      IFIRST(4)
                                     , STMAX(4)
                                                                       ENGY 110
      SRULC(4) , SRULT(4), CDC(5,4), SRC(5,4)
                                                                       ENGY 120
 7
                                                   .
                    , CRC(5,4)
                                  , SRT(5,4)
                                                                       ENGY 130
 8
     CDT(5,4)
                                     , NPN , XYZPOS(3,5)
                                                                       ENGY 140
      PFC(5,4)
                     , PFT(5,4)
 9
                                                                       ENGY 150
    , DC(3,3)
 Q
                     , PHI
                                     , THTA
                                                   , PSI
                                                                       ENGY 160
    , CGYCDL(3), EGYCDL(3)
 9
  COMMON / CMAIN /
                                                                       ENGY 170
      NITERS, NDISPS, STLN2 ,
                                                                       ENGY 180
 А
       IDSPDR, IDSPD1, IDSPD2, SPDISP, COEFF, XYZDSP(3),
                                                                       ENGY 190
 В
                                                                       ENGY 200
                                              , STRP(4) ,
 С
       STLN(4)
                   stlno(4)
                                     , DISTC(4)
                     , SPNGT(4)
                                                                       ENGY 210
 D
      SPNGC(4)
                                                      ,
                     , PFORC (4)
                                     , PFORT(4)
                                                                       ENGY 220
 E
      DISTT(4)
                                                      ,
                     , FOREVT(4)
                                     , IPOSC(4)
                                                                       ENGY 230
      FOREVC(4)
 F
                                                      ,
                     , INDULC(4)
                                                                       ENGY 240
 G
       IPOST(4)
                                     , INDULT(4)
                                                      ,
                                                                       ENGY 250
                                     AXSTIF(4)
       STR(4)
                     • AXIALF(4)
 Н
                                                      ,
                     , XYZDS (3)
                                     , SFCRCE(30)
                                                                       ENGY 260
      IPREV(4)
 T
                                                                       ENGY 270
                     SUMEM4(6)
                                     ,R24XYZ(3)
       CFCRCE(30)
 J
                     • FMVCTL(12,4) • FMVCTG(12,4)
                                                                       ENGY 280
       TM(36)
 Κ
       YSHREP(4), ZSHREP(4), YSHRMP(4), ZSHRMP(4), CGYXYZ(3),
                                                                       ENGY 290
 1
                                                                       ENGY 300
       YSHRFQ(4), ZSHRFQ(4), YSHRMQ(4), ZSHRMQ(4), EGYXYZ(3)
 Μ
                                                                       ENGY 310
  DATA IXYZ / 1HX, 1HY, 1HZ /
                                                                       ENGY 320
       ZERO OUT FOOTPAD CRUSH ENERGY
                                                                       ENGY 330
  DO 10 I = 1, 3
                                                                       ENGY 340
10 CGYXYZ(I) = 0.0
                                                                       ENGY 350
   IF ( IFPATN .EQ. 2 ) GO TO 50
   IF ( NFPCRL .EQ. 0 ) CO TO 50
                                                                       ENGY 360
       CALCULATE THE FORCE COMPONENT AT THE FOOTPAD JOINT
                                                                       ENGY 370
            NORMAL TO THE LANDING SURFACE
                                                                       ENGY 380
  RESULF = FORC2( IDSPDR )
                                                                       ENGY 390
                                                                       ENGY 400
       CALCULATE THE FOOTPAD CRUSH ENERGY FOR THIS STEP IN SURFACE
                                                                       ENGY 410
            COORDINATES AND ADD IT TO TOTAL ENERGY IN SURFACE
                                                                       ENGY 412
            COORDINATES
                                                                       ENGY 414
                                                                       ENGY 420
                                                                       ENGY 430
   K = NFPCR1
                                                                       ENGY 440
   J = NFPCRL
   DO 30 I =1,J
                                                                       ENGY 450
   IF ( RESULE .LT. FPCRLD(K) ) GO TO 30
                                                                       ENGY 460
                                                                       ENGY 470
   NFPCRL = NFPCRL -1
                                                                       ENGY 480
   CGYXYZ(IDSPDR) = CCYXYZ(IDSPDR) + FPCRLD(K)*FPCRST(K)
                                                                       ENGY 490
30 K = K - 1
   IF ( NFPCRL .EQ. J ) GO TO 50
                                                                       ENGY 500
   EGYXYZ(IDSPDR) = EGYXYZ(IDSPDR) + CGYXYZ(IDSPDR)
                                                                       ENGY 510
                                                                       ENGY 520
50 CONTINUE
                                                                       ENGY 530
```

```
358
```

С

C

С

С

Ċ

C C

С

С

		CALCULATE FOOTPAD CRUSH ENERGY FOR THIS STEP IN LANDER COORDINATES BY DIRECT TRANSFORMATION. CALCULATE ENERGY FOR THIS STEP DUE TO DISPLACEMENT OF FOOTPAD JOINT IN BOTH SURFACE AND LANDER COORDINATES	ENGY ENGY ENGY ENGY FNGY	540 550 560 565 570
		D0 410 I = 1, 3 CGYCDL(I) = 0.0 FORCL2(I) = 0.0 XYZDSL(I) = 0.0	ENGY ENGY ENGY ENGY	580 590 600 610
		DO 400 J = 1, 3 CGYCDL(I) = CGYCDL(I) + DC(J,I)*CGYXYZ(J) FORCL2(I) = FORCL2(I) + DC(J,I)* FORC2(J) XYZDSL(I) = XYZDSL(I) + DC(J,I)* XYZDS(J)	ENGY ENGY ENGY ENGY	620 630 640 650
C C C	400	CONTINUE CALCULATE TOTAL ENERGY IN LANDER COORDINATES AS THE SUM OF THE PREVIOUS TOTAL AND THE FOOTPAD CRUSH ENERGY FOR	ENGY ENGY ENGY ENGY	660 661 662 663
C C C		THIS STEP AND THE ENERGY DUE TO FOOTPAD JOINT DISPLACEMENT FOR THIS STEP EGYCDL(I) = EGYCDL(I) + XYZDSI(I) *(EORCL)(I)+EORCL2(I)) *-5	ENGY ENGY ENGY ENGY	664 665 666 670
С	410	EGYCDL(I) = EGYCDL(I) + CGYCDL(I) FORCL1(I) = FORCL2(I)	ENGY ENGY ENGY	680 690 700
		CALCULATE TOTAL ENERGY IN SURFACE COORDINATES AS THE SUM OF THE PREVIOUS TOTAL AND THE FOOTPAD CRUSH ENERGY FOR THIS STEP AND THE ENERGY DUE TO FOOTPAD JOINT DISPLACEMENT FOR THIS STEP	ENGY ENGY ENGY ENGY ENGY	720 722 722 724 730
С	60	DO 60 I = 1, 3 EGYXYZ(I) = EGYXYZ(I) + XYZDS(I)*(FORC1(I) + FORC2(I)) * .5 FORC1(I) = FORC2(I) CHECK ENERGY LIMIT	ENGY ENGY ENGY ENGY	740 750 760 770
		IF ( IGLDIR •EQ• •0 ) GO TO 80 IF ( ENGABS •GE• 0•) GO TO 64 IF ( EGYXYZ(IGLDIR) •GT• ENGA55 ) GO TO 80 GO TO 66	ENGY ENGY ENGY ENGY	780 790 800 810
	64 66	CONTINUE IF ( EGYXYZ( IGLDIR ) .LT. ENGABS ) GO TO 80 CONTINUE WRITE(6,71)	ENGY ENGY ENGY ENGY	820 830 840 842
	71 70	FORMAT(1H0, 33X 6(10H********) ) WRITE(6,70)EGYXYZ(IGLDIR),IXYZ(IGLDIR),ENGABS FORMAT(24H0**** THE TOTAL ENERGY (E14.7, 9H) IN THE A1.19H SURFACE COORDINATE.	ENGY ENGY ENGY ENGY	844 350 860 870
		38H DIRECTION EXCEEDS THE ENERGY CUTOFF (, E14.7,6H) ****) IGLDIR = 0	ENGY ENGY ENGY	880 885 390
	80	CONTINUE RETURN ENTRY ENERG1 DO 100 I = 1, 3 FORCL1(I) = $0.0$	ENGY ENGY ENGY ENGY ENGY	900 910 920 930 940
	100	FORC1(I) = 0.0 $FORC1(I) = 0.0$ $FORL = NFPCRL$	ENGY ENGY ENGY	950 960 970

RETURN END ENGY 980 ENGY 990

.

C         MULTIPLY TRANSFORMATION MATRIX TIMES STIFFNESS MATRIX         TRSM 100           C         STORE RESULT IN STIFFNESS MATRIX         TRSM 110           C         NORDR3 = NORDER / 3         TRSM 120           NORDR3 = NORDER / 3         TRSM 140           ICR=0         TRSM 150           D0 40 K = 1, NORDR3         TRSM 150           D0 30 IC = 1, NORDR3         TRSM 160           D0 30 J = 1,3         TRSM 170           JJ=+1/10(1C)         TRSM 120           D0 40 J = 1, NORDER         TRSM 220           A (ICR)=0.0         TRSM 200           D3 JI=.3         TRSM 200           JJ=+1/10(1C)         TRSM 200           D4 0 J = 1, NORDER         TRSM 200           D5 C         TRANSFORMATION MATRIX           C         MULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THE           C         TRANSFORMATION MATRIX           C         MULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THE           C         TRANSFORMATION MATRIX           C         TRANSFORMATION MATRIX           C         TRANSFORMATION MATRIX           C         MULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THE           C         TRANSFORMATION MATRIX           D0 50 ICR = K,		SUBROUTINE TRNFSM( STIFML, TRANSM,NORDER ) DIMENSION STIFML(NORDER,NORDER),TRANSM(3,3) DIMENSION A(12), IJ(4), IJC(12) DIMENSION IRC(12) DATA IJ / 0,3,6,9 / DATA IJC / 3*0, 3*3, 3*6, 3*9 / DATA IRC / 1,2,3,1,2,3,1,2,3,1,2,3 / ITINJ0=1	TRSM TRSM TRSM TRSM TRSM TRSM TRSM	10 20 30 40 50 60 70
NORDR3 = NORDER / 3         TRSM 130           D0 40 K = 1, NORDER         TRSM 130           D0 30 IC = 1, NORDR3         TRSM 150           D0 30 IC = 1, NORDR3         TRSM 160           D0 30 II = 1,3         TRSM 170           ICR=ICR+1         TRSM 180           A(ICR)=0.0         TRSM 200           JJ=J+IJ(IC)         TRSM 200           JJ=J+IJ(IC)         TRSM 200           JJ=J+IJ(IC)         TRSM 200           A(ICR)=A(ICR)+STIFNL(K,JJ)*TRANSM(J,I)         TRSM 220           A(ICR)=A(ICR)+STIFNL(K,JJ)*TRANSM(J,I)         TRSM 220           C         TRSM 200           A0 J = 1, NORDER         TRSM 250           C         MULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THE           C         TRSM 250           C         TRSM 250           C         TRSM 250           C         TRSM 250           C         TRANSFORMATION MATRIX         TRSM 250           C         TRSM 250         TRSM 250           C         TRSM 260         TRSM 260		MULTIPLY TRANSFORMATION MATRIX TIMES STIFFNESS MATRIX STORE RESULT IN STIFFNESS MATRIX	TRSM TRSM TRSM TRSM	100 110 120
DO       30 IC = 1, NORDR3       TRSM 160         DO       30 IE = 1,3       TRSM 170         ICR=ICR+1       TRSM 190         A(ICR)=0.0       TRSM 200         JJ=J+IJ(IC)       TRSM 200         IF (STIFML(K,JJ)+EQ.0.) GO TO 30       TRSM 210         A(ICR)=A(ICR)+STIFML(K,JJ)*TRANSM(J,I)       TRSM 220         CONTINUE       TRSM 200         DO       40 J = 1, NORDER       TRSM 200         CO       TRANSFORMATION MATRIX       TRSM 200         C       MULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THE       TRSM 200         C       TRANSFORMATION MATRIX       TRSM 200         C       TRANSFORMATION MATRIX       TRSM 200         DO       60 K = 1, NORDER       TRSM 200         DO       50 ICR = K, NORDER       TRSM 300         DO       50 JCR = K, NORDER       TRSM 300         DO 50 JCR = K, NORDER       TRSM 310         DO 50 JCR = K, NORDER       TRSM 340         JJ=J+IJC(ICR)       TRSM 340         DO 50 JCR = K, NORDER       TRSM 340         DO 50 JCR = K, N	C	NORDR3 = NORDER / 3 DO 40 K = 1, NORDER ICR=0	TRSM TRSM TRSM	130 140 150
A(ICR)=0.0       TRSM 190         D0 30 J=1.3       TRSM 200         JJ=J+IJ(IC)       TRSM 200         IF (STIFML(K,J).E0.0.) GO TO 30       TRSM 220         A(ICR)=A(ICR)+STIFML(K,JJ)*TRANSM(J,I)       TRSM 220         30 CONTINUE       TRSM 240         D0 40 J = 1, NORDER       TRSM 250         40 STIFML(K.J)=A(J)       TRSM 260         C       TRSM 270         C       MULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THE         C       MULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THE         D0 60 K = 1, NORDER       TPSM 300         D0 50 ICR = K, NORDER       TPSM 320         D0 50 ICR = K, NORDER       TPSM 320         D0 50 JE1,3       TRSM 360         JJ=+IJC(ICR)       TRSM 360         D0 60 J = I,3       TRSM 360         JJ=+IJC(ICR)       TRSM 360         D0 60 J = K, NORDER       TRSM 360         D0 60 J = K, NORDER       TRSM 360         D0 60 J = I,3       TRSM 400         D0 60 J = K, NORDER       TRSM 360         D0 60 J = K, NORDER       TRSM 360         D0 60 J = K, NORDER       TRSM 400         D0 70 I = 1, ICR       TRSM 400         D0 70 J = 1, ICR       TRSM 430 </td <td></td> <td>DO 30 IC =1, NORDR3 DO 30 I=1,3 ICR=ICR+1</td> <td>TRSM TRSM TRSM</td> <td>160 170 180</td>		DO 30 IC =1, NORDR3 DO 30 I=1,3 ICR=ICR+1	TRSM TRSM TRSM	160 170 180
IF (STIFML(K,JJ)*GQ*O*) GO TO 30       TRSM 220         A(ICR)=A(ICR)+STIFML(K,JJ)*TRANSM(J,I)       TRSM 230         DO 40 J = 1; NORDER       TRSM 250         40 STIFML(K,J)=A(J)       TRSM 260         C       TRSM 260         C       TRSM 270         C       TRSM 260         C       TRSM 260         C       TRSM 270         C       TRANSFORMATION MATRIX       TRSM 290         DO 60 K = 1; NORDER       TRSM 310         DO 50 ICR = K; NORDER       TPSM 320         DO 50 JCR = K; NORDER       TPSM 320         DO 50 JCR = K; NORDER       TPSM 320         DO 50 JCR = K; NORDER       TRSM 360         DO 60 J = K; NORDER       TRSM 360         DO 60 J = K; NORDER       TRSM 360         DO 60 J = K; NORDER       TRSM 400         ICR = NORDER - 1       TRSM 410         DO 70 I = 1; ICR       TRSM 420         ICR = NORDER - 1       TRSM 430         DO 70 J = 1C; NORDER       TPSM 440         TPSM 4400       TPSM 4400         ICR = NORDER - 1       <		A(ICR)=0.0 D0 30 J=1.3 JJ=J+IJ(IC)	TRSM TRSM TRSM	190 200 210
DO40J = 1; NORDERHRSM 26040STIFML(K,J)=A(J)TRSM 260CTRSM 270CTRANSFORMATION MATRIXTRSM 290CTRANSFORMATION MATRIXTRSM 300CTRANSFORMATION MATRIXTRSM 300D060 K = 1; NORDERTPSM 320D050 ICR = K; NORDERTPSM 320A(ICP)=0.0TPSM 350J=IRC(ICR)TRSM 350D0 50 J=1;3TRSM 350J=J+IJC(ICR)TRSM 37050A(ICR)=A(ICR)+STIFML(JJ;K)*TRANSM(J;I)TRSM 380D0 60 J = K; NCRDERTRSM 4360GSTIFML(J,K)=A(J)TRSM 410D0 70 I = 1; ICRTRSM 420ICC=I+1TRSM 420D0 70 J = IC;NORDFRTRSM 420CAS A RESULT OF THE ABOVE; STIFML CONTAINS THE STIFFNESSTRSM 450CAS A RESULT OF THE ABOVE; STIFML CONTAINS THE STIFFNESSTRSM 480CRETURNTRSM 500TRSM 450CRETURNTRSM 500TRSM 450CRETURNTRSM 500TRSM 450CRETURNTRSM 500TRSM 450CRETURNTRSM 500TRSM 450CRETURNTRSM 500TRSM 450CTRSM 500TRSM 450CTRSM 500TRSM 450CTRSM 500TRSM 450CTRSM 500TRSM 450CTRSM 500TRSM 500CTRSM 500TRSM 500CTRSM 500TRSM 500CTRSM 500TR	30	IF (STIFML(K,JJ).EQ.0.) GO TO 30 A(ICR)=A(ICR)+STIFML(K,JJ)*TRANSM(J,I) CONTINUE	TRSM TRSM TRSM	220 230 240
CMULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THETRSM 290CTRANSFORMATION MATRIXTRSM 300CD0 60 K = 1, NORDERTRSM 310D0 50 ICR = K, NORDERTPSM 320A(ICP)=0.0TPSM 340I = IRC(ICR)TPSM 350D0 50 J=1,3TRSM 360JJ=J+IJC(ICR)TRSM 360D0 60 J = K, NORDERTRSM 37050 A(ICR)=A(ICR)+STIFML(JJ,K)*TRANSM(J,I)TRSM 37050 A(ICR)=A(JCR)TRSM 340CTRSM 410D0 70 I = 1, ICRTRSM 440ICEI+1TRSM 430D0 70 J = IC,NORDERTPSM 440CAS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFFNESSTRSM 440CMATRIX TRANSFORMED TO THE GLOPAL COORDINATE SYSTEMTRSM 480CRETURNTRSM 480CRETURNTRSM 500	40 C	DO 40 J = 1, NORDER STIFML(K,J)=A(J)	TRSM TRSM TRSM	250 260 270
D0       60 K = 1, NORDER       TRSM 320         D0       50 ICR = K, NORDER       TPSM 330         A(ICP)=0.0       TRSM 340         I=IRC(ICR)       TRSM 350         D0       50 J=1,3       TRSM 360         JJ=J+IJC(ICR)       TRSM 360         50 A(ICR)=A(ICR)+STIFML(JJ,K)*TRANSM(J,I)       TRSM 370         50 A(ICR)=A(ICR)+STIFML(JJ,K)*TRANSM(J,I)       TRSM 370         50 D0       50 J = K, NORDER       TRSM 370         60 STIFML(J,K)=A(J)       TRSM 400         ICR = NORDER - 1       TRSM 410         D0       70 I = 1, ICR       TRSM 420         IC=I+1       TRSM 420         D0       TC,NORDER       TRSM 420         IC=I+1       TRSM 4400         D0       TC,NORDER       TRSM 4400         C       AS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFFNESS       TRSM 450         C       AS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFFNESS       TRSM 460         C       RETURN       TRSM 480       TRSM 490         C       RETURN       TRSM 500       TRSM 490		MULTIPLY THE RESULT OF THE ABOVE TIMES THE TRANSPOSE OF THE TRANSFORMATION MATRIX	TRSM TRSM TRSM	290 290 300 310
I = IRC (ICR)TRSM 350D0 50 J=1,3TRSM 360JJ=J+IJC(ICR)TRSM 37050 A(ICR)=A(ICR)+STIFML(JJ,K)*TRANSM(J,I)TRSM 380D0 60 J = K, NCRDERTRSM 39060 STIFML(J,K)=A(J)TRSM 400ICR = NORDER - 1TRSM 410D0 70 I = 1, ICRTRSM 420IC=I+1TRSM 430D0 70 J = IC,NORDERTPSM 440T0 STIFML(I,J)=STIFML(J,I)TRSM 450CAS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFFNESSTRSM 450CMATRIX TRANSFORMED TO THE GLOPAL COORDINATE SYSTEMTRSM 480CRETURNTRSM 450CRETURNTRSM 500	C	DO 60 K = 1, NORDER DO 50 ICR = K, NORDER A(ICR)=0.0	TRSM TPSM TDSM	320 330 340
50A(ICR)=A(ICR)+STIFML(JJ,K)*TRANSM(J,I)TRSM 38050G0 J = K, NCRDERTRSM 39060STIFML(J,K)=A(J)TRSM 4001CR = NORDER - 1TRSM 410D0 70 I = 1, ICRTRSM 420IC=I+1TRSM 430D0 70 J = IC,NORDERTPSM 44070STIFML(I,J)=STIFML(J,I)CAS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFFNESSCMATRIX TRANSFORMED TO THE GLOPAL COORDINATE SYSTEMCRETURNCTRSM 490TRSM 490		I = IRC(ICR) DO 50 J=1,3 JJ=J+IJC(ICR)	TRSM TRSM TRSM	350 360 370
ICR = NORDER - 1TRSM 410DO 70 I = 1, ICRTRSM 420IC=I+1TRSM 430DO 70 J = IC,NORDERTPSM 44070 STIFML(I,J)=STIFML(J,I)TRSM 450CAS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFFNESSTRSM 450CMATRIX TRANSFORMED TO THE GLOPAL COORDINATE SYSTEMTRSM 480CRETURNTRSM 450CRETURNTRSM 500	50 60	A(ICR)=A(ICR)+STIFML(JJ,K)*TRANSM(J,I) DO 60 J = K, NCRDER STIFML(J,K)=A(J)	TRSM TRSM TRSM	380 390 400
DO       70 J = IC,NORDER       TESM 440         70 STIFML(I,J)=STIFML(J,I)       TRSM 450         C       AS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFFNESS       TRSM 460         C       MATRIX TRANSFORMED TO THE GLOPAL COORDINATE SYSTEM       TRSM 480         C       RETURN       TRSM 450         C       RETURN       TRSM 500         C       AS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFFNESS       TRSM 480         C       MATRIX TRANSFORMED TO THE GLOPAL COORDINATE SYSTEM       TRSM 480         C       TRSM 500       TRSM 500		ICR = NORDER - 1 DO 70 I = 1, ICR IC=I+1	TRSM TRSM TRSM	410 420 430
C MATRIX TRANSFORMED TO THE GLOPAL COORDINATE SYSTEM TRSM 480 C RETURN TRANSFORMED TO THE GLOPAL COORDINATE SYSTEM TRSM 490 TRSM 500	70 C	DO 70 J = 1C,NORDER STIFML(I,J)=STIFML(J,I) AS A RESULT OF THE ABOVE, STIFML CONTAINS THE STIFENESS	TPSM TRSM TPSM TRSM	440 450 460 470
	c c	MATRIX TRANSFORMED TO THE GLOPAL COORDINATE SYSTEM	TRSM TRSM TRSM	480 490 500

```
SUBROUTINE TRALMG( FMG, FML, TM, NG )
                                                                             TRFM 10
                                                                             TRFM 20
С
          F \cdot M \cdot GLOBAL = (TM)T \cdot (F \cdot M \cdot LOCAL)
Ċ
            FM. VECTOR = FXP, FYP, FZP, FXQ, FYQ, FZQ, MXP..., MXQ, MYQ, MZQ
                                                                             TREM 30
                                                                             TRFM 40
      DIMENSION FMG(12), FML(12), TM(3,3)
                                                                             TREM 50
      N = C
      M = 0
                                                                             TRFM
                                                                                    60
      DC 30
                                                                             TRF/1
              I = 1, NG
                                                                                   70
C
C
                                                                             TRFM 80
                                                                             TEFM 90
           TRANSFORM LOCAL FORCE-MOMENTS TO GLOBAL FORCE-MOMENTS
                                                                             TRFM 100
С
      DO 20 J = 1, 3
M = M + 1
                                                                             TRFM 110
                                                                             TRFM 120
      FMG(M) = 0.0
                                                                             TRFM 130
      DO 20 K = 1, 3
                                                                             TRFM 140
      NK = N + K
                                                                             TREM 150
                                                                             TRFM 160
      FMG(M) = FMG(M) + TM(K,J) * FML(NK)
                                                                             TRFM 170
   20 CONTINUE
   30 N = N + 3
                                                                             TRFM 180
                                                                             TRFM 190
      RETURN
                                                                             TRFM 200
      END
```

```
362
```

```
SUBROUTINE STFMIT( STIFML, PO, DPQ, AE )
                                                                                 STAX 10
      DIMENSION STIFML(6,6)
                                                                                 STAX
                                                                                        20
000000
                                                                                 STAX
                                                                                        30
           COMPUTE THE LOCAL STIFFNESS MATRIX OF AN AXIAL STRUT
                                                                                 STAX
                                                                                        40
                                                                                 STAX
                                                                                        50
                      PO = CURRENT AXIAL FORCE
                                                                                 STAX
                                                                                        60
                      AE = CURRENT AXIAL STIFFNESS
                                                                                 STAX
                                                                                        70
                      DPQ= LENGTH OF THE PQ ELEMENT
                                                                                 STAX 80
       DO 100 I = 2, 6
K = I - 1
                                                                                 STAX 90
STAX 100
      DO 100 J = 1, K
                                                                                 STAX 110
  100 \text{ STIFML}(I,J) = 0.00
                                                                                 STAX 120
       STIFML(1,1) = AE
                                                                                 STAX 130
       STIFML(2,2) = PO / DPQ
                                                                                 STAX 140
       STIFML(3,3) = STIFML(2,2)
                                                                                 STAX 150
       STIFML(4,4) = AE
                                                                                 STAX 160
       STIFML(5,5) = STIFML(2,2)
                                                                                 STAX 170
       STIFML(6,6) = STIFML(2,2)
STIFML(4,1) = -AE
                                                                                 STAX 180
STAX 190
       STIFML(5,2) = -STIFML(2,2)
                                                                                 STAX 200
       STIFML(6,3) = -STIFML(2,2)
                                                                                 STAX 210
       DO 200 I = 2,6
                                                                                 STAX 220
  K = I - 1
DO 200 J = 1, K
200 STIFML(J,I) = STIFML(I,J)
                                                                                 STAX 230
                                                                                 STAX 240
                                                                                 STAX 250
       RETURN
                                                                                 STAX 260
                                                                                 STAX 270
       END
```

.

	SUBROUTINE TRANSM( P, Q, R, TM, TOTLPO )	TRAN 10
	DIMENSION $P(3)$ , $Q(3)$ , $TN(3,3)$ , $R(3)$	TRAN 20
	DOUBLE PRECISION X, Y, Z, TGTL , TMD(3,3), DPE, DER	TPAN 30
C	LI MI NI THE TRANSFORMATION MATRIX REQUIRED TO	TRAN 40
С	TM = L2 M2 N2 CONVERT GLOBAL COORDINATES TO LOCAL	TRAN 50
С	L3 M3 N3 COORDINATES	TRAN 60
	X = Q(1) - P(1)	TRAN .70
	Y = Q(2) - P(2)	TRAN 80
	Z = Q(3) - P(3)	TRAN 00
С	COMPUTE THE TOTAL LENGTH OF THE BAR PQ	TRAN 100
	TOTL = DSQRT(X*X + Y*Y + Z*Z)	TRAN 110
	TOTLPQ=TOTL	TRAN 120
С	COMPUTE THE DIRECTION COSINES OF THE LOCAL X AXIS PO	TRAN 130
	TMD(1,1) = X / TOTL	TRAN 140
	T(D(1,2) = Y / TOTL	TRAN 150
	$TMD(1,3) = \mathbb{Z} / TOTL$	TRAN 150
	X = R(1) - P(1)	TRA11 170
	Y = R(2) - P(2)	TRAN 180
	Z = R(3) - P(3)	TRAN 190
	DPE = TMD(1,1)*X + TMD(1,2)*Y + TMD(1,3)*Z	TRAN 200
_	DER = DSQRT( X*X + Y*Y + Z*Z - DPE*DPE )	TRAN 210
C	COMPUTE THE DIRECTION COSINES OF THU LOCAL Y AXIS ER	TRAN 220
C	WHERE E IS THE POINT OF INTERSECTION ON PU OF THE	TRAN 230
C	PERPENDICUEAR DRAWN FROM R TO PS	TRAN 240
	$TMD(2,1) = (\mathbf{X} - TMD(1,1) * DPE) / DEP$	TRAN 200
	$T_{MD}(2,2) = (7 - T_{MD}(1,2) + DPE T_{MD}) DER$	TRAN 200
~	IMD(2,3) = (2 - IMD(1,3)) * DPE / DEK	TRAN 170
C	COMPUTE THE DIRECTION COSINES OF THE LOCAL 2 AXIS FRXPG	TRAN 280
	IMD(3,1) = IMD(1,2)*IMD(2,3) - IMD(2,2)*IMD(1,3)	TRAN 290
	IMD(3,2) = IMD(2,1)*IMD(1,3) - IMD(1,1)*IMD(2,3)	TRAN 300
	$T = \{0, (3, 3)\} = T = T = 0$ (1,1)*T = $T = 0$ (2,2) = T = T = 0 (2,1)*T = 0 (1,2)	TDAN 010
		TRAN 320
		TRAN 350
		TDAN 340
		1KAN 320
	END	THAN 300

a

с		SUBROUTINE DMFSS(A,N,EPS,IRANK,TRAC)	DMFS DMFS	10 20
C C		SUBROUTINE DMFSS DETERMINES THE RANK (IRANK) AND LINEARLY	DMES	30
ĉ		SEMI-DEFINITE MATDIX (A) OF OPDED N AND DEPADES THE MATDIX	DMES	40 50
c		FOR CALCULATION OF THE LEAST SOLARS SOLUTION OF MINIMAL LENGTH		- 60
č		For exception of the EERST Subarts Section of Miniske EERST	DMES	70
č			DMES	80
С		DIMENSIONED DUMMY VARIABLES	DMES	90
		DIMENSION A(1), TRAC(1)	DMFS	100
		DOUBLE PRECISION SUM,A,TRAC,PIV,HOLD	DMFS	110
С		· · · · · ·	DMFS	120
С		TEST OF SPECIFIED DIMENSION	DMFS	130
~		IF(N)36,36,1	DMFS	140
Ċ			DMES	150
C	1	INITIALIZE TRIANGULAR PACTORIZATION	DMES	100
	*		DMES	180
		KPIV=C	DMES	190
		0=L	DMFS	200
		PIV=0.D0	DMFS	210
С			DMFS	220
C		SEARCH FIRST PIVOT ELEMENT	DMFS	230
		DO 3 K=1.N	DMFS	240
			DMFS	250
		$I \in A \subseteq \{k\} = A \subseteq J$	DMES	260
	2		DMES	280
	2		DMES	290
		KPIV=K	DMES	300
	3	CONTINUE	DMES	310
С			DMFS	320
С		START LOOP OVER ALL ROWS OF A	DMFS	330
		DO 32 I=1,N	DMFS	340
		ISUB=ISUE+I	DMFS	350
	7.		DMES	360
	4		DMES	310
C			DMES	300
č		PERFORM PARTIAL COLUMN INTERCHANGE	DMES	400
	5	JI=KSUB-KMI	DMES	410
		IDC=JI-ISUB	DMFS	420
		JJ=ISUB-IM1	DMFS	430
		DO 6 K=JJ,ISUB	DMFS	440
		KK=K+IDC	DMFS	450
			DMFS	460
	۷		DMFS	470
c	0		DMES	400
ĉ		PERFORM PARTIAL ROW INTERCHANGE	DMES	500
2		KK=KSUB	DNES	510
		DO 7 K=KPIV,N	DMFS	520
		II=KK-KMI	DMFS	530
		HOLD=A(KK)	DMFS	540
		A(KK)=A(II)	DMFS	550

		A(II)=HOLD	DMFS 560	ł
	7	KK=KK+K	DMFS 570	1
С			DMFS 580	)
С		PERFORM REMAINING INTERCHANGE	DMFS 590	J
		JJ=KPIV-1	DMFS 600	,
		II=ISUB	DMES 610	)
		$DO = 8  \text{K} = I \bullet J J$	DMES 620	,
		HOLD=A(II)	DMFS 630	)
		A(II) = A(JI)	DMES 640	)
		A(JI) = HOLD	DMES 650	)
		II=IJ+K	DMES 660	,
	8	JI=JI+1	DMFS 670	)
	9	IF(IRANK)22,10,10	DMFS 680	,
С			DMFS 690	)
С		RECORD INTERCHANGE IN TRANSPOSITION VECTOR	DMFS 700	)
	10	TRAC(KPIV)=TRAC(I)	DMFS 710	)
		TRAC(I)=KPIV	DMES 720	)
C ·			DMFS 730	)
С		MODIFY CURRENT PIVOT ROW	DMES 740	)
-		KK=IM1-IRANK	DMFS 750	١
		KMI=ISUB-KK	DMES 760	1
		PIV=0.00	DMES 770	)
		IDC=IRANK+1	DMES 780	)
		JI=ISUB-1	DMFS 790	)
		JK=KMI	DMFS 800	)
		JJ=ISUB-I	DMFS 810	)
		DO 19 K=I,N	DMFS 820	)
		SUM=0.DO	DMFS 830	)
С			DMFS 840	)
С		BUILD UP SCALAR PRODUCT IF NECESSARY	DMFS 850	)
		IF(KK)13,13,11	DMFS 860	)
	11	DO 12 J=KMI,JI	DMES 870	)
		SUM=SUM-A(J)*A(JK)	DMES 880	)
	12	JK = JK + 1	DMES 890	)
	13	JJ=J]+K	DMES 900	)
		IF(K-I)14,14,16	DMFS 910	)
	14	SUM=A(ISUB)+SUM	DMES 920	)
С			DMFS 93	5
C		TEST RADICAND FOR LOSS OF SIGNIFICANCE	DMES 940	.)
		IF(SUM-DABS(A(ISUB)*DBLE(EPS)))20,20,15	DMFS 95	5
	15	A(ISUB)=DSQRT(SUM)	DMES 96	5
		KPIV=I+1	DMFS 97	5
		GOTO 19	DMFS 98	5
	16	SUM=(A(JK)+SUM)/A(ISUP)	DMFS 99	0
-		A(JK)=SUM	DMF 5100	.)
C			DMFS101	D
С		SEARCH FOR NEXT PIVOT ROW	DMFS102	5 D
		IF(A(JJ))19,19,17	DMF S103	0
	17	TRAC(K) = TRAC(K) - SUM*SUM	DMFS104	0
		HOLD = TRAC(K)/A(JJ)	DMFS105	0
	<b>.</b> .	IF(PIV-HOLD)18,19,19	DMFS106	0
	18	PIV=HULU	DMFS107	0
		KPIV=K	DMFS108	0
		KSUE=JJ	DMFS109	0
	19		UMPSIIU OVECIAI	0.
				0



C C		CALCULATE MATRIX OF DEPENDENCIES U
	20	IF(IRANK)21,21,37
	21	IRANK=-1 GOTO 4
	22	IRANK=IM1
		I I = I SUB-I RANK
		JI=II DO 26 K-1 IDANK
		JT=JI-1
		JK = I SUB-1
		JJ=K-1
		DO 26 J=19N TDC=TRANK
		SUM=0.D0
		K M I = J I
		KK=JK TE(11)25•25•23
	23	D0 24 L=1,JJ
		IDC=IDC-1
		SUM=SUM=A(KMI)*A(KK)
	24	KK=KK-1
	25	A(KK)=(SUM+A(KK))/A(KMI)
c	26	JK=JK+J
c		CALCULATE I+TRANSPOSE(U)*U
		JJ=ISUB-I
		PIV=0.D0
		DO 31 K=I,N
		JJ=JJ+K
		IDC=0 DO 28 I=K•N
		SUM=0.DO
		KMI=JJ+IDC
		DO 27 L=II,KK
	27	SUM=SUM+A(L)*A(JK)
		A(KMI)=SUM
	28	IDC=IDC+J
		TRAC(K)=A(JJ)
С		
С		SEARCH NEXT DIAGONAL ELEMENT
	29	KPIV=K
		K SUB=JJ
	20	PIV=A(JJ)
	50	ΓΓ=ΓΓ+Ν ΚΚ=ΚΚ+Κ
	31	CONTINUE
	27	
	33	IF (IRANK) 35, 34, 35
	34	IRANK=N

DMFS1120 DMFS1130 DMFS1140 DMFS1150 DMFS1160 DMFS1170 DMFS1180 DMFS1190 DMES1200 DMFS1210 DMFS1220 DMFS1230 DMFS1240 DMFS1250 DMFS1260 DMFS1270 DMFS1280 DMFS1290 DMFS1300 DMFS1310 DMFS1320 DMFS1330 DMFS1340 DMFS1350 DMFS1360 DMFS1370 DMFS1380 DMFS1390 DMFS1400 DMFS1410 DMFS1420 DMFS1430 DMFS1440 DMFS1450 DMFS1460 DMFS1470 DMFS1480 DMFS1490 DMFS1500 DMFS1510 DMFS1520 DMFS1530 DMFS1540 DMFS1550 DMFS1560 DMFS1570 DMFS1580 DMES1590 DMFS1600 DMFS1610 DMFS1620 DMFS1630 DMFS1640 DMFS1650 DMFS1660 DMFS1670

.

367

	25	RETURN	DMF51680
c	22		DMFS1690
ĉ		FRROR RETURNS	DMFS1700
ĉ			DMFS1710
ĉ		RETURN IN CASE OF ILLEGAL DIMENSION	DMFS1720
C	36	IRANK = -1	DMFS1730
	50	RETURN	DMFS1740
С			DMFS1750
С		INSTABLE FACTORIZATION OF I+TRANSPOSE(U)*U	DMFS1760
	37	IRANK=-2	DMF51770
		RETURN	DMF51780
		END	DWESTIAO

c		SUBROUTINE DMLSS(A,N,IRANK,TRAC,INC,RHS,IER)	DMLS	10
č		SUBROUTINE DMLSS CALCULATES THE LEAST SQUARES SQUITTON OF	DMLS	20
С		MINIMAL LENGTH OF A SYSTEM OF SIMULTANEOUS LINEAR EQUATIONS	DMLS	40
С		WITH SYMMETRIC POSITIVE SEMI-DEFINITE COEFFICIENT MATRIX (A)	DMLS	50
С		WHOSE RANK (IRANK) IS KNOWN AND MAY BE LESS THAN THE ORDER (N)	DMLS	60
С			DMLS	70
С			DMLS	80
С		DIMENSIONED DUMMY VARIABLES	DMLS	90
		DIMENSION A(1),TRAC(1),RHS(1)	DMLS	100
		DOUBLE PRECISION SUM,A,RHS,TRAC,HOLD	DMLS	110
C			DMLS	120
С		TEST OF SPECIFIED DIMENSIONS	DMLS	130
		IDEF=N-IRANK	DMLS	140
	''		DMLS	150
	1 2	IF ( IRANN ) 23 3 3 5 7 2	DMLS	160
c	2		DMLS	170
c		CALCHLATE AUXILIARY VALUES	DMLS	100
C	3	ITEIRANK*(IRANK+1)/2	DMLS	200
	2		DMLS	210
		NP1=N+1	DMLS	220
		IFR=0	DMLS	230
С			DMLS	240
С		INTERCHANGE RIGHT HAND SIDE	DMLS	250
		J J = 1	DMLS	260
		I I = 1	DMLS	270
	4	DO 6 I=1,N	DMLS	280
		J=TRAC(II)	DMLS	290
		IF(J)31,31,5	DMLS	300
	5	HOLD=RHS(II)	DMLS	310
		RHS(11)=RHS(J)	DMLS	320
	,	RHS(J)=HOLD	DMLS	330
	6		DMLS	340
c		IF (JJ) 329 (9 (	DMLS	350
ĉ		DEDEORM STED 2 TE NECESSADY	DMLS	360
C	7	ISWEI	DMLS	370
	'		DMLS	200
С		11 (INC. IDEI 70,20,0	DMLS	390 400
č		CALCULATE X1 = X1 + $U$ * X2	DMLS	410
	8	ISTA=ITE	DMLS	420
		DO 10 I=1+IRANK	DMLS	430
		ISTA=ISTA+1	DMLS	440
		ATTIC	DMLS	450
		SUM=0•D0	DMLS	460
		DO 9 J=IX2.N	DMLS	470
		SUM=SUM+A(JJ)*RHS(J)	DMLS	480
	9		DMLS	490
	10	RHS(I)=RHS(I)+SUM	DMLS	500
~		GOIO(11,28,11),ISW	DMLS	510
ć			DMLS	520
C	1 1	CALCULATE XZ = TRANSPUSE(U) * X1	DMLS	530
	ΤT	101M-11L Do 15 1-1V2 N	DMLS	540
			DMLS	220

	12 13 14 15	JJ=ISTA SUM=0.D0 D0 12 J=1,IRANK JJ=JJ+1 SUM=SUM+A(JJ)*RHS(J) GOTO(13,13,14),ISW SUM=-SUM RHS(I)=SUM ISTA=ISTA+I
C C	16	GOTO(16,29,30),ISW INITIALIZE STEP (4) OR STEP (8) ISTA=IX2 IEND=N JJ=ITE+ISTA
C C	17	DIVISION OF X1 BY TRANSPOSE OF TRIANGULAR MATRIX SUM=0.D0 DO 20 I=ISTA,IEND JE(4(11))18.31.18
	18	RHS(I) = (RHS(I) - SUM) / A(JJ)
		IF(I-IEND)19,21,21
	19	JJ=JJ+ISTA SUM=0. D0
		DO 20 J=ISTA I
		SUM=SUM+A(JJ)*RHS(J)
~	20	1+LL=UL
C C		DIVISION OF YI BY TRIANGULAR MATRIY
C	21	SUM=0.DO
		II=IEND
		DO 24 I=ISTA,IEND
		RHS(II) = (RHS(II) - SUM) / A(JJ)
	- <b>- -</b>	IF(II=ISTA)25,22,22
	22	SUM=0.00
		DO 23 J=II, IEND
		SUM=SUM+A(KK)*RHS(J)
	23	
	24	U = U = -1
	25	IF(IDEF)26,30,26
_	26	GOTO(27,11,8),ISW
C		DEDEORM STED (5)
C	27	TSM=2
	~ '	GOTO 8
С		
C	•	PERFORM STEP (6)
	28	151A=1
		JJ=1
		ISW=2
~		GOTO 17
C		PERFORM STEP (8)
~		

DMLS 560 DMLS 570 DMLS 580 DMLS 590 DMLS 600 DMLS 610 DMLS 620 DMLS 630 DMLS 640 DMLS 650

DMLS 660 DMLS 670 DMLS 680 DMLS 690 DMLS 700

DMLS 710 DMLS 720 DMLS 730 DMLS 740 DMLS 750 DMLS 760 DMLS 770 DMLS 780 DMLS 790 DMLS 800 DMLS 810 DMLS 820

DMLS 830 DMLS 840

DMLS 850 DMLS 860 DMLS 870 DML3 880 DMLS 890 DMLS 900 DMLS 910 DMLS 920 DMLS 930 DMLS 940 DMLS 950 DMLS 960 DMLS 970 DMLS 980 DMLS 990

DMLS1000

DML51010 DMLS1020

DMLS1030 DMLS1040

DMLS1050 DMLS1060 DMLS1070 DMLS1080 DMLS1090 DMLS1100

DMLS1110

4

с	29	ISW=3 GOTO 16	DMLS1120 DMLS1130 DMLS1140
Ċ	30	REINTERCHANGE CALCULATED SOLUTION II=N JJ=-1 GOTO 4	DMLS1150 DMLS1160 DMLS1170 DMLS1180
c c	31 32	ERROR RETURN IN CASE OF ZERO DIVISOR IER=1 RETURN	DMLS1190 DMLS1200 DMLS1210 DMLS1220 DMLS1220
Ċ	33	ERROR RETURN IN CASE OF ILLEGAL DIMENSION IER=-1 RETURN END	DMLS1230 DMLS1240 DMLS1250 DMLS1260 DMLS1270

, SAVE1 , SAVE2 , SAVE3 , SAVE4 , SUBROUTINE STRUT ( IST STRT 10 SAVE5 , SAVE6 , SAVE7 , SAVE8 , ISAVE1, STRT 20 1 ISAVE2, ISAVE3, STRP , IPOSC , IPOST , STRT 30 2 , SRULT , STRT 40 3 IRET , SCMAX , STMAX , CDC C , SRT , SRC , PFT , PFC , FORCE , AXSTIF , IFIRST, ITER SRULC , SRT , PFC , STRT 50 4 5 STR STRT 60 , , ISTOP2 ) 6 CDT STRT 70 DIMENSION CDC(1), CDT(1), SRT(1), SRC(1), PFT(1), PFC(1) STRT 80 STRT 90 THIS SUBROUTINE DETERMINES THE FORCE AND AXIAL STIFFNESS OF A STRT 100 TYPICAL LANDING GEAR STRUT STRT 110 STRT 120 **STRT 130** INITIALIZE SUBROUTINE STRT 140 STRT 150 ISTOP2 = 0STRT 155 STRT 160 ABSTR=ABS(STR) **STRT 170** SPNGC=SAVE1 SPNGT=SAVE2 STRT 180 STRT 190 DISTC=SAVE3 DISTT=SAVF4 STRT 20.0 **STRT 210** PFORC=SAVE5 PFORT=SAVE6 STRT 220 STRT 230 FOREVC=SAVE7 FOREVT=SAVE8 STRT 240 STRT INDULC=ISAVE1 250 STRT 260 INDULT=ISAVE2 IJKC=1 STRT 270 STRT 280 IJKT=1 STRT 290 STRT 300 IPREV=ISAVE3 С STRT 310 С DETERMINE STROKING DIRECTION С STRT 320 IF(STR.LT.0.0) GO TO 500 STRT 330 STRT 340 IF(STR.GT.0.0)GO TO 400 C C STRT 350 STRT 360 * * * * * С ZERO STRUT STROKE STET 370 STRT 380 C C * * * * * STRT 390 STRT 400 FORCE=0.0 STRT 410 GO TO 2002 С STRT 420 С * * * * * STRT 430 С STRUT IN TENSION STRT 440 С STRT 450 ¥ × × * * С STRT 460 STRT 470 400 CONTINUE STRT 480 IF(ABSTR.GT.STMAX.AND.ITER.EQ.1) GO TO 5001 STRT 490 С WAS PREVIOUS STEP ON COMPRESSION SIDE STRT 500 С STRT 510 С IF(IPREV.EQ.0)GO TO 402 STRT 520 IF(IPREV.NE.-1)GO TO 402 STRT 530 STRT 540 IF(PFORC.EQ.0.0)GO TO 402

۵

С C C С C С

С

		IPRFV=+1	стрт	550
		IF(IRET-EQ-1) GO TO 651	STRT	560
		GO TO 626	CTDT	570
	402	CONTINUE	CTDT	570
	402-		SIKI	500
		IF(ABSTR GT DISTING TO 401	SIRI	590
			SIRI	600
			STRT	610
	1	G0 10 2001	STRT	620
	401	CONTINUE	STRT	630
		IF(STR•LT•STRP)GO TO 425	STRT	640
С			STRT	650
С		STRUT LOADING	STRT	660
С			STRT	670
		IF(SPNGT.EQ.0.0)G0 TO 405	STRT	680
		FORCE=SPNGT*(ABSTR-DISTT)	STRT	690
			STDT	700
			CTDT	710
			SIRI	719
			SIRI	720
			SIRI	730
		$I = \{1, 1, 1, 1\} = \{0, 1\} = \{0, 1, 0\} = \{1, 2, 3\}$	STRT	740
		DISTI=CDT(IJKT-I)-PFT(IJKT-I)/SRT(IJKT)	STRT	750
		GO TO 408	STRT	760
	407	DISTT=0•0	STRT	770
	408	CONTINUE	STRT	780
		IF (FORCE+LT+PFT(IJKT))GO TO 405	STRT	790
		FORCE=PFT(IJKT)	STRT	800
		IF(ABSTR+LT+CDT(IJKT))GO TO 2001	STRT	810
	405	CONTINUE	STRT	820
	406	IF(ABSTR.LT.CDT(IJKT))GO TO 410	STRT	830
		DISTT=CDT(IJKT)-PFT(IJKT)/SRT(IJKT+1)	STRT	840
		IJKT=IJKT+1	STRT	850
		60 TO 406	STRT	860
	410	CONTINUE	STRT	870
		EORCE = SRT(I,  KT) * (ARSTR-DISTT)	CTOT	880
				000
		GOTO 2201	51KI 6T7T	070
~			2171	900
0			SIRI	910
ĉ		STRUT UNEUADING	SIRI	920
C			STRI	630
	425	CONTINUE	STRT	940
		IF(IRET+EQ+1)GC TO 450	STRT	950
		IF(INDULT•NE•0)GC TO 475	STRT	960
	426	SPNGT=SRULT	STRT	970
		IF(PFORT•NE•PFT(IPOST))SPNGT=SRT(IPOST)	STRT	975
		FOREVT=PFORT	STRT	980
		GO TO 470	STRT	990
	450	CONTINUE	STRT	1000
		IF(INDULT•NE•C)GO TO 475	STRT	1010
	451	CONTINUE	STRT	1020
		FOREVT=PFORT	STRT	1030
		IF(PEORT.FQ.PET(IPOST)) 60 TO 465	STRT	1040
		SPNGI=SRI(IPOST)	STRT	1050
		GO TO 470	STRT	1060
	465	CONTINUE	STOT	
	+57	SPNGT=SPIII T	CTDT	1080
		IF/IDOST_GO 5) GO TO 470		1000
			3151	エリプロー

ſ	470 475	<pre>SPNGT=SRT(IPOST+1 ) CONTINUE INDULT=1 DISTT=ABS(STRP)-PFORT/SPNGT IF(IPREV.FQ1)GO TO 602 CONTINUE FORCE=SPNGT*(AESTR-DISTT) IF(ABSTR.LT.DISTT)FORCE=0.0 GO TO 2001</pre>		STRT1100 STRT1110 STRT1120 STRT1130 STRT1140 STRT1140 STRT1150 STRT1160 STRT1180
		* * * * * STRUT IN COMPRESSION * * * * *		STRT1200 STRT1210 STRT1220 STRT1220 STRT1230
C	500	CONTINUE IF(ABSTR.GT.SCMAX.AND.ITER.EQ.1) GO TO 500	00	STRT1240 STRT1250 STRT1260
C C		WAS PREVIOUS STEP ON TENSION SIDE		STRT1270 STRT1280
		IF(IPREV.EQ.0)GO TO 602 IF(IPREV.NE.+1)GO TO 602 IF(PFCRT.EQ.0.0)GO TO 602 IPREV=-1 IF(IRET.EQ.+1) GO TO 451		STRT1290 STRT1300 STRT1310 STRT1320 STRT1330
	602	GO TO 426 CONTINUE IPREV=-1 IF(AbSTR.GT.DISTC)GO TO 600 FORCE=0.0		STRT1340 STRT1350 STRT1360 STRT1370 STRT1380
	600	GO TO 2000 CONTINUE IF(ITER•NE•1•AND•PFORC•NE•PFC(IPOSC	)) GO TO 699	STRT1390 STRT1400 STRT1410
c	699	IF(STR.GT.STRP)GO TO 625 CONTINUE		STRT1420 STRT1430 STRT1440
Č C		STRUT LOADING		STRT1450 STRT1460
		IF(SPNGC.EQ.0.0)GO TO 605 FORCE=SPNGC*(ABSTR-DISTC) IF(FORCE.LT.FOREVC)GO TO 2000 FORFVC=0.0 INDULC=0		STRT1470 STRT1480 STRT1490 STRT1500 STRT1510
		SPNGC=0.0 IF(IJKC.EQ.1)GO TO 607 DISTC=CDC(IJKC-1)-PFC(IJKC-1)/SRC(IJKC) GO TO 608		STRT1520 STRT1530 STRT1540 STRT1550
	607 608	DISTC=0.0 CONTINUE IF(FORCE.LT.PFC(IJKC))G0 TO 605 FOPCE=PFC(IJKC) IF(A5STR.LT.CDC(IJKC))G0 TO 2000		STRT1560 STRT1570 STRT1580 STRT1590 STRT1600
	605 606	CONTINUE IF(ABSTR•LT•CDC(IJKC))GO TO 610 DISTC=CDC(IJKC)-PFC(IJKC)/SRC(IJKC+1) IJKC=IJKC+1 GO TO 606		STRT1610 STRT1620 STRT1630 STRT1640 STRT1650

6	510	CONTINUE	STRT1660
			STRI1670
		$\frac{1}{1} \left( \frac{1}{1} \left( \frac{1}{1} \right) - \frac{1}{1} \right) = \frac{1}{1} \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) = \frac{1}{1} \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) = \frac{1}{1} \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) = \frac{1}{1} \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) = \frac{1}{1} \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) = \frac{1}{1} \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) = \frac{1}{1} \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1} \right) \left( \frac{1}{1}$	SIR11680
c		60 10 2000	STRT1690
Ċ			SIKI1700
ĉ		STRUT UNEUADING	
<u>ر</u>	: 25	CONTINUE	STRT1720
C	121	LEVIENT CONTRACTOR (CONTRACTOR CONTRACTOR CONTRACT	51611750
			STRT1740
4	226		STRT1760
c	520	JEANC-SNOLC	STRT1765
		FOREVC=DECC	STRT1770
			STRT1780
4	550		STRT1790
,	000		STRTIADO
4	451		STRT1810
,	101		STRT1820
			STRT1830
		SPNGC=SRC(IPOSC)	STRT1840
		G0 10 670	STRT1850
f	565	CONTINUE	STRT1860
		SPNGC=SRULC	STRT187C
		IF(IPOSC+EQ+5) GO TO 670	STRT1880
		SPNGC=SRC(IPOSC+1))	STRT1890
ŧ	570	CONTINUE	STRT1900
		INDULC=1	STRT1910
		DISTC=ABS(STRP)-PF6RC/SPNGC	STRT1920
		$IF(IPREV \in Q + 1)GO TO 402$	STRT1930
	675	CONTINUE	STRT1940
		FORCE=SPNGC*(ABSTR-DISTC)	STRT1950
		IF(AESTR.LT.DISTC)FORCE=0.0	STRT1960
С			STRT1970
Ċ		UPDATE INDICATORS	STRT1980
С			STRT1990
2	000	CONTINUE	STRT2000
		PFORC=FORCE	STRT2010
		GO TO 2005	STRT2020
2	001	CONTINUE	STRT2030
		PFORT=FORCE	STRT2040
		GO TO 2005	STRT2050
2	002	CONTINUE	STRT2060
		PFORC=0.0	STRT2070
		PFORT=0•0	STRT2080
2	005	CONTINUE	STRT2090
		IF(ITER•NE•1) GO TO 60	STRT2100
		IPOSC=IJKC	STRT2110
		IPOST=IJKT	STRT2120
		SAVE1=SPNGC	STRT2130
		SAVE2=SPNGT	STRT2140
		SAVE3=DISTC	STRT2150
		SAVE4=DISTT	STRT2160
		SAVE5=PFORC	STRT2170
		SAVE6=PFORT	STRT2180
		SAVE7=FOREVC	STRT2190
		SAVE8=FOREVT	STRT2200

•

	60 100	ISAVE1=INDULC ISAVE2=INDULT STRP=STR ISAVE3=IPREV CONTINUE IF(STR-EQ.0.)GO TO 100 FORCE = (STR/ABSTR)*FORCE GO TO 101 FORCE=0. CONTINUE	STRT2210 STRT2220 STRT2230 STRT2240 STRT2250 STRT2260 STRT2260 STRT2270 STRT2280 STRT2290 STRT2290 STRT2300
C C	1.71	CALCULATE AXIAL STIFFNESS	STRT2302 STRT2304
č			STRT2306
		IF(STR.EQ.0.) GO TO 10	STRT2310
		IF(STR.GT.0.) GO TO 15	STRT2320
		IF(FORCE-EQ.0.)GO TO 12	STRIZ330
		IF(ABS(FORCE) • EQ • PFC(IJKC)) GO TO 12	STR12340
		IF (SPNGC-EQ+0+)GC TO 14	STR12520
		AXSTIF=SPNGC	STR12300
	• •		STRT2380
	14	AX511F=5RC(IJRC)	STRT2390
	15	G0 10 50 TE(E0PCE-E0-0-160 TO 12	STRT2400
	11	I = (I = 0, C = C = 0, P = I (L, K = 1)) GO TO 12	STRT2410
		F(SPNGT + EQ + 0 +) GQ TQ 13	STRT2420
		AXSTIF=SPNGT	STRT2430
		GO TO 50	STRT2440
	13	AXSTIF=SRT(IJKT)	STRT2450
		GO TO 50	STRT2460
	12	AXSTIF=0.	STRT2470
		GO TO 50	STRT2480
	10	IF(IFIRST+LT+0) GO TO 20	STRT2490
		AXSTIF=SRT(1)	SIR12500
		GO TO 50	STR12510
	20	AXSTIF=SRC(1)	STRT2520
	50	CONTINUE	STRT2540
~		RETURN	STRT2550
Ć		STRUT ROTTOMED OUT ON COMPRESSION SIDE	STRT2560
2		STRUT BUTTOMED OUT ON COMPACISION STOL	STRT2570
C	5000	CONTINUE	STRT2580
	20000		STRT2590
		GO TO 5002	STRT2600
С			STRT2610
Ċ		STRUT BOTTOMED OUT ON TENSION SIDE	STRT2620
С			STRT2630
	5001	CONTINUE	STR12640
		WRITE(6,9001)IST	STR12620
	5002	CONTINUE	51312000
		1510P2 = -1	STRT2680
~		KEIUKN	STRT 2690
C		FORMAT STATEMENTS	STRT2700
c c			STRT2710
C	<b>000</b> 0	FORMAT(///.10X.*STRUT*.13.* BOTTOMED OUT ON COMPRESSION SIDE*)	STRT2720
	9001	FORMAT(///,10X,*STRUT*,I3,* BOTTOMED OUT ON TENSION SIDE*)	STRT2730

•

STRT2740 END

OVERLAY ( GEARRT,5,1)

PPAG	PAM INDUT							INPT	10
	NSION SAVIS	)						INPT	20
DIME	NSTON /							INPT	30
COMM	INSIGN ,	/						INPT	40
1	LGRTYD. IDIA		TER. NOO	INT. NDEEL	D . NELE	M . NUMB		INDT	50
2	NSTEP . NITE		ENG	ARS. IEPA	TN. NEPC	RI FPCRI	D(3).	INPT	60
â	IGLDIR, NMA	TL , IST	• IMA	TL(4)	• MNO	• FPCRS	T(3),	INPT	70
4	INDPL(4)	, ELA	SBN(4)	, ELAS	AX(4)	,		INPT	80
5	EINERT(4)	, ARE	A(4)	, IRET	(4)	,		INPT	90
6	IFIRST(4)	, SCM	AX (4)	, STMA	X(4)	,		INPT	100
7	SRULC(4) ,	SRULT(4	), CDC(	5,4), SRC	(5,4),	1		INPT	110
8 0	CT(5,4)	, CRC	(5,4)	, SRT(	5,4)	,		INPT	120
9	PFC(5,4)	, PFT	(5,4)	, NPN	, XYZP	OS(3,5)		INPT	130
9,	DC(3,3)	, PHI		, THTA		, PSI		INPT	140
э,	CGYCDL(3),	EGYCDL(	3)					INPT	150
COMM	ON / CMAIN	1						INPT	160
А	NITERS, NDI:	SPS, STL	N2 ,					INPT	170
В	IDSPDR, IDS	PD1, IDS	PD2, SPD	ISP, COEF	F , XYZD	)SP(3),		INPT	180
С	STLN(4)	, STL	NO(4)		, STRP	P(4) ,		INPT	190
D	SPNGC(4)	, SPN	GT(4)	, DIST	C(4)	,		INPT	200
E	DISTT(4)	, PFO	RC (4)	, PFOR	Τ(4)	,		INPT	210
F	FOREVC(4)	, FOR	EVT(4)	, IPOS	⊂(4)	,		INPT	220
G	IPOST(4)	, IND	ULC(4)	, INDU	L1(4)	,		INPT	230
1-1	STR(4)	• AXI	ALF(4)	, AXSI	1 - (4)	,		INPT	240
I	IPREV(4)	• XYZ	DS (3)	• SFOR	CE(30)	9		INPT	250
J	CFORCE(30)	• SUM	FM4(6)	•R24XY	2(3)	•		INPT	260
ĸ	1M(36)	, FMV			CIG(12)4	+) 9 CGVVV7/2)			270
L.	YSHRFP(4)	250859(4	) YELDA	1P(4), 250 10(4), 764	RMP(4)	EGVYY7(3)	,		290
	ISHKEQ(4)9	7 205770	79 IOHNA 512 /	G(4/9 2.30	111111111111111111	201712(3)		INPT	200
DAT	A DEGRAD 7 D A BLANK 7 10	10292119	2157					INPT	310
WRI	EF(6.40)	/						INPT	320
AD EOR	ΛΔΤ ( 37H G	FAR ANAL	YSIS DAT	·	CARD CO	DDE //		INPT	330
1	28X31HBL	ANK - 0	1010 5/11	COMMENT C	ARDS	/		INPT	340
2	28X44H	1		GEAR AND	LOAD CAP	RD	1	INPT	350
3	23X47H	2		FRICTION	CARD		1	INPT	360
4	28X51H	3		APPLIED D	ISPLACE	MENT CARD		/ INPT	370
5	28X45H	4		STRUT MAT	ERIAL CA	ARDS	1	INPT	380
6	28X49H	5		SOLUTION	PARAMETE	ER CARD	1	INPT	390
7	28X41H	6		NODAL POI	NT CARDS	5 /	/	INPT	400
8	28X38H	7		FOOTPAD C	ARD	/		INPT	410
9	28X49H	8		MATERIAL	PARAMETE	ER CARDS	/	INPT	420
А	28X44H	9		MATERIAL	CRUSH CA	ARDS	1	INPT	430
В	28X51H	10		COMPRESSI	ON CRUSH	H DISTANCE	E CARDS	/INPT	440
С	28X51H	11		TENSION C	RUSH DI	STANCE CAP	RDS	/INPT	450
D	28X51H	12		COMPRESSI	ON SPRI	NG RATE CA	ARDS	/INPT	460
E	28X51H	13		TENSION S	PRING RA	ATE CARDS	CADEC	/ INP I	410
F	28X51H	14		COMPRESSI	UN PLAS	FORCE CAR	CARUS		400
G	28X51H	15		IENSIUN P	LASIIC I	FURCE CARI	, s /		470
н	28X42H	10		UATA TERM	IINATOR (	CARD ,	/ )		500 610
							=		520
	REAU IME IN	FUI FILE	ADD PLA	ACE ON UNI	I I FOR	CAND COD	<u> </u>	TULE 1	121

378

C C

С		PROCESSING	INPT	530
		REWIND 1	INPT	540
	90	READ(5,100)(ALPHA(I),I=1,8)	INPT	550
	100	FORMAT ( 8A10)	INPT	560
		IF ( FOF, 5 ) 103, 105	INPT	570
	103	CGNTINUE	INPT	580
		WRITE(6,290)	INPT	590
	290	FORMAT(39H1END OF JOBEND OF DATA SET ON UNIT 5 )	INPT	600
		STOP	INPT	610
	105	CONTINUE	INPT	620
		WRITE(6,1111)(ALPHA(I),I=1,8)	INPT	630
		DECODE (2, 451, ALPHA(1) ) I	INPT	640
	451	FORMAT (I2)	INPT	650
		IF ( I •EQ• 0 ) GO TO 90	INPT	660
]	111	FORMAT ( 1X8A10)	INPT	670
С		CHECK TO INSURE THAT CARD CODE IS RIGHT JUSTIFIED	INPI	680
		DECODE ( 2, 106, ALPHA(1) ) A, B	INPT	690
		IF ( B •NE• BLANK ) GO TO 107	INPT	700
		WRITE(1,106) B, A	INPT	710
	106	FORMAT ( 2A1 )	INPT	720
		GO TO 108	INPT	130
	107	CONTINUE	INPT	740
		WRITE(1,110) ALPHA(1)	INPT	750
		DECODE ( 2,280, ALPHA(1)) 1	INPT	760
		IF ( I •GT• 15 ) GO TO 120		770
	280	FORMAT (IZ)	INPT	780
	110	FORMAT ( A2 )		790
	108	WRITE(1,100)(ALPHA(1),1=1,8)	INPT	800
	0	GO TO 90		810
	120			020
~		REWIND I END OF INITIAL PEAD	INPI	840
C	0	END OF INITIAL READ		250
	150	READ(1,200) ICCODE		040
	200	FORMAT ( 12 )		000
	<b>2</b> 05	IF ( EOF) I ) 1000, 205		010
~	205	CONTINUE		800
ĉ		ONLY CARDS WITH CARD CODES OF 1-15 ARE ACCEPTABLE	INPT	900
ç		ONET CARDS WITH CARD CODES OF I ID ARE ACCELLADEE	TNPT	010
C		LE ( LCCODE (T. 15.) (() TO 1000	INPT	920
		$I = \left( I \left( C \left( O \right) E_{1} \left( T \right) \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( O \right) = \left( $	INPT	930
		$P_{1} = (1, 2, 0, 0) $ ( $1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,$	INPT	940
		WEITE(4,210)(ALPHA(1),I=1,0)	INPT	950
	210	$ = \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum_{n=1}^{\infty} \sum$	INPT	960
	210	CO TO 150	INPT	970
	220		INDT	080
	220	CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE = CONTINUE	A LCCODE INPT	990
c		CEAP AND 1000 CAPD	INPT	1000
Ç	,	READ ( 1.270) LERTYD. LDIND. STINZ. PSI. THTA. PHI	INPT	1010
	T	$\begin{array}{c} R L R D T I I I I I I I I$	INPT	1020
r		ERICTION CARD	INPT	1030
C	2	PEAD(1. 260) IDSPDR. SPDISP. COFFE	INPT	1040
	4	60  TO  150	INPT	1050
C		APPLIED DISPLACEMENT CARD	INPI	1060
C	3	READ(1, 250) (XYZDSP(I), I=1.3)	INPI	1070
	~	GO TO 150	INP	1080

C	STRUT MATERIAL CARDS	INPT1090
4	RFAD(1, 260) IST, IMATL(IST)	INPT1100
	GO TO 150	INPT1110
C	SOLUTION PARAMETER CARD	INPT1120
5	RFAD(1,260) NSTEP, NITER, NMATL, IGLDIR, ENGABS, TOL	INPT1130
-	60 10 150	INPT1140
C	NODAL POINT CARDS	INPT1150
6	$RFAD(1,240) NPN \rightarrow (XYZPOS(I,NPN),I=1,3)$	INPT1160
C,	GO TO 150	INPT1170
C	EDOTPAD CARD	INPT1180
7	RFAD(1+27C) IFPATN, NFPCRL, (FPCRLD(I), I=1,3), (FPCRST(I), I=1,3)	INPT1190
		INPT1200
c	MATERIAL PARAMETER CARDS	INPT1210
C g	READ(1,270) MNO • INDPL (MNO) • ELASBN (MNO) • ELASAX (MNO) •	INPT1220
0	BINERT (MNO) AREA (MNO)	INPT1230
	GO TO 150	INPT1240
C	MATERIAL CRUSH CARDS	INPT1250
<u>`</u> a	READ(1,230) MNO , IRET (MNO), IFIRST(MNO), SCMAX(MNO),	INPT1260
,	STMAX(MNO), SRULC (MNO), SRULT(MNO)	INPT1270
	GO TO 150	INPT1280
C	COMPRESSION CRUSH DISTANCE CARDS	INPT1290
10	READ(1,240) MNO = (CDC(1,MNO),I=1,5)	INPT1300
10		INPT1310
C	TENSION CRUSH DISTANCE CARDS	INPT1320
11	PEAD(1,240) MNO (CDT(I,MNO),I=1.5)	INPT1330
ΤŢ		INPT1340
c	COMPRESSION SPRING RATE CARDS	INPT1350
12	P = A D (1 - 240) MNO = (SRC(1 - MNO) + 1 = 1 + 5)	INPT1360
12		INPT1370
c	TENSION SPRING RATE CARDS	INPT1380
12	$PEAD(1-240) MNO = (SRT(1-MNO) \cdot I=1 \cdot 5)$	INPT1390
15		INPT1400
C	COMPRESSION PLASTIC FORCE CARDS	INPT1410
1.	DEAD(1.240) MNO (PEC(1.MNO) (I=1.5)	INPT1420
14		INPT1430
c	TENSION PLASTIC FORCE CARDS	INPT1440
ر ا د	PE(1,2/6) PEAD (PET(1,MNO), [=1,5])	INPT1450
15	READ(1)(240) MAC = F(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(1)(	INPT1460
100		INPT1470
C 1000	CONVERT FUER ANGLES TO RADIANS	INPT1480
C		INPT1490
	THIAS THIA/ DEGRAD	INPT1500
		INPT1510
C	FST - FST DISPLACEMENT STEP NUMBER	INPT1520
C		INPT1530
	$\frac{1}{10} \frac{1}{10} \frac{1}{10} = 0$	INPT1540
C		INPT1550
Ċ	SET ERICTION PLANE INDICATOR IDSPD1 AND IDSPD2	INPT1560
	SET FRICHOR FEARL INDICATOR INDICITION FEARL	INPT1570
C	$10001 \simeq 0$	INPT1580
	$\frac{1}{1}$	INPT1590
	UU 400 I = 1,2 Im / I FO, IOSPDP ) GO TO 400	INPT1600
	16 V I •LQ• 10360K / 00 10 400 Trodda = I	INPT1610
	$\frac{1}{100} = 1$	INPT1620
	IF ( 100F01 +NL+ 0 / 00 /0 +00	INPT1630
1.0	TUSPUL - I O CONTINUE	INPT1640
40		

С		SET NORMAL DISPLACEMENT	INPT1650
		XYZDSP(IDSPDR) = SPDISP	INPT1660
	410	CONTINUE	INPT1670
С			INPT1680
С		SET APPLIED DISPLACEMENT INCREMENTS	INPT1690
С			INPT1700
		DO 420 I = 1, 3	INPT1710
	420	XYZDS(I) = XYZDSP(I) / NSTEP	INPT1720
С		INITIALIZE THE ENERGY SUBROUTINE	INPT1730
		CALL ENERGI	INPT1740
		WRITE(6,320)	INPT1750
	320	FORMAT ( 1H1, 45X 36HINITIAL CONDITIONS FOR GEAR ANALYSIS ///)	INPT1760
С			INPT1770
С		GET THE LANDER TO SURFACE COORDINATE TRANSFORMATION MATRIX, DC	INPT1780
С			INPT1790
		CALL GEOM	INPT1800
C			INPTIBLO
С		CONVERT NODE COORDINATES TO SURFACE COORDINATES	INPT1820
С			INPT1830
		NPOINT= 4	INPT1840
		IF ( LGRTYP .EQ. 2 ) NPOINT≃ 5	INPT1850
		DO 700 I = 1, NPOINT	INPT1860
		bo 650 J = 1.3	INPT1870
		SAV(J) = 0.0	INPT1880
		DO 650 K = $1.3$	INPT1890
	650	SAV(J) = SAV(J) + DC(J,K)*XYZPOS(K,I)	INPT1900
		DO 700 J = 1.3	INPT1910
		IF ( ABS(SAV(J) ) •GT• 1•F-10 ) GO TO 690	INPT1920
		$XY7POS(J \cdot I) = .0$	INPT1930
			INPT1940
	600	$XYZPCS(J \cdot I) = SAV(J)$	INPT1950
	700	CONTINUE	INPT1960
		BETURN	INPT1970
	230	EORMAT ( 5X 315, 4E10.5 )	INPT1980
	240	FORMAT ( 5X 15, 10X 5E10.5)	INPT1990
	250	FORMAT ( 5X 15X 4E10.5)	LNPT2000
	260	FORMAT ( 5X 415, 5X 2E10, 5)	INPT2010
	270	FORMAT ( 5X 215. 5X 6510.5)	INPT2020
	Z I J	END CONTRACTOR OF CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRA CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR CONTRACT	INPT2020
			THE 12000

381

A
SUBROUTINE GEOM	GEOM 10
COMHON / CMAIN /	GEOM 20
1 LGRTYP, LDIND, INITER, NPOINT, NREFP, NELEM, NUMB	• GEOM 30
2 NSTEP , NITER , TOL , ENGABS, IFPATN, NEPCRL, FPCRL	D(3), GEOM 40
3 IGEDIR • NMATE • IST • IMATE(4) • MNO • FPCRS	T(3) GEOM 50
4 INDEL(4) ELASEN(4) ELASAX(4)	GEOM 60
5 $PINERT(A)$ , $APEA(A)$ , $IPET(A)$ .	GEOM 70
	GEOM PO
7 = SP(I)(4) = SP(I)(4) = OP(15.4) = SP(15.4) = SP(15	GEOM SU
$\begin{array}{c} \mathbf{P} = \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \right) \\ \mathbf{P} = \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \right) \\ \mathbf{P} = \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \right) \\ \mathbf{P} = \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \right) \\ \mathbf{P} = \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} = \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \left( \mathbf{P} \right) \\ \mathbf{P} \left( \mathbf{P} \right)$	GEOM 100
$O = DEC(S_{4}) \qquad \qquad DET(S_{4}) \qquad \qquad NDN \qquad VZDOS(2,S)$	GEOM 110
0 - DC(2-2) - DUT - THTA - DST	GEOM 120
0 . (CYCDI/2). ECYCDI/2)	
	GEOM 140
COMMONY / CMAIN /	GEOM 140
A WITERS, NUISES, STENZ , B INCRED, INCRED, INCRED, COFFE, VV7DCD/20,	GEOM 150
D IUSPUR, IUSPUI, IUSPUZ, SPUISP, CUEFF, XIZUSP(5),	
C SILN(4) , SILNO(4) , SIRP(4) ,	GEOM 170
D SPNG(4) , SPNG1(4) , DISIC(4) ,	GEOM 180
E DISTI(4) , PFORC (4) , PFORT(4) ,	GEON 190
+ FOREVC(4) + FOREVI(4) + IPOSC(4) +	GE0.4 200
G IPOST(4) , INDULC(4) , INDULT(4) ,	GEOM 210
H STR(4) , AXIALE(4) , AXSTIE(4) ,	GEOM 220
I IPREV(4) , XYZDS (3) , SFORCE(30) ,	GEOR 230
J CFORCE(30) , SUMFM4(6) ,R24XYZ(3) ,	GEOM 240
K TM(9,4) , FMVCTL(12,4) , FMVCTG(12,4) ,	GEO.4 250
L YSHRFP(4), ZSHRFP(4), YSHRMP(4), ZSHRMP(4), CGYXYZ(3)	, GEOM 260
M YSHRFQ(4), ZSHRFQ(4), YSHRMQ(4), ZSHRMQ(4), ESYXYZ(3)	GEOM 270
******	GEOM 280
CALCULATE THE DC TRAMSFORMATION MATRIX FOR CONVERTING	GEOM 290
LANDER COORDINATES TO SURFACE COORDINATES	GEOM 300
****	GEOM 310
CALCULATE THE TRIG. FUNCTIONS ONLY ONCE IN GEOM	GEOM 320
*****	GEOM 330
COSPHI = COS( PHI )	GEOM 340
SINPHI = SIN( PHI )	GEOM 350
COSTHA = COS(THTA)	GEON 360
SINTHA = SIN( THTA )	GEOM 370
CCSPSI = COS( PSI )	GEOM 380
SINPSI = SIN( PSI )	GEOM 390
DC(1,1) = COSTHA * COSPSI	GEOM 400
A = SINTHA * COSPSI	GEOM 410
B = COSPHI * SINPSI	GEOM 420
D((2,1) = SINPHI * A - B	GEOM 430
C = SINPHI * SINPSI	GEOM 440
DC(3,1) = COSPHI * A + C	GEOM 450
DC(1,2) = COSTHA * SINPSI	GEOM 460
DC(2,2) = SINTHA * C + COSPHI*COSPSI	GEOM 470
DC(3,2) = SINTHA * P - SINPHI*COSPSI	GEOM 480
D((1,3) = -SINTHA	GEOM 490
DC(2,3) = SINPHI*COSTHA	GEOM 500
DC(3,3) = COSPHI*COSTHA	GEOM 510
*****	GEOM 520
RETURN	GEOM 530
END	GEOM 540

С

OVERLAY ( GEARRT, 5,2)

C C

С

C C C

	<pre>PROGRAM INVTRP DIMENSION TSTIFM(6,6), CDLXYZ (6) COMMON / CMAIN / L LGRTYP, LDIND , INITER, NPOINT, NREFP , NELEM , NUMB , NSTEP , NITER , TOL , ENGABS, IFPATN, NFPCRL, FPCRLD(3), J IGLDIR, NMATL , IST , IMATL(4) , MNO , FPCRST(3), INDPL(4) , ELASBN(4) , ELASAX(4) , SINERT(4) , AREA(4) , IRET(4) , SRULC(4) , SRULT(4), CDC(5,4), SRC(5,4) , CCT(5,4) , CPC(5,4) , SRT(5,4) , PFC(5,4) , PFI(5,4) , NPN , XYZPOS(3,5) , DC(3,3) , PHI , THTA , PSI , CGYCDL(3), EGYCDL(3) COMMON / CMAIN / A NITERS, NDISPS, STLN2 , J IDSPDR, IDSPD1, IDSPD2, SPDISP, COEFF , XYZDSP(3), CSTLN(4) , STLNO(4) , STRP(4) , STRP(4) , SPNGT(4) , DISTC(4) , F FOREVC(4) , FOREVT(4) , IPOSC(4) , STRP(4) , STRP(4) , STRP(4) , AXIALF(4) , IPOSC(4) , STRP(4) , STRP(4) , STR(4) , AXIALF(4) , AXIFF(4) , STR(4) , AXIALF(4) , SPORT(4) , STRP(4) , XYZDS (3) , SFORCE(30) , J CFCRCE(30) , SUMFM4(6) , R24XYZ(3) , K TM(36) , FMVCTL(12,4) , FMVCTG(12,4) , J SHRFP(4), ZSHRFP(4), YSHRMQ(4), ZSHRMQ(4), EGYXYZ(3), M YSHRFP(4), ZSHRFP(4), YSHRMQ(4), ZSHRMQ(4), EGYXYZ(3), M YSHRFQ(4), ZSHRFQ(4), YSHRMQ(4), ZSHRMQ(4), EGYXYZ(3), M YSHRFQ(4), ZSHRFQ(1), YYZPOS(1,2), TM(10), STLNO(1) CALL TRANSM(XYZPOS(1,2), XYZPOS(1,4), XYZPOS(1,2), TM(10), STLNO(2) CALL TRANSM(XYZPOS(1,2), XYZPOS(1,4), XYZPOS(1,2), TM(10), STLNO(2) CALL TRANSM(XYZPOS(1,2), XYZPOS(1,4), XYZPOS(1,2), TM(10), STLNO(2) CALL TRANSM(XY</pre>	INVT INVT INVT INVT INVT INVT INVT INVT	$\begin{array}{c} 10\\ 20\\ 300\\ 400\\ 500\\ 600\\ 890\\ 100\\ 1200\\ 100\\ 1200\\ 100\\ 1200\\ 222\\ 220\\ 200\\ 300\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 330\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000\\ 000$
110	CALL TRANSM(XYZPOS(1,2),XYZPOS(1,4),XYZPOS(1,2),TM(17),STLNO(1) CALL TRANSM(XYZPOS(1,2),XYZPOS(1,4),XYZPOS(1,3),TM(10),STLNO(2) CALL TRANSM(XYZPOS(1,3),XYZPOS(1,4),XYZPOS(1,2),TM(19),STLNO(3) SAVE MAIN STRUT LENGTH STLNS = STLNO(2) DO 110 I = 1, 3 CDLXYZ(I) = XYZDS(I)	) INVT ) INVT INVT INVT INVT INVT	390 400 410 420 430 430
120 130	DO 120 I=1,3 STLN(I)=STLNO(I) CALL OUTPT1 CONTINUE UPDATE FOOTPAD JOINT COORDINATES	INVT INVT INVT INVT INVT INVT	450 460 470 480 490 500
	DO 140 I = 1, 3	INVT	520

383

a

```
140 XYZPOS(I,NPOINT) = XYZPOS(I,NPOINT) + CDLXYZ(I)
                                                                          INVT 530
                                                                          INVT 540
      IF ( NITERS .LE. 1 ) GO TO 145
      RMOVE = SQRT((BASEPX-XYZPOS(1,4))**2 + (BASEPY-XYZPOS(2,4))**2
                                                                          INVT 550
                 + (BASEPZ-XYZPOS(3,4))**2 )
                                                                          INVT 560
     1
      IF ( RMOVE .LT. RADRAG ) GO TO 145
                                                                          INVT 570
      WRITE(6,144)
                                                                          INVT 580
  144 FORMAT (43H1ERROR *** NO SOLUTION FOUND AT THIS LEVEL. )
                                                                          INVT 590
                                                                          INVT 600
      CALL OUTPT1
      RETURN
                                                                          INVT 610
  145 CONTINUE
                                                                          INVT 620
                                                                          INVT 630
С
          CALCULATE CURRENT STRUT LENGTH AND GET TRANS. M.
      CALL TRANSM(XYZPOS(1,1),XYZPOS(1,4),XYZPOS(1,2),TM( 1),STLN (1) ) INVT 640
      CALL TRANSM(XYZPOS(1,2),XYZPOS(1,4),XYZPOS(1,3),TM(10),STLN (2) ) INVT 650
      CALL TRANSM(XYZPOS(1,3),XYZPOS(1,4),XYZPOS(1,2),TM(19),STLN (3) ) INVT 660
                                                                          INVT 670
С
          CALCULATE STRUT FORCES AND AXIAL STIFFNESSES
                                                                          INVT 680
С
                                                                          INVT 690
С
                                                                          INVT 700
      MITERS = MITERS + 1
      ISTOP1 = 0
                                                                          INVT 705
                                                                          INVT 710
      DO 170 IST=1, NELEM
          COMPUTE STROKE
                                                                          INVT
                                                                               720
С
                                                                          INVT 730
      STR(IST) = STLN(IST) - STLNO(IST)
                                                                          INVT 740
C
         GET THE MATERIAL NO.
                                                                          INVT 750
      MNO = IMATL(IST)
      IF ( INDPL(MNO) •EQ• 2 ) GO TO 160
                                                                          INVT
                                                                               760
          COMPUTE AXIAL FORCE AND STIFFNESS - ELASTIC MATERIAL IN STRUT INVT 770
C
                                                                          INVT 780
      AXSTIF(IST) = AREA(MNO)*ELASAX(MNO)/STLN(IST)
      AXIALF(IST) = STR(IST)*AXSTIF(IST )
                                                                          INVT 790
                                                                          INVT 800
      GO TO 165
          COMPUTE AXIAL FORCE AND STIFFNESS - PLASTIC MATERIAL IN STRUT INVT 810
C
                                                                          INVT 820
  160 CONTINUE
                                                                          INVT 830
      CALL STRUT ( IST, SPNGC(IST) , SPNGT(IST) , DISTC(IST) ,
                         DISTT(IST) , PFORC(IST) , PFORT(IST) ,
                                                                          INVT 840
     1
                          FOREVC(IST), FOREVT(IST), INDULC(IST),
                                                                          INVT 850
     2
                          INDULT(IST), IPREV(IST) , STRP(IST) ,
                                                                          INVT 860
     3
                          IPOSC(IST) , IPOST(IST) , IRET(MNC)
                                                                          INVT 870
                                                               ,
     4
                                                                          INVT 880
                          SCMAX(MNO) , STMAX(MNO) , CDC(1,MNO) ,
     5
                                                                          INVT 890
                          SRULT(MNO) , SRULC(MNO) , SRT(1,MNC) ,
     6
                          SRC(1,MNO) , PFT(1,MNO) , PFC(1,MNO) ,
                                                                          INVT 900
     7
                          STR(IST) , AXIALF(IST), AXSTIF(IST),
                                                                          INVT 910
     8
                          IFIRST(MNO), NITERS , CDT(1,MNO),
                                                                          INVT 920
     G
                                                                          INVT 925
                         ISTOP2 )
     Δ
                                                                          INVT 930
      IF ( ISTOP2 .LT. 0 ) ISTOP1 = 1
                                                                          INVT 940
  165 CONTINUE
          SET LOCAL FORCE VECTOR WITH AXIAL FORCES
                                                                          INVT 950
С
      FMVCTL(1,IST) = -AXIALF(IST)
                                                                          INVT 260
      FMVCTL(4,IST) = AXIALF(IST)
                                                                          INVT 970
          CONVERT LOCAL FORCE VECTOR TO GLOBAL FORCE VECTOR
                                                                          INVT 980
С
      CALL TRALMG( FUVCTG(1,IST), FMVCTL(1,IST), TM( 9*IST+8), 2 )
                                                                          INVT 990
  170 CONTINUE
                                                                          INVT1000
                                                                          INVT1005
      IF ( ISTOP1 .EQ. 1 ) RETURN
C
                                                                          INVT1010
          SUM FORCES AT NODE POINT FOUR
                                                                          INVT1020
С
                                                                          INVT1030
С
                                                                          INVT1040
      DO 180 I = 1, 3
      SUMFM4(I) = 0.0
                                                                          INVT1050
```

	180	K = I + 3 DO 180 J = 1, NELEM SUMFM4(I) = SUMFM4(I) + FMVCTG(K,J)	INVT1060 INVT1070 INVT1080
C C C C		IF THREE FOOTPAD JOINT DISPLACEMENTS ARE APPLIED, GO TO OUTPUT AND ENERGY SECTION	INVT1090 INVT1100 INVT1105
c c c		IF ( LDIND •EQ• 2 ) GO TO 400 SKIP CONVERGENCE CHECK ON FIRST PASS CHECK FOR CONVERGENCE	INVIIII0 INVT1120 INVT1130 INVT1140
C		END OF LOOP IF ( IFIN •EQ• 1 ) GO TO 400 IF (NITERS•GT• NITER ) GO TO 400	INVT1150 INVT1160 INVT1170
C		COMPUTE FRICTION FORCE FRICTF = COEFF * SUMFM4( IDSPDR ) ABSFIR = SQRT( SUMFM4(IDSPD1)**2 + SUMFM4(IDSPD2)**2 ) IF ( NITERS_NE_ NITERC ) GO TO 190	INVT1180 INVT1190 INVT1200
с		IF ( ABSFIR •LE• AES( FRICTF ) ) GO TO 400 SET NORMAL FOOTPAD JOINT DISPLACEMENT INCREMENT TO ZERO CDLXYZ(IDSPDR ) = 0•0	INVT1220 INVT1230 INVT1240
C C		PICK UP THE COS OF THE ANGLE (ALF) BETWEEN THE MAIN STPUT AND THE NORMAL COSALE = ABS(TM( ICOS) )	INVT1250 INVT1260 INVT1270
	501	IF ( COSALF .LT. 1. ) GO TO 500 WRITE(6,501) FORMAT ( 100H1 ERROR *** DATA CASE TERMINATED, ANGLE BETWEEN MAIN STRUT AND THE NORMAL TO THE LANDING CUREACE IS	INVT1280 INVT1290 INVT1300
	502	WRITE(6,502) FORMAT ( 60X 14H+EQ+ 0 DEGREES ) CALL OUTPT1	INVT1310 INVT1320 INVT1330 INVT1340
	500	RETURN CONTINUE IF ( COSALF .GT. 0.) GO TO 505	INVT1350 INVT1360 INVT1370
	503	WRITE(6,501) WRITE(6,503 ) FORMAT ( 60X 15H+EQ+ 90 DEGREES )	INVT1380 INVT1390 INVT1400
	505	CALL OUTPT1 RETURN CONTINUE TANALE = SUBT( 1 = COSALEXX2 ) ( COSALE	INVT1410 INVT1420 INVT1430
С		<pre>TANALF = SURT( 1. = COSALF*2 ) / COSALF RADLIM = AMAX1( XYZDS(IDSPDR) / TANALF, XYZDS(IDSPDR) * TAMALF ) RADLIM = AMIN1(RADLIM,SQRT(STLNS**2-(STLNS-XYZDS(IDSPDR))**2)) SET OUTER RADIUS LIMIT</pre>	INVI1440 INVI1450 INVI1460 INVI1470
c		RADRAG = RADLIM * 1.05 RDSTEP = RADRAG / 100. DETERMINE THE CENTER OF SOLUTION CIRCLE	INVT1490 INVT1490 INVT1500
		BASEPX = XYZPOS(1,4) BASEPY = XYZPOS(2,4) BASEPZ = XYZPOS(3,4)	INVT1510 INVT1520 INVT1530
C C	190 ,	CONTINUE RDSTEP IS POSITIVE AND DELTF2 NEGATIVE UNTIL SOLUTION POINT IS PASSED	INVT1540 INVT1550 INVT1555
С		TEST FOR CONVERGENCE IF ( ABS(DELTF2) •LE• •0005*AES(FRICTF) ) GO TO 400 IF ( RDSTEP •LT• 0•0 ) GO TO 235	INVT1580 INVT1580 INVT1590

•

		IF ( DELTF2 .LT. 0.0 ) 60 TO 230	INVT1600
С		CALCULATE BACK STEP TO SOLUTION	INVT1610
		RDSTPS = RDSTEP	INVT1615
		RDSTEP = - DELTF2*RDSTEP / ( ABS(DELTF1 ) + DELTF2 )	INVT1620
С		SAVE DIFFERENCE IN RUSTEP	INVT1622
		RDSTPS = - (RDSTEP + RDSTPS)*•5	INVT1624
		GO TO 230	INVT1630
	235	CONTINUE	INVT1640
С		ONCE SOLUTION POINT IS PASSED, CONTINUE TAKING PACK SIFPS	INVT1650
С		UNTIL DELTF2 AGAIN BECOMES NEGATIVE AFTER WHICH ONE	INVT1660
С		MORE FORWARD STEP IS TAKEN	INVT1670
		RDSTFP = RDSTPS	INV11630
		IF ( DELTF2 .GT. 0.0 ) GO TO 230	INV11690
		RDSTEP = DELTE2 * RDSTEP / ( ABS(DELTE2 ) + DELTET )	
		IFIN=1	
	230		INV11720
		COSI - SUMPRATIDEDEDI / ABETE	14071750
	240	COS2 = SUMEM4(IDSPD2) / ABSEIN	INVT1750
1	240	COMPLETE NEXT SUIDING DISPLACEMENT INCREMENTS	INVT1760
C		CONVERSE COST $\neq$ DOL ACTOR ACTOR ACTOR	INVT1770
		CD[XYZ(I)SDD] = COST * DDSIED	INVT1780
		$V_{2} = V_{2}	INVT1720
		$X_1 Z_2 Z_3 = X_1 Z_2 Z_3 Z_1 Z_2 Z_2 Z_2 Z_2 Z_2 Z_2 Z_2 Z_2 Z_2 Z_2$	INVT1900
r		SAVE FRICTION FORCE UNPAI ANCE	INVT1810
C		DELTEL = DELTE2	INVT1820
			INVT1830
	400	CONTINUE	INVT184C
С			INVT1850
Ĉ		OUTPUT AND ENERGY SECTION	INVT1860
С			INVT1870
		CALL ENERGY( SUMFM4)	INVT1830
		CALL OUTPUT	INVT1890
		IF ( NITERS •LE• NITER ) GO TO 405	INVT1900
		WRITE(6+403)	INVT1910
	403	FORMAT (34H1MAXIMUM ITERATION NUMBER EXCEEDED )	INVT1220
		RETURN	INVT1930
	405	CONTINUE	INVI1940
		NITEPS =0	10011955
		IF ( NDISPS •GE• NSTEP ) RETURN	INVI1960
		IF ( IFPATN •EQ• 1 ) GO TO 425	INVT1970
		IF ( LDIND •EQ• 2 ) GO TO 130	INV11900
	425		INVT2000
		IF (SUMEM4(IDSPDR) •GE• 0•0 ) GU TU 428	INVI2000
			INVT2010
	425	PERMAT (ZIHINEGATIVE NORMAL EDA. )	INVT2020
	120		INVT2040
	428		LMVT2050
		IF V LUTRE CLOVE Z 7 GO TO 100 MITERS = 1	INVT2060
		XY70S(-1)SPD1 = 0.0	INVT2070
		XYZDS(IDSPD2) = 0.0	INVT2080
		$DO 430 I = 1 \cdot 3$	INVT2090
	430	O(CDLXYZ(I) = XYZDS(I)	INVT2100
C		SAVE MAIN STRUT LENGTH	INVT2110
-		STLNS = STLN(2)	INVT2120

```
NITERC = 2
                                                                          INVT2130
      IFIN = 0
                                                                          INVT2140
C
C
C
                                                                          INVT2150
          SET STRUT SAVE VARIABLES FOR THIS POSITION
                                                                          INVT2160
                                                                          INVT2170
      ISTOP1 = 0
                                                                          INVT2175
      DO 440 IST = 1, 3
                                                                          INVT2180
      MNO = IMATL(IST )
                                                                          INVT2190
      IF ( INDPL(MNO) •NE• 2 ) GO TO 440
                                                                          INVT2200
      CALL STRUT ( IST, SPNGC(IST) , SPNGT(IST) , DISTC(IST) ,
                                                                          INVT2210
                          DISTT(IST) , PFORC(IST) , PFORT(IST) ,
                                                                          INVT2220
     1
                          FOREVC(IST), FOREVT(IST), INDULC(IST),
                                                                          INVT2230
     2
                          INDULT(IST), IPREV(IST), STRP(IST)
                                                                          INVT2240
     3
                          IPOSC(IST) + IPOST(IST) + IRET(MNG) +
                                                                          INVT2250
     4
                          SCMAX(MNO) , STMAX(MNO) , CDC(1,MNO) ,
     5
                                                                          INVT2260
                          SRULT(MNO) , SRULC(MNO) , SRT(1,MNO) ,
                                                                          INVT2270
     6
                          SRC(1,MNO) , PFT(1,MNO) , PFC(1,MNO) ,
                                                                          INVT2280
     7
                          STR(IST) , AXIALF(IST), AXSTIF(IST),
                                                                          INVT2290
     8
                          IFIRST(MNO), NITERS
                                                                          INVT2300
                                               , CDT(1,MNO) ,
     9
                          ISTOP2 )
                                                                          INVT2305
     А
      IF ( ISTOP2 .LT. 0 ) ISTOP1 = 1
                                                                          INVT2310
  440 CONTINUE
                                                                          INVT2320
      IF ( ISTOP1 .EQ. 1 ) RETURN
                                                                          INVT2325
      GC TO 130
                                                                          INVT2330
      END
                                                                          INVT2340
```

OVERLAY ( GEARRT,5,3)

4

	PROGRAM CANTIL	CANT	10
	COMMON / CMAIN /	CANT	20
	1 LGRTYP, LDIND , INITER, NPOINT, NREFP , NELEM , NUMB ,	CANT	30
	2 NSTEP, NITER, TOL, ENGARS, IFPATN, NFPCRL, FPCRLD(3),	CANT	40
	3 IGLDIR, NMATL, IST, IMATL(4), MNO, FPCRST(3),	CANT	50
	4 INDPL(4) • FLASBN(4) • FLASAX(4)	CANT	60
	5 $BINERT(4)$ • $AREA(4)$ • $IRET(4)$ •	CANT	70
	6 IFIRST(4) + SCMAX(4) + STMAX(4) +	CANT	80
	7 SRU( $\zeta(4)$ - SRU( $\zeta(4)$ - CDC(5-4) - SRC(5-4) -	CANT	90
	(CDT(5,4)), $(CC(5,4))$ , $SBT(5,4)$ ,	CANT	100
	9 PEC(5.4) • PET(5.4) • NPN • XYZPOS(3.5)	CANT	110
		CANT	120
	9  (GYCDE(3), EGYCDE(3))	CANT	130
		CANT	140
	A NITEDC NDICCC CTIM2 -	CANT	150
	A NITERS, NUISES, STENZ , 5 IDEDDA INCOMI INCONS, CONTER AVV7050/31.	CANT	160
	B = 105F0K, 105F0I, 105F0Z, 5PUISF, COEFF, ALDSF( $3$ ),	CALL	170
	$\zeta = SIEN(4)$ , SIEN(4) , SIRP(4) ,	CANT	170
	$D = SPRG(4) \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad \qquad$	CANT	100
	E = DIST(4), PFURC (4), PFUR(4), P	CANT	190
	F FOREVC(4) , FOREVT(4) , IPOSC(4) ,	CANT	200
	G = IPOSI(4), $INDUEC(4)$ , $INDUEI(4)$ ,	CANT	210
	$\mathbf{H} = STR(4) \qquad \mathbf{\mathfrak{s}} \wedge XIALF(4) \qquad \mathbf{\mathfrak{s}} \rightarrow XSSIIF(4) \qquad \mathbf{\mathfrak{s}}$	CANI	220
	I IPREV(4) , XYZDS (3) , SEORCE(30) ,	CANT	230
	J CFORCE(30) , SUMEM4(6) ,R24XYZ(3) ,	CANT	240
	K TM(9,4) , FMVCTL(12,4) , FMVCTG(12,4) ,	CANT	250
	L YSHRFP(4), ZSHRFP(4), YSHRMP(4), ZSHRMP(4), CGYXYZ(3),	CANT	260
	M YSHRFQ(4), ZSHRFQ(4), YSHRMQ(4), ZSHRMQ(4), EGYXYZ(3)	CANT	270
	DIMENSION TSTIFM(15,15)	CANT	280
	DIMENSION CDLXYZ(15)	CANT	290
	DIMENSION STIFM6(6,6), STIFM(12,12), SARRY(15)	CANT	300
	DOUBLE PRECISION COL(12),SAVED(78),TRAC(12)	CANT	310
	NITERS = 0	CANT	320
	I1ST = 0	CANT	330
	I 2ND = 0	CANT	340
	IFIN = C	CANT	350
	NFLCH = 4	CANT	360
	ICOS = 1 + (IDSPDR-1) * 3	CANT	370
	IFRIST = 0	CANT	380
С	CALCULATE NODE POINT FOUR COORDINATES	CANT	390
	TL25 = 0.0	CANT	400
	$P_{0} = 100$ I = 1 3	CANT	410
	XY7POS(1.4) = XY7POS(1.5) - XY7POS(1.2)	CANT	420
	$T_{125} = T_{125} + XYZE_{CS}(1.44)**2$	CANT	420
	100 CONTINUE	CANT	440
	$\frac{125}{125} = SOPT(T125)$	CANT	450
		CANT	420
	$P_{2}(Y = 103 - 1 - 17) = P_{2}(Y = 1)$	CANT	470
	$\frac{1}{105} \frac{1}{3} \frac{1}{3} \frac{1}{5} $	CANT	420
~	CALCULATE INITIAL STRUCTURE AND LOCAL COODDINATE	CANT	490
č	TRANSFORMATION MATRICES	CANT	500
C	TRANSFORMATION PERIALS CALL TEANSFURYZOC(1.1.VYZOC(1.4V.YYZOC(1.2V.TM(1.1V.STMO/1)	ACANT	510
	CALL TRANSPORTATION (1) () (1) (1) (1) (1) (1) (1) (1) (1)	1 CANT	- E 2 0
	CALL TRANSMIXIZPUS(192)9X12PUS(19479R24X12( 1 7910/(192)9STLNU(2)	TCART	220

388

```
CALL TRANSM(XYZPOS(1,3),XYZPOS(1,4),XYZPOS(1,2),TM(1,3),STLNO(2) )CANT 530
CALL TRANSM(XYZPOS(1,5),XYZPOS(1,4),R24XYZ( 1 ),TM(1,4),STLNO(4) )CANT 540
      DO 120 I = 1, 3
                                                                               CANT 550
  120 CDLXYZ(I) = XYZDS(I)
                                                                                CANT 560
      DO 125 I = 4, 15
                                                                                CANT 570
  125 \text{ CDLXYZ}(I) = 0.0
                                                                                CANT 580
      DO 130 I = 1,4
                                                                                CANT 590
  130 \text{ STLN}(I) = \text{STLNO}(I)
                                                                                CANT 600
C
          SAVE TOTAL MAIN STRUT LENGTH = STRUT 2 + STRUT 4
                                                                                CANT 610
      STLNS = STLN(2) + STLN(4)
                                                                               CANT 620
С
                                                                               CANT 630
С
                PRINT INITIAL POSITIONS
                                                                               CANT 640
С
                                                                               CANT 650
                                                                               CANT 660
      CALL OUTPT1
      GO TO 150
                                                                               CANT 670
 1000 CONTINUE
                                                                               CANT 680
С
                                                                               CANT 690
С
           UPDATE COORDINATES OF NODE POINTS FOUR AND FIVE
                                                                               CANT 700
С
                                                                               CANT 710
                                                                               CANT 720
      DO 140
                I = 1, 3
      K = I + 3
                                                                               CANT 730
                                                                               CANT 740
      XYZPOS(I,NPOINT) = XYZPOS(I,NPOINT) + CDLXYZ(I)
                                                                               CANT 750
      XYZPOS(I, 4) = XYZPOS(I, 4) + CDLXYZ(K)
                                                                               CANT 760
  140 CONTINUE
                                                                                CANT 770
      IF ( NITERS .LE. 1 ) GO TO 145
                                                                                CANT 780
CANT 790
       IF ( I2ND • NE• 1 ) GO TO 147
      I2ND = 0
      CDLXYZ (1) = 0.0
                                                                                CANT 800
      CDLXYZ (2) = 0.0
                                                                                CANT 810
      CDLXYZ (3) = 0.0
                                                                                CANT 820
      RMOVE = SQRT((BASEPX-XYZPOS(1,5))**2 + (BASEPY-XYZPOS(2,5))**2
                                                                                CANT 330
                  + (BASEPZ-XYZPOS(3,5))**2 )
                                                                                CANT 840
     1
      IF ( RMOVE .LT. RADRAG ) GO TO 147
                                                                                CANT 850
      WRITE(6,144)
                                                                                CANT 860
  144 FORMAT (43H1ERROR *** NO SOLUTION FOUND AT THIS LEVEL. )
                                                                                CANT 870
                                                                                CANT 580
      CALL OUTPT1
      RETURN
                                                                                CANT 890
                                                                                CANT 900
  145 CONTINUE
      IF ( LDIND .EQ. 2 ) GO TO 146
                                                                               CANT 910
                 DETERMINE THE CENTER OF SOLUTION CIRCLE
С
                                                                                CANT 920
      BASEPX = XYZPOS(1,5)
                                                                                CANT 930
      BASEPY = XYZPOS(2,5)
                                                                                CANT 940
      BASEPZ = XYZPOS(3,5)
                                                                               CANT 950
  146 CDLXYZ (1) = 0.0
                                                                               CANT 960
                                                                               CANT 970
      CDLXYZ (2) = 0.0
       CDLXYZ (3) = 0.0
                                                                               CANT 980
  147 CONTINUE
                                                                                CANT 990
Ċ
                                                                                CANT1000
           CALCULATE BENDING MOMENTS AND SHEAR FORCES
С
                                                                                CANT1010
С
                                                                                CANT1020
       CALL BNDLDS
                                                                                CANT1030
           SET LOCAL FORCE/MOMENT VECTOR WITH BENDING LOADS
С
                                                                                CANT1040
      FMVCTL(2,2)= YSHRFP(2)
FMVCTL(5,2)= YSHRFQ(2)
FMVCTL(12,2)= ZSHRMQ(2)
                                                                                CANT1050
                                                                                CANT1060
                                                                                CANT1070
       FMVCTL(2,4) = YSHRFP(4)
                                                                                CANT1080
```

```
EMVCTL(5,4) = YSHRFQ(4)
                                                                           CANT1090
      FMVCTL(12,4) = ZSHRMQ(4)
                                                                           CANT1100
С
          CALCULATE CURRENT STRUT LENGTHS AND TRANSFORMATION MATRICES
                                                                           CANT1110
      CALL TRANSM(XYZPOS(1,1),XYZPOS(1,4),XYZPOS(1,2),TM(1,1),STLN(1) ) CANT1120
      CALL TRANSM(XYZPOS(1,2),XYZPOS(1,4),R24XYZ( 1 ),TM(1,2),STLN(2) ) CANT1130
      CALL TRANSM(XYZPOS(1,3),XYZPOS(1,4),XYZPOS(1,2),TM(1,3),STLN(3) ) CANT1140
      CALL TRANSM(XYZPOS(1,5),XYZPOS(1,4),R24XYZ( 1 ),TM(1,4),STLN(4) ) CANT1150
  150 CONTINUE
                                                                           CANT1160
С
                                                                           CANT1170
С
          CALCULATE STRUT FORCES AND AXIAL STIFFNESSES
                                                                           CANT1180
С
                                                                           CANT1190
      NITERS = NITERS + 1
                                                                           CANT1200
      ISTOP1 = 0
                                                                           CANT1205
      DO 190 IST = 1, NELEM
                                                                           CANT1210
С
               COMPUTE STROKE
                                                                           CANT1220
      STR(IST) = STLN(IST) - STLNO(IST)
                                                                           CANT1230
               GET MATERIAL NO.
С
                                                                           CANT1240
      MNO = IMATL(IST)
                                                                           CANT1250
      IF ( INDPL(MNO) •EQ• 2 ) GO TO 160 CANT1260
COMPUTE AXIAL FORCE AND STIFFNESS----ELASTIC MATERIAL IN STRUTCANT1270
С
      AXSTIF(IST) = AREA(MNO)*ELASAX(MNO) / STLN(IST)
                                                                           CANT1280
      AXIALF(IST) = STR(IST) * AXSTIF(IST)
                                                                           CANT1290
                                                                           CANT1300
      GO TO 170
          COMPUTE AXIAL FORCE AND STIFFNESS----PLASTIC MATERIAL IN STRUTCANT1310
С
  160 CONTINUE
                                                                           CANT1320
      CALL STRUT ( IST, SPNGC(IST) , SPNGT(IST) , DISTC(IST) ,
                                                                           CANT1330
                          DISTT(IST) , PFORC(IST) , PFORT(IST) ,
                                                                           CANT1340
     1
                          FOREVC(IST), FOREVT(IST), INDULC(IST),
                                                                           CANT1350
     2
                          INDULT(IST), IPREV(IST), STRP(IST),
                                                                           CANT1360
     3
                          IPOSC(IST) , IPOST(IST) , IRET(MNO)
                                                                           CANT1370
                                                                ,
     4
                          SCMAX(MNO) , STMAX(MNO) , CDC(1,MNO) ,
                                                                           CANT1380
     5
                          SRULT(MNO) , SRULC(MNO) , SRT(1,MNO) ,
                                                                           CANT1390
     6
                          SRC(1,MNO) , PFT(1,MNO) , PFC(1,MNO) ,
                                                                           CANT1400
     7
                          STR(IST)
                                     , AXIALF(IST), AXSTIF(IST),
                                                                           CANT1410
     8
                          IFIRST(MNO), NITERS
                                                   , CDT(1,MNO) ,
                                                                           CANT1420
     9
                          ISTOP2 )
                                                                           CANT1422
                                                                           CANT1425
      IF ( ISTOP2 .LT. 0 ) ISTOP1 = 1
  170 CONTINUE
                                                                           CANT1430
          SET LOCAL FORCE/MOMENT VECTOR WITH AXIAL FORCES
                                                                            CANT1440
C
      FMVCTL(1, IST) = -AXIALF(IST)
                                                                           CANT1450
                                                                           CANT1460
      FMVCTL(4,IST) = AXIALF(IST)
                                                                            CANT1470
  190 CONTINUE
      IF ( ISTOP1 .EQ. 1 ) RETURN
                                                                            CANT1475
                                                                            CANT1480
С
С
           CONVERT LOCAL FORCE/MOMENT VECTORS TO GLOBAL FORCE/MOMENT
                                                                            CANT1490
С
                                                                            CAMT1500
          VECTORS
                                                                            CANT1510
С
                                                                            CANT1520
      DO 210 IST = 1, NELEM
                                                                            CANT1530
      CALL TRALMG( FMVCTG(1,IST), FMVCTL(1,IST), TM(1,IST), 4 )
  210 CONTINUE
                                                                            CANT1540
                                                                            CANT1550
С
С
           SUM FORCE/MOMENTS AT MODE POINT FOUR
                                                                            CANT1560
                                                                            CANT1570
                          AND-SET UP SOLUTION ARRAY
С
          230 I = 1, 3
                                                                            CANT1580
      DO
                                                                            CANT1590
           = I + 9
      L
                                                                            CANT1600
      Κ
           = 1 + 3
```

SUMOK1=FMVCTG(K,1)+FMVCTG(K,3) CANT1610  $SUMOK_2 = FMVCTG(K, 2) + FMVCTG(K, 4)$ CANT1620 SUMFM4(I)=SUMOK1+SUMOK2 CANT1630 SUMOK1=FMVCTG(L,1)+FMVCTG(L,3) CANT1640 SUMOK2=FMVCTG(L,2)+FMVCTG(L,4) CANT1650 SUMFM4(K)=SUMOK1+SUMOK2 CANT1660 SARRY (K) = SUMFM4(I)CANT1670 SARRY (L) = SUMFM4(K) CANT1680 SARRY (I) = FMVCTG(I,4)CANT1690 230 CONTINUE CANT1700 IF ( IFRIST .EQ. 0 ) GO TO 280 CANT1710 GET MOMENTS AT NODES TWO AND FIVE FOR SOLUTION ARRAY С CANT1720 DO 240 I = 7,9 CANT1730 SARRY (I+6) = FMVCTG(I,2) SARRY (I) = FMVCTG(I,4) CANT1740 CANT1750 240 CONTINUE CANT1760 END OF LOOP C CANT1770 IF ( NITERS .GT. NITER ) GO TO 400 CANT1780 DO 260 I = 4, 15 CANT1790 CDLXYZ(I) = -SARRY(I)CANT1800 260 CONTINUE CANT1810 CHECK FOR CONVERGENCE CANT1820 Ċ DO 265 I = 4, 15CANT1830 IF ( ABS( CDLXYZ(I)) .GT. TOL ) GO TO 280 CANT1840 265 CONTINUE CANT1850 IF THREE FOOTPAD JOINT DISPLACEMENTS ARE APPLIED, CANT1860 С C GO TO OUTPUT AND ENERGY SECTION CANT1870 IF ( LDIND .EQ. 2 ) GO TO 400 CANT1880 IF ( IFIN .EQ. 1 ) GO TO 400 CANT1890 COMPUTE FRICTION FORCE CANT1900 С = COEFF * FMVCTG(IDSPDR,4) CANT1910 FRICTE ABSFIR = SQRT( FMVCTG(IDSPD1,4)**2 + FMVCTG(IDSPD2,4)**2) CANT1920 IF ( I1ST •NE• 0 ) GO TO 233 CANT1930 I1ST = 1CANT1940 IF ( ABSFIR .LE. ABS( FRICTF ) ) GO TO 400 CANT1950 PICK UP THE COS OF THE ANGLE (ALF) BETWEEN THE MAIN STRUT AND CANT1960 С С THE NORMAL CANT1970 COSALF = ABS(TM(ICOS, 2))CANT19S0 IF ( COSALF .LT. 1. ) GO TO 500 CANT1990 WRITE(6,501) CANT2000 501 FORMAT ( 100H1 ERROR *** DATA CASE TERMINATED, ANGLE BETWEEN MAINCANT2010 1 STRUT AND THE NORMAL TO THE LANDING SURFACE IS ) CANT2020 CANT2030 WRITE(6.502) 502 FORMAT ( 60X 14H.EQ. O DEGREES ) CANT2040 CALL OUTPT1 CANT2050 CANT2060 RETURN CANT2070 500 CONTINUE IF ( COSALF .GT. O.) GO TO 505 CANT2080 WRITE(6,501) CANT2090 CANT2100 WRITE(6,503 ) 503 FORMAT ( 60X 15H.EQ. 90 DEGREES ) CANT2110 CALL OUTPT1 CANT2120 CANT2130 RETURN 505 CONTINUE CANT2140 TANALF = SQRT( 1. - COSALF**2 ) / COSALF CANT2150 RADLIM = AMAX1( XYZDS(IDSPDR) / TANALF, XYZDS(IDSPDR) * TANALF ) CANT2160

		RADLIM=AMIN1(RADLIM,SQRT(STLNS**2-(STLNS-XYZDS(IDSPDR))**2))	CANT2170
С		SET OUTER RADIUS LIMIT	CANT2180
		RADRAG = RADLIM * 1.05	CANT2190
		RDSTEP = RADRAG / 50.	CANT2200
	233	CONTINUE	CANT2210
С		RDSTEP IS POSITIVE AND DELTF2 NEGATIVE UNTIL SOLUTION POINT	CANT2220
С		IS PASSED	CANT2230
_		DELIF2 = ABS( FRICTF ) - ABSFIR	CANT2240
C		TEST FOR CONVERGENCE	CANT2250
		IF ( ABS(DELIF2) LE • 0005*ABS(FRICIF) ) GO TO 400	CANT2260
		IF ( RDSIEP + LI + 0.00) GO TO 235	CANT2270
c		$\frac{1}{1} + \frac{1}{1} + \frac{1}$	CANT2280
C		RDSTDS = RDSTED	CANT2290
		RDSTFD = RESTER / ( ARS(RELTEL ) + RELTE2 )	
C		SAVE DIFFERENCE IN ROSTED	CANT2310
~		RDSTPS = -(RDSTEP + RDSTPS) + 5	CANT2320
		60 10 237	CANT2340
	235	CONTINUE	CANT2350
С		ONCE SOLUTION POINT IS PASSED, CONTINUE TAKING BACK STEPS	CANT2360
С		UNTIL DELTF2 AGAIN BECOMES NEGATIVE AFTER WHICH ONE	CANT2370
С		MORE FORWARD STEP IS TAKEN	CANT2380
		RDSTEP = RDSTPS	CANT2390
		IF ( DELTF2 .GT. 0.0 ) GO TO 237	CANT2400
		RDSTEP = DELTF2*RDSTEP / ( ABS(DELTF2 ) + DELTF1 )	CANT2410
		IFIN=1	CANT2420
	237	CONTINUE	CANT2430
		COS1 = FMVCTG(IDSPD1,4) / ABSFIR	CANT2440
_		COS2 = FMVCTG(IDSPD2,4) / ABSFIR	CANT2450
Ç		COMPUTE NEXT SLIDING DISPLACEMENT INCREMENTS	CANT2460
		CDLXYZ(IDSPD1) = -COS1 * RDSTEP	CANT2470
~		CDEXYZ(1DSPD2) = -COS2 * ROSTEP	CAN12480
C		SAVE FRICTION FORCE UNBALANCE	CANT2490
			CANT2500
		1200 - 1	CANT2510
	229	CD(XYZ(I) = 0.0	CANT2520
	280	CONTINUE	CANT2540
	200	IFRIST = 1	CANT2550
C		INITIALIZE THE SOLUTION STLEENESS MATRIX	CANT2560
-		PO = 290  J = 13  ,  15	CANT2570
		$p_0 290 J = 1 \cdot 9$	CANT2580
		TSTIFM(J,I) = 0.0	CANT2590
	290	TSTIFM(I,J) = 0.0	CANT2600
С		COMPUTE THE SM ELEMENT FOUR	CANT2610
		IST = 4	CANT2620
		CALL STEMCE( STIEM )	CANT2630
		CALL TRNFSM( STIFM, TM(1,4), 12 )	CANT2640
		DO 300 I = $4,12$	CANT2650
		DO $300 J = 1, 12$	CANT2560
		TSTIFM(I,J) = STIFM(I,J)	CANT2670
~	300	CONTINUE	CANT2680
C		COMPUTE THE SM. FOR ELEMENT TWO	CANT2690
		151 = 2	CANT2700
		CALL STEMCE( STIEM )	CANT2710
		CALL INNESMI SIIFA, IM(1,2), 12 )	CANT2720

.

```
CANT2730
                                                                           CANT2740
      K = J + 6
                                                                           CANT2750
      L = J + 3
                                                                           CANT2760
      M = J + 9
                                                                           CANT2770
      TSTIFM(I,J) = STIFM(I,J) + TSTIFM(I,J)
TSTIFM(I,K) = STIFM(I,K) + TSTIFM(I,K)
                                                                           CANT2780
                                                                           CANT2790
      TSTIFM(I,M) = STIFM(I,L)
                                                                           CANT2800
      N = I + 6
                                                                           CANT2810
      TSTIFM(N,J) = STIFM(N,J) + TSTIFM(N,J)
                                                                           CANT2820
      TSTIFM(N,K) = STIFM(N,K)
TSTIFM(N,M) = STIFM(N,L)
                      STIFM(N,K) + TSTIFM(N,K)
                                                                           CANT2830
                                                                           CANT2840
      N = I + 9
                                                                           CANT2850
      TSTIFM(N,J) = STIFM(I+3,J)
                                                                           CANT2860
      TSTIFM(N,K) = STIFM(I+3,K)
                                                                           CANT2870
      TSTIFM(N,M) = STIFM(I+3,L)
                                                                           CANT2880
  310 CONTINUE
                                                                           CANT2890
С
               COMPUTE THE SM. FOR ELEMENT ONE AND THREE
                                                                           CANT2900
      DO 320 IST = 1, 3, 2
                                                                           CANT2910
      CALL STFMIT ( STIFM6, AXIALF(IST), STLN(IST), AXSTIF(IST) }
                                                                           CANT2920
      CALL TRNFSM ( STIFM6, TM(1,IST), 6 )
                                                                           CANT2930
      DO 330 I = 4, 6
                                                                           CANT2940
      DO 330 J = 4.6
                                                                           CANT2950
      TSTIFM(I,J) = TSTIFM(I,J)+STIFM6(I,J)
                                                                           CANT2960
  330 CONTINUE
                                                                           CANT2970
  320 CONTINUE
                                                                           CANT2980
С
          ALTER THE SOLUTION VECTOR FOR KNOWN DISPLACEMENTS AT NODE FIVECANT2990
      DO
          340 I = 4, 15
                                                                           CANT3000
      DO 340 J = 1, 3
                                                                            CANT3010
  340 CDLXYZ(I) = CDLXYZ(I) - CDLXYZ(J)*TSTIFM(I,J)
                                                                           CANT3020
С
          THE 12 BY 12 MATRIX STORED IN TSTIFM IN COLUMNS 4-15 AND
                                                                            CANT3030
С
          ROWS 4-15 IS THE STIFFNESS MATRIX OF INTEREST. STORE THE
                                                                           CANT3040
С
          UPPER TRIANGULAR PORTION OF THIS MATRIX IN SAVED
                                                                           CANT3050
      JJ=0
                                                                           CANT3060
      DO 50 I=4,15
                                                                           CANT3070
      DO 50 J=4,I
                                                                           CANT3080
      JJ=JJ+1
                                                                           CANT3090
   50 SAVED(JJ)=TSTIFM(J,I)
                                                                           CANT3100
      DO 1 I=1,12
                                                                           CANT3110
    1 CDL(I)=CDLXYZ(I+3)
                                                                            CANT3120
      CALL DMFSS(SAVED, 12, 1.E-12, IRANK, TRAC)
                                                                           CANT3130
      IF(IRANK • LE • 0) GO TO 53
                                                                           CANT3140
      INC=100
                                                                           CANT3150
      GO TO 55
                                                                           CANT3160
   53 WRITE(6,54)IRANK
                                                                           CANT3170
   54 FORMAT(*1 BAD STIFFNESS MATRIX ----RANK = *,15)
                                                                           CANT3150
      CALL OUTPT1
                                                                           CAN:T3190
      RETURN
                                                                            CANT3200
   55 CONTINUE
                                                                           CANT3210
      CALL DMLSS(SAVED, 12, IRANK, TRAC, INC, CDL, IER)
                                                                            CANT3220
      DO 6 I=1,12
                                                                            CANT3230
    6 CDLXYZ(I+3)=CDL(I)
                                                                            CANT3240
С
                                                                            CANT3250
С
                 CORRECT FOR ROUND OFF ERROR
                                                                            CANT3260
С
                                                                            CANT3270
      DO 360 I = 4,6
                                                                            CANT3280
```

```
IF ( ABS(CDLXYZ(I)) \cdot LT \cdot 1 \cdot E - 11 ) CDLXYZ(I) = 0 \cdot 0
                                                                           CANT3290
 360 CONTINUE
                                                                           CANT3300
      IF ( LDIND .EG. 2 ) GO TO 1000
                                                                           CANT3310
      XYZDS(IDSPD1) = XYZDS(IDSPD1) + CPLXYZ(IDSPD1)
                                                                           CANT3320
      XYZDS(IDSPD2) = XYZDS(IDSPD2) + CDLXYZ(IDSPD2)
                                                                           CANT3330
                                                                           CANT3340
      GO TO 1000
 400 CONTINUE
                                                                           CANT3350
      CALL ENERGY( FMVCTG(1,4) )
                                                                           CANT3360
                                                                           CANT3370
      CALL OUTPUT
      IF ( NITERS .LE. NITER ) GO TO 405
                                                                           CANT3380
                                                                           CANT3390
      WRITE(6,403)
 403 FORMAT (34H1MAXIMUM ITERATION MUMBER EXCEEDED )
                                                                           CANT3400
      RETURN
                                                                           CANT3410
                                                                           CANT3420
 405 CONTINUE
                                                                           CANT3430
      IF ( NDISPS .GE. NSTEP ) RETURN
      IF( IFPATN •NE• 1 ) GO TO 430
                                                                           CANT3440
  427 CONTINUE
                                                                           CANT3450
      IF ( FMVCTG(IDSPDR,4) .GE. 0.0 ) GO TO 430
                                                                           CANT3460
                                                                           CANT3470
      WRITE(6,428)
  428 FORMAT (21HINEGATIVE NORMAL LOAD )
                                                                           CANT3480
                                                                           CANT3490
      RETURN
                                                                           CANT3500
  430 CONTINUE
      IF ( LDIND .EQ. 2 ) GO TO 432
                                                                           CANT3510
      XYZDS(IDSPD1) = 0.0
XYZDS(IDSPD2) = 0.0
                                                                           CANT3520
                                                                           CANT3530
  432 CONTINUE
                                                                           CANT3540
                                                                           CANT3550
      DO 434 I = 1, 3
  434 \text{ CDLXYZ(I)} = \text{XYZDS(I)}
                                                                           CANT3560
                                                                           CANT3570
      DO \ 436 \ I = 4, 15
  436 \text{ CDLXYZ(I)} = 0.0
                                                                           CANT3580
                                                                           CANT3590
      I1ST = C
      IFIN = \hat{U}
                                                                           CANT3600
                                                                           CANT3610
      I2ND = 0
           SAVE TOTAL MAIN STRUT LENGTH = STRUT 2 + STRUT 4
                                                                            CANT3520
С
                                                                            CANT3630
      STLNS = STLN(2) + STLN(4)
      NITERS = 1
                                                                            CANT3640
                                                                           CANT3650
С
          SET STRUT SAVE VARIABLES FOR THIS POSITION
                                                                           CANT3660
С
С
                                                                           CANT3670
                                                                            CANT3680
      ISTOP1 = 0
                                                                            CANT3690
      DO 440 IST = 1, 4
      MMO = IMATL(IST)
                                                                            CANT3700
      IF ( INDPL(MNO) •NE• 2 ) GO TO 440
                                                                            CANT3710
      CALL STRUT ( IST, SPNGC(IST) , SPNGT(IST) , DISTC(IST) ,
                                                                           CANT3720
                          DISTT(IST) , PFORC(IST) , PFORT(IST) ,
                                                                           CANT3730
     1
                          FOREVC(IST), FOREVT(IST), INDULC(IST),
                                                                           CANT3740
     2
                          INDULT(IST), IPREV(IST) , STRP(IST) ,
                                                                           CANT3750
     3
                                                                           C4NT3760
                          IPOSC(IST) , IPOST(IST) , IRET(MNO) ,
     4
                          SCMAX(MNO) , STMAX(MNO) , CDC(1,MNO) ,
                                                                            CANT3770
     5
                          SRULT(MNO) , SRULC(MNO) , SRT(1,MNO) ,
                                                                           CANT3750
     6
                          SRC(1,MNO) , PFT(1,MNO) , PFC(1,MNO) ,
                                                                           CANT3790
     7
                                    , AXIALF(IST), AXSTIF(IST),
                                                                           CANT3800
                          STR(IST)
     8
                          IFIRST(MNO), NITERS , CDT(1,MNO),
                                                                           CANT3810
     9
                          ISTOP2 )
                                                                           CANT3815
     А
                                                                           CANT3820
     IF ( ISTOP2 .LT. 0 ) ISTOP1 = 1
                                                                            CANT3830
  440 CONTINUE
```

IF ( ISTOP1 .EQ. 1 ) RETURN GO TO 280 END

CANT3840 CANT3850 CANT3860

BNDL SUBROUTINE BNDLDS 10 COMMON / CMAIN / BNDL 20 LGRTYP, LDIND , INITER, NPOINT, NREFP , NELEM , NUMB BNDL 1 30 NSTEP , NITER , TOL , ENGABS, IFPATN, NFPCRL, FPCRLD(3), BNDL 40 2 , IMATL(4) , MNO , FPCRST(3), ENDL 50 IGLDIR, NMATL , IST 3 INDPL(4) , ELASEN(4) , ELASAX(4) BNDL 60 4 9 • AREA(4) , IRET(4) BINERT(4) BNDL 70 5 , IFIRST(4) , SCMAX(4) , STMAX(4) BNDL 80 6 SRULC(4) , SRULT(4), CDC(5,4), SRC(5,4) BNDL 90 7 CDT(5,4) , CRC(5,4) , SRT(5,4) BNDL 100 8 , NPN , XYZPOS(3,5) PNDL 110 , PFT(5,4) 9 PFC(5,4) BNDL 120 , DC(3,3) , PHI , THTA , PSI 9 BNDL 130 9 , (GYCDL(3), EGYCDL(3) COMMON / CMAIN BNDL 140 1 NITERS, NDISPS, STLN2 , BNDL 150 Α В IDSPDR, IDSPD1, IDSPD2, SPDISP, COEFF, XYZDSP(3), BNDL 160 , STRP(4) , STLNO(4) BNDL 170 С STLN(4) BNDL 180 , SPNGT(4) , DISTC(4) D SPNGC(4) • , PFORT(4) , PFORC (4) BNDL 190 Ε DISTT(4) . FOREVT(4) , IPOSC(4) BNDL 200 F FOREVC(4) , , INDULC(4) , INDULT(4) IPOST(4) BNDL 210 G • BNDL 220 STR(4) , AXIALF(4) AXSTIF(4) Н , XYZDS (3) IPREV(4) sFORCE(30) BNDL 230 Ι , SUMEM4(6) ,R24XYZ(3) BNDL 240 J CFORCE(30) , FMVCTG(12,4) • FMVCTL(12,4) BNDL 250 Κ TM(36) YSHRFP(4), ZSHRFP(4), YSHRMP(4), ZSHRMP(4), CGYXYZ(3), BNCL 260 L BNDL 270 YSHRFQ(4), ZSHRFQ(4), YSHRNQ(4), ZSHRNQ(4), EGYXYZ(3) М DOUBLE PRECISION A, B, C, D, E, F, AL24, AL45, AL25, AL, DELTA, BNDL 250 3NDL 290 SINALF, COSALF, SINBET, COSBET, EB2, EB4, AI2, 1 AI4, TERM, TERM1, V2, V4, BM, RM1, RM2, RM3 BNDL 300 2 BNDL 310 A = XYZPOS(1,5) - XYZPOS(1,2)BNDL 320 B=XYZPOS(2,5)-XYZPOS(2,2) C=XYZPOS(3,5)-XYZPOS(3,2) BNDL 330 BNDL 340 D = XYZPOS(1,5) - XYZPOS(1,4)E = XYZPOS(2,5) - XYZPOS(2,4)BNDL 350 F = XYZPOS(3,5) - XYZPOS(3,4)BNDL 360 AL24 = DSQRT((A-D)**2 + (B-E)**2 + (C-E)**2)BNDL 370 BNDL 380 AL45 = DSQRT(D*D + E*E + F*F)BNDL 390 AL25 = DSQRT( A*A + B*B + C*C ) BNDL 40C AL = AL25DELTA ≈ DSGRT((B*F-C*E)**2 + (C*D-A*F)**2 + (A*E-0*D)**2)/AL25 BNDL 410 BNDL 420 SINALF = DELTA / AL24 COSALF = DSQRT( 1. - SINALF*SINALF ) BNDL 430 BNDL 440 SINBET = DELTA / AL45 COSBET = DSQRT( 1. - SINBET * SINPET ) BNDL 450 AL24 = AL24 * COSALFBNDL 460 AL45 = AL45 * COSEFT ENDL 470 BNDL 480 IM2=IMATL(2)**BNDI 490** IM4=IMATL(4) EB2=ELASBN(IM2) BNDL 500 BNDL 510 EB4=ELASEN(IM4) AI2=BINERT(IM2) BNDL 520 BNDL 530 AI4=BINERT(IM4) TERM=(AL24**3*AL45/(2•*AL)-AL24**4*AL45/(3•*AL*AL))/(EE2*AI2)+ PNDI 540 (AL24**2*AL/3. - AL24**3 + AL24**4/AL - AL24**5/(3.*AL*AL))/ ENDL 550 1

2 (EB4*AI4)	BNDL 560
TERM1=DELTA/(TERM*AL)	BNDL 570
V2=AL45*TERM1	BNDL 580
V4=AL24*TERM1	BNDL 590
BM=AL24*AL45*TERM1	BNDL 600
YSHRFP(2) = V2	BNDL 610
YSHRFQ(2) = -V2	BNDL 620
YSHRFP(4) = V4	BNDL 630
YSHRFQ(4) = -V4	ENDL 640
ZSHRMQ(2) = BM	BNDL 65(
ZSHRMQ(4) = BM	BNDL 660
RM1 = XYZPOS(1,4) + (C*C*D - A*C*F - A*B*E + B*B*D)/(AL25**2	) BNDL 67(
RM2 = XYZPOS(2,4) + (A*A*E - A*B*D - B*C*F + C*C*E)/(AL25**2	) PNDL 68(
RM3 = XYZPOS(3,4) + (B*B*F - B*C*E - A*C*D + A*A*F)/(AL25**2	) BNDL 690
$R_24XYZ(1) = RM1$	BNDL 700
R24XYZ(2) = RM2	BNDL 710
R24XYZ(3) = RM3	BNDL 720
RETURN	BNDL 73(
END	BNDL 74(

```
SUBROUTINE STEMCE( STIEML )
                                                                       STBN
                                                                            10
   COMMON / CMAIN /
                                                                       STBN
                                                                             20
       LGRTYP, LDIND , INITER, NPOINT, NREFP , NELEM , NUMB
  1
                                                                       STBN
                                                                             30
       NSTEP , NITER , TOL , ENGABS, IFPATN, NFPCRL, FPCRLD(3),
  2
                                                                       STBN
                                                                             40
       IGLDIR, NMATL, IST, IMATL(4), MNO, FPCRST(3),
                                                                       STBN
                                                                             50
  3
                                   , ELASAX(4)
       INDPL(4)
                     • ELASEN(4)
  4
                                                                       STBN
                                                                             6 C
                                                      ,
  5
       DIMERT(4)
                      , AREA(4)
                                      IRET(4)
                                                                       STBN
                                                                             70
                                                       9
                      SCMAX(4)
                                      , STMAX(4)
                                                                             80
       IFIRST(4)
                                                                       STBN
  6
                                                       ,
       SRULC(4) , SRULT(4), CDC(5,4), SRC(5,4) ,
  7
                                                                       STBN 90
                     • CRC(5•4) • SRT(5•4)
                                                                       STBN 100
      CDT(5,4)
  8
                                                      •
       PFC(3,4)
                                      , NPN , XYZPOS(3,5)
  9
                      , PFT(5,4)
                                                                       STBN 110
  0
      , DC(3,3)
                      , PHI
                                     , THTA
                                                    , PSI
                                                                       STEN 120
     , CGYCDL(3), EGYCDL(3)
  Ģ
                                                                       STBN 130
   COMMON / CMAIN /
                                                                       STEN 140
       MITERS, NDISPS, STLN2 ,
                                                                       STBN 150
  Δ
        ICSPOR, IDSPD1, IDSPD2, SPDISP, COEFF, XYZDSP(3),
  R
                                                                       STBN 160
                                                                       STBN 170
       STLN(4)
                     , STLNO(4)
                                              , STRP(4) ,
  С
                                      , DISTC(4)
                      spngt(4)
       SPNGC(4)
                                                                       STBN 180
  D
                                                      ,
                                      , PFCRT(4)
       DISTT(4)
                      , PFORC (4)
                                                                       STBN 190
  F
                                                       ,
                                      , IPCSC(4)
                                                                       STBN 200
  F
       FOREVC(4)
                      , FOREVT(4)
                                                       ,
                                      , INDULT(4)
  G
       IPOST(4)
                      , INDULC(4)
                                                                       STBN 210
                                                       ,
                      , AXIALF(4)
                                      , AXSTIF(4)
       STR(4)
                                                                       STEN 220
  н
                                                       ,
                      , XYZDS (3)
                                      , SFORCE(30)
                                                                       STEN 230
        IPREV(4)
   I
                                                       ,
                      , SUMEM4(6)
                                      ,R24XYZ(3)
                                                                       STEN 240
       CFORCE(30)
   .1
                                                    •
  Κ
       TE(9,4)
                      • FMVCTL(12,4)
                                      , FMVCTG(12,4)
                                                                       STBN 250
       YSHREP(4), ZSHREP(4), YSHRMP(4), ZSHRMP(4), CGYXYZ(3),
                                                                       STBN 260
  L
       YSHREQ(4), ZSHREQ(4), YSHRMQ(4), ZSHRMQ(4), EGYXYZ(3)
                                                                       STBN 270
  M
                                                                       STBN 280
       CALCULATE THE LOCAL STIFFNESS MATRIX FOR A BENDING STRUT
                                                                       STBN 290
                                                                       ST3N 300
                                                                       STBN 310
   DIMENSION STIFML(12,12)
                                                                       STEN 320
   MNO = IMATL(IST)
   DO 100 I = 2, 12
                                                                       STBN 330
       = I - 1
                                                                       STBN 340
   ĸ
   DO 100 J = 1, K
                                                                       STEN 350
                                                                       STBN 350
100 STIFML(I,J) = 0.0
   STIF#L(1,1) = AXSTIF(IST)
                                                                       STBN 370
   STIFML(2,2) = 6.*AXIALE(IST)/(5.*STLN(IST)) +
                                                                       STEN 380
                      12.*ELASEN(MMO)*PIMERT(MMO)/ STLN(IST)**3
                                                                       STEN 390
   1
    STIFML(3,3) = STIFML(2,2)
                                                                       STEN 400
    STIFML(4,4) = STIFML(1,1)
                                                                       STBN 410
   STIFML(5,5) = STIFML(2,2)
                                                                       STBN 420
    STIFML(6,6) = STIFML(2,2)
                                                                       STBN 430
    STIF\times L(7,7) = 0.0
                                                                       STBN 440
    STIFML(8,) = 2.*AXIALF(IST)*STLM(IST)/15.+4.*ELASBN(MMC)
                                                                       STBN 450
                                       *BINERT(MNG) / STLN(IST)
                                                                       STEN 460
   1
   STIFML(9,9) = STIFML(8,8)
                                                                       STEN 470
    STIFML(10,10) = 0.0
                                                                       STBN 480
                                                                       STBN 490
    STIFML(11,11) = STIFML(8,8)
    STIFML(12,12) = STIFML(8,8)
                                                                       STBN 500
    STIFML(4,1) = -STIFML(1,1)
                                                                       STBN 510
    STIFML(5,2) =-STIFML(2,2)
                                                                       STEN 520
    STIFML(6,3) =-STIFML(2,2)
                                                                       STBN 530
    STIFML(8,0) = -AXIALF(IST)*.1 - 6.*ELASEN(MNO)*BINERT(MNO)
                                                                       STEN 540
                                                                       STBN 550
   1
                                       / STLN(IST)**2
```

```
C
C
C
```

STBN	560
STBN	570
STBN	580
STBN	590
STEN	600
STBN	610
STEN	620
STRN	630
STEN	640
STBN	650
STBN	660
STBN	670
STBN	680
STBN	690
STEN	700
STBN	710
	STBN STBN STBN STBN STBN STBN STBN STBN

Ą

## THIS PAGE INTENTIONALLY LEFT BLANK

APPENDIX I

## PROGRAM LISTING

## LANDING LOADS AND MOTIONS PROGRAM

```
OVERLAY ( LLMPT5, 0, 0 )
                                                                             OVER 10
      PROGRAM LLMP
                      (INPUT,OUTPUT,TAPE5=INPUT,TAPE6=OUTPUT,TAPE3,TAPE4)LLMP
                                                                                    10
С
                                                                             LLMP
                                                                                    20
С
      LANDING LOADS AND MOTIONS PROGRAM
                                                                             LLMP
                                                                                    30
С
      MASTER AGREEMENT, CONTRACT NAS1-8137, TASK ORDER NUMBER FIVE
                                                                             LLMP
                                                                                    40
С
      MCDONNELL DOUGLAS ASTRONAUTICS COMPANY - EAST
                                                                             LLMP
                                                                                    50
C
                                                                             LLMP
                                                                                    60
          FORTRAN I/O UNIT 3 - SECONDARY TIME HISTORY DATA PLACED ON
THIS FILE (NFORC=0, NO SECONDARY TIME
С
                                                                             LLMP
                                                                                    70
С
                                                                             LLMP
                                                                                    80
C
C
                                 HISTORY OUTPUT - NFORC=1, SECONDARY
                                                                             LLMP
                                                                                    90
                                 HISTORY OUTPUT ON TAPE3).
                                                                             LLMP 100
С
          FORTRAN I/O UNIT 4 - WORK FILE ASSOCIATED WITH MULTIPLE DATA
                                                                             LLMP 110
С
                                 CASES.
                                                                             LLMP 120
¢
          FORTRAN I/O UNIT 5 - CARD INPUT DATA.
                                                                             LLMP 130
С
          FORTRAN I/O UNIT 6 - PRINTED OUTPUT.
                                                                             LLMP 140
С
                                                                             LLMP 150
      COMMON COMINT(1400)
                                                                             LLMP 160
      EQUIVALENCE ( COMINT ( 362 ), IDSETN )
                                                                             LLMP 170
      EQUIVALENCE ( COMINT (1399), NOCASE )
                                                                             LLMP 180
      EQUIVALENCE ( COMINT (1400), TYMIN )
                                                                             LLMP 190
      COMMON / THISV / TIMHSA(275)
                                                                             LLMP 200
      DO 1000 I = 1, 1400
                                                                             LLMP 210
 1000 \text{ COMINT(I)} = 0.0
                                                                             LLMP 220
      DO 2000 I = 1, 275
                                                                             LLMP 230
 2000 \text{ TIMHSA(I)} = 0.0
                                                                             LLMP 240
С
                                                                             LLMP 250
С
          CALL THE INPUT AND INITIALIZATION ROUTINES
                                                                             LLMP 260
С
                                                                             LLMP 270
    1 CONTINUE
                                                                             LLMP 280
LLMP 290
c
      CALL SECOND(TYMIN)
                                                                             LLMP 300
С
                                                                             LLMP 310
      CALL OVERLAY(6LLLMPT5,1,0,6HRECALL )
                                                                             LLMP 320
С
                                                                             LLMP 330
С
           CALL THE LANDER LANDING, LOADS, AND MOTION SIMULATION
                                                                             LLMP 340
С
                ROUTINES
                                                                             LLMP 350
      CALL OVERLAY(6LLLMPT5,2,0,6HRECALL )
                                                                             LLMP 360
С
                                                                             LLMP 370
      CALL SECOND(TYMOUT)
                                                                             LLMP 380
С
                                                                             LLMP 390
      RUNTYM=TYMOUT-TYMIN
                                                                             LLMP 400
      WRITE(6,3000)NOCASE,RUNTYM
                                                                              LLMP 410
      GO TO 1
                                                                             LLMP 420
С
                                                                             LLMP 430
C
C
                                                                             LLMP 440
      FORMAT STATEMENT
                                                                             LLMP 450
 3000 FORMAT(//10X,8HCASE NO ,I10,9H RAN FOR ,F7.3,11H CP SECONDS)
                                                                             LLMP 460
      END
                                                                             LLMP 470
```

## OVER 20 OVERLAY ( LLMPT5, 1, 0 ) READ 10 PROGRAM READIT READ 20 THIS PORTION OF THE PROGRAM READS THE INPUT DATA FROM READ 30 READ 40 SEQUENCED DATA CARDS. READ 50 THE CARDS MUST BE PREPARED IN THE FOLLOWING FORMAT READ 60 READ 70 COLUMN 1-4 11-20 21-30 31-40 41-50 51-60 READ 80 6-9 ICNTRL CARD DATA DATA DATA DATA DATA READ 90 SEQ. **READ 100** RT. JUST. **READ 110** READ 120 READ 130 DIMENSION GC(4,5), GCD(4,5), GCDD(4,5) **READ 140 READ 150** DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5), XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5), **READ 160** 1 **READ 170** XFPSDD(4,5), YFPSDD(4,5), ZFPSDD(4,5) 2 **READ 180** DIMENSION DUMMY(5),KONT(50), JKONT(20), NOCARD(500) **READ 190** READ 200 COMMON / THISV/ TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6), **READ 210** READ 220 INDFPC(6), 1 READ 230 strpds(10), strpms( 5), IPOCDS(10), IPOCMS( 5), URCDS (10), 2 URTDS (10), URCMS ( 5), URTMS ( 5), SETCDS(10), SETTDS(10), READ 240 3 SETCMS( 5), SETTMS( 5), INDCDS(10), INDTDS(10), INDCMS(10), READ 250 4 INDIMS( 5), PRFCDS (10), PRFTDS (10), PRFCMS ( 5), PRFTMS ( 5), READ 260 5 IPRDS (10), IPRMS ( 5), FRVDSC(10), FRVDST(10), FRVMSC( 5), READ 270 6 READ 280 FRVMST( 5), IPOTDS(10), IPOTMS( 5) L **READ 290** EQUIVALENCE ( NFTPDS, NOLEG ) **READ 300** COMMON COMINT(400) **READ 310** EQUIVALENCE ( COMINT( 1 ), ISAVCM ) READ 320 EQUIVALENCE ( COMINT( 2 ), JTEST **READ 330** 3 ), IBOTM EQUIVALENCE ( COMINT( ) **READ 340** 4 ), XSD EQUIVALENCE ( COMINT( ) READ 350 EQUIVALENCE ( COMINT( 8 ), XSDD } **READ 360** EQUIVALENCE ( COMINT( 12 ), YS ) EQUIVALENCE ( COMINT( 16 ), YSD READ 370 ) EQUIVALENCE ( COMINT( 20 ), YSDD **READ 380** ) **READ 390** EQUIVALENCE ( COMINT( 24 ); ZS ) EQUIVALENCE ( COMINT( 28 ), ZSD READ 400 ) EQUIVALENCE ( COMINT( 32 EQUIVALENCE ( COMINT( 36 READ 410 ). ZSDD ) READ 420 ), PHI ) EQUIVALENCE ( COMINT( 40 ), PHID READ 430 ١ READ 440 EQUIVALENCE ( COMINT( 44 ), WX ١ READ 450 EQUIVALENCE ( COMINT( 48 ), WXD ) READ 460 EQUIVALENCE ( COMINT( 52 ), THTA ) **READ 470** EQUIVALENCE ( COMINT( 56 ), THTAD ) EQUIVALENCE ( COMINT( 60 ), WY **READ 480** ) READ 490 EQUIVALENCE ( COMINT( 64 ), WYD ) READ 500 EQUIVALENCE ( COMINT( 68 ), PSI ١ EQUIVALENCE ( COMINT( 72 ), PSID EQUIVALENCE ( COMINT( 76 ), WZ **READ 510** ) READ 520

)

С

С

С

¢

¢

С

C

Ĉ

С

С С

C

С

С

FOUT		DEAD	= 2 0
EQUI	VALENCE ( COMINT( 80 )) W2D )		550
EQUI	VALENCE (COMINICIST ) GC )	READ	540
EQUI	VALENCE ( COMINT(104 ), GCD )	READ	550
EQUI	VALENCE ( COMINT(124 ), GCDD )	READ	560
EQUI	VALENCE ( COMINT(144 ), XFPS )	READ	570
EQUI	VALENCE ( COMINT(164 ), XFPSD )	READ	580
EQUI	VALENCE ( COMINT(184 ), XFPSDD )	READ	590
FOUT	VALENCE ( COMINT(204 ) YEPS )	READ	600
FOUT	VALENCE ( COMINT(264), ZEPS )	READ	610
FOUT	VALENCE ( COMINT(224 ), YEPSD )	READ	620
	VALENCE ( COMINICZET / ) (150 )	DEAD	630
EQUI	VALENCE ( COMINICZ44 /) TFPSDD /	DEAD	6.00
EQUI	VALENCE ( COMINI(284 ), ZFPSD )	READ	640
EQUI	VALENCE ( COMINT(304 ), ZEPSDD )	READ	650
EQUI	VALENCE ( COMINT( 324 ), TIME )	READ	660
EQUI	VALENCE ( COMINT( 325 ), HMAX )	READ	670
EQUI	VALENCE ( COMINT( 326 ), HMIN )	READ	680
EQUI	VALENCE ( COMINT( 327 ), EMIN )	READ	690
FQUI	VALENCE ( COMINT( 328 ) + EMAX )	READ	700
FQUI	VALENCE ( COMINT( 329 ) XSI )	READ	710
FOUT	VALENCE ( COMINT ( 337 ) HZ )	READ	720
EQUI		DEAD	720
EQUI	VALENCE ( COMINT( 330 )) COLERN	DEAD	740
EQUI	VALENCE ( COMINIC 339 ) IP )	DEAD	750
EQUI	VALENCE ( COMINIT 340 ), IVARH )	READ	750
EQUI	VALENCE ( COMINT( 341 ), IMTH )	READ	760
EQUI	VALENCE ( COMINT( 342 ), IPRNT )	READ	110
EQUI	VALENCE ( COMINT( 343 ), IFIN )	READ	780
EQUI	VALENCE ( COMINT( 344 ), IAD )	READ	790
EQUI	VALENCE ( COMINT( 352 ), IND )	READ	800
EQUI	VALENCE ( COMINT( 360 ), JCUT )	READ	810
FQUI	VALENCE ( COMINT( 361 ) + IPTCNT )	READ	820
EQUI	VALENCE ( COMINI ( 362 ) IDSEIN )	RFAD	830
FOUT	VALENCE ( COMINT ( 302 ), IVAL )	READ	840
	VALENCE ( COMINT 303 ) IVAL )	DEVD	950
EQUI	VALENCE ( COMINIC 364 ); X3 )	DEAD	020
	COMINI(364-367) USED BT XS	DEAD	000
EQUI	VALENCE ( COMINI (1399), NOCASE )	READ	870
COMM	10N	READ	880
1	CBMASS, CBIXX , CBIXZ , CBIYY , CBIYZ , CBIZZ , FPMASS, CBIXY,	READ	890
2	DC(3,3) , XFP(5), YFP(5), ZFP(5), WNX(5), WNY(5),	READ	900
3	WNZ(5), PX(5), PY(5), PZ(5), GM(5), OMEGA(5),	READ	910
4	GRAV, GRAVE, ZETA, FTS (6), FSXSI(5),	READ	920
5	FSYSI(5) , FSZSI(5) , SOILX(5) ,	READ	930
6	SOLLY(5) • SOLLZ(5) • PMSX(5•5) •	READ	940
7	DMSY(5,5) , $DMSY(5,5)$ , $DMSY(10,5)$ .	READ	950
0	$\frac{1}{2} \left[ \frac{1}{2} \left[ \frac{1}{2} \left[ \frac{1}{2} \left[ \frac{1}{2} \right] \right] \right] + \frac{1}{2} \left[ \frac{1}{2} \left[ \frac{1}{2} \left[ \frac{1}{2} \left[ \frac{1}{2} \right] \right] \right] \right] + \frac{1}{2} \left[ \frac{1}{2} \left[ \frac{1}{2} \left[ \frac{1}{2} \left[ \frac{1}{2} \right] \right] \right] \right]$	DEAD	060
8	$PUST(10,5) \qquad , PUSZ(10,5) \qquad , TLAS \qquad , FLIS \qquad , TLZS \qquad , TLAS \qquad $	DEAD	070
9	ILXL , ILYL , ILZL , SLU , XMSCB(5) ,	READ	910
С	YMSCB(5) , ZMSCB(5) , XDSCB(10) ,	READ	980
D	YDSCB(10) , ZDSCB(10) , ILEG , IMS ,	READ	990
Ε	FSTX , FSTY , FSTZ , PVCBX , PVCBY , PVCBZ ,	READ	1000
F	PVFPX , PVFPY , PVFPZ , NOLEG , SLOMS(5) ,	READ	1010
G	SLODS(10)	READ	1020
COMM	10N	READ	1030
1	PFCMS(5) , PFCDS(5) , PFTDS(5) ,	READ	1040
- 2	PETMS(5) SRCDS(5) SRCMS(5)	READ	1050
2	SRIDS(5) SRIMS(5) COFFDS. COFFMS. GAMDS -	RFAD	1060
ر ۱	CAMME , CHICKS, CHICKS, CHITKS, CHITKS, COMVAS, COMVAS,	READ	1070
4	CHAMING STULING STULING STULING STULING SUMANDS SUMANDS		1000
5	SIMADS, SIMAMS, CDCDS(5) , CDCMS(5) ,	READ	1080

с

```
READ1090
                                           , FRICDS, FRICMS, IRETDS,
          CDTDS(5)
                         , CDTMS(5)
     6
                                                                             READ1100
                                                  , STRKMS( 5)
          IRETMS, STROKE(10) , STRKDS(10)
     7
          LENGTH, CDXS, CDYS, CDZS, NTYPE, RAD(4), SS(4), ATTHCK(3),
                                                                             READ1110
     8
                         , ADIST , SOILP(3)
                                                   , NMODES,
                                                                             READ1120
     9
          ATTPRS(3)
          COEF , GAMMA , FRIC , SCMAX , STMAX , CDC(5), CDT(5),
                                                                             READ1130
     Α
          SRULT , SRULC , SRT(5), SRC(5), PFT(5), PFC(5), GFLEGS(5),
                                                                             READ1140
     В
          CURMSV, CURMSL, INDFXD, INDFYD, INDFZD, INDFXR, INDFYR,
                                                                             READ1150
     С
                                                                             READ1160
          INDFZR, TIMAX, DRAGST, IFP
     D
                                                                             READ1170
                                                                             READ1180
      COMMON
     1CMS, CDCONT, SOILNU, SLRHO,
                                      NOOUT, XOUT(10), YOUT(10), ZOUT(10),
                                                                             READ1190
     2MODEIN, POUTX(10,5), POUTY(10,5), POUTZ(10,5), PCGX(5), PCGY(5),
                                                                             READ1200
     3PCGZ(5),AEI1,AEI2,FORMS(5),FORDS(10),FORCE,NFORC,SAVMSX(5),
                                                                             READ1210
     4SAVMSZ(5),SAVDSX(10),SAVDSY(10),SAVDSZ(10),IQUOUT,GSINZT,
                                                                             READ1220
     5GCOSZT,STAB,STABVL,ISTAB,JCKSAB,VELX,VELY,VELZ,SAVMSY(5)
                                                                             READ1230
                                                                             READ1240
      COMMON
          SMXMSC(5), TMXMSC(5), SMXMST(5), TMXMST(5),
SMXDSC(10), TMXDSC(10), SMXDST(10), TMXDST(10)
                                                                             RFAD1250
     1
                                                                             READ1260
     2
         ,SLNGMS(5) ,SLNGDS(10), CURDSL, INLEG, IFPPRT,
IMPACT(5) ,IPRTFP(5), KOUNT(5) ,ANGX, ANGY, ANGZ
                                                                             READ1270
     3
                                                                             READ1280
     4
                                                                             READ1290
      INTEGER STOP
                                                                             READ1300
      DATA NEXT/4HNEXT/,STOP/4HSTOP/,NOGO/0/
                                                                             READ1310
      IF ( ISAVCM .EQ. 0 ) GO TO 4000
                                                                             READ1320
С
           INITIALIZE COMMONS BEFORE READING NEXT DATA CASE
                                                                             READ1330
С
C
                                                                             READ1340
                                                                             READ1350
      REWIND 4
      READ (4) (COMINT(I), I=1,1399), (TIMHSA(I), I=1,275)
                                                                             READ1360
                                                                             READ1370
 4000 CONTINUE
                                                                             READ1380
      DO 5000 I=1,50
                                                                             READ1390
 5000 KONT(I)=0
                                                                             READ1400
      DO 5001 I=1,20
                                                                             READ1410
 5001 JKONT(I)=0
                                                                             READ1420
      MMM=0
                                                                             READ1430
      WRITE(6,9000)
                                                                             READ1440
 9000 FORMAT(1H1,43X,
     *49HLANDING LOADS AND MOTIONS PROGRAM - LEGGED LANDER/38X,
                                                                             READ1450
     *61HMASTER AGREEMENT, CONTRACT NAS1-8137, TASK ORDER NUMBER FIVE
                                                                             READ1460
                                                                             READ1470
     */46X.
     *45HMCDONNELL DOUGLAS ASTRONAUTICS COMPANY - EAST ////63X,
                                                                             READ1480
     *10HINPUT DATA ////)
                                                                             READ1490
                                                                             READ1500
 9999 READ(5,9001)ICNTRL,NCARD,(DUMMY(I),I=1,5)
                                                                             READ1510
 9001 FORMAT(A4,1X,14,1X,5E10.3)
       IF(ICNTRL.EQ.NEXT) GO TO 7999
                                                                             READ1520
                                                                             READ1530
      IF(ICNTRL.EQ.STOP) STOP
                                                                             READ1540
      MMM=MMM+1
      NOCARD(MMM) = NCARD
                                                                             READ1550
                                                                             READ1560
      IF(NCARD.LE.0)GO TO 9004
      IF(NCARD.GT.42)G0 TO 9003
                                                                              READ1570
      GO TO (1,2,3,4,5,6,7,8,9,10,11,12,13,14,15,16,17,18,19,20,21,
                                                                              READ1580
              22,23,24,25,26,27,28,29,30,31,32,33,34,35,36,37,38,39,
                                                                              READ1590
     ×
                                                                              READ1600
     ×
              40,41,42
                                                                              READ1610
     ¥
             ),NCARD
                                                                              READ1620
 9003 CONTINUE
                                                                              READ1630
       JJJ=NCARD/100
       IF(JJJ.LE.0.0R.JJJ.GT.17)GO TO 9004
                                                                              READ1640
```

С

	GO TO (100,200,300,400,500,600,700,800,900,1000,1100,1200, * 1300,1400,1500,1600,1700	READ1650 READ1660
		READ1670
9004	CONTINUE	READ1680
	WRITE(6,9002) NCARD	READ1690
9002	FORMAT(24H INVALID CARD NUMBER +14)	READ1700
	NGGO=1	READ1710
	GO TO 9999	READ1720
c		PEAD1730
č	ASSTONMENT OF INDUIT DATA	
C C	ASSIGNMENT OF INPOT DATA	
C		READITOU
1	CONTINUE	READ1760
	KONT(NCARD)=KONT(NCARD)+1	READ1//O
	NOCASE =DUMMY(1)	READ1780
	GO TO 9999	READ1790
С		READ1800
2	2 CONTINUE	READ1810
	KONT(NCARD)=KONT(NCARD)+1	READ1820
	TIMAX =DUMMY(1)	READ1830
	IPTCNT =DUMMY(2)	READ1840
	INFEG =DUMMY(3)	READ1850
		READ1860
		READ1870
c		PEAD1880
۲. 		PEAD1890
2		READIOOO
	KONT(NCARD)=KONT(NCARD)+1	READIGUU
	NMODES =DUMMY(I)	READI910
	NOOUT =DUMMY(2)	READ1920
	GO TO 9999	READ1930
С		READ1940
4	+ CONTINUE	READ1950
	KONT(NCARD)≠KONT(NCARD)+1	READ1960
	INDFXD =DUMMY(1)	READ1970
	INDFYD =DUMMY(2)	READ1980
	INDFZD =DUMMY(3)	READ1990
	GO TO 9999	READ2000
C		RFAD2010
- -		READ2020
-		READ2030
		PEAD2030
		PEAD2040
		READ2050
		READ2000
-	GO TO 9999	READ2070
C		READ2080
6	5 CONTINUE	READ2090
	KONT(NCARD) = KONT(NCARD) + 1	READ2100
	HMAX =DUMMY(1)	READ2110
	HMIN =DUMMY(2)	READ2120
	EMAX =DUMMY(3)	READ2130
	EMIN =DUMMY(4)	READ2140
	IP =DUMMY(5)	READ2150
	GO TO 9999	READ2160
С		READ2170
-	7 CONTINUE	READ2180
	KONT(NCARD)=KONT(NCARD)+1	READ2190
	IVARH =DUMMY(1)	READ2200

с		IMTH CUTERR GO TO	9999	=DUMMY(2) =DUMMY(3)	
C	8	CONTIN KONT(N NFORC IQUOUT JCKSAE IDSETN GO TO	IUE ICARD) = KC 9999	DNT(NCARD)+1 =DUMMY(1) =DUMMY(2) =DUMMY(3) =DUMMY(4)	
	9	CONTIN KONT(N ZETA GRAV GRAVE GO TO	UE (CARD) = KC 9999	DNT(NCARD)+1 =DUMMY(1) =DUMMY(2) =DUMMY(3)	
	10	CONTIN KONT(N ANGX ANGY ANGZ GO TO	NUE NCARD)=K( 9999	DNT(NCARD)+1 =DUMMY(1) =DUMMY(2) =DUMMY(3)	
	11	CONTIN KONT(N WX WY WZ GO TO	NUE NCARD)=K( 9999	DNT(NCARD)+1 =DUMMY(1) =DUMMY(2) =DUMMY(3)	
c	12	CONTII KONT(I VELX VELY VELZ GO TO	NUE NCARD)=K( 9999	ONT(NCARD)+1 =DUMMY(1) =DUMMY(2) =DUMMY(3)	
c	13	CONTI KONT( CBMAS GO TO	NUE NCARD)=K S 9999	ONT(NCARD)+1 =DUMMY(1)	
	14	CONTI KONT( CBIXX CBIYY CBIZZ GO TO	NUE NCARD}=K 9999	ONT(NCARD)+1 =DUMMY(1) =DUMMY(2) =DUMMY(3)	
Ĺ	15	CONTI KONT( CBIXY CBIXZ	NUE NCARD)=K	ONT(NCARD)+1 =DUMMY(1) =DUMMY(2)	

READ2210 **READ2220 READ2230** READ2240 READ2250 READ2260 READ2270 READ2280 READ2290 **READ2300** READ2310 **READ2320 READ2330** READ2340 **READ2350** READ2360 **READ2370** READ2380 READ2390 READ2400 READ2410 READ2420 READ2430 READ2440 **READ2450 READ2460 READ2470** READ2480 READ2490 READ2500 READ2510 READ2520 READ2530 READ2540 **READ2550** READ2560 READ2570 READ2580 READ2590 READ2600 READ2610 READ2620 **READ2630** READ2640 **READ2650** READ2660 **READ2670** READ2680 **READ2690** READ2700 READ2710 READ2720 **READ2730 READ2740** READ2750 **READ2760** 

407

		CBIYZ =DUMMY(3)	READ2770
_		GO TO 9999	READ2780
С			READ2790
	100	CONTINUE	READ2800
			READ2010
			DEAD2020
			DEAD2850
			READ2850
		200((11)) = D0MMY(3)	PEAD2860
c		60 10 9999	READ2870
C			READ2880
	10		READ2890
			READ2900
			RFAD2910
c			READ2920
C	17	CONTINUE	READ2930
	1.	KONT (NCARD) = KONT (NCARD) +1	READ2940
		$P_{0} = 7000 = 1.4$	READ2950
7	000	RAD(I) = DUMMY(I)	READ2960
•	000	GO TO 9999	READ2970
С			READ2980
-	18	CONTINUE	READ2990
		KONT(NCARD)=KONT(NCARD)+1	READ3000
		DO 7001 I=1,4	READ3010
7	001	SS(I) =DUMMY(I)	READ3020
		GO TO 9999	READ3030
С			READ3040
	19	CONTINUE	READ3050
		KONT(NCARD)=KONT(NCARD)+1	READ3060
		DO 7002 I=1,3	READ3070
7	002	ATTHCK(I) =DUMMY(I)	READ3080
		GO TO 9999	READ3090
С			READ3100
	20	CONTINUE	READ3110
		KONT(NCARD)=KONT(NCARD)+1	READ3120
_		DO 7003 I=1,3	READSISU
(	003		READ3140
~		60 10 9999	READS150
C	<b>~ 1</b>		READ3170
	21		READ3180
			READ3190
			READ3200
~			READ3210
C	22	CONTINUE	READ3220
	22	KONTINOL = KONT(NCARD) + 3	READ3230
		PO = 70.04 [=1.3]	READ3240
7	004	SOILP(I) =DUMMY(I)	READ3250
•	504	GO TO 9999	READ3260
С			READ3270
	23	CONTINUE	READ3280
		KONT(NCARD)=KONT(NCARD)+1	READ3290
		NOLEG =DUMMY(1)	READ3300
		ILEG =DUMMY(2)	READ3310
		DRAGST =DUMMY(3)	READ3320

~		GO TO 9999	READ3330
C			READ3340
	200	CONTINUE	READ3350
		JKONT (JJJ)=JKONT (JJJ)+1	READ3360
		II=JKONT(JJJ)	READ3370
		XFP(II) =DUMMY(1)	READ3380
		YFP(II) =DUMMY(2)	READ3390
		ZFP(II) =DUMMY(3)	READ3400
		GO TO 9999	READ3410
С			READ3420
	300	CONTINUE	READ3430
		JTNOXL=(LLL)TNOXL=	READ3440
		II=JKONT(JJJ)	READ3450
		XMSCB(II) = DUMMY(1)	READ3460
		YMSCB(II) =DUMMY(2)	READ3470
		ZMSCB(II) =DUMMY(3)	READ3480
		GO TO 9999	READ3490
С			READ3500
	24	CONTINUE	READ3510
		KONT (NCARD) = KONT (NCARD) + 1	READ3520
		DO 7005 I=1.5	READ3530
7	005		PEAD2540
'	002		DEAD3550
c			
C	25	CONTINUE	READ3500
	22	$V_{0}$	PEAD2580
			READ9500
7	004		READ3590
1	006	= DOMMT(1)	READ3600
r		G0 T0 9999	READSOLU
C	26	CONTINUE	READ3620
	26		READ3630
		RONT(NCARD) = RONT(NCARD) + 1	READ3640
_			READ3650
1	007	CDCMS(1) = DOMMY(1)	READ3660
~		60 10 9999	READ3670
C			READ3680
	27	CONTINUE	READ3690
		KONI(NCARD)=KONI(NCARD)+1	READ3700
-		DO 7008 I=1,5	READ3710
1	008	CDTMS(1) = DUMMY(1)	READ3720
		GO TO 9999	READ3730
C			READ3740
	28	CONTINUE	READ3750
		KONI (NCARD)=KONI (NCARD)+1	READ3760
_		DO 7009 I=1,5	READ3770
(	009	SRCMS(I) =DUMMY(I)	READ3780
		GO TO 9999	READ3790
С			READ3800
	29	CONTINUE	READ3810
		KONT(NCARD)=KONT(NCARD)+1	READ3820
		DO 7010 I=1,5	READ3830
7	010	SRTMS(I) =DUMMY(I)	READ3840
-		GO TO 9999	READ3850
С			READ3860
	30	CONTINUE	READ3870
		KONT(NCARD)=KONT(NCARD)+1	READ3880

•

```
SCMXMS
                     =DUMMY(1)
      STMXMS
                     = DUMMY(2)
      SRUCMS
                     = DUMMY(3)
      SRUTMS
                     = DUMMY(4)
      GO TO 9999
С
   31 CONTINUE
      KONT (NCARD) = KONT (NCARD) +1
      IRETMS
                     =DUMMY(1)
      GO TO 9999
C
   32 CONTINUE
      KONT(NCARD)=KONT(NCARD)+1
                     =DUMMY(1)
      FRICMS
                     = DUMMY(2)
      COEFMS
                     = DUMMY(3)
      GAMMS
      AEII
                     =DUMMY(4)
                     = DUMMY(5)
      AEI2
      GO TO 9999
с
  400 CONTINUE
      JKONT(JJJ)=JKONT(JJJ)+1
      II = JKONT(JJJ)
      XDSCB(II)
                     =DUMMY(1)
      YDSCB(II)
                     =DUMMY(2)
                     =DUMMY(3)
      ZDSCB(II)
      GO TO 9999
С
   33 CONTINUE
       KONT(NCARD)=KONT(NCARD)+1
      DO 7011 I=1,5
 7011 PFCDS(I)
                     =DUMMY(I)
      GO TO 9999
С
    34 CONTINUE
       KONT(NCARD)=KONT(NCARD)+1
       DO 7012 I=1,5
                      =DUMMY(I)
 7012 PFTDS(I)
       GO TO 9999
С
    35 CONTINUE
       KONT(NCARD)=KONT(NCARD)+1
       DO 7013 I=1,5
                      =DUMMY(I)
 7013 CDCDS(I)
       GO TO 9999
С
    36 CONTINUE
       KONT(NCARD)=KONT(NCARD)+1
       DO 7014 I=1,5
  7014 CDTDS(I)
                      =DUMMY(I)
       GO TO 9999
С
    37 CONTINUE
       KONT(NCARD)=KONT(NCARD)+1
       DO 7015 I=1,5
  7015 SRCDS(I)
                      =DUMMY(I)
```

**READ3890 READ3900** RFAD3910 READ3920 **READ3930 READ3940** READ3950 **READ3960** READ3970 READ3980 **READ3990 READ4000** READ4010 **READ4020** READ4030 READ4040 READ4050 READ4060 READ4070 **READ4080 READ4090 READ4100** READ4110 **READ4120 READ4130** READ4140 **READ4150 READ4160 READ4170 READ4180 READ4190 READ4200** READ4210 **READ4220 READ4230** READ4240 READ4250 **READ4260 READ4270 READ4280 READ4290 READ4300 READ4310 READ4320 READ4330 READ4340 READ4350 READ4360** READ4370 **READ4380 READ4390 READ4400** READ4410 **READ4420** READ4430 **READ4440** 

READ4450 GO TO 9999 С READ4460 READ4470 38 CONTINUE KONT(NCARD)=KONT(NCARD)+1 READ4480 READ4490 DO 7016 I=1,5 7016 SRTDS(I) =DUMMY(I) READ4500 GO TO 9999 READ4510 С **READ4520 39 CONTINUE** READ4530 READ4540 KONT(NCARD)=KONT(NCARD)+1 READ4550 SCMXDS =DUMMY(1) =DUMMY(2) READ4560 STMXDS SRUCDS = DUMMY(3) READ4570 SRUTDS =DUMMY(4) READ4580 GO TO 9999 READ4590 **READ4600** С READ4610 40 CONTINUE KONT(NCARD)=KONT(NCARD)+1 READ4620 READ4630 IRETDS =DUMMY(1) GO TO 9999 READ4640 READ4650 С **41 CONTINUE** READ4660 KONT(NCARD)=KONT(NCARD)+1 READ4670 READ4680 FRICDS =DUMMY(1) READ4690 = DUMMY(2) COEFDS GAMDS =DUMMY(3) **READ4700** READ4710 GO TO 9999 С **READ4720 READ4730** 42 CONTINUE KONT(NCARD)=KONT(NCARD)+1 READ4740 **READ4750** MODEIN =DUMMY(1) **READ4760** GO TO 9999 **READ4770** С **READ4780** 500 CONTINUE **READ4790** JKONT(JJJ)=JKONT(JJJ)+1 II=JKONT(JJJ) READ4800 READ4810 =DUMMY(1) GM(II) =DUMMY(2) READ4820 OMEGA(II) **READ4830** GO TO 9999 READ4840 С READ4850 600 CONTINUE RFAD4860 JKONT(JJJ)=JKONT(JJJ)+1 **READ4870** II=JKONT(JJJ) WNX(II) =DUMMY(1) READ4880 **READ4890** WNY(II) =DUMMY(2) WNZ(II) =DUMMY(3) **READ4900** READ4910 GO TO 9999 С READ4920 READ4930 700 CONTINUE JKONT(JJJ)=JKONT(JJJ)+1 READ4940 **READ4950** II=JKONT(JJJ) **READ4960** PX(II) =DUMMY(1) **READ4970** = DUMMY(2) PY(II) =DUMMY(3) READ4980 PZ(II) READ4990 GO TO 9999 **READ5000** С

411

800	CONTINUE JKONT(JJJ)=JKO II=JKONT(JJJ) DO 7017 I=1,5 PMSX(II,I)	=DUMMY(I)	
C 900	GO TO 9999 CONTINUE	NT()+1	
7018	II=JKONT(JJJ) DO 7018 I=1,5 PMSY(II,I) GO TO 9999	=DUMMY(I)	
C 1000	CONTINUE JKONT(JJJ)=JKC II=JKONT(JJJ)	1+(LLL)TN	
7019 C	DO 7019 I=1,5 PMSZ(II,I) GO TO 9999	=DUMMY(I)	
1100	CONTINUE JKONT(JJJ)=JKO II=JKONT(JJJ)	DNT(JJJ)+1	
7020 C	PDSX(II,I) GO TO 9999	=DUMMY(I)	
1200	CONTINUE JKONT(JJJ)=JKO II=JKONT(JJJ) D0 7021 I=1,5	DNT(JJJ)+1	
7021 C	PDSY(II,I) GO TO 9999	=DUMMY(I)	
1300	CONTINUE JKONT(JJJ)=JKO II=JKONT(JJJ) DO 7022 J=1-5	DNT(JJJ)+1	
7022	PDSZ(II,I) GO TO 9999	=DUMMY(I)	
1400	CONTINUE JKONT(JJJ)=JKO II=JKONT(JJJ) D0 7023 I=1-5	1+(LLL)TAC	
7023 C	POUTX(II,I) GO TO 9999	=DUMMY(I)	
<u>ັ</u> 1500	CONTINUE JKONT(JJJ)=JKO II=JKONT(JJJ) DO 7024 I=1,5	DNT(JJJ)+1	
7024 C	POUTY(II,I) GO TO 9999	=DUMMY(I)	

READ5010 **READ5020 READ5030** READ5040 **READ5050 READ5060 READ5070** READ5080 READ5090 READ5100 READ5110 READ5120 READ5130 READ5140 READ5150 READ5160 READ5170 READ5180 **READ5190 READ5200** READ5210 READ5220 READ5230 READ5240 **READ5250 READ5260** READ5270 **READ5280** READ5290 **READ5300** READ5310 **READ5320** READ5330 **READ5340** READ5350 READ5360 **READ5370 READ5380 READ5390 READ5400 READ5410 READ5420 READ5430** READ5440 READ5450 READ5460 **READ5470 READ5480** READ5490 **READ5500** READ5510 READ5520 **READ5530** READ5540 READ5550 READ5560

٥

.

1600	JKONT(JJJ)=JKONT(JJJ)+1	READ5570 READ5580
	II=JKONT(JJJ)	READ5590
	DO 7025 I=1,5	READ5600
7025	POUTZ(II,I) =DUMMY(I)	READ5610
	GO TO 9999	READ5620
С		READ5630
1700	CONTINUE	READ5640
	JHONT(JJJ)=JKONT(JJJ)+1	READ5650
	II=JKONT(JJJ)	READ5660
	PCGX(II) =DUMMY(1)	READ5670
	PCGY(II) =DUMMY(2)	READ5680
	PCGZ(II) =DUMMY(3)	READ5690
	GO TO 9999	READ5700
С		READ5710
C	CHECK INPUT DATA	READ5720
C		READ5730
7999	CONTINUE	READ5740
	DO 8000 I=1,42	READ5750
	IF (KONT(I) • EQ • 1) GO TO 8000	READ5760
	IF (KONT(I) EQ.0. AND. JTEST. EQ.1) GO TO 8000	READ5770
	WRIE(6,8001)1	READ5780
0000		READ5790
8000	CONTINUE	READ5800
8001	PORMAT(745H INCORRECT NUMBER OF DATA CARDS - CARD NUMBER, 15)	READ5810
		READ5820
	$\prod_{i=1}^{n} \prod_{j=1}^{n} \prod_{i=1}^{n} \prod_{j=1}^{n} \prod_{j=1}^{n} \prod_{j=1}^{n} \prod_{i=1}^{n} \prod_{j=1}^{n} \prod_{j$	READ5830
	GO = IO - (8005 + 8003 + 8003 + 8005 + 8005 + 8006 + 8006 + 8012 + 1000000000000000000000000000000000	READ5840
:	8012,8012,8013,8013,8013,8015,1 I	PEADERAD
8002	LE(KONT(I)=E0-NOOUI) 60 TO 8007	READ5870
0002		READ5880
	WRITE (6+8008) II	READ5890
	GO TO 8007	READ5900
8003	LE(JKONT(L) = EQ - NOLEG) GO TO 8007	READ5910
0000	NOGOTI	READ5920
	WRITE (6,8008) II	READ5930
	GO TO 8007	READ5940
8004	IF(JKONT(I)•EQ•2*NOLEG) GO TO 8007	READ5950
	NOGO=1	READ5960
	WRITE(6,8008)II	READ5970
	GO TO 8007	READ5980
8005	IF(JKONT(I)•EQ•MODEIN) GO TO 8007	READ5990
	NOGO=1	READ6000
	WRITE(6,8008)II	READ6010
	GO TO 8007	READ6020
8006	IF(MODEIN•EQ•0•AND•JKONT(I)•EQ•0) GO TO 8007	READ6030
	IF(JKONT(I)•EQ•NOLEG) GO TO 8007	READ6040
	NOGO=1	READ6050
	WRITE(6,8008)II	READ6060
	GO 10 8007	READ6070
8012	IF (MODEIN • EQ. 0 • AND • JKONT (I) • EQ.0) GO TO 8007	READ6080
	IF(JKUNI(I)+EQ+2*NOLEG) GO TO 8007	READ6090
		READ6100
	WKIIE(0)0UU0)II	READ6110
		READ6120

8013 IF(MODEIN.EQ.O.AND.JKONT(I).EQ.O) GO TO 8007 READ6130 IF(JKONT(I).EQ.NOOUT) GO TO 8007 READ6140 NOGO=1 READ6150 WRITE(6,8008)II READ6160 8007 CONTINUE READ6170 8008 FORMAT(/64H DATA DECK DOES NOT AGREE WITH CONTROL INFORMATION - CAREAD6180 READ6190 *RD NUMBER 15) IF(NMODES.LE.MODEIN)GO TO 8011 READ6200 READ6210 NOGO=1WRITE(6,8010) READ6220 8010 FORMAT(/24H INCONSISTENT MODAL DATA ) READ6230 READ6240 8011 CONTINUE WRITE(6,8009)(NOCARD(I), I=1, MMM) READ6250 8009 FORMAT(/27H NEW CARDS READ THIS RUN - 20(1X,I4)/(27X,20(1X,I4))) READ6260 **READ6270** С READ6280 CALL DATAOT READ6290 С **READ6300** IF(NOGO.EQ.1) STOP READ6310 JTEST=1 READ6320 С READ6330 IF ( IDSETN .EQ. 0 ) GO TO 4010 READ6340 С IF ANOTHER DATA CASE IS INDICATED (IDSETN .NE. 0) THEN SAVE С READ6350 С BLANK COMMON AND THISV COMMON FOR REINITIALIZATION BEFORE READ6360 С **READ6370** THE NEXT DATA CASE IS READ READ6380 С READ6390 REWIND 4 READ6400 WRITE (4)(COMINT(I), I=1, 1400), (TIMHSA(I), I=1, 275) READ6410 ISAVCM = 1READ6420 4010 CONTINUE С READ6430 THIS PORTION OF THE PROGRAM INITIALIZES THE PROGRAM VARIABLES C C READ6440 READ6450 C DO NOT LET TIMAX .EQ. 0 READ6460 С READ6470 READ6480 IF ( TIMAX .EQ. 0.0 ) TIMAX = HMAX+HMAX READ6490 TIME=0.0 ISTAB=0 READ6500 IBOTM=0 READ6510 IF(INLEG.EQ.1)CUTERR=100.*GRAVE READ6520 IF(NFORC.NE.O) REWIND 3 READ6530 C C C C READ6540 INITIALIZATION FOR SUBROUTINE SOIL **READ6550** READ6560 READ6570 DO 62 I=1,3 62 ATTHCK(I)=SS(1)-ATTHCK(I) READ6580 DO 63 I=1,NOLEG READ6590 **READ6600** IMPACT(I)=0 READ6610 IPRTFP(I)=IPTCNT KOUNT(I)=0 READ6620 READ6630 63 ATTH(I)=SS(1) READ6640 ADIST=ATTHCK(3) READ6650 RADIAN=57.295779513 IF(NTYPE.NE.O) GO TO 68 READ6660 SOILP(1)=SOILP(1)/RADIAN READ6670 **READ6680** TANPHI=TAN(SOILP(1))

	CMS=29•*EXP(1•4*SOILP(3))*TANPHI	READ6690
	SLRHO=SUILP(2)/GRAVE	READ6700
	TANPH2=TAN(SOILP(1)/2.)	READ6710
	SOILNU=(3.14159265/3.)*(1.+TANPH2)/(1TANPH2)	READ6720
	IF(SOILP(3).GE.0.5) GO TO 67	READ6730
	CDCONT=(GRAV/GRAVE)*(4++80+*SOILP(3))*TANPHI/(RAD(4)**2)	READ6740
	GO TO 68	READ6750
67	CDCONT=4.*(GRAV/GRAVE)*EXP(4.83*SOILP(3))*TANPHI/(RAD(4)**2)	READ6760
68	CONTINUE	READ6770
•••		READ6780
	INITIALIZE THE CENTER BODY MASS MATRIX	READ6790
		READ6800
	DO 70 I = 1 + 6	READ6810
	DO 70 J = 1, 6	READ6820
70	$AM(I_{2},I) = 0.0$	READ6830
10	$AM(1,s_1) = CBMASS$	READ6840
	$AM(2 \cdot 2) = CBMASS$	READ6850
	AM(3,3) = CBMASS	READ6860
	$AM(4 \cdot 4) = CBIXX$	READ6870
	$AM(5 \cdot 5) = CBIYY$	READ6880
	AM(6,6) = CB177	READ6890
	AM(4,5) = -CBIZY	READ6900
	AM(4,6) = -CBIXZ	READ6910
	AM(5,6) = -CBIYZ	READ6920
	AM(5,4) = -CBIXY	READ6930
	AM(6,4) = -CBIXZ	READ6940
	AM(6,5) = -CBIYZ	READ6950
		READ6960
	INITIALIZATION FOR SUBROUTINE STRUT	READ6970
		READ6980
	STRUT LENGTHS	READ6990
		RFAD7000
	DO 90 I=1.NOLEG	READ7010
	DX = XMSCB(I) - XFP(I)	READ7020
	DY = YMSCB(I) - YFP(I)	READ7030
	DZ = ZMSCB(I) - ZFP(I)	READ7040
	SI OMS(I)=SQRI(DX*DX+DY+DY+DZ*DZ)	READ7050
	SLNGMS(I) = SLOMS(I)	READ7060
	DO 90 J=1.2	READ7070
	NNN=2*(1-1)+J	READ7080
	IF(ILEG•EQ•0) GO TO 84	READ7090
	CDX=DX/SLOMS(I)	READ7100
	CDY=DY/SLOMS(I)	READ7110
	CDZ=DZ/SLOMS(I)	READ7120
	DFL=SLOMS(I)-DRAGST	READ7130
	DDX=XDSCB(NNN)-(XFP(I)+CDX*DEL)	READ7140
	DDY=YDSCB(NNN)-(YFP(I)+CDY*DEL)	READ7150
	DDZ=ZDSCB(NNN)-(ZFP(I)+CDZ*DEL)	READ7160
	GO TO 85	READ7170
84	CONTINUE	READ7180
	DDX=XDSCB(NNN)-XFP(I)	READ7190
	DDY=YDSCB(NNN)-YFP(I)	READ7200
	DDZ=ZDSCB(NNN)-ZFP(I)	READ7210
85	SLODS(NNN)=SQRT(DDX*DDX+DDY*DDY+DDZ*DDZ)	READ7220
	SLNGDS(NNN)=SLODS(NNN)	READ7230
90	CONTINUE	READ7240

415

4

 $\circ \circ \circ \circ \circ$ 

с с с

C		MAIN CTRUTC		R	EAD7250
Ċ		MAIN STRUIS			EAD 7200
C		DO 97 1-1 NOLEC		א מ	EADT210
		STROME(1)=0.0			EAD7200
					EAD7200
				P	EAD7310
					EAD7320
		SETTMS(I)=0.0		R	FAD7330
		PRECMS(I)=0.0		R	FAD7340
				R	EAD7350
		ERVMSC(I)=0.0		R	EAD7360
		FRVMST(I)=0.0		R	EAD7370
		IPOCMS(I)=1		F	EAD7380
		IPOTMS(I)=1		R	EAD7390
		INDCMS(I)=0		F	EAD7400
		INDTMS(I)=0		F	EAD7410
		IPRMS(I)=0		F	EAD7420
	87	CONTINUE		R	EAD7430
С				A	EAD7440
С		DRAG STRUTS		A	EAD7450
С				F	READ7460
		II=2*NOLÉG		F	₹EAD7470
		DO 86 I=1,II		F	₹EAD7480
		STRPDS(I)=0.0		F	READ7490
		URCDS(I)=0.0		F	READ7500
		URTDS(I)=0.0		F	READ7510
		SETCDS(I)=0.0		ŀ	READ 7520
		SETTDS(I)=0.0		ŀ	READ7530
		PRFCDS(I)=0.0			READ7540
		PRFTDS(I)=0.0		F	READ 7550
		FRVDSC(I)=0.0			KEAD 7560
		FRVDST(1)=0.0		ł	KEAD 7570
		IPOCDS(I)=1		F	CEAD 7580
		IPOTDS(1)=1		ł	(EAD 7590
		INDCDS(I)=0		ł	(EAD 7600
				t i i i i i i i i i i i i i i i i i i i	CEAU / 010
	~ ~	IPRDS(I)=0		r	
c	86	CONTINUE		1	
ĉ	88	CONVERT ANGLUAR	QUANTITIES TO RADIANS	1	READ7650
ĉ		CONVERT ANODEAR	doanti i i con la la la la la la la la la la la la la		READ7660
C					READ7670
					READ7680
					READ7690
			1		READ7700
		DO 88 I=1.NMODES			READ7710
		OMEGA(I) = OMEGA(I)	)*2.*3.14159265		READ7720
c					READ7730
ĉ		INITIALIZE ANGUL	AR QUANTITIES		READ7740
c					READ7750
		COX=COS(ANGX/RAD	DIAN)		READ7760
		SIX=SIN(ANGX/RAD	DIAN)		READ7770
		COY=COS(ANGY/RAD	DIAN)		READ7780
		SIY=SIN(ANGY/RAD	DIAN)		READ7790
		COZ=COS(ANGZ/RAD	DIAN)		READ7800

.

	SIZ=SIN(ANGZ/RADIAN)	READ7810
	SIZT=SIN(ZETA)	READ7820
	COZT=COS(ZETA)	READ7830
	THTA=ASIN(~SIZT*COZ*COY~COZT*(SIZ*SIX~COZ*SIY*COX))	READ7840
	AAN=-SIZT*SIZ*COY+COZT*(COZ*SIX+SIZ*SIY*COX)	READ7850
	BBD=SIZT*SIY+COZT*COY*COX	READ7860
	IF(ABS(BBD).LT.1.E-09)GO TO 104	READ7870
	PHI=ATAN2(AAN,BBD)	READ7880
	GO TO 101	READ7890
1	04 IF(AAN.GT.0.0)PHI=+3.14159265/2.	READ7900
	IF(AAN.LT.0.0)PHI=-3.14159265/2.	READ7910
1	01 CONTINUE	READ7920
	AAN=SIZ*COX+COZ*SIY*SIX	READ7930
	BBD=COZT*COZ*COY-SIZT*(SIZ*SIX-COZ*SIY*COX)	READ7940
	IF(ABS(BBD).LT.1.E-09)GO TO 102	READ7950
	PSI=ATAN2(AAN,BBD)	READ7960
	GO TO 103	READ7970
1	02 IF(AAN.GT.0.0)PSI=+3.14159265/2.	READ7980
	IF(AAN.LT.0.0)PSI=-3.14159265/2.	READ7990
1	03 CONTINUE	READ8000
C		READ8010
С	INITIALIZE VELOCITIES	READ8020
С		READ8030
	ZETA=-ZETA	READ8035
	GSINZT=SIN(ZETA)*GRAV	READ8040
	GCOSZT=COS(ZETA)*GRAV	READ8050
	XSD=VELX*COS(ZETA)+VELZ*SIN(ZETA)	READ8060
	YSD=VELY	READ8070
	ZSD=-VELX*SIN(ZETA)+VELZ*COS(ZETA)	READ8080
	RETURN	READ8090
	END	READ8100

.
SUBROUTINE DATAOT	DATA	10
	DATA	20
THIS SUBROUTINE LISTS THE INPUT DATA	DATA	30
	DATA	40
DIMENSION NOTION; NOMB(25)	DATA	50
COMMON / THISV/ TIMHSA(1), ATTH(5),AM(6.6),AME1(3),INDERI(6),	DATA	70
1 INDEPC(6).	DATA	80
2 STRPDS(10), STRPMS( 5), IPOCDS(10), IPOCMS( 5), URCDS (10),	DATA	90
3 URTDS (10), URCMS ( 5), URTMS ( 5), SETCDS(10), SETTDS(10),	DATA	100
4 SETCMS( 5), SETTMS( 5), INDCDS(10), INDTDS(10), INDCMS(10),	DATA	110
5 INDTMS( 5), PRFCDS (10), PRFTDS (10), PRFCMS ( 5), PRFTMS ( 5),	DATA	120
6 IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5),	DATA	130
L FRVMST(5), IPOTDS(10), IPOTMS(5)	DATA	140
EQUIVALENCE ( NFIPDS; NOLEG )	DATA	150
COMMON COMINITADO		170
FOULVALENCE ( COMINIC 4 ), YSD )	DATA	180
EQUIVALENCE ( COMINT( 28 ) ZSD )	DATA	190
EQUIVALENCE ( COMINT( 36 ), PHI )	DATA	200
EQUIVALENCE ( COMINT( 44 ), WX )	DATA	210
EQUIVALENCE ( COMINT( 52 ), THTA )	DATA	220
EQUIVALENCE ( COMINT( 60 ), WY )	DATA	230
EQUIVALENCE ( COMINT( 68 ), PSI )	DATA	240
EQUIVALENCE ( COMINT( 76 ), WZ )	DATA	250
EQUIVALENCE ( COMINT( 324 ), TIME )	DATA	260
EQUIVALENCE ( COMINI( 325 ), HMAN )	DATA	280
FOUVALENCE ( COMINT( 320 )) FMIN )	DATA	290
EQUIVALENCE ( COMINT( 328 ) EMAX )	DATA	300
EQUIVALENCE ( COMINT ( 337 ), HZ )	DATA	310
EQUIVALENCE ( COMINT( 338 ), CUTERR)	DATA	320
EQUIVALENCE ( COMINT( 339 ), IP )	DATA	330
EQUIVALENCE ( COMINT( 340 ), IVARH )	DATA	340
EQUIVALENCE ( COMINT( 341 ), IMTH )	DATA	350
EQUIVALENCE ( COMINT( 342 ), IPRNT )	DATA	360
EQUIVALENCE ( COMINI( 343 ), IFIN )	DATA	370
EQUIVALENCE ( COMINI( 361 )) IPICNI )	DATA	380
FOULVALENCE ( COMINT ( 1392), NOCASE )		400
COMMON	DATA	410
1 CBMASS, CBIXX, CBIXZ, CBIYY, CBIYZ, CBIZZ, FPMASS, CBIXY,	DATA	420
2 DC(3,3) , XFP(5), YFP(5), ZFP(5), WNX(5), WNY(5),	DATA	430
3 WNZ(5), PX(5), PY(5), PZ(5), GM(5), OMEGA(5),	DATA	440
4 GRAV , GRAVE , ZETA , FTS (6) , FSXSI(5) ,	DATA	450
5 FSYSI(5) , FSZSI(5) , SOILX(5) ,	DATA	460
$\begin{array}{cccc} 6 & \text{SOLY}(5) & \text{, SOLZ}(5) & \text{, PMSX}(5,5) & \text{,} \\ \end{array}$	DATA	4/0
( MMST(5)5) ) MMSZ(5)5) ) MUSX(10)5) ) C DNSV(10.5) DNS7(10.5) - FLVS - FLVS - FLVS		480
0 FUSICIUSDE S FUSICIUSDE S FLAS S FLAS S FLAS S	DATA	500
C YMSCB(5) • ZMSCB(5) • XDSCB(10) •	DATA	510
D YDSCB(10) , ZDSCB(10) , ILEG , IMS ,	DATA	520
E FSTX , FSTY , FSTZ , PVCBX , PVCBY , PVCBZ ,	DATA	530
F PVFPX , PVFPY , PVFPZ , NOLEG , SLOMS(5) ,	DATA	540
G SLODS(10)	DATA	550

.

5

c c c c

	COMMON	· · · · · · · · · · · · · · · · · · ·		DATA	560
2	PFCMS(5) , PFCDS(5)	, PFTDS(5) ,		DATA	570
2	PFTMS(5) , SRCDS(5)	, SRCMS(5) ,		DATA	580
3	SRTDS(5) SRTMS(5)	, COEFDS, COEFMS,	GAMDS ,	DATA	590
4	GAMMS , SRUCDS, SRUCMS, SRUTDS	, SRUTMS, SCMXDS,	SCMXMS,	DATA	600
	5 STMXDS, STMXMS, CDCDS(5)	, CDCMS(5) ,		DATA	610
e	CDTDS(5) , CDTMS(5)	, FRICDS, FRICMS,	IRETDS,	DATA	620
-	7 IRETMS, STROKE(10) , STRKDS	(10) , STRKMS(	5) 🦻	DATA	630
8	LENGTH, CDXS, CDYS, CDZS, NTYPE	, RAD(4), SS(4),	ATTHCK(3),	DATA	640
ç	ATTPRS(3) , ADIST , SOILP(	3). , NMODES,	<b>—</b>	DATA	650
/	A COEF , GAMMA , FRIC , SCMAX	, STMAX , CDC(5),	CDT(5),	DATA	660
ŧ	B SRULT , SRULC , SRT(5), SRC(5)	, PFT(5), PFC(5),	GFLEGS(5),	DATA	670
(	CURMSV, CURMSL, INDEXD, INDEYL	, INDEZD, INDEXR,	INDFYR,	DATA	680
l	D INDEZRO TIMAX O DRAGSTO IEP			DATA	690
	COMMON			DATA	/00
	ICMS, CDCONT, SOILNU, SLRHO, NOOL	T,XOUT(10),YOUI(1	$(10)_{0}$	DATA	/10
	2MODEIN, POUIX(10,5), POUIY(10,5), POU	112(10,5),PCGX(5),	PCGY(5),	DATA	720
	3PCGZ(5),AEI1,AEI2,FORMS(5),FORDS(	0) FORCE, NFORC, SA	VMSX(5),	DATA	730
4	\$\$AVMS2(5), \$AVDSX(10), \$AVDSY(10), \$	VDSZ(10), IQUOUT, G	SINZT,	DATA	740
1	5GCOSZT 9STAB 9STABVL 9ISTAB 9JCKSAB 9V	LX OVELY OVELZ OSAVM	SY(5)	DATA	750
	COMMON			DATA	760
	1 SMXMSC(5), IMXMSC(5), SMXMS	(5), IMXMSI(5),		DATA	780
	Z SMADSC(107) MADSC(107) SMADS	IN CONTRACTOR			700
	S SLNGMS(S) SLNGUS(IU) CORDSL	INLEG & IPPPRIS	4 11 6 7	DATA	190
~ '	4 IMPACI(5) 9IPRIFP(5) & KOUNI(	ANGX, ANGT,	ANGZ	DATA	800
C	DO 5000 I-1 100			DATA	010
5 0 0 0				DATA	020
5000	NO(1) = 1 $DO = 5001  I = 1 \cdot 25$			DATA	020
5001				DATA	040
- 500I	$NOMB(1) = 1 \times 100$			DATA	860
ĉ	PROCRAM CONTROL DATA			DATA	870
c	PROGRAM CONTROL DATA			ΠΔΤΔ	880
C	WEITE (6.101)NO(1) NOCASE			DATA	890
	WRITE(6,102)NO(2), TIMAX, IPTCNT, IN	EG+IFPPRT		DATA	900
	WRITE(6.103)NO(3) • NMODES • NOOUT			DATA	910
	WRITE(6,104)NO(4) INDEXD, INDEYD, I			DATA	920
	WRITE(6,105)NO(5), INDEXR, INDEYR, I	NDFZR		DATA	930
	WRITE(6,106)NO(6) +HMAX +HMIN + FMAX +	MIN. TP		DATA	940
	WRITE(6,107)NO(7), IVARH, IMTH, CUTE	R		DATA	950
	WRITE(6.108)NO(8).NFORC.IQUOUT.JC	SAB IDSETN		DATA	960
	IE (NEORC • NE • O • AND • IDSETN • NE • O ) WRI	F(6,3000)		DATA	970
c		,		DATA	980
č	INITIAL CONDITIONS			DATA	990
Ċ				DATA	1000
C	WRITE(6.109)NO(9).7ETA.GRAV.GRAVE			DATA	1010
	WRITE(6,110)NO(10), ANGX, ANGY, ANG7			DATA	1020
	WRITE(6,111)NO(11) .WX .WY .W7			DATA	1030
	WRITE(6,112)NO(12) VFLX VFLY VFL7			DATA	1040
С				DATA	1050
c	CENTER BODY DATA			DATA	1060
с				DATA	1070
-	WRITE(6,113)NO(13),CBMASS			DATA	1080
	WRITE(6,114)NO(14),CBIXX,CBIYY,CB	IZZ		DATA	1090
	WRITE(6,115)NO(15),CBIXY,CBIXZ,CB	[YZ		DATA	1100
	IF(NOOUT.EQ.0)GO TO 2			DATA	1110

.

```
DO 1 I=1,NOOUT
                                                                      DATA1120
1 WRITE(6,1100)NUMB(1),I,XOUT(I),YOUT(I),ZOUT(I)
                                                                      DATA1130
                                                                      DATA1140
  GO TO 3
2 WRITE(6,1101)
                                                                      DATA1150
3 CONTINUE
                                                                      DATA1160
                                                                      DATA1170
  FOOTPAD DATA
                                                                      DATA1180
                                                                      DATA1190
  WRITE(6,116)NO(16), FPMASS
                                                                      DATA1200
  WRITE(6,117)NO(17),(RAD(I),I=1,4)
                                                                      DATA1210
  WRITE(6,118)NO(18),(SS(I),I=1,4)
                                                                      DATA1220
  WRITE(6,119)NO(19),(ATTHCK(I),I=1,3)
                                                                      DATA1230
                                                                      DATA1240
  WRITE(6,120)NO(20),(ATTPRS(I),I=1,3)
                                                                      DATA1250
                                                                      DATA1260
  SOIL DATA
                                                                      DATA1270
  IF(NTYPE.EQ.0)WRITE(6,121)NO(21),NTYPE
                                                                      DATA1280
  IF(NTYPE.NE.O)WRITE(6,221)NO(21),NTYPE
                                                                      DATA1290
  WRITE(6,122)NO(22),(SOILP(I),I=1,3)
                                                                      DATA1300
                                                                      DATA1310
  LEG DATA
                                                                      DATA1320
                                                                      DATA1330
  WRITE(6,123)NO(23),NOLEG,ILEG,DRAGST
                                                                      DATA1340
                                                                      DATA1350
  DO 4 I=1,NOLEG
4 WRITE(6,1200)NUMB(2),I,XFP(I),YFP(I),ZFP(I)
                                                                      DATA1360
                                                                      DATA1370
  DO 5 I=1.NOLEG
5 WRITE(6,1300)NUMB(3), I, XMSCB(I), YMSCB(I), ZMSCB(I)
                                                                      DATA1380
  WRITE(6,124)NO(24),(PFCMS(I),I=1,5)
                                                                      DATA1390
  WRITE(6,125)NO(25), (PFTMS(I), I=1,5)
                                                                      DATA1400
  WRITE(6,126)NO(26),(CDCMS(I),I=1,5)
                                                                      DATA1410
  WRITE(6,127)NO(27),(CDTMS(I),I=1,5)
                                                                      DATA1420
                                                                      DATA1430
  WRITE(6,128)NO(28),(SRCMS(I),I=1,5)
                                                                      DATA1440
  WRITE(6,129)NO(29),(SRTMS(I),I=1,5)
                                                                      DATA1450
  WRITE(6,130)NO(30), SCMXMS, STMXMS, SRUCMS, SRUTMS
                                                                      DATA1460
  WRITE(6,131)NO(31), IRETMS
  WRITE(6,132)NO(32), FRICMS, COEFMS, GAMMS, AEI1, AEI2
                                                                      DATA1470
  II=2*NOLEG
                                                                      DATA1480
  DO 6 I=1,II
                                                                      DATA1490
6 WRITE(6,1400)NUMB(4),I,XDSCB(I),YDSCB(I),ZDSCB(I)
                                                                      DATA1500
  WRITE(6,133)NO(33),(PFCDS(I),I=1,5)
                                                                      DATA1510
  WRITE(6,134)NO(34), (PFTDS(I), I=1,5)
                                                                      DATA1520
  WRITE(6,135)NO(35),(CDCDS(I),I=1,5)
                                                                      DATA1530
  WRITE(6,136)NO(36),(CDTDS(I),I=1,5)
                                                                      DATA1540
  WRITE(6,137)NO(37),(SRCDS(I),I=1,5)
                                                                      DATA1550
  WRITE(6,138)NO(38),(SRTDS(I),I=1,5)
                                                                      DATA1560
  WRITE(6,139)NO(39),SCMXDS,STMXDS,SRUCDS,SRUTDS
                                                                      DATA1570
  WRITE(6,140)NO(40), IRETDS
                                                                      DATA1580
  wRITE(6,141)NO(41), FRICDS, COEFDS, GAMDS
                                                                      DATA1590
                                                                      DATA1600
  MODAL DATA
                                                                      DATA1610
                                                                      DATA1620
                                                                      DATA1630
  WRITE(6,142)NO(42),MODEIN
  IF(NMODES.EQ.0)WRITE(6,242)
                                                                      DATA1640
  IF(MODEIN.EQ.0)RETURN
                                                                      DATA1650
                                                                      DATA1660
  DO 8 I=1,MODEIN
8 WRITE(6,1500)NUMB(5),I,GM(I),OMEGA(I)
                                                                      DATA1670
```

C C

С

С

С

С

С

С

С

C C

C

			DATA1680
	~		DATA1690
	9	WRITE (051000)NOMD(0) 51 9WNA(1) 9WNT(1) 9WNZ(1)	DATA1700
		DO IO I=I,MODEIN	DATAITOO
	10	WRITE(6,1700)NUMB(7), I, PX(I), PY(I), PZ(I)	DATA1710
		DO 11 I=1,NOLEG	DATA1720
	11	WRITE(6 + 1800)NUMB(8) + I + (PMSX(I + J) + J = 1 + MODEIN)	DATA1730
		DO 12 I=1,NOLEG	DATA1740
	12	WRITE(6,1900)NUMB(9),I,(PMSY(I,J),J=1,MODEIN)	DATA1750
		DO 13 I=1-NOLEG	DATA1760
	12		DATA1770
	1 9		DATA1780
	1 /		DATA1700
	14	WKITE(692100)NOMD(11)919(PDSA(193)93=19MODEIN)	DATA1750
			DATA1000
	15	WRITE(6,2200)NOMB(12), I, (PDSY(I,J), J=I, MODEIN)	DATAIOIU
		DO 16 I=1,II	DATA1820
	16	WRITE(6,2300)NUMB(13),I,(PDSZ(I,J),J=1,MODEIN)	DATA1830
		IF(NOOUT.EQ.0)GO TO 21	DATA1840
		DO 17 I=1,NOOUT	DATA1850
	17	WRITE(6,2400)NUMB(14),I,(POUTX(I,J),J=1,MODEIN)	DATA1860
		DO 18 I=1,NOOUT	DATA1870
	18	WRITE $(6, 2500)$ NUMB $(15)$ , I, (POUTY $(I, J)$ , J=1, MODEIN)	DATA1880
		DO 19 I=1 NOOUT	DATA1890
	19	WRITE $(6, 2600)$ NUMB $(16)$ , $I_{2}(POUTZ(I_{2}J), J=1, MODEIN)$	DATA1900
	21		DATA1910
	21		DATA1020
		DU 20 1-13MODEIN	DATA1920
	20	WRITE(6,2700)NUMB(17),1,PCGX(1),PCGY(1),PCGZ(1)	DATA1950
		RETURN	DATA1940
r			
C			DATA1950
c		FORMAT STATEMENTS	DATA1950
c c		FORMAT STATEMENTS	DATA1950 DATA1960 DATA1970
c c	101	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//,	DATA1950 DATA1960 DATA1970 DATA1980
c c	101	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6)	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990
C C C	101	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2,	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990 DATA2000
	101 102	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2)	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990 DATA2000 DATA2010
	101 102	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2)	DATA1950 DATA1960 DATA1970 DATA1980 DATA1980 DATA2000 DATA2010 DATA2020
	101 102 103	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDEXD = ,I2,10H INDEZD = ,I2)	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990 DATA2000 DATA2010 DATA2020 DATA2030
	101 102 103 104	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2)	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990 DATA2000 DATA2010 DATA2010 DATA2030 DATA2040
	101 102 103 104 105	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2)	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990 DATA2000 DATA2010 DATA2020 DATA2030 DATA2050 DATA2050
	101 102 103 104 105 106	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4, 8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3,	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990 DATA2000 DATA2010 DATA2020 DATA2020 DATA2030 DATA2050 DATA2050
	101 102 103 104 105 106	FORMAT STATEMENTS FORMAT (5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4, 8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,F10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4, 0H LWADH = ,I2)	DATA1950 DATA1960 DATA1970 DATA1980 DATA1980 DATA2000 DATA2000 DATA2010 DATA2020 DATA2030 DATA2030 DATA2040 DATA2050 DATA2060
	101 102 103 104 105 106	FORMAT STATEMENTS FORMAT (5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3)	DATA1950 DATA1960 DATA1970 DATA1970 DATA2000 DATA2000 DATA2010 DATA2020 DATA2030 DATA2030 DATA2050 DATA2050 DATA2060 DATA2070
	101 102 103 104 105 106 107 108	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,16) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2,	DATA1950 DATA1960 DATA1970 DATA1970 DATA1990 DATA2000 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2050 DATA2050 DATA2060 DATA2070 DATA2080
	101 102 103 104 105 106 107 108	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,16) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,12,9H INLEG = ,12, * 10H IFPPRT = ,12) FORMAT(I4,10H NMODES = ,12, 9H NOOUT = ,12) FORMAT(I4,10H INDFXD = ,12,10H INDFYD = ,12,10H INDFZD = ,12) FORMAT(I4,10H INDFXR = ,12,10H INDFYR = ,12,10H INDFZR = ,12) FORMAT(I4,8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,12) FORMAT(I4,9H IVARH = ,12,8H IMTH = ,12,10H CUTERR = ,E10.3) FORMAT(I4,9H NFORC = ,12,10H IQUOUT = ,12,10H JCKSAB = ,12, *10H IDSETN = ,12)	DATA1950 DATA1960 DATA1970 DATA1980 DATA1980 DATA2000 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2050 DATA2050 DATA2060 DATA2080 DATA2080
C C C	101 102 103 104 105 106 107 108	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(//10X,26H* * INITIAL CONDITIONS * *//,I4, 8H ZETA = ,E10.3	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990 DATA2010 DATA2010 DATA2010 DATA2030 DATA2030 DATA2030 DATA2040 DATA2050 DATA2050 DATA2060 DATA2070 DATA2080 DATA2090 DATA2100
	101 102 103 104 105 106 107 108	<pre>FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2,9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,F10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(//10X,26H* * INITIAL CONDITIONS * *//,I4, 8H ZETA = ,E10.3 * 8H GRAV = ,E10.3, 9H GRAVE = ,E10.3)</pre>	DATA1950 DATA1960 DATA1970 DATA1980 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2030 DATA2040 DATA2050 DATA2050 DATA2050 DATA2060 DATA2070 DATA2080 DATA2090 DATA2100 DATA2110
	101 102 103 104 105 106 107 108 109	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4, 8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,F10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(//10X,26H* * INITIAL CONDITIONS * *//,I4, 8H ZETA = ,E10.3 * , 8H GRAV = ,E10.3, 9H GRAVE = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,8H ANGY = ,E10.3,8H ANGZ = ,E10.3)	DATA1950 DATA1960 DATA1970 DATA1980 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2030 DATA2030 DATA2050 DATA2050 DATA2050 DATA2050 DATA2050 DATA2070 DATA2070 DATA2100 DATA2100 DATA2120
	101 102 103 104 105 106 107 108 109	<pre>FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3,8H HMIN = ,E10.3,8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(/10X,26H* * INITIAL CONDITIONS * *//,I4, 8H ZETA = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,8H ANGY = ,E10.3,8H ANGZ = ,E10.3) FORMAT(I4,6H WX = ,E10.3,6H WY = ,E10.3,6H WZ = ,E10.3)</pre>	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2050 DATA2050 DATA2050 DATA2070 DATA2090 DATA2000 DATA2100 DATA2100 DATA2120 DATA2130
	101 102 103 104 105 106 107 108 109 109	<pre>FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3,8H HMIN = ,E10.3,8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(//10X,26H* * INITIAL CONDITIONS * *//,I4, 8H ZETA = ,E10.3 * , 8H GRAV = ,E10.3, 9H GRAVE = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,6H WY = ,E10.3,6H WZ = ,E10.3) FORMAT(I4,8H VELX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3)</pre>	DATA1950 DATA1960 DATA1970 DATA1970 DATA1990 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2050 DATA2050 DATA2050 DATA2050 DATA2060 DATA2060 DATA2000 DATA2000 DATA2100 DATA2110 DATA2120 DATA2130 DATA2140
	101 102 103 104 105 106 107 108 109 110 111 112 113	<pre>FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,16) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3,8H HMIN = ,E10.3,8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(//10X,26H* * INITIAL CONDITIONS * *//,I4, 8H ZETA = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,9H GRAVE = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,8H ANGY = ,E10.3,8H ANGZ = ,E10.3) FORMAT(I4,8H VELX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3) FORMAT(//10X,24H* * CENTER BODY DATA * *,//,I4,10H CBMASS = ,</pre>	DATA1950 DATA1960 DATA1970 DATA1970 DATA1990 DATA2000 DATA2010 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2050 DATA2050 DATA2050 DATA2050 DATA2050 DATA200 DATA2100 DATA2120 DATA2120 DATA2130 DATA2140 DATA2150
	101 102 103 104 105 106 107 108 107 108 109 110 111 112 113	<pre>FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(//10X,26H* * INITIAL CONDITIONS * *//,I4, 8H ZETA = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,8H ANGY = ,E10.3,8H ANGZ = ,E10.3) FORMAT(I4,8H VELX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3) FORMAT(//10X,24H* * CENTER BODY DATA * *,//,I4,10H CBMASS = , * E10.3)</pre>	DATA1950 DATA1960 DATA1970 DATA1970 DATA1990 DATA2000 DATA2010 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2050 DATA2050 DATA2050 DATA2050 DATA2000 DATA2100 DATA2120 DATA2150 DATA2150 DATA2160
	101 102 103 104 105 106 107 108 107 108 109 110 111 112 113	FORMAT STATEMENTS FORMAT(5h CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,16) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3,8H HMIN = ,E10.3,8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H JCKSAB = ,I2, * 10H IDSETN = ,I2) FORMAT(//10X,26H* * INITIAL CONDITIONS * *//,I4,8H ZETA = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,9H GRAVE = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,8H ANGY = ,E10.3,8H ANGZ = ,E10.3) FORMAT(I4,8H VELX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3) FORMAT(//10X,24H* * CENTER BODY DATA * *,//,I4,10H CBMASS = , * E10.3] FORMAT(//10X,24H * = ,E10.3,9H CBIYY = ,E10.3,9H (BIZZ = ,E10.3)	DATA1950 DATA1960 DATA1970 DATA1970 DATA1990 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2040 DATA2050 DATA2050 DATA2050 DATA2050 DATA2070 DATA2080 DATA2080 DATA2090 DATA2100 DATA2120 DATA2150 DATA2150 DATA2170
	101 102 103 104 105 106 107 108 109 110 111 112 113 114	<pre>FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,16) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(I4,9H ARAX = ,E10.3,9H GRAVE = ,E10.3) FORMAT(I4,9H NERX = ,E10.3,9H GRAVE = ,E10.3,9H VELZ = ,E10.3) FORMAT(I4,9H VELX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) </pre>	DATA1950 DATA1960 DATA1960 DATA1970 DATA1980 DATA2000 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2040 DATA2050 DATA2050 DATA2050 DATA2050 DATA2060 DATA2070 DATA2080 DATA2090 DATA2100 DATA2120 DATA2150 DATA2180
	101 102 103 104 105 106 107 108 109 110 111 112 113 114	<pre>FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,16) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3,8H HMIN = ,E10.3,8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(I4,8H ANGX = ,E10.3,9H GRAVE = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,9H GRAVE = ,E10.3,8H ANGZ = ,E10.3) FORMAT(I4,8H VELX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.</pre>	DATA1950 DATA1960 DATA1960 DATA1970 DATA1980 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2030 DATA2040 DATA2050 DATA2050 DATA2050 DATA2060 DATA2060 DATA2070 DATA2080 DATA2090 DATA2100 DATA2120 DATA2120 DATA2150 DATA2150 DATA2180
	101 102 103 104 105 106 107 108 109 110 111 112 113 114 115 1100	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,16) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H NMODES = ,I2, 9H NOOUT = ,I2) FORMAT(I4,10H INDFXD = ,I2,10H INDFYD = ,I2,10H INDFZD = ,I2) FORMAT(I4,10H INDFXR = ,I2,10H INDFYR = ,I2,10H INDFZR = ,I2) FORMAT(I4, 8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(//10X,26H* * INITIAL CONDITIONS * *//,I4, 8H ZETA = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,8H ANGY = ,E10.3,8H ANGZ = ,E10.3) FORMAT(I4,6H WX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3) FORMAT(I4,8H VELX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3) FORMAT(/10X,24H* * CENTER BODY DATA * *,//,I4,10H CBMASS = , * E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,5H I = ,I2,8H XOUT = ,E10.3,8H YOUT = ,E10.3,8H ZOUT = * E10.3)	DATA1950 DATA1960 DATA1970 DATA1980 DATA1990 DATA2000 DATA2010 DATA2010 DATA2020 DATA2030 DATA2030 DATA2030 DATA2050 DATA2050 DATA2050 DATA2050 DATA2090 DATA2100 DATA2100 DATA2120 DATA2150 DATA2150 DATA2160 DATA2160 DATA2170 DATA2190 DATA2190
	101 102 103 104 105 106 107 108 109 110 111 112 113 114 115	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,I6) FORMAT(I4,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(I4,10H INDEXD = ,I2,9H NOOUT = ,I2) FORMAT(I4,10H INDEXD = ,I2,10H INDEYD = ,I2,10H INDEZD = ,I2) FORMAT(I4,10H INDEXR = ,I2,10H INDEYR = ,I2,10H INDEZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3,8H HMIN = ,E10.3,8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(I4,9H IVARH = ,I2,8H IMTH = ,I2,10H CUTERR = ,E10.3) FORMAT(I4,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(I4,9H ANGX = ,E10.3,9H GRAVE = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,9H GRAVE = ,E10.3,8H ANGZ = ,E10.3) FORMAT(I4,8H ANGX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3) FORMAT(I4,8H VELX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY =	DATA1950 DATA1960 DATA1970 DATA1970 DATA1990 DATA2000 DATA2010 DATA2010 DATA2010 DATA2030 DATA2030 DATA2030 DATA2050 DATA2050 DATA2050 DATA2050 DATA2050 DATA2000 DATA2000 DATA2100 DATA2100 DATA2120 DATA2130 DATA2140 DATA2150 DATA2160 DATA2180 ,DATA2180 DATA2230 DATA2230
	101 102 103 104 105 106 107 108 107 108 109 110 111 112 113 114 115 1100	<pre>FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * I4,10H CASE NO =,16) FORMAT(14,9H TIMAX = ,E10.3,10H IPTCNT = ,I2,9H INLEG = ,I2, * 10H IFPPRT = ,I2) FORMAT(14,10H INDEXD = ,I2,9H NOOUT = ,I2) FORMAT(14,10H INDEXD = ,I2,10H INDEYD = ,I2,10H INDEZD = ,I2) FORMAT(I4,10H INDEXR = ,I2,10H INDEYR = ,I2,10H INDEZR = ,I2) FORMAT(I4,8H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,I2) FORMAT(14,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(14,9H NFORC = ,I2,10H IQUOUT = ,I2,10H JCKSAB = ,I2, *10H IDSETN = ,I2) FORMAT(I4,8H ANGX = ,E10.3,9H GRAVE = ,E10.3) FORMAT(I4,8H NELX = ,E10.3,9H ANGY = ,E10.3,8H ANGZ = ,E10.3) FORMAT(I4,8H VELX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3) FORMAT(I4,8H VELX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(I4,9H CBIXY = ,E10.3,9H CBIYY = ,E10.3,9H CBIYY = ,E10.3,9H CBIYY = ,E10.3,9H CBIYY = ,E1</pre>	DATA1950 DATA1960 DATA1970 DATA1970 DATA1980 DATA2000 DATA2010 DATA2010 DATA2010 DATA2010 DATA2030 DATA2030 DATA2050 DATA2050 DATA2050 DATA2050 DATA2050 DATA2050 DATA2000 DATA2100 DATA2120 DATA2120 DATA2130 DATA2140 DATA2150 DATA2160 DATA2190 DATA2190 DATA2200 DATA2200 DATA2200 DATA2200 DATA2200 DATA2200
	101 102 103 104 105 106 107 108 107 108 107 108 107 108 107 111 112 113 114 115 1100 1101 116	FORMAT STATEMENTS FORMAT(5H CARD/2X,2HNO//10X,28H* * PROGRAM CONTROL DATA * *//, * 14+10H CASE NO ==16) FORMAT(14+9H TIMAX = ,E10.3,10H IPTCNT = ,12,9H INLEG = ,12, * 10H IFPPRT = ,12) FORMAT(14+10H INDEXD = ,12,9H NOOUT = ,12) FORMAT(14+10H INDEXD = ,12,10H INDEYD = ,12,10H INDEZD = ,12) FORMAT(14+10H INDEXR = ,12,10H INDEYR = ,12,10H INDEZR = ,12) FORMAT(14,9H HMAX = ,E10.3, 8H HMIN = ,E10.3, 8H EMAX = ,E10.3, * 8H EMIN = ,E10.3,6H IP = ,12) FORMAT(14,9H IVARH = ,12,8H IMTH = ,12,10H CUTERR = ,E10.3) FORMAT(14,9H NFORC = ,12,10H IQUOUT = ,12,10H JCKSAB = ,12, *10H IDSETN = ,12) FORMAT(14,9H NFORC = ,12,10H IQUOUT = ,12,10H JCKSAB = ,12, * 0 H GRAV = ,E10.3,9H GRAVE = ,E10.3) FORMAT(14,8H ANGX = ,E10.3,9H GRAVE = ,E10.3) FORMAT(14,8H ANGX = ,E10.3,8H ANGY = ,E10.3,8H VELZ = ,E10.3) FORMAT(14,8H VELX = ,E10.3,8H VELY = ,E10.3,8H VELZ = ,E10.3) FORMAT(14,8H VELX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXX = ,E10.3,9H CBIYY = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXX = ,E10.3,9H CBIYZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXX = ,E10.3,9H CBIXZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXX = ,E10.3,9H CBIXZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXX = ,E10.3,9H CBIXZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXX = ,E10.3,9H CBIXZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXX = ,E10.3,9H CBIXZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXX = ,E10.3,9H CBIXZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXX = ,E10.3,9H CBIXZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXY = ,E10.3,9H CBIXZ = ,E10.3,9H CBIZZ = ,E10.3) FORMAT(14,9H CBIXY = ,E10.3,9H CBIXZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(14,9H CBIXY = ,E10.3,9H CBIXZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(14,9H CBIXY = ,E10.3,9H CBIXZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(14,9H CBIXY = ,E10.3,9H CBIXZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(14,9H CBIXY = ,E10.3,9H CBIXZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(14,9H CBIXY = ,E10.3,9H CBIXZ = ,E10.3,9H CBIYZ = ,E10.3) FORMAT(14,9H CBIYY = ,E10.3,9H CBIYA = ,E10.3,9H CB	DATA1950 DATA1960 DATA1970 DATA1970 DATA1990 DATA2010 DATA2010 DATA2010 DATA2010 DATA2030 DATA2030 DATA2030 DATA2050 DATA2050 DATA2050 DATA2050 DATA2050 DATA2050 DATA2100 DATA2120 DATA2120 DATA2150 DATA2150 DATA2150 DATA2160 DATA2170 DATA2180 ,DATA2190 DATA2190 DATA2200 DATA2200 DATA2200 DATA2200 DATA2200 DATA2200

•

421

Ą

```
118 FORMAT(I4,9H SS(I) = ,4(2X,E10.3))
119 FORMAT(I4,13H ATTHCK(I) = ,3(2X,E10.3))
                                                                          DATA2240
                                                                          DATA2250
120 \text{ FORMAT}(I4, 13H \text{ ATTPRS}(I) = ,3(2X, E10.3))
                                                                          DATA2260
121 FORMAT(//10X,17H* * SOIL DATA * *,//,14, 9H NTYPE = ,12,/,4X,
                                                                          DATA2270
           25H (PRIMARY SOIL MECHANICS))
                                                                          DATA2280
221 FORMAT(//10X,17H* * SOIL DATA * *,//,14,9H NTYPE = ,12,/,4X,
                                                                          DATA2290
   ¥
           27H (SECONDARY SOIL MECHANICS))
                                                                          DATA2300
122 FORMAT(I4,12H SOILP(I) = ,E10.3,3H , ,E10.3,3H , ,E10.3)
                                                                          DATA2310
123 FORMAT(//10X,16H* * LEG DATA * *,//,14, 9H NOLEG = ,12,
                                                                          DATA2320
           8H ILEG = ,I2,10H DRAGST = ,E10.3)
                                                                          DATA2330
   ¥
1200 FORMAT(I4,5H I = ,12,7H XFP = ,E10.3,7H YFP = ,E10.3,7H ZFP = ,
                                                                          DATA2340
                                                                          DATA2350
   *
           E10.3)
                                                                           DATA2360
1300 FORMAT(I4,5H I = ,I2,9H XMSCB = ,E10.3,9H YMSCB = ,E10.3,
           9H ZMSCB = ,E10.3)
                                                                           DATA2370
   ¥
124 FORMAT(14,9H PFCMS = ,5(2X,E10.3))
                                                                           DATA2380
125 FORMAT(I4,9H PFTMS = ,5(2X,E10.3))
                                                                           DATA2390
126 FORMAT(I4,9H CDCMS = ,5(2X,E10.3))
                                                                           DATA2400
                                                                           DATA2410
127 FORMAT(I4,9H CDTMS = ,5(2X,E10.3))
128 FORMAT(I4,9H SRCMS = ,5(2X,E10.3))
                                                                           DATA2420
129 FORMAT(I4,9H SRTMS = ,5(2X,E10.3))
                                                                           DATA2430
130 FORMAT(I4,10H SCMXMS = ,E10.3,10H STMXMS = ,E10.3,10H SRUCMS = ,
                                                                          DATA2440
                                                                           DATA2450
   ¥
           E10.3,10H SRUTMS = E10.3
 131 FORMAT(14,10H IRETMS = ,12)
                                                                           DATA2460
 132 FORMAT(I4,10H FRICMS = ,E10.3,10H COEFMS = ,E10.3,9H GAMMS = ,
                                                                           DATA2470
                                                                           DATA2480
           E10.3,8H AEI1 = ,E10.3,8H AEI2 = ,E10.3)
   ¥
1400 FORMAT(I4,5H I = ,12,9H XDSCB = ,E10.3,9H YDSCB = ,E10.3,
                                                                           DATA2490
                                                                           DATA2500
   ¥
           9H ZDSCB = +E10.3)
133 FORMAT(I4,9H PFCDS = ,5(2X,E10.3))
                                                                           DATA2510
134 FORMAT(I4,9H PFTDS = ,5(2X,E10.3))
                                                                           DATA2520
135 FORMAT(I4,9H CDCDS = ,5(2X,E10.3))
                                                                           DATA2530
                                                                           DATA2540
 136 FORMAT(I4,9H CDTDS = ,5(2X,E10.3))
                                                                           DATA2550
 137 FORMAT(I4,9H SRCDS = ,5(2X,E10.3))
 138 FORMAT(I4,9H SRTDS = ,5(2X,E10.3))
                                                                           DATA2560
 139 FORMAT(I4,10H SCMXDS = ,E10.3,10H STMXDS = ,E10.3,10H SRUCDS = ,
                                                                           DATA2570
                                                                           DATA2580
           E10.3.10H SRUTDS = .E10.3)
   ¥
 140 FORMAT(I4,10H IRETDS = ,I2)
                                                                           DATA2590
 141 FORMAT(I4,10H FRICDS = ,E10.3,10H COEFDS = ,E10.3,9H GAMDS = ,
                                                                           DATA2600
                                                                           DATA2610
           E10.3)
   ×
 142 FORMAT(//10X+18H* * MODAL DATA * * \cdot // \cdot 15 \cdot 10H MODEIN = \cdot 12)
                                                                           DATA2620
 242 FORMAT(/5X,25HRIGID CENTER BODY ASSUMED)
                                                                           DATA2630
1500 FORMAT(I5,5H I = ,12,6H GM = ,E10.3,9H OMEGA = ,E10.3)
                                                                           DATA2640
1600 FORMAT(I5,5H I = ,12,7H WNX = ,E10.3,7H WNY = ,E10.3,7H WNZ = ,
                                                                           DATA2650
                                                                           DATA2660
   ¥-
           E10.3)
1700 FORMAT(I5,5H I = ,I2,6H PX = ,E10.3,6H PY = ,E10.3,6H PZ = ,E10.3)DATA2670
1800 \text{ FORMAT}(15,5H \text{ I} = ,12,8H \text{ PMSX} = ,5(2X,E10,3))
                                                                           DATA2680
                                                                           DATA2690
1900 FORMAT(I5,5H I = ,I2,8H PMSY = ,5(2X,E10.3))
2000 FORMAT(I5,5H I = ,12,8H PMSZ = ,5(2X,E10.3))
                                                                           DATA2700
2100 FORMAT(I5,5H I = ,12,8H PDSX = ,5(2X,E10.3))
                                                                           DATA2710
2200 FORMAT(I5,5H I = ,I2,8H PDSY = ,5(2X,E10.3))
                                                                           DATA2720
2300 FORMAT(I5,5H I = ,I2,8H PDSZ = ,5(2X,E10.3))
                                                                           DATA2730
2400 FORMAT(I5,5H I = ,I2,9H POUTX = ,5(2X,E10.3))
                                                                           DATA2740
2500 FORMAT(I5,5H I = ,12,9H POUTY = ,5(2X,E10.3))
                                                                           DATA2750
2600 FORMAT(I5,5H I = ,I2,9H POUTZ = ,5(2X,E10.3))
                                                                           DATA2760
2700 FORMAT(15,5H I = ,12,8H PCGX = ,E10.3,8H PCGY = ,E10.3,8H PCGZ = ,DATA2770
                                                                           DATA2780
    ¥
           E10.3)
3000 FORMAT(/5X,50HWARNING - PROGRAM CONTROL DATA INDICATES THIS CASE,/DATA2790
```

¥	55HIS PART OF A MULTIPLE DATA SET RUN AND A SECONDARY TIME	•/DATA2800
¥	52HHISTORY TAPE IS TO BE GENERATED. THIS COMBINATION OF,/	DATA2810
*	42HPROGRAM OPTIONS MUST BE HANDLED WITH CARE.,)	DATA2820
END		DATA2830

•

OVERLAY ( LLMPT5, 2, 0 )	OVER	30
PROGRAM LIMPEX	LLEX	10
COMMON / L777RM / IPTATI IPTOTI I LOCNAM(90)	LLEX	20
DIMENSION XSI(8) TAD(8) AND(8)	LLEX	30
FOULVALENCE ( TIMESA(1), LTIMES )	LLEX	40
DIMENSION TIMES(275)	LLEX	50
DIMENSION TIMESB(275)	LLEX	60
DIMENSION IMHSA(1)	LLEX	70
FOULVALENCE ( $1MHSA(1) + TIMHSA(1)$ )	LLEX	80
DIMENSION $GC(4,5)$ , $GCD(4,5)$ , $GCDD(4,5)$	LLEX	90
DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),	LLEX	100
1 XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),	LLEX	110
2 XFPSDD(4,5), YFPSDD(4,5), ZFPSDD(4,5)	LLEX	120
DIMENSION XMN1(8)	LLEX	130
DIMENSION INOUTV(5)	LLEX	140
COMMON / THISV/ TIMHSA(1), ATTH(5),AM(6,6),AME1(3),INDFPI(6),	LLEX	150
1 INDFPC(6),	LLEX	160
2 STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10),	LLEX	170
3 URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10),	LLEX	180
4 SETCMS( 5), SETTMS( 5), INDCDS(10), INDTDS(10), INDCMS(10),		190
5 INDTMS( 5), PRFCDS (10), PRFTDS (10), PRFCMS ( 5), PRFIMS ( 5),		200
6 IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5),		210
L FRVMST(5), IPOIDS(10), IPOIMS(5)		220
EQUIVALENCE ( NFIPDS, NOLEG )		240
COMMON COMINI(400)	LLEX	250
EQUIVALENCE ( COMINIC 3 / FIDUM )	LIFX	260
EQUIVALENCE ( COMINIT 4 /) ASD )	LLEX	270
EQUIVALENCE ( COMINIC 12 ), XSC )	LLEX	280
EQUIVALENCE ( COMINT( 12 / ) 13 )	LLEX	290
EQUIVALENCE ( COMINT( 20 ) SDD )	LLEX	300
FOUTVALENCE ( COMINT( 24 )) ZS )	LLEX	310
EQUIVALENCE ( COMINT ( 28 ) • ZSD )	LLEX	320
EQUIVALENCE ( COMINT ( 32 ) + ZSDD )	LLEX	330
EQUIVALENCE ( COMINT( 36 ), PHI )	LLEX	340
EQUIVALENCE ( COMINT( 40 ), PHID )	LLEX	350
EQUIVALENCE ( COMINT( 44 ), WX )	LLEX	360
EQUIVALENCE ( COMINT( 48 ), WXD )	LLEX	370
EQUIVALENCE ( COMINT( 52 ), THTA )	LLEX	380
EQUIVALENCE ( COMINT( 56 ), THTAD )	LLEX	390
EQUIVALENCE ( COMINT( 60 ), WY )	LLEX	400
EQUIVALENCE ( COMINT( 64 ), WYD )	LLEX	410
EQUIVALENCE ( COMINT( 68 ), PSI )	LLEX	420
EQUIVALENCE ( COMINT( 72 ), PSID )	LLEX	430
EQUIVALENCE ( COMINT( 76 ), WZ )		440
EQUIVALENCE ( COMINT( 80 ), W2D )		420
EQUIVALENCE ( COMINT( 84 ), GC )		400
EQUIVALENCE ( COMINT(104 ), GCD )		480
EQUIVALENCE ( COMINT/144 ), VEDS )		490
EQUIVALENCE ( COMINT/144 /) AFFS /	LIFY	500
EQUIVALENCE ( COMINITION ), AFFOR )	LLEX	510
FOULVALENCE ( COMINICION ), AFSUD /	LLEX	520

\$

EQUI	VALENCE ( COMINT(224 ), YFPSD )	LLEX 530
EQUI	VALENCE ( COMINT(244 ), YEPSDD )	LLEX 540
EOUI	VALENCE ( COMINT/266 ) 7505 )	LIEY 550
E001	VALENCE ( COMINIZEDA ) ZERCE )	
	VALENCE ( COMINI(204 )) ZEPSD )	LLEA 500
EQUI	VALENCE ( COMINI(304 )) ZEPSDD )	LLEX 570
EQUI	VALENCE ( COMINT( 324 ), TIME )	LLEX 580
EQUI	VALENCE ( COMINT( 325 ), HMAX )	LLEX 590
EQUI	VALENCE ( COMINT( 326 ), HMIN )	LLEX 600
EQUI	VALENCE ( COMINT( 327 ), EMIN )	LLEX 610
EQUI	VALENCE ( COMINT( 328 ), EMAX )	LLEX 620
FQU	VALENCE ( COMINT( 329 ) XSI )	LLEX 630
FOUL	VALENCE (COMINT (337)) $H_{7}$	LIEX 640
E001	VALENCE ( COMINE ( 320 ), CHEEDD	
	VALENCE ( COMINIC 338 7) CULERRY	
EQU.	VALENCE ( COMINI( 339)) IP )	LLEX 660
EQU	VALENCE ( COMINI( 340 ), IVARH )	LLEX 670
EQU	VALENCE ( COMINT( 341 ), IMTH )	LLEX 680
EQUI	VALENCE ( COMINT( 342 ), IPRNT )	LLEX 690
EQU	IVALENCE ( COMINT( 343 ), IFIN )	LLEX 700
EQU:	VALENCE ( COMINT( 344 ), IAD )	LLEX 710
EQU	VALENCE ( COMINT( 352 ), IND )	LLEX 720
FQU	VALENCE ( COMINT( 360 ) JULIT )	1 LEX 730
FOU	VALENCE ( COMINT( 361 ) IPTONT )	11 EX 740
FOU	VALENCE ( COMINT( 363 ), IVAL )	LIEX 750
EQUI		
EQU.	VALENCE ( COMINIC 364 7) X5 7	
	COMINI(364-367) USED BY XS	LLEX //O
EQU	VALENCE ( COMINT( 368 ), CURDT )	LLEX 780
EQU	(VALENCE ( COMINT( 369 ), XMN1 )	LLEX 790
X	INI USES COMINT(369-376)	LLEX 800
COM	10N	LLEX 810
1	CBMASS, CBIXX , CBIXZ , CBIYY , CBIYZ , CBIZZ , FPMASS, CBIXY,	LLEX 820
2	DC(3,3) , XFP(5), YFP(5), ZFP(5), WNX(5), WNY(5),	LLEX 830
3	WN7(5) • PX(5) • PY(5) • P7(5) • GM(5) • OMEGA(5) •	LLEX 840
4	GRAV - GRAVE - ZETA - ETS (6) - ESXSI(5) -	11 EX 850
т 5	FSYSI(5) FSYSI(5) SOILX(5)	LIEX 860
4		LIEX 870
7		
/	PMS(15,5) , PMS(15,5) , PDS(10,5) ,	LLEX 800
8	PDSY(10,5) , PDSZ(10,5) , FLXS , FLYS , FLZS ,	LLEX 890
9	ILXL , ILYL , ILZL , SLO , XMSCB(5) ,	LLEX 900
С	YMSCB(5) , ZMSCB(5) , XDSCB(10) ,	LLEX 910
D	YDSCB(10) , ZDSCB(10) , ILEG , IMS ,	LLEX 920
£	FSTX , FSTY , FSTZ , PVCBX , PVCBY , PVCBZ ,	LLEX 930
F	PVFPX , PVFPY , PVFPZ , NOLEG , SLOMS(5) ,	LLEX 940
G	SLODS(10)	LLEX 950
Сом	40N	LLEX 960
1	PECMS(5) PECDS(5) PETDS(5)	LLEY 970
2		
2	CREDUCED COERCE CARDO	LLEX 900
3	SRIDS(5) , SRIMS(5) , CUEPDS, CUEPMS, GAMDS,	LLEX 990
4	GAMMS , SKUCDS, SKUCMS, SKUTDS, SRUTMS, SCMXDS, SCMXMS,	LLEX1000
5	STMXDS, STMXMS, CDCDS(5) , CDCMS(5) ,	LLEX1010
6	CDTDS(5) , CDTMS(5) , FRICDS, FRICMS, IRETDS,	LLEX1020
7	IRETMS, STROKE(10) , STRKDS(10) , STRKMS( 5) ,	LLEX1030
8	LENGTH, CDXS, CDYS, CDZS, NTYPE, RAD(4), SS(4), ATTHCK(3),	LLEX1040
9	ATTPRS(3) , ADIST , SOILP(3) , NMODES,	LLEX1050
А	COEF , GAMMA , FRIC , SCMAX , STMAX , CDC(5), CDT(5).	LLEX1060
в	SRULT , SRULC , SRT(5), SRC(5), PFT(5), PFC(5), GFLEGS(5),	LLEX1070
c	CURMSV. CURMSI. INDEXD. INDEXD. INDEXD. INDEXR. INDEXR.	LIEX1080
<u> </u>	CONTRACTOR AND ADD ADD ADD ADD ADD ADD ADD ADD ADD	

.

С

с

425

```
D
          INDFZR, TIMAX, DRAGST, IFP
                                                                            LLEX1090
     COMMON
                                                                            LLEX1100
                                     NOOUT,XOUT(10),YOUT(10),ZOUT(10),
     1CMS, CDCONT, SOILNU, SI RHO,
                                                                            LLEX1110
     2MODEIN, POUTX(10,5), POUTY(10,5), POUTZ(10,5), PCGX(5), PCGY(5),
                                                                            LLEX1120
     3PCGZ(5),AEI1,AEI2,FORMS(5),FORDS(10),FORCE,NFORC,SAVMSX(5),
                                                                            LLEX1130
     4SAVMSZ(5), SAVDSX(10), SAVDSY(10), SAVDSZ(10), IQUOUT, GSINZT,
                                                                            LLEX1140
     5GCOSZT,STAB,STABVL,ISTAB,JCKSAB,VELX,VELY,VELZ,SAVMSY(5)
                                                                            LLEX1150
      COMMON
                                                                            LLEX1160
          SMXMSC(5), TMXMSC(5), SMXMST(5), TMXMST(5),
SMXDSC(10), TMXDSC(10), SMXDST(10), TMXDST(10)
                                                                            LLEX1170
     1
                                                                            LLEX1180
     2
         slngms(5) slngds(10), curdsl, inleg, ifpprt,
                                                                            LLEX1190
     3
          IMPACT(5) , IPRTFP(5), KOUNT(5) , ANGX, ANGY, ANGZ
                                                                            LLEX1200
     4
      DIMENSION AHOLD(6),
                                                                            LLEX1210
                             AMINV(6,6), FLTL(6)
      DIMENSION GF(5), GFO(5)
                                                                            LLEX1220
      DIMENSION AMFP(5), FO(3), FE(6)
                                                                            LLEX1230
      DIMENSION AME2(3)
                                                                            LLEX1240
               PRECALCULATE SOME CONSTANTS USED LATER IN MODE WORK
                                                                            LLEX1250
С
      DIMENSION PYZ(5), PYZ2(5), PXZ(5), PXZ2(5), PXY(5), PXY2(5),
                                                                            LLEX1260
         WNZNY(5),WNXNZ(5),WNYNX(5),PYNZ2(5),PZNX2(5),PXNY2(5),W2GM(5)
                                                                            LLEX1270
     1
                SET INTEGRATION LIST UPDATE INDICATOR
                                                                            LLEX1280
С
      INDIVR = 0
                                                                            LLEX1290
      LTIMHS = 273
                                                                            LLEX1300
С
          PRINT COUNT
                                                                            LLEX1310
      KPTCNT = 99999
                                                                            LLEX1320
                                                                            LLEX1330
С
          PITCH ANGLE LIMIT
      ATIPMX = 1.55
                                                                            LLEX1340
                                                                            LLEX1350
          TIME HISTORY UPDATE INDICATOR
С
      INDBPU = 1
                                                                            LLEX1360
           INTEGRATION LOOP FIRST PASS INDICATOR
С
                                                                            LLEX1370
                                                                            LLEX1380
      INDIFP = 0
      IF ( NMODES .EQ. 0 ) GO TO 110
                                                                            LLEX1390
      DO 100 I = 1, NMODES
                                                                            LLEX1400
                   PY(I)+PZ(I)
      PYZ(I) =
                                                                            LLEX1410
      PYZ2(I) =
                   PYZ(I) + PYZ(I)
                                                                            LLEX1420
                   PX(I)+PZ(I)
                                                                            LLEX1430
      PXZ(I) =
      PXZ2(I) =
                   PXZ(I)+PXZ(I)
                                                                            LLEX1440
      PXY(I) =
                   PX(I)+PY(I)
                                                                            LLEX1450
      PXY_2(I) =
                   PXY(I)+PXY(I)
                                                                            LLEX1460
      WNZNY(I) =
                   WNZ(I)-WNY(I)
                                                                            LLEX1470
      WNXNZ(I) =
                   WNX(I)-WNZ(I)
                                                                            LLEX1480
                   WNY(I)-WNX(I)
      WNYNX(I) =
                                                                            LLEX1490
                 OMEGA(I)*OMEGA(I)*GM(I)
      W2GM(I) =
                                                                            LLEX1500
                                                                            LLEX1510
                  2.*( PY(I)- PZ(I) )
      PYNZ2(I) =
      PZNX2(I) = 2.*(PZ(I) - PX(I))
                                                                            LLEX1520
  100 PXNY2(I) = 2 \cdot (PX(I) - PY(I))
                                                                            LLEX1530
С
           SET MODAL ANALYSIS MASS TO ZERO
                                                                            LLEX1540
          105 I = 1, 3
                                                                            LLEX1550
      DO
  105 \text{ AME1(I)} = 0.0
                                                                            LLEX1560
  110 CONTINUE
                                                                            LLEX1570
C
                                                                            LLEX1580
                                                                            LLEX1590
           AT PRESENT ALL FOOT PADS ARE OF THE SAME MASS
С
                                                                            LLEX1600
С
      DO 70 I = 1, NFTPDS
                                                                            LLEX1610
   70 AMFP(I)
                                                                            LLEX1620
               = FPMASS
                INDFAR=3 IF ROTATIONS ARE NOT INTEGRATED
С
                                                                            LLEX1630
      INDFAR = INDFXR+INDFYR+INDFZR
                                                                            LLEX1640
```

С PRECALCULATE MOMENT OF INERTIA CONSTANTS LLEX1650 LLEX1660 CIZZYY = CBIZZ-CBIYY CIXXZZ = CBIXX-CBIZZ LLEX1670 LLFX1680 CIYYXX = CBIYY-CBIXX С LLEX1690 SET INTEGRATION VARIABLE LIST AND CUTOFF LIST COUNTERS LLEX1700 С LLEX1710 С LLEX1720 IPTATL = 0LLEX1730 IPTOTL = 0С INITIALIZE THE INTEGRATION ROUTINES LLEX1740 LLEX1750 CALL SETUP С LLEX1760 ARE INTEGRATION V.LLEX1770 С XS, XSD, AND XSDD IF ( INDFXD .NE. 0 ) GO TO 210 LLEX1780 LLEX1790 CALL INUPD ( 364 ) LLEX1800 CALL INUPD ( 4 ) CALL INUPD ( 8 LLEX1810 ) GO TO 220 LLEX1820 210 IF ( INDFYD .NE. 0 ) ARE INTEGRATION V.LLEX1830 YS, YSD, AND YSDD C CALL INUPD ( 12 LLEX1840 LLEX1850 CALL INUPD ( 16 ) CALL INUPD ( 20 ) 220 IF ( INDFZD •NE• 0 ) LLEX1860 GO TO 230 LLEX1870 ZS, ZSD, AND YSDD ARE INTEGRATION V.LLEX1880 С LLEX1890 CALL INUPD (24 ) LLEX1900 CALL INUPD (28 ) CALL INUPD ( 32 LLEX1910 ) ILEX1920 230 IF ( INDFXR .NE. 0 ) GO TO 240 WX, WXD, PHID, AND PHI ARE INTEGRATION V.LLEX1930 C CALL INUPD ( 44 ) LLEX1940 LLEX1950 CALL INUPD ( 48 ) CALL INUPD ( 40 ) CALL INUPD ( 40 ) CALL INUPD ( 36 ) 240 IF ( INDFYR •NE• 0 ) GO TO 250 LLEX1960 LLEX1970 LLEX1980 WY, WYD, THTAD, AND THTA ARE INTEGRATION VLLEX1990 С CALL INUPD ( 60 LLEX2000 ) CALL INUPD ( 64 CALL INUPD ( 56 LLEX2010 ) LLEX2020 ) CALL INUPD ( 52 LLEX2030 ) LLEX2040 250 IF ( INDFZR .NE. 0 ) GO TO 260 WZ, WZD, PSID, AND PSI ARE INTEGRATION V.LLEX2050 С LLEX2060 CALL INUPD ( ) 76 CALL INUPD ( LLEX2070 80 ) CALL INUPD ( LLEX2080 72 ) CALL INUPD ( 68 LLEX2090 ) LLEX2100 260 CONTINUE GC(I), GCD(I), AND GCDD(I) ARE INTEGRAT. VLLEX2110 C 11 FX2120 IF ( NMODES .EQ. 0 ) GO TO 290 LLEX2130 DO 280 I = 1, NMODES LLEX2140 J = 4 * ICALL INUPD ( 80+ J ) LLEX2150 LLEX2160 CALL INUPD (100+ J ) 280 CALL INUPD (120+ J ) LLEX2170 LLEX2180 290 CONTINUE LLEX2190 1 CONTINUE LLEX2200 С

```
CALL GEOM
                                                                           LLEX2210
С
                                                                           LLEX2220
      IF ( INDIFP .EQ. 1) GO TO 64
                                                                           LLEX2230
С
                                                                           LLEX2240
С
                INITIALIZE FOOT PAD POSITIONS IN SURFACE COORDINATES
                                                                           LLEX2250
C
C
                     ALL SO SET UP FOOT PAD CUTOFF VARIABLES AND
                                                                           LLEX2260
                     INTEGRATION QUANTIES
                                                                           LLEX2270
С
                                                                           LLEX2280
      XS = 100 \cdot *GRAVE
                                                                           LLEX2290
      YS = 0.0
                                                                           LLEX2300
      ZS = 0.0
                                                                           LLEX2310
               I = 1, NFTPDS
      DO 60
                                                                           LLEX2320
                       DC(1,1)*XFP(I) + DC(1,2)*YFP(I) + DC(1,3)*ZFP(I)
      Α
                Ξ
                                                                           LLEX2330
     1
                       ATTH(I)
                                                                           11 FX2340
   60 \times S = AMIN1(\times S, A)
                                                                           LLEX2350
      XS = -XS
                                                                           LLEX2360
С
                                                                           LLEX2370
С
      PRINT LANDER INITIAL CONDITONS
                                                                           LLEX2380
С
                                                                           LLEX2390
      WRITE(6,5010)XS, YS, ZS, XSD, YSD, ZSD, WXIN, WYIN, WZIN
                                                                           LLEX2400
 5010 FORMAT(1H1,/20X,46HLANDER INITIAL CONDITIONS (SURFACE COORDINATE +LLEX2410
     ¥
             7HSYSTEM),//37X,1H(,10X,1HY,
                                                                           LLEX2420
             10X,1HZ//6X,21HINITIAL C.G. POSITION,6X,3(F9.2,2X),//
     ¥
                                                                           LLEX2430
     ¥
             6X,21HINITIAL C.G. VELOCITY,6X,3(F9.2,2X)//
                                                                           LLEX2440
            6X,24HINITIAL ANGULAR VELOCITY,3X,3(F9.2,2X))
     ¥
                                                                           LLEX2450
               I = 1, NFTPDS
      DO 62
                                                                           LLEX2460
      XFPS(1,I) = XS+DC(1,1)*XFP(I)+DC(1,2)*YFP(I)+DC(1,3)*ZFP(I)
                                                                           LLEX2470
                                                                           LLEX2480
      J = 4*Ⅰ
      IF ( XFPS(1,I)-ATTH(I) .LT. CUTERR ) GO TO 61
                                                                           LLEX2490
          XFPS( I ) IS A CUTOFF VARIABLE
С
                                                                           LLEX2500
      CALL LOC(140+J,ATTH(I)+(.5+GRAVE*2.5E-06)*CUTERR,1)
                                                                           LLEX2510
      INDFPI(I) = IPTATL
                                                                           LLEX2520
      INDFPC(I) = 0
                                                                           LLEX2530
      GO TO 62
                                                                           LLEX2540
С
                XFPS(I),YFPS(I),ZFPS(I),XFPSD(I),YFPSD(I), ZFPSD(I)
                                                                           LLEX2550
С
                  XFPSDD(I), YFPSDD(I), ZFPSDD(I) ARE INTEGRATION VAR.
                                                                           LLEX2560
   61 CALL INUPD ( 140 + J )
                                                                            LLEX2570
      INDFPI(I) = IPTOTL
                                                                            LLEX2580
      CALL INUPD ( 160 + J )
                                                                            LLEX2590
      CALL INUPD ( 180 + J )
                                                                            LLEX2600
      CALL INUPD ( 200 + J )
                                                                            LLEX2610
      CALL INUPD ( 220 + J )
                                                                            LLEX2620
      CALL INUPD ( 240 + J )
                                                                            LLEX2630
      CALL INUPD ( 260 + J )
                                                                            LLEX2640
      CALL INUPD ( 280 + J )
                                                                            LLEX2650
      CALL INUPD (300 + J)
                                                                            LLEX2660
      YFPs(1,1) = YS+DC(2,1)*XFP(1)+DC(2,2)*YFP(1)+DC(2,3)*ZFP(1)
                                                                            LLEX2670
      ZFPS(1,1) = ZS+DC(3,1)*XFP(1)+DC(3,2)*YFP(1)+DC(3,3)*ZFP(1)
                                                                            LLEX2680
             WY * ZFP(I) - WZ * YFP(I)
WZ * XFP(I) - WX * ZFP(I)
                                                                            LLEX2690
      Δ
          =
      В
          =
                                                                            LLEX2700
             WX * YFP(I)
                              WY * XFP(I)
      C
                                                                            LLEX2710
          Ξ
      XFPSD(1,I) = XSD + DC(1,1)*A + DC(1,2)*B + DC(1,3)*C
                                                                            LLEX2720
      YFPSD(1,I) = YSD + DC(2,I)*A + DC(2,2)*B + DC(2,3)*C
                                                                            LLEX2730
      ZFPSD(1,I) = ZSD + DC(3,1)*A + DC(3,2)*B + DC(3,3)*C
                                                                            LLEX2740
С
          SET THIS FOOT PAD*S CONTACT INDICATOR
                                                                            LLEX2750
      INDFPC (I) = 2
                                                                            LLEX2760
```

```
LLEX2770
   62 CONTINUE
С
                                                                              LLEX2780
                SAVE NON INTEGRATION TIMEHISTORY VARIABLES IN CASE
С
                                                                              LLEX2790
С
                            OF INITIAL BACKUP
                                                                              LLEX2800
      DO 63 I = 1, LTIMHS
                                                                              LLEX2810
   63 \text{ TIMHSC}(I) = \text{TIMHSA}(I)
                                                                              LLEX2820
   64 CONTINUE
                                                                              LLEX2830
С
                                                                              LLEX2840
С
           PROCESS THE FT. PAD. CALLS
                                                                              LLEX2850
С
                                                                              LLEX2860
      DO 380 I = 1 • 6
                                                                              LLEX2870
  380 \text{ FLTL(I)} = 0.0
                                                                              LLEX2880
      IF ( NMODES .EQ. 0 ) GO TO 395
                                                                              LLEX2890
      DO 390 I = 1, 3
                                                                              LLEX2900
      AME2(I) = 0.0
                                                                              LLEX2910
                                                                              LLEX2920
  390 \text{ GF(I)} = 0.0
  395 CONTINUE
                                                                              LLEX2930
      DO 430 IFP = 1, NFTPDS
                                                                              LLEX2940
      IF ( INDFPC(IFP) .LE. 0 ) GO TO 405
                                                                              LLEX2950
С
                                                                              LLEX2960
С
      CONTACTING FOOTPAD
                                                                              LLEX2970
С
                                                                              LLEX2980
      IF(INDIFP.EQ.1) CALL FTPAD
                                                                              LLEX2990
С
                                                                              LLEX3000
      IF(IBOTM.NE.O)GO TO 3110
                                                                              LLEX3010
С
           SUM UP THE RESULTAN LEG FORCES
                                                                              LLEX3020
      FLTL(1) = FLTL(1) + FLXS
                                                                              LLEX3030
      FLTL(2) = FLTL(2) + FLYS
                                                                              LLEX3040
      FLTL(3) = FLTL(3) + FLZS
                                                                              LLEX3050
      FLTL(4) = FLTL(4) + TLXL
                                                                              LLEX3060
      FLTL(5) = FLTL(5) + TLYL
                                                                              11 EX3070
                                                                              LLEX3080
      FLTL(6) = FLTL(6) + TLZL
                                                                              LLEX3090
      IF ( NMODES .EQ. 0 ) GO TO 406
С
           SUM THE GEN. FORCES TO ELASTIC EFFECTS OF THE MAIN AND
                                                                              LLEX3100
                                                                              LLEX3110
С
                DRAG STRUTS
                                                                              LLEX3120
      DO 404 I = 1, NMODES
  404 \text{ GF(I)} = \text{GF(I)} + \text{GFLEGS(I)}
                                                                              LLEX3130
                                                                              LLEX3140
  406 CONTINUE
        F.P. ACCEL. = (GRAVITY LOADS -STRUT LOADS+STOIL LOADS)/F.P. MASSLLEX3150
C
       XFPSDD(1,IFP) =(-GCOSZT*AMFP(IFP)+FSXSI(IFP)+SOILX(IFP))/AMFP(IFP)LLEX3160
                                         +FSYSI(IFP)+SOILY(IFP))/AMFP(IFP)LLEX3170
       YFPSDD(1,IFP) =(
       ZFPSDD(1,IFP) =( GSINZT*AMFP(IFP)+FSZSI(IFP)+SOILZ(IFP))/AMFP(IFP)LLEX3180
                INTEGRATE TO GET XFPS, YFPS, AND ZFPS
                                                                              LLEX3190
С
       CALL INTEG ( XFPSDD(1,IFP), XFPSD(1,IFP) )
                                                                              LLEX3200
      CALL INTEG ( XFPSD (1, IFP), XFPS (1, IFP) )
                                                                              LLEX3210
      CALL INTEG ( YFPSDD(1, IFP), YFPSD(1, IFP) )
                                                                              LLEX3220
      CALL INTEG ( YFPSD (1,IFP), YFPS (1,IFP) )
CALL INTEG ( ZFPSDD(1,IFP), ZFPSD(1,IFP) )
                                                                              LLEX3230
                                                                              LLEX3240
      CALL INTEG ( ZFPSD (1, IFP), ZFPS (1, IFP) )
                                                                              LLEX3250
                                                                              LLEX3260
       GO TO 430
                                                                              LLEX3270
С
С
      NONCONTACTING FOOTPAD
                                                                              LLEX3280
С
                                                                              LLEX3290
  405 CONTINUE
                                                                              LLEX3300
      XXX = XFP(IFP)
                                                                              LLEX3310
       YYY=YFP(IFP)
                                                                              LLEX3320
```

	ZZZ=ZFP(IFP)	LLEX3330
	IF(NMODES+EQ+0)GO TO 320	LLEX3340
	IF(ILEG•EQ•0)GO TO 304	LLEX3350
	RRRT=DRAGST/SLNGMS(IFP)	LLEX3360
	XXX=XMSCB(IFP)+RRRT*(XFP(IFP)~XMSCB(IFP))	LLEX3370
	YYY=YMSCB(IFP)+RRRT*(YFP(IFP)-YMSCB(IFP))	LLEX3380
	777 = 7MSCB(IFP) + RRRT*(7FP(IFP) - 7MSCB(IFP))	LLEX3390
204		LLEX3400
504	VMST=VMSCB(IED)	LLFX3410
		LLEX3420
		LLEX3430
	NNI=2*(I+P-I)+1	
	NN2=NN1+1	LLEX3450
	XDS1T=XDSCB(NN1)	LLEX3460
	YDS1T=YDSCB(NN1)	LLEX3470
	ZDS1T=ZDSCB(NN1)	LLEX3480
	XDS2T=XDSCB(NN2)	LLEX3490
	YDS2T=YDSCB(NN2)	LLEX3500
	7DS2T=7DSCB(NN2)	LLEX3510
	DO 305 I=1.NMODES	LLEX3520
	$xMST = xMST + PMSX(IFP \cdot I) * GC(3 \cdot I)$	LLEX3530
	YMST=YMST+PMSY(IFP,I)*GC(1.I)	LLEX3540
	TMST = TMST + PMST (IFP + I) * GC (1 + I)	LLEX3550
	$\sum_{i=1}^{N} (1 - \sum_{i=1}^{N} (1 - \sum_{i$	LLEX3560
		LLEX3570
	10511 = 10511 + PD51(NN191) + 0C(191)	LLEX3580
	XDS21=XDS21+PDSX(NN2,1)*GC(1,1)	
	YDS21=YDS21+PDSY(NN2,1)*GC(1,1)	
	ZDS2T=ZDS2T+PDS2(NN2,1)*GC(1,1)	LLEX3010
305	CONTINUE	LLEX3620
310	CONTINUE	LLEX3630
	DXMS=XXX-XMST	LLEX3640
	DYMS=YYY-YMST	LLEX3650
	DZMS=ZZZ-ZMST	LLEX3660
	DXDS1=XXX-XDS1T	LLEX3670
	DYDS1=YYY-YDS1T	LLEX3680
	DZDS1=ZZZ-ZDS1T	LLEX3690
	DXDS2=XXX-XDS2T	LLEX3700
	DYDS2=YYY-YDS2T	LLEX3710
	DZDS2=ZZZ-ZDS2T	LLEX3720
	FUNMS = DXMS**2+DYMS**2+DZMS**2-SLNGMS(IFP)**2	LLEX3730
	IF(ILEG_NE_O)FUNMS=DXMS**2+DYMS**2+DZMS**2-DRAGST**2	LLEX3740
	FUNDS1=DXDS1**2+DYDS1**2+DZDS1**2-SLNGDS(NN1)**2	LLEX3750
	FUNDS2=DXDS2**2+DYDS2**2+DZDS2**2-SLNGDS(NN2)**2	LLEX3760
		LLEX3770
	IF (FRPAILTAC ADOD5*GRAVE) GO TO 320	LLEX3780
		LLEX3790
		LLEX3800
		LIEX3810
		LLEX3820
	DELX=-FUNMS*AAA-FUNDS1*BBB-FUNDS2*CCC	
	DELY=4.*DXMS*(DZDSI*FUNDS2-DZDS2*FUNDS1)-FUNMS*BBB+4.*DZMS*	LLEX2050
	DELZ=4+*DXMS*(DYDS2*FUNDS1-DYDS1*FUNDS2)+4+*DYMS*(DXDS1*FUNDS2-	
	* DXDS2*FUNDS1)-FUNMS*CCC	LLEX38/0
	XXX=XXX+DELX/DELL	LLEX3880

```
YYY=YYY+DELY/DELL
                                                                          LLEX3890
      ZZZ=ZZZ+DELZ/DELL
                                                                          LLEX3900
     GO TO 310
                                                                          LLEX3910
                                                                          LLEX3920
 320 CONTINUE
      IF(ILEG.EQ.0)GO TO 321
                                                                          LLEX3930
      IF(NMODES.EQ.0)GO TO 321
                                                                          LLEX3935
     CX=(XXX-XMSCB(IPF))/DRAGST
                                                                          LLEX3940
     CY=(YYY-YMSCB(IFP))/DRAGST
                                                                          LLEX3950
     CZ=(ZZZ-ZMSCB(IFP))/DRAGST
                                                                          LLEX3960
      DDD=SLNGMS(IFP)-DRAGST
                                                                          LLEX3970
      XXX=XXX+CX*DDD
                                                                          LLEX3980
      YYY=YYY+CY*DDD
                                                                          LLEX3990
      ZZZ=ZZZ+CZ*DDD
                                                                          LLEX4000
 321 CONTINUE
                                                                          LLEX4010
     XFPS(1,IFP)=XS+DC(1,1)*XXX+DC(1,2)*YYY+DC(1,3)*ZZZ
                                                                          LLEX4020
                                                                          LLEX4030
      YFPS(1, IFP)=YS+DC(2,1)*XXX+DC(2,2)*YYY+DC(2,3)*ZZZ
      ZFPS(1, IFP)=ZS+DC(3,1)*XXX+DC(3,2)*YYY+DC(3,3)*ZZZ
                                                                          LLEX4040
      NNN=2*(IFP-1)+1
                                                                          LLEX4050
     FORMS(IFP)=0.0
                                                                          LLEX4060
      FORDS(NNN)=0.0
                                                                           LLEX4070
     FORDS(NNN+1)=0.0
                                                                           LLEX4080
      IF(NFORC.EQ.0) GO TO 430
                                                                           LLEX4090
      SAVMSX(IFP)=0.0
                                                                           LLEX4100
      SAVMSY(IFP)=0.0
                                                                           LLEX4110
      SAVMSZ(IFP)=0.0
                                                                           LLEX4120
      SAVDSX(NNN)=0.0
                                                                           LLEX4130
      SAVDSX(NNN+1)=0.0
                                                                           LLEX4140
                                                                           LLEX4150
      SAVDSY(NNN)=0.0
      SAVDSY(NNN+1)=0.0
                                                                           LLEX4160
      SAVDSZ(NNN)=0.0
                                                                           LLEX4170
                                                                           LLEX4180
      SAVDSZ(NNN+1)=0.0
                                                                           LLEX4190
  430 CONTINUE
                                                                           LLEX4200
      INDIFP = 1
С
                                                                           LLEX4210
                                                                           LLEX4220
C
C
          SET INDEPD TO INDICATE THE HIGHEST FOOTPAD STATUS
                                                                           LLEX4230
      INDFPD = INDFPC(1)
                                                                           LLEX4240
      DO 435 IFP = 2,6
                                                                           LLEX4250
  435 INDFPD = MINO( INDFPD, INDFPC(IFP) )
                                                                           LLEX4260
                                                                           1 L EX4270
С
               UPDATE THE MASS MATRIX WITH THE ELASTIC MODE
С
                                                                           LLEX4280
С
                    CONTRIBUTIONS
                                                                           LLEX4290
      IF ( NMODES .EQ. 0 ) GO TO 480
                                                                          LLEX4300
      DO 450 I = 1,NMODES
                                                                          LLEX4310
      AME2(1)=AME2(1) + ( PYZ2(I)+WNX(I)*GC(1,I))*GC(1,I)
                                                                          LLEX4320
      AME2(2)=AME2(2) + ( PXZ2(I)+WNY(I)*GC(1,I))*GC(1,I)
                                                                           LLEX4330
  450 AME2(3)=AME2(3) + ( PXY2(I)+WNZ(I)*GC(1,I))*GC(1,I)
                                                                           LLEX4340
      DO 470 I = 1, 3
                                                                           LLEX4350
      J = I + 3
                                                                           LLEX4360
  470 \text{ AM}(J_{9}J) = \text{AM}(J_{9}J) - ( \text{AME1}(I) - \text{AME2}(I))
                                                                           LLEX4370
                                                                           LLEX4380
  480 CONTINUE
      IF ( INDFPD .GT. 1 ) GO TO 540
                                                                           LLEX4390
                                                                           LLEX4400
С
С
               PROCESS THE NONCONTACTING FOOTPAD MASS CONTRIBUTIONS
                                                                           LLEX4410
С
                                                                           LLEX4420
      DO 490 I = 1, 3
                                                                           LLEX4430
```

```
DO 490 J = 1, 3
                                                                          LLEX4440
          = .1+3
      к
                                                                          LLEX4450
      AM(I_{*}K) = 0.0
                                                                          LLEX4460
  490 AM(K \cdot I) = 0 \cdot 0
                                                                          LLEX4470
      DO 530 IFP=1,NFTPDS
                                                                          LLEX4480
      IF ( INDFPC(IFP) .GT. 0 ) GO TO 520
                                                                          LLEX4490
          500 I = 1, 3
      DO
                                                                          LLEX4500
С
               SET UPPER-RIGHT HAND CORNER OF THE MASS MATRIX
                                                                          LLEX4510
      AM(I,4)=AM(I,4)+ AMFP(IFP)*(-DC(I,2)*ZFP(IFP)+DC(I,3)*YFP(IFP) )
                                                                          LLEX4520
      AM(1,5)=AM(1,5)+ AMFP(IFP)*( DC(1,1)*ZFP(IFP)-DC(1,3)*XFP(IFP) )
                                                                          LLEX4530
  500 AM(I,6)=AM(I,6)+ AMFP(IFP)*(-DC(I,1)*YFP(IFP)+DC(I,2)*XFP(IFP) )
                                                                          LLEX4540
C
               SET LOWER-LEFT HAND CORNER OF THE MASS MATRIX
                                                                          LLEX4550
      AM(5,1) = AM(5,1) + AMFP(IFP) * ZFPS(1,IFP)
                                                                          LLEX4560
      AM(6,1) = AM(6,1) - AMFP(IFP) * YFPS(1,IFP)
                                                                          LLEX4570
      AM(6,2) = AM(6,2) + AMFP(IFP) * XFPS(1,IFP)
                                                                          LLEX4580
      IF (INDFPC(IFP).LT. 0 ) GO TO 530
                                                                          LLEX4590
      INDFPC(IFP) = -1
                                                                          LLEX4600
      AK = AMFP(IFP)
                                                                          LLEX4610
  510 A = YFP(IFP)*YFP(IFP)
                                                                          LLEX4620
      B = ZFP(IFP) * ZFP(IFP)
                                                                          LLEX4630
      С
         = XFP(IFP) * XFP(IFP)
                                                                          LLEX4640
      AM(1,1) = AM(1,1) + AK
                                                                          LLEX4650
      AM(4,4) = AM(4,4) + AK*(A + B)
                                                                          LLEX4660
      AM(5,5) = AM(5,5) + AK*(C + B)
                                                                          LLEX4670
      AM(6,6) = AM(6,6) + AK*(C + A)
                                                                          LLEX4680
      AM(5,4) = AM(5,4) - AK* XFP(IFP) * YFP(IFP)
                                                                          LLEX4690
      AM(6,4) = AM(6,4) - AK* XFP(IFP) * ZFP(IFP)
                                                                          LLEX4700
      AM(6,5) = AM(6,5) - AK* YFP(IFP) * ZFP(IFP)
                                                                          LLEX4710
      GO TO 530
                                                                          LLEX4720
  520 IF ( INDFPC(IFP) .GT. 1 ) GO TO 530
                                                                          LLEX4730
      AK = -AMFP(IFP)
                                                                          LLEX4740
      INDFPC(IFP) = 2
                                                                          LLEX4750
      GO TO 510
                                                                          LLEX4760
  530 CONTINUE
                                                                          LLEX4770
      AM(2,2) = AM(1,1)
                                                                          LLEX4780
      AM(3,3) = AM(1,1)
                                                                          LLEX4790
      AM(4,5) = AM(5,4)
                                                                          LLEX4800
      AM(4,6) = AM(6,4)
                                                                          LLEX4810
      AM(5,6) = AM(6,5)
                                                                          LLEX4820
      AM(4,2) = -AM(5,1)
                                                                          LLEX4830
      AM(4,3) = -AM(6,1)
                                                                          LLEX4840
      AM(5,3) = -AM(6,2)
                                                                          LLEX4850
С
                                                                          LLEX4860
С
          IF NECESSARY, DEFIND THE MASS MATRIX INVERSE
                                                                          LLEX4870
С
                                                                          LLEX4880
      CALL MATINV( AM(1,1), AMINV(1,1), 6 )
                                                                          11 FX4890
  540 CONTINUE
                                                                          LLEX4900
С
                                                                          LLEX4910
С
          SET UP THE FORCE/TORQUE ARRAY
                                                                          LLEX4920
С
                                                                          LLEX4930
С
               LEG FORCE/TORQUE EFFECTS
                                                                          LLEX4940
           600 I = 1, 6
      DO
                                                                          LLEX4950
  600 FTS(I) =FLTL(I)
                                                                          LLEX4960
               PLANET GRAVITY EFFECTS
C
                                                                          LLEX4970
      FTS(1) = FTS(1) - GCOSZT* AM(1,1)
                                                                           LLEX4980
      FTS(3) = FTS(3) + GSINZT* AM(1,1)
                                                                          LLEX4990
```

		∧ = WX ★ WX	LLEX5000
		B = WY * WY	LLEX5010
		C = WZ * WZ	LLEX5020
		IF ( INDFAR .EQ. 3 ) GO TO 610	LLEX5030
		FO(1) = -WX*(WZ*CBIXY-WY*CBIXZ)-(C-B)*CBIYZ-WY*WZ*CIZZYY	LLEX5040
		FO(2) = -WX*(WX*CBIYZ-WZ*CBIXY) - (A-C)*CBIXZ-WX*WZ*CIXXZZ	
_		FO(3) = -WZ*(WY*CBIXZ-WX*CBIYZ)-(B-A)*CBIXT-WX*WT*CITTAA	LLEX5070
C		RIGID BODY INERITA LOAD EFFECTS	LLFX5080
		$F_{1,3}(4) = F_{1,3}(4) + F_{0}(2)$	LLEX5090
		FTS(6) = FTS(6) + FO(3)	LLEX5100
	610	CONTINUE	LLEX5110
		IF ( NMODES .EQ. 0 ) GO TO 640	LLEX5120
С		INERTIA LOADS DUE TO ELASTIC MODES	LLEX5130
		DO $625$ I = 1,6	LLEX5140
	625	FE(I) = 0.0	LLEX5160
		DO 630 I = 1, NMODES $E_{1} = E_{1} + (DY(I) + WN(I) * G((I \cdot I))) * G(D(I \cdot I))$	LLEX5170
		FE(1) = FE(1) + (PYZ(1) + WNY(1) * GC(1)) / GCD(1) / FE(2) = FE(2) + (PYZ(1) + WNY(1) * GC(1)) / FC(1) / FZ(1) / FZ(1) + WNY(1) * GC(1) / FZ(1) / FZ	LLEX5180
		FF(3) = FE(3) + (PXY(I) + WNZ(I) * GC(1 + I)) * GCD(1 + I)	LLEX5190
		FE(4) = FE(4) + (PYNZ2(I) + WNZNY(I)*GC(1,I))*GC(1,I)	LLEX5200
		FE(5) = FE(5) + (PZNX2(I) + WNXNZ(I)*GC(1,I))*GC(1,I)	LLEX5210
	630	FE(6) = FE(6) + (PXNY2(I) + WNYNX(I) *GC(1,I)) *GC(1,I)	LLEX5220
		FE(4) = FE(4) + WX*(FE(1) + FE(1))	
		FE(5) = FE(5) + WY*(FE(2) + FE(2))	LLEX5250
		FE(6) = FE(6) + WZ*(FE(5) + FE(5))	LLEX5260
		FTS(4) = FTS(5) = FF(5)	LLEX5270
		FTS(6) = FTS(6) - FE(6)	LLEX5280
	640	CONTINUE	LLEX5290
С			LLEX5300
С		CALCULATE THE CENTER BODY ACCELERATIONS AND ANGULAR RATES OF	
С		CHANGE	LLEX5320
		DO = 670 I = 1 + 6	LLEX5340
			LLEX5350
	670	AHOLD(I) = AHOLD(I) + AMINV(I,J) * FTS(J)	LLEX5360
	010	XSDD = AHOLD(1)	LLEX5370
		YSDD = AHOLD(2)	LLEX5380
		ZSDD = AHOLD(3)	LLEX5390
		WXD = AHOLD(4)	11 FX5410
		WYD = AHOLD(5)	LLEX5420
			LLEX5430
			LLEX5440
		IF(INDFZD.NE.0)ZSDD=0.0	LLEX5450
		IF(INDFXR.NE.O)WXD=0.0	LLEX5460
		IF(INDFYR.NE.0)WYD=0.0	LLEX54/0
		IF(INDFZR.NE.O)WZD=0.0	
		IF ( NMODES .EQ. 0 ) GO TO /50	LIEVSSOO
Ç		DRAG STRUT	LLEX5510
c		FLASTIC-ROTATIONAL COUPLINGS (GFO)	LLEX5520
c		MAIN STRUT	LLEX5530
		DO 730 I = 1, NMODES	LLEX5540
		GFO(I) = A*( PYZ(I)+WNX(I)*GC(1,I))+ B*(PXZ(I)+WNY(I)*GC(1,I))	LLEX5550

1	GE(I) = GEO(I) + GE(I)	+ C*(PXY(I)+WNZ(I)*GC(1,I))	LLEX5560
c		1	
~ 720			
750	CONTINUE		
150	LE ( INDEXD NE O ) CO TO 1000	1	
	THE CONDEND OF GO TO 1000		
	CALL INTEG ( XSDD) XSD )		
1000	LE ( INDEVD NE 0 \ CO TO 1010		
1000	IF ( INDEYD •NE• 0 ) GO TO 1010		LLEX5640
	CALL INTEG (YSDD, YSD)		LLEX5650
	CALL INTEG (YSD , YS )		LLEX5660
1010	IF ( INDFZD .NE. 0 ) GO TO 1020	i	LLEX5670
	CALL INTEG ( ZSDD, ZSD )	1	LLEX5680
	CALL INTEG ( ZSD , ZS )		LLEX5690
1020	IF ( INDFXR .NE. 0 ) GO TO 1030		LLEX5700
	CALL INTEG ( WXD , WX )		LLEX5710
	CALL INTEG ( PHID, PHI )		LLEX5720
1030	IF ( INDFYR .NE. 0 ) GO TO 1040		LLEX5730
	CALL INTEG ( WYD , WY )		LLEX5740
	CALL INTEG ( THTAD, THTA )		LLEX5750
1040	IF ( INDFZR .NE. 0 ) GO TO 1050		LLEX5760
	CALL INTEG ( WZD , WZ )		LLEX5770
	CALL INTEG ( PSID, PSI )		LLEX5780
1050	CONTINUE		LLEX5790
	IF ( NMODES .EQ. 0 ) GO TO 1070		LLEX5800
	DO 1060 I= 1, NMODES		LLEX5810
	CALL INTEG ( GCDD(1.1) . GCD(1.1)	, -	LLEX5820
1060	CALL INTEG ( GCD (1.1)) GCD(1917)		LLEX5830
1000	CONTINUE	,	
c 1070	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	****	
•	IE ( IPTATE .EQ. 0 ) 60 TO 2010		LLEX5860
с			LLEX5870
č	CHECK FOR A VARIABLE PAST OR A	AT IT*S CUTOFE VALUE.	1 L E X 5880
č			LLEX5890
•			LI EX5900
c		AT CUTOEE	LLEX5010
C	TE ( ICUT GT. 0 ) GO TO 3000		LLEX5920
	I = ( JCUT = E0 - 0 ) GO TO 2010		
c			
C	JOUT VARIADLE P	PAST 11*5 CUTUFF VALUE	
2005	$\frac{1}{2005} 1 = 1945$		
2005	$\frac{11}{1000} = \frac{11}{1000} = $		LLENDYOU
2004	DU 2000 I = 529 LIIMMS		
2006	IIMHSA(I) = IIMHSC(I)		LLEX5980
~	INDBPU = 1		LLEX5990
C			LLEX6000
C	RESET FOOTPAD CONTACT INDICATO	OR IF THEY WERE JUST CHANGED	LLEX6010
C	AFIER THE CALL TO UPDATE.	•	LLEX6020
C			LLEX6030
	IF ( INDIVR •EQ• 0 ) GO TO 1		LLEX6040
	DO 2007 I = 1, INDIVR		LLEX6050
	IFP = INOUTV(I)		LLEX6060
2007	INDFPC(IFP) = 2		LLEX6070
	INDIVR = 0		LLEX6080
	GO TO 1		LLEX6090
2010	CONTINUE		LLEX6100
С	**************	****	LLEX6110

LLEX6120 IF ( IFIN .NE. 0 ) GO TO 2000 LLEX6130 IF ( IVARH .EQ. 1 ) GO TO 1910 LLEX6140 IF ( IVAL .LT. 0 ) GO TO 2000 ( INDIVR .EQ. 0 ) GO TO 1920 LLEX6150 1910 IF LLEX6160 IK= 1, INDIVR DO 424 IFP = INOUTV(IK) LLEX6170 THIS FOOTPAD HAS JUST LEFT THE LANDING SURFACE. REMOVE LLEX6180 С ITS DISPLACEMENT, VELOCITY, AND ACCELATION VARIABLES LLEX6190 С LLEX6200 С FROM THE INTEGRATION LIST = INDFPI(IFP) LLEX6210 .1 K = J+9 LLEX6220 IF ( K .GT. IPTOTL ) GO TO 421 LLEX6230 DO 420 I = K, IPTOTL LOCNAM(J) = LOCNAM(I) LLEX6240 LLEX6250 LLEX6260 420 J = J + 1LLEX6270 421 CONTINUE IPTOTL = IPTOTL - 9LLEX6280 С SET XFPS AS A CUTOFF VARIABLE LLEX6290 LLEX6300 CALL LOC(140+4*IFP,ATTH(IFP)+(.5+GRAVE*2.5E-06)*CUTERR,1) С LLEX6310 LLEX6320 С UPDATE THE INTEGRATION VARIABLE POINTERS С LLEX6330 LLEX6340 DO 423 I = 1, NFTPDS IF ( INDFPC(I) .LE. 0 ) GO TO 423 LLEX6350 IF ( INDFPI(I).GT. INDFPI(IFP)) INDFPI(I) = INDFPI(I) - 9 LLEX6360 LLEX6370 423 CONTINUE IF ( IK .EQ. INDIVR ) GO TO 425 LLEX6380 LLEX6390 K = IK + 1DO 422 I = K, INDIVR LLEX6400 J = INOUTV(I)LLEX6410 IF ( INDFPI(J) .GT. INDFPI(IFP)) INDFPI(J) = INDFPI(J) - 9 LLEX6420 422 CONTINUE LLEX6430 LLEX6440 425 CONTINUE LLEX6450 UPDATE THE CUTOFF VARIABLE POINTER С INDFPI(IFP) = IPTATL LLEX6460 LLEX6470 424 CONTINUE С LLEX6490 1920 CONTINUE IPRNT .EQ. O EACH TIME THE VARIABLE (TIME) IS A MULTIPLE LLEX6500 С LLEX6510 С OF HMAX. LLEX6520 IF (IPRNT .NE. 0 ) GO TO 2000 LLEX6530 KPTCNT = KPTCNT + 1LLEX6540 IF(IFPPRT.EQ.0)GO TO 1999 LLEX6550 IPRINT=IPRTFP(1) LLEX6560 DO 1998 I=1,NFTPDS 1998 IF(IPRTFP(I).LT.IPRINT)IPRINT=IPRTFP(I) LLEX6570 1999 CONTINUE LLEX6580 IF(IFPPRT.EQ.0) IPRINT=IPTCNT LLEX6590 IF (KPTCNT.LT. IPRINT) GO TO 2000 LLEX6600 LLEX6610 KPTCNT = 0TIMTST = TIME LLEX6620 LLEX6630 С CALL OUTPUT LLEX6640 LLEX6650 IF ( HZ .LT. 1.E-32 ) STOP LLEX6660 C LLEX6670 C CHECK FOR TIME MAX.

IF ( TIME.GE. TIMAX ) GO TO 3110 CHECK FOR STABILITY LLEX6680 С LLEX6690 IF(ISTAB.NE.0)GO TO 3110 LLEX6700 С 2000 CONTINUE LLEX6720 LLEX6730 IF ( IFIN .NE. 0 ) GO TO 2060 TE ( IVARH .EQ. 1 ) GO TO 2060 LLEX6740 LLEX6750 ( IVAL .GE. 0 ) GO TO 2060 IF IF THE INTEGRATION IS GOING TO BE BACKED UP TO REDUCE THE С LLEX6760 STEP SIZE, THEN THE TIME HISTORY COMMON MUST BE BACKED С 1 LEX6770 UP AT THIS TIME LLEX6780 C LLEX6790 2020 CONTINUE DO 2030 I = 1,45 LLEX6800 2030 TIMHSA(I) = TIMHSC(I)LLEX6810 DO 2031 I = 52, LTIMHS LLEX6820 LLFX6830 2031 TIMHSA(I) = TIMHSC(I)LLEX6840 INDBPU = 1RESET FOOTPAD CONTACT INDICATOR IF THEY WERE JUST CHANGED LLEX6850 С С AFTER THE CALL TO UPDATE. LLEX6860 IF ( INDIVR .EQ. 0 ) GO TO 2060 LLEX6870 LLEX6880 DO 2037 I = 1, INDIVR IFP = INOUTV(I)LLEX6890 LLEX6900 2037 INDFPC(IFP) = 2 LLEX6910 INDIVR = 0LLEX6920 С С LLEX6930 CALL INTEGRATION UPDATE С LLEX6940 С LLEX6960 2060 CALL UPDAT LLEX6970 IF ( IFIN .NE. 0 ) GO TO 1 LLEX6980 INDIVR=0 DO 2070 IFP = 1, NFTPDS LLEX6990 IF ( INDFPC(IFP) .LE. 0 ) GO TO 2070 CHECK TO SEE IF FOOT PAD HAS LEFT THE GROUND LLEX7000 LLEX7010 С IF(XFPS(1,IFP).LT.ATTH(IFP)+2.*CUTERR) GO TO 2070 LLEX7020 LLEX7030 С SAVE AND UPDATE INDICATORS FOR INTEGRATION VARIABLES TO BE LLEX7040 C С REMOVE FROM THE INTEGRATION LIST IF THIS STEP LLEX7050 IS ACCEPT BY THE CUTOFF ROUTINE AND THE VARIABLE STEP Ç LLEX7060 C SIZE OPTION 11 FX7070 С LLEX7080 LLEX7090 INDIVR = INDIVR + 1INOUTV(INDIVR) = IFPLLEX7100 Ç SET THE CONTACT INDICATOR LLEX7110 INDFPC(IFP) = 0LLEX7120 RECALCULATE THE F.P. LANDER COORDINATES С LLEX7130 LANDER COORDINATES = DC**(-1)*(SURFACE COORDINATES F.P. LLEX7140 С - SURFACE COORDINATES C.G.) LLEX7150 С = XFPS(1, IFP) - XS LLEX7160 А = YFPS(1, IFP) - YSLLEX7170 В = ZFPS(1, IFP) - ZSLLEX7180 C XFP(IFP)=DC(1,1)*A + DC(2,1)*B + DC(3,1)*C LLEX7190 LLEX7200 YFP(IFP)=DC(1,2)*A + DC(2,2)*B + DC(3,2)*C ZFP(IFP)=DC(1,3)*A + DC(2,3)*B + DC(3,3)*C LLEX7210 2070 CONTINUE LLEX7220 IF ( NMODES .EQ. 0 ) GO TO 2090 LLEX7230

```
DO 2080 I = 1, 3
                                                                              LLEX7240
 2080 \text{ AME1(I)} = \text{AME2(I)}
                                                                              LLEX7250
 2090 CONTINUE
                                                                              LLEX7260
      DO 2095 I = 1. LTIMHS
                                                                              LLEX7270
      IF ( INDBPU .EQ. 1 ) GO TO 2095
                                                                              LLEX7280
      TIMHSC(I) = TIMHSB(I)
                                                                              LLEX7290
 2095 \text{ TIMHSB}(I) = \text{TIMHSA}(I)
                                                                              LLEX7300
      INDBPU = 0
                                                                              LLEX7310
      GO TO 1
                                                                             LLEX7320
C
                     С
                                                                             LLEX7340
С
          MAKE SURE THIS TIME POINT IS PRINTED
                                                                              LLEX7350
С
                                                                              LLEX7360
 3000 IF ( TIMTST .NE. TIME ) CALL OUTPUT
                                                                              LLEX7370
      TIMTST = TIME
                                                                              LLEX7380
      I = (IAD(JCUT) - 140)
                                                                              LLEX7390
           FIND THE SUBSCRIPT OF THE FOOT PAD
С
                                                                              LLEX7400
С
           ADD X-Y-ZFPS(I/4), X-Y-ZFPSD(I/4), AND X-Y-ZFPSDD(I/4)
                                                                              LLEX7410
      TO INTEGRATION LIST
CALL INUPD( 140 + I )
С
                                                                              LLEX7420
                                                                              LLEX7430
      J = I/4
                                                                              LLEX7440
С
                                                                              LLEX7450
С
           UPDATE THE CUTOFF VARIABLE POINTERS
                                                                              LLEX7460
С
                                                                              LLEX7470
      DO 3005 K = 1, NFTPDS
                                                                              LLEX7480
      IF ( INDFPC(K) .GT. 0 ) GO TO 3005
                                                                              LLEX7490
      IF ( INDFPI(K) \bulletGT \bullet INDFPI(J)) INDFPI(K) = INDFPI(K) - 1
                                                                              LLEX7500
 3005 CONTINUE
                                                                              LLEX7510
      UPDATE THE INTEGRATION VARIABLE POINTER INDFPI(J) = IPTOTL
С
                                                                              LLEX7520
                                                                              LLEX7530
      CALL INUPD( 160 + I )
                                                                              LLEX7540
      CALL INUPD( 180 + I )
                                                                              LLEX7550
      CALL INUPD( 200 + I )
                                                                              LLEX7560
      CALL INUPD( 220 + I )
                                                                              LLEX7570
      CALL INUPD( 240 + I )
CALL INUPD( 260 + I )
                                                                              LLEX7580
                                                                              LLEX7590
      CALL INUPD( 280 + I )
                                                                              LLEX7600
      CALL INUPD( 300 + I )
                                                                              LLEX7610
          = WY * ZFP(J) - WZ * YFP(J)
= WZ * XFP(J) - WX * ZFP(J)
= WX * YFP(J) - WY * XFP(J)
      Α
                                                                              LLEX7620
      B
                                                                              LLEX7630
      С
                                                                              LLEX7640
      XFPSD(1,J) = XSD + DC(1,1)*A + DC(1,2)*B + DC(1,3)*C
                                                                              LLEX7650
      YFPSD(1,J) = YSD + DC(2,1)*A + DC(2,2)*B + DC(2,3)*C
                                                                              LLEX7660
      ZFPSD(1,J) = ZSD + DC(3,1)*A + DC(3,2)*B + DC(3,3)*C
                                                                              LLEX7670
      INDFPC( J )
                     = 1
                                                                              LLEX7680
      I = JCUT + 1
                                                                              LLEX7690
      IF ( I .GT. IPTATL ) GO TO 3015
                                                                              LLEX7700
      DO 3010 K= I , IPTATL
                                                                              LLEX7710
      IAD(JCUT) = IAD(K)
                                                                              LLEX7720
      XSI(JCUT) = XSI(K)
                                                                              LLEX7730
      IND(JCUT) = IND(K)
                                                                              LLEX7740
      XMN1(JCUT) = XMN1(K)
                                                                              LLEX7750
 3010 \text{ JCUT} = \text{JCUT} + 1
                                                                              LLEX7760
 3015 CONTINUE
                                                                              LLEX7770
      IPTATL = IPTATL-1
                                                                              LLEX7780
C
                                                                              LLEX7790
```

C	CHECK FOR SIMULTANEOUS CUTOFF VARIABLES	LLEX7800
<b>C</b>	15 / 187411 - FO. 0 ) 60 TO 2020	LLEX7820
		LLEX7830
	LE ( JCUT +GT+ 0 ) GO TO 3000	LLEX7840
		LLEX7850
с		LLEX7860
Ċ	RESET THE INTEGRATION PRINT AND STEP SIZE CONDITIONS	LLEX7870
С		LLEX7880
3030	CONTINUE	LLEX7890
	CALL SETUP	LLEX7900
	DO $3040 I = 1, 6$	LLEX7910
3040	TIMHSA(I) = TIMHSB(I)	LLEX7920
	DO 3045 I = 43, 45	LLEX7930
3045	TIMHSA(I) = TIMHSB(I)	LLEX7940
	DO 3050 I = 58, LTIMHS	LLEX7950
3050	TIMHSA(I) = TIMHSB(I)	LLEX7960
	DO 3060 I = 1, LTIMHS	LLEX7970
3060	TIMHSC(I) = TIMHSA(I)	LLEX7980
	INDBPU = 1	LLEX7990
	GO TO 1	LLEX8000
C	***************************************	*LLEX8010
C		LLEX8020
C	END OF CASE	LLEX8030
C		LLEX8040
3110	CONTINUE	LLEX8050
C		LLEX8060
~	IF(IBOTM•NE•O)CALL OUTPUT	LLEX8070
C		LLEXGUOU
~	CALL SUMMRY	LLEX8090
Ç		
	REIORN	LLEXOIIO
3200		
2010	WRITE (0)3210)	
2210	A A AND ROTA COULD NOT BE SUCCESSED AND THE SAME THE COUNTY	LIEX8150
	I ATT AND BOTH COOLD NOT BE SUCCESSIVELY COT BACK /	LLEX8160
0000	CALL OUTFOIL	LIEX8170
9000		11 EX8180

a

THIS SUBROUTINE CONTAINS THE LOGIC FOR DETERMINING THE SOILFTPD 20FORCES AND LANDING GEAR STRUT FORCESFTPD 40FOLLOWING NOTATION USED -FTPD 70S.C.S SURFACE COORDINATE SYSTEMFTPD 70L.C.S LANDER COORDINATE SYSTEMFTPD 70CONSIDERATION.FTPD 100STRUT FORCES ON CENTER BODY IN S.C.S.FTPD 110STRUT FORCES ON CENTER BODY IN S.C.S.FTPD 120STRUT FORCES ON CENTER BODY IN S.C.S.FTPD 120DIMENSION XFPSI(4,5), YFPSI(4,5), ZFPSI(4,5),FTPD 1701XFPSDI(4,5), YFPSI(4,5), ZFPSI(4,5),FTPD 120COMMON X THISV/ TIMHSA(1), ATTH(5), AMM(646), AME(13), INDFPI(6),FTPD 220COMMON / THISV/ TIMHSA(1), ATTH(5), AMM(646), AME(13), INDFPI(6),FTPD 220STRUTS (10), URCMS (5), URCMS (5), SETCDS(10), STDS(10), INDCMS(10),FTPD 220STRUTS (5), PRECOS (10), PRETOS (10), FRVDS(10), FTPD 240SETCMS(5), SETCMS(5), INCDS(10), INDCMS(10),FTPD 240STRUPS(10), STRPMS (5), INDCDS(10), FRVDS(10), FRVDS(5),FTPD 240SUMON COMINT(40)SETCMS(5), FRVDSC(10), FRVDS(10), FTPD 240FTPD 240COMMON COMINT(400)FTPD 240FTPD 300EQUIVALENCE (COMINT (3), IBOTM )FTPD 300EQUIVALENCE (COMINT (4), XSD )FTPD 300EQUIVALENCE (COMINT (4), SSD )FTPD 300EQUIVALENCE (COMINT (4), SSD )FTPD 300EQUIVALENCE (COMINT (4), SSD )FTPD 300EQUIVALENCE (COMINT (4), SCD )FTPD 300EQUIVALENCE (COMINT (4), SCD )FTPD 300EQUIVALENCE (COMINT (4), SCD )FTPD 300EQUIVALENCE (C	SUBROUTINE FTPAD	FTPD	10
THIS SUBROUTINE CONTAINS THE LOGIC FOR DETERMINING THE SOILFTPD 30FORCES AND LANDING GEAR STRUT FORCESFTPD 50FOLLOWING NOTATION USED -S.C.S SURFACE COORDINATE SYSTEMFTPD 70L.C.S LANDER COORDINATE SYSTEMFTPD 90THE FOLLOWING INFORMATION IS RETURNED FOR THE LANDING GEAR UNDERFTPD 100CONSIDERATION.SREUT FORCES ON CENTER BODY IN S.C.S.FTPD 110STRUT FORCES ON CENTER BODY IN S.C.S.FTPD 120STRUT TORQUES ON CENTER BODY IN S.C.S.FTPD 140GENERALIZED FORCESFTPD 140DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),FTPD 160DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),FTPD 120COMMON Y THISY/TIMHSA(1), ATTH(5),AM(6,6),AMEL(3),INDFPI(6),FTPD 200COMMON Y THISY/TIMHSA(1), ATTH(5),AM(6,6),AMEL(3),INDFPI(6),FTPD 220STRUTS (10), URCMS (5), URCDS(10), IPOCMS(5), URCDS (10), FTPD 240SETTMS(5), SETTMS(5), IPOCDS(10), FRVDST(10), FTPD 240STRUTS (10), URCMS (5), URCDS (10), FTPD 240SETTMS(5), PRECDS (10), PRECDS (10), FRVDST(10), FTPD 240SUDTMS (5), PRECDS (10), PRETDS (10), FRVDST(10), FRVMSC (5), FTPD 260FTPD 210COMMON COMINT(400)FTPD 230EQUIVALENCE (COMINT( 4), XSD )FTPD 330EQUIVALENCE (COMINT( 6), WY )FTPD 330EQUIVALENCE (COMINT( 6), WY )FTPD 340EQUIVALENCE (COMINT( 64), SCD )FTPD 340EQUIVALENCE (COMINT( 64), SCD )FTPD 340EQUIVALENCE (COMINT( 64), SCD )FTPD 340EQUIVALENCE (COMINT( 64), SCD )FTPD 350EQUIVALENCE (COMINT( 64), SCD )FTPD 350EQUIVA		FTPD	20
FORCES AND LANDING GEAR SIRUI FORCES         FIPD 40           FOLLOWING NOTATION USED -         FTPD 50           S.C.S SURFACE COORDINATE SYSTEM         FTPD 70           L.C.S LANDER COORDINATE SYSTEM         FTPD 90           THE FOLLOWING INFORMATION IS RETURNED FOR THE LANDING GEAR UNDER         FTPD 100           CONSIDERATION.         FTPD 100           STRUT FORCES ON CENTER BODY IN S.C.S.         FTPD 110           STRUT FORCES ON FOOTPAD IN S.C.S.         FTPD 120           STRUT FORCES ON FOOTPAD IN S.C.S.         FTPD 130           STRUT FORCES ON FOOTPAD IN S.C.S.         FTPD 140           GENERALIZED FORCES         FTPD 160           DIMENSION XFPS(4,5), YFPSD(4,5), ZFPS(4,5),         FTPD 170           T XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),         FTPD 180           OCMMON / THISV/ TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDEPI(6),         FTPD 230           C XFPSD(10), STRPMS (5), IDPODS(10), INDCMS(10), FTPD 240         SETEMS(5), SETTMS (5), INDCDS(10), INDCMS(10), FTPD 240           4 SETEMS(5), PRCDS (10), PRFTDS (10), INDCMS(10), INDCMS(10), FTPD 260         FTPD 300           COMMON COMINIT(40)         FTPD 300           FTPD 20(10), IPOTMS(5)         FTPD 300           COMMON COMINIT(40)         FTPD 300           EQUIVALENCE (COMINIT 4), XSD )         FTPD 300           E	THIS SUBROUTINE CONTAINS THE LOGIC FOR DETERMINING THE SOIL	FTPD	30
FIED       50         FOLLOWING NOTATION USED -       FIED       50         S.C.S SURFACE COORDINATE SYSTEM       FIED       70         L.C.S LANDER COORDINATE SYSTEM       FIED       70         THE FOLLOWING INFORMATION IS RETURNED FOR THE LANDING GEAR UNDER       FIED       90         CONSIDERATION.       FRETURNED FOR THE LANDING GEAR UNDER       FIED       90         STRUT FORCES ON CENTER BODY IN S.C.S.       FIED       10         STRUT TORQUES ON CENTER BODY IN L.C.S.       FIED       10         GENERALIZED FORCES       FIED       100         DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),       FIED       160         DIMENSION XFPS(4,5), YFPS(4,5), ZFPS0(4,5),       FIED       100         COMMON / THISV/ TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),       FIED       20         COMMON / THISV/ TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),       FIED       20         GUIVALENCE (COMINT (S), INOCDS(10), INOCDS(10), INOCDS(10), FIED       20       20         GUIVALENCE (COMINT (A), SETTMS (S), FRODS(10), FRVMSC (S), FRETMS (S), FIED       20         COMMON COMINT(400)       FIED       20       FIED       20         COMMON COMINT(400)       FIED       20       FIED       20       20       20       20	FORCES AND LANDING GEAR STRUE FORCES	FIPD	40
FOLLOWING NOTATION USED -       FIPD 60         S.C.S SURFACE COORDINATE SYSTEM       FIPD 70         L.C.S LANDER COORDINATE SYSTEM       FIPD 90         THE FOLLOWING INFORMATION IS RETURNED FOR THE LANDING GEAR UNDER       FIPD 100         CONSIDERATION.       STRUT FORCES ON CONTRATE BODY IN S.C.S.       FIPD 120         STRUT FORCES ON CENTER BODY IN L.C.S.       FIPD 1100         DIMENSION XFPS(4,5), YFPS0(4,5), ZFPS0(4,5),       FIPD 120         DIMENSION XFPS(4,5), YFPSD(4,5), ZFPS0(4,5),       FIPD 120         OKOMON / THISV/TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),       FIPD 120         OKOMON / THISV/TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),       FIPD 220         OKOMON / THISV/TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),       FIPD 220         STRUS (10), STRPMS (5), URDS (10), INDCDS(10), INDCMS(10),       FTPD 250         SUDTMS (5), PRFCDS (10), FRVDS(10), FRVDS(10), FRVMS( 5),       FIPD 220         STRUS (10), STRPMS (5), INDCDS(10), INDCMS(10),       FTPD 250         SUDTMS (5), PRFCDS (10), FRVDS(10), FRVMS( 5),       FTPD 260         G INDTMS (5), PRFCDS (10), FRVDS(10), FRVMS( 5),       FTPD 260         G INDTMS (5), PRFCDS (10), FRVDS(10), FRVMS( 5),       FTPD 260         G UIVALENCE (COMINT (3), IBOTM )       FTPD 300         EQUIVALENCE (COMINT (4), XSD )       FTPD 320		FTPD	50
S.C.S SURFACE COORDINATE STSTEM       FIPD 80         L.C.S LANDER COORDINATE STSTEM       FIPD 80         THE FOLLOWING INFORMATION IS RETURNED FOR THE LANDING GEAR UNDER       FIPD 100         CONSIDERATION.       FIPD 100         STRUT FORCES ON CENTER BODY IN S.C.S.       FIPD 120         STRUT TORQUES ON CENTER BODY IN S.C.S.       FIPD 140         GENERALIZED FORCES       FIPD 170         DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),       FIPD 170         1       XFPSD(4,5), YFPSD(4,5), ZFPS(4,5),       FIPD 170         2       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FIPD 170         1       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FIPD 120         COMMON / THISV/ TIMHSA(1), ATH(5), AM(6,6), AME1(3), INDFPI(6),       FTPD 220         2       STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10), FTPD 240       FTPD 240         3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETDS(10), FTPD 240       FTPD 250         4       SETCMS(5), SETTMS(5), IPOCDS(10), INDCMS(10), FTPD 240       FTPD 320         5       INDTMS (5), FRECS (10), PRTDS (10), FRVMSC(5), FTPD 260       FTPD 320         6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVMSC(5), FTPD 270       FTPD 320         6       IPRDS (10), IPRMS (5), IPROTSC (10), FRVMSC(5), FTPD 320       FTPD 320         6 <td>FOLLOWING NOTATION USED -</td> <td>FTPD</td> <td>60</td>	FOLLOWING NOTATION USED -	FTPD	60
Laces - LANDER COURDINATE SYSTEM FIPD 80 FTPD 90 THE FOLLOWING INFORMATION IS RETURNED FOR THE LANDING GEAR UNDER FTPD 110 STRUT FORCES ON CENTER BODY IN s.c.s. FTPD 120 STRUT FORCES ON FOOTPAD IN S.c.s. FTPD 140 GENERALIZED FORCES STRUT TORQUES ON CENTER BODY IN L.C.S. FTPD 140 GENERALIZED FORCES DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5), FTPD 160 DIMENSION STPS(4,5), YFPS(4,5), ZFPS(4,5), FTPD 170 XFPSD(4,5), GCO(4,5), GCD(4,5) COMMON / THISV/ TIMHSA(1), ATTH(5),AM(6,6),AME1(3),INDFPI(6), FTPD 200 COMMON / THISV/ TIMHSA(1), ATTH(5),AM(6,6),AME1(3),INDFPI(6), FTPD 210 TNDFPC(6), STRMS(5), IPOCDS(10), IPOCMS(5), URCDS (10), FTPD 230 3 URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10), FTPD 230 3 URTDS (10), URCMS (5), URTMS (5), SETCDS(10), INDCMS(10), FTPD 250 5 INDINS(5), PRFCDS (10), PRFTDS (10), PRFCMS (5), PRFTMS (5), FTPD 260 COMMON COMINT(40) FTPD 5(10), IPOTMS(5) COMMON COMINT(40) FTPD 520 EQUIVALENCE (COMINT( 3), IBOTM ) FTPD 320 EQUIVALENCE (COMINT( 4), XSD ) FTPD 320 EQUIVALENCE (COMINT( 10), IPOTMS(5) FTPD 320 EQUIVALENCE (COMINT( 10), YSD ) FTPD 320 EQUIVALENCE (COMINT( 10), WX ) FTPD 320 EQUIVALENCE (COMINT( 10, WX ) FTPD 320 EQUIVALENCE (COMINT( 14, WX ) FTPD 370 EQUIVALENCE (COMINT( 14, WX ) FTPD 370 EQUIVALENCE (COMINT( 14, WX ) FTPD 370 EQUIVALENCE (COMINT( 164, YFPSD ) FTPD 430 EQUIVALENCE (COMINT( 164, YFPSD ) FTPD 430 EQUIVAL	S-C-S- SURFACE COORDINATE SYSTEM	FIPD	70
THE FOLLOWING INFORMATION IS RETURNED FOR THE LANDING GEAR UNDER       FIPD 100         CONSIDERATION.       FIPD 100         STRUT FORCES ON CENTER BODY IN S.C.S.       FIPD 120         STRUT FORCES ON CENTER BODY IN S.C.S.       FIPD 140         GENERALIZED FORCES       FIPD 140         DIMENSION XFPS(4,5), YFPS(4,5), ZFPSD(4,5),       FIPD 160         DIMENSION XFPS(4,5), YFPSD(4,5), ZFPSD(4,5),       FIPD 170         1       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FIPD 170         1       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FIPD 170         2       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FIPD 170         1       NCPPS(1,5), GCD(4,5), ZFPSD(4,5),       FIPD 190         DIMENSION GC(4,5), GCD(4,5), GCDD(4,5)       FIPD 210       FIPD 210         1       INDFPC(6),       FIPD 220       FIPD 230         2       STRPMS(15), INDCDS(10), INDCMS(10), FIPD 240       SETCMS(5), SETIMS(5), INDCDS(10), FRVDST(10), FRVMSC(5), FIPD 260         6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5), FIPD 260       FIPD 300         6       IPRDS (10), IPRMS (5), INDCMS(10), FRVMSC(5), FIPD 260       FIPD 300         6       IPRDS (10), IPRTMS (5)       FIPD 320         6       IPRDS (10), IPRTMS (5)       FIPD 320         6       IPRDS (10)	L.C.S LANDER COORDINATE SYSTEM	FIPD	80
THE FOLLOWING INFORMATION IS RETORNED FOR THE LANDING GEAR UNDER FIPD 110         STRUT FORCES ON CENTER BODY IN S.C.S.       FTPD 120         STRUT FORCES ON CENTER BODY IN S.C.S.       FTPD 130         STRUT FORCES ON CENTER BODY IN L.C.S.       FTPD 140         GENERALIZED FORCES       FTPD 160         DIMENSION XFPS(4,5), YFPSD(4,5), ZFPS(4,5),       FTPD 170         1       XFPSD1(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         2       XFPSD1(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         1       XFPSD1(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         2       XFPSD1(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         2       XFPSD1(4,5), GCD(4,5), GCD(4,5),       FTPD 200         COMMON / THISV/ TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),       FTPD 220         2       STRPDS(10), STRPMS(5), IPOCDS(10), INDCMS(5), URDMS(10), FTPD 240       FTPD 270         3       URTDS (10), PRRMS(5), IPOTMS(5)       FTPD 170         4       STCMS(5), SETTMS(5), IPOTDS(10), FRVDST(10), FRVMSC(5), FTPD 280       FTPD 320         5       INDTMS(5), PRFCDS (10), FNCDST(10), FRVMSC(5), FTPD 280       FTPD 340         COMMON COMINT(400)       FTPD 320       FTPD 340         EQUIVALENCE ( COMINT( 4), XSD )       FTPD 340       FTPD 340         EQUIVALENCE ( COMINT( 4), X	THE FOLLOWING INCOMMITION TO DETURNED FOR THE LANDING SEAR UNDER	FIPD	,90
CONSIDERATION:       FIPD 101         STRUT FORCES ON CENTER BODY IN S.C.S.       FTPD 120         STRUT TORCUES ON CENTER BODY IN L.C.S.       FTPD 140         GENERALIZED FORCES       FTPD 160         DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),       FTPD 170         1       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         2       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         1       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         2       XFPSD(4,5), GCD(4,5), GCD(4,5),       FTPD 170         1       INOFPC(6),       FTPD 210         2       STRPDS(10), STRPMS(5), ICODS(10), IPOCMS(5), URCDS (10), FTPD 230         3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10), FTPD 240         4       SETCMS(5), SETTMS(5), INOCDS(10), INDTDS(10), INOCMS(10), FTPD 260         6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5), FTPD 260         7       INDTMS(5), PRECDS (10), FRVDST(10), FRVMSC(5), FTPD 270         2       GOMINA COMINT(40)         4       SETCMS(10), IPRMS (5), IBOTM       FTPD 310         5       INDTMS (5), FTPD 320       FTPD 320         6       IPRDS (10), IPRMS (5), ZSETDS (10), FRVMSC(5), FTPD 320       FTPD 340         5       INDTMS (5), FTPD 320       FTPD 340	CONSIDERATION IS REFORMED FOR THE LANDING GEAR UNDER	FIPU	100
STRUT FORCES ON FOOTPAD IN S.C.S.       FTPD 130         STRUT TORQUES ON CENTER BODY IN L.C.S.       FTPD 140         GENERALIZED FORCES       FTPD 150         DIMENSION XFPS(4,5), YFPSD(4,5), ZFPS(4,5),       FTPD 160         1       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         1       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 180         2       XFPSDD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 180         2       XFPSDD(4,5), GCD(4,5), GCD(4,5),       FTPD 200         COMMON / THISV/ TIMHSA(1), ATTH(5), AM(6,6), AMEI(3), INDFPI(6),       FTPD 220         3       URTDS (10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10), FTPD 230         3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10), FTPD 250         5       INDTMS(5), SETTMS(5), INOTS(10), FRVDST(10), FRVMSC(5), FTPD 260         6       IPRDS (10), IPOMS (5), IPOTMS(5)       FTPD 320         COMMON COMINT(400)       FTPD 320         EQUIVALENCE ( COMINT( 12 ), YS )       FTPD 330         EQUIVALENCE ( COMINT( 24 ), ZSD )       FTPD 390         EQUIVALENCE ( COMINT( 44 ), WX )       FTPD 390         EQUIVALENCE ( COMINT( 44 ), WX )       FTPD 390         EQUIVALENCE ( COMINT( 44 ), XFPS )       FTPD 390         EQUIVALENCE ( COMINT( 44 ), XFPS )       FTPD 390	STRUT FORCES ON CENTER BODY IN S.C.S.		120
STRUT TORQUES ON CENTER BODY IN L.C.S.       FTPD 140         GENERALIZED FORCES       FTPD 160         DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),       FTPD 170         1       XFPSD(4,5), YFPS(4,5), ZFPSD(4,5),       FTPD 180         2       XFPSD(4,5), YFPS(4,5), ZFPSD(4,5),       FTPD 190         DIMENSION SC(4,5), GCD(4,5), GCD(4,5),       FTPD 210         COMMON / HISV/ TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),       FTPD 230         3       URTDS (10), STRPMS(5), INDCDS(10), IPOCMS(5), URCDS (10), FTPD 240         4       SETDS(10), STRPMS(5), INDCDS(10), INDTDS(10), INDCMS(10), FTPD 240         5       INDTMS(5), PRECDS (10), PRECMS (5), PRETMS (5), FTPD 260         6       IPRDS (10), IPRMS (5), IPOCDS(10), FRVDST(10), FNEMS(5), FTPD 260         7       IFMMS (5), IPOTDS(10), IPOTMS(5)         COMMON COMINT(400)       FTPD 330         EQUIVALENCE ( COMINT( 4), XSD )       FTPD 350         EQUIVALENCE ( COMINT( 24), ZSD )       FTPD 360         EQUIVALENCE ( COMINT( 24), YFPSD )       FTPD 370         EQUIVALENCE ( COMINT( 24), YFPSD )	STRUT FORCES ON CENTER DUD IN SACAS	FTPD	120
GENERALIZED FORCES       FTPD 150         DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),       FTPD 160         1       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         2       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         0       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 170         0       XFPSD(4,5), GCD(4,5), GCD(4,5),       FTPD 170         0       MENSION GC(4,5), GCD(4,5),       GCD(4,5),         0       MENSION GC(4,5), GCD(4,5),       FTPD 220         2       STRPDS1(1), STRPMS(5), IPOCDS(10), IPOCMS(5), MECDS 100),       FTPD 220         3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10),       FTPD 220         4       SETCMS(5), SETTMS(5), IPOCDS(10), INDTDS10), INDCMS(10),       FTPD 260         5       INDTMS(5), IPOTDS(10), IPOTMS(5)       FTPD 320         COMMON COMINT(400)       FTPD 310       FTPD 320         EQUIVALENCE (COMINT(3), IBOTM )       FTPD 330         EQUIVALENCE (COMINT(4), YSD )       FTPD 340         EQUIVALENCE (COMINT(4), YSD )       FTPD 390         EQUIVALENCE (COMINT(76), WZ )       FTPD 370         EQUIVALENCE (COMINT(76), WZ )       FTPD 370         EQUIVALENCE (COMINT(76), WZ )       FTPD 370         EQUIVALENCE (COMINT(76), WZ )       FTPD 370 </td <td>STRUT TORQUES ON CENTER BODY IN LACASA</td> <td>FTPD</td> <td>140</td>	STRUT TORQUES ON CENTER BODY IN LACASA	FTPD	140
DIMENSION         TPD 1600           DIMENSION         XFPS(4,5), YFPS(4,5), ZFPS(4,5),         FTPD 170           1         XFPS(4,5), YFPS(4,5), ZFPS(4,5),         FTPD 180           2         XFPS(4,5), GCD(4,5), GCD(4,5),         FTPD 190           DIMENSION         GC(4,5), GCD(4,5), GCD(4,5),         FTPD 200           COMMON / THISV/ TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),         FTPD 220           2         STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10),         FTPD 220           3         URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10),         FTPD 240           4         SETCMS(5), SETTMS(5), INDCDS(10), INDTDS(10), INDCMS(10),         FTPD 240           5         INDTMS(5), PRFCDS (10), PRTDS (10), PRECMS (5),         FTPD 270           5         INDTMS(5), IPOTDS(10), IPOTMS(5)         FTPD 270           C         FRWST(5), IPOTDS(10), IPOTMS(5)         FTPD 300           EQUIVALENCE (COMINT(4), XSD )         FTPD 330           EQUIVALENCE (COMINT(12), YS)         FTPD 370           EQUIVALENCE (COMINT(44), WX )         FTPD 370           EQUIVALENCE (COMINT(44), WX )         FTPD 370           EQUIVALENCE (COMINT(60), WY )         FTPD 370           EQUIVALENCE (COMINT(64), KPSD )         FTPD 370           EQUIVALENCE (COMINT(144), KPSD )	GENERALIZED EORCES	FTPD	150
DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),       FTPD 170         1       XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),       FTPD 180         2       XFPSD(4,5), GCD(4,5), GCD(4,5),       FTPD 190         DIMENSION GC(4,5), GCD(4,5), GCD(4,5),       FTPD 200         COMMON / THISV/ TIMHSA(1), ATTH(5),AM(6,6),AME1(3),INDFPI(6),       FTPD 210         1       INDFPC(6),       FTPD 230         2       STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10),       FTPD 240         3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10),       FTPD 240         4       SETCMS(5), SETTMS(5), INDCDS(10), INDTDS(10), INDCMS(10),       FTPD 260         6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMST(5),       FTPD 270         C       FRVMST(5), IPOTDS(10), IPOTMS(5)       FTPD 320         COMINON COMINT(400)       FTPD 310       FTPD 320         EQUIVALENCE (COMINT 1 2 ), YS       FTPD 320         EQUIVALENCE (COMINT 1 2 ), YS       FTPD 370         EQUIVALENCE (COMINT 1 2 ), YS       FTPD 390         EQUIVALENCE (COMINT 1 2 ), YS       FTPD 390         EQUIVALENCE (COMINT 1 2 ), YS       FTPD 390         EQUIVALENCE (COMINT 1 4 ), XSD       FTPD 390         EQUIVALENCE (COMINT 1 6 ), WX       FTPD 390         EQUIVALENCE (COMINT 1 6 ), WZ		FTPD	160
1         XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),         FTPD 180           2         XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),         FTPD 190           COMMON / THISV/ TIMHSA(1), ATTH(5),AM(6,6),AME1(3),INDFPI(6),         FTPD 200           1         INDFPC(6),         FTPD 200           2         STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10),         FTPD 230           3         URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10),         FTPD 240           4         SETCMS(5), SETTMS(5), IPOCDS(10), INDCDS(10), INDCDS(10),         FTPD 250           5         INDTMS (5), PRFCDS (10), PRFTDS (10), FRVDSC(10), FRVMSC(5),         FTPD 260           6         IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5),         FTPD 280           COMMON COMINT(400)         IPOTMS(5)         FTPD 310           EQUIVALENCE (COMINT (4), XSD )         FTPD 340           EQUIVALENCE (COMINT (4), YSD )         FTPD 370           EQUIVALENCE (COMINT (6), WX )         FTPD 370           EQUIVALENCE (COMINT (60), WY )         FTPD 420           EQUIVALENCE (COMINT (6), WZ )         FTPD 420           EQUIVALENCE (COMINT (64), KFPS )         FTPD 420           EQUIVALENCE (COMINT (64), KFPS )         FTPD 420           EQUIVALENCE (COMINT (64), KFPS )         FTPD 420           EQUIVALENCE (COMINT	DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),	FTPD	170
2       XFPSDD(4,5), YFPSDD(4,5), ZFPSDD(4,5)       FTPD 190         DIMENSION GC(4,5), GCD(4,5), GCD(4,5), GCD(4,5), GCD(4,5), GCD(4,5), FTPD 210       FTPD 210         1       INDFPC(6),       FTPD 200         2       STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10), FTPD 230         3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10), FTPD 240         4       SETCMS(5), SETTMS(5), INDCDS(10), INDTDS(10), INDCMS(10), FTPD 240         5       INDTMS(5), PRFCDS (10), PRFTDS (10), FRVDS(10), INDCMS(10), FTPD 250         6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVDS(10), FRVMSC(5), FTPD 270         COMMON COMINT(400)       FTPD 300         EQUIVALENCE (COMINT 3), IBOTM )       FTPD 300         EQUIVALENCE (COMINT 12), YS )       FTPD 320         EQUIVALENCE (COMINT 16), YSD )       FTPD 350         EQUIVALENCE (COMINT 16), WX )       FTPD 370         EQUIVALENCE (COMINT 76), WZ )       FTPD 380         EQUIVALENCE (COMINT 76), WZ )       FTPD 420         EQUIVALENCE (COMINT 76), YFPSD )       FTPD 420         EQUIVALENCE (COMINT 76), WZ )       FTPD 370         EQUIVALENCE (COMINT 76), YFPSD )       FTPD 420         EQUIVALENCE (COMINT 76), YFPSD )       FTPD 420         EQUIVALENCE (COMINT 76), YFPSD )       FTPD 420         EQUIVALENCE (COMINT 76), YFPS	XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),	FTPD	180
DIMENSION GC(4,5), GCD(4,5), GCDD(4,5)       FTPD 200         COMMON / THISV/ TIMES(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),       FTPD 210         INDFPC(6),       FTPD 220         2       STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10), FTPD 240         4       SETCMS(5), SETTMS(5), INDCDS(10), INDCMS(10), INDCMS(10), FTPD 240         4       SETCMS(5), SETTMS(5), INDCDS(10), INDCMS(10), FTPD 240         6       IPRCS (10), PRFCDS (10), PRFCMS (5), PRFTMS (5), FTPD 260         6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5), FTPD 270         COMMON COMINT(400)       FTPD 270         EQUIVALENCE (COMINT(1 3), IBOTM )       FTPD 300         EQUIVALENCE (COMINT(1 6), YSD )       FTPD 310         EQUIVALENCE (COMINT(1 6), YSD )       FTPD 350         EQUIVALENCE (COMINT(44), WX )       FTPD 370         EQUIVALENCE (COMINT(60), WY )       FTPD 370         EQUIVALENCE (COMINT(164, SCD )       FTPD 370         EQUIVALENCE (COMINT(164, SCD )       FTPD 370         EQUIVALENCE (COMINT(64, WX )       FTPD 370         EQUIVALENCE (COMINT(64, KCC )       FTPD 470         EQUIVALENCE (COMINT(164, KCC )       FTPD 470         EQUIVALENCE (COMINT(164, KFPSD )       FTPD 470         EQUIVALENCE (COMINT(164, KFPSD )       FTPD 470         EQUIVALEN	XFPSDD(4,5), YFPSDD(4,5), ZFPSDD(4,5)	FTPD	190
COMMON / THISV/ TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),       FTPD 210         1       INDFPC(6),       FTPD 230         2       STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10), FTPD 230         3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10), FTPD 240         4       SETCMS(5), SETTMS(5), INDCDS(10), INDTDS(10), INDCMS(10), FTPD 250         5       INDTMS(5), PRFCDS (10), PRFCDS (10), PRCMS (5), FTPD 260         6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5), FTPD 270         COMMON COMINT(400)       FTPD 310         EQUIVALENCE ( COMINT( 4), XSD )       FTPD 320         EQUIVALENCE ( COMINT( 12), YS )       FTPD 340         EQUIVALENCE ( COMINT( 24), ZS )       FTPD 370         EQUIVALENCE ( COMINT( 24), ZS )       FTPD 370         EQUIVALENCE ( COMINT( 76), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76), WZ )       FTPD 470 <td>DIMENSION GC(4,5), GCD(4,5), GCDD(4,5)</td> <td>FTPD</td> <td>200</td>	DIMENSION GC(4,5), GCD(4,5), GCDD(4,5)	FTPD	200
1       INDFPC(6),       FTPD 220         2       STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10), FTPD 230         3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10), FTPD 240         4       SETCMS(5), PRFCDS (10), INDTDS(10), INDCMS(10), FTPD 250         5       INDTMS(5), PRFCDS (10), PRFDS (10), FRVMS(5), FTPD 260         6       IPRS (10), IPRMS (5), FRVDSC(10), FRVMSC(5), FTPD 270         1       FRVMST(5), IPOTDS(10), IPOTMS(5)         COMMON COMINT(400)       FTPD 290         EQUIVALENCE (COMINT(4), XSD)       FTPD 310         EQUIVALENCE (COMINT(24), YS)       FTPD 320         EQUIVALENCE (COMINT(24), YS)       FTPD 340         EQUIVALENCE (COMINT (24), ZS)       FTPD 370         EQUIVALENCE (COMINT (4), WX)       FTPD 370         EQUIVALENCE (COMINT (60), WY)       FTPD 370         EQUIVALENCE (COMINT (60), WY)       FTPD 400         EQUIVALENCE (COMINT (144), SCD)       FTPD 410         EQUIVALENCE (COMINT (144), SCD)       FTPD 420         EQUIVALENCE (COMINT (144), SCD)       FTPD 420         EQUIVALENCE (COMINT (24), GCD)       FTPD 440         EQUIVALENCE (COMINT (144), XFPSD)       FTPD 440         EQUIVALENCE (COMINT (244), YFPSD)       FTPD 450         EQUIVALENCE (COMINT (244), YFPSD)       FTPD 440<	COMMON / THISV/ TIMHSA(1), ATTH(5), AM(6,6), AME1(3), INDFPI(6),	FTPD	210
2       STRPDS(10), STRPMS(5), IPOCDS(10), IPOCMS(5), URCDS (10), FTPD 230         3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10), FTPD 240         4       SETCMS(5), SETTMS(5), INDCDS(10), INDCDS(10), INDCMS(10), FTPD 240         5       INDTNS(5), PRFCDS (10), PRFCDS (10), PRFCMS (5), PRFTMS (5), FTPD 260         6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5), FTPD 270         COMMON COMINT(40)       FTPD 280         EQUIVALENCE (COMINT(3), IPOTDS(10), IPOTMS(5)       FTPD 290         EQUIVALENCE (COMINT(4), XSD)       FTPD 310         EQUIVALENCE (COMINT(24), YSD)       FTPD 340         EQUIVALENCE (COMINT (24), YSD)       FTPD 350         EQUIVALENCE (COMINT (60), WY)       FTPD 370         EQUIVALENCE (COMINT (60), WZ)       FTPD 370         EQUIVALENCE (COMINT (76), WZ)       FTPD 370         EQUIVALENCE (COMINT (164), SCD)       FTPD 370         EQUIVALENCE (COMINT (164), SCD)       FTPD 370         EQUIVALENCE (COMINT (164), SCD)       FTPD 400         EQUIVALENCE (COMINT (164), SCD)       FTPD 420         EQUIVALENCE (COMINT (164), SCD)       FTPD 420         EQUIVALENCE (COMINT (164), SCPS)       FTPD 420         EQUIVALENCE (COMINT (164), YFPS)       FTPD 420         EQUIVALENCE (COMINT (164), YFPS)       FTPD 440	INDFPC(6),	FTPD	220
3       URTDS (10), URCMS (5), URTMS (5), SETCDS(10), SETTDS(10), FTPD 240         4       SETCMS(5), SETTMS(5), INDCDS(10), INDTDS(10), INDCMS(10), FTPD 250         5       INDTMS(5), PRFCDS (10), PRFCDS (10), FRVMSC(5), FTPD 260         6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5), FTPD 270         L       FRVMST(5), IPOTDS(10), IPOTMS(5)         COMMON COMINT(400)       FTPD 290         EQUIVALENCE (COMINT (4), XSD )       FTPD 310         EQUIVALENCE (COMINT (4), XSD )       FTPD 320         EQUIVALENCE (COMINT (24), ZS )       FTPD 340         EQUIVALENCE (COMINT (24), ZS )       FTPD 350         EQUIVALENCE (COMINT (44), WX )       FTPD 370         EQUIVALENCE (COMINT (60), WY )       FTPD 370         EQUIVALENCE (COMINT (76), WZ )       FTPD 370         EQUIVALENCE (COMINT (164), GCC )       FTPD 400         EQUIVALENCE (COMINT (164), SCD )       FTPD 420         EQUIVALENCE (COMINT (164), XFPSD )       FTPD 420         EQUIVALENCE (COMINT (164), XFPSD )       FTPD 420         EQUIVALENCE (COMINT (244), YFPSD )       FTPD 420         EQUIVALENCE (COMINT (244), YF	STRPDS(10), STRPMS( 5), IPOCDS(10), IPOCMS( 5), URCDS (10),	FTPD	230
4       SETCMS(5), SETTMS(5), INDCDS(10), INDCMS(10), FTPD 250         5       INDTMS(5), PRFCDS (10), PRFCDS (10), PRFCMS (5), PRFTMS (5), FTPD 270         C       IPRDS (10), IPRMS (5), FRVDS(10), FRVDST(10), FRVMSC(5), FTPD 270         C       FRVMST(5), IPOTDS(10), IPOTMS(5)         COMMON COMINT(400)       FTPD 280         EQUIVALENCE (COMINT(4), SSD )       FTPD 300         EQUIVALENCE (COMINT(12), YS )       FTPD 320         EQUIVALENCE (COMINT(24), ZS )       FTPD 340         EQUIVALENCE (COMINT (24), ZS )       FTPD 370         EQUIVALENCE (COMINT (60), WY )       FTPD 370         EQUIVALENCE (COMINT (60), WY )       FTPD 370         EQUIVALENCE (COMINT (60), WY )       FTPD 370         EQUIVALENCE (COMINT (76), WZ )       FTPD 370         EQUIVALENCE (COMINT (76), WZ )       FTPD 370         EQUIVALENCE (COMINT (76), WZ )       FTPD 370         EQUIVALENCE (COMINT (164), SCD )       FTPD 400         EQUIVALENCE (COMINT (164), SCD )       FTPD 410         EQUIVALENCE (COMINT (184), SFPS )       FTPD 420         EQUIVALENCE (COMINT (184), XFPS )       FTPD 440         EQUIVALENCE (COMINT (244), YFPS )       FTPD 450         EQUIVALENCE (COMINT (244), YFPS )       FTPD 450         EQUIVALENCE (COMINT (244), YFPSD )       FTPD 450	URTDS (10), URCMS ( 5), URTMS ( 5), SETCDS(10), SETTDS(10),	FTPD	240
5       INDTMS( 5), PRFCDS (10), PRFTDS (10), FRVDST(10), FRVMSC( 5), FTPD 260         6       IPRDS (10), IPRMS ( 5), FRVDSC(10), FRVDST(10), FRVMSC( 5), FTPD 270         L       FRVMST( 5), IPOTDS(10), IPOTMS( 5)         COMMON COMINT(400)       FTPD 280         EQUIVALENCE ( COMINT( 3), IBOTM )       FTPD 300         EQUIVALENCE ( COMINT( 12 ), XSD )       FTPD 310         EQUIVALENCE ( COMINT( 12 ), YSD )       FTPD 340         EQUIVALENCE ( COMINT( 24 ), ZSD )       FTPD 350         EQUIVALENCE ( COMINT( 24 ), ZSD )       FTPD 350         EQUIVALENCE ( COMINT( 24 ), ZSD )       FTPD 360         EQUIVALENCE ( COMINT( 24 ), ZSD )       FTPD 370         EQUIVALENCE ( COMINT( 24 ), ZSD )       FTPD 370         EQUIVALENCE ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 390         EQUIVALENCE ( COMINT( 124 ), GCD )       FTPD 410         EQUIVALENCE ( COMINT( 124 ), GCD )       FTPD 420         EQUIVALENCE ( COMINT( 124 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 124 ), YFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 124 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 124 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 244 ), ZF	SETCMS( 5), SETTMS( 5), INDCDS(10), INDTDS(10), INDCMS(10),	FTPD	250
6       IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5), FTPD 270         L       FRVMST(5), IPOTDS(10), IPOTMS(5)       FTPD 280         COMMON COMINT(400)       FTPD 290         EQUIVALENCE ( COMINT(3), IBOTM )       FTPD 300         EQUIVALENCE ( COMINT(4), XSD )       FTPD 310         EQUIVALENCE ( COMINT(12), YS )       FTPD 330         EQUIVALENCE ( COMINT(24), ZS )       FTPD 340         EQUIVALENCE ( COMINT(28), ZSD )       FTPD 360         EQUIVALENCE ( COMINT (60), WY )       FTPD 370         EQUIVALENCE ( COMINT (60), WY )       FTPD 370         EQUIVALENCE ( COMINT (76), WZ )       FTPD 390         EQUIVALENCE ( COMINT (104), GCD )       FTPD 400         EQUIVALENCE ( COMINT (144), XFPS )       FTPD 410         EQUIVALENCE ( COMINT (164), SCDD )       FTPD 430         EQUIVALENCE ( COMINT (164), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT (164), XFPSD )       FTPD 450         EQUIVALENCE ( COMINT (244), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT (244), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT (244), YFPSD )       FTPD 47	INDTMS( 5), PRFCDS (10), PRFTDS (10), PRFCMS ( 5), PRFTMS ( 5)	, FTPD	260
L       FRVMST(5), IPOIDS(10), IPOTMS(5)       FTPD 280         COMMON COMINT(400)       FTPD 300         EQUIVALENCE ( COMINT( 4 ), XSD )       FTPD 310         EQUIVALENCE ( COMINT( 12 ), YS )       FTPD 320         EQUIVALENCE ( COMINT( 16 ), YSD )       FTPD 330         EQUIVALENCE ( COMINT( 16 ), YSD )       FTPD 340         EQUIVALENCE ( COMINT( 28 ), ZSD )       FTPD 360         EQUIVALENCE ( COMINT( 28 ), ZSD )       FTPD 370         EQUIVALENCE ( COMINT( 64 ), WX )       FTPD 370         EQUIVALENCE ( COMINT( 64 ), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 164 ), GCD )       FTPD 410         EQUIVALENCE ( COMINT( 124 ), GCD )       FTPD 420         EQUIVALENCE ( COMINT( 164 ), XFPS )       FTPD 430         EQUIVALENCE ( COMINT( 124 ), GCD )       FTPD 430         EQUIVALENCE ( COMINT( 164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 124 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 284 ), ZFPSD )       FTPD 470         EQUIVALENCE ( COMINT( 284 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT( 284 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT( 364 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT( 364 ), ZFPSD )       FTPD 500	IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5),	FTPD	270
COMMON COMINT(400)       FIPD 290         EQUIVALENCE ( COMINT( 1 ), XSD )       FTPD 300         EQUIVALENCE ( COMINT( 12 ), YS )       FTPD 320         EQUIVALENCE ( COMINT( 16 ), YSD )       FTPD 320         EQUIVALENCE ( COMINT( 16 ), YSD )       FTPD 340         EQUIVALENCE ( COMINT( 28 ), ZSD )       FTPD 350         EQUIVALENCE ( COMINT( 44 ), WX )       FTPD 360         EQUIVALENCE ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE ( COMINT( 60 ), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 400         EQUIVALENCE ( COMINT( 104 ), GCD )       FTPD 400         EQUIVALENCE ( COMINT( 124 ), GCDD )       FTPD 410         EQUIVALENCE ( COMINT( 124 ), SCPSD )       FTPD 420         EQUIVALENCE ( COMINT( 184 ), XFPSD )       FTPD 430         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 450 <t< td=""><td>FRVMST(5), IPOIDS(10), IPOTMS(5)</td><td>FTPD</td><td>280</td></t<>	FRVMST(5), IPOIDS(10), IPOTMS(5)	FTPD	280
EQUIVALENCE ( COMINT( 3 ), IBOTM )       FTPD 300         EQUIVALENCE ( COMINT( 12 ), YS )       FTPD 310         EQUIVALENCE ( COMINT( 12 ), YS )       FTPD 320         EQUIVALENCE ( COMINT( 16 ), YSD )       FTPD 330         EQUIVALENCE ( COMINT( 24 ), ZS )       FTPD 340         EQUIVALENCE ( COMINT( 28 ), ZSD )       FTPD 370         EQUIVALENCE ( COMINT( 28 ), ZSD )       FTPD 370         EQUIVALENCE ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 380         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 390         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 400         EQUIVALENCE ( COMINT( 164 ), GCD )       FTPD 410         EQUIVALENCE ( COMINT( 144 ), SCPS )       FTPD 420         EQUIVALENCE ( COMINT( 164 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT( 164 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT( 184 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 500 <td>COMMON COMINT(400)</td> <td>FTPD</td> <td>290</td>	COMMON COMINT(400)	FTPD	290
EQUIVALENCE ( COMINT( 4 ), XSD )       FTPD 310         EQUIVALENCE ( COMINT( 12 ), YS )       FTPD 320         EQUIVALENCE ( COMINT( 16 ), YSD )       FTPD 330         EQUIVALENCE ( COMINT( 28 ), ZSD )       FTPD 340         EQUIVALENCE ( COMINT( 28 ), ZSD )       FTPD 360         EQUIVALENCE ( COMINT( 28 ), ZSD )       FTPD 370         EQUIVALENCE ( COMINT( 44 ), WX )       FTPD 370         EQUIVALENCE ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 164 ), GCD )       FTPD 400         EQUIVALENCE ( COMINT(104 ), GCD )       FTPD 410         EQUIVALENCE ( COMINT(164 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT(164 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT(164 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT(184 ), XFPSD )       FTPD 420         EQUIVALENCE ( COMINT(24 ), YFPS )       FTPD 430         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 470         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 450         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 510 <td>EQUIVALENCE ( COMINIC 3 ), IBOTM )</td> <td>FTPD</td> <td>300</td>	EQUIVALENCE ( COMINIC 3 ), IBOTM )	FTPD	300
EQUIVALENCE ( COMINT( 12 ), YS       FTPD 320         EQUIVALENCE ( COMINT( 12 ), YS       FTPD 330         EQUIVALENCE ( COMINT( 24 ), ZS       FTPD 340         EQUIVALENCE ( COMINT( 28 ), ZSD )       FTPD 350         EQUIVALENCE ( COMINT( 44 ), WX )       FTPD 370         EQUIVALENCE ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE ( COMINT( 60 ), WZ )       FTPD 380         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 380         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 380         EQUIVALENCE ( COMINT( 104 ), GCD )       FTPD 400         EQUIVALENCE ( COMINT( 104 ), GCD )       FTPD 410         EQUIVALENCE ( COMINT( 144 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT( 164 ), XFPSD )       FTPD 430         EQUIVALENCE ( COMINT( 164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 184 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT( 244 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT( 244 ), ZFPSD )       FTPD 510         EQUIVALENCE ( COMINT( 343 ), IFIN )       FTPD 510         EQUIVALENCE ( COMINT( 343 ), IFIN )       FTPD 530	EQUIVALENCE ( COMINI( 4 ), XSD )		310
EQUIVALENCE       ( COMINT( 16 ), TSD )       FTPD 350         EQUIVALENCE       ( COMINT( 28 ), ZSD )       FTPD 350         EQUIVALENCE       ( COMINT( 28 ), ZSD )       FTPD 360         EQUIVALENCE       ( COMINT( 44 ), WX )       FTPD 370         EQUIVALENCE       ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE       ( COMINT( 76 ), WZ )       FTPD 370         EQUIVALENCE       ( COMINT( 164 ), GCD )       FTPD 400         EQUIVALENCE       ( COMINT(104 ), GCD )       FTPD 410         EQUIVALENCE       ( COMINT(124 ), GCD )       FTPD 420         EQUIVALENCE       ( COMINT(124 ), GCD )       FTPD 420         EQUIVALENCE       ( COMINT(164 ), XFPS )       FTPD 420         EQUIVALENCE       ( COMINT(164 ), XFPSD )       FTPD 420         EQUIVALENCE       ( COMINT(204 ), YFPS )       FTPD 440         EQUIVALENCE       ( COMINT(224 ), YFPSD )       FTPD 450         EQUIVALENCE       ( COMINT(224 ), YFPSD )       FTPD 470         EQUIVALENCE       ( COMINT(264 ), ZFPSD )       FTPD 480         EQUIVALENCE       ( COMINT(284 ), ZFPSD )       FTPD 490         EQUIVALENCE       ( COMINT(364 ), ZFPSD )       FTPD 500         EQUIVALENCE       ( COMINT(364 ), ZFPSD )       FTPD 510 <td>EQUIVALENCE ( COMINIC 12 ), 15 )</td> <td>F I P D</td> <td>320</td>	EQUIVALENCE ( COMINIC 12 ), 15 )	F I P D	320
EQUIVALENCE       (COMINT(24), ZSD)       FTPD 350         EQUIVALENCE       (COMINT(44), WX)       FTPD 360         EQUIVALENCE       (COMINT(60), WY)       FTPD 370         EQUIVALENCE       (COMINT(76), WZ)       FTPD 390         EQUIVALENCE       (COMINT(164), GCD)       FTPD 400         EQUIVALENCE       (COMINT(124), GCDD)       FTPD 420         EQUIVALENCE       (COMINT(144), XFPS)       FTPD 420         EQUIVALENCE       (COMINT(164), XFPSD)       FTPD 420         EQUIVALENCE       (COMINT(164), XFPSD)       FTPD 420         EQUIVALENCE       (COMINT(184), XFPSD)       FTPD 440         EQUIVALENCE       (COMINT(204), YFPS)       FTPD 440         EQUIVALENCE       (COMINT(244), YFPSD)       FTPD 450         EQUIVALENCE       (COMINT(284), ZFPSD)       FTPD 470         EQUIVALENCE       (COMINT(284), ZFPSD)       FTPD 470         EQUIVALENCE       (COMINT(284), ZFPSD)       FTPD 500         EQUIVALENCE       (COMINT(304), ZFPSD)       FTPD 510         EQUIVALENCE       (COMIN	EQUIVALENCE ( COMINIC 16 ), ISD )	FTPD	340
EQUIVALENCE ( COMINT( 24 ), WX )       FTPD 360         EQUIVALENCE ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE ( COMINT( 60 ), WZ )       FTPD 370         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 380         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 390         EQUIVALENCE ( COMINT( 104 ), GCD )       FTPD 400         EQUIVALENCE ( COMINT(124 ), GCD )       FTPD 410         EQUIVALENCE ( COMINT(124 ), GCD )       FTPD 420         EQUIVALENCE ( COMINT(164 ), XFPS )       FTPD 430         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(184 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 450         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPS )       FTPD 470         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(304 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(324), TIME )       FTPD 510         EQUIVALENCE ( COMINT(364 ), ZFPSD )       FTPD 520         EQUIVALENCE ( COMINT(364 ), IFIN )       FTPD 520         EQUIVALENCE ( COMINT(364 ), IFIN )       FTPD 530         EQUIVALENCE ( COMINT(364 ), XS )       FTPD 540         EQUIVALENCE ( COMINT(364 ), XS )       FTPD 550	FOLIVALENCE (COMINIC 24 ), ZSD )	FTPD	350
EQUIVALENCE ( COMINT( 60 ), WY )       FTPD 370         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 380         EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 390         EQUIVALENCE ( COMINT( 84 ), GC )       FTPD 390         EQUIVALENCE ( COMINT( 104 ), GCD )       FTPD 400         EQUIVALENCE ( COMINT( 124 ), GCDD )       FTPD 410         EQUIVALENCE ( COMINT( 144 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT( 164 ), XFPSD )       FTPD 430         EQUIVALENCE ( COMINT( 164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT( 204 ), YFPS )       FTPD 440         EQUIVALENCE ( COMINT( 224 ), YFPSD )       FTPD 460         EQUIVALENCE ( COMINT( 244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT( 284 ), ZFPSD )       FTPD 470         EQUIVALENCE ( COMINT( 284 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT( 304 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT( 304 ), ZFPSD )       FTPD 510         EQUIVALENCE ( COMINT( 364 ), ZFPSD )       FTPD 510         EQUIVALENCE ( COMINT( 364 ), IFIN )       FTPD 520         EQUIVALENCE ( COMINT( 364 ), IFIN )       FTPD 520         EQUIVALENCE ( COMINT( 364 ), SS )       FTPD 540         COMINT( 364 - 367) USED BY XS       F	FOULVALENCE (COMINT ( 44 ) + WX )	FTPD	360
EQUIVALENCE ( COMINT( 76 ), WZ )       FTPD 380         EQUIVALENCE ( COMINT( 84 ), GC )       FTPD 390         EQUIVALENCE ( COMINT(104 ), GCD )       FTPD 400         EQUIVALENCE ( COMINT(124 ), GCDD )       FTPD 410         EQUIVALENCE ( COMINT(124 ), GCDD )       FTPD 420         EQUIVALENCE ( COMINT(144 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 430         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(184 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPSD )       FTPD 450         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 440         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 480         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT(324 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(324 ), ZFPSD )       FTPD 510         EQUIVALENCE ( COMINT(343 ), JFIN )       FTPD 520         EQUIVALENCE ( COMINT(361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT(364 ), XS )       FTPD 540         COMINT(364-367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT( 50 ) WY )	FTPD	370
EQUIVALENCE ( COMINT( 84 ), GC )       FTPD 390         EQUIVALENCE ( COMINT(104 ), GCD )       FTPD 400         EQUIVALENCE ( COMINT(124 ), GCDD )       FTPD 410         EQUIVALENCE ( COMINT(124 ), SCDD )       FTPD 420         EQUIVALENCE ( COMINT(144 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 430         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(184 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 450         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 440         EQUIVALENCE ( COMINT(264 ), ZFPS )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT(324 ), ZFPSD )       FTPD 510         EQUIVALENCE ( COMINT(324 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(324 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(364 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(364 ), ZFPSD )       FTPD 510         EQUIVALENCE ( COMINT(364 ), ZFPSD )       FTPD 520         EQUIVALENCE ( COMINT(364 ), ZFPSD )       FTPD 520         EQUIVALENCE ( COMINT(361 ), IPTCNT ) <td< td=""><td>EQUIVALENCE ( COMINT( 76 ), WZ )</td><td>FTPD</td><td>380</td></td<>	EQUIVALENCE ( COMINT( 76 ), WZ )	FTPD	380
EQUIVALENCE ( COMINT(104 ), GCD )       FTPD 400         EQUIVALENCE ( COMINT(124 ), GCDD )       FTPD 410         EQUIVALENCE ( COMINT(144 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 430         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 450         EQUIVALENCE ( COMINT(224 ), YFPSD )       FTPD 460         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPS )       FTPD 480         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT 324), TIME )       FTPD 510         EQUIVALENCE ( COMINT 343 ), JFIN )       FTPD 520         EQUIVALENCE ( COMINT 361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT 364 ), XS )       FTPD 540         COMINT (364 - 367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT( 84 ), GC )	FTPD	390
EQUIVALENCE ( COMINT(124 ), GCDD )       FTPD 410         EQUIVALENCE ( COMINT(144 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 430         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(184 ), XFPSDD )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 450         EQUIVALENCE ( COMINT(224 ), YFPSD )       FTPD 460         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPS )       FTPD 480         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT 324), TIME )       FTPD 510         EQUIVALENCE ( COMINT 343 ), JFIN )       FTPD 520         EQUIVALENCE ( COMINT 361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT 364 ), XS )       FTPD 540         COMINT (364 - 367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT(104 ), GCD )	FTPD	400
EQUIVALENCE ( COMINT(144 ), XFPS )       FTPD 420         EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 430         EQUIVALENCE ( COMINT(184 ), XFPSD )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 450         EQUIVALENCE ( COMINT(224 ), YFPSD )       FTPD 460         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 480         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(304 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT 324), TIME )       FTPD 510         EQUIVALENCE ( COMINT 343 ), IFIN )       FTPD 520         EQUIVALENCE ( COMINT 361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT 364 ), XS )       FTPD 540         COMINT (364 - 367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT(124 ), GCDD )	FTPD	410
EQUIVALENCE ( COMINT(164 ), XFPSD )       FTPD 430         EQUIVALENCE ( COMINT(184 ), XFPSDD )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 450         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 450         EQUIVALENCE ( COMINT(224 ), YFPSD )       FTPD 460         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPS )       FTPD 480         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT(324), TIME )       FTPD 510         EQUIVALENCE ( COMINT(324), JFIN )       FTPD 520         EQUIVALENCE ( COMINT(361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT(361 ), IPTCNT )       FTPD 540         COMINT(364 - 367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT(144 ), XFPS )	FTPD	420
EQUIVALENCE ( COMINT(184 ), XFPSDD )       FTPD 440         EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 450         EQUIVALENCE ( COMINT(224 ), YFPSD )       FTPD 460         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPS )       FTPD 480         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 480         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT(324), TIME )       FTPD 510         EQUIVALENCE ( COMINT(324), TIME )       FTPD 520         EQUIVALENCE ( COMINT(361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT(364 ), XS )       FTPD 540         COMINT(364-367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT(164 ), XFPSD )	FTPD	430
EQUIVALENCE ( COMINT(204 ), YFPS )       FTPD 450         EQUIVALENCE ( COMINT(224 ), YFPSD )       FTPD 460         EQUIVALENCE ( COMINT(244 ), YFPSD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 480         EQUIVALENCE ( COMINT(264 ), ZFPS )       FTPD 490         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT(324), TIME )       FTPD 510         EQUIVALENCE ( COMINT( 324), TIME )       FTPD 520         EQUIVALENCE ( COMINT( 361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT( 364 ), XS )       FTPD 540         COMINT(364-367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT(184 ), XFPSDD )	FTPD	440
EQUIVALENCE ( COMINT(224 ), YFPSD )       FTPD 460         EQUIVALENCE ( COMINT(244 ), YFPSDD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 480         EQUIVALENCE ( COMINT(264 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 500         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 510         EQUIVALENCE ( COMINT(324), TIME )       FTPD 510         EQUIVALENCE ( COMINT( 324), TIME )       FTPD 520         EQUIVALENCE ( COMINT( 361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT( 364 ), XS )       FTPD 540         COMINT(364-367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT(204 ), YFPS )	FTPD	450
EQUIVALENCE ( COMINT(244 ), YFPSDD )       FTPD 470         EQUIVALENCE ( COMINT(264 ), ZFPS )       FTPD 480         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 510         EQUIVALENCE ( COMINT(324), TIME )       FTPD 510         EQUIVALENCE ( COMINT(343 ), JFIN )       FTPD 520         EQUIVALENCE ( COMINT(361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT(364 ), XS )       FTPD 540         COMINT(364-367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT(224 ), YFPSD )	FTPD	460
EQUIVALENCE ( COMINT(264 ), ZFPS )       FTPD 480         EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT(324), TIME )       FTPD 510         EQUIVALENCE ( COMINT(324), TIME )       FTPD 520         EQUIVALENCE ( COMINT( 343 ), JFIN )       FTPD 520         EQUIVALENCE ( COMINT( 361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT( 364 ), XS )       FTPD 540         COMINT(364-367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT(244 ), YFPSDD )	FTPD	470
EQUIVALENCE ( COMINT(284 ), ZFPSD )       FTPD 490         EQUIVALENCE ( COMINT(304 ), ZFPSDD )       FTPD 500         EQUIVALENCE ( COMINT(324), TIME )       FTPD 510         EQUIVALENCE ( COMINT(343 ), JFIN )       FTPD 520         EQUIVALENCE ( COMINT(361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT(364 ), XS )       FTPD 540         COMINT(364-367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINT(264 ), ZFPS )	FTPD	480
EQUIVALENCE ( COMINI (304 ), ZEPSDD )       FTPD 500         EQUIVALENCE ( COMINT ( 324), TIME )       FTPD 510         EQUIVALENCE ( COMINT ( 343 ), IFIN )       FTPD 520         EQUIVALENCE ( COMINT ( 361 ), IPTCNT )       FTPD 530         EQUIVALENCE ( COMINT ( 364 ), XS )       FTPD 540         COMINT ( 364 - 367) USED BY XS       FTPD 550	EQUIVALENCE ( COMINI(284 ), ZEPSD )	FTPD	490
EQUIVALENCE ( COMINT( 3247) TIME )FTPD 510EQUIVALENCE ( COMINT( 343 ), IFIN )FTPD 520EQUIVALENCE ( COMINT( 361 ), IPTCNT )FTPD 530EQUIVALENCE ( COMINT( 364 ), XS )FTPD 540COMINT(364-367) USED BY XSFTPD 550	EQUIVALENCE ( COMINICOU ) CEPSOD )	FTPD	500
EQUIVALENCE ( COMINT( 361 ), IPTCNT )FTPD 530EQUIVALENCE ( COMINT( 364 ), XS )FTPD 540COMINT(364-367) USED BY XSFTPD 550	FOUTVALENCE ( COMINT( 343 ) IFIN )	FTPD	520
EQUIVALENCE ( COMINT( 364 ), XS ) COMINT(364-367) USED BY XS FTPD 550	FOULVALENCE ( COMINT( 361 ) IPTONT )	FTPD	520
COMINT (364–367) USED BY XS FTPD 550	FOULVALENCE ( COMINT( 364 ) XS )	FTPD	540
	COMINT(364-367) USED BY XS	FTPD	550

С

	EQUIVALENCE ( COMINT( 368 ), CURDT )	FTPD 560
	COMMON	FTPD 570
]	CBMASS, CBIXX , CBIXZ , CBIYY , CBIYZ , CBIZZ , FPMASS, CBIXY,	FTPD 580
2	2 DC(3,3) , XFP(5), YFP(5), ZFP(5), WNX(5), WNY(5),	FTPD 590
	3 WNZ(5), PX(5), PY(5), PZ(5), GM(5), OMEGA(5),	FTPD 600
2	GRAV , GRAVE , ZETA , FTS (6) , FSXSI(5) ,	FTPD 610
	5 FSYSI(5) , FSZSI(5) , SOILX(5) ,	FTPD 620
6	5 SOILY(5) , SOILZ(5) , PMSX(5,5) ,	FTPD 630
-	7 PMSY(5,5) , PMSZ(5,5) , PDSX(10,5) ,	FTPD 640
8	B PDSY(10,5) , PDSZ(10,5) , FLXS , FLYS , FLZS ,	FIPD 650
ç	FIXL, TLYL, TLZL, SLO, XMSCB(5)	FTPD 660
(	C YMSCB(5) , ZMSCB(5) , XDSCB(10) ,	FTPD 670
[	YDSCB(10) , ZDSCB(10) , ILEG , IMS ,	FTPD 680
ļ	E FSTX , FSTY , FSTZ , PVCBX , PVCBY , PVCBZ ,	FIPD 690
1	F PVFPX , PVFPY , PVFPZ , NOLEG , SLOMS(5) ,	FTPD 700
(	G SLODS(10)	FTPD 710
	COMMON	FIPD 720
	1 PFCDS(5) , PFCDS(5) , PFTDS(5) ,	FIPD 730
	2 PFIMS(5) , SRCDS(5) , SRCMS(5) ,	FIPD 740
	3 SRIDS(5) , SRIMS(5) , COEPDS, COEPMS, GAMDS,	FIPD 750
4	4 GAMMS, SRUCDS, SRUCMS, SRUTDS, SRUTMS, SCMXDS, SCMXMS,	FIPD 760
	5 STMXDS, SIMXMS, CDCDS(5) , CDCMS(5) ,	FIPD 770
	CD(DS(5)), $CD(MS(5))$ , $FR(COS)$ , $FR(CMS)$ , $RE(DS)$	FTPD 700
	7 IRE[MS, SIRURE(10) , SIRUBS(10) , SIRUMS(2) ,	FTPD 790
i	B LENGTH CDXS CDYS CDZS , NTYPE , RAD(4), SS(4) , ATTHCK(3),	FIPD 800
	9 ATTPRS(3) , ADIST, SUILP(3) , NMODES,	FTPD 810
	A COEF, GAMMA, FRIC, SCMAX, SIMAX, COC(5), CD1(5),	FIPD 820
	$B \qquad SRULT , SRULC , SRT(5), SRC(5), PFI(5), PFC(5), GFLEGS(5),$	FIPD 830
1	C CURMSV, CURMSL, INDEXD, INDEXD, INDEXD, INDEXR, INDEXR,	FIPD 840
i	D INDEZR, TIMAX, DRAGSI, IEP	FIPD 850
	COMMON	FIPD 860
		FIPD 870
	2MODEIN, POULA(10,5), POULT(10,5), POULZ(10,5), PCGA(5),	FTPD 800
	3PCGZ(5),AEII,AEIZ,FURMS(5),FURUS(10),FURCE,NFURC(5AVMSX(5),	FTPD 890
		FIPD 900
	GOUNCEL STABSTABVL STABSJCKSADSVELASVELTSVELZSAVMST()	FTPD 910
		FTPD 920
	I SMAMSC(3), IMAMSC(3), SMAMSI(3), IMAMSI(3),	FIFD 950
	2 SMXDSC(10), IMXDSC(10), SMXDS(10), IMXDS(10) SMXDSC(5), SMXDS(10), SMXDS(10), IMXDS(10)	FTPD 940
	5 (SLINGMS(5)) (SLINGDS(10)) COUNDS() INLEG () IFFRI) MDACT(5) IDDTED(5), KOUNT(5) ANGY ANGY, ANG7	ETPD 060
	4 IMPACI(5) SIEKTER(5), KOUNI(5) SANGKS ANGES ANGE	FTPD 900
		ETDD 980
	SUBROUTINE INTITALIZATION	ETDD 900
	LEVINODES SO DISO TO 11	FTPD1000
		FTPD1020
10		ETDD1020
11		
ττ		ETDD1050
	* * * * *	FTPD1020
	DETERMINE LANDING GEAR STRUT LOADS	ETDD1070
	A X X X X	
		1 1 F D 1 0 0 0
	* * * * *	F1PD1090
		ETEDIIO

.

C C C

С		¥	*	¥	*	×								FTPD1120
С														FTPD1130
		IMS	5=1											FTPD1140
С														FTPD1150
С		CEN	ITE	R B	ODY	STRUT	ATT	ACH PC	OINT	INL	C.S.			FTPD1160
С														FTPD1170
		ХT	= XM	SCE	(IF	P)								FTPD1180
		YT:	= YM	SCB	(IF	P)								FTPD1190
		ZT:	=ZM	SCE	(IF	P)								FTPD1200
		IF	(NM	ODE	S.E	Q.0)GO	то	101						FTPD1210
		DO	10	οI	=1,	NMODES								FTPD1220
		ХT	=XT	+PM	ISX (	IFP,I)	*GC(	1,I)						FTPD1230
		Y۲	=YT	+PM	ISY (	IFP,I)	*GC(	1,I)						FTPD1240
	100	ZT	=ZT	+P№	ISZ (	IFP,I)	*GC(	1,I)						FTPD1250
	101	COI	ITN	NUE	-									FTPD1260
		XTI	4S=	хт										FTPD1270
		YTI	∕IS=	ΥT										FTPD1280
		ZTI	∙ns=	ΖT										FTPD1290
С														FTPD1300
С		CEI	NTE	RE	ODY	STRUT	ATT	ACH PO	OINT	IN S	5.C.S.			FTPD1310
С														FTPD1320
		PV	свх	=XS	5+DC	(1,1)*	XT+D	C(1,2	)*YT+	HDC (	1,3)*Z	т		FTPD1330
		P۷	СВҮ	=YS	5+DC	(2,1)*	XT+D	C(2,2)	)*YT+	FDC ( )	2,3)*Z	Ť		FTPD1340
		PV	CBZ	=ZS	S+DC	(3,1)*	XT+D	C(3,2)	)*YT+	FDC ( )	3,3)*Z	т		FTPD1350
С														FTPD1360
С		FO	OTP	AD	END	STRUT	ATT	ACH PO	OINT	IN :	S.C.S.			FTPD1370
С														FTPD1380
		P۷I	FPX	=	XFP	S(1,IF	P)							FTPD1390
		PVI	FPY	=	YFP	S(1,IF	P)							FTPD1400
		PVI	FPZ	=	ZFP	S(1,IF	P)							FTPD1410
С														FTPD1420
С		SE	τυ	Ρl	OAD	-STROK	E CU	JRVE						FTPD1430
С														FTPD1440
		DO	11	5 1	[=1,	5								FTPD1450
		ΡF	C(I	) = F	PFCM	S(I)								FTPD1460
		PF	T(I	) = F	PFTM	S(1)								FTPD1470
		SR	C(I	)=5	SRCM	S(I)								FTPD1480
		SR	T ( ŀ	)=5	SRTM	S(1)								FTPD1490
		CD		)=(	DCM	S(I)								FTPD1500
		CD	T ( 1	}=(	DTM	S(1)								FTPD1510
	115	CO	NTI	NUE										FIPD1520
		C0	EF=	COE	EFMS									FTPD1530
		GA	MMA	= 64	AMMS								,	F1PD1540
		SR		= SF		S								FIPD1550
		SK		= 51		5								FTPD1560
		SC	MAX	= 50	_141 X 141 EM X M	5								FIPD1570
		51		-01	CMC	3								F 1PU1500
		- F K	 	1.01	CMD AC ( T	501								
c		36	0-3	LOP	1311	1								FTPD1600
ĉ		<u>د</u> ۲	<b>т</b> , ,	<b>.</b> .		CATOR								
č		SE	ιU	۳.	INDI	CATURS	J							FTDD1420
Ç		ст	00-		стр	DMCIT	D١							FTPD1640
				=		MS(TEE	ις / Σ							FTPD1650
		0 7 0		-	HDT	MCITE	, ,							FTDD1440
		יים	STC	-	SET	CMS(IF	P)							FTPD1670
		~ 1	$\sim$	-	<u> </u>									

	DISTT= SETTMS(IFP)	FTPD1680
	PFORC= PRFCMS(IFP)	FIPD1690
	PFORT= PRFTMS(IFP)	F 1PD1700
	FOREVC= FRVMSC(IFP) FOREVT- FRVMSC(IFP)	ETPD1720
		FTPD1730
	IPOSC= IPOCMS(IFP)	FTPD1740
	INDULC= INDCMS(IFP)	FTPD1750
	INDULT= INDTMS(IFP)	FTPD1760
	IPREV= IPRMS(IFP)	FTPD1770
	IRET=IRETMS	FTPD1780
С		FTPD1790
	CALL STRUT(STRP, SPNGC, SPNGT, DISTC, DISTT, PFORC, PFORT, F	FOREVC, FTPD1800
	* FOREVT, IPOSC, IPOST, INDULC, INDULT, IPREV, IRE	
Ç		F (PD1020
C	SAVE INDICATORS	ETPD1850
C		FTPD1850
		FTPD1860
	FORMS(IFP)=FORCE	FTPD1870
	$IE(TIME \cdot EQ \cdot 0 \cdot 0) = GO TO 117$	FTPD1880
	IF(IFIN•NE•0) GO TO 117	FTPD1890
	STRPMS(IFP)=STROKE(IFP)	FTPD1900
	SLNGMS(IFP)≈CURMSL	FTPD1910
	URCMS(IFP) = SPNGC	F1PD1920
	URTMS(IFP) =SPNGT	F1PD1930
	SETCMS(IFP) =DISTC	FTPD1940
	SETIMS(IFP) =DIST DECMC(IFP) =DECDC	ETPD1960
		FTPD1970
	EDVMSC(JED) -FOREVC	FTPD1980
	FRVMSC(IFP) =FOREVT	FTPD1990
	IPO(MS(IFP) = IPOSC	FTPD2000
	IPOTMS(IFP) = IPOST	FTPD2010
	INDCMS(IFP) =INDULC	FTPD2020
	INDTMS(IFP) = INDULT	FTPD2030
	IPRMS(IFP) = IPREV	FIPD2040
Ç	SAVE MAXIMUM STROKE	FTPD2050
	IF(STROKE(IFP).GT.SMXMSC(IFP))GU TU IIB	F1FD2000
	SMXMSC(IFP)=STRORE(IFP)	ETPD2080
		FTPD2090
	JUSTENSTERNELLEDI LT. SMYMSTITEPINGO TO 117	FTPD2100
	SMXMST(IEP)=STROKE(IEP)	FTPD2110
	TMXMST(IFP)=TIME	FTPD2120
	117 CONTINUE	FTPD2130
С		FTPD2140
С	C SAVE FORCES AND TORQUES	FTPD2150
С	C	FTPD2160
	FX=-DC(1,1)*FSTX-DC(2,1)*FSTY-DC(3,1)*FSTZ	F1PD2170
	FY=-DC(1,2)*FSTX-DC(2,2)*FSTY-DC(3,2)*FSTZ	F1FU2100
	FZ = -DC(1,3)*FSIX-DC(2,3)*FSIY-DC(3,3)*FSIZ	F1702170
	IF (NFORCEQ.0) GU TU II6	FTP02210
		FTPD2220
	SAVMST(IFP)=FT	FTPD2230

С

с с с

FTPD2240 116 CONTINUE FSXSI(IFP)=FSTX FTPD2250 FTPD2260 FSYSI(IFP)=FSTY FTPD2270 FSZSI(IFP)=FSTZ FTPD2280 FLXS=-FSTX FTPD2290 FLYS=-FSTY FLZS=-FSTZ FTPD2300 FTPD2310 TLXL=YT*FZ-ZT*FY TLYL=ZT*FX-XT*FZ FTPD2320 TLZL=XT*FY-YT*FX FTPD2330 IF(NMODES.EQ.0)GO TO 121 FTPD2340 DO 120 I=1,NMODES FTPD2350 120 GFLEGS(I)=GFLEGS(I)+FX*PMSX(IFP,I)+FY*PMSY(IFP,I)+FZ*PMSZ(IFP,I) FTPD2360 FTPD2370 121 CONTINUE FTPD2380 С FTPD2390 C C * * * ¥ ¥ FTPD2400 DRAG STRUT FTPD2410 C * * * × ÷. FTPD2420 С FTPD2430 IMS=0 FTPD2440 DO 2500 III=1.2 FTPD2450 NNN=2*(IFP-1)+IIIFTPD2460 SRBEND=0. FTPD2470 C FTPD2480 С CENTER BODY STRUT ATTACH POINT IN L.C.S. FTPD2490 С FTPD2500 XT = XDSCB(NNN)FTPD2510 YT = YDSCB(NNN)FTPD2520 ZT = ZDSCB(NNN)FTPD2530 IF(NMODES.EQ.0)GO TO 2101 FTPD2540 DO 2100 I=1,NMODES FTPD2550 XT=XT+PDSX(NNN,I)*GC(1,I) YT=YT+PDSY(NNN,I)*GC(1,I) FTPD2560 FTPD2570 2100 ZT=ZT+PDSZ(NNN,I)*GC(1,I) FTPD2580 2101 CONTINUE FTPD2590 С FTPD2600 С CENTER BODY ATTACH POINT IN S.C.S. FTPD2610 С PVCBX=XS+DC(1,1)*XT+DC(1,2)*YT+DC(1,3)*ZT FTPD2620 FTPD2630 PVCBY=YS+DC(2,1)*XT+DC(2,2)*YT+DC(2,3)*ZT PVCBZ=ZS+DC(3,1)*XT+DC(3,2)*YT+DC(3,3)*ZT FTPD2640 FTPD2650 IF(ILEG.EQ.0)GO TO 2000 FTPD2660 С C C CANTILEVER GEAR FTPD2670 FTPD2680 FOOTPAD END STRUT ATTACH POINT IN S.C.S. FTPD2690 С FTPD2700 С FTPD2710 DEL=CURMSL-DRAGST FTPD2720 IF(DEL.LT.0.0) DEL=0.0 FTPD2730 PVFPX = XFPS(1, IFP)+CDXS*DEL FTPD2740 PVFPY = YFPS(1,IFP)+CDYS*DEL FTPD2750 PVFPZ = ZFPS(1, IFP)+CDZS*DEL FTPD2760 C C FTPD2770 MAIN STRUT BENDING FTPD2780 С IF(AEI1.LE.0.0.0R.AEI2.LE.0.0) GO TO 2000 FTPD2790

	DX=PVFPX-PVCBX	ETPD2800
	DY=PVFPY-PVCBY	ETPD2810
	D7 = PVFP7 - PVCB7	ETDD2820
	$S \leq I = S \cap R I (D \vee N \vee D \vee N \vee D \vee D \vee D \vee D \vee D \vee D \vee $	ETD02820
	SY=DY/SSI	FTPD2050
		FTPD2040
		TTFD2850
		F1PD2860
		FIPD2870
	SDUIN=1+U-SDUIM**2	F 1PD2950
	SNZ=SDUIN**2	FTPD2960
	DZ=DRAGS1**2	FTPD2970
	D3=DRAGST**3	FTPD2980
	AAA=(D3*DEL)/(3*CURMSL)*(1DRAGST/CURMSL)	FTPD2990
	BBB=D2*(D2/CURMSL-D3/(3•*CURMSL**2)+CURMSL/3•-DRAGST)	FTPD3000
	SRBEND=1•/(SN2*((AAA/AEI1)+(BBB/AEI2)))	FTPD3010
2000	CONTINUE	FTPD3020
C,		FTPD3030
C	SET UP LOAD-STROKE CURVE	FTPD3040
С		FTPD3050
	DO 2115 I=1,5	FTPD3060
	PFC(I)=PFCDS(I)	FTPD3070
	PFT(I)=PFTDS(I)	FTPD3080
	IF(ILEG.EQ.O) GO TO 2114	FTPD3090
	IF(AEI1.LE.0.0.0R.AEI2.LE.0.0) GO TO 2114	FTPD3100
	<pre>SRC(I)=(SRBEND*SRCDS(I))/(SRBEND+SRCDS(I))</pre>	FTPD3110
	<pre>SRT(I)=(SRBEND*SRTDS(I))/(SRBEND+SRTDS(I))</pre>	FTPD3120
	CDC(I)=CDCDS(I)+PFCDS(I)/SRBEND	FTPD3130
	CDT(I)=CDTDS(I)+PFTDS(I)/SRBEND	FTPD3140
	GO TO 2115	FTPD3150
2114	SRC(I) = SRCDS(I)	FTPD3160
	SRT(I) = SRTDS(I)	FTPD3170
	CDC(I) = CDCDS(I)	FTPD3180
	CDT(I) = CDTDS(I)	ETPD3190
2115	CONTINUE	ETPD3200
~117	COFF=COFFDS	FTPD3210
		ETDD220
		ETD02220
		F1FD3230
	STMAX=STMADS	FIPD3250
		F 1PD3260
		F 1PD3270
~	SLU=SLUDS(NNN)	F1PD3280
Ċ		FTPD3290
C	SET UP INDICATORS	FTPD3300
C		FTPD3310
	STRP= STRPDS(NNN)	FTPD3320
	SPNGC= URCDS(NNN)	FTPD3330
	SPNGI= URTDS(NNN)	FTPD3340
	DISTC= SETCDS(NNN)	FTPD3350
	DISH = SETTDS(NNN)	FTPD3360
	PFORC= PRFCDS(NNN)	FTPD3370
	PFORT= PRFTDS(NNN)	FTPD3380
	FOREVC= FRVDSC(NNN)	FTPD3390
	FOREVT = FRVDST(NNN)	FTPD3400
	IPOSC= IPOCDS(NNN)	FTPD3410
	IPOST= IPOTDS(NNN)	FTPD3420

a

ß

	INDULC= INDCDS(NNN) INDULT= INDTDS(NNN)	FTPD3430 FTPD3440
c	IPREV= IPRDS(NNN) IRET=IRETDS	F 1PD3450 FTPD3460 FTPD3470
c	CALL STRUT(STRP,SPNGC,SPNGT,DISTC,DISTT,PFORC,PFORT,FOREVC, * FOREVT,IPOSC,IPOST,INDULC,INDULT,IPREV,IRET,NNN)	FTPD3480 FTPD3490
C C	SAVE INDICATORS	FTPD3500 FTPD3510 FTPD3520
_	IF(IBOTM•NE•O)IFP=NNN*100 IF(IBOTM•NF•O) RETURN	FTPD3530
	STRKDS(NNN)=STROKE(NNN) FORDS(NNN)=FORCE	FTPD3550 FTPD3560
	IF(TIME.EQ.0.0) GO TO 2127 IF(IFIN.NE.0) GO TO 2127	FTPD3570 FTPD3580
	STRPDS(NNN)=STROKE(NNN) SLNGDS(NNN)=CURDSL	FTPD3590 FTPD3600
	URCDS(NNN) = SPNGC URTDS(NNN) = SPNGT SETCOS(NNN) = DISTC	FTPD3610 FTPD3620
	SETTDS(NNN) =DISTT PRECDS(NNN) =PFORC	FTPD3640 FTPD3640 FTPD3650
	PRFTDS(NNN) =PFORT FRVDSC(NNN) =FOREVC	FTPD3660 FTPD3670
	FRVDST(NNN) =FOREVT IPOCDS(NNN) =IPOSC	FTPD3680 FTPD3690
	IPOTDS(NNN) = IPOST INDCDS(NNN) = INDULC	FTPD3700 FTPD3710
	INDTDS(NNN) = INDULT IPRDS(NNN) = IPREV	FTPD3720 FTPD3730
С	SAVE MAXUMUM STROKE IF(STROKE(NNN).GT.SMXDSC(NNN))GO TO 2128	FTPD3740 FTPD3750
	SMXDSC(NNN)=SIROKE(NNN) TMXDSC(NNN)=TIME	FTPD3760 FTPD3770
2128	GO TO 2127 IF(STROKE(NNN) • LT • SMXDST(NNN))GO TO 2127 SMXDST(NNN) = STROKE(NNN)	FTPD3780 FTPD3790 FTPD3800
2127	TMXDST(NNN)=TIME CONTINUE	FTPD3810 FTPD3820
с	IF(ILEG.EQ.0)GO TO 2119	FTPD3830 FTPD3840
c c	SAVE FORCES AND TORQUES - CANTILEVER	FTPD3850 FTPD3860
	FX=-DC(1,1)*FSTX-DC(2,1)*FSTY-DC(3,1)*FSTZ FY=-DC(1,2)*FSTX-DC(2,2)*FSTY-DC(3,2)*FSTZ F7=-DC(1,2)*FSTX-DC(2,2)*FSTY-DC(3,2)*FSTZ	FTPD3870 FTPD3880
	FZ = DC(1) ST = ST = DC(2) ST = ST = DC(3) ST = ST = ST = ST = ST = ST = ST = ST	FTPD3890 FTPD3900
	SAVDSX(INNN)=FX SAVDSZ(INNN)=FZ	FTPD3910 FTPD3920 FTPD3930
2118	CONTINUE FLXS=FLXS-FSTX	FTPD3940 FTPD3950
	FLYS=FLYS-FSTY FLZS=FLZS-FSTZ TLXL=TLXL+YT*FZ-ZT*FY	FTPD3960 FTPD3970 FTPD3980

C C C

445

\$

	TLYL=TLYL+ZT*FX-XT*FZ	FTPD3990				
	TLZL=TLZL+XT*FY-YT*FX	FTPD4000				
	IF(NMODES.EQ.0)GO TO 2117	FTPD4010				
	DO 2116 I=1,NMODES	FTPD4020				
2116	GFLEGS(I)=GFLEGS(I)+FX*PDSX(NNN,I)+FY*PDSY(NNN,I)+FZ*PDSZ(NNN,I)	FTPD4030				
2117	CONTINUE	FTPD4040				
	FA=FSTX*CDXS+FSTY*CDYS+FSTZ*CDZS	FTPD4050				
	FAX=FA*CDXS	FTPD4060				
	FAY=FA*CDYS	FTPD4070				
	FAZ=FA*CDZS	FTPD4080				
	FLX=FSTX-FAX	F1PD4090				
	FLY=FSTY-FAY	FTPD4100				
	FLZ=FSTZ-FAZ	FTPD4110				
	RAT2=(CURMSL-DRAGST)/CURMSL	F1PD4120				
	IF(RAT2.LT.0.0)RAT2=0.0	F1PD4150				
	RATI=1-RAT2	FTPD4140				
	FSXSI(IFP)=FSXSI(IFP)+RATI*FLX	FTPD4150				
	FSYSI(IFP)=FSYSI(IFP)+RAT1*FLY	F1PD4100				
	FSZSI(IFP)=FSZSI(IFP)+RAT1*FLZ	FIPD4170				
	F XX = FAX + RA / Z + L X	FTPD4100				
		FTPD4190				
	FZZ=FAZ+RAT2*FLZ	F1PD4200				
	FX = DC(10,1) * FXX + DC(10,2) * FYY + DC(10,2) * FZZ	FTPD4210				
	F Z = DC(33,1) * F X X + DC(3,2) * F Y Y + DC(3,3,2) * F Z Z	FTFD4220				
	FY = DC(2) [] * FXX + DC(2) ] * FY + DC(2) ] * FZZ	FTPD4240				
		FTPD4250				
	SAVMSX(IFP)= SAVMSX(IFP)+FX	ETPD//260				
	SAVMSY(IFP) = SAVMSY(IFP) + FY	ETPD4200				
	SAVMSZ(IFP)=SAVMSZ(IFP)+FZ	FTPD4280				
2122		ETDD4200				
		FTPD4300				
		FTPD4310				
		ETPD4320				
		FTPD4330				
		ETPD4340				
		FTPD4350				
		FTPD4360				
2200	CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCII-CCIECCI	FTPD4370				
2200	General and the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second se	ETPD4380				
2201		ETPD4390				
c	00 10 2000	FTPD4400				
	SAVE FORCES AND TOROUSS - INVERTED TRIDOD	FTPD4410				
c	SAVE FORCES AND FORGES - INVERTED TRIFOD	FTPD4420				
2110	CONTINUE	FTPD4430				
2119		FTPD4440				
	$F_{A} = DC(1) + 1/4F_{A} + DC(2) + 1/4F_{A} + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + DC(3) + $	FTPD4450				
	$F_1 = DC(1, 2) + F_1 + DC(2, 2) + F_1 + DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + F_1 - DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3) + DC(3, 3$	FTPD4460				
	E = D C C (1) = D C (2)	FTPD4470				
	SAVDST (MMA) = F7	FTPD4490				
		FTPD4500				
21 22		FTPD4510				
2123		FTPD4520				
	F SASINIFF /-F SASINIF / THISTA F SVSI (IFD)=F SVSI (IFD)+F STY	FTPD4530				
	FC7C1(IFD)=FC7C1(IFD)+FCT7	FTPD4540				

FTPD4550 FLXS=FLXS-FSTX FTPD4560 FLYS=FLYS-FSTY FTPD4570 FLZS=FLZS-FSTZ FTPD4580 TLXL=TLXL+YT*FZ-ZT*FY FTPD4590 TLYL=TLYL+ZT*FX-XT*FZ TLZL=TLZL+XT*FY-YT*FX FTPD4600 FTPD4610 IF(NMODES.EQ.0)GO TO 2121 DO 2120 I=1,NMODES FTPD4620 2120 GFLEGS(I)=GFLEGS(I)+FX*PDSX(NNN,I)+FY*PDSY(NNN,I)+FZ*PDSZ(NNN,I) FTPD4630 FTPD4640 2121 CONTINUE FTPD4650 2500 CONTINUE С FTPD4660 FTPD4670 * * * С ¥ DETERMINE SOIL FORCES FTPD4680 С * * * * FTPD4690 С × 3000 CONTINUE FTPD4700 IF(XFPS(1,IFP)-ATTH(IFP).GT.0.0)GO TO 5000 FTPD4710 FTPD4720 ¢ FTPD4730 С FOOTPAD IMPACT PRINT CONTROL FTPD4740 С IF(IFPPRT.EQ.0)GO TO 4010 FTPD4750 FTPD4760 IF(IFIN.NE.O)GO TO 4010 IF(IPTCNT.EQ.1)GO TO 4010 FTPD4770 FTPD4780 IF(IMPACT(IFP).NE.0)GO TO 4000 FTPD4790 IMPACT(IFP)=1 FTPD4800 IPRTFP(IFP)=1 FTPD4810 4000 CONTINUE FTPD4820 IF(IMPACT(IFP).EQ.2)GO TO 4010 FTPD4830 KOUNT(IFP)=KOUNT(IFP)+1 IF (KOUNT (IFP) • LE • IFPPRT) GO TO 4010 FTPD4840 IPRTFP(IFP)=IPTCNT FTPD4850 FTPD4860 KOUNT(IFP)=0 FTPD4870 IMPACT(IFP)=2 4010 CONTINUE FTPD4880 FTPD4890 C FTPD4900 CALL SOIL FTPD4910 С CHECK MAGNITUDE OF IN PLANE SOIL FORCE FTPD4920 С FTPD4930 С FF=SQRT(SOILY(IFP)**2+SOILZ(IFP)**2) FTPD4940 IF(FF.LT.0.00001) GO TO 3200 FTPD4950 FTPD4960 CFFY=SOILY(IFP)/FF FTPD4970 CFFZ=SOILZ(IFP)/FF FH=SQRT(FSYSI(IFP)**2+FSZSI(IFP)**2) FTPD4980 IF(FH.GT.0.00001) GO TO 3102 ETPD4990 FTPD5000 HDOTF=0.0 FTPD5010 GO TO 3103 FTPD5020 3102 CFHY=FSYSI(IFP)/FH CFHZ=FSZSI(IFP)/FH FTPD5030 FTPD5040 HDOTF=CFHY*CFFY+CFHZ*CFFZ 3103 VS=SQRT(YFPSD(1,IFP)**2+ZFPSD(1,IFP)**2) FTPD5050 AMV=FPMASS*VS/CURDT FTPD5060 IF(FF.LE.(-FH*HDOTF+AMV)) GO TO 3200 FTPD5070 IF (HDOTF.LE.O.)FF=-FH*HDOTF+AMV FTPD5080 FTPD5090 IF (HDOTF .GT .O.) FF = AMV FTPD5100 IF(FF.LT.0.)FF=0.0

SOILY(IFP)=CFFY*FF SOILZ(IFP) =CFFZ*FF CONTINUE RETURN	FTPD5110 FTPD5120 FTPD5130 FTPD5140 FTPD5150		
* * * * *	FTPD5160		
FOOTPAD OFF GROUND	FTPD5170		
* * * * *	FTPD5180		
	FTPD5190		
CONTINUE	FTPD5200		
SOILX(IFP)=0.0	FTPD5210		
SOILY(IFP)=0.0	FTPD5220		
SOILZ(IFP)=0.0	FTPD5230		
	FTPD5240		
FOOTPAD IMPACT PRINT CONTROL	FTPD5250		
	FTPD5260		
IF(IFPPRT-EQ-0)GO TO 5020	FTPD5270		
IF(IFIN•NE•O)GO TO 5020	FTPD5280		
IF(IMPACT(IFP).NE.1)GO TO 5010	FTPD5290		
KOUNT(IFP)=KOUNT(IFP)+1	F1PD5300		
IF(KOUNT(IFP).LE.IFPPRT)GO TO 5020	FTPD5310		
IPRTFP(IFP)=IPTCNT	FIPD5320		
KOUNT(IFP)=0	FTPD5330		
IMPACT(IFP)=0	F1PD5340		
CONTINUE	FTPD5350		
RETURN	FTPD5360		
END	FTPD5370		
	SOILY(IFP)=CFFY*FF SOILZ(IFP) =CFFZ*FF CONTINUE RETURN * * * * * FOOTPAD OFF GROUND * * * * * * CONTINUE SOILX(IFP)=0.0 SOILY(IFP)=0.0 FOOTPAD IMPACT PRINT CONTROL IF(IFPPRT.EQ.0)GO TO 5020 IF(IFPPRT.EQ.0)GO TO 5020 IF(IMPACT(IFP).NE.1)GO TO 5010 KOUNT(IFP)=KOUNT(IFP)+1 IF(KOUNT(IFP)+LE.IFPPRT)GO TO 5020 IPRTFP(IFP)=IPTCNT KOUNT(IFP)=0 IMPACT(IFP)=0 CONTINUE RETURN END		

CC	SUBROUTINE GEOM THIS SUBROUITNE DETERMINES THE DIRECTION COSINE MATRIX AND DETERMINES THE TIME DERIVATES OF THE EULER ANGLES COMMON COMINT(400) EQUIVALENCE ( COMINT( 36 ), PHI ) EQUIVALENCE ( COMINT( 40 ), PHID ) EQUIVALENCE ( COMINT( 44 ), WX ) EQUIVALENCE ( COMINT( 52 ), THTA ) EQUIVALENCE ( COMINT( 56 ), THTAD ) EQUIVALENCE ( COMINT( 56 ), THTAD ) EQUIVALENCE ( COMINT( 60 ), WY ) EQUIVALENCE ( COMINT( 68 ), PSI ) EQUIVALENCE ( COMINT( 72 ), PSID ) EQUIVALENCE ( COMINT( 76 ), WZ ) COMMON	GEOM GEOM GEOM GEOM GEOM GEOM GEOM GEOM	10 20 30 50 60 70 80 90 100 110 120 130
С	1 (BMASS, CBIXX, CBIXZ, CBIYY, CBIYZ, CBIZZ, FPMASS, CBIXY, 2 DC(3,3) COSPHI = COS( PHI ) SINPHI = SIN( PHI ) COSTHA = COS( THTA ) SINTHA = SIN( THTA ) COSPSI = COS( PSI ) DC(1,1) = COSTHA * COSPSI A = SINTHA * COSPSI B = COSPHI * SINPSI DC(1,2) = SINPHI * A - B C = SINPHI * SINPSI DC(1,3) = COSPHI * A + C DC(2,1) = COSTHA * SINPSI DC(2,2) = SINTHA * C + COSPHI*COSPSI DC(2,3) = SINTHA * B - SINPHI*COSPSI DC(3,1) = - SINTHA DC(3,2) = SINPHI*COSTHA *********	GEOM GEOM GEOM GEOM GEOM GEOM GEOM GEOM	150 160 170 180 200 220 240 250 260 2780 260 2780 260 310 320 340 350
	IF (ABS(COSTHA)-0.1E-10) 100,100,200 200 CONTINUE PSID =(WZ*COSPHI + WY*SINPHI) / COSTHA PHID = WX + SINTHA* PSID THTAD = WY*COSPHI - WZ*SINPHI RETURN 100 CONTINUE PSID = 0.0 PHID = WX THTAD = SQRT(WY*WY+WZ*WZ) RETURN END	GEOM GEOM GEOM GEOM GEOM GEOM GEOM GEOM	360 370 380 390 400 410 420 430 440 450 460 470

C C

449

4

•

-	SUBROUTINE STRUT(STRP, SPNGC, SPNGT, DISTC, DISTT, PFORC, PFORT, FOREVC, FOREVT, IPOSC, IPOST, INDULC, INDULT, IPREV,	STRU STRU	10 20
ź	2 IRET,NNN )	STRU	30
		STRU	40
	THIS SUBROUTINE DETERMINES THE FORCES IN A TYPICAL	STRU	50
	LANDING GEAR STRUT	STRU.	60
		STRU	70
	THE STRUT FORCES ACTING ON THE FOOTPAD IN THE SURFACE	STRU	80
	COORDINATE SYSTEM ARE RETURNED	STRU	90
		STRU	100
	DIMENSION GC(4,5), GCD(4,5), GCDD(4,5)	STRU	110
	DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),	STRU	120
	1 XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),	STRU	130
4	2 XFPSDD(4,5), YFPSDD(4,5), ZFPSDD(4,5)	STRU	140
	COMMON 7 THISV7 TIMHSA(I), ATTH(5), AM(6,6), AMEI(3), INDEPI(6),	SIRU	150
1	I = INDEPC(0)	SIRU	100
;	2 STRPDS(10), STRPMS(5), TPOCDS(10), TPOCMS(5), URCDS (10),	SIKU	10
-	$\beta_{1}$ or $\beta_{2}$ or $\beta_{3}$ or	CTRI	190
	5 INDIMS( 5), PRECOS (10), DRETOS (10), PRECOS ( 5), PRETOS ( 5),	STRU	200
;	TRANS (10), TRANS (5), FRVDSC(10), FRVDST(10), FRVMSC(5),	STRU	210
ì	[FRVMST(5), IPOTDS(10), IPOTMS(5)]	STRU	220
	COMMON COMINT(400)	STRU	230
	EQUIVALENCE ( COMINT( 3 ), IBOTM )	STRU	240
	EQUIVALENCE ( COMINT( 84 ), GC )	STRU	250
	EQUIVALENCE ( COMINT(104 ), GCD )	STRU	260
	EQUIVALENCE ( COMINT(124 ), GCDD )	STRU	270
	EQUIVALENCE ( COMINT(144 ), XFPS )	STRU	280
	EQUIVALENCE ( COMINT(164 ), XFPSD )	STRU	290
	EQUIVALENCE ( COMINT(184 ), XEPSDD )	STRU	300
	EQUIVALENCE ( COMINT(204 ), YEPS )	STRU	310
	EQUIVALENCE ( COMINT(224 ), YFPSD )	STRU	320
	EQUIVALENCE ( COMINT(244 ), YEPSDD )	STRU	330
	EQUIVALENCE ( COMINT(264 ), ZFPS )	STRU	340
	EQUIVALENCE ( COMINI(284), ZEPSD )	STRU	350
	EQUIVALENCE ( COMINI(304 ), ZEPSDD )	STRU	270
	CONTRACTOR (COMINICISES ), CORDI )	SIRU	200
	COMMON	STRU	200
	$ \begin{array}{c} 1 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \\$	STRU	400
	2 = WN7(5) + DY(5) + DY(5) + D7(5) + GM(5) + OM(5) +	STRU	410
	(RAV) $(RAVE)$ $(TETA)$ $(TETS)$ $(A)$ $(RAVE)$ $(RAVE$	STRU	420
	S ESYSI(5) • ESZSI(5) • SOUX(5) •	STRU	430
	6 SOILY(5) • SOILZ(5) • PMSX(5,5) •	STRU	440
	7 PMSY(5,5) , PMSZ(5,5) , PDSX(10,5) ,	STRU	450
	8 PDSY(10,5) , PDSZ(10,5) , FLXS , FLYS , FLZS ,	STRU	460
,	9 TLXL , TLYL , TLZL , SLO , XMSCB(5) ,	STRU	470
	C YMSCB(5) , ZMSCB(5) , XDSCB(10) ,	STRU	480
I	D YDSCB(10) , ZDSCB(10) , ILEG , IMS ,	STRU	490
	E FSTX , FSTY , FSTZ , PVCBX , PVCBY , PVCBZ ,	STRU	500
	F PVFPX , PVFPY , PVFPZ , NOLEG , SLOMS(5) ,	STRU	510
,	G SLODS(10)	STRU	520
	COMMON	STRU	530
	1 PFCMS(5) , PFCDS(5) , PFTDS(5) ,	STRU	540
	2 PFTMS(5) • SRCDS(5) • SRCMS(5) •	STRU	550

¥

SRTDS(5) , SRTMS(5) , COEFDS, COEFMS, GAMDS , GAMMS , SRUCDS, SRUCMS, SRUTDS, SRUTMS, SCMXDS, SCMXMS, STRU 560 3 STRU 570 4 , CDCMS(5) STRU 580 5 STMXDS, STMXMS, CDCDS(5) , FRICDS, FRICMS, IRETDS, STRU 590 6 CDTDS(5) , CDTMS(5) , STRKDS(10) , STRKMS( 5) IRETMS, STROKE(10) STRU 600 7 9 LENGTH, CDXS, CDYS, CDZS, NTYPE, RAD(4), SS(4), ATTHCK(3), STRU 610 8 , NMODES, STRU 620 9 ATTPRS(3) , ADIST , SOILP(3) STRU 630 COEF , GAMMA , FRIC , SCMAX , STMAX , CDC(5), CDT(5), A SRULT , SRULC , SRT(5), SRC(5), PFT(5), PFC(5), GFLEGS(5), STRU 640 В CURMSV, CURMSL, INDFXD, INDFYD, INDFZD, INDFXR, INDFYR, STRU 650 C STRU 660 D INDFZR, TIMAX, DRAGST, IFP STRU 670 COMMON NOOUT,XOUT(10),YOUT(10),ZOUT(10), STRU 680 1CMS, CDCONT, SOILNU, SLRHO, 2MODEIN, POUTX(10,5), POUTY(10,5), POUTZ(10,5), PCGX(5), PCGY(5), **STRU 690** 3PCGZ(5),AEI1,AEI2,FORMS(5),FORDS(10),FORCE,NFORC,SAVMSX(5), STRU 700 STRU 710 4SAVMSZ(5),SAVDSX(10),SAVDSY(10),SAVDSZ(10),IQUOUT,GSINZT, 5GCOSZT, STAB, STABVL, ISTAB, JCKSAB, VELX, VELY, VELZ, SAVMSY(5) STRU 720 STRU 730 COMMON SMXMSC(5), TMXMSC(5), SMXMST(5), TMXMST(5), SMXDSC(10), TMXDSC(10), SMXDST(10), TMXDST(10) STRU 740 1 **STRU 750** 2 STRU 760 ,SLNGMS(5) ,SLNGDS(10), CURDSL, INLEG, IFPPRT, 3 STRU 770 IMPACT(5) , IPRTFP(5), KOUNT(5) , ANGX, ANGY, ANGZ 4 STRU 780 INITIALIZE SUBROUTINE **STRU 790 STRU 800** STRU 810 DX=PVCBX-PVFPX STRU 820 DY=PVCBY-PVFPY STRU 830 DZ=PVCBZ-PVFPZ STRU 840 SLNGTH=SQRT(DX*DX+DY*DY+DZ*DZ) STRU 850 CDX=DX/SLNGTH STRU 860 CDY=DY/SLNGTH STRU 870 CDZ=DZ/SLNGTH STRU 880 STR=SLNGTH-SLO STRU 890 ABSTR=ABS(STR) **STRU 900** VELST=(STR-STRP)/CURDT STRU 910 ABVEL=ABS(VELST) STRU 920 IF(IMS.NE.1)GO TO 10 STRU 930 CDXS=CDX STRU 940 CDYS=CDY STRU 950 CDZS=CDZ STRU 960 CURMSV=VELST STRU 970 CURMSL=SLNGTH STRU 980 **10 CONTINUE** IF(IMS.EQ.0)CURDSL=SLNGTH STRU 990 IJKC=IPOSC STRU1000 STRU1010 IJKT=IPOST STRU1020 STRU1030 DETERMINE STROKING DIRECTION STRU1040 STRU1050 IF(STR.LT.0.0)GO TO 500 STRU1060 IF(STR.GT.0.0)GO TO 400 STRU1070 STRU1080 * ¥ ¥ STRU1090 ZERO STRUT STROKE STRU1100 ¥ ¥ ¥ ¥ × STRU1110

C C C

c c

С

С

C C C C C C

```
FORCE=0.0
      GO TO 2002
С
        ¥
           * *
C
C
C
C
      ÷¥-
                  *
      STRUT IN TENSION
         ¥
            ¥
               ¥
                  ¥
С
  400 CONTINUE
      IF(ABSTR.GT.STMAX)GO TO 5001
С
С
      WAS PREVIOUS STEP ON COMPRESSION SIDE
С
      IF(IPREV.EQ.0)GO TO 402
      IF(IPREV.NE.-1)GO TO 402
      IF(PFORC.EQ.0.0)GO TO 402
      IPREV=+1
      IF(IRET.EQ.1) GO TO 651
      GO TO 626
  402 CONTINUE
      IPREV=+1
      IF (ABSTR.GT.DISTT) GO TO 401
      FORCE=0.0
      GO TO 2001
  401 CONTINUE
      IF(STR.LT.STRP)GO TO 425
С
С
      STRUT LOADING
С
      IF (SPNGT.EQ.0.0)GO TO 405
      FORCE=SPNGT*(ABSTR-DISTT)
      IF(FORCE.LT.FOREVT)GO TO 2001
      FOREVT=0.0
      INDULT=0
      SPNGT=0.0
      IF(IJKT.EQ.1)GO TO 407
      DISTT=CDT(IJKT-1)-PFT(IJKT-1)/SRT(IJKT)
      GO TO 408
  407 DISTT=0.0
  408 CONTINUE
      IF (FORCE.LT.PFT(IJKT))GO TO 405
      FORCE=PFT(IJKT)
      IF(ABSTR.LT.CDT(IJKT))GO TO 2001
  405 CONTINUE
  406 IF(ABSTR.LT.CDT(IJKT))GO TO 410
      DISTT=CDT(IJKT)-PFT(IJKT)/SRT(IJKT+1)
      IPOST=IPOST+1
      IJKT=IPOST
      GO TO 406
  410 CONTINUE
      FORCE=SRT(IJKT)*(ABSTR-DISTT)
      IF(FORCE.GT.PFT(IJKT))FORCE=PFT(IJKT)
      GO TO 2001
С
      STRUT UNLOADING
С
С
  425 CONTINUE
```

STRU1120 STRU1130 STRU1140 STRU1150 STRU1160 STRU1170 STRU1180 STRU1190 STRU1200 STRU1210 STRU1220 STRU1230 STRU1240 STRU1250 STRU1260 STRU1270 STRU1280 STRU1290 STRU1300 STRU1310 STRU1320 STRU1330 STRU1340 STRU1350 STRU1360 STRU1370 STRU1380 STRU1390 STRU1400 STRU1410 STRU1420 STRU1430 STRU1440 STRU1450 STRU1460 STRU1470 STRU1480 STRU1490 STRU1500 STRU1510 STRU1520 STRU1530 STRU1540 STRU1550 STRU1560 STRU1570 STRU1580 STRU1590 STRU1600 STRU1610 STRU1620 STRU1630 STRU1640 STRU1650 STRU1660

STRU1670

```
IF(IRET.EQ.1)GO TO 450
     IF(INDULT.NE.O)GO TO 475
 426 SPNGT=SRULT
      IF(PFORT.NE.PFT(IPOST))SPNGT=SRT(IPOST)
      FOREVT=PFORT
     GO TO 470
 450 CONTINUE
     IF(INDULT.NE.0)GO TO 475
  451 CONTINUE
      FOREVT=PFORT
      IF(PFORT.EQ.PFT(IJKT))GO TO 465
      SPNGT=SRT(IJKT)
      GO TO 470
  465 CONTINUE
      SPNGT=SRULT
      IF((IJKT+1).GT.5)GO TO 470
      SPNGT=SRT(IJKT+1)
  470 CONTINUE
      INDULT=1
      DISTT=ABS(STRP)-PFORT/SPNGT
      IF(IPREV.EQ.-1)GO TO 602
  475 CONTINUE
      FORCE=SPNGT*(ABSTR-DISTT)
      IF(ABSTR.LT.DISTT)FORCE=0.0
      GO TO 2001
С
C
C
      * * * * *
      STRUT IN COMPRESSTION
С
      * * * *
                  ×
С
  500 CONTINUE
      IF(ABSTR.GT.SCMAX)GO TO 5000
С
¢
      WAS PREVIOUS STEP ON TENSION SIDE
С
      IF(IPREV.EQ.0)GO TO 602
      IF(IPREV.NE.+1)GO TO 602
      IF (PFORT.EQ.0.0)GO TO 602
      IPREV=-1
      IF(IRET.EQ.+1)GO TO 451
      GO TO 426
  602 CONTINUE
      IPREV=-1
      IF(ABSTR.GT.DISTC)GO TO 600
      FORCE=0.0
      GO TO 2000
  600 CONTINUE
      IF(STR.GT.STRP)GO TO 625
С
С
      STRUT LOADING
С
      IF(SPNGC.EQ.0.0)GO TO 605
      FORCE=SPNGC*(ABSTR-DISTC)
      IF(FORCE.LT.FOREVC)GO TO 2000
      FOREVC=0.0
      INDULC=0
```

STRU1680 STRU1690 STRU1700 STRU1705 STRU1710 STRU1720 STRU1730 STRU1740 STRU1750 STRU1760 STRU1770 STRU1780 STRU1790 STRU1800 STRU1810 STRU1820 STRU1830 STRU1840 STRU1850 STRU1860 STRU1870 STRU1880 STRU1890 STRU1900 STRU1910 STRU1920 STRU1930 STRU1940 STRU1950 STRU1960 STRU1970 STRU1980 STRU1990 STRU2000 STRU2010 STRU2020 STRU2030 STRU2040 STRU2050 STRU2060 STRU2070 STRU2080 STRU2090 STRU2100 STRU2110 STRU2120 STRU2130 STRU2140 STRU2150 STRU2160 STRU2170 STRU2180 STRU2190 STRU2200 STRU2210 STRU2220
	SPNGC=0.0	STRU2230
	IF(IJKC.EQ.1)GO TO 607	STRU2240
	DISTC=CDC(IJKC-1)-PFC(IJKC-1)/SRC(IJKC)	STRU2250
	GO TO 608	STRU2260
607	DISTC=0.0	STRU2270
608	CONTINUE	STRU2280
	IF(FORCE.LT.PFC(IJKC))G0 T0 605	STRU2290
	FORCE=PFC(IJKC)	STRU2300
	IF(ABSTR.LT.CDC(IJKC))GO TO 2000	STRU2310
605	CONTINUE	STRU2320
606	IF(ABSTR.LT.CDC(IJKC))G0 T0 610	STRU2330
	DISTC=CDC(IJKC)-PFC(IJKC)/SRC(IJKC+1)	STRU2340
	IPOSC=IPOSC+1	STRU2350
	IJKC=IPOSC	STRU2360
	GO TO 606	STRU2370
610	CONTINUE	STRU2380
	FORCE=SRC(IJKC)*(ABSTR-DISTC)	STRU2390
	IF(FORCE.GT.PFC(IJKC))FORCE=PFC(IJKC)	STRU2400
	GO TO 2000	STRU2410
C		STRUZ420
C	STRUT UNLOADING	STRUZ430
C (OF		S1RU2440 STRU2450
625	CONTINUE	STR02450
	IF (INCH C NE ONCO TO CZE	STR02400
626		STRU2470
020	JENGC-SKOLC	STRU2485
	FOREVC=DEORC	STRU2490
		STRU2500
650		STRU2510
0,00	$IF(INDULC \cdot NF \cdot 0)GO TO 675$	STRU2520
651	CONTINUE	STRU2530
	FOREVC=PFORC	STRU2540
	IF(PFORC.EQ.PFC(IJKC))GO TO 665	STRU2550
	SPNGC=SRC(IJKC)	STRU2560
	GO TO 670	STRU2570
665	CONTINUE	STRU2580
	SPNGC=SRULC	STRU2590
	IF((IJKC+1).GT.5) GO TO 670	STRU2600
	SPNGC=SRC(IJKC+1)	STRU2610
670	CONTINUE	STRU2620
	INDULC=1	STRU2630
	DISTC=ABS(STRP)-PFORC/SPNGC	STRU2640
	IF(IPREV.EQ.+1)GO TO 402	STRU2650
675	CONTINUE	STR02660
	FORCE=SPNGC*(ABSTR-DISTC)	STRU2670
~	IF (ABSTR•LI•DISTC)FORCE=0•0	51RU2000 STRU2000
C		STR52090
C	UPDATE INDICATORS	STRU2700 STRU2710
	CONTINUE	STRU2710
2000		STRU2720
	CO TO 2005	STRU2750
2001		STRU2750
2001		STRU2760
	60  TO  2005	STRU2770

2002 2005	CONTINUE PFORC=0.0 PFORT=0.0 CONTINUE STROKE(NNN)=STR	STRU2780 STRU2790 STRU2800 STRU2810 STRU2820
c c c	DETERMINE FRICTION AND DAMPING FORCES	STRU2830 STRU2840 STRU2850 STRU2860
c c	TOTAL STRUT FORCE FORCE=SIGN(1.,STR)*FORCE+FRIFOR ESTX=CDX*EOPCE	STRU2870 STRU2880 STRU2890 STRU2900
C	FSTY=CDZ*FORCE FSTZ=CDZ*FORCE RETURN	STRU2910 STRU2920 STRU2930 STRU2940
C C 5000	STRUT BOTTOMED OUT ON COMPRESSION SIDE	STRU2950 STRU2960 STRU2970
c	IBOTM=-1 RETURN	STRU2980 STRU2990 STRU3000
C C 5001	STRUT BOTTOMED OUT ON TENSION SIDE	STRU3010 STRU3020 STRU3030 STRU3040
	RETURN END	STRU3050 STRU3060

SUBROUTINE SOIL	SOIL 10
	SOIL 20
THIS SUBROUTINE DETERMINES THE SOIL FORCES ACTING ON	SOIL 30
A FOUTPAD	SOIL 40
THE EVALUATION OF THE FOOTPAD ATTENUATION SYSTEM FORCES	SOIL 50
IS ALSO INCLUDED HERE	SOIL 70
	SOIL 80
THE SOIL/ATTENUATION FORCES ACTING ON THE FOOTPAD IN THE	SOIL 90
SURFACE COORDINATE SYSTEM ARE RETURNED	SOIL 100
	SOIL 110
DIMENSION GC(4,5), GCD(4,5), GCDD(4,5)	SOIL 120
DIMENSION $XFPS(4,5)$ , $YFPS(4,5)$ , $ZFPS(4,5)$ ,	SOIL 130
1 XFPSU(4,5), YFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), YFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), XFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), XFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), XFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), XFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), XFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), ZFPSU(4,5), br>xFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), ZFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xFPSU(4,5), xF	SOIL 140
2 XFPSDD(4,57, FFPSDD(4,57, ZFPSDD(4,57)	SOIL 190
COMMON / THISV/ TIMHSA()), ATTH(5), AM(6,6), AME1(3), INDEPI(6),	SOIL 100
1 INDFPC(6),	SOIL 180
2 STRPDS(10), STRPMS( 5), IPOCDS(10), IPOCMS( 5), URCDS (10	), SOIL 190
3 URTDS (10), URCMS ( 5), URTMS ( 5), SETCDS(10), SETTDS(10	), SOIL 200
4 SETCMS( 5), SETTMS( 5), INDCDS(10), INDTDS(10), INDCMS(10	), SOIL 210
5 INDTMS( 5), PRFCDS (10), PRFTDS (10), PRFCMS ( 5), PRFTMS (	5), SOIL 220
6 IPRDS (10), IPRMS (5), FRVDSC(10), FRVDST(10), FRVMSC(5)	), SOIL 230
L = FRVMS((5), FPOIDS(10), FPOIMS(5)	SUIL 240
$EOUTVALENCE (COMINT(B_{4}), CC)$	SOIL 290
EQUIVALENCE ( COMINT(104 ), GCD )	SOIL 200
EQUIVALENCE ( COMINT(124 ), GCDD )	SOIL 280
EQUIVALENCE ( COMINT(144 ), XFPS )	SOIL 290
EQUIVALENCE ( COMINT(164 ), XFPSD )	SOIL 300
EQUIVALENCE ( COMINT(184 ), XFPSDD )	SOIL 310
EQUIVALENCE ( COMINT(204 ), YFPS )	SOIL 320
EQUIVALENCE ( COMINT(224 ), YFPSD )	SOIL 330
EQUIVALENCE ( COMINT(244 ), YEPSDD )	SOIL 340
EQUIVALENCE ( COMINI(264 ), ZEPS )	SUIL 350
EQUIVALENCE ( COMINT(204 ), ZEPSD )	SOIL 300
FOUTVALENCE ( COMINT(324 ) TIME )	SOIL 380
EQUIVALENCE ( COMINT( 343) • IFIN )	SOIL 390
EQUIVALENCE ( COMINT(368 ), CURDT )	SOIL 400
COMMON	SOIL 410
1 CBMASS, CBIXX, CBIXZ, CBIYY, CBIYZ, CBIZZ, FPMASS, CB	IXY, SOIL 420
2 DC(3,3) , XFP(5), YFP(5), ZFP(5), WNX(5), WNY(5),	SOIL 43(
3 WNZ(5), PX(5), PY(5), PZ(5), GM(5), OMEGA(5),	SOIL 440
4 GRAV , GRAVE , ZEIA , FIS (6) , FSXSI(5) ,	SOIL 450
5 FSTSI(5) 9 FSZSI(5) 9 SUILX(5) 9	SOIL 460
7 PMSY(5.5) • PMS7(5.5) • PDSX(10.5) •	SOT 480
8 PDSY(10,5) , PDSZ(10,5) , FLXS , FLYS , FL7S ,	SOIL 490
9 TLXL , TLYL , TLZL , SLO , XMSCB(5)	SOIL 500
C YMSCB(5) , ZMSCB(5) , XDSCB(10) ,	SOIL 510
D YDSCB(10) , ZDSCB(10) , ILEG , IMS ,	SOIL 52(
E FSTX , FSTY , FSTZ , PVCBX , PVCBY , PVCBZ ,	SOIL 530
F PVFPX , PVFPY , PVFPZ , NOLEG , SLOMS(5) ,	SOIL 540
G SLODS(10)	SOIL 550

.

с

SOIL 560 COMMON SOIL 570 , PFTDS(5) PFCMS(5) , PFCDS(5) 1 9 , SRCDS(5) sRCMS(5) SOIL 580 PFTMS(5) 2 , , COEFDS, COEFMS, GAMDS , SRTMS(5) 3 SRTDS(5) SOIL 590 GAMMS , SRUCDS, SRUCMS, SRUTDS, SRUTMS, SCMXDS, SCMXMS, SOIL 600 4 , CDCMS(5) 5 STMXDS, STMXMS, CDCDS(5) SOIL 610 CDTDS(5) , CDTMS(5) , FRICDS, FRICMS, IRETDS, SOIL 620 6 • STRKMS( 5) • IRETMS, STROKE(10) STRKDS(10) SOIL 630 7 LENGTH, CDXS, CDYS, CDZS, NTYPE, RAD(4), SS(4), ATTHCK(3), SOIL 640 8 ATTPRS(3) , ADIST , SOILP(3) , NMODES, SOIL 650 9 COEF , GAMMA , FRIC , SCMAX , STMAX , CDC(5), CDT(5), SOIL 660 А SRULT , SRULC , SRT(5), SRC(5), PFT(5), PFC(5), GFLEGS(5), SOIL 670 R CURMSV, CURMSL, INDFXD, INDFYD, INDFXD, INDFXR, INDFYR, SOIL 680 C D INDFZR, TIMAX, DRAGST, IFP SOIL 690 COMMON SOIL 700 SOIL 710 SOIL 720 1CMS, CDCONT, SOILNU, SLRHO, NOOUT, XOUT(10), YOUT(10), ZOUT(10), 2MODEIN,POUTX(10,5),POUTY(10,5),POUTZ(10,5),PCGX(5),PCGY(5), SOIL 730 3PCGZ(5),AEI1,AEI2,FORMS(5),FORDS(10),FORCE,NFORC,SAVMSX(5), 4SAVMSZ(5),SAVDSX(10),SAVDSY(10),SAVDSZ(10),IQUOUT,GSINZT, SOIL 740 SOIL 750 5GCOSZT,STAB,STABVL,ISTAB,JCKSAB,VELX,VELY,VELZ,SAVMSY(5) COMMON SOIL 760 SMXMSC(5), TMXMSC(5), SMXMST(5), TMXMST(5), SMXDSC(10), TMXDSC(10), SMXDST(10), TMXDST(10) SOIL 770 1 SOIL 780 2 slngms(5) ,slngds(10), curdsl, inleg, ifpprt, SOIL 790 3 SOIL 800 IMPACT(5) , IPRTFP(5), KOUNT(5) , ANGX, ANGY, ANGZ 4 SOIL 810 SOIL 820 SUBROUTINE INITIALIZATION SOIL 830 SOIL 840 KJI=1 SOIL 850  $I_JK = 1$ PIE=3.14159265 SOIL 860 SOIL 870 PIE2=PIE/2. DEPTH=XFPS(1,IFP)-ATTH(IFP) SOIL 880 ADEPTH=ABS(DEPTH) SOIL 890 SOIL 900 FPSTR=ATTH(IFP)-ADEPTH SOIL 910 IF(DEPTH.GE.0.0)GO TO 2000 SOIL 920 SOIL 930 FOOTPRINT AREA SOIL 940 SOIL 950 IF(FPSTR.LE.SS(4)) GO TO 101 SOIL 960 100 IJK=IJK+1 SOIL 970 IF(IJK.GT.4) GO TO 101 IF(FPSTR.LT.SS(IJK)) GO TO 100 SOIL 980 SOIL 990 IF(FPSTR.EQ.SS(IJK))GO TO 102 DRDD = (RAD(IJK) - RAD(IJK-1)) / (SS(IJK-1) - SS(IJK))SOIL1000 RR=RAD(IJK)-DRDD*(FPSTR-SS(IJK)) SOIL1010 GO TO 104 SOIL1020 101 RR=RAD(4) SOIL1030 GO TO 103 SOIL1040 102 RR=RAD(IJK) SOIL1050 SOIL1060 103 DRDD=0.0 104 AREA=PIE*RR*RR SOIL1070 IF(NTYPE.EQ.0) GO TO 500 SOIL1080 SOTI 1090 * * SOIL1100 SECONDARY SOIL MECHANICS SOIL1110

C C C

C C

С

C C

С



с		*	*	¥	¥	*	SOIL1120
С							SOIL1130
		PRI	ESS	=AD	EPT	H*SOILP(2)	SOIL1140
			(PR	ESS	•GT	•SOILP(3)) PRESS=SOILP(3)	SOIL1150
r		50	ILX	(15	P)=	PRESS*AREA	SOIL1160
ĉ		CH	ECV	MA	GNT		SOILII70
ĉ		Cn	ECK	MM	GNI	TUDE OF SUIL FURCE	SULLIDU
C		s۵	TI C	D-F	БМΛ		SUIL1190
		10	100		г ма D I	T = 0 = 0 = 0 = 0	50111210
		TE	150		11E	D). GT. SOILCR-0+0	SOIL1210
		DD	FSC	- < 0		(TED)/AREA	SOIL1220
	150	co	NTI	MUE	ILA		SOIL1250
	100	ŇВ	AR≈	SQR	T(Y	EPSD(],IEP)**2+7EPSD(1,IEP)**2)	SOIL 1250
		IF	(VB	AR.	LT.	0.000001)GO TO 200	SOIL1260
		CO	EF=	SOI	LPI	1)	SOIL1270
		CY	=CO	EF*	YFP	SD(1,IFP)/VBAR	SOIL1280
		CZ	=CO	EF*	ZFP	SD(1,IFP)/VEAR	SOIL1290
		GO	то	21	0		SOIL1300
	200	CΥ	=0.	0			SOIL1310
		CZ	=0•	0			SOIL1320
	210	CO	NTI	NUE			SOIL1330
		SO	ILY	(IF	P)=	-CY*SOILX(IFP)	SOIL1340
		SO	ILZ	(1F	P)=	-CZ*SOILX(IFP)	SOIL1350
_		GO	10	10	00		SOIL1360
Ċ		.,					SOIL1370
C		*	*	*	*		S01L1380
Č		PR	IMA	RY	SÕI	LMECHANICS	SOIL1390
ĉ		*	*	*	*	*	SOIL1400
C	600	<u> </u>	<u>ыт</u> т	N11 1			SOIL1410
	500	- CO - VB		SOD	TIX	EDSD(1, [ED)**2+VEDCD(1, [ED)**2+7EDSD(1, [ED)**2)	SOIL1420
		 		SOP	T ( V	FOC(1,1F)**2+1FOC(1,1F)**21	50111440
		тн	FTA	1=0		F 5 D ( 1 9 1) F ) ** Z ( Z ( F 5 D ( 1 9 1) F ) ** Z )	SOTE 1450
		TE		AR.	ĠT.	0.000001) THETAL = ASIN (VHOR (VBAR)	50111460
		IF.	IXE	PSD	(1.	$IEP1 = GT_{0} = THETAI = PIE - THETAI$	50111470
		co.	ST=	ABS	ico	S(THETAL))	50111480
		ŠĪ	NT=	SIN	(TH	ETAL)	SOIL1490
		ΑT	ніс	K=R	R*A	DEPTH	S0IL1500
		ΑT	нта	=AR	EA*	COST+ATHICK*SINT	SOIL1510
		AP	ERP	=AR	EA*	SINT+ATHICK*COST	SOIL1520
		FΡ	H I =	1	2•*	THETAL/PIE	SOIL1530
		AL	AM	= .	25*	(APERP/ATHTA)*(1EXP(-50.*THETAL))*(1.+SINT)	SOIL1540
		IF	(TH	ETA	L.L	T.PIE2) GO TO 505	SOIL1550
		FΡ	H I =	0•			SOIL1560
		AL	AM	≖.	5*(	APERP/ATHTA)	SOIL1570
	505	CO	NTI	NUE			SOIL1580
		ΙF	(ТН	ETA	L. G	T.(3.*PIE/4.))ALAM =0.0	SOIL1590
		CD	=•8	+CD	CON	T*RR*RR*FPHI	SOIL1600
		AΑ	A = C	D*S	LRH	O*ATHTA*VBAR*VBAR	SOIL1610
		83	8=3	•*S	OIL	NU*RR*RR*DRDD*((ATHTA/AREA)**1.5)*VBAR*VBAR*COST	SOIL1620
		CC	C=C	MS*	SLR		S01L1630
		FA	P=C	CC*	ADE		SUIL1640
		15	(VD	AK		0.0000017 GO 10 510 N/1 JEDNA JE O 0000018 CO EO EO/	SULL650
		11	(AB	SIX	- HS	U(1)1FF7))•L1•U•U00001) GU 10 506	SUIL1660
		30	1N – X	r M 3	いしょ	911 F / / AUG(AF MGU(191F M/)	SUIL16/0

		GO TO 507
	506	SGN=+1.0
	507	CONTINUE
		CX=XFPSD(1,IFP)/VBAR
		IF(VHOR•LT•1•E-10) GO TO 508
		CY=YFPSD(1,IFP)/VHOR
		CZ=ZFPSD(1,IFP)/VHOR
		GO TO 509
	508	CY=0•0
		CZ=0•0
	509	CONTINUE
		CH=VHOR/VBAR
		FH=(CH+CX*ALAM )*FAP
		SOILX(IFP)=(-CX+SGN*CH*ALAM )*FAP
		IF(SOILX(IFP) • LT • 0 • )SOILX(IFP) = 0 • 0
		SOILY(IFP)=-CY*FH
		SOILZ(IFP) = -CZ*FH
		GO TO 520
	510	CONTINUE
		$C_7 = 0.0$
		$SO(1) \times (1EP) = EAP$
		SOILY(IFP)=0.0
		SOILZ(IFP)=0.0
	520	CONTINUE
		PRESS=SOILX(IFP)/AREA
С		
С		* * * * *
С		FOOTPAD ATTENUATION
С		* * * * *
С		
	1000	CONTINUE
		IF(FPSTR•GE•ADIST)GO TO TOOT
		IF (PRESSOLE AT IPRS(3)) RETURN
		A [ H(IFP) = A [ H(K(3) ]
	1001	
	1001	
		CRERES-RTTPRS(RST)
	1002	101004
	1002	KUT=KUT+1
		GO TO 1001
	1003	CRPRES=ATTPRS(3)
	1004	CONTINUE
		IF(CRPRES.GT.PRESS) RETURN
		SOILX(IFP)=CRPRES*AREA
		IF(NTYPE.EQ.0)GO TO 1015
		ADD=CRPRES/SOILP(2)
		GO TO 1020
	1015	CONTINUE
		IF(VBAR.LT.0.000001) GO TO 1016
		FAP=SOILX(IFP)/(-CX+SGN*CH*ALAM)
		ADD = (FAP - AAA - BBB) / CCC
	1	
	1016	ADD=SOILX(IFP)/CCC

SOIL1680 SOIL1690 S0IL1700 SOIL1710 SOIL1720 SOIL1730 SOIL1740 SOIL1750 SOIL1760 SOIL1770 SOIL1780 SOIL1790 SOIL1800 SOIL1810 SOIL1820 SOIL1830 SOIL1840 SOIL1850 SOIL1860 SOIL1870 SOIL1880 SOIL1890 SOIL1900 SOIL1910 SOIL1920 SOIL1930 SOIL1940 SOIL1950 SOIL1960 SOIL1970 SOIL1980 SOIL1990 SOIL2000 SOIL2010 S0IL2020 SOIL2030 SOIL2040 SOIL2050 S0IL2060 SOIL2070 SOIL2080 SOIL2090 SOIL2100 SOIL2110 SOIL2120 SOIL2130 SOIL2140 SOIL2150 SOIL2160 SOIL2170 SOIL2180 SOIL2190 SOIL2200 SOIL2210 SOIL2220 SOIL2230

Δ

IF(ABS(ADD).GT.ADEPTH)ADD=ADEPTH IF(IFIN.EQ.O)ATTH(IFP)=FPSTR+ADD SOILY(IFP)=-CY*SOILX(IFP) SOILZ(IFP)=-CZ*SOILX(IFP) RETURN FOOTPAD OFF SURFACE 2000 CONTINUE •

## C C

С

SOILX(IFP)=0.0 SOILY(IFP)=0.0 SOILZ(IFP)=0.0 RETURN END SOIL2240 SOIL2250 SOIL2260 SOIL2270 SOIL2280 SOIL2300 SOIL2310 SOIL2310 SOIL2330 SOIL2330 SOIL2350 SOIL2350 SOIL2350 SOIL2370

	SUBROUTINE INITUP ( IYD, CUTVAL, IDIRT )	INIT	10
	COMMON / IZZZRM / IPTATL, IPTOTL, LOCNAM(90)	INIT	20
	DIMENSION XS(8), IAD(8), IND(8)	INIT	30
	COMMON COMINT (400)	INIT	40
	FQUIVALENCE ( COMINT( 329 ) X5 )	INIT	50
	EQUIVALENCE ( COMINT( 344 ), IAD )	INIT	60
	EQUIVALENCE ( COMINT( 352 ), IND )	INIT	70
	ENTRY LOC	INIT	80
	IPTATL = IPTATL + 1	INIT	90
	IF ( IPTATL +LE+ 8 ) GO TO 3	INIT	100
	WRITE (6,1)	INIT	110
1	FORMAT ( 65HJOB TERMINATED, MORE THAN EIGHT CALLS TO	LOINIT	120
٦,	*()	INIT	130
	STOP	INIT	140
3	CONTINUE	INIT	150
	IAD(IPTATL) = IYD	INIT	160
	XS( IPTATL ) = CUTVAL	INIT	170
	IND(IPTATL) = IDIRT	INIT	180
	RETURN	INIT	190
	ENTRY INUPD	INIT	200
	IPTOTL = IPTOTL + 1	INIT	210
	IF ( IPTOTL •LE• 90 ) GO TO 4	INIT	220
	WRITE (6,2)	INIT	230
2	FORMAT(67HJOB TERMINATED, MORE THAN NINETY CALLS TO I	NUPINIT	240
ł	*D)	INIT	250
	STOP	INIT	260
4	LOCNAM(IPTOTL) = IYD	INIT	270
	RETURN	INIT	280
	END	INIT	290

	SUBROUTINE RKCUT (YD, Y)	RKCT	10
	COMMON / IZZZRM / IPTATL, IPTOTL, LOCNAM(90)	RKCI	20
	$DIMENSION X(8) \times MNI(8) \times S(8) \times IAU(8) \times IV(8) \times IV$	RKCT	20 40
	CONTROL CONTROL ( $CONTROL 324$ ). T	RKCT	50
	FOULVALENCE (COMINT (325), HMAX)	RKCT	60
	FOULVALENCE ( COMINT( 326 ) HIMAN )	RKCT	70
	EQUIVALENCE ( COMINT( 327 ), EMIN )	RKCT	80
	EQUIVALENCE ( COMINT( 328 ) EMAX )	RKCT	90
	EQUIVALENCE ( COMINT( 329 ), XS )	RKCT	100
	EQUIVALENCE ( COMINT ( 337 ), HZ )	RKCT	110
	EQUIVALENCE ( COMINT( 338 ), CUTERR)	RKCT	120
	EQUIVALENCE ( COMINT( 339 ), IP )	RKCT	130
	EQUIVALENCE ( COMINT( 340 ), IVARH )	RKCT	140
	EQUIVALENCE ( COMINT( 341 ), IMTH )	RKCT	150
	EQUIVALENCE ( COMINT( 342 ), IPRNT )	RKCT	160
	EQUIVALENCE ( COMINT( 343 ), IFIN )	RKCT	170
	EQUIVALENCE ( COMINT( 344 ), IAD )	RKCT	180
	EQUIVALENCE ( COMINT( 352 ), IND )	RKCT	190
	EQUIVALENCE ( COMINT( 360 ), J )	RKCI	200
	EQUIVALENCE ( COMINT( 363 ), IVAL )	RKCI	210
	EQUIVALENCE ( COMINI ( 368 ), H )	RACI	220
	EQUIVALENCE (COMIN)(369), XMNI)	PRCT	250
	XMNI USES COMINI(369-376)	RKCI	240
		DVCT	250
-		DKCT	200
T		RKCT	280
		RKCT	290
		RKCT	300
	$HZ = HAX + 2 \bullet O(0 + * (-IP))$	RKCT	310
	$HD_{2}=HZ/2.00$	RKCT	320
	H=HD2	RKCT	330
	IALP=4	RKCT	340
	IPRNT=0	RKCT	350
	IFIN=0	RKCT	360
	IVAL=0	RKCT	370
	ISCNT = 0	RKCT	380
	IBI1=4	RKCT	390
	IBI2=2	RKCT	400
	IBU1=1	RKCT	410
	IF(IVARH)3,2,3	RKCT	420
2	IBU2=2	RKCT	430
	EMAX= ABS(EMAX)		440
	EMIN= ABS(EMIN)		450
~		RKCI RKCT	400
3			410
			400
	ENIKT INIEU CO. TO/(O.EE.EE.20).IBI1	RKCT	500
20	V//)=V/)+H*VD/])	RKCT	510
דכ	1947-11210010117	RKCT	520
41	FR = ARS(H7*(YD(1)-YD(3)))	RKCT	530
- I	IF(Y(1))43,44,43	RKCT	540
43	IF(ER-ER/ ABS(Y(1))) 44,44,4	RKCT	550

С

4	ER=ER/ ABS(Y(1))	RKCT	560
44	IF(ER-EMAX)45946946	RKCT	570
45	1F(FR-EMIN)50+48+48	RKCT	580
1.0		PKCT	590
40		DKCT	100
	GO 10 50	RKCI	600
46	IVAL=-8300000	RKCT	610
	IF ( HZ •EQ• HMIN ) IVAL = 1	RKCT	620
50	YD(3)=YD(1)	RKCT	630
	RETURN	RKCT	640
55	Y(4) = Y(2) + H * YD(1)	RKCT	650
	$YD(3) = YD(3) + 2 \cdot 00 * YD(1)$	RKCT	660
	RETURN	RKCT	670
60	Y(4) = Y(2) + H/6 = 00*(YD(3) + YD(1))	RKCT	680
•••	YD(3) = YD(1)	RKCT	690
		RKCT	700
		DVCT	710
100		DECT	720
100		DECT	720
		DYCT	750
	IF(IPRNI)II0,119,110	RNCI	740
115	IPT1=IPT2	RKCT	750
	IBU3=2	RKCT	760
110	IF(IALP-1)118,120,118	RKCT	770
120	IPT1=IPT1-1	RKCT	780
	IALP=4	RKCT	790
	H=HD2	RKCT	800
	IFIN=0	RKCT	810
121	DO 122 IMVER = 1, IPTOTL	RKCT	820
	KMVER = LOCNAM( IMVER )	RKCT	830
122	COMINT( KMVER ) = COMINT( KMVER + 3 )	RKCT	840
150	IPRNT=IPT1	RKCT	850
<b>1</b> 2 0	IBII=IAI P	RKCT	860
		RKCT	870
	RETURN	RKCT	880
118		RKCT	890
TIO	[O I O (125 - 126 - 127) + IA] P	RKCT	900
1 26	LE ( 18/12 - 1) 30/2 - 30/1 - 30/2	RKCT	910
202		RKCT	920
502		RKCT	030
0.01		PKCT	040
301		DVCT	050
			920
125	I=I+HD2		900
	GO TO 121	RKCI	970
127	GO TO(130,131),1BU1	RKCI	980
130	DO 132 IMVER = 1, IPTOTL	RKCT	990
	KMVER = LOCNAM( IMVER )	RKCT	1000
	COMINT( KMVER + 1 ) = COMINT( KMVER )	RKCT	1010
132	COMINT( KMVER ) = COMINT( KMVER + 3 )	RKCT	1020
	T=T+HD2	RKCT	1030
	GO TO 150	RKCT	1040
131	IF(IVAL)135,136,135	RKCT	1050
136	IF(ISCNT-1)137,137,138	RKCT	1060
137	ISCNT=ISCNT+1	RKCT	1070
	GO TO 130	RKCT	1080
138	HIPT1=IPT1/2	RKCT	1090
	XIPT1=IPT1	RKCT	1100
	XIPT1=XIPT1/2•00	RKCT	1110

140	IF(XIPT1-HIPT1)130,140,130 IPT2=IPT2/2 IPT1=IPT1/2
	ISCNT=0
	IVAL=0
120	
1 3 7	
	IBU1=1
	GO TO 150
135	ISCNT=0
	IF(IVAL)160,160,130
160	IF(IPT1)130,161,161
161	IF(IBU3-1)163,165,163
163	
165	IP 1=2*(IP 1+1)
	TPT2-2*1PT2 T-T_W7
	IF ( $HZ \bullet LT \bullet HMIN$ ) $HZ = HMIN$
	HD2=HZ/2.00
	H=HD2
	DO 170 IMVER = 1, IPTOTL
	KMVER = LOCNAM( IMVER )
170	COMINT(KMVER) = COMINT(KMVER+1)
	LE(LEIN)200+250+200
200	J=0
200	IERROR = 1
	RETURN
250	K = 1
260	IF ( K •LE• IPTATL ) GO TO 300
270	IF(K-1)280,200,280
280	
	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
200	XMNI(I) = COMINT(KMVER)
270	GO = TO = 200
300	XU=XS(K)+•5*CUTERR
	XL=XS(K)5*CUTERR
	IF(IND(K))500,310,500
310	KMVER = IAD(K)
	X(K) = COMINT(KMVER)
	IF(X(K)-XL) 320,320,400
320	IF(IVAL)20093309530
0 و و	N=N+1 TE(K=9)260,280,280
	IF(X - 3)200,280,200
410	J=K
	IERROR = 1
	RETURN
500	KMVER = IAD(K)
	X(K) = COMINT( KMVER )

)

RKCT1120 **RKCT1130** RKCT1140 RKCT1150 RKCT1160 RKCT1170 RKCT1180 RKCT1190 RKCT1200 RKCT1210 **RKCT1220 RKCT1230** RKCT1240 RKCT1250 RKCT1260 RKCT1270 RKCT1280 RKCT1290 RKCT1300 RKCT1310 RKCT1320 RKCT1330 RKCT1340 RKCT1350 RKCT1360 **RKCT1370** RKCT1380 RKCT1390 RKCT1400 RKCT1410 **RKCT1420** RKCT1430 RKCT1440 **RKCT1450** RKCT1460 RKCT1470 RKCT1480 RKCT1490 RKCT1500 RKCT1510 RKCT1520 RKCT1530 RKCT1540 RKCT1550 RKCT1560 RKCT1570 RKCT1580 RKCT1590 RKCT1600 RKCT1610 RKCT1620 RKCT1630 RKCT1640 RKCT1650 **RKCT1660** RKCT1670

\$

	IF(X(K)-XU)510,320,320	RKCT1680
510	IF(X(K)-XL)600,600,410	RKCT1690
600	IF ( IVAL •LT• 0 ) GO TO 200	RKCT1700
	IF ( IERROR •NE• 0 ) GO TO 1054	RKCT1710
	WRITE(6,1051)K	RKCT1720
1051	FORMAT(1H0,28H******** CUTOFF PASSED BY ,I1,56HTH CUTOFF	VARIABLRKCT1730
· ]	LE ON THE INITIAL CALL TO CUT *********** )	RKCT1740
	STOP	RKCT1750
1054	CONTINUE	RKCT1760
	HN=HZ/2.00*((XS(K)-XMN1(K))/(X(K)-XMN1(K)))	RKCT1770
	T = T - HZ	RKCT1780
	HZ=HN	RKCT1790
	HD2=HZ/2.00	RKCT1800
	H=HD2	RKCT1810
	IALP=4	RKCT1820
	IVAL=0	RKCT1830
	IBI2=2	RKCT1840
	IBU1=1	RKCT1850
	IBI1=IALP	RKCT1860
	DO 640 IMVER = 1, IPTOTL	RKCT1870
	KMVER = LOCNAM(IMVER)	RKCT1880
640	COMINT(KMVER) = COMINT( KMVER+1 )	RKCT1890
	IFIN=1	RKCT1900
	IPT1=IPT2	RKCT1910
	ISCNT=0	RKCT1920
	J=-1	RKCT1930
	IERROR = 1	RKCT1940
	RETURN	RKCT1950
	END	RKCT1960

•

รเ	JBROUTINE OU	ITPUT								OUTP	10
DI	MENSION DIS	P(3),VEL	(3),A	CC	EL(3),	EL(3)	•ELD(3)•E	ELDD	(3),	OUTP	20
¥	AVE	L(9),AAC	CEL(9	),,	AA(3),	BB(3)	,CC(3),DD	)(3),	EE(3),	OUTP	30
¥	AOU	IT(3,10),	NAC(1	.0)						OUTP	40
DI	MENSION GC(	4,5), GC	D(4,5	;),	GCDD(	4,5)				OUTP	50
D	MENSION XF	PS(4,5),	YFPS	6(4	,5), ZI	FPS(4	,5),			ουτρ	60
1	XFP	SD(4,5),	YFPS	SD (	4,5),	ZFPSD	(4,5),			OUTP	70
2	XFPS	DD(4,5),	YFPS	SDD	(4,5),	ZFPS	DD(4,5)			OUTP	80
D	MENSION BOU	T(75)								OUTP	90
C	OMMON / THIS	V/ TIMHS	A(1):	A	TTH(5)	AM(6	,6),AME1(	(3),1	INDFPI(6),	OUTP	100
1	INDFPC(6)	•								OUTP	110
2	STRPDS(10	), STRPM	S( 5)	•	IPOCDS	(10),	IPOCMS(	5),	URCDS (10),	OUTP	120
3	URTDS (10	), URCMS	( 5)	9	URTMS	(5),	SETCDS(1	10),	SETTDS(10);	OUTP	130
4	SETCMS( 5	), SETTM	S( 5)	,	INDCDS	(10),	INDTDS(1	10),	INDCMS(10),	OUTP	140
5	INDIMS( 5);	PRFCDS	(10)	P	RETDS	(10),	PRFCMS (	5)	PRFTMS ( 5)	, OUTP	150
6	IPRDS (10	), IPRMS	( 5)	,	FRVDSC	(10),	FRVDST(1	.0),	FRVMSC( 5),	OUTP	160
L	FRVMST( 5	), IPOTD	S(10)	وا	IPOTMS	(5)				OUTP	170
E	QUIVALENCE (	NFTPDS,	NOLE	ΞG	)					OUTP	180
C	OMMON COMINT	(400)								OUTP	190
E	QUIVALENCE (	COMINT(	3	9	Івотм	)				OUTP	200
ΕĊ	JUIVALENCE (	COMINT(	4		XSD	)				OUTP	210
Ē	QUIVALENCE (	COMINT	8	,	XSDD	ý				OUTP	220
E	QUIVALENCE (	COMINT	12		YS	)				OUTP	230
E	QUIVALENCE (	COMINT	16		YSD	ý				OUTP	240
E	QUIVALENCE (	COMINT	20		YSDD	ý				OUTP	250
Ē	QUIVALENCE (	COMINT	24		ZS	j				OUTP	260
E	QUIVALENCE (	COMINT	28		750	, ,				OUTP	270
Ē	QUIVALENCE (	COMINT	32		ZSDD	ý				OUTP	280
F	SUIVALENCE (	COMINT	36		PHI	ń				OUTP	290
F	QUIVALENCE (	COMINTI	40		PHID	ý				OUTP	300
Ē	QUIVALENCE (	COMINT	44		WX	ý				OUTP	310
Ē	QUIVALENCE (	COMINT	48		WXD	,				OUTP	320
E	QUIVALENCE (	COMINT(	52	,	THTA	, ,				OUTP	330
Ē	QUIVALENCE (	COMINT	56		THTAD	, i				OUTP	340
Ē	QUIVALENCE (	COMINT(	60	,	WY	j				OUTP	350
F	QUIVALENCE (	COMINTI	64	).	WYD	, )				OUTP	360
Ē	QUIVALENCE (	COMINT	68	) .	PSI	Ś				OUTP	370
Ē	QUIVALENCE (	COMINT	72	) .	PSID	ì				OUTP	380
Ē	QUIVALENCE (	COMINT	76	),	WZ	5				OUTP	390
E	QUIVALENCE (	COMINT(	80	),	WZD	, ,				OUTP	400
Ē	QUIVALENCE (	COMINT(	84	) .	GC	, i				OUTP	410
Ē	QUIVALENCE (	COMINT	104	),	GCD	) )				OUTP	420
Ē	QUIVALENCE (	COMINT	124	),	GCDD	j				OUTP	430
E	QUIVALENCE (	COMINT	144	),	XFPS	ý				OUTP	440
Ē	QUIVALENCE (	COMINT	164		XEPSD	, j				OUTP	450
E	QUIVALENCE (	COMINT	184	, . , .	XEPSDD	i i				OUTP	460
Ē	QUIVALENCE (	COMINT	204	),	YEPS	ý				OUTP	470
F	QUIVALENCE /	COMINTI	224	).	YEPSD	ý				OUTP	480
F	QUIVALENCE (	COMINT	244	).	YEPSDD	, j				OUTP	490
Ē	QUIVALENCE (	COMINT	264	, ,	ZEPS	j.				OUTP	500
F	QUIVALENCE (	COMINT	284	) .	ZEPSD	, )				OUTP	510
Ē	QUIVALENCE (	COMINT	304	),	ZEPSDD	, j				OUTP	520
F	QUIVALENCE	COMINT	324	).	TIME	, ,				OUTP	530
Ē	QUIVALENCE (	COMINT	325	).	НМАХ	)				OUTP	540
Ē	QUIVALENCE	COMINTI	326	).	HMIN	)				OUTP	550
-						•					- • •

A

EQUIVALENCE ( COMINT( 327 ), EMIN )	OUTP 560
EQUIVALENCE ( COMINT( 328 ), EMAX )	OUTP 570
EQUIVALENCE ( COMINT( 329 ), XSI )	OUTP 580
FOUTVALENCE ( COMINT ( 337 ) HZ )	OUTP 590
FOULVALENCE ( COMINIC 338 ), CUIEPP)	OUTP 600
	OUTP 600
EQUIVALENCE ( COMINI ( 339 ), IP )	001P 610
EQUIVALENCE ( COMINI ( 340 ), IVARH )	OUTP 620
EQUIVALENCE ( COMINT( 341 ), IMTH )	OUTP 630
EQUIVALENCE ( COMINT( 342 ), IPRNT )	OUTP 640
EQUIVALENCE ( COMINT( 343 ), IFIN )	OUTP 650
EQUIVALENCE ( COMINT( 344 ), IAD )	OUTP 660
EQUIVALENCE ( COMINT( 352 ), IND )	OUTP 670
	OUTD 680
EQUIVALENCE ( COMINIC 500 ), JOICH )	0010 000
EGUIVALENCE ( COMINIC 361 ), IP(CN) )	00TP 690
EQUIVALENCE ( COMINT( 364 ), XS )	OUTP 700
COMINT(364-367) USED BY XS	OUTP 710
COMMON	OUTP 720
1 CBMASS, CBIXX , CBIXZ , CBIYY , CBIYZ , CBIZZ , FPMASS, CBIXY,	OUTP 730
2 DC(3.3) . XEP(5), YEP(5), ZEP(5), WNX(5), WNY(5),	OUTP 740
$ = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac{1}{2} = \frac$	OUTP 750
$ = \begin{cases} g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} \\ g_{AA} & g_{AA} & g_{AA} \\ g_{AA} & g_{AA} & g_{AA} \\ g_{AA} & g_{AA} & g_{AA} & g_{AA} \\ g_{AA} & g_{AA} & g_{AA} & g_{AA} \\ g_{AA} & g_{AA} & g_{AA$	0017 750
$\begin{array}{c} 4 \\ \hline \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 1 \\ 0 \\ 0$	0018 780
5 FSTST(5) , FSZST(5) , SULX(5) ,	OUTP 770
6 SOILY(5) , SOILZ(5) , PMSX(5,5) ,	OUTP 780
7 PMSY(5,5) , PMSZ(5,5) , PDSX(10,5) ,	OUTP 790
8 PDSY(10,5) , PDSZ(10,5) , FLXS , FLYS , FLZS ,	OUTP 800
9 TLXL , TLYL , TLZL , SLO , XMSCB(5) ,	OUTP 810
C YMSCB(5) • 7MSCB(5) • XDSCB(10) •	OUTP 820
D = VDSCB(10) $ZDSCB(10)$ $USCB(10)$ $USCB(10)$	OUTP 930
E ECTY ECTY ECTY DUCDY DUCDY DUCDY	OUTP 050
E FSIX , FSIT , FSIZ , PVCBA , FVCBT , FVCBZ ,	0019 840
F PVFPX , PVFPY , PVFPZ , NOLEG , SLOMS(5) ,	OULD 820
G SLODS(10)	OUTP 860
COMMON	OUTP 870
1 PFCMS(5) , PFCDS(5) , PFTDS(5) ,	OUTP 880
2 PFTMS(5) • SRCDS(5) • SRCMS(5) •	OUTP 890
3 SRTDS (5) • SRTMS (5) • COFFDS • COFFMS • GAMDS •	OUTP 900
4 GAMMS + SRUCDS+ SRUCMS+ SRUTDS+ SRUTMS+ SCMXDS+ SCMXMS+	OUTP 910
E STMYDS, CTMYMS, CDC/CS() - CDC/MS(5)	OUTP 020
CONSCIENCES COURSES COURSES FRICKS INFIDE	OUTP 920
6 CDIDS(5) , CDIMS(5) , FRICDS, FRICMS, IREIDS,	OUTP 930
/ IRETMS, SIRVE(10) , SIRKDS(10) , STRKMS(5) ,	OUTP 940
8 LENGTH, CDXS, CDYS, CDZS, NTYPE, RAD(4), SS(4), ATTHCK(3),	OUTP 950
9 ATTPRS(3) , ADIST , SOILP(3) , NMODES,	OUTP 960
A COEF , GAMMA , FRIC , SCMAX , STMAX , CDC(5), CDT(5),	OUTP 970
B SRULT , SRULC , SRT(5), SRC(5), PFT(5), PFC(5), GFLEGS(5),	OUTP 980
CURMSV. CURMSI. INDEXD. INDEXD. INDEXD. INDEXR. INDEXR.	OUTP 990
D INDEZE TIMAX - DRAGST, IED	OUTPIOOD
	00171000
COMMON	00101010
ICMS, CDCON1, SOILENU, SERHO, NOOUT, XOUT(IU), YOUT(IU), ZOUT(IU),	00191020
2MODEIN,POUTX(10,5),POUTY(10,5),POUTZ(10,5),PCGX(5),PCGY(5),	OUTP1030
3PCGZ(5),AEI1,AEI2,FORMS(5),FORDS(10),FORCE,NFORC,SAVMSX(5),	OUTP1040
4SAVMSZ(5),SAVDSX(10),SAVDSY(10),SAVDSZ(10),IQUOUT,GSINZT,	OUTP1050
5GCOSZT,STAB,STABVL,ISTAB,JCKSAB,VELX,VELY,VELZ,SAVMSY(5)	OUTP1060
COMMON	OUTP1070
SMYMSC(5), TMYMSC(5), SMYMST(5), TMYMST(5),	OUTPIORO
I SHARDECIJI HAMBECIJI SHARDECIJI HAMBELIJI	
3 (SLNGMS(S) (SLNGUS(IU)) CUKUSLA INLEG (IPPKK)	00121100
4 IMPACT(5) 9IPRTFP(5)9 KOUNT(5) 9ANGX9 ANGX9 ANGZ	0011110

С

	RADIAN=57.295779513 DUM1=PHI*RADIAN DUM2=WX*RADIAN DUM3=WXD*RADIAN DUM4=THTA*RADIAN DUM5=WY*RADIAN DUM5=WY*RADIAN DUM6=WYD*RADIAN DUM7=PSI*RADIAN DUM8=WZ*RADIAN DUM9=WZD*RADIAN AVEL(1)=0. AVEL(2)=WZ AVEL(3)=-WY AVEL(6)=WX AVEL(7)=WY
	AVEL(7)-WT AVEL(8)=-WX
	AACCEL (1)=0.
	AACCEL(2)=W2D AACCEL(3)=-WYD
	AACCEL(4)=-WZD AACCEL(5)=0•
	AACCEL(6)=WXD AACCEL(7)=WYD
	AACCEL(8) =-WXD
	DISP(1)=XS
	DISP(2)=YS DISP(3)=75
	VEL(1)=XSD
	VEL(2)=750 VEL(3)=ZSD
	ACCEL(1)=XSDD ACCEL(2)=YSDD
	ACCEL(3)=ZSDD
	CENTER OF GRAVITY MOTIONS
	WRITE(6,350)
	IF (NMODES-EQ.0)GO TO 90
	DO 50 1=1,3 EL(I)=0.
50	ELD(I)=0
20	DO 51 I=1,NMODES
	EL(2) = EL(2) + PCGY(I) + GC(1,I)
	EL(3)=EL(3)+PCGZ(I)*GC(1,1) FLD(1)=ELD(1)+PCGX(I)*GCD(1,1)
	ELD(2)=ELD(2)+PCGY(I)*GCD(1,I)
	ELD(3)=ELD(3)+PCGZ(1)*GCD(1,1) ELDD(1)=ELDD(1)+PCGX(1)*GCDD(1,1)
	<pre>ELDD(2)=ELDD(2)+PCGY(I)*GCDD(1+I)</pre>

OUTP1120 OUTP1130 OUTP1140 OUTP1150 OUTP1160 OUTP1170 OUTP1180 OUTP1190 OUTP1200 OUTP1210 OUTP1220 OUTP1230 OUTP1240 OUTP1250 OUTP1260 OUTP1270 OUTP1280 OUTP1290 OUTP1300 OUTP1310 OUTP1320 OUTP1330 OUTP1340 OUTP1350 OUTP1360 OUTP1370 OUTP1380 OUTP1390 OUTP1400 OUTP1410 OUTP1420 OUTP1430 OUTP1440 OUTP1450 OUTP1460 **OUTP1470** OUTP1480 OUTP1490 OUTP1500 OUTP1510 OUTP1520 OUTP1530 OUTP1540 OUTP1550 OUTP1560 OUTP1570 OUTP1580 OUTP1590 OUTP1600 OUTP1610 OUTP1620 OUTP1630 OUTP1640 OUTP1650 OUTP1660 OUTP1670

٨

C C С

	51	ELDD(3)=ELDD(3)+PCGZ(I)*GCDD(1,I) CONTINUE	OUTP1680 OUTP1690
		CALL GMPRD(DC+EL+AA+3+3+1)	OUTP1700
		DO 52 I=1,3	OUTP1710
	52	DISP(I)=DISP(I)+AA(I)	OUTP1720
		CALL GMPRD(AVEL ,EL,AA,3,3,1)	OUTP1730
		DO 53 I=1,3	OUTP1740
	53	BB(I)=ELD(I)+AA(I)	OUTP1750
		CALL GMPRD(DC,BB,CC,3,3,1)	OUTP1760
		DO 54 I=1,3	OUTP1770
	54	VEL(I)=VEL(I)+CC(I)	OUTP1780
		CALL GMPRD(AVEL,AA,BB,3,3,1)	OUTP1790
		CALL GMPRD(AVEL,ELD,AA,3,3,1)	OUTP1800
		CALL GMPRD(AACCEL,EL,CC,3,3,1)	OUTP1810
		DO 55 I=1,3	OUTP1820
	55	$DD(I) = ELDD(I) + CC(I) + 2 \cdot *AA(I) + BB(I)$	OUTP1830
		CALL GMPRD(DC,DD,AA,3,3,1)	OUTP1840
		DO 56 I=1,3	OUTP1850
	56	ACCEL(I)=ACCEL(I)+AA(I)	OUTP1860
	<b>9</b> 0	CONTINUE	OUTP1870
		WRITE(6,1001)DISP(1),VEL(1),ACCEL(1),DUM1,DUM2,DUM3,	OUTP1880
	+	<pre>t DISP(2),VEL(2),ACCEL(2),DUM4,DUM5,DUM6,</pre>	OUTP1890
	+	<pre>f DISP(3),VEL(3),ACCEL(3),DUM7,DUM8,DUM9</pre>	OUTP1900
		WRITE(6,1022)	OUTP1910
		DO 100 I=1,NFTPDS	OUTP1920
		IF(INDFPC(I) + LE + 0)GO TO 100	OUTP1930
		WRITE(6,1003)I,	OUTP1940
	;	<pre>* XFPS(1,I),XFPSD(1,I),XFPSDD(1,I),</pre>	OUTP1950
	+	<pre></pre>	OUTP1960
	•	<pre># ZFPS(1,I),ZFPSD(1,I),ZFPSDD(1,I)</pre>	OUTP1970
	100	CONTINUE	OUTP1980
C			OUTP1990
С		ACCELERATION AND INTEGRATED QUANTITIES	OUTP2000
С			OUTP2010
	_	DO 60 I=1,3	OUTP2020
	60	- * * / T \ = / DC / 1 + T \ ¥ * C C EL / 1 \ + DC / 2 + T \ ¥ * C C EL / 2 \ + DC / 2 + T \ ¥ * C C EL / 2 \ \ / C P * V E	
			OUTP2030
		IF (NOOUT • EQ.0) GO TO 202	OUTP2030 OUTP2040
		IF(NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT	OUTP2030 OUTP2040 OUTP2050
		IF(NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3	OUTP2030 OUTP2040 OUTP2050 OUTP2060
		IF (NOOUT • EQ • 0) GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0.	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070
	102	IF (NOOUT • EQ • 0) GO TO 202 DO 200 J=1•NOOUT DO 102 I=1•3 ELD(I)=0• ELDD(I)=0•	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080
	102	IF (NOOUT • EQ•0)GO TO 202 DO 200 J=1•NOOUT DO 102 I=1•3 ELD(I)=0• EL(1)=XOUT(J) EL(1)=XOUT(J)	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090
	102	IF(NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J)	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100
	102	IF (NOOUT • EQ • 0) GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) EL(3)=ZOUT(J)	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2110
	102	IF (NOOUT • EQ • 0) GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) EL(3)=ZOUT(J) IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 ELD(I)=0. EL(2)=YOUT(J) EL(3)=ZOUT(J) IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 DO 102 I=1,3 IF (NMODES • EQ • 0) GO TO 111 IF (NMODES • 0) GO TO 111 IF (NMODES • 0) GO TO 10 IF (NMODES • 0) GO TO 10	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2100 OUTP2120
	102	IF (NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) EL(3)=ZOUT(J) IF (NMODES.EQ.0)GO TO 111 DO 110 I=1,NMODES EL(1)=F(1)=F(1)=F(1)=F(1)=F(1)=F(1)=F(1)=F	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2100 OUTP2120 OUTP2120 OUTP2130
	102	IF (NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. ELDD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) EL(3)=ZOUT(J) IF (NMODES.EQ.0)GO TO 111 DO 110 I=1,NMODES EL(1)=EL(1)+POUTX(J,I)*GC(1,I) EL(2)=FOUTY(J,I)*GC(1,I)	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2110 OUTP2120 OUTP2130 OUTP2140
	102	<pre>IF(NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) EL(3)=ZOUT(J) IF(NMODES.EQ.0)GO TO 111 DO 110 I=1,NMODES EL(1)=EL(1)+POUTX(J,I)*GC(1,I) EL(2)=EL(2)+POUTY(J,I)*GC(1,I)</pre>	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2110 OUTP2120 OUTP2130 OUTP2130 OUTP2150 OUTP2150
	102	<pre>IF(NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. ELDD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) EL(3)=ZOUT(J) IF(NMODES.EQ.0)GO TO 111 DO 110 I=1,NMODES EL(1)=EL(1)+POUTX(J,I)*GC(1,I) EL(2)=EL(2)+POUTY(J,I)*GC(1,I) EL(3)=EL(3)+POUTZ(J,I)*GC(1,I)</pre>	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2100 OUTP2120 OUTP2130 OUTP2140 OUTP2150 OUTP2160
	102	<pre>IF(NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. ELDD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) EL(3)=ZOUT(J) IF(NMODES.EQ.0)GO TO 111 DO 110 I=1,NMODES EL(1)=EL(1)+POUTX(J,I)*GC(1,I) EL(2)=EL(2)+POUTY(J,I)*GC(1,I) EL(3)=EL(3)+POUTZ(J,I)*GC(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I)</pre>	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2100 OUTP2120 OUTP2130 OUTP2130 OUTP2150 OUTP2160 OUTP2170 OUTP2180
	102	<pre>IF (NOOUT • EQ • 0) GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. ELDD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) IF (NMODES • EQ • 0) GO TO 111 DO 110 I=1,NMODES EL(1)=EL(1)+POUTX(J,I)*GC(1,I) EL(2)=EL(2)+POUTY(J,I)*GC(1,I) EL(3)=EL(3)+POUTZ(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(2)=ELD(2)+POUTY(J,I)*GCD(1,I) ELD(2)=ELD(2)+POUTY(J,I)*GCD(1,I) ELD(2)=ELD(2)+POUTY(J,I)*GCD(1,I)</pre>	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2100 OUTP2120 OUTP2130 OUTP2140 OUTP2150 OUTP2160 OUTP2180 OUTP2180
	102	<pre>IF(NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. ELDD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) IF(NMODES.EQ.0)GO TO 111 DO 110 I=1,NMODES EL(1)=EL(1)+POUTX(J,I)*GC(1,I) EL(2)=EL(2)+POUTY(J,I)*GC(1,I) EL(3)=EL(3)+POUTZ(J,I)*GC(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GC(1,I) ELD(2)=ELD(2)+POUTY(J,I)*GC(1,I) ELD(3)=ELD(3)+POUTZ(J,I)*GCD(1,I) ELD(3)=ELD(3)+POUTZ(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(3)=ELD(3)+POUTZ(J,I)*GCD(1,I)</pre>	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2100 OUTP2120 OUTP2130 OUTP2130 OUTP2150 OUTP2160 OUTP2170 OUTP2180 OUTP2190 OUTP2190
	102	<pre>IF(NOOUT.EQ.0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) IF(NMODES.EQ.0)GO TO 111 DO 110 I=1,NMODES EL(1)=EL(1)+POUTX(J,I)*GC(1,I) EL(2)=EL(2)+POUTY(J,I)*GC(1,I) EL(3)=EL(3)+POUTZ(J,I)*GC(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GC(1,I) ELD(2)=ELD(2)+POUTY(J,I)*GC(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GC(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I)</pre>	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2100 OUTP2120 OUTP2130 OUTP2140 OUTP2150 OUTP2150 OUTP2170 OUTP2180 OUTP2190 OUTP2200 OUTP2200
	102	<pre>IF (NOUT • EQ•0)GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) EL(3)=ZOUT(J) IF (NMODES • EQ•0)GO TO 111 DO 110 I=1,NMODES EL(1)=EL(1)+POUTX(J,I)*GC(1,I) EL(2)=EL(2)+POUTY(J,I)*GC(1,I) EL(3)=EL(3)+POUTZ(J,I)*GC(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(2)=ELD(2)+POUTY(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(2)=ELDD(2)+POUTY(J,I)*GCDD(1,I) ELDD(2)=ELDD(2)+POUTY(J,I)*GCDD(1,I) ELDD(3)=EIDD(3)+POUTZ(J,I)*GCDD(1,I)</pre>	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2100 OUTP2120 OUTP2120 OUTP2130 OUTP2140 OUTP2150 OUTP2160 OUTP2170 OUTP2180 OUTP2190 OUTP2200 OUTP2210
	102	<pre>IF (NOOUT • EQ • 0) GO TO 202 DO 200 J=1,NOOUT DO 102 I=1,3 ELD(I)=0. EL(1)=XOUT(J) EL(2)=YOUT(J) EL(3)=ZOUT(J) IF (NMODES.eQ.0)GO TO 111 DO 110 I=1,NMODES EL(1)=EL(1)+POUTX(J,I)*GC(1,I) EL(2)=EL(2)+POUTY(J,I)*GC(1,I) EL(3)=EL(3)+POUTZ(J,I)*GC(1,I) ELD(1)=ELD(1)+POUTX(J,I)*GCD(1,I) ELD(2)=ELD(2)+POUTY(J,I)*GCD(1,I) ELD(1)=ELDD(1)+POUTX(J,I)*GCD(1,I) ELDD(1)=ELDD(1)+POUTX(J,I)*GCDD(1,I) ELDD(2)=ELDD(2)+POUTY(J,I)*GCDD(1,I) ELDD(3)=ELDD(3)+POUTZ(J,I)*GCDD(1,I) ELDD(3)=ELDD(3)+POUTZ(J,I)*GCDD(1,I) ELDD(3)=ELDD(3)+POUTZ(J,I)*GCDD(1,I) ELDD(3)=ELDD(3)+POUTZ(J,I)*GCDD(1,I) ELDD(3)=ELDD(3)+POUTZ(J,I)*GCDD(1,I) ELDD(3)=ELDD(3)+POUTZ(J,I)*GCDD(1,I)</pre>	OUTP2030 OUTP2040 OUTP2050 OUTP2060 OUTP2070 OUTP2080 OUTP2090 OUTP2100 OUTP2100 OUTP2120 OUTP2130 OUTP2140 OUTP2140 OUTP2150 OUTP2160 OUTP2170 OUTP2180 OUTP2190 OUTP2200 OUTP2210 OUTP2230

	111	CONTINUE CALL GMPRD(AVEL,EL,BB,3,3,1)	OUTP2240 OUTP2250
		CALL GMPRD(AVEL,BB,CC,3,3,1)	OUTP2260
		CALL GMPRD(AACCEL,EL,BB,3,3,1)	OUTP2270
		CALL GMPRD(AVEL,ELD,DD,3,3,1)	OUTP2280
		DO 120 I=1,3	OUTP2290
	120	EE(I)=ELDD(I)+BB(I)+2.*DD(I)+CC(I)	0072300
		AOUT(1,J)=(DC(1,1)*XSDD+DC(2,1)*YSDD+DC(3,1)*ZSDD+EE(1))/GRAVE	OUTP2310
		AOUT(2,J)=(DC(1,2)*XSDD+DC(2,2)*YSDD+DC(3,2)*ZSDD+EE(2))/GRAVE	OUTP2320
		AOUT(3,J)=(DC(1,3)*XSDD+DC(2,3)*YSDD+DC(3,3)*ZSDD+EE(3))/GRAVE	001P2330
	200	CONTINUE	001P2340
	202	CONTINUE	001P2350
		IF (NMODES • EQ • 0) GO TO 220	0012360
			00172370
~		DRINT C C ACCELERATION AND INTEGRATED ONANTITIES	00172300
		WRINE COO ACCELERATION AND INTEGRATED GOANTITIES	OUTP2400
		$WRITE(6,1007)\Delta\Delta(1),\Delta\Delta(2),\Delta\Delta(3),XS,XSD,XSDD$	OUTP2410
		WRITE(6,1008)YS,YSD,YSDD	OUTP2420
		WRITE(6,1009)ZS+ZSD+ZSDD	OUTP2430
		DO 201 I=1.NMODES	0UTP2440
	201	WRITE(6,1010)I,GC(1,1),GCD(1,1),GCDD(1,1)	OUTP2450
		GO TO 300	OUTP2460
С		PRINT C.G. ACCELERATION, SECONDARY POINTS, AND INTEGRATED QUANT.	OUTP2470
	210	CONTINUE	OUTP2480
		DO 95 I=1,10	0UTP2490
	95	NAC(I)=I	OUTP2500
-		WRITE(6,1005)	OUTP2505
		WRITE(6,1007)AA(1),AA(2),AA(3),XS,XSD,XSDD	OUTP2510
		WRITE(6,1011)NAC(1),AOUT(1,1),AOUT(2,1),AOUT(3,1),YS,YSD,YSDD	OUTP2520
		WRITE(6,1012)NAC(2),AOUT(1,2),AOUT(2,2),AOUT(3,2),ZS,ZSD,ZSDD	OUTP2530
		DO 211 I=1,NMODES	OUTP2540
			00192550
	211	WRITE(6,1013)NAC(11),AOUT(1,11),AOUT(2,11),AOUT(3,11),1,GC(1,1),	001P2560
	,		0012270
		IF (II • GE • NUUUI) GU 10 300	00172500
			00122550
	-1-	$\frac{1}{2} = 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1$	OUTP2610
	212	WRIE(b)IU(4)NAC(1)4AC(1)4AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)44AC(1)444AC(1)444AC(1)444AC(1)444AC(1)444AC(1)444AC(1)444AC(1)444AC(1)444AC(1)444AC(1)444AC(1)444AC(1)4444AC(1)4444AC(1)4444444444	OUTP2620
c		DDINT CAG. ACCELERATIONS AND SECONDARY POINTS	OUTP2630
C	220	CONTINUE	OUTP2640
	220	WRITE(6.1004)	0UTP2650
		$WRITE(6 \cdot 1006) AA(1) \cdot AA(2) \cdot AA(3)$	0UTP2660
		IF (NOOUT • EQ • 0) GO TO 300	OUTP2670
		$DO 96 I = 1 \cdot 10$	0UTP2675
	96	NAC(I)=I	OUTP2676
		DO 221 I=1,NOUT	0UTP2680
	221	WRITE(6,1014)NAC(I),AOUT(1,I),AOUT(2,I),AOUT(3,I)	0UTP2690
С			OUTP2700
С		STRUT INFORMATION	OUTP2710
С			OUTP2720
	300	CONTINUE	OUTP2730
		WRITE(6,1020)	0012740
		DO 205 1=1.NFIPDS	0012/50
		11=2*(1-1)+1	00172760
A =	0	8	

•

		II2=II+1	OUTP2770
		CRUSH=SS(1)-ATTH(I)	OUTP2780
	205	WRITE(6,1021)I, CRUSH , FORMS(I), STRKMS(I), II, FORDS(II), STRKDS(II)	»OUTP2790
	*	LI2 FORDS(II2) STRKDS(II2)	OUTP2800
c		112, 0, 0, 0, 112, , 0, 1, 0, 0, 112,	OUTP2810
ř		CHECK LANDER STABLITY	OUTP2820
È		CHECK EADER STADIETT	OUTP2830
C			OUTP2840
c			OUTP2850
C			0UTP2860
~		CALL STABLE	00112000
C			001-2010
		IF (ISTAB • NE • 0) GO TO 400	001F2800
		WRITE (6,1015) STAB, STABVE	0012890
	400	CONTINUE	00122900
		IF(NFORC.eq.0) RETURN	00122910
С			001P2920
С		WRITE TIME HISTORY QUANTITIES ON TAPE FOR USE WITH LANDING	OUTP2930
С		LOADS PROGRAM	OUTP2940
С		1. NFORC=1 RESULTS IN TIME HISTORY TAPE BEING GENERATED	OUTP2950
С		2. TIME HISTORY INFORMATION WRITTEN ON TAPE3	OUTP2960
С			OUTP2970
		BOUT(1)=TIME	0UTP2980
		BOUT(2)=HMAX	OUTP2990
		BOUT(3)=XSDD	OUTP3000
		BOUT(4)=YSDD	OUTP3010
		BOUT (5)=7SDD	OUTP3020
			OUTP3030
			OUTP3040
			OUTP3050
			OUTP3060
			OUTP3070
			OUTP3080
		BOUT(11)=w2	OUT P3090
		BOUT(12)=WXD	00179090
		BOUT(13)=WYD	001F3100
		BOUT(14)=W2D	00179110
		NSP=14	00193120
		DO 450 I=1,NMODES	001P3130
		NNL=NSP+3*I	001P3140
		BOUT(NNL-2)=GC(1,I)	OUTP3150
		BOUT(NNL-1)=GCD(1,I)	OUTP3160
	450	BOUT(NNL)=GCDD(1,I)	OUTP3170
		NSP=14+3*NMODES	OUTP3180
		DO 451 I=1,NOLEG	OUTP3190
		NNL=NSP+3*I	OUTP3200
		BOUT(NNL-2) = SAVMSX(I)	OUTP3210
		BOUT(NNL-1) = SAVMSY(I)	OUTP3220
	451	BOUT(NNL)=SAVMSZ(I)	OUTP3230
		II=2*NOLEG	OUTP3240
		NSP=NSP+3*NOLEG	0UTP3250
		$DO 452 I=1 \times I I$	OUTP3260
			0UTP3270
			OUTP3280
		$\frac{DOUT(NNL-2) - SAVDSA(1)}{DOUT(NNL-1) - SAVDSA(1)}$	OUTP3290
			OUTP3200
	452	BUUI(NNL)=SAVDS2(1)	00119900
			00153310
		WRITE(3)(BOUT(I))I=I)NSP)	00189920

OUTP3330 RETURN OUTP3340 С OUTP3350 ENTRY SUMMRY OUTP3360 С OUTP3370 С THIS PORTION OF THE SUBROUTINE PRINTS THE SUMMARY INFORMATION С OUTP3380 OUTP3390 FILE 3 IF (NFORC • NE • 0) END OUTP3400 С OUTP3410 WRITE(6,2000) OUTP3420 WRITE(6,2006) WRITE(6,2001)(I,TMXMSC(I),SMXMSC(I),TMXMST(I),SMXMST(I),I=1,NOLEG)OUTP3430 OUTP3440 WRITE(6,2007) OUTP3450 NN=2*NOLEG OUTP3460 WRITE(6,2001)(I,TMXDSC(I),SMXDSC(I),TMXDST(I),SMXDST(I),I=1,NN) OUTP3470 IF(IBOTM.NE.0)GO TO 1900 OUTP3480 IF(ISTAB.EQ.0)WRITE(6,2003) OUTP3490 IF(ISTAB.EQ.1)WRITE(6,2004) IF(ISTAB.EQ.2)WRITE(6,2005) OUTP3500 OUTP3510 RETURN OUTP3520 1900 CONTINUE IF(IFP.GT.10)GO TO 1950 OUTP3530 OUTP3540 IF(IBOTM.EQ.+1)GO TO 6000 OUTP3550 WRITE(6,6201)IFP RETURN OUTP3560 OUTP3570 6000 WRITE(6,6200) IFP RETURN OUTP3580 OUTP3590 1950 CONTINUE OUTP3600 IEP=IEP/100 IF(IBOTM.EQ.+1)GO TO 6100 OUTP3610 OUTP3620 WRITE(6,6203) IFP OUTP3630 RETURN 6100 WRITE(6,6202) IFP OUTP3640 OUTP3650 RETURN OUTP3660 С С FORMAT STATEMENTS OUTP3670 OUTP3680 С OUTP3690 350 FORMAT(/) 1000 FORMAT(/, 90H0* * * * * * * * * * * * * * * * * * * * * * * * * * OUTP3700 //15X,7HTIME = ,E13.6, OUTP3710 OUTP3720 *14H STEP SIZE = ,E13.6) 1001 FORMAT(//3X,19HCENTER BODY MOTIONS//6X,39HTRANSLATION - SURFACE COOUTP3730 *ORDINATE SYSTEM, 10X, 8HROTATION/91X, 25HVELOCITY AND ACCELERATION/ OUTP3740 OUTP3750 *16X,4HDISP,8X,3HVEL,7X,5HACCEL,48X,24HLANDER COORDINATE SYSTEM/ *10X,2HX ,3(1X,E10.3),14X,20HEULER ANGLES PHI OUTP3760 ,E10.3,3X,2HX , OUTP3770 *2(1X,E1C.3)/10X,2HY ,3(1X,E10.3),28X,6HTHETA ,E10.3,3X,2HY , OUTP3780 *2(1X,E10.3)/10X,2HZ ,3(1X,E10.3),28X,6HPSI •E10•3•3X•2HZ • *2(1X,E10.3)) OUTP3790 OUTP3800 1003 FORMAT(6X,7HFOOTPAD,13,13X,1X,3(1X,E10.3),1X, OUTP3810 *3(1X,E10.3),1X,3(1X,E10.3)) 1004 FORMAT(/3X,53HACCELERATIONS IN EARTH G*S - LANDER COORDINATE SYSTEOUTP3820 OUTP3830 *M/ *30X,1HX,10X,1HY,10X,1HZ) OUTP3840 1005 FORMAT(/3X,53HACCELERATIONS IN EARTH G*S - LANDER COORDINATE SYSTEOUTP3850 OUTP3860 *M,17X,33HCENTER BODY INTEGRATED QUANTITIES/ OUTP3870 *30X,1HX,10X,1HY,10X,1HZ,49X,4HDISP,8X,3HVEL,7X,5HACCEL) 1006 FORMAT(6X,18HCENTER OF GRAVITY ,3(1X,E10.3)) OUTP3880

1007 FORMAT(6X,18HCENTER OF GRAVITY ,3(1X,E10.3) ,26X,13HCENTER BODY	X,OUTP3890
*2X,3(1X,E10.3))	OUTP3900
1008 FORMAT(83X,13HCENTER BODY Y,2X,3(1X,E10.3))	OUTP3910
1009 FORMAT(83X,13HCENTER BODY Z,2X,3(1X,E10.3))	OUTP3920
1010 FORMAT(83X,4HMODE,I4,7X,3(1X,E10.3))	OUTP3930
1011 FORMAT(6X,6HPOINT ,I2,10X,3(1X,E10.3),26X,13HCENTER BODY Y,2X,	OUTP3940
*3(1X,E10.3))	OUTP3950
1012 FORMAT(6X,6HPOINT ,I2,10X,3(1X,E10.3),26X,13HCENTER BODY Z,2X,	OUTP3960
*3(1X,E10.3))	OUTP3970
1013 FORMAT(6X,6HPOINT ,12,10X,3(1X,E10.3),26X,4HMODE,I4,7X,3(1X,	OUTP3980
*E10.3))	OUTP3990
1014 FORMAT(6X,6HPOINT ,12,10X,3(1X,E10.3))	OUTP4000
1015 FORMAT(/3X,16HLANDER STABILITY/6X,20HSTABILITY ANGLE = ,	OUTP4010
*E10.3/6X,20HPITCHING VELOCITY = ,E10.3)	OUTP4020
1020 FORMAT(/3X,30HLANDING GEAR STRUT INFORMATION/6X,3HLEG,6X,	OUTP4030
*5HCRUSH918X94HLOAD9	OUTP4040
*6X,6HSTROKE,20X,4HLOAD,6X,6HSTROKE,20X,4HLOAD,6X,6HSTROKE)	OUTP4050
1021 FORMAT(7X, I1, 4X, E10.3, 2X, 10HMAIN STRUT, 2(1X, E10.3), 12H DRAG STF	RUTOUTP4060
*,12,	OUTP4070
*2(1X,E10.3),12H DRAG STRUT,12,2(1X,E10.3))	OUTP4080
1922 FORMAT(/3X,43HFOOTPAD MOTIONS - SURFACE COORDINATE SYSTEM	0UTP4090
*/47X,1HX,33X,1HY,	OUTP4100
*33X,1HZ/34X,4HDISP,8X,3HVEL,7X,5HACCEL,7X,4HDISP,8X,3HVEL,7X,	OUTP4110
*5HACCEL,7X,4HDISP,8X,3HVEL,7X,5HACCEL)	OUTP4120
2000 FORMAT(//3X,19HSUMMARY INFORMATION/)	OUTP4130
2001 FORMAT(7X, I2, 2X, 22(2X, E10, 3), 2X, 2(2X, E10, 3))	OUTP4140
2003 FORMAT(/,6X,24HEND OF CASE - TIME LIMIT)	OUTP4150
2004 FORMAT(/,6X,31HEND OF CASE - PITCH INSTABILITY)	OUTP4160
2005 FORMAT(/,6X,29HEND OF CASE - YAW INSTABILITY)	OUTP4170
2006 FORMAT(3X,26HMAXIMUM MAIN STRUT STROKES,/	OUTP4180
*19X,11HCOMPRESSION,17X,7HTENSION/6X,5HSTRUT,5X,4HTIME,7X,	OUTP4190
*6HSTROKE,9X,4HTIME,7X,6HSTROKE)	OUTP4200
2007 FORMAT(3X,26HMAXIMUM DRAG STRUT STROKES)/	OUTP4210
*19X,11HCOMPRESSION,17X,7HTENSION/6X,5HSTRUT,5X,4HTIME,7X,	OUTP4220
*6HSTROKE,9X,4HTIME,7X,6HSTROKE)	OUTP4230
6200 FORMAT(//10X+35HMAIN STRUT BOTTOMED ON TENSION SIDE+/	0UTP4240
* 30X •8HSTRUT = •12)	OUTP4250
6201 FORMAT(//10X .39HMAIN STRUT BOTTOMED ON COMPRESSION SIDE ./	OUTP4260
*	OUTP4270
6202 FORMAT(//10X+35HDRAG STRUT BOTTOMED ON TENSION SIDE+/	OUTP4280
* 30X * 8HSTRUT = (12)	OUTP4290
6203 FORMAT (//IOX-39HDRAG STRUT BOTTOMED ON COMPRESSION SIDE /	OUTP4300
* 30X 8HSTRUT = .12)	0UTP4310
END	OUTP4320
	0000 4760

SUBROUTINE STABLE	STAB	10
	STAB	20
THIS ROUTINE CHECKS THE STABILITY OF THE LANDER	STAB	30
THE FOLLOWING FLAG IS RETURNED	STAB	40
1. ISTAD=U - LANDER STADLE	STAD	50
2. ISTABEL - PITCH INSTABILITY	STAB	70
J. ISTAD-2 - TAW INSTADILITY	STAB	80
DIMENSION DX(5), $DY(5)$ , $DZ(5)$ , $SGN(5)$ , $ANG(5)$	STAB	90
	STAB	100
DIMENSION GC(4,5), GCD(4,5), GCDD(4,5)	STAB	110
	STAB	120
DIMENSION XFPS(4,5), YFPS(4,5), ZFPS(4,5),	STAB	130
1 XFPSD(4,5), YFPSD(4,5), ZFPSD(4,5),	STAB	140
2 XFPSDD(4,5), YFPSDD(4,5), ZFPSDD(4,5)	STAB	150
	STAB	160
COMMON / THISV/ TIMHSA(1), ATTH(5),AM(6,6),AME1(3),INDFPI(6),	STAB	170
1 INDFPC(6),	STAB	180
2 STRPDS(10), STRPMS( 5), IPOCDS(10), IPOCMS( 5), URCDS (10),	STAB	190
3 URTDS (10), URCMS ( 5), URTMS ( 5), SETCDS(10), SETTDS(10),	STAB	200
4 SETCMS(5), SETIMS(5), INDCDS(10), INDIDS(10), INDCMS(10),	STAD	210
5 INDIMS( 5), PRECOS (IO), PRETOS (IO), PRECMS ( 5), PRETMS ( 5),	STAD	220
6 IPRES (10), IPRMS (5), PRUDSCID), PRUDSCIDI, PRUDSCIDI,	CTAR	200
E PROMINE () ( POTOS (10) ( POTOS ( ))	STAD	250
COMMON COMINT(400)	STAB	260
	STAR	270
EQUIVALENCE ( COMINT( 8), XSDD )	STAB	280
FQUIVALENCE ( COMINT( 12 ), YS )	STAB	290
EQUIVALENCE ( COMINT( 16 ), YSD )	STAB	300
EQUIVALENCE ( COMINT( 20 ), YSDD )	STAB	310
EQUIVALENCE ( COMINT( 24 ), ZS )	STAB	320
EQUIVALENCE ( COMINT( 28 ), ZSD )	STAB	330
EQUIVALENCE ( COMINT( 32 ), ZSDD )	STAB	340
EQUIVALENCE ( COMINT( 36 ), PHI )	STAB	350
EQUIVALENCE ( COMINT( 40 ), PHID )	STAB	360
EQUIVALENCE ( COMINT( 44 ), WX )	STAB	370
EQUIVALENCE ( COMINT( 48 ), WXD )	STAB	380
EQUIVALENCE ( COMINI( 52 )) THIA )	STAD	590
EQUIVALENCE ( COMINIT 50 ), HIAD )	STAR	410
EQUIVALENCE ( COMINT( 50 ), WT )	STAR	420
FOULVALENCE ( COMINIT 64 ), WID )	STAB	430
FOULVALENCE ( COMINT( 72 ), PSID )	STAB	440
EQUIVALENCE ( COMINT( 76 ), WZ )	STAB	450
EQUIVALENCE ( COMINT( 80 ), WZD )	STAB	460
EQUIVALENCE ( COMINT( 84 ), GC )	STAB	470
EQUIVALENCE ( COMINT(104 ), GCD )	STAB	480
EQUIVALENCE ( COMINT(124 ), GCDD )	STAB	490
EQUIVALENCE ( COMINT(144 ), XFPS )	STAB	500
EQUIVALENCE ( COMINT(164 ), XFPSD )	STAB	510
EQUIVALENCE ( COMINT(184 ), XFPSDD )	STAB	520
EQUIVALENCE ( COMINT(204 ), YFPS )	SIAB	530
EQUIVALENCE ( COMINT(264 ), ZEPS )	STAB	540
EQUIVALENCE ( CUMINI(224 ); YEPSD )	STAB	220

С

С

<b>F 0</b> 1 1		
EQUI	VALENCE ( COMINI(244 ), YFPSDD )	STAB 560
EQUI	VALENCE ( COMINT(284 ), ZFPSD )	STAB 570
EQUI	VALENCE ( COMINT(304 ), ZFPSDD )	STAB 580
EQUI	VALENCE ( COMINT( 324 ), TIME )	STAB 590
EQUI	VALENCE ( COMINT( 325 ), HMAX )	STAB 600
EQUI	VALENCE ( COMINT( 326 ) HMIN )	STAB 610
FQUI	VALENCE ( COMINT( 327 ) EMIN )	STAB 620
FOUT	VALENCE ( COMINT( 328 ), EMAX )	STAB 430
FOUT	VALENCE ( COMINT( 320 ), VCL )	STAD 000
5001	VALENCE ( COMINIT / 227 / X31 )	STAD 640
E001	VALENCE (COMINI (222)) = CUTEDDA	STAB 650
	VALENCE ( COMINT( 330 ), CULERN)	STAD 660
EQUI	VALENCE ( COMINT( 559 / ) IF )	STAD 670
EQUI	VALENCE ( COMINIC 540 ) IVARD )	STAD SOU
	VALENCE (COMINIC 241 ) IMPRID	STAD 690
EQUI	VALENCE (COMINT(342 )) IFIN	STAB 700
EQUI	IVALENCE ( COMINI( 343 ), IFIN )	STAB 710
EQUI	VALENCE ( COMINI( 344 ), IAD )	STAB 720
EQUI	VALENCE (COMINIC 352), IND )	STAB 730
EQUI	VALENCE ( COMINT( 360 ), JCUT )	STAB 740
EQUI	IVALENCE ( COMINT( 361 ), IPTCNT )	STAB 750
EQUI	VALENCE ( COMINT ( 362 ), IDSETN )	STAB 760
EQUI	IVALENCE ( COMINT( 363 ), IVAL )	STAB 770
EQUI	IVALENCE ( COMINT( 364 ), XS )	STAB 780
	COMINT(364-367) USED BY XS	STAB 790
COMM	10N	STAB 800
1	CBMASS, CBIXX , CBIXZ , CBIYY , CBIYZ , CBIZZ , FPMASS, CBIXY,	STAB 810
2	DC(3,3) , XFP(5), YFP(5), ZFP(5), WNX(5), WNY(5),	STAB 820
3	WNZ(5), PX(5), PY(5), PZ(5), GM(5), OMEGA(5),	STAB 830
4	GRAV , GRAVE , ZETA , FTS (6) , FSXSI(5) ,	STAB 840
5	FSYSI(5) , FSZSI(5) , SOILX(5) ,	STAB 850
6	SOILY(5) , SOILZ(5) , PMSX(5,5) ,	STAB 860
7	PMSY(5,5) , PMSZ(5,5) , PDSX(10,5) ,	STAB 870
8	PDSY(10,5) , PDSZ(10,5) , FLXS , FLYS , FLZS ,	STAB 880
9	TLXL , TLYL , TLZL , SLO , XMSCB(5) ,	STAB 89 <b>0</b>
С	YMSCB(5) , ZMSCB(5) , XDSCB(10) ,	STAB 900
D	YDSCB(10) , ZDSCB(10) , ILEG , IMS ,	STAB 910
Ε	FSTX , FSTY , FSTZ , PVCBX , PVCBY , PVCBZ ,	STAB 920
F	PVFPX , PVFPY , PVFPZ , NOLEG , SLOMS(5) ,	STAB 930
G	SLODS(10)	STAB 940
COM	MON	STAB 950
1	PFCMS(5) , PFCDS(5) , PFTDS(5) ,	STAB 960
2	PETMS (5) SRCDS (5) SRCMS (5)	STAB 970
2	SRIDS(5) SRIMS(5) COEEMS, GAMDS	STAB 980
5	GAMMS - SRUCDS - SRUCMS - SRUTDS - SRUTMS - SCMXDS - SCMXMS -	STAB 990
5	STMXDS. STMXMS. CDCDS(5) CDCMS(5)	STAB1000
6	(DTDS(5) - CDTMS(5) - EPICDS - EPICMS - IRETDS	STABLOLO
7	IPETMS, STROKE(10) , STREDS(10) , STREMS( 5)	STABL020
γ Q	LENGTH CONSTITUTES ANTYDE - DADIAL STANDAU - ATTHCK/21.	STAB1020
0	ATTORS(2) ADJET COLLO 7 NITEL 9 NAU(4/) SU(4/) 7 ATTORS(2/)	STABLOOD
7	COFE CAMMA FRIC SCMAY STMAY COCISIS COTISIS	STABLORD
A	CULE , GAMMA , ENIC , SCHAR , STMAR , CULLI, CULLI), CDUET , CDUEC , CRTIEN, CRCIEN, DETIEN, DECISI, GELEGGIEN,	STABLODU
D C	CHEMICAL FUNCTION STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATES STATE	STABLOOU
	LUNESTS CURESLS INDERES INDEEDS INDEEDS INDERES INDEEDS	STABLORO
υ	INNERVA ITMAN A NEARONA IER	STABL000
c ••••	MON	STABLUSU
		STADILUU
TCWR	ACTION A 2 CRUCA NOT A NOT A 201 CT A 2	STADILIO

С

С

475

2MODEIN, POUTX(10,5), POUTY(10,5), POUTZ(10,5), PCGX(5), PCGY(5), 3PCGZ(5), AEI1, AEI2, FORMS(5), FORDS(10), FORCE, NFORC, SAVMSX(5), STAB1120 STAB1130 4SAVMSZ(5), SAVDSX(10), SAVDSY(10), SAVDSZ(10), IQUOUT, GSINZT, STAB1140 STAB1150 5GCOSZT, STAB, STABVL, ISTAB, JCKSAB, VELX, VELY, VELZ, SAVMSY(5) STAB1160 COMMON SMXMSC(5), TMXMSC(5), SMXMST(5), TMXMST(5), STAB1170 1 SMXDSC(10), TMXDSC(10), SMXDST(10), TMXDST(10) STAB1180 2 ,SLNGMS(5) ,SLNGDS(10), CURDSL, INLEG , IFPPRT, STAB1190 3 ANGZ STAB1200 IMPACT(5) ,IPRTFP(5), KOUNT(5) ,ANGX, ANGY, 4 STAB1210 STAB1220 SUBROUTINE INITIALIZATION STAB1230 STAB1240 DO 50 I=1,NOLEG STAB1250 50 ANG(I)=2000. ANX=GSINZT*YSD STAB1260 STAB1270 ANY=-(GCOSZT*ZSD+GSINZT*XSD) ANZ=GCOSZT*YSD STAB1280 STAB1290 AMAG=SQRT(ANX**2+ANY**2+ANZ**2) STAB1300 ANX=ANX/AMAG STAB1310 ANY=ANY/AMAG STAB1320 ANZ=ANZ/AMAG STAB1330 C C DETERMINE FOOTPAD POSITIONS RELEATIVE TO PLANE OF MOTION STAB1340 STAB1350 С DO 100 I=1.NOLEG STAB1360 STAB1370 DX(I) = XFPS(1,I) - XSSTAB1380 DY(I) = YFPS(1,I) - YSDZ(I) = ZFPS(1,I) - ZSSTAB1390 DDOTN=ANX*DX(I)+ANY*DY(I)+ANZ*DZ(I) STAB1400 SGN(I)=SIGN(1.,DDOTN) STAB1410 STAB1420 IF(ABS(DDOTN) • LE • 1 • E-05)SGN(I) = 0 • STAB1430 100 CONTINUE STAB1440 С STAB1450 С DETERMINE STABILITY ANGLE С STAB1460 STAE1470 DO 200 I=1,NOLEG IF(SGN(I).NE.0.)GO TO 110 STAB1480 STAB1490 XL = DX(I)STAB1500 YL = DY(I)STAB1510 ZL=DZ(I)GO TO 120 STAB1520 STAB1530 110 I1=I+1 STAB1540 IF(I.EQ.NOLEG)I1=1 ASGN=SGN(I)*SGN(I1) STAB1550 STAB1560 IF(ASGN•NE•-1•0)GO TO 200 ANUM=ANX*DX(I)+ANY*DY(I)+ANZ*DZ(I) STAB1570 ADEN=ANZ*(DX(I1)-DX(I))+ANY*(DY(I1)-DY(I))+ANZ*(DZ(I1)-DZ(I)) STAB1580 STAB1590 ALAMB=-ANUM/ADEN XL=DX(I)+ALAMB*(DX(I1)-DX(I))STAB1600 STAB1610 YL=DY(I)+ALAMB*(DY(I1)-DY(I))STAB1620 ZL=DZ(I)+ALAMB*(DZ(I1)-DZ(I))STAB1630 120 CONTINUE GDOTL=-GCOSZT*XL+GSINZT*ZL STAB1640 ANG(I)=ACOS( GDOTL /(GRAV*SQRT(XL**2+YL**2+ZL**2))) STAB1650 STAB1660 GCLDN=GSINZT*YL*ANX-(GCOSZT*ZL+GSINZT*XL)*ANY+GCOSZT*YL*ANZ STAB1670 ANG(I)=ANG(I)*SIGN(1.,GCLDN)

C C С

	200	CONTINUE	STAB1680
С			STAB1690
С		DETERMINE LANDER STABILITY	STAB1700
С			STAB1710
		KOUNT=0	STAB1720
		DO 300 I=1,NOLEG	STAB1730
		IF(ANG(I).EQ.2000.) GO TO 301	STAB1740
		IF(ANG(I).GT.0.)GO TO 320	STAB1750
		GO TO 300	STAB1760
	301	KOUNT=KOUNT+1	STAB1770
	300	CONTINUE	STAB1780
		IF(KOUNT •EQ•NOLEG)GO TO 310	STAB1790
С		PITCH INSTABILITY	STAB1800
		ISTAB=1	STAB1810
		RETURN	STAB1820
С		YAW INSTABILITY	STAB1830
	310	CONTINUE	STAB1840
		ISTAB=2	STAB1850
		RETURN	STAB1860
С		LANDER STABLE	STAB1870
	320	CONTINUE	STAB1880
		STAB=ANG(I)*57.295779513	STAB1890
		ROTX=DC(1,1)*WX+DC(1,2)*WY+DC(1,3)*WZ	STAB1900
		ROTY=DC(2,1)*WX+DC(2,2)*WY+DC(2,3)*WZ	STAB1910
		ROTZ=DC(3,1)*WX+DC(3,2)*WY+DC(3,3)*WZ	STAB1920
		STABVL=(ROTX*ANX+ROTY*ANY+ROTZ*ANZ)*57•295779513	STAB1930
		STABVL=SIGN(1.,(STAB-PRSTAB))*ABS(STABVL)	STAB1940
		IF(TIME•EQ•O•O)STABVL=O•O	STAB1950
		PRSTAB=STAB	STAB1960
		RETURN	STAB1970
		END	STAB1980

```
SUBROUTINE MATINV ( A, B, N )
                                                                                MATV
                                                                                       10
                                                                                 MATV
C
                                                                                       20
           COMPUTE THE SIMPLE MATRIX INVERSE OF A AND STORE IT IN B
                                                                                 MATV
                                                                                       30
C
C
C
                                                                                MATV
                 A IS NOT ALTERED
                                                                                       40
                                                                                MATV
                                                                                       50
                 A MUST BE NON-SINGULAR
C
C
                 A SHOULD HAVE DOMINANT DIAGONAL ELEMENTS AND BE OF
                                                                                 MATV
                                                                                       60
                      SMALL ORDER. IF THESE CONDITIONS ARE NOT MET,
                                                                                MATV
                                                                                       70
C
                                                                                       80
                      THEN A DIFFERENT INVERSE ROUTINE SHOULD BE USED.
                                                                                 MATV
           A = MATRIX OF ORDER (N.N)
B = INVERSE OF ORDER (N.N)
С
                                                                                 MATV
                                                                                       90
С
                                                                                 MATV 100
      DIMENSION A(N+1), B(N+1)
                                                                                MATV 110
С
           COPY A TO B
                                                                                 MATV 120
      DO 100 I = 1, N
DO 100 J = 1, N
                                                                                 MATV 130
                                                                                 MATV 140
                                                                                 MATV 150
  100 B(I_{,J}) = A(I_{,J})
С
                                                                                 MATV 160
С
           CONVERT B TO INVERSE BY G.E. ON BI WHERE I IS OVER B
                                                                                 MATV 170
c
c
                                                                                 MATV 180
                                                                                 MATV 190
           IP = THE COL. BEING REDUCED TO ZEROS
                                                                                 MATV 200
           400 IP = 1, N
      DO
              =
                        B(IP, IP)
                                                                                 MATV 210
      Н
           FROM HERE ON CONSIDER THE IP COL. OF B TO BE THE IP COL. OF I MATV 220
¢
                                                                                 MATV 230
       B(IP,IP) = 1.
С
           REDUCE DIAGONAL ELEMENT OF B TO 1, UPDATE I
                                                                                 MATV 240
                                                                                 MATV 250
       DO
           200 J = 1, N
  200 B(IP,J) = B(IP,J) / H
                                                                                 MATV 260
                                                                                 MATV 270
С
           REDUCE THE IP COL. OF B TO ZERO
       DO 400 J = 1, N
IF ( J •EQ• IP)GO TO 400
                                                                                 MATV 280
                                                                                 MATV 290
           UPDATE IP COL. OF I
                                                                                 MATV 300
С
                                                                                 MATV 310
       H = B(J_{*}IP)
                                                                                 MATV 320
       B(J,IP) = 0.0
DO 300 K = 1, N
                                                                                 MATV 330
                                                                                 MATV 340
       B(J_{9}K) = B(J_{9}K) - B(IP_{9}K)*H
  300 CONTINUE
                                                                                 MATV 350
                                                                                 MATV 360
  400 CONTINUE
                                                                                 MATV 370
       RETURN
                                                                                 MATV 380
       END
```

	SUBROUTINE GMPRD(A,B,R,N,M,L)	GMPD	10
	THIS SUBROUTINE MULTIPLIES TWO GENERAL MATRICES TO FORM A GENERAL	GMPD	30
	RESULTANT MATRIX	GMPD	40
	DIMENSION $A(1) \cdot B(1) \cdot B(1)$	GMPD	50 60
		GMP D	70
	IR=0	GMPD	80
	I K = – M	GMPD	90
	DO 10 K=1.	GMPD	100
	IK=IK+M	GMPD	110
	DO 10 J=1,N	GMPD	120
	IR=IR+1	GMPD	130
		GMPD	140
	IB=IK	GMPD	150
	R(IR)=0	GMPD	160
	DO 10 I=1,M	GMPD	170
	JI=JI+N	GMPD	180
	IB=IB+1	GMPD	190
10	R(IR) = R(IR) + A(JI) + B(IB)	GMPD	200
	RÉTURN	GMPD	210
	END	GMPD	220

# THIS PAGE INTENTIONALLY LEFT BLANK

#### NATIONAL AERONAUTICS AND SPACE ADMINISTRATION WASHINGTON, D.C. 20546

OFFICIAL BUSINESS

FIRST CLASS MAIL



POSTAGE AND FEES PAID NATIONAL AERONAUTICS AND SPACE ADMINISTRATION

POSTMASTER: If Undeliverable (Section 158 Postal Manual) Do Not Return

"The aeronautical and space activities of the United States shall be conducted so as to contribute ... to the expansion of human knowledge of phenomena in the atmosphere and space. The Administration shall provide for the widest practicable and appropriate dissemination of information concerning its activities and the results thereof." —National Aeronautics and Space Act of 1958

# NASA SCIENTIFIC AND TECHNICAL PUBLICATIONS

TECHNICAL REPORTS: Scientific and technical information considered important, complete, and a lasting contribution to existing knowledge.

TECHNICAL NOTES. Information less broad in scope but nevertheless of importance as a contribution to existing knowledge.

TECHNICAL MEMORANDUMS Information receiving limited distribution because of preliminary data, security classification, or other reasons.

CONTRACTOR REPORTS Scientific and technical information generated under a NASA contract or grant and considered an important contribution to existing knowledge. TECHNICAL TRANSLATIONS Information published in a foreign language considered to merit NASA distribution in English.

SPECIAL PUBLICATIONS Information derived from or of value to NASA activities. Publications include conference proceedings, monographs, data compilations, handbooks, sourcebooks, and special bibliographies.

### TECHNOLOGY UTILIZATION

PUBLICATIONS: Information on technology used by NASA that may be of particular interest in commercial and other non-aerospace applications. Publications include Tech Briefs, Technology Utilization Reports and Technology Surveys.

Details on the availability of these publications may be obtained from:

SCIENTIFIC AND TECHNICAL INFORMATION DIVISION NATIONAL AERONAUTICS AND SPACE ADMINISTRATION Woshington, D. C. 20546